Vascular Surgery Exam Prep

Audible Bleeding

2024-02-04

Table of contents

Pr	eface Usag Add Com	e ge	 12 12 12 13 							
Αι	Authors and Contributors									
I	He	ead and Neck	17							
1	Cere	ebrovascular	18							
	1.1	Available Guidelines	18							
	1.2	Presentation	18							
	1.3	Evaluation	19							
	1.4	Management	20							
		1.4.1 Medical - Asymptomatic Patients	20							
		1.4.2 Medical - Symptomatic Patients	21							
		1.4.3 Surgical - Carotid Endarterectomy	22							
		1.4.4 Endovascular - Carotid Artery Stenting	27							
		1.4.5 Summary of Prospective Trials	28							
	1.5	Uncommon Carotid Disease	30							
		1.5.1 Complicated Extracranial Occlusive Disease	30							
		1.5.2 Vertebrobasilar Disease	31							
		1.5.3 Carotid artery dissection	32							
		1.5.4 Internal Carotid Artery Aneurysms	32							
		1.5.5 Carotid Body Tumors (CBT)	32							
2	Tra	uma - Cerebrovascular	34							
	2.1	Neck Exposure Techniques	34							
		2.1.1 Carotid Artery	34							
		2.1.2 Vertebral Artery	36							
	2.2	Tenets of Cerebrovascular Injury	38							
	2.3	Blunt Trauma	38							
		2.3.1 Evaluation	38							
		2.3.2 Management	39							
		-								

2.4	Penetrating Trauma	41
	2.4.1 Management \ldots	41
2.5	Internal Jugular Vein Injuries	44

II Upper Extremity

3	Upp	er Extre	emity	46
	3.1	Overvi	ew	46
		3.1.1	Anatomy	46
		3.1.2	Etiology	49
		3.1.3	Evaluation	49
	3.2	Vaso-o	cclusive Disease	50
		3.2.1	Demographics	50
		3.2.2	Management	51
	3.3	Vasosp	astic Disorders	54
		3.3.1	Iatrogenic or Vasopressor Induced Ischemia	54
		3.3.2	Raynaud's Phenomenon	54
		3.3.3	Ergotism	55
		3.3.4	Buerger's Disease	56
		3.3.5	Large Artery Vasculitis	57
		3.3.6	Other Vasculitis syndromes	60
	3.4	Aneury	ysmal Disease	60
		3.4.1	Subclavian Artery Aneurysms	60
		3.4.2	Axillary Artery Aneurysms	61
		3.4.3	Brachial Artery Aneurysms	62
		3.4.4	Radial Artery Pseudoaneurysm	63
	3.5	Occupa	ational Vascular Disease	63
		3.5.1	Hand-Arm Vibration Syndrome	63
		3.5.2	Hypothenar hammer syndrome	64
		3.5.3	Environmental Exposures	65
	3.6	Sports	Medicine	66
		3.6.1	Quadrilateral space syndrome	66
		3.6.2	Humeral head compression of axillary artery	67
		3.6.3	Thoracic Outlet Syndrome	67
	3.7	Vascula	ar Trauma-Upper Extremity	67
		3.7.1	Subclavian Artery Trauma	68
		3.7.2	Axillary Artery Trauma	68
		3.7.3	Brachial Artery Trauma	69
		3.7.4	Radial/ulnar artery trauma	69
	3.8	Upper	Extremity Lymphedema	70
		3.8.1	Etiology	70
		3.8.2	Management	70

4	Tho	oracic Outlet	71
	4.1	Overview	71
		4.1.1 Demographics	71
		4.1.2 Evaluation \ldots	74
		4.1.3 Management	76
	4.2	Venous (vTOS) \ldots	77
		4.2.1 Demographics	77
		4.2.2 Evaluation \ldots	78
		4.2.3 Management	79
		4.2.4 Summary	80
	4.3	Arterial (aTOS)	81
		4.3.1 Presentation	81
		4.3.2 Evaluation \ldots	82
		4.3.3 Management	82
	4.4	Neurogenic (nTOS)	83
		4.4.1 Demographics	83
		4.4.2 Management	84
5	Hen	nodialysis	85
	5.1	Planning access creation	85
	5.2	Techniques for access creation	90
		5.2.1 Anesthetic Considerations	91
		5.2.2 Endovascular AV Fistula	92
	5.3	Maintenance and complications	92
		5.3.1 Failure to mature	92
		5.3.2 Access Failure	93
		5.3.3 Steal Syndrome	95
		5.3.4 Neuropathy and Neuropraxia	97
		5.3.5 Perioperative neuropraxia	97
		5.3.6 Bleeding and Aneurysm	98
	5.4	Dialysis Catheters	98
	5.5	Peritoneal Dialysis	99

III Thorax

6	Aort	tic Dissection 1	.01
	6.1	Demographics	.01
		6.1.1 Classification and Terminology	.02
		6.1.2 Presentation $\ldots \ldots \ldots$.06
	6.2	Evaluation	.06
	6.3	Management	.06

7	TAA	AA	109
	7.1	Demographics	109
		7.1.1 Anatomy	109
		7.1.2 Etiology	111
	7.2	Management	113
		7.2.1 Ascending aorta and arch	113
		7.2.2 Isolated Thoracic Aneurysms	114
		7.2.3 Thoracoabdominal Aneurysms	115
8	Aor	topathies	128
-	8.1	General Considerations	128
	8.2	Demographics	129
	8.3	Evaluation	130
	8.4	Management	131
		8.4.1 Medical Management	131
		8.4.2 Surgery	131
	8.5	Summary	133
	8.6	Additional Resources	133
9	Trai	uma - Endo	134
•	9.1	Blunt Thoracic Aortic Injury (BTAI)	134
	0.1	911 Demographics	134
		9.1.2 Evaluation	137
		9.1.3 Management	138
	9.2	Blunt Abdominal Aortic Injuries	141
	-	9.2.1 Demographics	141
		9.2.2 Evaluation	141
		9.2.3 Management	142
	9.3	Blunt Axillo-Subclavian Injuries	142
		9.3.1 Demographics	142
		9.3.2 Management	143
IV	/ Δ h	adomen	145
	- - U		- J

10	AAA	A contract of the second se	146
	10.1	Demographics	146
		10.1.1 Etiology	147
	10.2	Evaluation	150
		10.2.1 Imaging	150
	10.3	Management	151
		10.3.1 EVAR	156
		10.3.2 Open Repair \ldots	159

		10.3.3 Ruptured Aneurysms	65
	10.4	Iliac Artery Aneurysms	68
		10.4.1 Demographics	68
		10.4.2 Presentation $\ldots \ldots \ldots$	68
		10.4.3 Management	68
			-0
11	Nes		70
	11.1	Acute Mesenteric Ischemia	70 70
		11.1.1 Presentation and Evaluation	70 71
		11.1.2 Management	(1 76
	11.0	11.1.3 Other Etiologies	() 77
	11.2	Chronic Mesenteric Ischemia	(`(
		11.2.1 Presentation and Evaluation	70
	11.0	11.2.2 Management	78 70
	11.3	Visceral Aneurysms	(9 00
		11.3.1 SMA Aneurysm	50
	11 4	11.3.2 Splenic Aneurysm	50
	11.4	Median Arcuate Ligament Syndrome (MALS)	30
	11.5	SMA Syndrome	31
	11.6	Uterine Fibroids	31
12	Renz	al 12	82
	12.1	Renovascular Ischemic Disease	82
	12.1	12.1.1 Pathophysiology	82
		12.1.2 Evaluation	85
		12.1.3 Management	87
	12.2	Fibromuscular Dysplasia (FMD)	94
	12.2	12.2.1 Demographics	94
		12.2.2 Evaluation	97
		12.2.3 Management 19	99
	12.3	Renal Artery Aneurysms	03
		12.3.1 Demographics	03
		12.3.2 Anatomy	04
		12.3.3 Presentation	04
		12.3.4 Management 20	04
	12.4	Renal Artery Dissection 20	05
	12.1	12.4.1 Demographics	05
		12.4.2 Evaluation 20	05
		12.4.3 Management 20	05
	12.5	Nutcracker Syndrome 20	06
		12.5.1 Demographics	06
		12.5.2 Evaluation	06
		12.5.3 Management	06
			-

	12.6	Renal	Vein Thrombosis $\ldots \ldots 20'$	7
		12.6.1	Evaluation $\ldots \ldots 20'$	7
		12.6.2	$Management \dots \dots$	7
	12.7	Renal .	rteriovenous Fistula	7
		12.7.1	Presentation $\ldots \ldots 20'$	7
		12.7.2	Evaluation $\ldots \ldots 20'$	7
		12.7.3	$Management \dots \dots$	8
12	т		torial 20	^
13	12 1	Conoro	Principles 20	9 0
	12.1	Bogie I	$\frac{1}{2} 111111111111111111111111111111111$	9 0
	13.2	Dasic I	Supersolies Control	1
		10.2.1	Vigeonal Detetion 21	า ก
		10.2.2	VISCEIAL ROLATION	2 1
	199	13.2.3 Manag	Exposure of Infrarenal Aorta	4 1
	19.9	Manag	$\begin{array}{c} \text{alient} & \dots & $	4
		13.3.1	$\begin{array}{c} \text{Cenac Artery} & \dots & $	4 r
		13.3.2	$ \begin{array}{c} \text{Ommon Hepatic Artery (CHA)} \\ \text{Controllerational Actions (CDA)} \\ \end{array} $	о с
		13.3.3	$Gastroduodenal Artery (GDA) \dots \dots$	0
		13.3.4	Spienic Artery \dots	0
		13.3.5	Superior Mesenteric Artery (SMA) $\ldots \ldots \ldots$	0
		13.3.0	Devel Arteries	ð o
		13.3.7		ð
14	Trau	ma - V	enous 22	0
	14.1	Genera	Principles	0
		14.1.1		
			Anatomy	0
		14.1.2	Anatomy	0 0
		$14.1.2 \\ 14.1.3$	Anatomy 220 Non-operative management 220 Hemorrhage Control 220	0 0 1
		$14.1.2 \\ 14.1.3 \\ 14.1.4$	Anatomy	0 0 1 1
		$14.1.2 \\ 14.1.3 \\ 14.1.4 \\ 14.1.5$	Anatomy	$0\\0\\1\\2$
	14.2	14.1.2 14.1.3 14.1.4 14.1.5 Infrare	Anatomy	$0\\0\\1\\2\\2$
	14.2	14.1.2 14.1.3 14.1.4 14.1.5 Infrare 14.2.1	Anatomy	$ \begin{array}{c} 0 \\ 0 \\ 1 \\ 1 \\ 2 \\ 2 \\ 2 \end{array} $
	14.2	14.1.2 14.1.3 14.1.4 14.1.5 Infrare 14.2.1 14.2.2	Anatomy	$ \begin{array}{c} 0 \\ 0 \\ 1 \\ 2 \\ 2 \\ 2 \end{array} $
	14.2	14.1.2 14.1.3 14.1.4 14.1.5 Infrare 14.2.1 14.2.2 14.2.3	Anatomy	$egin{array}{c} 0 \\ 0 \\ 1 \\ 1 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \end{array}$
	14.2	$\begin{array}{c} 14.1.2\\ 14.1.3\\ 14.1.4\\ 14.1.5\\ \text{Infrare}\\ 14.2.1\\ 14.2.2\\ 14.2.3\\ 14.2.4\end{array}$	Anatomy	$egin{array}{c} 0 \\ 0 \\ 1 \\ 1 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \end{array}$
	14.2	$\begin{array}{c} 14.1.2 \\ 14.1.3 \\ 14.1.4 \\ 14.1.5 \\ \text{Infrare} \\ 14.2.1 \\ 14.2.2 \\ 14.2.3 \\ 14.2.4 \\ 14.2.5 \end{array}$	Anatomy 221 Non-operative management 221 Hemorrhage Control 222 Primary repair 222 Ligation 222 Nal IVC 222 Anatomy 222 Non-operative Management 222 Non-operative Management 222 Exposure 222 Hemorrhage Control 222 Primary Repair 222	$egin{array}{c} 0 \\ 0 \\ 1 \\ 1 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2$
	14.2	$\begin{array}{c} 14.1.2 \\ 14.1.3 \\ 14.1.4 \\ 14.1.5 \\ \text{Infrare} \\ 14.2.1 \\ 14.2.2 \\ 14.2.3 \\ 14.2.4 \\ 14.2.5 \\ 14.2.6 \end{array}$	Anatomy	$egin{array}{cccc} 0 \\ 0 \\ 1 \\ 1 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 3 \end{array}$
	14.2 14.3	14.1.2 14.1.3 14.1.4 14.1.5 Infrare 14.2.1 14.2.2 14.2.3 14.2.4 14.2.5 14.2.6 Renal/	Anatomy221Non-operative management221Hemorrhage Control222Primary repair222Ligation222al IVC222Anatomy222Non-operative Management222Exposure222Hemorrhage Control222Primary Repair222Ligation222Suprarenal IVC223	$egin{array}{cccc} 0 \\ 0 \\ 1 \\ 1 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 3 \\ 3 \end{array}$
	14.2 14.3	$\begin{array}{c} 14.1.2\\ 14.1.3\\ 14.1.4\\ 14.1.5\\ \text{Infrare}\\ 14.2.1\\ 14.2.2\\ 14.2.3\\ 14.2.4\\ 14.2.5\\ 14.2.6\\ \text{Renal}/\\ 14.3.1\\ \end{array}$	Anatomy221Non-operative management221Hemorrhage Control22Primary repair22Ligation22Nal IVC22Anatomy22Non-operative Management22Exposure22Hemorrhage Control22Primary Repair22Ligation22Anatomy22Anatomy22Anatomy22Anatomy22Anatomy22Anatomy22Anatomy22Anatomy22Anatomy22Anatomy22Anatomy22Anatomy22Anatomy22Anatomy22	$egin{array}{cccc} 0 \\ 0 \\ 1 \\ 1 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 3 \\ 3 \\ 3 \end{array}$
	14.2 14.3	$\begin{array}{c} 14.1.2\\ 14.1.3\\ 14.1.4\\ 14.1.5\\ \text{Infrare}\\ 14.2.1\\ 14.2.2\\ 14.2.3\\ 14.2.4\\ 14.2.5\\ 14.2.6\\ \text{Renal}/\\ 14.3.1\\ 14.3.2\\ \end{array}$	Anatomy221Non-operative management221Hemorrhage Control222Primary repair222Ligation222Lal IVC222Anatomy222Non-operative Management222Hemorrhage Control222Hemorrhage Control222Ligation222Suprarenal IVC222Anatomy223Non-operative Management224Hemorrhage Control225Hemorrhage Control225Ligation225Suprarenal IVC225Non-operative Management225Non-operative Management225Suprarenal IVC225Non-operative Management225Non-operative Management225 <td>$egin{array}{cccc} 0 \ 0 \ 1 \ 1 \ 2 \ 2 \ 2 \ 2 \ 3 \ 3 \ 3 \ 3 \ 3 \ 3$</td>	$egin{array}{cccc} 0 \ 0 \ 1 \ 1 \ 2 \ 2 \ 2 \ 2 \ 3 \ 3 \ 3 \ 3 \ 3 \ 3$
	14.2 14.3	$\begin{array}{c} 14.1.2\\ 14.1.3\\ 14.1.4\\ 14.1.5\\ \text{Infrare}\\ 14.2.1\\ 14.2.2\\ 14.2.3\\ 14.2.4\\ 14.2.5\\ 14.2.6\\ \text{Renal}/\\ 14.3.1\\ 14.3.2\\ 14.3.3\\ \end{array}$	Anatomy	$egin{array}{ccccc} 0 \ 0 \ 1 \ 1 \ 2 \ 2 \ 2 \ 2 \ 2 \ 3 \ 3 \ 3 \ 3 \ 3$
	14.2 14.3	$\begin{array}{c} 14.1.2\\ 14.1.3\\ 14.1.4\\ 14.1.5\\ \text{Infrare}\\ 14.2.1\\ 14.2.2\\ 14.2.3\\ 14.2.4\\ 14.2.5\\ 14.2.6\\ \text{Renal}/\\ 14.3.1\\ 14.3.2\\ 14.3.3\\ 14.3.4\end{array}$	Anatomy	$egin{array}{cccccccccccccccccccccccccccccccccccc$
	14.2	$\begin{array}{c} 14.1.2\\ 14.1.3\\ 14.1.4\\ 14.1.5\\ \text{Infrare}\\ 14.2.1\\ 14.2.2\\ 14.2.3\\ 14.2.4\\ 14.2.5\\ 14.2.6\\ \text{Renal}/\\ 14.3.1\\ 14.3.2\\ 14.3.3\\ 14.3.4\\ 14.3.5\end{array}$	Anatomy	$egin{array}{cccccccccccccccccccccccccccccccccccc$

14.4	Retroh	epatic IVC
	14.4.1	Anatomy
	14.4.2	Management
	14.4.3	Exposure
	14.4.4	Hemorrhage Control
	14.4.5	Primary Repair
	14.4.6	Ligation
14.5	Suprah	epatic IVC \ldots \ldots \ldots \ldots 226
	14.5.1	Anatomy
	14.5.2	Non-operative Management
	14.5.3	Exposure
	14.5.4	Hemorrhage Control
	14.5.5	Primary Repair
	14.5.6	Ligation
14.6	Portal	Vein and Superior Mesenteric Vein Injuries
	14.6.1	Exposure
	14.6.2	Management

V Lower Extremity

Med	lical Management and Claudication	229
15.1	Pathogenesis of peripheral arterial disease	229
	15.1.1 Risk Factors	230
	15.1.2 Landmark Studies	233
15.2	Intermittent Claudication	234
	15.2.1 Evaluation \ldots	235
	15.2.2 Management	237
CLT	1	243
16.1	Guidelines	243
16.2	Demographics	244
	16.2.1 Etiology and Presentation	244
	16.2.2 Natural History	245
	16.2.3 WIFi Classification	246
	16.2.4 Differential Diagnosis	247
16.3	Evaluation	247
	16.3.1 Physiologic Testing	248
	16.3.2 Imaging	249
16.4	Management	250
	16.4.1 Medical Managment	250
	16.4.2 Endovascular and Surgical Management	251
	16.4.3 Complications	255
	Med 15.1 15.2 CLT 16.1 16.2 16.3 16.4	Medical Management and Claudication 15.1 Pathogenesis of peripheral arterial disease 15.1.1 Risk Factors 15.1.2 Landmark Studies 15.2 Intermittent Claudication 15.2.1 Evaluation 15.2.2 Management 15.2.2 Management 15.2.1 Evaluation 15.2.2 Management 16.1 Guidelines 16.2 Demographics 16.2 Demographics 16.2.1 Etiology and Presentation 16.2.2 Natural History 16.2.3 WIFi Classification 16.2.4 Differential Diagnosis 16.3 Evaluation 16.3.1 Physiologic Testing 16.3.2 Imaging 16.4.1 Medical Managment 16.4.2 Endovascular and Surgical Management 16.4.3 Complications

17	Acut	e Limb Ischemia	256
	17.1	Overview	256
		17.1.1 Demographics	256
		17.1.2 Evaluation	259
		17.1.3 Management	259
	17.2	Compartment Syndrome	262
		17.2.1 Etiology	262
		17.2.2 Evaluation	263
		17.2.3 Management	264
	17.3	Blue Toe Syndrome	265
		17.3.1 Etiology and Presentation	265
		17.3.2 Evaluation	265
		17.3.3 Management	266
	17.4	Non-atheromatous Popliteal Artery Disease	266
		17.4.1 Popliteal Artery Entrapment Syndrome (PAES)	266
		17.4.2 Cystic Adventitial Disease	267
	17.5	Amputations	267
		17.5.1 Above-knee, trans-femoral amputation (AKA)	267
		17.5.2 Below-knee, trans-tibial amputation (BKA)	268
		17.5.3 Transmetatarsal amputation (TMA)	268
10	Von	Nus Disease	260
10	18 1	Chronic Vonous Insufficiency	209
	10.1	18.1.1 Terminology and Presentation	209
		18.1.2 Evaluation	209
		18.1.2 Evaluation	$271 \\ 971$
		18.1.4 Other Complications of Superficial Vancus Disease	271
	189	Acute Deep Vanous Thromhosis (DVT)	210
	10.2	18.2.1 Demographics	213
		18.2.2 Evaluation	280
		18.2.2 Evaluation	283
	18.3	Phleomasia	289
	18.4	Deep Venous Obstruction	290
	10.1	18.4.1 Etiology	290
		18.4.2 Evaluation	291
		1843 Management	293
	18.5	Pelvic Congestive Syndrome	295
	10.0	1851 Presentation	295
		18.5.2 Evaluation and Management	296
	18.6	Superior Vena Cava (SVC) Syndrome	296
	10.0	18.6.1 Management	297
	18.7	Lymphedema	297
	2	18.7.1 Demographics	297

		$18.7.2 \text{Evaluation} \dots \dots \dots \dots \dots \dots \dots \dots \dots $		298
		18.7.3 Management		298
19 ٦	Frau	na - Peripheral		299
1	9.1	Demographics		300
		19.1.1 Epidemiology		300
		19.1.2 Types of Injuries		300
		19.1.3 Presentation		300
1	9.2	Evaluation		301
1	9.3	Management		302
		19.3.1 General Considerations		303
		19.3.2 Iliac Vessels		306
		19.3.3 Femoral Vessels		307
		19.3.4 Popliteal Artery		308
		19.3.5 Tibial Vessels		309
		19.3.6 Subclavian and Proximal Axillary Arteries		310
		19.3.7 Distal Axillary and Brachial Arteries		311
		19.3.8 Radial and Unar Artery		311
1	9.4	Venous Iniury		311
-		19.4.1 Lower Extremity Venous Injury		312
		19.4.2 Upper Extremity Venous Injury		312
		opportantion of the infation o	•	<u> </u>

VI General

20	Vasc	cular Lab	314
	20.1	Overview	314
		20.1.1 Waveforms	315
	20.2	Extracranial	319
	20.3	Intracranial	324
	20.4	Upper Extremity	326
		20.4.1 Hemodialysis Access	328
	20.5	Lower Extremity	329
		20.5.1 Pathologies	332
	20.6	Abdominal	335
		20.6.1 Aorta	335
		20.6.2 Mesenteric Vasculature	337
		20.6.3 Portal Vein	338
		20.6.4 Renal arteries	338
	20.7	Venous	339
		20.7.1 Reflux	342
		20.7.2 May Thurner Syndrome	343
		20.7.3 Nutcracker Syndrome	343

		20.7.4 Thrombosis	343
		20.7.5 Vein Mapping	345
	20.8	Artifacts	346
	20.9	Accreditation and Credentialing	350
21	Radi	iation Safety	352
	21.1	Important Terms	352
	21.2	Radiation effects	353
	21.3	Risk factors	353
	21.4	Reduction strategies	354
		21.4.1 Pregnancy considerations	355
		21.4.2 Regulation	356
	21.5	Contrast Reduction	357
		21.5.1 CO2 Angiography	357
22	Fnd	ovascular Access	350
	22.1	Complications	364
	22.1		501
23	Rapi	id Review	366
	23.1	Venous Disease	366
	23.2	Vascular Lab	367
		23.2.1 TCD	367
		23.2.2 Basics	368
		23.2.3 Sizes and Velocities	368
	23.3	Zebras	370
	20.0	200100	510
Re	feren	ices	373

References

Preface

This content was developed to accompany our Audible Bleeding Exam Prep podcast series. Each chapter covers a key domain of vascular surgery, essential for board review preparation, and is associated with an Audible Bleeding episode which you can access through the link embedded into the text.

Editors: Adam P Johnson; Matt Smith; and Audible Bleeding Team

Version 01.26

Usage

This e-book is intended to be a high level, easily accessible review for exam preparation and paired with our podcast content to help streamline your studying.

Please consider this a "living document." We plan to post regular updates and respond to your feedback. We are utilizing an open source annotations software called hypothes. is to allow you to take note along side our e-book. Feel free to post your notes publicly if you think other readers would benefit from your additions, or create a small private group for your local trainees of faculty members. We will work to incorporate relevant notes and additions into future editions.

We are excited to provide our readers and listeners a free and open opportunity to contribute to this work. However, we do expect that you follow a Contributor Code of Conduct. Public annotations and feedback are considered contributions and by contributing to this ebook, you agree to abide by its terms.

This ebook is (and always will be) free, and is licensed under a Creative Commons Attribution-NonCommercial-NoDerivs 3.0 United States License.

Additional Resources

We hope you use this e-book as a high level review for your exam preparation. This resource is by no means comprehensive. We have done our best to reference and link to more comprehensive resources throughout the text so that you can dive deeper into topics that are beyond the scope of this review. We try to prioritize open access publications and free, high quality, unbiased materials. A few highly recommended additional resources include:

- Rutherford's Vascular Surgery and Endovascular Therapy
 - No vascular surgeon bookshelf is complete without a copy of this book sitting on it!
 - Of note, our review book references the 9th edition of Rutherford's, however there is a new edition (10th) has just been released in June 2022.
 - Many institutional libraries carry this book and can be accessed for free. We would recommend checking with that resource prior to purchasing yourself.
 - Some industry representatives offer text book purchasing programs, so check with them prior to purchasing yourself.
- VESAP6
 - This is the premier question bank for preparing for the vascular surgery boards.
 - Some industry representatives offer text book purchasing programs and include this as an option, so check with them prior to purchasing yourself.
 - A new version of VESAP is released every 2-3 years, and once a new one is release, the older versions are no longer accessible. Make sure to check the release schedule so that you can time your purchase to align when you will need to do the majority of your exam preparation.

• SVS Education Website

- Check this site regularly for free content through the SVS. This is available to all candidate members of the SVS, so sign up!
- VSCORE
 - This is a curated curriculum developed in collaboration with Association for Program Directors in Vascular Surgery (APDVS)
 - Check with your program director to confirm whether this is available at your training program.

Comments, Questions or Contributions

Please visit our github page or send us an email.

This book is built on Quarto.

Authors and Contributors

This book is the result of a broad community of vascular surgeons and trainees. They have all freely donated their time to contribute to this work and we owe them a debt of gratitude. Follow them on twitter or other social media platforms to keep up to date with what they have going on.

Danielle Bajakian, MD; Assistant Professor of Vascular Surgery; Director, Critical Limb Ischemia Program; Columbia University, New-York Presbyterian

Mohamed Barkat, MSc, FRCS (Vasc); Consultant Vascular Surgeon; Manchester University NHS Foundation Trust; @mo_barkat

Jason Bingham, MD; General and Bariatric Surgeon, Director of Research, Associate Program Director of General Surgery Residency Program; *Madigan Army Medical Center; @Bing-hamMD*

Einar Brevik, MD; Consultant Vascular Surgeon; Haukeland University Hospital

Sarah J. Carlson, MD, MSc, FACS; Professor of Vascular Surgery, Associate Chief for Surgery Operations; *Boston University, Boston VA Medical Center*; @sarahcarlsonmd

Marlin "Wayne" Causey, MD; Chief of Vascular Surgery; *Brooke Army Medical Center;* Co-Host of WarDocs: The Military Medicine Podcast.

Matthew Chia, MD, MS; Vascular Surgery Resident; Northwestern; @chia_md

Anahita Dua, MD, MS, MBA, FACS; Assistant Professor of Vascular Surgery; Director of Vascular Laboratory; Co-director of the Peripheral Arterial Disease Center; *Harvard University, Massachusetts General Hospital*; @AnahitaDua

Julie Duke, MD, RPVI; Vascular Surgeon; Vascular Surgical Associates of Atlanta, GA

Matthew Edwards, MD; Richard H. Dean Professor and Chairman of Vascular and Endovascular Surgery; *Atrium Health Wake Forest Baptist*; @msedwards0427

Sharif Ellozy, MD; Associate Professor of Vascular Surgery, Vascular Surgery Fellowship Program Director; *Weill Cornell Medical College, New-York Presbyterian*; @SharifEllozy

Adham Elmously, MD; Cardiothoracic Surgery Fellow; *Houston Methodist;* @elmous-lyMD

Leanna Erete, MBBS; Vascular Surgery Trainee; London Deanery; @leanna_erete

Alex Forsyth, MD; Vascular Surgery Resident; University of Massachusetts; @alex_forsyth_

Rachael Forsythe, MBChB, PhD; Consultant Vascular Surgeon; *Royal Infirmary Edinburgh*; @ROForsythe

Nicholas Greaves MBChB, PhD, FRCS (Vasc); Consultant Vascular Surgeon; Manchester University NHS Foundation Trust

Manj Gohel, MBChB, MD, FRCS, FEBVS; Consultant Vascular Surgeon; *Nuffield Health*; @ManjGohel

Eilidh Gunn, MBChB, MRCSEd; Clinical Research Fellow, Vascular Surgery; *Royal Infirmary Edinburgh*; @eil1_g

Michael Jenkins, BSc, MBBS, MS, FRCS, FEBVS; Consultant Vascular Surgeon; St Mary's Hospital, Imperial College Healthcare NHS Trust and Chelsea and Westminster Hospitals; @mpj21165

Adam P. Johnson, MD, MPH; Assistant Professor of Vascular Surgery; *Duke University* School of Medicine; @adam_mdmph

Nedal Katib, MB, BCh, BAO, MS, FRACS; Consultant Vascular Surgeon; Prince of Wales Hospital, Gosford and Wyong Hospitals

Wen Kawaji, MD, ScM; General Surgery Resident; MedStar Health @WenKawaji

MAJ Kevin Kniery, MD, MPH; Vascular Surgeon; *Brooke Army Medical Center;* @Kniery_Bird

Melissa Kirkwood, MD; Chief of the Division of Vascular and Endovascular Surgery; UT Southwestern Medical Center;

Young Lee, MD; Vascular Surgery Fellow; Emory University; @youngclee86

Joseph Lombardi, MD, MBA; Professor and Chief of Vascular Surgery; *Cooper University Hospital*; @VascSurgMD

Rafael Demarchi Malgor, MD, MBA, FACS; Associate Professor of Surgery; Program Director, Robert B. Rutherford fellowship and residency in Vascular Surgery; Division of Vascular Surgery and Endovascular Therapy; *The University of Colorado, Anschutz Medical Center;* @RafaMalgor

M. Asharf Mansour, MD; Academic Chair, Vascular Surgeon; *Michigan State University*, Spectrum Health

Nicolas J Mouawad, MD, MPH, MBA, FSVS, FRCS, FACS, RPVI; Chief and Medical Director of Vascular and Endovascular Surgery; *McLaren Health System - Bay Region*, *Michigan*, USA; Associate Clinical Professor of Surgery, *Michigan State University*, *Central Michigan University*; @NickMouawadMD Cullen McCarthy, MD; Vascular Surgeon; Cardio-Thoracic & Vascular Surgical Associates, Mobile Infirmary

Mia Miller, MD; Fellow; University of Minnesota

Andrew Nickinson, MBBS, MD, MRCS(Eng); Vascular Surgery Trainee; Wessex/Thames Valley Deanery @AndrewNickinson

Anna Ohlsson, MD; Assistant Professor of Surgery, Associate Program Director for Vascular Surgery training programs; University of Pennsylvania, Corporal Michael J. Crescenz/Philadelphia VA Medical Center

Alaska Pendleton, MD, MPH; Vascular Surgery Resident; Massachusetts General Hospital

Todd Rassmussen, MD; Professor of Surgery, Vice-Chair of education; *Mayo Clinic;* @trasmussen_md

Maggie Reilly, MD; Vascular Surgery Resident; Northwestern

Nicole Rich, MD, MPH; Vascular Surgeon; Ascension Medical Group Saint Thomas Rutherford Surgery

Drena Root; Vascular Center Technical Director; Massachusetts General Hospital

Nakia Sarad, DO, MS; General Surgery Resident; New York Presbyterian/Queens - Weill Cornell; @NakSaradDO

Sherene Shalhub, MD; Associate Professor of Surgery, Division of Vascular Surgery; University of Washington; @shereneshalhub

Kush J Sharma, MD, RPVI; Vascular Surgeon; Desert Vein and Vascular Institute

Sammy Siada, DO, RPVI; Assistant Clinical Professor of Vascular and Endovascular Surgery; UCSF Fresno; @SammySiada

Aminder Singh, MBBS, MRes, MRCS; Vascular Surgery Trainee, PhD Candidate; *East of England, University of Cambridge*; @AminderASingh

Matthew R. Smeds, MD, FACS, DFSVS; Professor of Surgery, Division of Vascular and Endovascular Surgery; *St Louis University*; @mattsmeds

Matthew Smith, MD, PhD; Assistant Professor of Surgery, Division of Vascular Surgery; University of Washington

Matt Spreadbury, MD; Vascular Surgery Resident; Haukeland University Hospital

Part I

Head and Neck

1 Cerebrovascular

Authors: Adam P Johnson, Nicole Rich, and Kevin Kniery Contributors: Eilidh Gunn, Ronnachai Buppanharun, Henri Pickard

1.1 Available Guidelines

Society for Vascular Surgery clinical practice guidelines for management of extracranial cerebrovascular disease (AbuRahma et al. 2022)

1.2 Presentation

1. What is the definition of crescendo TIAs?

The presenting symptoms of extracranial carotid disease is of the utmost importance to understand because it is a major factor in determining treatment and prognosis. Focal neurologic symptoms that occur suddenly and then resolve within 24 hours are called a transient ischemic attack (TIA). If the deficit persists beyond 24 hours, then it is considered a stroke. Strokes are then graded according to the National Institute of Health Stroke Scale (NIHSS).(Fischer et al. 2010)

A particularly confusing distinction is between a crescendo TIA and a stroke in evolution. A crescendo TIA is frequent repetitive neurological attacks with complete resolution of the deficit between the episodes, usually over a 24 hour period. The result of which is the same neurological deficit.(R. Naylor et al. 2023) If the patient exhibits progressive deterioration in neurological function between TIA episodes this is classified as a stroke in evolution not a crescendo TIA.(Ricotta 2019) These are both particularly concerning findings and may be an indication for an emergent intervention.

2. Who needs to be screened for carotid disease?

Not everyone should be screened for carotid disease, however, there is merit in screening high risk patients as only 15% of patients experience a warning TIA prior to a stroke.(Rockman 2019) The purpose of carotid bifurcation imaging is to detect "strokeprone" carotid bifurcation plaque and identify a high-risk patient likely to benefit from therapy designed to reduce stroke risk. Traditionally, the clinical sign associated with asymptomatic carotid stenosis is the presence of a neck bruit on auscultation. Focal ipsilateral carotid bruits in symptomatic patients has a sensitivity of 63% and a specificity of 61% for high-grade carotid stenosis (range, 70%-99%).(Sauve 1994) As these figures suggest the absence of carotid bruit does not absolutely exclude the presence of a significant carotid bifurcation lesion, therefore additional criteria are used to determine which patients would benefit from screening:

- Evidence of clinically significant peripheral vascular disease regardless of age
- Patients aged >65 years with a history of one or more of the following atherosclerotic risk factors:
 - CAD
 - Smoking
 - Hypercholesterolemia

In general, the more risk factors present, the higher the yield of screening for the patient. Presence of a carotid bruit, AAA or family history of disease alone is not sufficient to warrant imaging.(Aburahma and Perler 2022)

1.3 Evaluation

1. Ultrasound Imaging

- US is considered the best, least invasive, and cost effective method for detecting carotid stenosis on screening.(Shaalan et al. 2008) For more see Section 20.2
- 70-99% stenosis of ICA Duplex diagnostic criteria:
sec-extracranial
 - PSV > = 230 cm/sec
 - EDV >100 (EDV > 140 cm/sec most sensitive for stenosis >80%)
 - Internal/Common Carotid PSV Ratio > 4
- + 50-69% stenosis of ICA Duplex diagnostic criteria:
 - PSV 125-229 cm/sec
 - EDV 40-100
 - Internal/Common Carotid PSV Ratio 2-4
- US has a low sensitivity for 50-69% stenosis. A negative ultrasound in symptomatic patients necessitates additional imaging.
- Velocity-based estimation of carotid artery stenosis may need to be adjusted in certain circumstances:

- Higher velocities often seen in women than in men.
- Higher velocities often seen in the presence of contralateral carotid artery occlusion.
- High carotid bifurcation, severe arterial tortuosity, extensive vascular calcification, and obesity may also reduce the accuracy of US imaging.

2. Other Imaging Modalities

- CTA
 - Pro fast, sub-millimeter spatial resolution, visualize surrounding structures.
 - Con cost, contrast exposure.
- MRA
 - Pro no contrast administered; can analyze plaque morphology.
 - Con Does not visualize calcium in plaque; overestimates the degree of stenosis (False positive for 50-69% to be read as >70%).
- Catheter-based digital subtraction imaging (DSA)
 - Still considered by many the gold-standard imaging modality.
 - Reserved for individuals with conflicting less-invasive imaging or those considered for CAS.
 - Con cost, contrast, and risk of stroke.

1.4 Management

1.4.1 Medical - Asymptomatic Patients

First line management for asymptomatic patients with carotid stenosis is medical optimization of risk factors. The benefit of prophylactic intervention (CEA/CAS) of even high grade stenosis is estimated at a 1-2% stroke reduction risk per year. (A. Ross Naylor 2015) In addition, keep in mind that intervention (CEA/CAS) has only demonstrated a benefit in asymptomatic patient with life expectancy greater than 3 years. (Bulbulia and Halliday 2017; Halliday et al. 2010; Rosenfield et al. 2016)

Take a Listen

Check out our conversation with Prof Ross Naylor where he discusses in depth his decision making for asymptomatic carotid disease and other controversies in carotid disease

management.

Hypertension

- Blood pressure reduction, with a target of <140/90 mmHg through lifestyle interventions and anti-hypertensive treatment is recommended in individuals who have hypertension with asymptomatic carotid atherosclerosis or those with TIA or stroke after the hyperacute period.
- Each 10-mm Hg reduction in blood pressure among hypertensive patients decreases the risk of stroke by 33% in patients aged 60-79.(Aiyagari and Gorelick 2009)

Diabetes

• In diabetic patients glycaemic control to nearly normoglycemic levels (target hemoglobin A1C <7%) is recommended to reduce microvascular complications and, with a smaller evidence base, macrovascular complications other than stroke.(Ricotta 2019)

Hyperlipidemia

- Statin agents are recommended targeting LDL of 100 mg/dL, for those with coronary heart disease or symptomatic atherosclerotic disease, and LDL of 70 mg/dL for very high-risk persons with multiple risk factors.
- High dose statin therapy in patients with TIA/stroke reduce future rates of stroke or cardiovascular events but not overall mortality at 5 years. (Karam, Loney-Hutchinson, and McFarlane 2008)

Smoking Cessation - Encouraging smoking cessation is paramount for these patients. Smoking and carotid disease have been found to have a dose dependent relationship based on pack-years. Risk of stroke has been shown to reduce to baseline within 2-4 years of smoking cessation.(Hicks et al. 2015; Kawachi et al. 1993; Petrik et al. 1995; P. W. Wilson et al. 1997) For more on smoking cessation, see Section 15.2.2.1

Antiplatelet therapy - Single agent antiplatelet therapy, most often aspirin, is often all that is needed, but this should be weighted with other cardiovascular risk factors. A number of trials have showed limited benefit of adding an additional antiplatelet or routinely testing for clopidogrel resistance. (Ricotta 2019)

1.4.2 Medical - Symptomatic Patients

Patients who present with a TIA should be placed on dual antiplatelet therapy, often aspirin and clopidogrel, which are continued through the workup and treatment of carotid disease. Patients often also have their blood pressure medications reduced to allow permissive hypertension and maximize perfusion to at risk cerebral tissue. While we are not often directly involved in the management of acute stroke, vascular surgeons should be aware of best practices in this area. Management of acute stroke currently follow these timelines (Powers et al. 2018)

- < 4.5 hours from onset of symptoms medical throm bolysis with tPA unless contraindicated.
 - Patient age >80 and diabetes are contraindications to tPA administration after 3hrs.
 - Other contraindications include hypertension, intracranial hemorrhage, recent stroke or cranial trauma, spine/brain surgery within 3 months, GI bleed within 21 days.
 - BP management post stroke with thrombolysis includes treatment of BP to <185/110 prior to administration of thrombolysis, then keep <180/105 for 24hrs post administration.
- <6 hours from onset of symptoms catheter directed therapy
- After 6 hours, there is limited benefit to thrombolytic therapy.
 - BP management post stroke without thrombolysis includes an initial period of permissive hypertension, treat only if >220/120. Slowly lower BP to target of 140/90 over 7-14d particularly important to go slow in patients with intra- or extra-cranial large vessel stenosis.
 - Contraindications to permissive hypertension include active CAD, heart failure, aortic dissection, hypertensive encephalopathy.

1.4.3 Surgical - Carotid Endarterectomy

1.4.3.1 Indications

- Recommendations on when to operate after a TIA/stroke.
 - Acute stroke with a fixed neurologic deficit of >6h duration When the patient is medically stable, intervention in 14 days or less after stroke is preferable. (Rothwell et al. 2004; Meershoek and de Borst 2018)
 - Consider urgent intervention in a medically stable patient with mild-moderate neurologic deficit, if there is a significant area of ischemic penumbra at risk for progression on imaging that is contributing to the patient's presentation.

- Stroke in evolution (fluctuating / evolving neurological deficit) or crescendo TIA (repetitive transient ischemia with improvement between events) If the patient's neurological status is not stabilized by medical intervention then an urgent CEA should be considered.
- In this clinical context CEA is preferred to CAS based on an increased embolic potential of carotid lesions. (Rantner et al. 2017)
- What is the only emergent indication for CEA?
 - Crescendo TIAs or a stroke in evolution with a surgically correctable lesion identified on imaging.

1.4.3.2 Techniques

- General concepts
 - Patch angioplasty or eversion endarterectomy are recommended over primary arterial closure due to decreased ipsilateral stroke, perioperative carotid occlusion, return to the operating room, restenosis and 1-year stroke rates.(Goodney et al. 2010; K. Rerkasem and Rothwell 2009) Patch repair is considered the standard of care surgical management for most extracranial carotid lesions.(Arnold and Perler 2019)
- Neuromonitoring/Shunting options during a carotid endarterectomy(Chongruksut, Vaniyapong, and Rerkasem 2014; Wiske et al. 2018)
 - Local/regional anesthesia with direct neurological monitoring the benefit being that the patient is awake and moving to command throughout the case. Indications for shunting include lateralizing deficits, seizure, lack of consciousness and severe anxiety. Improved neuromonitoring, however, has not been shown to reduce myocardial infarction rate with CEA.(A. Rerkasem et al. 2021)
 - Stump pressure- Clamp the inflow and place a butterfly attached to a-line tubing into the internal carotid artery. If the stump pressure is >40mmHg the surgeon can chose to proceed without shunt placement, if <40mmHg then a shunt should be placed prior to proceeding. There remains some controversy with some literature advocating cutoff as high as 50mmHg or as low as 30mmHg.(Arnold and Perler 2019; Chia et al. 2020)
 - **EEG Neuromonitoring** EEG tech places neuromonitoring, monitored by the technician and neurologist remotely intraoperatively. Generally the surgeon should clamp the ICA for 3 minutes before proceeding, if the patient develops neurological deficit/EEG abnormalities with slowing and decreased alpha and beta waves, then unclamp, await normalization of EEG, then proceed.

- Cerebral Oximetry 15% decrease in either hemisphere should be shunted.
- Transcranial Doppler (TCD) severe slowing in the middle cerebral artery. An RCT demonstrated that TCD and cerebral oximetry changes most accurately predicted cerebral ischemia. Detection of a greater than 50% drop in middle cerebral artery velocity using transcranial Doppler is 100% sensitive for detecting cerebral ischemia. (Moritz et al. 2007b) For more see Section 20.3.
- Non-selective shunting the surgeon uses a shunt for all CEAs as standard practice, this is often done for all CEAs performed under general anesthesia
- Techniques to reach high internal carotid lesions?(Beretta et al. 2006)
 - A good knowledge of the available imaging while preparing for the procedure is essential when considering the level of the carotid bifurcation.
 - Nasotracheal intubation will allow for jaw closure and neck extension to facilitate exposure of higher lesions.
 - Division of the posterior belly of the digastric muscle, stylopharyngeal muscle, styloglossus muscle, stylomandibular ligament or the styloid process. Care should be taken to identify and preserve the glossopharyngeal nerve.
 - Division of the occipital artery.
 - Dissection along the posterior parotid gland.
 - ENT surgeon assisted mandible subluxation with assistance if previous techniques fail.
- What is the best technique for a patient with an anatomically kinked internal carotid artery?
 - Eversion carotid endarterectomy (only indication for selection for this technique over patch endarterectomy), transection allows for reduction of redundancy at reanastamosis.
 - Otherwise, no advantage has been shown between eversion or patch, both can be shunted.
- Nerve Injuries where you would encounter these and, if injured, what deficit would be seen?
 - **Hypoglossal** Most commonly injured. Usually passes above the bifurcation of the carotid artery. If injured the tongue will deviate toward the side of injury on protrusion.

- **Glossopharyngeal** Commonly injured during high dissections. Found under the posterior digastric muscle. Injury can be devastating for patients resulting in difficulty swallowing and high aspiration risk.
- **Vagus** Usually found adjacent and lateral to the common carotid. Injury can occur with carotid clamping. Patients develop hoarseness due to innervation of the laryngeal muscles from the recurrent laryngeal nerve which is a branch of the vagus.
- **Marginal Mandibular** Branch of the facial nerve. Runs inferior to the angle of the mandible. Commonly a retraction injury during high dissections leading to the corner of the ipsilateral lip drooping, can be confused with a significant neurological deficit following the c ase.
- Completion duplex for CEA
 - Can identify technical issues with carotid endarterectomy such as clamp site injury, thrombosis, intimal flap or dissection, or missed disease.(Lipski et al. 1996)
 - Acute thrombus may be hypoechoic and difficult to see on B-mode, color flow necessary to determine luminal narrowing or occlusion. For more see Section 20.2.
 - Revision required if residual plaque identified with severe velocity elevation.(Weinstein et al. 2015)
 - Elevated velocities can be seen for many reasons that do not require revision, including incorrect technique (compression with probe), tortuosity, size discrepancy, ICA spasm.(Ricco et al. 2013; Weinstein et al. 2015)
 - One study demonstrated that 2.3% of carotids required revision for concerning findings on completion duplex ultrasound.(Ascher et al. 2004)

1.4.3.3 Complications

- Post-operative stroke What to do if the patient develops neuro deficits following carotid endarterectomy?
 - Two main concerns- an intimal flap causing thrombosis or an embolic event resulting in a stroke. Perform a thorough neurological exam to confirm in operated carotid territory.
 - If in OR perform duplex, very low threshold to re-explore and confirm patency of ICA with duplex or doppler. If ICA open and concerned for distal embolization, then perform a cerebral angiogram and proceed with thrombolysis or thrombectomy.(Fletcher et al. 2016)

- If in Recovery or on the floor many would consider CTA first line vs duplex to look for thrombosis.
- ESRD patients have higher rates of perioperative stroke, but also have higher rates of stroke if not revascularized. (Klarin et al. 2016)
- Risk factors and how to manage hyperperfusion syndrome?
 - Defined as an ipsilateral headache, hypertension, neurological deficits likely in distribution of the carotid which can progress to seizures and intracranial hemorrhage (75-100% mortality). Can present 2-3 days post surgery.(Kaku, Yoshimura, and Kokuzawa 2004)
 - Patients with uncontrolled hypertension are at risk for hyperperfusion syndrome, clinical practice guidelines from SVS recommend strict BP control following CEA, with target pressure being less than 140/80mmHg or 20mmHg from preoperative baseline.(Abou-Chebl et al. 2004; Aburahma and Perler 2022)
- Carotid artery patch infections
 - Rare complication (<1%) presenting as phlegmon, pseudoaneurysm, sinus tract or carotid cutaneous fistula.(Patrick A. Stone et al. 2011)
 - Treatment is excision and replacement with autologous tissue.(Fatima et al. 2019)

1.4.3.4 Outcomes and Surveillence

- Recommend follow up US duplex of bilateral carotids at 30 days. If stenosis >50% is identified, then further imaging is indicated.
- Ipsilateral restenosis
 - 0-3mo likely due to inadequate endarterectomy or clamp injury.(Kang et al. 2014)
 - 1-3y likely neointimal hyperplasia, smooth and regular. If < 80% stenosis and asymptomatic, then duplex surveillance and antiplatelet. If progresses, then often endo treatment first.(Garzon-Muvdi et al. 2016)
 - Risk factors for restenosis include hyperlipidemia(LaMuraglia et al. 2005), female gender, primary closure without patch, younger age, and certain plaque characteristics-abundance of smooth muscle, absence of macrophages/lymphocytes, and lipid core <10%.(Hellings et al. 2008; Pauletto et al. 2000)
- Contralateral stenosis:
 - The risk of progression for moderate stenosis at the initial surveillance to severe stenosis in the contralateral artery can be as high as 5 times.

• Requires post-operative surveillance.

1.4.4 Endovascular - Carotid Artery Stenting

1.4.4.1 Indications

Limited data is available on CAS in asymptomatic patients - currently is not supported by guidelines or considered reimbursable.

Consider CAS in symptomatic patients with >50% stenosis who are poor candidates for CEA due to severe uncorrectable medical comorbidities and/or anatomic considerations

- Ipsilateral neck dissection or XRT equivalent periprocedural stroke rate to CEA, but increased later stroke rate. CEA higher rates of cranial nerve damage (9%). (Giannopoulos et al. 2018)
- Contralateral vocal cord paralysis.
- Lesions that extend proximally to the clavicle or distal to C2.

In patients aged >70 the risk of intra or post procedure stroke was the highest, if undergoing CAS, presumably due to calcified disease in the arch.

- Lesion-specific characteristics are thought to increase the risk of cerebral vascular events after CAS and include a "soft" lipid-rich plaque identified on noninvasive imaging, extensive (15 mm or more) disease, a pre-occlusive lesion, and circumferential heavy calcification.
- This risk can be reduced, but not eliminated, by using flow-reversal embolic protection rather than distal filter protection.

1.4.4.2 Technique

Transfemoral Approach vs Transcarotid approach

- ROADSTER Trial single arm study with flow reversal for cerebral protection. Suggested lower rates of post-op stroke in patients undergoing TCAR.
- TCAR contraindications include previous CCA intervention, CCA disease at entry site, $<5 \mathrm{cm}$ CCA (clavicle to bifurcation) for sheath, CCA $< 6 \mathrm{mm}$, contralateral recurrent laryngeal or vagus nerve injury.(Kwolek et al. 2015)
- Patient needs to be preloaded with dual antiplatelet medications for at least 5-7d, should delay procedure if has not been appropriately loaded with antiplatelets.(Brott et al. 2010; Kwolek et al. 2015; Valls et al. 2017)

• Studies with TCD have shown that there may be reduced embolization rates with TCAR over transfermoral stenting. (Plessers et al. 2016) Complications

Post stenting ICA thrombosis should first be treated with abciximab and catheter directed thrombolysis, followed by endo-salvage with suction thrombectomy, repeat angioplasty or stenting. May require eventual conversion to CEA.(Coelho et al. 2019; A. R. Naylor et al. 2018)

💡 Take a Look

Check out our How I Do It Presentation where Dr Christopher Henry reviews their approach to TCAR at Baylor Scott & White.

1.4.4.3 Outcomes and Surveillance

Dual-platelet therapy should be continued for 1 month after the procedure, and aspirin should be continued indefinitely.

In stent restenosis (>50%) - 4-times higher rate of stroke than no recurrence of disease, but more benign than de novo disease. (Clavel et al. 2019)

- Stenosis should be confirmed with CTA. No clear difference between treatments in literature - PBA, Cutting, DBA, stenting, open conversion.(Arhuidese et al. 2017)
- Repeat angioplasty or stent have low incidence of periprocedural stroke but failed to improve long term stroke/death/MI or patency rates.(Chung et al. 2016)

1.4.5 Summary of Prospective Trials

- 1. Asymptomatic Carotid Atherosclerosis Study (ACAS)(Walker 1995)
 - Compared medical management with CEA in asymptomatic patients with > 60% stenosis.
 - 5-year stroke and death rate was 5.1% vs 11%.
 - In women, the benefit of CEA was not as certain as 5y stroke and death rates were 7.3% vs. 8.7%.
 - This trial was conducted pre-statin and clopidogrel era.
- 2. North American Symptomatic Carotid Endarterectomy Trial (NASCET) (North American Symptomatic Carotid Endarterectomy Trial Collaborators 1991)

- Compared medical management vs CEA for symptomatic patients with moderate (50-69%) and severe stenosis (>70%)
- Only moderate impact for patients with moderate stenosis (50-69%)
- Symptomatic patients with >70 % stenosis benefited from CEA, at 18 months 7% major stroke in surgical arm, and a 24% stroke rate in medical arm. 29% reduction in 5-year risk of stroke or death.
 - Patients with severe >70% stenosis had such a dramatic effect the trial was stopped early for this subset and all referred for endarterectomy.
- No benefit is shown in symptomatic patients with < 50% stenosis
- European studies have shown similar results
 - ACST = ACAS
 - ECST = NASCET.

💡 Take a Listen

Check out our conversation with Dr. William Jordan where he discusses this landmark NASCET paper.

- Carotid Revascularization Endarterectomy versus Stenting Trial (CREST). (Silver et al. 2011)
 - Compared CEA vs. CAS in both symptomatic and asymptomatic patients.
 - Composite endpoint of 30-day stroke, MI, death equivalent between CEA and CAS.
 - CAS had a significantly higher incidence of stroke and death than CEA and CEA higher incidence of MI.
 - Follow up at 10 years demonstrated no difference in composite stroke/MI/death but increased rate of stroke/death in stented patients likely attributable to increased periprocedural stroke. (Brott et al. 2016)
 - CAS patients with stroke had a higher impact on QOL than CEA patients with MI.(Brott et al. 2010; Redfern, Rodseth, and Biccard 2011; Mantese et al. 2010)
 - Subanalyses identified that older patients (>70y) had better outcomes after CEA than CAS, the QOL impact of stroke was more significant than that of MI, and anatomic characteristics of carotid lesions (longer, sequential, remote) were predictive of increased stroke and death after CAS.(Voeks et al. 2011)
 - Unfortunately, this study provides a benchmark to strive for, but no other large trials have achieved these results.

- 4. ROADSTER(Kwolek et al. 2015)
 - Single arm feasibility trial of transcarotid carotid stenting.
 - The results of the ROADSTER trial demonstrate that the use of the ENROUTE Transcarotid NPS is safe and effective at preventing stroke during CAS. The overall stroke rate of 1.4% is the lowest reported to date for any prospective, multicenter clinical trial of CAS (i.e. CREST). No direct comparison.
 - Recent multi-institutional post market surveillance study demonstrated found TCAR outcomes to be "broadly similar" to carotid endarterectomy.(Kashyap et al. 2019)

💡 Take a Listen

Check out our conversation with Dr Jeffrey Jim where he discusses details of comparative effectiveness research and how it has been applied to TCAR.

1.5 Uncommon Carotid Disease

1.5.1 Complicated Extracranial Occlusive Disease

Occluded Carotid: What is the management of an occluded carotid?

• Leave it alone.

What if an occluded carotid on imaging is still causing TIAs?

- Ongoing symptoms in the setting of carotid occlusion is known as carotid artery stump syndrome. Symptoms are likely due to emboli from friable intima of the internal carotid or from the external carotid through collaterals. Treatment is external carotid endarterectomy and ligation of the internal carotid.(Hrbáč et al. 2012)
- The addition of oral anticoagulation is likely to reduce the rate of recurrent stroke.

Simultaneous coronary and carotid disease

- Patients with symptomatic carotid stenosis (50-99%) will benefit from CEA before or concomitant with CABG. The timing of the intervention depends on the clinical presentation and institutional experience (grade 2C).(Aburahma and Perler 2022)
- Patients with severe bilateral asymptomatic carotid stenosis, including stenosis and contralateral occlusion, should be considered for CEA before or concomitant with CABG (grade 2C).(Aburahma and Perler 2022)

• Patients undergoing simultaneous CEA/CABG demonstrate highest mortality. (A. R. Naylor et al. 2003)

What about tandem lesions in the carotid in a symptomatic patient, carotid bulb and carotid siphon lesion (high ICA)? How should you treat this?

- Treat carotid bulb and extracranial disease first, likely to be the embolic source.
- Treatment of intracranial disease has been shown to increase rates of stroke and death.(Chimowitz et al. 2011). Extracranial to intracranial bypass has increased risk of stroke over BMT.(E. B. S. Group 1985)

What if the patient has severe vertebrobasilar insufficiency and carotid artery disease?

- Should undergo carotid revascularization first to improve flow. Then reassess for vertebrobasilar insufficiency.
- Vertebrobasilar insufficiency characterized by dizziness, ataxia, nausea, vertigo and bilateral weakness. (Lima Neto et al. 2017a; Nouh, Remke, and Ruland 2014)

1.5.2 Vertebrobasilar Disease

Most common etiology is atherosclerosis, but can also be caused by embolism, dissection or migraines.(Lima Neto et al. 2017b)

Symptomatic primary vertebral disease with >60% stenosis should be treated. Vertebral divided into segments. V1 preforaminal (lower than C6), V2 foraminal (C6-C2), V3 (C2-dura), V4 (intracranial).(M. Morasch 2019; Berguer, Morasch, and Kline 1998)

Disease of V1 should be treated with vertebral transposition of bypass with vein.(Rangel-Castilla et al. 2015)

• Most common complication is Horner's Syndrome due to disruption of the sympathetic chain (20%), next is TIA (1-3%).(Coleman et al. 2013; C. A. Ramirez et al. 2012)

Disease of V2 is most difficult to access and should be treated with bypass from ICA to V3.(Berguer 1985)

RCT of BMT vs endo treatment did not show superiority of endo treatment, recurrent symptoms most common, periprocedural stroke was rare.(George A. Antoniou et al. 2012; Coward et al. 2007)

1.5.3 Carotid artery dissection

Patients with carotid dissection should be initially treated with antithrombotic therapy (antiplatelet agents or anticoagulation) (GRADE 1, Level of Evidence C).

Indications for endovascular treatment of carotid artery dissection (Cohen et al. 2012; Markus et al. 2019; Pham et al. 2011)

- Ongoing symptoms on best medical therapy
- Contraindication to antithrombotics
- Pseudoaneurysm

1.5.4 Internal Carotid Artery Aneurysms

Defined as dilation of the bulb 200% of the internal carotid artery or 150% of the common carotid artery. Like popliteal aneurysms the major risk is thromboembolization.(Bush, Long, and Atkins 2019; Fankhauser et al. 2015; Pourier et al. 2018)

Treatment is resection with primary repair or interposition. Distorted anatomy increased the risk for cranial nerve injury (12%).(Welleweerd et al. 2015)

Redundant ICA - should be repaired if patient symptomatic due to kinking. Treated with resection and primary repair with shortening and straightening.(Ballotta et al. 2005)

Fibromuscular Dysplasia (FMD) (Jeffrey W. Olin and Sealove 2011) - for more, see Section 12.2

- Asymptomatic disease can often be treated with antiplatelets alone.
- Duplex ultrasound demonstrates classic "chain of lakes" appearance. For more see Section 20.2
- Associated with intracranial aneurysms (13%), therefore may be reasonable to also pursue brain imaging if identified. However, no formal guidelines exist.(Lather et al. 2017)

1.5.5 Carotid Body Tumors (CBT)

1.5.5.1 Demographics

One quarter (25%) are associated with a germline mutation or familial history. Genetic testing recommended in patients with multifocal tumors, associated paraganglioma, pheochromocytoma or family history. SDHD gene mutation is most common associated with familial paraganglioma.(Davila et al. 2016; Kruger et al. 2010) Blood supply primarily from the external carotid artery. (Robertson et al. 2019; Davila et al. 2016)

Shamblin criteria utilized to determine difficulty of resection and is associated with rates of carotid reconstruction or cranial nerve injury. (Chow, Moore, and Lamuraglia 2019; Shamblin et al. 1971)

- 1. Type 1 abut but do not involve internal or external carotid arteries or cranial nerves can be easily dissected.
- 2. Type 2 partially surround external carotid artery or cranial nerves and may be dissected free but risk injury to these structures.
- 3. Type 3 fully surround either internal or external carotid artery or cranial nerves and pose a high risk of injury and need for reconstruction.

1.5.5.2 Management

Risk of injury or revascularization during resection determined by Shamblin criteria as discussed above. In patients with Shamblin Type 3 or with involvement of the internal carotid, these patients should be evaluated for carotid artery ligation.

- When involved with a CBT or other skull base tumors, cerebral perfusion should be thoroughly evaluated prior to surgical resection to determine whether a patient will tolerate carotid ligation.
- Evaluation includes balloon occlusion testing with xenon-enhanced SPECT imaging. Low risk for ligation if no neurologic deficits (after 30min and hypotensive challenge) and normal diffusion of SPECT (ratio of >0.9). Angiography with contralateral occlusion evaluates collaterals.(Sugawara et al., n.d.; Tansavatdi et al. 2015)

Embolization if tumor arterial inflow may help reduce risk of perioperative blood loss but is not associated with reduced cranial nerve injury. Resection should take place 24-36hrs after embolization.(Power et al. 2012)

2 Trauma - Cerebrovascular

Authors: Kevin Kniery, Nicole Rich, Nakia Sarad, and Todd Rasmussen

These trauma episodes were developed in collaboration with Behind the Knife: The Premier Surgery Podcast.

Vascular trauma requires close collaboration between vascular surgeons and trauma surgeons. The decision of which specialty should manage which injuries vary across different centers. Through these trauma chapters we will discuss what management decisions the majority of vascular surgeons should be comfortable managing as a part of a multidisciplinary trauma team.

? Take a Listen

Check our debate between leading trauma and vascular surgeons about how to best develop a collaborative team to manage vascular trauma.

For relevant images and a more in-depth discussion of this topic, please review **Chapter 20: Neck and Thoracic Outlet**(Magee and Weaver 2022) in Dr. Rasmussen's 4th Edition of *Rich's Vascular Trauma*.

2.1 Neck Exposure Techniques

2.1.1 Carotid Artery

Anatomy

- Right common carotid originates from brachiocephalic artery
 - External landmark: right sternoclavicular joint
- Left common carotid originates from aortic arch in superior mediastinum
- Common carotid artery (CCA) is within the carotid sheath
 - Contents:
 - Common and Internal carotid (medially)

- Internal jugular vein (laterally)
- Vagus nerve (posteriorly)
- Anterior margin: sternocleidomastoid (SCM) and omohyoid muscle
- Posterior margin: longus colli and longus capitis muscles
- Medial margin: esophagus/trachea
- Common carotid bifurcates to external and internal at level of superior border of thyroid cartilage
 - External carotid artery
 - Medial to internal carotid artery (majority of course)
 - First branch is superior thyroid artery (near bifurcation)
 - Terminate in parotid gland, divide into superficial temporal and maxillary arteries
 - Internal carotid artery
 - No extracranial branches
 - Lateral to external carotid artery until level of skull base, where it crosses medially
 - Enters the skull base through the carotid canal behind the styloid process
 - Facial vein is the anatomical landmark that approximates the location of the carotid bifurcation (deep to it).
 - Hypoglossal nerve (CN XII) and posterior belly of digastric muscle superficially cross the internal and external carotid arteries at the angle of the mandible
 - Glossopharyngeal nerve (CN IX) crosses in front of internal carotid artery, superior to hypoglossal nerve (CN XII)

Exposure

- Longitudinal incision along anterior border of SCM, extending from suprasternal notch to mastoid process
 - For proximal common carotid artery control, would do combination of SCM incision with median sternotomy
- Incise platysma and expose anterior border of SCM, retract laterally
 - Identify and avoid Accessory nerve (CN XI) that enters SCM

- Small branches of external carotid should be ligated to adequately mobilize the SCM and expose carotid sheath
- Carotid sheath is visualized and incised longitudinally
 - Omohyoid muscle may require division if proximal exposure required
- Identify Internal jugular vein, mobilize, and retract laterally
- Identify Vagus Nerve and avoid injury (posterior between internal jugular vein and common carotid)
- Place vessel loops around CCA, internal jugular, and vagus nerve
- Identify facial vein (overlying CCA bifurcation), Hypoglossal nerve (CN XII) and Ansa cervicalis
 - Ligate facial vein to further mobilize the internal jugular vein laterally and expose the underlying carotid bifurcation
 - Identify and protect Hypoglossal nerve (CN XII)
 - Ansa cervicalis may be ligated if necessary for exposure
- Dissect carotid bifurcation
 - Careful to not stimulate carotid body and cause hemodynamic instability
 - If access to distal internal carotid artery is required, may consider subluxation of mandible or mandibular osteotomy to gain appropriate exposure

2.1.2 Vertebral Artery

Anatomy

- First cephalad branch of subclavian artery
- Divided into 3 parts
 - Part I (proximal)
 - Origin at subclavian artery to C6 (enters transverse foramen)
 - External landmarks: found between two heads of SCM (sternal and clavicular) and clavicle
 - Vertebral artery runs between anterior scalene and longus colli muscles
 - Part II Courses through bony vertebral canal (transverse foramen) from C6 to C1
• Part III (distal) - Courses outside vertebral canal from C1 to base of skull (enters through foramen magnum) and joins contralateral vertebral artery to form basilar artery (part of circle of Willis)

Exposure

- Will focus on exposure of Proximal Vertebral Artery
- Supraclavicular transverse incision extending between the sternal and clavicular heads of the sternocleidomastoid (SCM) muscle
- Incise platysma and continue dissection into base of triangle
- Carotid sheath is first vascular structure identified
 - Jugular vein lateral
 - Common carotid medial
 - Vagus nerve posterior
- Identify scalene fat pad between two heads of SCM and clavicle, dissect to expose anterior scalene muscle
 - Careful of phrenic nerve as it runs on surface of anterior scalene muscle
- Divide anterior scalene muscle to visualize the subclavian artery and two of its branches (thyrocervical trunk and internal mammary artery). Control each branch.
- Proximal VA is located deep to supraclavicular artery and anterior to groove between C7 vertebral body and transverse process
- Proximal VA is between anterior scalene (laterally) and longus colli (medially)
- Best found through palpation with tip of index finger within groove of C7 vertebral body and transverse process
- Avoid injury to vertebral venous plexus as it is anterior to VA
- Use right angle clamp to dissect out VA

Detailed exposure techniques can be found here(E. Kwon, Grabo, and Velmahos 2019)

2.2 Tenets of Cerebrovascular Injury

- Control of catastrophic bleeding with manual compression and resuscitation
- Ensure that injury is not causing airway compromise
- Evaluate the neurological status of patient have a thorough baseline neuro exam and look for signs of ischemia, hemispheric stroke, arousability/mental status
- Prevent secondary injury maintain blood pressure to maintain cerebral perfusion pressure and prevent hypoxia
- After appropriate exposure and control, recommend that patient be heparinized during carotid artery repair. Be prepared for bleeding, especially in patients with polytrauma. Resuscitate appropriately until repair completed.

2.3 Blunt Trauma

- Mechanism of Injury: history of a motor vehicle collision, blunt force to neck, severe hyperextension/rotation/flexion injuries, etc.
- External signs of injury: marks on neck or around the thoracic inlet (i.e. "seatbelt sign"), unilateral neurodeficits (suggestive of hemispheric ischemia or stroke)
- Accompanying injuries: concomitant closed head injuries (TBIs), complex facial fractures, cervical vertebral body and transverse foramen fractures

2.3.1 Evaluation

- CTA is gold standard. Limitation, image is static.
- US duplex combines B mode ultrasound with pulse doppler. Limitation in visualization of injury above the angle of the mandible (distal internal carotid and proximal thoracic outlet)
- These imaging modalities are used to assess visibility of defect (i.e. dissection, thrombus, intimal flap) and degree of flow limitation by measured velocity.

Denver Grading Scale for Blunt Vascular Carotid Injury (BCVI)(Biffl et al. 2001)

- Grade I: Luminal irregularity or dissection with < 25% luminal narrowing
- Grade II: Dissection or intraluminal hematoma with > 25% luminal narrowing, intraluminal thrombus, or raised intimal flap
- Grade III: Pseudoaneurysm
- Grade IV: Occlusion
- Grade V: Transection with free extravasation

2.3.2 Management

Considerations

- Consider the overall injury severity of the trauma patient with any concomitant injuries that may prevent antiplatelet/anticoagulation initiation (i.e. TBI) or intervention(Bromberg et al. 2010)
 - Oftentimes have other blunt vascular injuries, such as a ortic injuries, that need to be addressed as well
 - Can consider delay of intervention (approx. 7 days) to decrease risk of neurological events
- If there is a change in the neurological exam after initiation of medical management, considered as failure of medical therapy and surgical intervention is indicated
- If unable to obtain a reliable neurological exam, can consider transcranial doppler or EEG if accessible, but is not mandatory
- Continue supportive measures to prevent secondary injuries, i.e. optimize cerebral perfusion pressure
- Repeat imaging with change in neurological status
- Serial imaging important in assessing whether injury has evolved
- Endovascular repair usually involves transfemoral approach for access with arch aortogram, selection of defective carotid vessel, and placement of the endovascular treatment (ie: stent graft, bare metal stent, coil embolization of pseudoaneurysm)

Carotid Artery Injury Treatments

- Grade I and II injuries are recommended to be treated medically with antiplatelet/anticoagulation therapy if patient is able to tolerate
 - Repeat CTA imaging in 5-7 days while inpatient to ensure that injury has not evolved in acute phase
 - Follow-up CTA imaging within 1 month of discharge as outpatient to assess healing. Imaging needs to be continued until there is resolution of lesion.
 - Grade I lesions typically improve over time, Grade II lesions or higher have a greater chance of evolving (approximately 60%)
- Accessible Grade III and V carotid injuries should be repaired.

- Grade III pseudoaneurysms should be repaired endovascularly with a covered stent (ie stent graft that covers the opening of the pseudoaneurysm) or bare metal stent (maintains integrity of the wall). Through these stents, coils can be placed to thrombose the pseudoaneurysm. Flow needs to be maintained.
- Open approach can be considered for Zone II injuries
- Grade I and IV carotid injuries, and inaccessible Grade II and III carotid injuries should be treated with medical management (antiplatelet/anticoagulation)
 - Grade IV injuries without associated injuries that preclude use of anticoagulation, would heparinize and proceed with anticoagulation (avoid bolus of heparin, goal PTT 50-70 within first 24 hours). Monitor for bleeding. Transition to oral anticoagulant for 30-90 days.
 - Grade IV injuries with contraindications for anticoagulation, would recommend dual antiplatelet therapy (aspirin and clopidogrel)
 - Overall, anticoagulation is preferred if there are no contraindications. If there are, then dual antiplatelet therapy is preferred.
- Grade V and persistent Grade III should be repaired surgically

Vertebral Artery Injury Treatments

- Special Considerations
 - Attempt to determine whether the dominant or diminutive vertebral artery is involved
 - Treatment is again based on severity with grading system (Denver Grading Scale), consider location and extent
 - Consider any associations with a basilar artery defect or posterior circulation stroke
 - Grade I-IV vertebral injuries should be treated with medical management as first line, approximately 90% self-resolve
 - Endovascular interventions can be considered for those patients with severe injury and/or who are symptomatic
 - Usually performed by neuro-interventionalist as the injury is located within the skull base
 - Serial imaging still needs to be performed to evaluate healing (similar manner to carotid injury)

- In the setting of trauma, if there is an expanding hematoma due to vertebral artery injury and open approach is performed to control bleeding, vertebral artery is usually ligated in this scenario
 - Vertebral reconstructions are not well-suited for traumas

Guidelines and treatment algorithm can be found here(Geddes et al. 2016)

2.4 Penetrating Trauma

Levels(Roon and Christensen 1979)

- Zone I: thoracic inlet (sternal notch) to level of cricothyroid cartilage
- Zone II: cricothyroid cartilage to angle of mandible
- Zone III: angle of the mandible to the base of the skull

Images of the neck zones can be found here

2.4.1 Management

Injury to Zone I and Zone III

- If hemodynamically stable, requires imaging prior to consideration of intervention/repair
 - Imaging: CTA and/or angiogram
 - CTA positive » Endovascular intervention vs. OR
 - Consider concomitant injury to aerodigestive tract as well
- If hemodynamically unstable with/without hard signs (aerodigestive or neurovascular injuries)
 - Tamponade » Secure Airway for Air leak/hematoma » OR

Injury to Zone II (violation of platysma)

- Direct to OR for exploration, especially with high suspicion of carotid injury and neurovascular deficits (hard sign)
- Repair carotid artery if injured, even if thrombosed.
 - The risk of reperfusion injury causing an intracranial hemorrhage was traditionally considered a reason to not repair.

- However, overall mortality and final neurostatus has shown to be better with intervention, even if patient started with neurodeficits.
- Requires communication with operative team (i.e. anesthesia) to mitigate hypoperfusion
- Consider concomitant injury to aerodigestive tract
- Scenario: Patient has large cerebral infarct on same side as carotid injury.
 - Continue to OR to explore
 - Preferred management is to repair and attempt reperfusion of hemisphere, possibly perfusing a previously ischemic penumbra
 - Ligation usually not an option.
 - Unless carotid is completely occluded and there is no retrograde bleeding after attempted thrombectomy » there is potential for ligation, but not readily considered.
 - Observation without repair can risk evolution of the injury into pseudoaneurysm with need to repair and reoperate days/weeks later Outcome of this repair is likely a stroke

Repair Techniques

- 1. Appropriate exposure of carotid artery (see above for details)
- 2. Gain proximal and distal control
- 3. Identify injury
- 4. Remove burden of thrombus (if present)
 - Thrombus burden must be removed prior to repair
 - Via flushing from back-bleeding and forward-bleeding of internal carotid artery Via Fogarty or thrombectomy catheter, carefully
 - Recommend 2 French or 3 French size
 - Pass only 1-2 cm up into distal internal carotid
- 5. Debride any devitalized tissue prior to repair
 - May cause defect to be larger, but repair requires healthy tissue planes

- 6. Repair carotid Heparinization required when repairing carotid artery and/or clamping. Be efficient, as these patients usually have significant polytrauma and other areas that are susceptible to bleeding which will need to be controlled when heparin administered. Repair options, include:
 - Primary Repair
 - Patch Repair Used for short-segment injuries. Conduits include:
 - Dacron or bovine pericardium
 - Autologous vein Not first-line due to extra time required for harvest. Used if there is concern for contamination (i.e. shot gun wound with large soft tissue injury, aerodigestive injury, etc)
 - Interposition Graft Used for long-segment injuries. Conduits include:
 - Autologous vein (great saphenous vein) Good option when there is concomitant aerodigestive injury. Consider shunting while waiting for harvest
 - Synthetic (i.e. Dacron/PTFE)
 - Transposition of Internal Carotid Artery to External Carotid artery Can avoid need for shunting

Consideration of Shunting

- In most cases, shunting not required because adds extra level of complexity to trauma case that may not have the luxury of time
- Cases where shunting can be used:
 - Interposition graft repair for long-segment carotid injury that requires vein harvest > shunting can be performed to reperfuse the affected brain hemisphere while waiting for conduit preparation
- Typical shunts used: Argyle or Sundt

Assessment of Repair

- Assess repair with removal of thrombus and once back-bleeding restored
- Duplex US
 - Best option when time is limited to assess flow
 - Usually also used most often with elective carotid repairs

- If ultrasound is not available, then a continuous wave doppler is at least necessary to demonstrate monophasic signal with diastolic flow in the revascularized internal carotid artery.
- Intraoperative arteriograms
 - If have luxury of time to perform and set up is appropriate, used for completion studies
 - Documents complete evacuation of distal thrombus
 - Performed by placing 18-gauge butterfly needle in CCA (remove air) and inject contrast to view repair and distal internal carotid to ensure lack of thrombus

Consideration of Drains

- If there is contamination (aerodigestive injury), then can place a small JP drain (7mm) to protect the repair
- Buttressing repair in contaminated cases usually not used as the neck is highly vascular
 - Can consider a muscle flap to separate arterial injury from esophageal injury (commonly not seen) with gross contamination

Guideline Algorithm for Penetrating Neck Trauma(Sperry et al. 2013)

2.5 Internal Jugular Vein Injuries

- If patient has significant TBI, consider repairing Internal Jugular venous injuries to maintain venous outflow to that side of the brain especially if there is a transduced venous pressure gradient. Cases include patients who have intracranial hypertension and require a decompressive craniectomy.
- Otherwise, able to ligate without much consequential effects in most patients due to appropriate amounts of collateral outflow.

Part II

Upper Extremity

3 Upper Extremity

Authors: Kush Sharma and Ashraf Mansour Contributors: Hattan Alghamdi

3.1 Overview

3.1.1 Anatomy

What are the zones of the upper extremity?(Karl A. Illig 2019)

Division of the upper extremity into three zones:

- 1. Intrathoracic Zone: Inclusive of the aortic arch and its branches, innominate artery (dividing into right subclavian and right common carotid), left subclavian artery and left common carotid artery, as well as the innominate veins, and SVC.
 - Arch classification helps determine difficulty of accessing the right innominate from a femoral approach. Type 1 innominate does not break the plane of the outer curve, Type 2 it is between the outer curve and inner curve and Type 3 it falls below the inner curve of the aortic arch.
- 2. Thoracic Outlet: Extends from the base of neck to the axilla. Includes the subclavian, proximal vertebral, and proximal axillary arteries, as well as the corresponding veins).
- 3. Axilla to Fingers: The arm.

What are some common exposures for major upper extremity arteries?(Karl A. Illig 2019; Gary G Wind and R. James Valentine 2013)

- 1. Proximal Right Subclavian Artery: Access through a median sternotomy.
- 2. **Proximal Left Subclavian Artery:** An anterolateral thoracotomy is required in emergent settings for proximal left subclavian artery control.
 - For proximal, mid and distal exposure is required a "trap door" exposure is used which is a combined anterolateral thoracotomy, third space sternotomy, and left supraclavicular incision.

- 3. Mid and Distal Subclavian Artery: Access through a supraclavicular Incision. After division of the platysma and clavicular head of the SCM, a fat pat of varying thickness is encountered which contains the omohyoid muscle.
 - This should be divided and placed supero-laterally. At this point, the anterior scalene muscle is exposed medially with the phrenic nerve running in a lateral to medial direction, which should be identified and preserved.
 - In a carotid/subclavian bypass division of the anterior scalene should be performed as close to the first rib as possible. Once divided the subclavian artery is exposed.
- 4. Axillary Artery: Exposure is gained through an infraclavicular incision, below the middle third of the clavicle.
 - Pectoralis major is divided and pectoralis minor is freed at the lateral margin of the wound. The axillary vein should be identified and followed supero-deeply to identify and expose the artery. Anatomically bound by the first rib proximally and the lateral edge of the teres major muscle distally.
 - For exposure of the first part of the axillary artery, the ipsilateral arm is abducted approximately 90 degrees and a horizontal skin incision 2 cm below the middle third of the clavicle should be made. The underlying pectoralis major is split by bluntly separating the fibers, exposing the tough clavipectoral fascia. At the lateral wound margin, the pectoralis minor can be freed and laterally retracted.
 - The axillary vein is the first structure to be encountered in the fascia and the artery lies just superior and deep to the vein. Make sure to protect the nerves of the brachial plexus which lie deep to the first part of axillary artery and are at risk of injury during blind placement of occluding arterial clamps.
- 5. Brachial Artery, Upper Arm: The brachial artery's superficial location makes it vulnerable to injury and accounts for most vascular injuries of the upper extremities.
 - Brachial artery exposure involves a 5-8 cm longitudinal incision in the groove between the biceps/triceps muscles on the medial aspect of the arm. In the lower half of the arm, take care to avoid basilic vein damage as it runs in the subcutaneous tissue.
 - The neurovascular bundle is exposed by incising the deep fascia at the medial border of the biceps muscle, which then should be retracted anteriorly. The basilic vein should be identified and retracted into the posterior wound.
 - On opening the brachial sheath the median nerve is the most superficial structure and should be identified, protected and retracted. The brachial artery lies deep to the median nerve and is flanked by two brachial veins. Be mindful that the ulnar nerve lies posteriorly.

- 6. Brachial Artery, Antecubital Fossa: Exposure of the brachial artery in the antecubital fossa requires a transverse skin incision 1 cm distal to the midpoint of the antecubital crease.
 - After deepening, mobilize the basilic vein medially to avoid injury. The medial antebrachial cutaneous nerve should be identified and protected. The bicipital aponeurosis should be divided to exposure the brachial artery, which is flanked by two deep veins crossing tributaries.
 - To isolate the brachial artery ligation and division of these tributaries is required. The brachial artery bifurcates at the radial tuberosity into its radial and ulnar branches.
 - Immediately after the ulnar artery arises from the bifurcation origin, it gives off a short common interosseous branch, which bifurcates at the hiatus in the proximal interosseous membrane.
- 7. Radial Artery at the Wrist: This requires a 2-3 cm longitudinal incision generally between the palpable radial artery and cephalic vein.
 - Incision of the antebrachial fascia medially to the radius should expose the radial artery. Two veins accompany the artery and should be dissected away to isolate the artery.
 - The superficial radial nerve and its medial/lateral branches course between the cephalic vein and radial artery in this area.
- 8. Ulnar Artery: The ulnar artery courses beneath the superficial flexor muscles of the proximal forearm, emerging near the ulnar border at the midpoint between the elbow and the wrist.
 - In the distal forearm, the ulnar artery runs just deep to the antebrachial fascia and is easily exposed through a longitudinal incision placed radially to the flexor carpi ulnaris.
 - The palmar branch of the ulnar nerve courses superficially to the antebrachial fascia and should be preserved during arterial exposure to avoid injury.

What common aberrant upper extremity/arch anatomy is important to be aware of?

- 1. Bovine Arch: Where the innominate and left common carotid arteries share a common origin.(Layton et al. 2006)
- 2. Aberrant Vertebral Artery: Where the vertebral artery arises directly from the aortic arch.

- 3. Aberrant Right Subclavian Artery: Where there is no innominate artery and instead the right common carotid arises directly from the aorta.
 - Confusingly the right subclavian artery, then originates from the aortic arch distal to the left subclavian artery.
 - To supply the right arm it therefore, needs to pass behind the esophagus and anterior to the spine.

Of note, thoracic aortic disease is associated with aberrant anatomy, with high rates of bovine arch (25%), isolated L vertebral (6-8%), and aberrant R subclavian (1-2%).(Dumfarth et al. 2015)

3.1.2 Etiology

How does evaluation of upper extremity ischemia differentiate from lower extremity ischemia?

Upper extremity ischemia <5% of patients with limb ischemia are a result of underlying atherosclerosis, which is in contrast to lower extremity.(Shuja 2019)

Vast majority of cases caused by embolic, autoimmune, or connective tissue disorders.

How can upper extremity disease be classified?

Anatomic Location:

• Large vs. Small Vessel

Disease Process:

- Vaso-occlusive disease often requires endovascular/surgical management.
- Vasospastic disease is more responsive to pharmacological management.

3.1.3 Evaluation

How should patients be evaluated who have concern for upper extremity disease?

- 1. Detailed H+P evaluation (pulse palpation, auscultation at supraclavicular/infraclavicular fossa may reveal a bruit concerning for subclavian artery stenosis, upper extremity neurovascular/skin exam)
- 2. Brachial/forearm blood pressures and if suspected claudication, measured at rest and 2-5 minutes after exercise. Look for a gradient of >20 mmHg is considered significant

- 3. Some or all of 6 P's of acute limb ischemia with symptoms occurring within 14 days are deemed acute
- 4. Doppler insonation of radial, ulnar, palmar, and digital arteries
- 5. Vascular Lab Evaluation
 - 1. Segmental Pressure Measurements
 - 2. Duplex Ultrasound (look for large vessel occlusive disease)
- 6. Other Imaging
 - 1. CTA/MRA
- 7. Clinical Lab tests
 - 1. Inflammatory disorders-CBC, ESR, CRP, ANA, RF
 - 2. Hypercoagulable screening

3.2 Vaso-occlusive Disease

3.2.1 Demographics

What are causes and symptoms associated with subclavian/axillary occlusive disease? (Jack L Cronenwett, Alik Farber, and Erica L. Mitchell 2020)

Etiology: Atherosclerosis is the most common cause of subclavian/axillary occlusive disease. Left SCA > Right involvement. Less common causes include Takayasu disease, giant cell arteritis, or arterial TOS

Symptoms: Upper extremity arm/hand ischemia or neurologic symptoms due to subclavian-vertebral steal.

- Because significant collaterals, minimal pain on exertion even with subclavian occlusion.
- Innominate disease can present with complex steal syndrome from the vertebral artery.(Rodriguez 2016)

What are causes and symptoms associated with brachial/forearm occlusive disease?

Etiology: MCC of brachial artery occlusion is cardiac origin embolus.

• Atherosclerosis RARELY affects the brachial artery. Forearm occlusive disease can be seen in advanced ESRD or DM where calcific atherosclerosis of radial/ulnar arteries is present.

- Distal axillary/proximal brachial stenosis can be from repetitive trauma from crutch use.
- Radiation arteritis may be seen with radiation after breast cancer therapy. Lesions characteristically tapered with smooth border. May occur alongside a brachial plexopathy.(Goldstein et al. 2010; Modrall and Sadjadi 2003)
- Less common causes include Buerger's disease or Raynaud Phenomenon

3.2.2 Management

3.2.2.1 Chronic Limb Threatening Ischemia

How/when is upper extremity occlusive disease treated?

Indications for treatment

- 70% stenosis and associated symptoms or deficit
- Lesion with ipsilateral ICA requiring treatment
- Pre-occlusive lesion in a good surgical candidate (>5y life expectancy)

Innominate or SCA Occlusive Disease

- Endovascular with balloon expandable stent via femoral or ipsilateral brachial artery. (Chatterjee et al. 2013; Bradaric et al. 2015; Saha et al. 2017; Palchik et al. 2008; Mordasini et al. 2011) Preferred in:
 - Short segment or ostial disease with adequate distance to the vertebral artery origin.
 - History of neck surgery or radiation.
- Surgery:
 - Bypass from aortic arch through median sternotomy Most durable option, particularly for dense R innominate disease, so preferred in patients who can tolerate.
 - Endarterectomy not appropriate if disease involves the origin.(Aiello and Morrissey 2011; Mansukhani et al. 2018; Berguer, Morasch, and Kline 1998; Daniel et al. 2014; Reul et al. 1991; Berguer et al. 1999; Byrne et al. 2007)
 - Complications include compression of the graft on sternal closure, so proximal anastomosis should be placed to the side of the aorta.

- Ipsilateral CCA to subclavian artery (bypass or transposition) Recent NSQIP study showed no difference in stroke and death for CS bypass vs transposition, therefore procedure should be determined by anatomic constraints.(Cinà et al. 2002; Vliet et al. 1995) Small studies and systematic reviews have shown that patency of carotid subclavian bypass is better with synthetic graft than with autologous vein.(AbuRahma, Robinson, and Jennings 2000; Illuminati et al. 2018; Ziomek et al. 1986)
 - An additional indication for carotid subclavian bypass is as a staged revascularization prior to TEVAR for an eurysmal disease requiring coverage of the LSA
 - How does the exposure differentiate in transposition vs bypass?
 - Arterial transposition via a short, transverse cervical incision above the clavicle between two heads of SCM (bypass is lateral to entire SCM)
 - Sub-platysmal flaps created and avoid EJ vein damage
 - Omohypoid divided between heads of SCM and IJ mobilized laterally (bypass IJ is mobilized medially to expose CCA and care must be taken to avoid phrenic nerve in more lateral approach)
 - CCA is reflected medially with vagus nerve
 - On the left side, the thoracic duct is identifiable and divided followed by dividing the vertebral vein
 - Subclavian artery and proximal branches identified (anterior scalene is in lateral dissection)
 - What are some common complications after carotid subclavian bypass in order of highest to lowest incidence?
 - 1. Phrenic nerve palsy (most common) most often managed conservatively.
 - 2. Recurrent laryngeal palsy
 - 3. Lymphatic leak, seen more on the L with thoracic duct injury
 - 4. Neck hematoma (Voigt et al. 2019)

• When carotid-subclavian bypass compared to transposition?

 Vertebral artery takes origin from the subclavian artery in a very proximal position or is dominant over the contralateral side, then bypass preferred. (M. D. Morasch 2009a)

- 2. For coronary-subclavian steal with patent internal mammary artery to coronary artery bypass graft, then bypass (a carotid-subclavian transposition requires a more proximal clamp with occlusion of inline antegrade flow to the coronary bypass during the procedure) (Cua et al. 2017)
- Contralateral CCA bypass with anterior or retropharyngeal tunneling.

Brachial/forearm Occlusive disease

Endovascular: PTA evidence is an ecdotal with stents for lesions unresponsive to PTA or dissection following angioplasty. (Cheun et al. 2019; Nasser et al. 2014; Dineen, Smith, and Arko 2007)

Surgery: GSV vein bypass remains standard for revascularization with bypasses to superficial or deep palmar arch have good patency rates. Tunneling is subcutaneous if to distal ulnar or superficial palmar arch whereas anatomical to distal radial artery over the anatomic snuffbox.(B. B. Chang et al. 2003; Masden, Seruya, and Higgins 2012; Spinelli et al. 2010)

3.2.2.2 Acute Limb Ischemia

What is the procedure of choice in acute limb ischemia of the upper extremity?

With acute presentation of upper limb ischemia and a localizing examination, may be reasonable to proceed directly to OR for embolectomy to minimize ischemia time.(Wahlberg, Goldstone, and Olofsson 2006; Henke 2009)

What are some postoperative complications of an embolectomy for acute limb ischemia?

Brachial sheath hematoma - parasthesias and weakness in the median nerve distribution require emergent re-exploration and decompression. Even small hematomas that are not readily identifiable on clinical exam, can cause compression on the median nerve and are a surgical emergency.(D. D. Tran and Andersen 2011)

Compartment Syndrome - If prolonged ischemia or no hematoma found, and forearm compartments appear tense, then should proceed with fasciotomies to treat forearm compartment syndrome. Rarely performed prophylactically due to significant morbidity.(Gelberman et al. 1981; Kistler, Ilyas, and Thoder 2018; Leversedge et al. 2011)

- Three compartments forearm volar (flexor, superficial, and deep), mobile wad (lateral), and dorsal (extensor, superficial, and deep). Volar compartments most susceptible to ischemia and compartment syndrome, most vulnerable muscle are flexor digitorum profundas and flexor pollicus longus.(Ronel, Mtui, and Nolan 2004)
- Volar/Henry approach decompresses lateral and volar compartments with a single incision, includes carpal tunnel release.

3.3 Vasospastic Disorders

3.3.1 latrogenic or Vasopressor Induced Ischemia

You are called to the intensive care unit for a septic patient with acute bilateral upper limb ischemia, what is the management strategy?

Critically ill patients on vasopressors with bilateral upper extremity ischemia are often managed best with supportive measures and attempts to wean vasopressors (especially norepinephrine).(Gregory J. Landry et al. 2018)

Limited role for radial artery embolectomy, only in clear ischemia isolated to the distribution of an occluded radial artery, often secondary to line placement. (Valentine, Modrall, and Clagett 2005)

3.3.2 Raynaud's Phenomenon

What is Raynaud's and what causes it?

Exaggeration of normal physiologic response with episodic pallor or cyanosis of the fingers caused by small digital artery vasoconstriction occurring in response to cold or emotional stress. There is an abnormality with sympathetic nervous system, resulting in a multifactorial problem involving a combination of vascular, neural, and humoral factors.(Shuja 2019; Gregory J. Landry 2019)

What are the subtypes of Raynaud's phenomenon and what is the underlying pathology?

- 1. Primary: Raynaud's disease-idiopathic form that is a benign process not associated with structural vascular change. Triggers include (cold, emotional stress, caffeine) resulting in digital smooth muscle contraction and temporary digital hypoperfusion.
- 2. Secondary: Fixed vascular obstruction to blood flow decreasing threshold for cold induced vasospasm or progress to tissue loss. Diseases associated include mixed connective tissue disease, SLE, and rheumatoid arthritis, and scleroderma (accounts for 80-90% of cases). In setting of lower digital blood pressure, symptomatic digital ischemia or tissue loss under low stress conditions. With cold/emotional stress, vasoconstrictive response of digital artery smooth muscle further causes arterial closure and resultant symptoms

What are diagnostic criteria for Raynaud's?

• Clinical (Progression of ischemia with white -> blue -> red finger discoloration. Episodes can be self-limited and may last from less than a minute, but generally not longer than 10-20 minutes

- Qualitative testing for severity of cold sensitivity in Raynaud's syndrome can be useful. Most basic test is cold sensitivity and recovery after ice water immersion. >10 minutes return to baseline pressure concerning for Raynaud's
- Segmental pressures with finger systolic blood pressure can differentiate purely vasospastic vs occlusive disease. Difference of more than 15 mm Hg between fingers or absolute finger pressure <70 mm Hg may indicate occlusive disease
- Serologic evaluation (ANA/RF)

What are appropriate treatments for Raynaud's phenomenon?

- 1. Medical-cold/tobacco avoidance lead to improvement in around half of patients. Calcium channel blocker (nifedipine ER 30mg qday) and losartan (50mg BID) have been shown to be most effective.(Gregory J. Landry 2013; Wigley and Flavahan 2016) Other drugs include alpha blocker, sildenafil, fluoxetine (SSRI), reserpine, cilostazol, captopril.
 - OUTCOMES ARE POOR IN PATIENTS WITH ARTERIAL OBSTRUCTION. If there is an asymmetric vascular examination, then further non-invasive vascular imaging is needed.
- 2. Surgical-thoracic sympathectomy (used for treatment of digital artery vasospasm/digital ischemic ulceration). For vasospasm, thoracic sympathectomy is initially successful, but symptoms return generally within 3-6 months.
- 3. Immunosuppression/immunomodulation for connective tissue disorders associated with secondary Raynaud phenomenon

See Landry's Review in JVS for a good treatment algorithm. (Gregory J. Landry 2013)

3.3.3 Ergotism

What is Ergotism?

Ergot is a parasitic fungal disease that has a particular prevalence for infecting rye plants and ergot alkaloids have been linked to epidemic poisonings that manifested as ergotism from consumption of rye. **Modern day is rare.**(Stanley, Veith, and Wakefield 2014)

What causes Ergotism and how do patients present?

- Ergotamine is chemically like endogenous catecholamines/indolamines and when applied clinically, it behaves as an agonist to alpha-adrenergic, sertoninergic, and dopaminergic receptors. Can be seen with ergot alkaloid migraine medications. Despite limited bioavailability, vasocontrictive effects have been reported to last for 24 hours or longer
- Gangrenous-mild limb pain followed by burning pain/shooting and

• Convulsive-heaviness in limbs and head associated with diarrhea. Could result in tonicclonic spasms

How can you diagnose Ergotism and what is the process for treating this disease?

Upper extremity ischemia (i.e. digital ulceration) in the setting of ergot alkaloid use (typically for migraines). The treatment includes:

- Volume expansion and IV heparin as anticoagulation
- IV infusion of nitroprusside, nitroglycerin, iloprost or combination
- Infusion of Ca 2+ channel blockers
- Surgical: for thrombosis, consider thrombolysis. Avoid or delay amputations as much as possible to maximize amount of tissue recovery.

3.3.4 Buerger's Disease

How is Buerger's disease categorized?

Non-atherosclerotic, segmental, inflammatory disease of small/medium sized arteries in distal extremities (upper and lower) of tobacco users distinct from either atherosclerosis of immune arteritis(Jack L Cronenwett, Alik Farber, and Erica L. Mitchell 2020; Le Joncour et al. 2018)

What clinical criteria can help diagnose Buerger's?

Diagnosis of Buerger's disease requires 5 criteria(Akar 2019)

- 1. Smoking history
- 2. Onset before age 45 or 50
- 3. Distal extremity ischemia with infrapopliteal arterial occlusions and often upper limb involvement
- 4. Classic angiographic findings of segmental occlusions of distal arteries with corkscrew collaterals
- 5. Absence of atherosclerotic risk factors (besides smoking, of course), autoimmune disease, diabetes or proximal embolic source.

Although rarely are biopsy sent, pathology will show panvasculitis within small and mediumsized arteries, highly cellular intraluminal thrombus (contains CD68, Cd3, Cd8 inflammatory cells), preserved elastic lamina structure, unlike atherosclerotic lesions.(Akar 2019; M. Kobayashi et al. 1999)

What is important about diagnosing Buerger's

- Typically a diagnosis of exclusion
- Must rule out proximal embolic source, trauma, local lesions (e.g. pop entrapment or cystic adventitial disease), autoimmune disease, hypercoagulable status, atherosclerosis

What physical exam and non-invasive/invasive imaging findings of Buerger's?

- Distal, but not proximal arterial disease (palpable brachial/popliteal but absent/reduced at ankle or wrist)
- Digital brachial index (DBI) < 0.6 and flat/reduced digital waveforms
- CTA/MRA/DSA or duplex characteristic findings of serpiginous/corkscrew collaterals, occlusion of distal calf/pedal arteries, and normal proximal arteries.(K. Busch 2011; Fujii et al. 2011)

What is the mainstay treatment in Buerger's disease?

- 1. Smoking cessation! Only treatment to improve symptoms and reduce amputation risk if achieved before onset of gangrene or tissue loss. May even require inpatient admission for intensive smoking cessation therapy.(Hooten, Bruns, and Hays 1998) It is important to remember following treatments will likely fail without smoking cessation.(J. W. Olin 2000)
- 2. If smoking cessation does not improve, medical management with antiplatelet agents, immunomodulators, vasodilators (Calcium channel blockers and cilostazol(Dean and Satiani 2001)), anticoagulants(Kubota et al. 1997), and IPC.(Montori et al. 2002)
- 3. Endovascular-distal small vessel intervention
- 4. Surgical-upper extremity autogenous vein bypass-limited success due to poor outflow
- 5. Sometimes can consider upper extremity sympathectomy, but unproven benefit
- 6. Amputation-reported in 30-40% who are followed longer than 5 years (Jeffrey W. Olin 2018)

3.3.5 Large Artery Vasculitis

What are common characteristics for patients who are suspected to have a large vessel vasculitis? (Shanmugam 2019; Weyand and Goronzy 2003)

- Affect aorta and major branches
- Present with non-specific heterogenous symptoms making the diagnosis challenging. Most commonly, they present with systemic or constitutional symptoms (fatigue, fever, weight loss, arthralgias)

• Frequently, diagnosis made with presence of constitutional symptoms, elevated inflammatory markers (ESR/CRP), and dedicated imaging (MRA, CTA, DUS, or PET)

How can you differentiate Takayasu arteritis vs giant cell arteritis?

3.3.5.1 Takayasu arteritis

- 1. Involves a and major arch branches (Ehlert and Abularrage 2019a)
- 2. Young patients (20-40 years) and female in 80-90% of cases, Asian populations
- 3. American College of Rheumatology (ACR) Criteria(Maz et al. 2021)
 - 1. Onset <40 years
 - 2. Claudication of an upper extremity
 - 3. Decreased brachial pulse
 - 4. > 10 mmHg SBP between arms
 - 5. Bruit over subclavian arteries or aorta
 - 6. Arteriographic evidence of narrowing/occlusion in aorta/primary branches/or large upper/lower extremity arteries

3.3.5.2 Giant cell arteritis

- 1. Aorta and main branches, but predilection for carotid artery branches(Bongartz and Matteson 2006)
- 2. Diagnosis:
 - 1. Age at disease onset > 50 years
 - 2. New headache
 - 3. Temporal artery abnormality
 - 4. Elevated ESR (>50) can be normal in up to a quarter of cases.
 - 5. Abnormal (temporal) artery biopsy (gold standard test)
- 3. Other symptoms include jaw pain with mastication or visual changes
- 4. Associated with Polymyalgia rheumatic, characterized by morning stiffness in shoulders/hips occurring in 40-50% of patients.(Weyand and Goronzy 2014)
- 5. Arteriography/MRA/CTA/PET may be used to assess large vessel involvement classic finding is smooth, tapering stenosis.

3.3.5.3 Management

What is the medical treatment for GCA and when do you consider surgical treatment?

Medical-steroid therapy - In as many as 50% of patients who have a large vessel vasculitis refractory to glucocorticoid therapy alone, patients will trial immunomodulators or cytotoxic trugs (i.e. methotrexate, azathioprine, mycophenolate, tocilizumab, or leflunomide)

Intervention - Avoid intervention in the acute setting. Once in remission, treatment of symptomatic arterial lesions or those at an eurysm size threshold should be considered and as many as 50-70% with large vessel vasculitis will require intervention.

- Endovascular angioplasty/stent/stent graft for large vessel vasculitis have all been described, however higher restenosis in endovascular compared to open treatment
- Open Surgery (gold standard) lesions are long, fibrotic and therefore less amenable to endovascular treatment. Bypass grafts from aorta-CCA are the most common (CEA should be avoid due to pathology involved)
 - Upper extremity bypass with autogenous vein to the brachial artery
 - Aortic aneurysms should be managed with open surgery(Janssen et al. 2008)
 - Mid aortic syndrome likely needs open repair from uninvolved aorta-usually thoracic aorta-to bifurcation bypass with jump graft to involved visceral vessels.(Ehlert and Abularrage 2019a)

3.3.5.4 Surveillance

How should patients be monitored with active large artery vasculitis?

- Lab data tracked at least monthly for 6 months with close follow-up to ensure appropriate response to medical treatment and enable physicians to assess for adverse effects of medical treatment
- Repeat tests after remission reached and imaging choice to evaluate large vessels (DUS/CTA/MRA)

3.3.6 Other Vasculitis syndromes

Polyarteritis Nodosum - Focal necrotizing lesions primarily affecting medium-sized muscular arteries, peak incidence in 40s. Multiple saccular aneurysms secondary to inflammatory destruction of vessel media.(Shanmugam 2019; Weyand and Goronzy 2003)

Behçet Disease - recurrent oral pathos ulcers, genital ulcers and uveitis.(Shanmugam 2019; Weyand and Goronzy 2003)

3.4 Aneurysmal Disease

3.4.1 Subclavian Artery Aneurysms

How are subclavian aneurysms caused and how can they present? (Baig and Timaran 2019)

Etiology

- Degenerative atherosclerotic or due to aberrant right subclavian with degenerative changes in proximal subclavian known as "Kommerell diverticulum" (Tanaka, Milner, and Ota 2015)
- Traumatic blunt, penetrating, iatrogenic with attempted catheter placement
- Radiation induced, sometimes seen after radiation for breast cancer.(Mohan et al. 2016)
- Thoracic outlet obstruction no need to investigate for aTOS if there is another plausible cause.

Presentation

- Rare, comprises 1% of peripheral aneurysms.
- Exam-pulsatile supraclavicular mass or bruit, absent/diminished pulses, signs of microembolization ("blue finger")
- Most discovered incidentally, however referred chest, neck, shoulder pain, upper extremity ischemia due to thrombosis or distal embolism, brachial plexus compression, hoarseness from right recurrent laryngeal nerve compression
- Dysphagia from esophageal compression in aberrant right subclavian artery.(Tanaka, Milner, and Ota 2015)
- Rarely rupture

Evaluation

What are diagnostic studies and treatment modalities for subclavian aneurysms?

• CXR-mediastinal mass may suggest neoplasm

• MRA/CTA important to delineate extent of an eurysm and proximity to ipsilateral vertebral artery

Management

- Open Repair-resection/endoaneurysmorrhaphy with end to end (small aneurysms) or interposition prosthetic graft(Vierhout et al. 2010)
 - Proximal-median sternotomy with supraclavicular fossa extension for adequate proximal control for right side, however supraclavicular with left anterolateral thoracotomy for left subclavian aneurysm
 - Mid-Distal-supraclavicular/infraclavicular generally adequate for control where again resection of the clavicle may be needed
- Endovascular Repair-transbrachial/transfemoral approach with covered stent(Maskanakis et al. 2018)
 - Must consider vertebral artery origin. Can cover vertebral artery if contralateral vertebral artery is patent and of adequate size, however posterior circulation stroke may occur when the contralateral vertebral artery is highly stenotic, hypoplastic or occluded.
- Hybrid Repair-embolization/coils of proximal subclavian artery combined with subclavian transposition or carotid-subclavian bypass
- For aberrant subclavian artery aneurysm, resection or exclusion of the aneurysmal artery with vascular reconstruction of the subclavian artery is recommended. Especially in the setting of dysphagia lusoria, subclavian artery reconstructed by interposition graft where proximal anastomosis is on ascending aorta. Alternatively, left posterolateral thoracotomy for proximal aneurysm resection and right supraclavicular incision for reconstruction of subclavian artery by end to side to the right CCA has been reported.(Tanaka, Milner, and Ota 2015)

3.4.2 Axillary Artery Aneurysms

How are axillary aneurysms caused and how can they present?

Etiology

- Blunt/penetrating trauma
- Congenital (infrequently reported)
- Post-traumatic axillary aneurysms (repeated abduction/external rotation downward toward humeral head in baseball pitchers)

Presentation

• Exam-pulsatile supraclavicular mass or bruit, absent/diminished pulses, signs of microembolization ("blue finger")

What are diagnostic studies and treatment modalities for axillary aneurysms?

Evaluation

- Ultrasound
- CTA/MRA of upper extremity

Management

- Open Repair-resection with interposition vein grafting or prosthetic if inadequate vein is present.
- Endovascular repair-covered stent graft can be placed with occasional embolization with micro coils to isolate sac and prevent retrograde endoleaks

3.4.3 Brachial Artery Aneurysms

How are brachial aneurysms caused and how can they present?

Etiology

- False aneurysms secondary to repetitive trauma
- Iatrogenic complications Seen in 1-3% of brachial artery access increased risk in older age, female patients, and larger sheaths.(Treitl et al. 2015) More on appropriate access technique can be found in Chapter 22
- IV drug abuse infected pseudoaneurysms in antecubital fossa.
- Connective tissue disorders (ex. type IV Ehlers Danlos)

Presentation

- Exam: pulsatile mass
- Local pain or symptoms of median nerve compression
- Hand/digital ischemia from thrombosis/distal embolization

What are diagnostic studies and treatment modalities for brachial aneurysms?

Evaluation

• Duplex Ultrasound

• CTA/MRA of upper extremity may be needed to delineate extent of aneurysm

Management

- Open Repair (preferred)-resection with patch or interposition vein grafting
- Endovascular repair-rare and generally in a traumatic setting
- Iatrogenic injuries-due to access and nonoperative treatment for small/asymptomatic pseudoaneurysms that are likely to thrombose spontaneously. Direct suture repair with evacuation of hematoma is possible. Thrombin injection is less favorable due to location and short neck.
- Neurological symptoms from median nerve compression require an urgent repair, open surgery best for decompression.

3.4.4 Radial Artery Pseudoaneurysm

How do you manage a patient who presents with a radial artery pseudoaneurysm after a coronary angiogram on year ago?

Small pseudoaneurysms <3cm have a high rate of spontaneous thrombosis. However, larger psuedoaneurysms or those with symptoms require treatment. Often best managed with open excision and primary repair or interposition graft.(Tosti et al. 2017)

Can also be seen as a result of trauma.(Bagir, Sayit, and Tanrivermis Sayit 2017)

3.5 Occupational Vascular Disease

There are some occupational vascular disorders than contribute to vascular disease in the upper extremity. Hand arm vibration syndrome and hypothenar hammer are of particular importance.(Eskandari and Morgan 2020)

3.5.1 Hand-Arm Vibration Syndrome

Etiology

- Vibrating handheld machines (i.e. pneumatic hammers and drills, grinders, and chain saws)
- Linear relationship between exposure over years and onset of this syndrome
- Exact mechanism unknown, but thought that endothelial damage with sympathetic hyperactivity -> finger blanching attack

Presentation

- Various stages seen where early results in slight tingling/numbness and lateral, the tips of one or more fingers experience attacks of blanching that is usually precipitated by cold
- Blanching typically lasts 1 hour and terminates with reactive hyperemia, but prolonged exposure can cause bluish black cyanosis of fingers

Evaluation

- Detailed history with use of vibrating tools/symptoms of Raynaud phenomenon
- Objectively: cold induced ischemia with recording time until digital temperature recovers
- Digital occlusion with noninvasive digit pressures or duplex scanning

Management

- Avoidance of vibratory tools
- Nifedipine (Ca2+ channel blocker) in advanced cases
- IV prostanoid (i.e. prostacyclin) for digital gangrene
- Surgery-cervical sympathetectomy or digital sympathectomy rarely needed

3.5.2 Hypothenar hammer syndrome

Etiology

- Repetitive use of palm of hand in activities that involve pushing, pounding, twisting particularly to the ulnar artery as it exits guyots canal and crosses the hook of the hamate.(Ferris et al. 2000)
- Name comes from reports of mechanics, factory workers, carpenters or laborers who habitually use there hands as a hammer are ad risk for disease
- Repetitive trauma leads to thrombotic occlusion, aneurysm formation or both

Presentation

- Asymmetrical distribution involving dominant upper extremity where cyanosis and pallor can occur and digits affected are ulnar distribution in nature
- Cool/mottled digits or severe cases with ischemic ulcers

Evaluation

• Duplex ultrasound

- CTA or MRA
- Arteriography (gold standard) with corkscrew pattern typically in affected vessels

Management

- Conservative-smoking cessation/hand protection/cold avoidance.(Carr et al. 2019)
- Medical-calcium channel blockers/antiplatelet
- Surgical (severe digital ischemia/aneurysm)-ligation if adequate collateral or interposition vein graft has good long term patency.

3.5.3 Environmental Exposures

Exposure to what environmental agents can result in upper extremity ischemia?

Acrosteolysis

- Exposure to polyvinyl chloride can result in ischemic hand symptoms similar to those of Raynaud syndrome
- Angiography-damage to digital arteries with multiple stenosis/occlusions or hyper vascularity adjacent to areas of bone resorption
- Treatment-supportive

Electrical burns

- $\bullet~<\!\!1000$ V cause injuries limited to immediate skin/soft tissue, however >1000 V cause damage from entry to exit point
- Results in arterial necrosis with thrombus or bleeding and gangrene of digits develop
- Initially can be occlusion/thrombosis or spasm, however later damage can cause an eurysmal degeneration
- Treatment-dependent on soft tissue/bone injuries as well. Can have reconstruction with free flap due to local vascular damage or occlusion of major artery requiring bypass grafting

Extreme thermal injuries

- Workers at risk with chronic exposure to cold (slaughterhouse, canning factory, and fisheries)
- Raynaud syndrome symptoms due to vasomotor disturbances in the hands when exposed to extreme chronic thermal trauma
- Treatment-Supportive

3.6 Sports Medicine

How can athletes specifically be affected by upper extremity ischemia?

Athletes who engage in strenuous or exaggerated hand/shoulder activity may be susceptible to upper extremity ischemia from arterial injury manifested by Raynaud syndrome, symptoms of sudden arterial occlusion or digital embolization.

3.6.1 Quadrilateral space syndrome

Anatomy

- Bordered by teres minor superiorly, humeral shaft laterally, and teres major inferiorly, and long head of triceps muscle medially
- Posterior humeral circumflex artery and axillary nerve in space

Etiology

- Compression of posterior humeral circumflex occurs with abduction/external rotation
- Typically seen with chronic overhand motion athletes (pitchers/volleyball players)
- Vascular-repetitive mechanical trauma to posterior circumflex humeral artery
- Neurogenic-fixed structural impaction of quadrilateral space by fibrous bands or spaceoccupying lesions

Presentation

• Muscle atrophy, paresthesias, poorly localized shoulder pain and pain in quadrilateral space

Management

- Medical: Oral anti-inflammatory medications, PT, limitation of activities
- Surgery: decompression with neurolysis/excision of fibrous bands or other space occupying lesions

3.6.2 Humeral head compression of axillary artery

Anatomy

• 3rd portion of axillary artery compressed by head of humerus

Etiology

• Arm is abducted and externally rotated with downward compression of humeral head to axillary artery

Presentation

• Arm fatigue, loss of pitch velocity, finger numbress, Raynaud, cutaneous embolization

Evaluation

- Provocative maneuvers with impedance of flow through axillary artery on ultrasonography
- Arteriography with rest and provocative position

Management

- Supportive with avoidance of throwing motion
- Surgical-saphenous vein patch for no improvement or structural injury may require resection with saphenous vein bypass anatomically or extra-anatomic tunneling above pectoralis minor muscle

3.6.3 Thoracic Outlet Syndrome

Thoracic outlet syndrome and cervical rib abnormalities are covered extensively in another chapter. For more information see Chapter 4

3.7 Vascular Trauma-Upper Extremity

This is discussed in detail in Chapter 19, so we will go over some important specifics for upper extremity vascular injury. (David S. Kauvar and Kraiss 2020)

3.7.1 Subclavian Artery Trauma

What is the management strategy for an iatrogenic placement of a subclavian artery catheter?

Critically ill patients can be treated effectively with the use of a closure device or covered stent placement. (Yoon et al. 2015; V. Tran et al. 2009; Cohen et al. 2014) Cut down in the mid subclavian artery is very difficult to perform and results in increased blood loss and worse outcomes.

Trauma to the proximal subclavian should be accessed on R through a median sternotomy or trapdoor incision. On the left, the proximal subclavian is accessed best through an anterolateral thoracotomy.

3.7.2 Axillary Artery Trauma

What is the mechanism and management of upper extremity axillary artery trauma?

Etiology

• Predominantly in penetrating trauma with equal incidence in proximal/middle/distal divisions and brachial plexus injury in >1/3rd of arterial injury

Evaluation

- Physical exam with deficiencies in upper extremity pulses/ischemic changes, but may not be present given collateral flow from axillary artery to upper extremity
- High index of suspicion with location of injury proximity to course of axillary artery
- Upper extremity Doppler or CTA if patient is stable for diagnosis

Management

- Primary repair or treated with interposition graft
- If hemodynamically stable, can consider covered stent based on location to thoracic outlet via femoral/brachial approach. Discussed in more detail in Section 9.3

3.7.3 Brachial Artery Trauma

What is the mechanism and management of upper extremity brachial artery trauma?

Etiology

- Frequently associated with humerus fractures/elbow dislocation. Seen often in children.
- Penetrating trauma

Evaluation

- Pulse deficit in majority (>75% of cases)
- Upper extremity Doppler of CTA

Management

- Given course, can be extensively mobilized and repaired in end-to-end fashion in 50% of cases. Otherwise, treatment with an interposition graft
- Need to differentiate injury from spasm.

3.7.4 Radial/ulnar artery trauma

What is the mechanism and management of upper extremity radial/ulnar artery trauma?

Etiology

• Associated with significant soft tissue pattern

Evaluation

- Pulse deficit in >80% of patients
- Doppler based Allen test-confirm radial/ulnar contribution to palmar arch

Management

- If Allen test reveals a patent palmar arch, the injured artery can be ligated
- If palmar arch is not patent in the absence of contribution of the injured artery, it should be repaired
- If both are damaged, preference to ulnar artery as dominant contribution to hand
- Generally, repair can be done in an end to end fashion given mobility of the vessel

3.8 Upper Extremity Lymphedema

3.8.1 Etiology

Often due to breast cancer therapy and axillary lymph node dissection. (Morrell et al. 2005)

3.8.2 Management

Initial treatment includes decongestive therapy, such as daily massage for 2-4 weeks. Once the girth and symptoms stabilize, then can transition to long term management with compression sleeve.(Mondry, Riffenburgh, and Johnstone 2004)

Severe cases my require lymph node transfer.(Warren et al. 2007)

Lymphangiosarcoma in the setting of lymphedema (Stewart-Treves syndrome) without distant disease is best managed with wide local excision and adjuvant chemoradiation. Prognosis is poor.(Sharma and Schwartz 2012)

4 Thoracic Outlet

Author: Nedal Katib and Matt Smith

4.1 Overview

4.1.1 Demographics

Thoracic Outlet Syndrome (TOS) is a constellation of signs and symptoms relating to the compression of the neurovascular structures that occurs as these structures travel between the thoracic aperture and the upper limb.

Types: Neurogenic, Venous and Arterial

- vTOS 2-3%
- aTOS 1%
- nTOS >95% (Humphries and Freischlag 2019b)

4.1.1.1 Anatomy

Understanding the anatomy of what is collectively referred to as the thoracic outlet is the best way to thoroughly appreciate this topic.

Anatomy from anterior to posterior

- 1. Subclavian vein
- 2. Phrenic nerve
- 3. Anterior scalene muscle attachment to the first rib
- 4. The subclavian artery
- 5. The brachial plexus
- 6. The middle scalene muscle

Three spaces where the neurovascular structures are at risk of compression

- 1. Interscalene Triangle Appreciating the attachments of the Anterior and Middle Scalene Muscles on the first rib becomes important in the diagnosis of the various types and also the ultimate surgical management of the compression.
 - Anterior Scalene Attachments: Anterior Tubercles of the four 'typical' cervical vertebrae (3-6) AND the scalene tubercle on the upper surface of the first rib.
 - Phrenic nerve runs along anterior scalene muscle and injury can cause ipsilateral diaphragm paralysis.
 - Middle Scalene Attachments: The posterior tubercles and intertubercular lamellae of all the cervical vertebrae AND the Quadrangular area between the neck and subclavian groove of the first rib. (McMinn 2019)
 - Long thoracic nerve runs along middle scalene muscle and injury can cause winged scapula.
 - The First Rib The broadest and flattest of the ribs and is an 'Atypical Rib' because it only one articular facet.
 - The upper surface of the first rib has the scalene and quadrangular tubercles for attachments of the anterior and middle scalene muscles respectively. There are also three grooves for the Subclavian Vein, artery and the Lower Trunk of the Brachial Plexus.
 - The Inferior Surface is smooth and inferior and medially has an attachment for the suprapleural membrane, Sibson's fascia AKA scalenus minimus muscle, which is tethered to the C7 vertebrae.
- 2. Costoclavicular Passage This is the passage of the subclavian vein largely as it emerges through the tight space created by the clavicle, the subclavius muscle and the costoclavicular ligament and also more posteriorly this can also compress the artery and nerves as the space can also be narrow in relation with the scapula and subscapularis. (Gary G Wind and R. James Valentine 2013)
 - Subclavius Muscle Attached to the costochondral junction of the first rib and is inserted into the subclavian groove on the inferior surface of the clavicle. (McMinn 2019)
- 3. Subcoracoid Space This space is best appreciated by intimate knowledge of three things(Gary G Wind and R. James Valentine 2013)
 - 1. The Coracoid Process
 - Arising from the Scapula as a 'process', this broad-based bony landmark offers attachment to muscles and ligaments.
• The relevant attachments being the pectoralis minor muscle occupying the medial border for about 2cm behind its tip. The tip itself having a medial and lateral facet for the short head of biceps and the coracobrachialis muscles respectively.

2. Pectoralis Minor Muscle:

• Attached to the bone of the third, fourth and fifth ribs AND the medial border of the coracoid process.

3. Clavipectoral Fascia:

• A sheet of fascial membrane that fills the space between the clavicle and pectoralis minor splitting and encompassing the subclavius muscle. Its superior portion is what can be thickened and become a tight band referred to as the costocoracoid ligament.

Anomalous Anatomy

- **Phrenic Nerve Anomaly** The phrenic nerve normally runs anterior to the Subclavian Vein. A rare anomaly is the nerve compressing the vein anteriorly and in very rare circumstances due to the timing of development can run through the vein itself.
- **Cervical Rib** Anomalous anatomy can also cause TOS especially when patients have a Cervical Rib and anomalous first ribs or a congenital band attaching to the first rib.
 - Incidence of anomalous first ribs and cervical ribs is 0.76% and 0.75% respectively.
 - Incidence of bands are as high as 63% in the general population. (Humphries and Freischlag 2019b)

4.1.1.2 Etiology

nTOS

- 1. Scalene Triangle compression most common cause of brachial plexus and neurogenic TOS
- 2. Cervical Rib and Anomalous First Rib

aTOS

- 1. Cervical Rib and Anomalous First Rib most common
- 2. Scalene Triangle compression

vTOS

- 1. Costoclavicular Passage most common
- 2. Subcoracoid Space

4.1.1.3 Presentation

- Identify symptoms and thoroughly interrogate timing
- Exclude history of trauma clavicle fracture and malunion
- Associated symptoms like headache, visual disturbance, neurology in the upper limb
 - Exclude Carpal Tunnel and Antecubital Tunnel Syndromes if symptoms are isolated to the arm or forearm or hand
- Patients with vTOS may present acutely and have acute or sub-acute upper limb DVT
- Patients with aTOS need to be investigated and assessed urgently given risk of ischemia.

4.1.2 Evaluation

4.1.2.1 Clinical Examination

Provocative maneuvers are largely used for nTOS. While these are described and mentioned in most texts their utility largely is beyond the scope of a vascular surgeon's assessment and diagnosis of nTOS.

Adson Test

- Extended abducted and externally rotated arm palpate radial pulse
- Rotate and laterally flex the neck to the ipsilateral side while inhaling deeply.
- A positive test results in reduction or complete obliteration of radial pulse

Roos Test / EAST test

- Patient seated and both arms abducted 90 degrees and externally rotated and elbows flexed at 90 degrees.
- Open and close hands for 3 minutes or until pain or paraesthesia sets in.

Elveys Test

- Abduct both arms to 90 degrees with elbows extended and dorsiflex both wrists.
- If pain is elicited as wrists dorsiflexed then test is positive.

• A further maneuver is then performed, laterally flex the head on each side, if pain is elicited on the contralateral side to which the head is flexed then test is positive. (Humphries and Freischlag 2019b)

4.1.2.2 Imaging

Vascular Lab

- Digital Brachial Index (DBI)
- Arterial Duplex can identify subclavian artery aneurysms.
 - Can also be performed with provocative maneveurs
- Venous Duplex

Axial Imaging

- CTV commonly performed in acute upper limb DVT and suspicion of vTOS
- CTA for the evaluation of aTOS and excluding other causes of embolisation
- MRI for further evaluation of the anatomy and related neurovascular compression

Angiography

• In the setting of vTOS, some advocate for upper extremity venogram with provocative maneuvers to evaluate upper arm collaterals and compression

Other studies

- Electromyography and nerve conduction studies for nTOS performed after evaluation with neurologist and physiotherapist.
- In the setting of nTOS, scalene block can be diagnostic if the patient has temporary relief.

4.1.3 Management

After the appropriate evaluation, if decompression of thoracic outlet is deemed appropriate, there are multiple approaches to the first rib resections.

Transaxillary

- Advantages:
 - Cosmetically more appealing as it has a limited hidden scar
 - Allows for removal of a significant portion of the rib anteriorly and posteriorly
- Disadvantages:
 - Difficult to visualize the anatomy, dependent on good assistance
 - Risk of injury to T1 nerve root, phrenic nerve, long thoracic, brachial plexus, subclavian vein and arterial with limited exposure to repair
 - Not able to approach cervical ribs, scalene triangle or patch vein.

Take a Look

Check out Dr. Freischlag's operative video where she reviews her approach to a transaxillary rib resection.

Supraclavicular

- Advantages
 - Good for scalene triangle access and debulking and cervical rib resection
 - Required for aTOS if arterial reconstruction necessary
- Disadvantages
 - Unable to decompress venous compression or visualize vein adequately
 - Cosmetically less appealing

💡 Take a Look

Check out our How I Do It Presentation where Dr Westley Ohman and Dr. Robert Thompson review their approach to a supraclavicular rib resection.

Infraclavicular(Siracuse et al. 2015)

• Advantages

- Good access for venous decompression
- Allows for excision of subclavius muscle and costoclavicular ligament
- Disadvantages
 - Unable to expose subclavian artery or decompress brachial plexus.
 - Difficult to access most posterior aspect of rib
 - Cosmetically less appealing

Paraclavicular

- Advantages
 - Useful if mixed etiology TOS to adequately decompression all neurovascular structures
- Disadvantages
 - Requires two incisions one above and below the clavicle

4.1.3.1 Complications

- Post operative patients with hemodynamic instability and ipsilateral effusion on xray should go back to OR for exploration and hemorrhage control. (Rinehardt, Scarborough, and Bennett 2017)
- Chyle leak often managed with adequate drainage and medium chain fatty acid diet. Persistent leak should undergo wound exploration and ligation of thoracic duct (most often encountered on the left) or VATS thoracic duct ligation.(Delaney et al. 2017) IR embolization has also been shown to be effective.(Cope, Salem, and Kaiser 1999)

4.2 Venous (vTOS)

4.2.1 Demographics

- Incidence: 2/100,000 persons
- Age: 18 years to 30 years (Karl A. Illig and Doyle 2010)
- M>F
- Paget Schroetter Syndrome

- First defined by Hughes in 1949 in reference to Sir James Paget who in a hundred years earlier defined acute arm swelling and pain as possibly related to vasospasm and then von Schroetter who in 1884 attributed to the presentation to subclavian and axillary vein thrombosis. (Humphries and Freischlag 2019a)
- Now vTOS and Paget Schroetter Syndrome are synonymous.
- Paget Schroetter Syndrome accounts for 10-20% of all upper extremity deep vein thrombosis. (Sekhar 2018)

4.2.1.1 Presentation

- Upper Limb edema, pain and cyanosis. Edema affects the shoulder, arm and hand and characteristically non pitting.
- Collateral vein dilatation over the shoulder, neck and anterior chest wall to accommodate for the increased venous hypertension. (Humphries and Freischlag 2019a)
- Pain on exertion of the upper limb described as stabbing, aching or tightness.
- The reported incidence of PE following upper limb DVT is $<\!\!12\%$. (Humphries and Freischlag 2019a)

4.2.2 Evaluation

Upper extremity venogram is a very useful modality for diagnosis. Compression likely will not be seen in normal orientation, however can be demonstrated with provocative maneuvers including 90 degree arm abduction and external rotation. (Moriarty et al. 2015)

- A differential diagnosis for Upper Limb DVT
 - vTOS
 - Congenital Phrenic Nerve anomaly
 - History of fracture, clavicular fracture and malunion
 - Repetitive arm provocative maneuvers, check occupation and history of bodybuilding
 - Pectoralis Minor Hypertrophy.
- Exclude Pulmonary Embolism
- Exclude Venous Gangrene and Phlegmasia of the upper limb

4.2.3 Management

Limited evidence due to lack of RCT's. Majority of evidence based on retrospective studies.

- Prevent immediate risk
- Return patient to unrestricted use of the affected extremity
- Prevent recurrence of thrombosis without the need of long-term anticoagulation
- Prevent long term Post Phlebitic Limb Syndrome

4.2.3.1 Medical Management

As per ACCP Guidelines: Initial management is anticoagulation regardless of etiology. (Kearon, Akl, Ornelas, Blaivas, Jimenez, Bounameaux, Huisman, King, Morris, Sood, Stevens, et al. 2016)

• The limitations of anticoagulation alone are that the slow recanalization of the thrombus may lead to eventual valvular damage and intravenous scarring. (Sekhar 2018)

Catheter Directed Thrombolysis (CDT) has been considered superior to anticoagulation alone in minimizing valvular damage due to residual clot and carries a lower risk of intracranial hemorrhage.(Urschel and Patel 2008)

- Systemic Lysis non favored due to risk of intracranial hemorrhage. (Grunwald and Hofmann 2004)
- Optimal timing of CDT Within 14 days of onset of thrombosis. Excellent results have been reported following CDT if initiated before 14 days. (J. J. Wilson, Zahn, and Newman 1990)

Patient should be maintained in a compression sleeve until definitive decompression can be performed.

4.2.3.2 Surgery

After initial management patients are generally divided into two groups, unsuccessful or successful thrombolysis.

• Persistent stenosis or signs of extrinsic compression, on venography, has generally been perceived as a significant risk of recurrent thrombosis.

Surgery for vTOS remains to be mainly Rib Resection and decompression of the subclavian vein with or without venolysis and patch plasty either surgical or endovenous.

• Surgical treatment of severe resistant subclavian vein stenosis in the setting of vTOS is rib resection by paraclavicular approach and vein patch plasty. (Melby et al. 2008)

Venous occlusion in vTOS may be treated with jugular turn down or venous bypass to IJ of SVC if patients remain symptomatic. (Vemuri et al. 2016)

4.2.4 Summary

4.2.4.1 Controversy

There is a lack of consensus around the necessity of surgical rib resection, the timing and the requirement for vein patch plasty. Options post recanalization:

- Deferring surgical decompression for 1-3 months after thrombolysis to allow for healing of the venous endothelium and resolution of the acute inflammatory process. (Humphries and Freischlag 2019a)
- Decompression during the same admission, as the thrombolysis, with the main benefit being to reduce the risk of re-occlusion. (Humphries and Freischlag 2019a; Molina, Hunter, and Dietz 2007)
- Post decompression venography and treatment 2 weeks post rib resection may help to prevent recurrence and long term vein patency. (K. Z. Chang et al. 2012)

4.2.4.2 Landmark papers

- 1. Lugo J et al Acute Paget Schroetter syndrome: does the first rib routinely need to be removed after thrombolysis? Annals of Vascular Surgery 2015 (Lugo et al. 2015)
 - Systematic literature review analysis. Patients divided into three groups
 - 1. First Rib resection (FRR) n=448
 - 2. First Rib resection and endovenous venoplasty (FRR and PLASTY) n=68
 - 3. No further intervention after Thrombolysis n=168
 - Symptom relief after initial follow up more likely in FRR (95%) and FRR and PLASTY (93%) compared to no rib removed (54%) p < 0.0001
 - Results showed superior patency with FRR and PLASTY and FRR compared to anticoagulation alone.
 - Conclusion was that patients are more likely to experience greater long-term results with FRR compared to no FRR.

- 2. Sajid MS et al Upper limb vein thrombosis: a literature review to streamline the protocol for management. Acta Haematology 2007 (Sajid et al. 2007)
 - Comprehensive review identifying the key papers on this topic and allows for a clear view of the best management strategy.
- 3. Freischlag J The art of caring in the treatment of thoracic outlet syndrome. Diagnostics 2018.(Freischlag 2018)
 - Review of key publications guiding her Dr. Freischlag's care of these patients.

💡 Take a Listen

Check our interview with Dr Julie Freischlag where she discusses her career and management of patients with thoracic outlet syndrome.

4.3 Arterial (aTOS)

4.3.1 Presentation

Most common: Hand ischemia due to arterial compression or microembolization with subclavian artery aneurysm and pulsatile supraclavicular mass (Boll and Valentine 2019)

Less common: Exertional pain, unilateral Raynaud's Phenomena, retrograde embolisation and neurological symptoms

Differential Diagnosis

- Trauma
- Primary and Secondary Raynaud's Phenomena
- Small Vessel Vasculitis
- Connective Tissue Disorders
- Thromboangiitis Obliterans. For more see Section 3.3.4
- Arterial Embolization Aortic or Central Source
- Radiation Arteritis
- Atherosclerotic / Dissection causes

The different anatomical abnormalities causing aTOS (Boll and Valentine 2019)

• Cervical Rib (60%)

- Anomalous First Rib (18%)
- Fibrocartilaginous band (15%)
- Clavicular Fracture (6%)
- Enlarged C7 transverse process (1%)

4.3.2 Evaluation

Clinical Examination

- Audible Bruit / Palpable thrill over the supraclavicular fossa
- Pulsatile mass
- Distal ischemic lesions in the distal hand Splinter hemorrhages
- Positive Adson Test

Most useful studies are pulse volume recordings (PVR) and duplex to identify an eurysm or sites of embolization. Stress test is not reliable for diagnosis. (Vemuri et al. 2017; E. Criado, Berguer, and Greenfield 2010)

4.3.3 Management

Symptomatic patients are generally indicated for treatment. Unlike asymptomatic patients in whom it may be appropriate to manage conservatively. (Boll and Valentine 2019) Symptoms are classified according to the Scher staging system.

- Scher Staging of aTOS(Scher et al. 1984)
 - Stage 0: Asymptomatic
 - Stage 1: Stenosis of Subclavian Artery with minor post stenotic dilatation with no intimal disruption
 - Stage 2: Subclavian artery aneurysm with intimal damage and mural thrombus
 - Stage 3: Distal embolisation from subclavian artery disease

Supraclavicular rib resection is the most suitable for adequate arterial reconstruction. Transaxillary has been argued to offer more complete rib resection however arterial repair is not possible in this approach.

Subclavian artery repair is necessary in Scher Stages 2 and 3 and in some cases 1. Arterial repair with conduit either GSV, Femoral Vein or prosthetic have been described. Ringed PTFE offers good patency and resistance to kinking in this functional anatomical location.

4.4 Neurogenic (nTOS)

4.4.1 Demographics

Neurogenic TOS is largely a clinical diagnosis with symptoms and signs pertaining to nerve compression most commonly the lower trunk of the brachial plexus.

- $F{>}M-70\%$ Female
- Ages 20-40
- Occupational Exposure
- Trauma history often associated with recent history of neck trauma.

4.4.1.1 Presentation

- Symptoms (Sadeghi-Azandaryani et al. 2009; Sanders, Hammond, and Rao 2007)
 - Paraesthesia (98%)
 - Trapezius pain (92%)
 - Neck, shoulder or arm pain (88%) usually involves nerve roots so isolated hand symptoms is rare.
 - Supraclavicular pain with or without occipital headache (76%)
 - Chest pain (72%)
 - Weakness
 - Swelling
- Positional Effects
 - Reproducible exacerbation of symptoms
 - Lying supine with arms overhead
 - Overhead activities occupational or recreational
- Weakness and Muscle Atrophy
 - Hypothenar atrophy
 - Drop-off in athletic performance
 - Inability to carry out activities of daily living

4.4.2 Management

As the patient has already seen multiple specialists and physiotherapists, the role of the vascular surgeon with nTOS is often limited but should focus on:

- Exclude other causes
- Confirm diagnosis Neurophysiologic Tests (EMG and NCS)
- Seek alternate opinion
- Trial of Physiotherapy and non-operative management patients should be evaluated and undergo a 6 week course of physical therapy. This physical therapy focuses on scalene and pectoralis stretching improving mobility of the shoulder and strengthening the arm. Many improve with physical therapy. (Balderman et al. 2019)
- Anterior scalene lidocaine block may provide temporary symptom relief (~7 days) and may help identify those patients most likely to benefit from surgical decompression. (Salhan et al. 2016; Lum, Brooke, Likes, et al. 2012)
- Botulinum injection may give an average of 6 weeks of relief. (Salhan et al. 2016)
- Be selective in patients who may require surgery

Surgery with Rib resection often is accomplished with transaxillary or supraclavicular approach, particularly if scalenectomy or cervical rib resection is necessary.

5 Hemodialysis

Authors: Young Lee and Matthew Smeds

5.1 Planning access creation

Management of Dialysis access is an important topic of discussion, not only because it is a significant part of board examinations, but also because healthcare costs continue to rise for ESRD patients, particularly during the transition from CKD to ESRD. This is attributed to use of dialysis catheters and frequent hospitalizations for arteriovenous access failures and related procedures.

The National Kidney Foundation-Dialysis Outcomes Quality Initiative (NKF-KDOQI) and SVS has provided guidelines in the follow areas: (Foundation 2015; Sidawy et al. 2008)

- Timing of referral to access surgeons
- Operative strategies to maximize placement of autogenous AV accesses
- First choice for autogenous access
- Choice of AV access when a patient is not a suitable candidate for a forearm autogenous access
- The role of monitoring and surveillance of AV access management
- Conversion of a prosthetic AV access to a secondary autogenous AV access
- Management of nonfunctional or failed AV access

This brings us to the question, who needs dialysis access?

Patients should be referred to a vascular surgeon for access when their creatinine clearance is <25mL/min which is CKD stage 4. You want to provide adequate time for your autogenous access to mature, so the ideal time for access creation would be > 6 months before anticipated need of dialysis. This allows for time for any subsequent interventions if your access is not maturing.

Should prosthetic access also be placed several months before anticipated dialysis?

Prosthetic access patency is limited by duration of access placement, thus, if a patient requires prosthetic access, placement should be delayed until about 3-6 weeks before initiation of dialysis.(Foundation 2015)

For dialysis access creation, which site should be considered and used first?

Due to the easier accessibility and lower infection rates, upper extremity access sites are used first. Furthermore, you want to place your access as far distally in the extremity as possible to preserve the proximal arm for future accesses.

What are some important considerations in a patient's history when planning a dialysis access?

It is important to find out recent history of:

- Previous access procedures
- Peripheral IV lines
- Sites of any existing or previous indwelling catheters including pacemakers and defibrillators
- Trauma or surgery to the upper extremities

Moreover, you also want to consider the patient's quality of life, thus, noting which extremity is dominant is important. If possible, you want to create your dialysis access in the non-dominant arm so that when the patient is receiving dialysis multiple times a week, they are able to use their dominant arm during their dialysis sessions.

However, placing a fistula with lowest rate of failure is preferred to using non-dominant hand. Therefore, placement of access on the patients dominant side would be preferred over placement of access with high risk of failure (i.e. on the side of an implantable cardiac device).(Sgroi et al. 2019)

As with any preoperative planning, physical examination is extremely important. Central venous stenosis can cause problems such as prolonged bleeding after dialysis sessions at the puncture site. What are some signs of central venous stenosis?

Unilateral arm swelling or edema and prominent venous collaterals are signs of central venous stenosis. Central venous stenosis can lead to venous hypertension which affects access patency and function, and also causes disabling edema. Placement of AV access in a patient with undiagnosed SVC stenosis or obstruction can cause SVC syndrome.(Kalra, Bjarnason, and Gloviczki 2019)

Beyond signs of central venous stenosis, when examining a patient, an Allen's test should always be performed to evaluate palmar arch patency.

Preoperative planning should also include arterial and venous assessments. What are your size requirements for the artery and the vein to be used in your dialysis access creation?

First, you want equal pressure gradients in bilateral upper extremities. If there is a pressure differential >20mmHg or pulses are non-palpable then patients should undergo a duplex ultrasound to evaluate for arterial insufficiency and and the artery should be greater than or equal to 2mm. Segmental pressures have limited applicability. Patients with correctable lesions on duplex may consider undergoing angiography and intervention to optimize inflow. (Sidawy et al. 2008)

A venous duplex should also be done to evaluate for diameter, distensibility and continuity. A vein mapping is useful to determine the size of the patient's superficial veins at various points in the forearm and upper arm. The vein should ideally be at least 3mm, but 2.5mm will likely dilate after regional anesthesia and be sufficient for access particularly in smaller patients.(Huber et al. 2002; Smith, Gohil, and Chetter 2012)

Autogenous access should always be considered first due to higher patency rates, lower infection rates, and longer duration of access survival. What are the different configurations of autogenous accesses?

The first and best option would be direct arteriovenous anastomosis. However, if that is not possible, then venous transposition should be considered next follow by venous translocation.

Venous transposition is for deeper veins such as the basilic vein, which is transposed so the vein lies just below the skin for easier access for puncture during dialysis sessions. This can be done in either a one-stage or two-stage procedure. In a 2-stage procedure, the direct arteriovenous anastomosis is created during the first stage and once the vein has arterialized 4-6 weeks later, the second stage of transposition is done when the vein is easier to mobilize. Translocation procedures include harvesting the femoral or saphenous vein and using it as a conduit for AV access creation in the upper extremity.

When can a venous transposition be done in a one stage procedure?

Many surgeons will do a brachiobasilic arteriovenous fistula creation with the transposition in one-stage. However, if the vein is <4mm, it is generally recommended to do a two-stage procedure so that if the fistula fails to mature, the patient does not have to undergo a second operation with extensive dissection of the vein. When a patient has not been initiated on dialysis and there is marginal vein that would require transposition, but carry a significant risk of graft failure then a first-stage BB fistula may be appropriate.

When comparing single- to two-stage brachiobasilic fistula, two-stage creation had high rates of primary patency at 2-years. However, there was no difference in infection rate, steal syndrome, hematoma, pseudoaneurysm, stenosis, failure rate, primary or secondary patency at 1-year or secondary patency at 2-years.(Jun Yan Wee et al. 2018)

It was mentioned earlier that the dialysis access should be created as distally as possible on the extremity. What are some of the most distal locations?

The snuffbox fistula, which is the posterior radial branch to cephalic direct access and Brescia-Cimino-Appell (often shortened to just Cimino) fistula which is the radial-cephalic wrist direct access are two of the most distal fistulas that can be created. Overall, these have comparable patency and ischemia risk. While snuff box fistulas carry a 10% failure rate at 1mo, the majority are able to be converted to RC fistulas.(Siracuse et al. 2019)

💡 Take a Listen

Check out our episode on the history of AV Access where we interview Dr. Kenneth Appell, one of the surgeons who pioneered this surgical technique.

What are your arterial and venous options in the upper extremity?

In the forearm, you have your radial, ulnar, and brachial arteries and cephalic and basilic veins. In the upper arm, you have your brachial or proximal radial arteries and cephalic, basilic, brachial and axillary veins.

If adequate vein is available, most common access sites distal to proximal include:

- 1. Snuff box
- 2. Radiocephalic
- 3. Basilic transposition in forearm
- 4. Brachiocephalic fistula these have a few advantages over more distal fistulas, including higher maturation rates, decreased time to maturity, higher patency and functional primary patency.(Nguyen et al. 2007)
- 5. Brachiobasilic fistula(Gilmore 2006)
- 6. Vein transposition from a remote site Femoral vein transposition can be used for upper extremity dialysis access in patients with previous upper extremity access complicated by infection, thus not a candidate for prosthetic, but no central stenosis. There is a high rate of hand ischemia due to larger conduit (up to 43%).(Huber et al. 2004)

If you need to use a prosthetic graft, what would you use?

PTFE is the most commonly used prosthetic graft. Either a 6mm graft or a tapered 4-7mm graft can be used. The tapered grafts help to ensure that the size of your arterial anastomosis isn't too large to minimize chances of steal. However, recent publications found that there is no difference between tapered and non-tapered grafts in terms of primary patency, steal syndrome, complication rates or need for reintervention. (Roberts et al. 2019; Han, Seo, and Ryu 2017) Both the 6mm PTFE graft and the tapered 4-7mm PTFE grafts are standard sizes used for dialysis access creation.

Overall AVG have benefit of earlier cannulation, however they have lower patency rates and higher infection rates. There is no difference between grafts and fistulas for perioperative morbidity and mortality.(Macsata and Sidawy 2019)

Most prosthetic access will fail because of the development of venous outflow stenosis as a result of smooth muscle proliferation at vein graft interface. Thus, a major cause of graft failure is due to thrombosis. Heparin-bonded PTFE (HB-PTFE) grafts are an option, however, studies have shown no improvement in graft patency with HB-PTFE compared to standard PTFE grafts.(Shemesh et al. 2015; Davies, Anaya-Ayala, and El-Sayed 2016)

There are also early cannulation prosthetic grafts, which is another option for patients who need more immediate dialysis access. These grafts are constructed in 3 layers with an elastomeric membrane in the middle between two layers of ePTFE material. This allows for cannulation as early as 24 hours from implantation because the graft configuration minimizes dialysis needle bleeding. This graft options allows for avoidance of a central venous catheter. Studies have shown similar patency rates compared to standard ePTFE, earlier cannulation and catheter removal, and decreased catheter related complications.(Julien Al Shakarchi and Inston 2019; J. K. Wagner et al. 2019)

For patients with central venous stenosis or occlusion, what is another alternative upper extremity access creation?

Two overall strategies for managing patients with severe central venous stenosis or occlusion are lower extremity AV access of hybrid catheter/grafts. Patency of femoral vein transposition is better than hybrid catheter/graft placement.(Brownie 2016; Glickman 2011) We will cover specifics of these techniques later.

For lower extremity AV access, femoral vein transposition has been shown to have the best outcomes, followed by saphenous vein transposition and graft placement. Lower extremity access should be avoided in patients with peripheral vascular disease or lower extremity ischemic symptoms-when in doubt perform duplex and segmental pressures to fully evaluate. Veins should also be evaluated to ensure patency more proximally.(Parekh, Niyyar, and Vachhara-jani 2016)

For patients where lower extremity access may not be appropriate, the hemodialysis reliable outflow (HeRO) device can come to the rescue. These devices may not have been encountered in your training, so we will go into detail here. This device is composed of 2 components: a graft which is made of 6mm PTFE with a titanium coupler at one end, and a venous outflow component of a 19 Fr silicone catheter reinforced with a nitinol braid to prevent kinking. The graft portion is anastomosed to an artery, usually brachial, and is tunneled subcutaneously and the venous component is percutaneously placed into the right atrium via the IJ or subclavian vein. The two components are connected with a titanium coupler at the deltopectoral groove. If you need more immediate dialysis, the super HeRO comes to the rescue in which the graft portion is the early cannulation graft. For patients with thoracic central venous occlusion, there is also the SURFACER inside-out access catheter system. This is a device used to cross central occlusions from the right femoral vein. It allows placement of a tunneled central venous catheter, but it can also be utilized as an adjunct for HeRO placement. In a multicenter study, the results from the SAVE (Surfacer System to Facilitate Access in Venous Obstructions) registry showed that in 29 or 30 patients with thoracic central venous occlusion, a central venous catheter was successfully placed. Moreover, there were no device-related adverse events, catheter malposition, or intra-or postprocedural complications. (Gallieni et al. 2020)

Hybrid catheter/grafts are good alternatives in patients with previous line infections, central stenosis, peripheral vascular disease and no suitable vein in the upper extremities. They have shown fewer bacteremia episodes than catheter, low primary patency rate and acceptable secondary patency rates.(J. Al Shakarchi et al. 2015) Absolute contraindications include donor artery <3mm, inability to dilate outflow vein to 19f, allergy to device materials (ePTFE, silicone, titanium, nitinol), and current active infection.(Medical, n.d.)

5.2 Techniques for access creation

The techniques of arteriovenous fistula creation are common across access sites. Can you go through the techniques?

First the vein is identified and the distal end is transected and flushed with heparin. By flushing with the heparin, you are able to access the caliber and extent of the vein as well as identify any side branches

Then after distal and proximal control of your artery, a 4-6mm arteriotomy is made. The length is limited to decrease incidence of arterial steal. The artery is then flushed with heparin to avoid thrombosis during the anastomosis and an anastomosis is created between the side of the artery and the end of the vein. A 6-0 or 7-0 nonabsorbable continuous suture should be used to create the anastomosis to avoid future dilation of the anastomosis.

For prosthetic accesses, the length of the arteriotomy does not have to be limited to 4 to 6mm since the incidence of arterial steal is limited by the graft diameter. Both arterial inflow and venous outflow vessels need to be dissected prior to graft anastomoses. The graft should be tunneled close to the surface of the skin to allow for easier cannulation. Meticulous attention to sterile technique is important to avoid graft infections. And as with the fistula creations, a 6-0 or 7-0 nonabsorbable continuous suture should be used to create the anastomosis to avoid future dilation of the anastomosis.

💡 Take a Listen

Check out our episode on the basics of AV Access placement.

What are some other options if an access is not able to be created in the upper extremity?

Autogenous accesses can also be created in the lower extremity. Femoral artery to femoral vein or saphenous vein anastomosis can be created. Both veins have to be transposed. Synthetic grafts have high infection rates (as high as 22% in some series) and only a 50% patency at 6 months.(G. A. Antoniou et al. 2009; Lazarides et al. 2018; Pike et al. 2019) Risk of lower extremity ischemia increased with burden of PVD, AV fistula vs graft, distal arterial inflow, and large femoral vein mismatch, but can be tempered by banding/narrowing of femoral vein at the time of surgery.(G. A. Antoniou et al. 2009; Bourquelot et al. 2012; Gradman, Laub, and Cohen 2005)

Access creation in the chest wall or cervical region is also possible with axillary artery to ipsilateral axillary vein loop access, axillary artery to contralateral axillary or jugular vein straight access (ie necklace access) and brachial artery to jugular vein straight access. Keep in mind that for these fistulas, the central veins must be patent.

5.2.1 Anesthetic Considerations

Are there benefits to different anesthetic techniques used during access placement, such as regional anesthesia?

Regional anesthesia for AVF creation has been associated with higher perioerative flow and lower rates of vasospasm resulting in higher rates of short term patency. Regional anesthesia has not been associated with perioperative rates of major morbidity or mortality. (Aitken et al. 2016; Siracuse et al. 2014)

Due to underlying comorbidities, it is often best to avoid general anesthesia in these patients. Many access procedures can be performed under regional or straight local anesthetic. There, it is very important to understand the pharmacokinetics and maximum doses of common local anesthetics.(Neal et al. 2018) Historically, it has been taught that the inclusion of epinephrine allows for higher dosages, but more recent reports do not include this adjustment.(Sztajnkrycer 2019) Common local anesthetics include:

- Bupivacaine Long acting. Max SubQ Dose is 2mg/kg.
- Lidocaine Medium acting. Max SubQ Dose is 4.5 mg/kg.
- Mepivacaine Medium acting. Max SubQ Dose is 4.4 mg/kg.
- Ropivacaine Long acting. Max SubQ Dose is 3 mg/kg.

Check out this Calculator

MedCalc has a useful calculator for estimating maximum allowable volume of anesthetic based on patient weight and dose percentage.

5.2.2 Endovascular AV Fistula

What is the role for endovascular creation of AV fistula?

Endovascular approach to fistula creation without open surgery is another option for dialysis access creation. The results of the NEAT study (Novel Endovacular Access Trial), prospective multicenter study which showed that 98% of the 80 patients enrolled had EndoAVFs created. Of these, 87% were physiologically suited for dialysis, and functional usability (2 needle cannulation) was 64%. Primary patency was 69% and and cumulative patency was 84%. Total complication rate was 8% and this is a promising alternative to surgical arteriovenous fistula creations. (Lok et al. 2017)

5.3 Maintenance and complications

When is the newly created dialysis access ready for use?

A good way to remember this is the rule of 6's. Fistulas should be created about 6 months prior to start of hemodialysis. It is ready to use when the fistula is 6mm in diameter, has a flow of 600ml/min, is 6mm from the surface of the skin and usually takes 6 weeks to mature.

Prosthetic AV accesses can be used as early as 2 weeks postoperatively. If you use the 3-layer early cannulation grafts, the access can be used as early as 24 hours after access creation. This is great because it offers the potential for avoidance of dialysis catheters in patients who need dialysis immediately.(Glickman et al. 2015)

5.3.1 Failure to mature

What are some reasons why an access may fail to mature?

Sometimes your access may have arterial inflow stenosis. This is difficult to detect clinically because there will be a palpable thrill, however, due to the stenosis, the flow is not sufficient enough for dialysis. Dialysis access duplex is a useful way to assess flow volumes and identify areas of stenosis that could be further assessed with fistulogram. In the absence of arterial inflow issues, collateral or large venous branches can divert blood away from the main access channel resulting in insufficient flow. This can be resolved by coiling or ligating the large venous branches that limit maturation of the fistula.

If the newly created AV fistula is not maturing, what are some secondary procedures to help with maturation?

First a duplex ultrasound can help to identify whether this is an inflow, outflow, or conduit issue. Once the likely source of the issue is identified, then there are multiple open or endovascular techniques to assist maturation.

Endovascular procedures include arterial and venous angioplasties to improve inflow or outflow issues.(Sidawy et al. 2008) In particular, balloon assisted maturation (BAM) has been described to decrease maturation time. A prospective randomized controlled study by Elkassaby et al. showed that BAM significantly decreased maturation time and had higher successful functional maturation. However, BAM was also shown to have increased fistula complication rates.(Elkassaby et al. 2021) Open procedures include vein patches, interposition vein grafts, vein transposition to proximal arteries, branch ligations, and vein superficialization.

5.3.2 Access Failure

Once a dialysis access is created, maintenance of the access is extremely important. The flow disturbances and hemodynamic changes associated with AV access creation causes intimal hyperplasia leading to venous outflow stenosis. This can ultimately lead to access thrombosis and failure. What are some methods of detecting access failure?

One way of detecting a well functioning access is a strong thrill at the arterial anastomosis which continues a few centimeters into the outflow vein. If you feel a pulsation near the venous outflow, then a stenosis or thrombosis is likely. If you feel a thrill distal to the area of pulsation, then you have likely localized your area of stenosis. It is important to note that you may feel a pulsation at a pseudoaneurysm independent of venous outflow issues.

Other signs of outflow stenosis are collateral veins or upper extremity edema.(Padberg, Calligaro, and Sidawy 2008) This is indicative of venous hypertension likely secondary to stenosis. You will typically see this in the shoulder area or anterior chest as a results of subclavian vein stenosis/thrombosis. Moreover, these high venous pressures as a result of the stenosis can result in excessive and prolonged bleeding after removal of needles from the dialysis puncture sites. This is often the first sign of elevated venous pressures. These patients should undergo a fistulogram to evaluate for underlying outflow stenosis.(Caro Monroig et al. 2018) Sometimes, a hematoma from dialysis access cannulation can result in compression of the fistula. In these cases, the hematoma should be drained to relieve the compression.

The most common cause of graft failure in upper extremity fistulas is venous outflow stenosis and in grafts is venous anastomotic intimal hyperplasia. (Berman and Gentile 2001; Padberg, Calligaro, and Sidawy 2008) Diagnostic fistulogram is a very useful tool to diagnose and treat potential access complications. Diagnostic fistulogram is often performed with proximal access of the fistula a few centimeters distal to the anastomosis. Compression of venous outflow is necessary to evaluate for inflow and anastomotic issues.(Bountouris et al. 2018)

Finally, even patients with normal physical exam can have issues while on the hemodialysis circuit, such as recirculation. When in doubt, if there are recurrent issues with hemodialysis, the patient should undergo a fistulogram to evaluate for occult venous outflow stenosis that may be missed on other diagnostic modalities. (Sidawy et al. 2008)

What are some endovascular interventions for a failing access?

First line therapy for outflow stenosis is a simple balloon angioplasty of the stenosed area. (Berman and Gentile 2001) Insufflation times are generally up to 2-3 minutes. Treatment of stenosis 2/2 intimal hyperplasia often require high pressures of 20 ATM or more. However, this is a double edge sword because this can lead to trauma in the veins stimulating a further intimal hyperplasia process. Some advocate a cutting balloon before high pressure dilation. There are also studies showing improved primary patency with drug coated balloons with no difference in survival compared to plain angioplasty.(Han, Seo, and Ryu 2017; X. Chen et al. 2020; Moreno-Sánchez et al. 2020; Yin et al. 2021) Stenting is also an option to treat residual stenosis or dissections after balloon angioplasty. Covered stents have shown good patency results.

Some unique situations include, recurrent cephalic arch stenosis which may be best treated with bare metal stenting. (Shemesh et al. 2008) Proximal occlusions (i.e. subclavian) may be best treated primarily with a covered stent. (Agarwal 2015; Anaya-Ayala et al. 2011)

If endovascular interventions fail, what are some open options for managing a failing access?

Generally an interposition graft or patch angioplasty is performed and the results of the two techniques are largely equivalent.

If an AV access has ultimately failed and thrombosed, what are your endovascular options at this point?

Some endovascular options are catheter directed thrombolysis with about 2-4mg of tPA injected into the clot, followed by balloon angioplasty (typically an 8mm by 8cm high pressure balloon). A mechanical thrombectomy device, such as angiojet, can also be used in combination to thrombolysis.

Alternatively, an open thrombectomy with a thromboembolectomy balloon and patch angioplasty of venous stenosis areas can also be used. Moreover, a hybrid approach of open thrombectomy with percutaneous interventions of venous stenosis areas has been described. Thrombosed fistulas are difficult to salvage, but thrombosed AV grafts have a high likelihood of successful recannalization and should be managed aggressively.

Immediate postoperative thrombosis of an AV graft is likely technical - inadequate inflow (i.e. small brachial artery) or outflow stenosis/occlusion. Open revision and thrombectomy is often the best option.(Paulson, Ram, and Zibari 2002)

What are some ways to manage failing AV access in the setting of ipsilateral vTOS?

Performing a first rib resection in a hemodialysis patient is high risk and controversial. However, if there is a prominent external jugular vein, then there are reports of external to internal jugular vein transposition offering more in line drainage and access salvage. (DeGiovanni, Son, and Salehi 2020)

5.3.3 Steal Syndrome

Earlier, you mentioned steal syndrome, can you explain to us what this is?

Steal syndrome is also known as Access Related Hand Ischemia (ARHI). It is an uncommon but devastating complication of access creation. All patients with arteriovenous fistulas have some degree of physiologic steal or reversal of flow in part of the artery distal to the fistula. However, this is not sufficient enough to cause ischemia. Rather, ischemia results from inadequate collateral circulation and inability of peripheral arteries to meet the increased demand. Diseased vessels do not dilate and stenosis of arteries leads to decreased distal perfusion pressure. Furthermore, hypotension during dialysis further decreases perfusion causing symptoms. Steal can be limb threatening and is graded as follows:

- Grade 0 asymptomatic, no flow augmentation or steal
- Grade 1 asymptomatic, mild ischemia with signs of cool extremity and flow augmentation with access occlusion (May be seen in over half of AV access patients)(Leake et al. 2015)
- Grade 2 moderate/intermittent ischemia that is experienced only during dialysis and patients feel effort induced ischemic pain.
- Grade 3 severe, ischemic pain at rest with tissue loss.

What are some symptoms and signs of Steal syndrome?

Symptoms include coolness, parasthesias, rest pain, and weakness. Signs of steal include cool to touch, pallor, cyanosis, delayed capillary refill, absent pulses/signals, diminished sensation, weak grip, and in severe cases ulceration or gangrene. If the patient shows improvement with access compression, diagnosis is confirmed.

Patients with acute pain and loss of pulses immediately after AVF creation should first have arterial injury/thrombosis ruled out, but then after diagnosed with steal syndrome should undergo fistula ligation.(H. Schanzer and Eisenberg 2004; Yevzlin, Chan, and Asif 2016)

When is an intervention necessary to treat steal syndrome?

You do not need to intervene for grade 0 and 1.(Leake et al. 2015) For grade 3 an intervention is mandatory. The goal of treatment includes symptom resolution and access preservation, and this is achieved by reducing access flow and increasing distal arterial flow.

The most important finding on duplex ultrasound to evaluate for steal syndrome is fistula flow rate. High flow rates suggest a primary problem with the fistula, low or normal flow suggests underlying PVD and inflow insufficiency as etiology.(Julien Al Shakarchi et al. 2016; Leake et al. 2015) Approximately 5% of steal is secondary to inflow stenosis.

What are your intervention options for resolving steal syndrome?

There are multiple ways to manage steal syndrome, detailed as follows:(Leake et al. 2015; Gupta et al. 2011)

- AV Fistula Banding is a simple option to reduce access flow. This is done by suture plication, placement of single narrowing tie or wrap by constrictive cuff to cause a stenosis in the AV access near the arterial anastomosis.
 - A minimally invasive approach is used by the MILLER banding which uses an endoluminal 4 or 5mm balloon as a sizer and a suture is placed around the access with the balloon inflated.
 - This procedure increases arterial inflow towards the hand. One technique describes using intra-operative finger pressures and plication to result in intraoperative increase in finger pressure by 20% or a minimum of 90mmHg. Flow should be maintained at 600ml/min.
- Revision using distal inflow (RUDI) involves ligation of the fistula at the arterial anastomosis and reestablishment of flow via a more distal artery by bypass or vein translocation. This allows for decreased flow through the access by reducing the fistula diameter and by taking inflow form a smaller vessel. However, ultimately, the fistula is placed at risk.
- Proximalization of arterial inflow (PAI) involves ligation of AV anastomosis, and the inflow is moved to a more proximal level with a prosthetic interposition. Dialysis can be continue via the vein. The main advantage is the native artery's continuity.(Zanow, Kruger, and Scholz 2006)
- Distal revascularization-interval ligation (DRIL) is ultimately considered the best option by many vascular surgeons due to the excellent results shown. There is an arterial bypass created originating proximal to the access and ending distal to the access, with ligation of the artery distal to the anastomosis. This prevents retrograde flow from distal vessels and allows for a low resistance pathway for arterial supply to the hand. DRIL may be particularly useful in patients with distal brachial, proximal ulnar/radial disease, as this can bypass the underlying stenosis as well.(Leake et al. 2015)

• Distal radial artery ligation (DRAL) can be performed for palmar arch steal syndrome from radio-cephalic av accesses, to prevent reversal of flow in the palmar arch. However, the ulnar artery patency needs to evaluated first.

5.3.4 Neuropathy and Neuropraxia

What complication often presents like steal syndrome, but with easily palpable distal pulses?

Access creation can result in neuropathy. It is important to note that over 2/3s of the patients have preexisting peripheral neuropathy. Neuropathy is also graded:

- Grade 0 asymptomatic
- Grade 1 mild intermittent changes (pain, paresthesia, numbress with sensory deficit)
- Grade 2 moderate persistent sensory changes
- Grade 3 severe sensory changes with progressive motor loss (motion, strength, muscle wasting).

Ischemic Monomelic Neuropathy (IMN) is rare but occurs acutely after AV access creation. Within hours of surgery, patients develop acute pain, weakness, or paralysis of hand and forearm muscles with prominent sensory loss. However, the hand is warm with palpable pulse or audible signal in distal radial and ulnar arteries. It is important to note that pain out of proportion is what differentiates IMN from ARHI. Treatment is access ligation or emergent augmentation of flow.

5.3.5 Perioperative neuropraxia

What are some common technical complications after placement of AV access?

Post operative numbress of the thumb, index and middle finger and weakness of abduction and grip strength may suggest median nerve neuropraxia and will likely resolve. However, you need to rule out steal syndrome. (Talebi et al. 2011; Vahdatpour et al. 2012)

Postoperative hematoma can often just be observed. However, signs of median nerve compression mandate immediate operative brachial sheath evacuation or the neurologic deficits may become permanent. (Padberg, Calligaro, and Sidawy 2008)

Prosthetic grafts can results in seroma from ultrafiltration of the graft and most resolve without intervention.

5.3.6 Bleeding and Aneurysm

What presentations require urgent revision?

Fistulas that present with bleeding from a visible ulcer, wet scabbing, frank infection or acute pseudoaneurysm often require urgent open revision. Some access may be salvaged, but in acute presentations with significant risk of life threatening hemorrhage the safest option may be to ligate the access and place a temporary dialysis access line.(Galbusera, Remuzzi, and Boccardo 2009)

Many fistulas develop aneurysms over time. When do fistula aneurysms require revision?

- Pseudoaneurysms often result from trauma due to repeated punctures or poor technique
- True aneurysms result from long standing hemodynamically significant outflow stenosis.(Hossny 2014; M. S. Patel et al. 2015)

Both can lead to cannulation difficulties, increased risk of thrombosis, pain, bleeding and cosmetic deformities. AV aneurysms should be revised if they develop skin thinning, ulceration or bleeding. Early skin changes can be observed. Intervention often requires revision with open plication/excision or ligation. When isolated to a short segment, attempts should be made to salvage the fistula.(Al-Jaishi et al. 2017; Al-Thani et al. 2017; Pasklinsky et al. 2011)

5.4 Dialysis Catheters

While ideally every patient would have surgically placed access, many patients still receive dialysis through catheters. What is the difference between an acute and chronic hemodialysis catheter?

Chronic catheters have a subcutaneous cuff at the exit site and tunneled to the vein. This decreases infection rates and makes them less likely to become dislodged. Tunneled hemodialysis catheters can be used up to 12 months.

If catheters cause so much problems such as infection and central venous stenosis, what would be an indication for them?

The most common indication would be for urgent hemodialysis. But other indications include patient who are not operative candidates due to advanced comorbidities, or patients who are unable to have an AVF or AVG due to anatomic feasibility. Temporary dialysis access may also be needed in patients who have just had a peritoneal dialysis catheter placement or in chronic peritoneal dialysis catheter patients requiring abdominal or inguinal surgery.

Which site is the most ideal site for a hemodialysis catheter?

The right internal jugular vein with a tip in at the cavo-atrial junction is preferred because it has the best patency. Subclavian veins are avoided due to high risk of stenosis and femoral veins are avoided due to infection risk.

If all traditional access sites are inaccessible, translumbar IVC catheters may be used as a last resort as salvage. Overall complications are similar to other permanent dialysis access sites, however obesity is a relative contraindication due to potential migration into the soft tissues.

Every procedure has potential complications. What are the immediate complications of catheter placement?

When placed in the internal jugular veins, there is always a chance of a pneumothorax or hemothorax. Wire embolism can occur is control of the wire is lost during the procedure. If the guidewire is placed too far, then there is always a chance of arrhythmia. Thus, the best place for the wire is through the IVC. With a left internal jugular vein approach, there is always a risk of thoracic duct laceration. If a leak is apparent, then the catheter needs to be removed immediately and a pressure dressing applied.

How do you manage infected hemodialysis lines?

Infected hemodialysis lines should be removed. Ideally you should avoid replacing any lines until blood cultures have been negative for 48hrs, particularly subsequent permanent lines(V. A. 2006. W. Group 2006)

5.5 Peritoneal Dialysis

Although not often managed by the vascular surgeon, we should still be aware of another method of dialysis in renal failure patients. What are the indications for peritoneal dialysis?

Peritoneal dialysis is better tolerated and less disruptive to daily life, so is often a better option for young, active patients with less comorbidities. PD can be considered first-line or in patients who are not candidates for HD, but require high patient engagement and education and so may not be appropriate for patients with other chronic diseases limiting independence, unstable housing, cognitive decline or poor management other medical issues.(Ansari 2011)

From a technical standpoint, PD may be difficult in patients who have had multiple previous open abdominal surgeries, but can be tolerated in patients who have had minimally invasive or minor abdominal procedures. (Sinnakirouchenan and Holley 2011) Long dwell times with glucose solution can sometimes lead to hyperglycemia. The major reason for conversion from PD to HD is PD catheter infection. (S. Li et al. 2017)

Part III

Thorax

6 Aortic Dissection

Authors: Matt Spreadbury, Adham Elmously, Einar Brevik, and Joseph Lombardi

6.1 Demographics

What is an aortic dissection?

It's when a tear occurs in the intima that results in separation of layers of the intima and media and allows blood to flow through the false lumen.

How common are they and how serious are they?

Acute dissections occur around 3/100000 - 2-3x more common than ruptured a ortic aneurysm. For Type A dissections, early mortality 1-2% per hour - if untreated, 20% die within 6 hours, 50% within 24 hours, 70% first week.

Main cause of death in type A is a ortic rupture into the pericardium, acute a ortic regurgitation, and coronary ostia compromise. While patients with descending thoracic a ortic dissections are more likely to die from end organ compromise due to obstruction of visceral or extremity vessels in the acute phase of the disease.

The time frame is also important.

- Hyperacute <24 hours
- Acute < 2 weeks
- Subacute 2 weeks 3 months -> TEVAR
- Chronic >3 months -> Chronic aneurysmal degeneration/ partial false lumen thrombosis (highest risk) = operative treatment

6.1.1 Classification and Terminology

When we think about a rtic dissections there are a few classifications, how can we break it down?

Historically, there are the Stanford and Debakey Criteria.

Anatomical Stanford

- Type A involves the ascending aorta, 2/3 (most common)
- Type B arises from distal to L subclavian, 1/3

Debakey

- A
 - 1 ascending + descending
 - 2 ascending only
- B distal or at the LSCA.
 - 3a Descending aorta above diaphragm
 - 3b Descending aorta above and below diaphragm

How about the new system proposed by Dr Lombardi, the SVS-STS classification system?

The new system published in 2020 keeps A and B and adds a number system which divides the aorta into zones from 0 proximally to 12 distally in the mid SFA. (Lombardi, Hughes, et al. 2020)



Type A is now JUST the ascending aorta to the innominate, also called Zone 0.

Type B is now an entry tear in Zone 1 or greater and distally to whichever zone the dissection lands in.

6.1.1.1 Penetrating Aortic Ulcer and Intramural Hematoma

This anatomical classification is based on reading the CT angio. What else could we see on a CT angio that we have to know about?

So aside from the aortic dissection its self, you could see a bleb of contrast sticking out. That could be an **penetrating aortic ulcer**. That is an atherosclerotic plaque that penetrates the internal elastic lamina of the aortic wall. (Ciccone et al. 2016)

Another key finding can be an **intramural hematoma** which is a hyper-dense crescent shaped hemorrhage within the aortic wall. There is no identifiable direct communication between the true and false lumen. IMH are classified in the same way but with the abbreviation IMH p-d zones.

Society for Vascular Surgery and Society of Thoracic Surgery Reporting Standards for Type B Aortic Dissections

	DEFINITIONS			
	Aortic Dissection	Tear in the intima that results in separation of layers of the media and allows blood to flow through the false lumen		
	Intramural Hematoma	There is no identifiable direct communication between true and false lumen. Characterized by hyper- dense, crescent-shaped hemorrhage within aortic wall		
	Penetrating Aortic Ulcer	Atherosclerotic lesion that penetrates the internal elastic lamina of the aortic wall. Often diagnosed in presence of intramural hematoma		
	Aortic Dissection Acuity		CHRONICITY	Time from Onset of Symptoms
	High Risk	Refractory pain or HTN Bloody pleural effusion Aortic diameter > 40 mm Radiographic only malperfusion Readmission Entry tear: Lesser curve location False lumen diameter > 22mm	Hyperacute	< 24 hours
			Acute	1-14 days
			Subacute	15-90 days
			Chronic	> 90 days
	Complicated	Rupture Malperfusion		
JVS Journal of Vascular Surgery orcur Publication of the Society for Vascular Surgery Copyright © 2020 by the Society for Vascular Surgery*				🏏 @JVascSurg

Whats the significance of these two in combination?

There is a higher chance of a rupture if a penetrating a ortic ulcer is seen with intramural hematoma.

When a patient presents with an aortic dissection how can we classify them clinically?

- Uncomplicated
 - Stable hemodynamics
 - No evidence of malperfusion
 - Pain controlled
- Complicated
 - End organ ischemia / malperfusion
 - Rupture or impending rupture
- High risk for early complication or continued growth (Bogerijen et al. 2014; Reutersberg et al. 2018)
 - Uncontrollable pain / hypertension
 - Bloody pleural effusion

- A ortic diameter $>40\mathrm{mm}$ / False lumen diameter $>22\mathrm{mm}$
- Readmission
- Radiographic only malperfusion
- Entry tear on the lesser curve
- Number vessels originating from false lumen and length of dissection.(Brunkwall et al. 2014; Kamman et al. 2017; Nienaber et al. 2009)

What is the danger of a false lumen? How does it lead to symptoms and malperfusion? Likewise which arteries commonly branch off the true lumen?

The false lumen can lead to end organ ischemia as the intimal flap can cover the ostia of branching vessels. This can be a static or a dynamic obstruction.

Likewise it also leads to weakening in the wall of the aorta which can become a threatened rupture or rupture if the diameter of the false lumen is larger than 22mm.

The celiac trunk, SMA, right renal typically come of the true lumen. Left renal comes off the false.

Also the dissection most commonly goes down into the left common iliac rather than the right. You might be able to detect down stream effects of this on clinical exam with reduced left sided groin pulse.

The presence of multiple false lumens is associated with increased risk of a ortic dissection related death.(Sueyoshi et al. 2013)

What kind of patients get aortic dissections?

Hypertension (older patients) vs cocaine or Meth (younger patients)

Marfans, Loeys-Dietz, Vascular Ehlers Danlos (Type 4), Turners, arteritis, Bicuspid aortic valve.

We also have a traumatic cause of aortic dissections. That being called blunt thoracic aortic injury:

- Grade 1: intima tear
- Grade 2: IMH
- Grade 3: Pseudo aneurysm
- Grade 4: Aortic rupture.

For more on the management of blunt aortic trauma, see Chapter 9

6.1.2 Presentation

How do these patients present?

Signs and symptoms – Chest pain 90% tearing pain radiating between the shoulder blades. Chest pain extending to the abdomen abdomen? Think mesenteric ischemia or aortic tear Type A - Stroke 5-10%, Syncope 15%, tamponade, carotid dissection, paralysis. Others: MI, Hypovolemic shock, leg ischemia

6.2 Evaluation

What is the workup?

Physical Exam –Asymmetric pulses / blood pressure differences / Diastolic murmur,

Investigations - CXR, EKG, D-dimer + Troponin, CTA, ECHO for type A.

The big distinction is to find out if this is a type A or type B because the treatment strategy is completely different.

- Type A need an emergent operation
- Type B starts with medical management, follow up CT angiogram +/- Trans esophageal echo in the OR. Reevaluate at 24 hours.

6.3 Management

What are the details of Type A treatment?

Operative treatment with a 30% operative mortality. Let cardiothoracic take the lead on this one. However vascular surgeons should be involved in the management of type A as after the repair, a type A can become a functional type B.

Type B is in the realm of vascular surgery. What is the first management step after we have diagnosed a type B dissection?

Invasive impulse therapy aimed to reduce a ortic wall stress.(G. C. Hughes, Andersen, and McCann 2013; Nienaber and Clough 2015) That means reducing the force of transmitted impulse down the aorta with blood pressure goals of 100-120mmHg and Hr < 60bpm.

How would you achieve that?

Start with a beta-blocker (esmolol or labetalol) first then a vasodilator (nitroprusside). This is to stop the sympathetic surge after vasodilation that could increase pressure and thus tearing forces inside the aorta worsening the dissection.

Initial CT, 72 hours, 3 months x 4, q6 months x2, q12 month. (Descending thoracic aorta that dilates first.)

Why isn't open surgery indicated for type B dissections?

Open surgery is not recommended due to the high mortality 30% if <48 hours. 18% if >49 hours.(Trimarchi et al. 2006)

In the acute setting mortality can be up to 50% with a 20% paraplegia risk. Its been described as sowing tissue paper.

What is the management plan for a complicated Type B aortic dissection?

Start with invasive medical management and plan for TEVAR. The goal with TEVAR being to direct the blood flow into the true lumen and seal the entry tear. If there was a dynamic obstruction (flap occludes branching vessels.) Then TEVAR would reestablish the true lumen hence removing the dynamic obstruction. Endovascular fenestration can also equalize the pressure in the true and false lumen. (Lombardi, Gleason, et al. 2020)

For a static occlusion there could be a thrombus or stenosis in the branched vessel so a stent might be indicated.

What are the major risks of TEVAR in the management of Type B aortic dissections?

- Retrograde type A (reported 2% in literature however it can be around 20% in some experiences). Carries a high mortality (ranging from 7-50%). Factors associated with retrograde dissection include:
 - Over-sized endograft every 1% over 9% results in an increase in retrograde dissection of 14%.(Canaud et al. 2014)
 - Proximal landing zone more than $40 \text{mm} (2\% \rightarrow 18\%)$
 - Periprocedural hypertension
 - Underlying aortopathy.(Tjaden et al. 2018)
- Paraplegia (reported around 5%)
- Stent induced new entry.

💡 Take a Listen

Check out Dr Arko's conversation on the BackTable podcast to learn more about the complications of TEVAR management in dissections.

Is there a role for TEVAR in uncomplicated type B dissections?

The INSTEAD and INSTEAD XL trials looked at uncomplicated Type B dissections. There was NO statistical difference at 2 years comparing OMT vs TEVAR but at 5 years there was good aortic remodeling and better long term survival in patients treated in the sub-acute stage.

Timing for TEVAR is a difficult choice. In chronic dissections the septum thickens leading to a potentially difficult TEVAR. Anecdotally, TEVAR is best at 2 weeks to 3 months.

It should be noted, that both the thoracic and abdominal aorta often continue to dilate even after repair so follow up surveillance is paramount.(Famularo, Meyermann, and Lombardi 2017)

If repair must extend into Zone 2, then pre-TEVAR carotid subclavian bypass should seriously be considered in elective/urgent inpatient setting to reduce the risk of periprocedural stroke (1.9% vs 14.3%).(Bradshaw et al. 2017; Teixeira et al. 2017) Revascularization of the L subclavian with a L subclavian bypass is particularly important with a LIMA CAMB or a dominant L vertebral and a transposition is not appropriate.(M. D. Morasch 2009b; Matsumura et al. 2009)
7 TAAA

Authors: Rachael Forsythe, Mohamed Barkat, Nicholas Greaves, and Michael Jenkins

Contributors: Sharif Ellozy, Leanna Erete

This episode was developed as a collaboration between Audible Bleeding and The Rouleaux Club in the United Kingdom.

Treatment for thoracoabdominal aneurysms has become a bit of sub-specialization within vascular surgery, with the best outcomes achieved at high-volume centers with close partnerships with cardiothoracic surgery. (Cowan et al. 2003) While the majority of vascular surgeons may not manage these patients in their day to day practice, and many trainees have limited exposure, this topic is still fair game for examinations and should be reviewed. However, endovascular techniques are still evolving and practice varies widely between centers and so covering this topic may not be particularly high yield.

Take a Listen

Check out our conversation with Dr. Tom Forbes where we discuss some challenges and opportunities in specialized aortic training and centralization of services.

This chapter was developed by surgeons in the United Kingdom, where manufactured custom devices have broader regulatory approval and are more widely available. The availability of these devices in the United States is more limited and so management decisions are different depending on what devices are available. However, the basic principles of open management and overall clinical decision making are shared no matter what side of the Atlantic ocean you practice.

7.1 Demographics

7.1.1 Anatomy

Can you take us through the Crawford classification to start off?

The Crawford classification is relatively recent (1986) and is a very practical classification, depending on body cavity and how to get to an aneurysm.(Crawford et al. 1986) This classification doesn't follow an expected pattern as getting either more extensive or less extensive.

- 1. **Type I** These extend from the left subclavian down to just below the diaphragm, which distinguishes it from an isolated thoracic aneurysm, which you can get to just from the chest. Going into a second body cavity is a very important differentiating marker.
- 2. **Type II** These are the biggies, they extend from the left subclavian all the way down to your bifurcation. So both abdominal and thoracic exposure, all the visceral, renal arteries and a lot of intercostal and lumbers—so big impact for cord supply, et cetera.
- 3. **Type III** These extend from the mid chest down to and involving the viscera, renals, and bifurcation. These differentiate from Type 2 because you may not need to utilize full support and may risk a clamp and go approach.
- 4. **Type IV** These are characterized by being accessible from the abdomen, in most patients, although it often does require division of the crus. In most patients, because there are some anatomical situations with body habitus, which means that going into the left chest is useful even for type four aneurysm.
- 5. **Type V** This was an additional classification that was added by Dr. Hazim Safi's group later, which is a bit like a type III at the top of the type I at the bottom effectively.(Safi and Miller 1999)

Take a Listen

Check out our conversation with Dr. Hazim Safi where he discusses the history of TAAA repair and much, much more.

There appears to have been an increase in the incidence of thoracoabdominal aneurysms. Do you think it's a true increase or does this relate to having scans for other reasons picked up by accident and then going on from that? What sort of many referrals do you get in a year and what's your turn down?

The overall incidence of infrarenal atherosclerotic abdominal aneurysms is going down every year in the national UK vascular registry, likely related to smoking reduction for a big group of patients. I think thoracoabdominal aneurysms are going up partly because we are now imaging more and more people and therefore we are imaging the chest and seeing them.

So that's an artificial increase in incidence, but also the rate of aortic dissection is on the rise and chronic post dissection aneurysms are increasing. In addition, I think it's a small group, but there is more knowledge about connective tissue disease, genetic studies and family screening, which perhaps is also a small part of annual increase. So over the next 10-20 years, aneurysms are not going to go away, and there is a huge number now within the national UK screening program under surveillance for small aneurysms.

7.1.2 Etiology

Are there major differences in presentation and etiology between thoracoabdominal aneurysms and abdominal aortic aneurysms from your experience?

On presentation, the majority of aneurysms are asymptomatic so they are most often found incidentally. Rarely, they get symptomatic and can get tender as they approach a time when the wall is going to breach pre-rupture. I think thoracoabdominal aneurysms tend to be a bit more symptomatic than infrarenal but may have some atypical symptomatic presentations.

- Some that extend into the arch may present with hoarseness from recurrent laryngeal tension or bronchial compression.
- There is a group that cause dilatation of the crus. The crus of the diaphragm acts like an extrinsic wrap and can be extremely tight. These patients present with excruciating pain, radiating around their costal margin, which is a presentation not often thought of as a sign of an aneurysm.
- The larger extent and more chronic aneurysms may even present with things like weight loss and general poor health.

The other group that is a little different than infrarenal aneurysms, is the post dissection aneurysms. The majority of these are known and followed closely, however there was a time when patients with a type A dissections that extended all the way down were only followed with with echo of the ascending and the rest was forgotten about. Many of them got lost from cardiothoracic follow up. That group is a different group because we already know they've got an aneurysm. So there are subtle differences.

And then the connective tissue group, I suppose, is a bit different because they tend to have more extensive aneurysms rather than just confined to the infrarenal segment, which is by far and way, the most common for abdominal aneurysms.

Do you think patients with thoracic aneurysms as a result of dissection or connective tissue disease have a different threshold for intervention compared to the non-connective tissue disorder patients?

I think we need to be aware that the connective tissue disorders we talk about—Marfans, Loeys Dietz, vascular Ehlers Danlos—are probably the tip of the iceberg. There are likely a number of other cases that we don't have genetic sequencing for, but behave differently than classic atherosclerotic aneurysms. The data from a dimension point of view is not quite as robust as one would hope, but the threshold that most people would agree for non connective tissue disorders would be about six centimeters. That takes into account the increased risks of operating both in the chest and the abdominal segment.

I think most people would agree that five centimeters is a better cutoff for connective tissue patients. Some cardiac surgeons would repair a young Marfan ascending aneurysm at four and a half centimeters. It does vary a little bit geographically, a bit like the threshold for infrarenal

aneurysms, which varies between Europe, the States and the UK. So you've got to look at the patient in front of you and make a decision. I would certainly increase the threshold for someone who was not so fit and perhaps lower it a bit for a younger patient.

For more about connective tissue related aortopathies, see Chapter 8

Also, be aware if it's something odd anatomically—a saccular bulge, an eccentric penetrating aortic ulcer (PAU), or something where you think there could be a mycotic element—those are all very different. You can't be reassured by a dimension in an axial plane that is safe. These findings up the ante, in terms of whether you would repair at an earlier threshold because they do not behave like a conventional fusiform aneurysm.

Can you talk us through the Ishimaru zones of the aorta?

Ishimaru is a useful classifications. There are a lot classifications in medicine, but it's nice to find one that's actually useful. This classification made sure that everyone was on the same page when reporting proximal seals zones for thoracic devices. That is important for two reasons:

- 1. The extent of coverage
- 2. The complications increased, the more proximally you go. The main complication for that is stroke.

So historically, Ishimaru decided to classify these zones as(Ishimaru 2004):

- **Zone 0**, the first most proximal zone, is the ascending aorta up to and including the brachiocephalic trunk
- **Zone 1** is between the brachiocephalic trunk and including the left common carotid artery
- Zone 2 is between the left common carotid artery and including the left subclavian artery
- Zone 3 is the aortic segment just distal to the left subclavian artery.
- **Zone 4**, added by some people, is distal to the T4 level, which is much lower down the thoracic aorta

Most thoracic stenting will go really to zone three or perhaps into zone two. The more proximal you go, the more work needs to be done in terms of either extra anatomical debranching or using some form of fenestrated or branched arch device with an increase in stroke risk. What this allowed people to do is compare different series. You're not just saying these were a group of TEVAR patients, but you could define exactly how proximally they go.

And the same applies for the Crawford classification. It allows comparison within thoracoabdominal groups, and that's important with both survival and complication rates. Because if your series is mainly type II thoracoabdominal aneurysms, you're going to have a very different outcomes from someone who's got mainly type IV aneurysms. Both classifications allow you to look at and compare data between groups.

For a discussion of the new SVS/STS classification system for a ortic dissections, please see Chapter 6

7.2 Management

7.2.1 Ascending aorta and arch

The cardiac surgeons use many different techniques for managing an ascending aortic aneurysm or dissections, which is beyond the scope of this review. However, in preparation for management of aneurysms or dissections that extend beyond the arch, they often utilize an elephant trunk down the descending aorta. Can you briefly summarize the difference between a conventional elephant trunk and a frozen elephant trunk, and when to use one over the other?

This is particularly pertinent to aortic dissection. The the main purpose of repair in Type A dissection is to protect the heart. People die of either rupture into pericardium causing tamponade or stripping the coronary ostia off and getting myocardial ischemia. So the priority of repair in acute ascending aortic dissection is primarily to protect the heart and to some extent to ensure a true lumen flow distally. Therefore, it was very popular because it was the least invasive to do a short interposition ascending repair, but that left problems for later.

As people became more adept with cardiac and cerebral protection, it became more popular to do a more extensive repair in the first sitting. This involves an arch repair and an acceptance that eventually the descending thoracic aorta will still need to be repaired, but at a later stage. The ascending and arch is done from a median sternotomy, and it's really difficult to get beyond the left subclavian from that position.

So when the arch was done, the elephant trunk came from leaving an extra piece of Dacron within the descending thoracic aorta in the true lumen of a dissection or in the main lumen of an aneurysm by a double sewing technique on the distal anastomosis and then inverting it down the descending thoracic aorta. The benefit of that was when returning to repair descending thoracic segment via left thoracotomy, you couldquickly open the aorta and clamp that Dacron for a ready, made proximal anastomosis. This was much easier, because it meant you didn't have to dissect all the way up to the left subclavian with scar tissue and a previous anastomosis.

Now, what industry realized is that they could help with this procedure by developing a Dacron device sized to be an arch replacement which had three or four ready-sewn 10 millimeter branches with an extra pipe for rewarming. These branches are used as bypasses to the innominate, carotid and subclavian and attached to that piece of Dacron was a stent graft, which could be placed distally in lieu of what was previously a floppy piece of Dacron.

And two manufacturers, JOTEC and Terumo Aortic, have made these devices, which facilitate and make things much easier. Therefore, so-called FET or frozen elephant trunk, has now become quite popular. I don't quite know why it's called frozen, but I think it perhaps means that the thoracic segment is stiff with a supported stent graft rather than just a floppy Dacron segment.

7.2.2 Isolated Thoracic Aneurysms

For isolated thoracic aneurysm treated with TEVAR, what is the optimal landing zone?

So I think the isolated, thoracic aneurysm is a perfect application for TEVAR because if you can get a good seal zone, proximally and distally, it provides an an endoluminal approach and saves a thoracotomy, which is a massive difference for these patients. Unfortunately I suppose those cartoons you see on the industry advertisements are vanishingly rare, when you've got a perfect proximal and distal landing zone and a really straight thoracic aorta.

I think for the majority of cases, you need a 2cm seal zone proximally and distally. However, if you have a torturous aorta, you likely need a longer seal zone. You want to land in an area where there is a good parallel walled walled segment. You have to be careful of so-called "bird-beaking" when a stiff, less conformable device lands perfectly on the outer curve, but holds off on the inner curve. This results in a lip, which protrudes and allows blood to get under that and cause stent graft crushing.(Marrocco-Trischitta et al. 2019)

Modern day outcomes are very good, because I think modern devices have become much more conformable. We've learned many lessons to avoid too much oversizing and how to taper stent grafts when there's mismatch between proximal and distal landing zones.

These have all significantly improved things, but there's still no getting away from the fact that some anatomy is not well adjusted to the currently available stent grafts. Gothic arches, torturous and large sacks, are a problem and can allow a stent graft to move away from the center line to the outer curve. (Iwakoshi et al. 2019) If that sac doesn't shrink, you can imagine that draws on both the proximal and distal seals zone, and reduces them.

I think you can sometimes get away with a shorter landing zone. If, for example, you're doing a post traumatic aneurysm where it's only on one segment of the outer wall where there's a problem. But for the vast majority of true aneurysms, you need that seal zone and you do need to oversize to achieve that.

For more on the endovascular management of traumatic aneurysms, see Chapter 9

Do you utilize rapid ventricular pacing when deploying these stents?

I think the majority of aneurysms at the left subclavian area, you can just drop the blood pressure to a systolic of 70 or 80 pharmacologically. Once you go more proximal to that, especially if you've got a custom device, with a scallop or branched device where you need an

absolutely critical landing zone, you then really need to achieve a short transient period of circulatory arrest, either with adenosine, which is perhaps sometimes a bit unreliable, or rapid pacing.

Rapid Pacing can be difficult and does come with complications though, such as ventricular puncture. There are some newer techniques in terms of caval occlusion balloons, which basically stop venous blood returning to the heart with a reduction in output. Those are gaining popularity certainly in Germany and they can be quite effective for that. I think having tip capture on devices and better, more accurate deployment and better imaging does allow you to be more accurate in deployment, but there are still problems.

These devices have been inserted to very torturous iliac systems sometimes and they retain some energy. When you deliver them by removing that sheath, sometimes that energy is still there and they can jump forward as well as back. This can cause problems because you end up covering a vessel that you didn't intend to.

As we have described TEVAR is a great option for the majority of isolated, thoracic aneurysms, but is there still a place for open repair?

I'm not so sure in isolated, thoracic aneurysms, unless for some reason they're not suitable for TEVAR. I think if you've got problems with a mycotic aneurysm or a fistula into the bronchus or esophagus, these are our big problems with high mortality and your duty bound to go for open repair.

Patients with hemoptysis and a previous aortic repair should carry a high suspicion for an aortobronchial fistula. TEVAR is considered preferred repair, due to high morbidity and mortality with open repair. Bronchoscopy should be avoided due to high risk of rebleeding. (Bailey et al. 2011; Léobon et al. 2002; Quintana et al. 2006)

Certainly with connective tissue patients, open has advantages. In some unique situations it is possible to bridge between repairs. I think in some of the younger connective tissue patients, they will end up having certain segments repaired at different stages of their life. Although it's probably now accepted that "all endovascular" is not a good option for these patients, if you've got Dacron proximally and distally already, bridging those segments with an endovascular device might work well. Although the Dacron will slowly dilate, it won't dilate as much as aortic tissue in these patients. So there are options, but I think in reality, isolated, thoracic aneurysms on reasonable rare and TEVAR can be used for the majority of them.

7.2.3 Thoracoabdominal Aneurysms

Moving on to thoracoabdominal aneurysms. Earlier we used the Crawford classification to describe the anatomical extent of the thoracoabdominal aneurysm. So let's talk firstly, about type 1, 2, 3, and 5, which involved the thoracic of water to a greater or lesser extent as well as the abdominal aorta. We'll come back to type IV later on in this podcast.

So type II open thoracoabdominal aneurysm is arguably the most invasive operation a patient can undergo and carries 30 day mortality in excess of 10 to 15%. What is the size threshold for type II that you currently use in your practice and what is the evidence for this?

So, as I suggested earlier, the evidence is that it's a bit historical, and most people would actually say about six centimeters. I think that can be brought down a little bit for connective tissue patients and can go up a little bit for atheromatous patients. But in general, you're talking about a younger segment of patients. This is not an operation for people in their eighties. Outcomes stratified by age greater than 50 found a higher rate of death, paraplegia, renal failure, cardiac complications and length of stay.(Coselli et al. 2017) These patients need to be in very good shape because it is a big onslaught physiologically on them. It's not just getting them off the table, it's getting them out of intensive care and out of hospital and to recover back to their baseline.

I think that's one of the things that can be really difficult about this disease. You are taking patients who are effectively functioning quite well, and they're often relatively asymptomatic and you're putting them through a prophylactic operation to try and prevent rupture, which no one quite knows if or when it'll happen.

So to that end, can you talk us through the decision making process when you're assessing a patient with a thoracoabdominal aneurysm types 1, 2, 3, or 5?

I think it is really dependent on anatomy, physiology, and the patient in front of you. Unfortunately, it is a bit of a bespoke assessment for an individual patient. So firstly, I think you have got to look at what are the options for that patient, which goes from conservative to endovascular, to open surgery.

The endovascular domain has changed hugely now with the increasing availability of branched devices. When I started, there weren't custom devices. There were some FEVAR devices just coming out, and we did use long covered stents going through devices with TEVAR above. However, there wasn't the option for what we've now got from an endovascular perspective.

💡 Take a Listen

The availability of fenestrated and branched devices is very different between the United Kingdom, Europe and the United States. In the UK, custom fenestrated and branched devices have received CE mark approval, which is similar to FDA approval in the United States. These devices do not have FDA approval in the United States, so access to them is far more limited.

Check out our conversation with Dr. Gustavo Oderich where we discuss the advances in endovascular techniques for branched and fenestrated devices in the United States.

Check out our conversation with Dr. Benjamin Starnes where we discuss the process of an investigation device exemption (IDE), which is how the majority of custom devices are performed in the United States. So I think you've got to choose the patients that are either unsuitable for endovascular, or are we really going to benefit from a durability of an open approach. Those are the two big differences, and they are differences. One, you're going for gold standard approach because you think they're life expectancy warrants that. The other one is much higher risk, because they haven't got another endovascular option. These range from people with very tortuous anatomy, to difficult renal arteries, such as early divisions, that make them unsuitable as target vessels. Another reason is complex dissections, which are not suitable for an endovascular approach. Therefore, it is quite variable, but the first thing you must consider is whether this patient is going to get through this and get out of hospital. And that really is looking at them in the eye and considering what's their quality of life or what's their family support, and are they up for this?

The second one then is anatomical difficulties— Is there a shaggy aorta, atheromatous disease, or calcification of a target vessel, etc. Severe aortic wall thrombus is associated with solid organ infarction (24%), acute kidney injury, without dialysis (21%) and delayed oral intake by 3.4d, however rarely resulted in mortality (0.5%).(Ribeiro et al. 2017) Rarely is this a hundred percent or 0% decision, but it's about building up relative contraindications. Some of them are pretty simple, like access problems for endovascular, and it's a combination of those which then will sway you in one way or the other. And if anything, that decision-making is probably more pertinent for type III and type IV, where there really is a different option and these patients are n their sixties and seventies with classic atheromatous disease. The really, young fit non-connective tissue patient with a type II TAAA is a rare beast.

For connective tissue disease, many textbooks prefer an open approach over an endovascular approach do you still agree with this?

Yes, I do. I think certainly, what were you know about patients who've had total endovascular approaches for connective tissue diseases is that they don't last. They may last five years or so and then you get progressive aortic dilatation and loss of those of seal zones. Whether that is the natural history of the native disease or whether it is regional force hooks, barbs or whatever is related to the device is unclear.

But when you sew these pensions and just touching the back end of a needle on the anterior wall can create a massive problem or just being a little bit clumsy and you get a radial tear. It's very clear that a stiff device sitting in those aortas for many years with 60-70 beats per minute going through it and often hypertension, isn't going last.

However there are some caveats, as I mentioned earlier, bridging between previous prosthetic open repairs is a possibility. And sometimes in a lifesaving situation, such as when someone has presented with a rupture, you may have to use an endovascular approach. However, be prepared then to treat that as a bridge and go back to do a definitive repair when things have calmed down and in the cold light of day.

So we don't really have the scope to dive into complex endovascular repair in this episode, but can you describe the broad principles of your operative approach to open thoracoabdominal aneurysm repair?

In terms of the patient, for anything juxtarenal or type IV, I tend to have a patient supine on their back, but with a break in the table at the level between the costal margin and the anterior superior iliac spine. By breaking the table at that level, you can increase the exposure to that segment of the aorta.

However, there are some patients even with a type IV repair that demand a left thoracoabdominal incision. For a type four, if you're doing a left thoracoabdominal approach, it's usually for someone with a very narrow acute angle of the costal margin, such as a more petite frame female patient or a crus that is very high on the aorta. And if you look at CT scans, which is something I've realized over many years, that not everyone is the same. So where the celiac and SMA are can be in a different proportion, depending on where the costal margin and and skeleton is, which can make them easier or harder to get to. And certainly in a situation, if there is a rupture within the left side of the abdomen, you wouldn't really want to go into that without getting proximal control. Because as you do the visceral rotation, you could have catastrophic bleeding.

So under those circumstances, I'd also go into the left chest with a reasonably low segment so that you can place a clamp ready before doing the visceral rotation. So for type IV the one above then, the higher you go, the higher your thoracotomy and the more tipped over you need the patient. So for a type II, you're going probably fifth intercostal space and the shoulders are at 90 degrees and the pelvis is 60 degrees. For type three, you're slightly further over, so you're down to about 60 degrees and you're perhaps in the seventh intercostal space. So the lower you go, then right up to being pretty much supine.

The left arm is put over top. You mark the scapular and you go into the chest and abdomen. I tend to go intraperitonealy into the abdomen, so I can see the bowel, although there are people who do a retroperitoneal approach. I tend to divide the diaphragm down to the central ligament and mark that. Then effectively you are looking at your clamp zone. You dissect around your clamp zone, both proximally and distally. We don't tend to expose the celiac and SMA in particular, or the right renal. It's important to find the left renal artery. For the visceral rotation, you effectively extend the left colon up, find the white line and get under that and get right onto psoas. The key is to get right down on the psoas early on, and then it is a relatively bloodless field. Everything has taken medially to leave the ureters out of the way. I always take for left kidney up. You will see in the textbooks and some advocate to keep the left kidney down. I that's utter madness to be honest and I can't see any point in it.

So for me, everything rotates up then out of the way. And it it looks a bit strange, like a postmortem. It looks very odd to see the whole left side abdominal cavity when all you've got is diaphragm muscle, psoas muscle and nothing else left. It does give you a very safe exposure then to the aorta. I don't tend to go round the aorta with slings, I think that can cause problems. I go around with my fingers and be very aware of the right sided lumbers or intercostals because you really don't want to cause a problem there because you can't easily get into the right chest from that exposure. So you have to be a bit careful so you can see exactly where you're getting round.

So in specifically with a type II TAAA, many people establish a left heart bypass. Do you always operate in that way and do you use a cardiac surgeon and the setting?

I think type II you are mandated to have some sort of adjunct to keep the rest of the body perfused while you're doing the proximal anastomosis. I think in an elective setting left heart bypass is the right way forward, because it does give you more flexibility and it allows you then to sequentially clamp and keep the legs, the viscera or the kidneys perfused. There are some negative aspects to left heart bypass, though, in terms of needing a much higher ACT and bleeding is the enemy here. The more you anti-coagulate, the more problems you've got. It's also not pulsatile flow and you're putting all your cells through a pump and you've got to be aware of increased problems around stroke and everything else that goes with that. I think it is very convenient for that set up, because you can get to the inferior pulmonary vein quite easily and femoral cannulation is quite easy for you to return and then you've got the option then of individually cannulating celiac and SMA for perfusion, whereas usually renals are cooled down.

If you've got a perfusionist there as well, you can get cold perfusate for the renal arteries. So it gives you more flexibility, and if things go wrong, you've then got an option to switch to conventional bypass if needed or even drain out and go to circulatory arrest and cool-down.

However, in a more emergency setting, an approach that tends to work quite well for us is to do an ax-fem approach. You use an ax-fem graft onto the right subclavian artery, not tunneled but on the outside of a body onto the right femoral and use the other limb as a single cannula, usually for the SMA, because the SMA is the king vessel with all of this. Most of the other things can go, but if you lose the SMA, you've got a dead patient. And that allows you to do an emergency procedure, a mid type III approach, without some of the problems associated with left heart bypass. It allows you to have a lower ACT; it gives you pulsatile flow; you're not smashing your cells up all the time; and it's relatively quick and straightforward to do. You don't have to mobilize a perfusionist and everything else that goes with that. It's worked well for us in emergency settings. It's better than a clamp and go approach because with a clamp and go approach, for the type III repair, you need to complete in 15 minutes, your inlay anastomosis at the top, the clock is already on. If you get down to the visceral segment and you've got to do some removal of thrombus to get your Carrel patch on or you've got to reimplant the left kidney separately, it's a lot of time pressure and there's no opportunity for something to go wrong there. That's where this gives you a bit more breathing space that you've got the lower body perfused while you're doing your proximal anastomosis.

7.2.3.1 Complications

As you've mentioned, open repair is a high tariff, high risk operation, but apart from death, what are the specific complications to the open approach? The big problems are bleeding and clamp times. Those are the things that cause problems for those patients who survive. So problems with large transfusion which cause problems with ventilation afterwards. The most common complications after a TAAA are pulmonary with 8.5% of one cohort requiring a tracheostomy. (Coselli et al. 2016) Renal ischemia is an issue of people will quote a 40 to 50 minute renal warm ischemic time, you increase your risk of renal failure. They may need to go on the hemofilter for temporary support, but these patients have already lost a lot of nephrons, so they haven't got a huge amount of capacity to lose more. That's why eGFR or high creatinine is a really important prognostic marker of outcome in this patients. I think it's like a barometer on their micro-circulation.

The big one and the Achilles heel of these approaches, whether it's endovascular or open is spinal cord ischemia and that is because of the segmental of blood supply to the cord comes from all the intercostals and lumbers. If you've got that aneurysm extending both proximally and distally involving the internal iliacs the left subclavian, then you're taking out your crucial collateral supply. Important aspects to reduce ischemia include:

- Getting the legs back in circuit early, whether that's removing the large sheaths, during a endovascular repair or getting the leg supply earlier in an open approach.
- Keeping the left subclavian in circuit.
- Maintain the mean arterial pressure in theater
- Place a spinal drain and ensure it is working.

Post TEVAR leg weakness needs to undergo urgent evaluation. Differential diagnosis includes spinal ischemia due to subclavian or intercostal artery coverage, distal embolization or stenosis of the TEVAR graft. Stenosis should be treated with angioplasty and extension.(Buth et al. 2007)

So the major issue is the spinal cord ischemia, can you explain in more detail the blood supply for the spinal cord and how this plays into the occurrence of spinal cord ischemia?

So it's complicated actually. There are anterior and posterior spinal arteries, but I think depending on what level you are in the cord, they get a blood supply from a more dominant approach. So higher up in the cord—cervical and higher thoracic—vertebral arteries are important, so hence the importance of a left subclavian bypass.

The lower you go, then they become a bit less important and the actual segmental arteries, intercostals, become dominant. Much is talked about the artery of Adamkiewicz. I think, yes, there may be a dominant vessel at about the level of the 9th-12th intercostal, but it's rarely one, absolutely single one that if you preserve that one, then the rest don't matter. I think they all matter.

And then the more distally you go, collateral supply from internal iliacs, median sacral, all those contribute to a collateral circulation that should be seen in some ways all connected.

And that is why, more recently in endovascular approaches, staging repair is really useful because you can allow the surviving arteries then to remodel and increase flow. It's why, during a type II repair, when you get the whole aorta open, it's very important to block the intercostals early—either by sewing them off or putting a little Pruitt balloon in and to stop losing blood from that collateral supply. It keeps the pressure in the spinal cord. Also the differential position of that blood supply is why patients then get a particularly prominant motor problems and sometimes preservation of sensation.

So purely for exam purposes and a bit of physiology. Can you remind us how to calculate the spinal cord perfusion?

All that is really is MAP minus the CSF pressure. So it's your mean arterial pressure, take out the CSF pressure and that's what you've got perfusing the spinal cord. It is a useful equation to remember because it allows you to decide what's the way to enhance spinal perfusion as much as possible, which is to increase MAP and decrease your CSF pressure. We do that by enhancing the mean arterial pressure as much as possible, and draining off CSF with a spinal drain to reduce your CSF pressure and allow more blood into the cord.

I think people perhaps get a bit hung up on spinal drainage and it's importance. It is just as important to maintain oxygenation and hemoglobin over 100 (which is equal to 10 g/dL in the USA), reduce blood loss and all those other things are as crucial. Spinal drainage is an adjunct, but it shouldn't be seen in isolation. It's one of many adjuncts to try and help prevent spinal ischemia.

So in regards to the mechanism for spinal cord ischemia during open thoracoabdominal aneurysm repair, do you think it's related the blood loss, clamp time or or implantation of the intercostals?

I think it's multifactorial. So I think the majority is due to ischemia scheme in terms of a watershed situation. Segments of a cord just not getting enough blood. And the very reason the more typical area around T7, 8, 9, to 10, is because it is in the middle between the vertebral artery and internal iliac artery supply. This is probably a kin to a splenic flexure of the colon, it's between two territories.

However, there is also potentially problems with microemboli or trashing of a spinal arteries, which can occur in some patients. It's a slightly confusing situation, but I think in terms of intraoperatively—blood loss, dumps in blood pressure, with hypotensive episodes really don't help.

When do you remiplant intercostal arteries?

Again, that is a bit confusing. Some people would argue if there's massive back bleeding, you don't need to be reimplant because they're getting collateral supply from somewhere.

I think it will depend on how the case is going. We do use motor evoked potentials (MEPs) but that's quite a specialist situation and you're very reliant on the readings you get. They can be confused by leg ischemia, cooling down or edema. But if you have already got the legs back in circuits and got the MAP up and things weren't improving and there was a sizable intercostals that could be reimplanted, well, then that's worth doing. And that's why the technique of using Pruitt balloon occlusion methods in pairs rather than ligating is good because you can then just reimplant those. If you don't need to implant them, you can just ligate and remove them. Whereas if you've already ligated them, you've got a problem because you've already destroyed the ostia.

We tend to leave a branch already attached in the mid thoracic aorta to save an anastomosis. You can, if you've got really good intercostals, fillet the branch to open it up and sew it on like a long patch longitudinally along the paired intercostals and then plumb it back on the other end so it's a circuit in series with your main aortic graft. That probably decreases the resistance to flow and that might remain patent for a bit longer, rather than having a big 10mm graft going into a single pair of intercostals with what are likely to have quite a high resistance, and they probably don't last as long.

Several spinal protection protocols have been described to reduce risk of spinal ischemia or mitigate the impact of spinal ischemia in patients who develop weakness.(Estrera et al. 2009; Yanase et al. 2012) An easy way to remember the important components are COPS:

- C Cerebral spine drain status keep for 7 days, pressure less than 5
- O Oxygen delivery supplemental oxygen, increase hgb, increase cardiac index
- PS Patient Status Keep BP greater than a MAP of 90mmHg

So do you think we're good at predicting preoperatively who may get spinal cord ischemia? You mentioned a few risk factors already that can predict who will or won't post-operatively.

I think we're utterly hopeless at it actually. I really don't think that there's a good method. You can see people having an infrarenal repair getting paralysis and you can completely replace the whole aorta from the arch down to the iliacs with prosthetic and have people walking around without a problem at all. So I don't really understand it and I think that comes into the business about whether it is about emboli, flow, or collaterals. I think it's a combination of things. When people started embolizing intercostals preoperatively prior to endo repairs, people thought this is a mad thing—you want to keep them perfused surely. But I think it does show the importance of maximizing the collateral supply, and I think some people have a better collateral supply than others. And of course, what we don't know is when you get down to the more microscopic level of actual arterial supply at cord level, what that is like in an individual patient that you can't see on imaging. If that is already compromised, I suspect those patients are more at risk than whether they have a macro vascular problem when things go wrong terms of hypotension or blood loss.

Many reports state that the highest risk factor for spinal cord ischemia includes length of aortic coverage or repair, followed by whether they have had a prior repair, preop hemoglobin, and intraoperative hypotension. (Bisdas et al. 2015)

We haven't really touched on complex endovascular repairs yet, but can you explain when you might choose a hybrid approach to this thoracoabdominal aneurysm repair?

So visceral hybrid operation came in really as a bridge between open surgery and endovascular. This is a bit dated now, because it is before custom devices and we only had thoracic stents at this time, the era of the Talent stent, which was a very early thoracic device made by Medtronic. Perhaps not the perfect name because it wasn't a particularly talented, it was a difficult thing to deploy, but it actually got us into the thoracic aorta. What we realized was that if you could operate in one cavity, such as the abdomen, but still excluding any aneurysm in the chest without cross camping and without rendering that patient ischemic over a large part of their body—they could withstand that hit much better than open cavity surgery with cross clamp and massive reperfusion.

So the concept of a visceral hybrid was to do an extra-anatomical bypass of the celiac, SMA and both renals from either the distal aorta or the iliac vessels. So patients would only get sequential ischemia of one organ at a time, say the right kidney or the celiac territory, and during that 10 minute anastomosis, the remainder of the body and organs were refused. And finally, at the end that, you could then put in a thoracic device as a freebie, because you'd already perfused the organs from distally. You'd ligated those target vessels and then you would just excluding the aneurysm so that you were transferring blood from above the repair to below the repair and perfusing the vital organs retrograde.

So it was very attractive and we started doing it quite early at St. Mary's and initially we had some very good results in the first 30 or so patients.(M. P. Jenkins et al. 2011) And as always, then it allowed you to consider older, sicker patients but it wasn't easy surgery. And actually I think because it was doable in terms of just an abdominal approach and relatively familiar territory for anyone who's done occlusive disease for the viscerals or renals, I think a number of centers started doing small numbers and not getting so good results. And then when better, custom made devices came in, I think it's purpose became less and less because why would you do that if you had an opportunity to do something which was even less invasive.

So what we've been left with now to consider a hybrid type approach effectively are those patients who are physiologically not fit enough for an open approach and those patients who are anatomically not suitable for a total endovascular approach. In some ways, they are the worst of both worlds in that outcome group. And so I don't think you'll ever be able to compare outcomes legitimately with the fitter group having a fully open approach and the anatomically suitable patients having a total endovascular.

But I must say we still see quite a lot in our multidisciplinary reviews coming back. What surprises me is the durability of those grafts. The visceral grafts remain patent for many years, some of them are 15 years now. Occasionally there will be an accessory or small diseased renal artery, and the renal graft will go down, but the others have been extremely durable. And even though we've had to re-line the stenting portion or extend it, the grafts have maintained really well. A unique situation that uses a similar technique is with mycotic aneurysms of the visceral segment of the aorta. The mainstay of treatment is a debranching of the aorta and debridement of the infected aorta with in situ revascularization or ax fem-fem bypass. There is an evolving role for FEVAR, such as with most aneurysmal disease.(Sörelius et al. 2016; Sule and Dharmaraj 2016)

So moving on to Type IV thoracoabdominal aneurysm repair. These can be repaired using endovascular or open techniques. Can you talk us through your approach to open Type IV repair in terms of exposure and any tips such as how to minimize the visceral ischemia?

So I think an open Type IV from a subcostal approach is a doable and durable procedure. And what I mean by that is that the hit to the patient is not enormous, and a reasonably fit patient can get through that reasonably well. You have to be a little bit selective, but not super selective. For example, in my first hundred, I had one 30 day death and one in hospital death, which was someone at about three months who just didn't get better, so that puts it into perspective that actually it's an operation with a very good outcome, if you get it right at the beginning.

The approach I mentioned earlier, nothing too special. This is a supine patient. Getting the break on the table in the right position to allow you to extend the abdomen. You lower both ends of a table to open up that segment of the abdomen and get better exposure. I tend to do the vast majority via a subcostal approach, some people would call it rooftop, then extending on the left side down below the costal margin a bit. And it gives you really good exposure even if you have to go down to the iliacs. It does cause a bit more of a problem getting to the right iliac bifurcation. The left is easy, because it's right there next to you. So that's an added complication.

The visceral rotation needs to be done carefully because what you do not want is to create blood loss right at beginning. And as I mentioned earlier, it's getting under the left colon along the white line onto psoas and then getting up under psoas and I always tell the registrars when they're doing it, you've got to go as far as possible on psoas from below. And when you think you've gone as far as possible, you've got to go another five centimeters. So you get right up to the diaphragm from below. And then you can get into the supra-colic compartment around the splenic flexure and get your fingers either side of the lienorenal ligament and take that off the lateral and the posterior abdominal wall. The more and more you do, the more you are freeing and the whole of the viscera comes up towards the right side of a patient and eventually they will rotate up and you've exposed the whole left wall of the aorta.

The first thing I then do is ensure that I've got a clamp zone, and I do that by dividing the crus which sometimes can be really tense at that level. The crus can act as an extrinsic wrap around the dilated aorta, to an extent that the distal end of the crus is almost ligamentous, causing a tight band on the aorta. I tend to get my finger underneath that before it with a diathermy and opening it, and then very carefully choosing the clamp zone and getting my fingers right round that area, so I know I've got a healthy clamp zone.

The next thing I then do is look for the left renal artery. I've already tended to make a decision about what I'm going to do with the left renal artery, which depends on the number of things. So it depends on, to some extent, the age of a patient, where the left renal is in terms of a clock face. So if you're looking at an axial cut to the CT, the more the left renal is around before three o'clock, up to two o'clock, the easier it is to incorporate because you've got less aortic tissue there. If the left renal is beyond 3:30 and posterior on the sidewall of the aorta or if the left renal is more distal compared to the right renal, then it's a long way away from the right renal artery and you're going be leaving a larger patch of aorta. This is usually bad news, unless you have a more elderly patient.

You're making this decision about whether you're going to try and incorporate the left renal artery into a big patch or whether you're going to reimplant it separately or perform a jump graft. So you've already thought about that, but it's crucial to find the left renal because when you've clamped and you've done your left aortotomy, you've got to get that aortotomy underneath and posterior to the left renal artery so that the orifice goes up with that segment of the aorta.

Then I tend to get the iliacs out. The left is easy, it's there. I have no qualms on the right by going back over through the peritoneum to find the right iliac artery to expose that. I don't go around any of them. I just find a very discrete camp zone. You don't need much, just enough to put a clamp on.

And then I tend to take the fat and lymphatic tissue off the left side of aorta. So when you open the aorta, you've not got a lot of immediate bleeding. If you've got a retro-aortic left renal vein, you've got a line gate that cause otherwise you'd be going through it. And surprisingly, on CT, the main left renal vein is anterior, but there is often a vein there which has stretched across, either a lumbar or phrenic vein, which you should deal with prior to your aortotomy.

It's about getting everything set up before you cross clamp, because once you cross clamp, the clock is on. So it's getting the whole team ready for that moment and making your anatomy as perfect as possible before you do that. Obviously it goes without saying that you have cell salvage and we use a Belmont with the big bucket on the top of it, so that you can rapidly give blood back when you need it.

Can you talk a little bit more about a Carrel patch and when would you choose to do a jump graft to the left renal rather than encompass all the vessel vessels onto that patch?

I tend to think of a Carrel patch more in a way with a type III repair, because with a type III repair, you've done your proximal anastomosis and then what you're doing is you're putting on a patch of aorta to include the celiac, SMA, and right renal, which tend to be together, whereas sometimes the left renal is a bit away. The more tissue you leave there, the more chance that aortic tissue will dilate to over time, hence the patch aneurysm. If your patient has got some form of connective tissue disease, we reimplant directly.(De Rango et al. 2011; Afifi et al. 2017) There is a very nice graft available with four side bunches already put on and you can go straight onto the ostia. So you're basically leaving no aortic tissue at all.

For type IV, you are doing an oblique anastomosis and you have to be aware that the right side of the aorta is really left there. It's a compromise, these patients with atherosclerotic type IV aneurysms, they may be in their sixties, seventies or even into their eighties. Even the good candidates, they're not going to live 25-30 years, and therefore over time that segment may begin to dilate a bit. I think that is acceptable. Half of that segment is Dacron and the other bit may dilate.

I think if you've got a Carrel patch, which really balloons out, that's something which is different. We don't have good data as to when those will give away and rupture. I think what is clear is that if you've got dehiscence between your suture line and your graft, that is a dangerous situation, akin to a false aneurysm. And although it may look similar on CT, that's a different beast and they're at risk at any stage. True aneurysmal dilation of a segment that is left behind is a bit different and I think that's a bit safer. In essence, what you're trying to do is a compromise between leaving as little aortic tissue as possible, but also getting the patient alive through their first repair safely.

If I know I'm not going to incorporate a left renal, then I will sew on a six or eight millimeter side branch onto the tube golf before I cross clamp, because that saves you one anastomosis. And then now my favorite technique is just to amputate the left renal artery and sew that directly onto your branch and leave that in a bit of a lazy C configuration in the paracolic gutter, so as the rotation comes back, it's not under tension. The other technique is to have very short graft, right from your aortic prosthetic, but you've got to really judge how that will sit when the kidney is back in its anatomic position. So it can't come off the anterior aspect of the tube graft. It needs to be further down and that's awkward to judge and I think we're probably not good at that so the lazy C is a better approach.

I much prefer that to trying to either reimplant the native artery directly onto the Dacron, as they tend to restenose or try to bring the aortic sidewall down on to the Dacron. I think that's fraught with difficulty. You've often got calcification at the ostium, and it's also very difficult because you've got the rest of your anastomosis really close and sometimes not enough room to bring your clamp down, which is exactly what you want to do. The left kidney will take a bit of an extra ischemic hit, but you want to be perfuse the right renal and viscera. You've got to have to bring your camp down below your proximal anastomosis.

When it comes to late complications from TAAA repair, 8% are from progression of aortic disease and 3% are related to the graft repair. Risk factors associated with late events include female sex, partial aneurysm resection, expansion of remaining native aorta or initial aneurysm rupture. (Clouse et al. 2003)

Have you ever had to treat the Carrel patch aneurysm in your career?

We have a number from previous Type III repairs and I think these are very suitable for fenestrated or branched devices. You've got a beautiful lending zone, proximally and distally. Why not? These tend to be well-suited to that and it's a much easier approach mixing endo and open, changing from one to the other, I think that's fine. It's just at a different stage of a patient's life and repair

Any final summary or closing words about aortic surgery?

We have a bit of a tipping point with open aortic surgery at the moment, certainly in the UK. Looking at registry data, many are tipping back to open now. Whether that was as a result of NICE guidelines, I suspect not, perhaps that just put it more into focus. I think many people are seeing more problems with patients surviving after what was very good endovascular surgery at the time, but they've outlived the repair. Now we've got problems with patients where you got to wonder whether that wasn't perfect anatomy and should they have had open surgery?

Now the difficulty we've got at the moment is the patients that perhaps weren't that suited for endovascular repair, did well initially from it and they are paying the price later on. But those very patients are the ones that are also more difficult from an open approach. They are patients with shorter, more tortuous necks or more calcification. There is a concern that there is an era of surgeons now that haven't done as much open aortic surgery, who may not be as confident with adverse anatomy.

That lack of confidence influences your decision-making, so patients don't get a full appraisal of what options are available. The default then may be to go with a less than perfect endovascular option. I agree, this may be less easy to kill a patient upfront with an endo option but you perhaps pay for that later on. This is a very difficult thing to get around and we've got to get to a compromise of finding the patients with the likely better life expectancy and sending them to groups with a larger open practice and get them through a big operation safely and capitalize on those benefits.

The really lucky ones are the ones that are anatomically really suited for endovascular repair and their sack really shrinks and their seal zones remain good. They are the best of both worlds as they tend to also be fitter with less extensive disease and get the benefits from a less invasive approach. Not all patients are the same and we've got to now accept that we've gone down the line of working with industry, which they didn't necessarily focus on durability. What they wanted to do is to extend the applicability, to treat more patients and get better, lower profile devices to go for these percutaneous cardiology market in the United States. And we're now paying the price of some of those decisions, so we've got to reboot and take a stance on which patients are going to actually benefit from which approach and have a better evaluation of that going forward.

8 Aortopathies

Authors: Anna Ohlsson and Sherene Shalhub

Contributor: Wen Kawaji

8.1 General Considerations

What are the common genetic aortopathies?

There are several well-known genetic disorders which account for genetic aortopathies. The most well-known are Marfan syndrome, Loeys-Dietz Syndrome, and Vascular Ehlers-Danlos Syndrome.

There are less commonly known ones such as Familial Thoracic Aortic Aneurysms and Dissections due to pathogenic variants smooth muscle cells genes such as *ACTA2*. There are others in which the causative gene is not known.(Black 2019)

Why are they such a big deal?

These are cases in which the building blocks of the aortic wall are defective. What I mean by this, is that these patients have pathogenic variants in the genes that affect cell signaling or smooth muscle cell structure that lead to suboptimal composition of the aortic wall. These alterations ultimately lead to cystic medial necrosis in the aortic wall.

As such they are at more risk for aortic aneurysms and dissections that can lead to the premature death of the patient.

To put the frequency in perspective, Marfan syndrome occurs in 1:5000 of the population while Vascular Ehlers-Danlos syndrome (also known as VEDS) occur in 1:50000 of the population.

Let's dive into them then – what are the defining features of each and the high yield information?

The high yield information is being able to pair the genetic syndrome and phenotype with its associated genetic mutation. A useful exercise following this broadcast is to list the disorders in a table and write out their associated gene mutation, what protein defect or deficit occurs, the typical phenotype, and the common vascular pathology associated.

But before we dive in, I want you to keep in mind some of shared features. One is that the associated aneurysms and dissections tend to occur at younger ages and dissect at lower blood pressures than what we see with sporadic dissections (these are the dissections that are not familial or associated with a syndrome)

One is that these are inherited in an autosomal dominant matter but there can be variation in how the pathogenic variants are expressed among affected people and even within families. The other thing to remember, is that in roughly half of these cases, the affected patient is the first in their family to have a given pathogenic mutation. The flip side of this, is in half the cases, there is a family history of aortic aneurysms, dissections, and sudden death.

8.2 Demographics

We will start with Marfan syndrome.

<u>Marfan syndrome</u> is caused by pathogenic variants in the FBN1 gene (also known as fibrillin-1 gene). These variants lead to improper formation of the microfibrils that maintain elastin, a key component of the arterial wall.

These patients are prone to aneurysmal degeneration and dissections of the aortic root but can also dissect the descending thoracic aorta. They commonly have lens dislocations (ectopia lentis). They have common skeletal features such as being tall, thin, with long arms and legs, scoliosis, pectus deformities (carnitatum or excavatum), and club feet. They can also have a history of spontaneous pneumothorax and mitral valve prolapse.

How is Marfan syndrome similar or different from the other genetically triggered aortopathies that you mentioned?

Loeys Dietz Syndrome is similar to Marfan syndrome in all the features including the aortic root aneurysms. They don't seem to have lens dislocation and they have other unique features such as bifid uvula or cleft palate, and hypertelorism (which is an abnormally increased distance between the eyes). What is different about Loeys Dietz Syndrome from Marfan syndrome is that they can have arterial aneurysms of other arteries instead of the aorta, such as the SMA, axillary, or other peripheral arteries.

<u>Vascular Ehlers-Danlos syndrome</u> has some shared features to Marfan Syndrome with both such as spontaneous pneumothorax, but these patients tend to be short and can have easy bruising. They also have similar features to Loeys Dietz syndrome in terms of arterial aneurysms. Common features of VEDS would be thin translucent skin where you can easily see their veins, thin lips, thin bridge of the nose, large eyes, easy bruising, acrogeria – or an aged appearance of the hands

However, unlike Marfan and Loeys-Dietz, the majority of VEDS patients tend to not have aortic root aneurysms. One thing to remember about VEDS is that it is a *subtype* of Ehlers Danlos syndromes. It's very important to distinguish it from the other subtypes because most of the other 12 Ehlers-Danlos syndromes are not associated with arterial pathology. So people with vascular EDS are prone to arterial, uterine, and intestinal rupture and their average lifespan is 48 due to these highly morbid pathologies. 25% of patients with vEDS will have experienced some clinical manifestation by age 20, and that number is close to 90% by age 40.

I remember learning about classic Ehlers-Danlos presenting with hypermobile skin and joints. Is this something you see with Vascular EDS as well?

Patient's with vEDS don't have the same hypermobile skin or joint laxity as we classically think of with classic Ehlers-Danlos. In fact, some vEDS patients report losing confidence in their physicians who ask them about joint and skin hypermobility because it suggests to them that their doctor doesn't know about their disease process. These patients often know more than most of the doctors they meet about their condition, and it's a source of constant frustration for them. It can also be a problem if the severity of the disease is underestimated, as we discussed they can present much younger than most patients with highly morbid issues – like arterial rupture.

You mentioned arterial pathology in Loeys-Dietz and vEDS. Can you tell me more about that?

In both types you can see subclavian, carotid, SMA, and iliac artery aneurysms and dissections, as well as less frequently vertebral, SFA, and popliteal aneurysms and dissections.

8.3 Evaluation

How do you diagnose these genetically triggered aortopathies?

There are clinical diagnostic criteria for each, but ultimately genetic and laboratory testing is very important for the final diagnosis.

Ghent criteria is used to clinically diagnose <u>Marfan syndrome</u>. The big ones are aortic root dilation, known family history of Marfan's or not, the diagnosis of ectopia lentis which clinically is manifested as iridodonesis (lens shimmering). Additionally, genetic testing for pathogenic FBN1 variants is also diagnostic.

To date, there are 5 types of <u>Loeys-Dietz</u> as of last check. These are due to pathogenic variants in the TGF-B signaling pathway, such as TGF-beta receptors and SMAD3 genes.

<u>Vascular EDS</u> is caused by a mutation in the COL3A1 gene which encodes a defective type of III procollagen. The defect in the procollagen makes it unable to properly fold into a triple helix that forms the normal collagen structure. This causes the defective procollagen to be degraded intracellularly and as a result there is an overall deficit in type III collagen which is an important component of arterial walls and other structures. The confirmatory test for VEDS is collagen testing which can confirm the collagen III defect.

8.4 Management

8.4.1 Medical Management

How would you manage these patients?

Medical optimization and surveillance is key to try to extend the time as much as possible before they get a dissection and avoid it if at all possible.

We start with lifestyle modification. Avoid "burst" exertions such as sprinting and weightlifting. Anything that very strenuous. That's not to say that they shouldn't exercise. Light exercise is encouraged, but this would be activities like light jogging, swimming laps, or biking.

In order to minimize aortic shear stress, a resting heart rate of under 70 beats per minute and an exercising heart rate under 100 should be the goal. This can be accomplished with beta blockers. Propranolol has been shown to significantly decrease the rate of aortic growth in Marfans patients with a baseline aortic root diameter under 4cm. There is research into the use of Losartan in murine models that suggests it inhibits TGF-beta in the aortic wall, which is an important pathway that contributes to the breakdown of the wall. However, randomized controlled studies have failed to show an increased benefit of Losartan over beta blockers in Marfans patients. ACE inhibitors are also being tested and are shown to decrease the risk of type b aortic dissection over 6 years.

In vascular EDS instead of propranolol, celiprolol has been studied by the French and shown to reduce vascular rupture from 50 down to 20% in vEDS, although the mechanism of this is not yet clear and does not appear to be necessarily the same as decreasing shear stress as in Marfan's syndrome.(Ong et al. 2010) In general taking care of these patients involves trying to minimize complications from procedures and interventions. For instance, use ultrasound for any line that is necessary and avoid arterial lines, intramuscular injections, or other invasive lines if possible to minimize the chance of a complication. Patients are advised to wear medical bracelets notifying that they have vEDS.

We also discuss the importance of forming a care team based on their needs. This usually includes a cardiologist, a cardiac surgeon, a vascular surgeon, and a primary care physician.

8.4.2 Surgery

What about surgical treatment for those who need it?

For patients with Marfan's, prophylactic surgery is recommended for a ortic root dilation >5cm or thoracic aorta >5.5cm. Often times the thoracic and abdominal aorta are involved. Open and endovascular surgery are options for these patients. Open procedures often include open thoracoabdominal aortic aneurysm repairs, open cardiac surgery for arch replacement, or cervical debranching procedures. Endovascular procedures can include regular TEVARs or branched TEVARs which require extensive aortic coverage.

Open surgery can be well tolerated and is ideal in the sense that you can replace the entire aorta which avoids the future complications from continued aneurysmal degeneration, loss of proximal or distal seal zones, or device issues that can plague endovascular methods. However open surgery, of course caries higher complication risk and morbidity up front and does share some complications with endovascular treatments as well.

Sometimes these patients will have hybrid procedures and often their care will require multiple surgery teams including cardiac and vascular surgery. An important thing to be up front with all of these patients about is that this is a long term relationship with their surgeon, as they often require multiple staged procedures, things aren't fixed in one procedure, and even after they have been surgically addressed there is a lifetime of maintenance and surveillance.

Ultimately, the decision for open vs. endovascular approaches will vary between patients based on their specific anatomy and arterial issues, what their body can tolerate, and ultimately what their goals of care are.(Lum, Brooke, Arnaoutakis, et al. 2012) Some may require having their entire aorta replaced, while others may only need ongoing medical therapy and surveillance and it's important to set expectations early.

What about VEDS, when surgery cannot be avoided? How do you mitigate the risk of complications?

The tissue is very fragile. So using instruments that are the least traumatic is key – like fogarty clamps for vessels. Sutures often must be pledgeted to reinforce them. Leave no tension on anastomoses or suture lines. Always keep a backup plan in mind– when arteries cannot be repaired, can they safely be ligated or embolized. Generally any large bore access for endovascular treatment is avoided because access site complications are high and can lead to devastating consequences. Some have described a technique of open exposure and buttressed repair, when endovascular approaches are required.(Black 2019)

In situations of extremis, like a rupture, these patients' tissues have been known to completely breakdown. This is often their first presentation of their aortopathy so there is no time for previously discussed work up or considerations. Try to avoid the worst case scenario, but of course sometimes it's the only option left to get out of the OR. Be upfront with the patient about how complications may arise, set expectations, and think about goals of care early.

Some specific considerations for TEVAR in these patients is that it may be important to have cardiac surgery readily available because they are high risk for retrograde dissection. Also, avoid aggressive oversizing, consider not ballooning if no leak is identified and avoid excessive wire manipulation.

8.5 Summary

We discussed earlier that these aortopathies can have shared phenotypic characteristics, some of which can be used in a clinical diagnosis, but are all of these genetic aortopathies syndromic?

Let's start by saying that all patients with Marfan syndrome and VEDS can have the syndromic features we just talked about. However, it's not always the case and the absence of these feature does not exclude the diagnosis. In fact, we recently treated a middle-aged woman with an aortic dissection who had Marfan Syndrome confirmed with genetic testing. She had been diagnosed prior to her dissection because her daughter had undergone genetic testing. However, on meeting her, I would not have guessed she had Marfan Syndrome, had I not known. She was average height, obese, and had no other relevant physical findings on exam or history.

This ties into another genetic aortopathy that we have not discussed yet which are the familial thoracic aortic aneurysms and dissections. They do not have any syndromic features. For example, patients with ACTA2 pathogenic variants that cause alpha actin mutations which again contribute to degeneration of the arterial wall. These patients tend to present 10 years younger than sporadic thoracic aortic aneurysms, generally in their late 50s compared to late 60s, and women seem to be less often effected than men.

Dr. Shalhub, I know vascular genetics is one of your passions. Is there anything else you want people to remember from this broadcast?

Don't forget the family. Once you've made the diagnosis in one of them, remember it is autosomal dominant, so it's important to make sure the family understands and that they are set up with the appropriate care team and monitoring. They may not all develop the same medical issues, however as we discussed, ongoing medical management and lifestyle changes are the key.

8.6 Additional Resources

VEDS Research Collaborative study:

- https://www.vedscollaborative.org/get-involved
- https://depts.washington.edu/vedscoll/

9 Trauma - Endo

Authors: Kevin Kniery, Marlin "Wayne" Causey, Nakia Sarad, and Todd Rasmussen

These trauma episodes were developed in collaboration with Behind the Knife: The Premier Surgery Podcast. In addition, Dr. Causey is a co-host for WarDocs: Military Medicine Podcast.

Vascular trauma requires close collaboration between vascular surgeons and trauma surgeons. The decision of which specialty should manage which injuries vary across different centers. Through these trauma chapters we will discuss what management decisions the majority of vascular surgeons should be comfortable managing as a part of a multidisciplinary trauma team.

? Take a Listen

Check our debate between leading trauma and vascular surgeons about how to best develop a collaborative team to manage vascular trauma.

For relevant images and a more in-depth discussion of this topic, please review Chapter 11: Resuscitative Endovascular Balloon Occlusion of the Aorta(Yi, Fox, and Moore 2022) and Chapter 17: Blunt Thoracic Aortic Injury(Demetriades, Talving, and Inaba 2022) in Dr. Rasmussen's 4th Edition of *Rich's Vascular Trauma*.

Although open surgical treatment is the mainstay of trauma management, there are some unique injuries where endovascular management has become the standard of care in centers with the resources and expertise to perform it.(Katsanos et al. 2009)

9.1 Blunt Thoracic Aortic Injury (BTAI)

9.1.1 Demographics

9.1.1.1 Guidelines

SVS Guidelines for Blunt Thoracic Aortic Injury (W. A. Lee et al. 2011)

9.1.1.2 Epidemiology

- Rare, but let hal. Less than 1% of all blunt traumas but second leading cause of death in blunt trauma.
- Due to high velocity mechanism of injury, usually are polytrauma patients with other severe injuries (i.e. intracranial injury)
- Common mechanisms
 - High-speed motor vehicle collision
 - Motorcycle collisions
 - Pedestrian vs. motor vehicle
 - Falls from heights
- Most patients have unsurvivable devastating injuries (75%) and pass prior to arrival to hospital for definitive care
- Rates of paraplegia, rates of blood utilization, and mortality rates have decreased with prominence of endovascular repair of injuries

? Further Reading

Seminal Papers in Blunt Thoracic Aortic Injury (BTAI) outlining the evolution of management from open to endovascular treatment.

- Fabian et al. (1997). Prospective study of blunt aortic injury: Multicenter Trial of the American Association for the Surgery of Trauma. The Journal of Trauma, 42(3), 374–380(T. C. Fabian et al. 1997)
- Demetriades et al. (2008). Diagnosis and treatment of blunt thoracic aortic injuries: Changing perspectives. The Journal of Trauma, 64(6), 1415–1418(Demetriades et al. 2008)
- Alarhayem et al. (2021). Timing of repair of blunt thoracic aortic injuries in the thoracic endovascular aortic repair era. Journal of Vascular Surgery, 73(3), 896–902(Alarhayem et al. 2021)

9.1.1.3 Anatomy

- Aortic arch » relatively mobile
 - Not common location for blunt injury
- Descending Thoracic aorta
 - Tethered
 - Junction just distal to left subclavian artery is where most injuries occur (within $0.5\text{-}2\mathrm{cm})$
 - Tear injury due to deceleration

9.1.1.4 Grading Systems

SVS BTAI Grading System(W. A. Lee et al. 2011)

- Grade I: Intimal Tear
- Grade II: Intramural Hematoma
- Grade III: Pseudoaneurysm
- Grade IV: Rupture

Harborview Grading System(Heneghan et al. 2016; Starnes et al. 2012; Quiroga et al. 2019) - Classification scheme helps dictate which patients can be managed operatively vs. non-operatively

- Minimal Injury
 - Absence of a rtic external contour abnormality
 - Intimal tear and/or thrombus of <10mm in length or width
 - Treatment: No intervention and surveillance with follow-up imaging
 - Antiplatelet therapy
 - CTA at least within 30 days of injury, progression is rare (5-7%) and often identified early(Osgood et al. 2014)
- Moderate Injury
 - External contour abnormality, includes pseudoaneurysms
 - Intimal tear >10mm in length or width

- Treatment: Delayed repair, semi-elective
 - Stabilization of concomitant injuries(Azizzadeh et al. 2009)
 - Impulse control
 - Short-acting beta blocker (i.e. esmolol)
 - SBP < 120 (coordinate with neurosurgery for TBI patients)
 - Repair within 1 week (TEVAR)
 - For those patients with severe TBI who are unable to undergo anti-impulse therapy due to maintenance of cerebral perfusion pressure, will recommend repair sooner
- Repeat CTA chest within 5-7 days for follow-up
- Severe Injury
 - Active extravasation (free contrast extravasation or hemothorax at thoracotomy)
 - Hemodynamic instability
 - Left subclavian artery (LSA) hematoma > 15 mm
 - Treatment: Emergent repair
 - BTAI takes first priority

🖡 Take a Listen

Check the BTK episode with Dr. Benjamin Starnes discussing blunt thoracic aortic trauma and this classification scheme.

9.1.2 Evaluation

- Be aware that patients likely have other injuries
- Recommend CTA chest, abdomen, pelvis
- Perform CTA follow-through to upper/lower extremities if those areas are of concern for injury

9.1.3 Management

Overview

- Endovascular repair of BTAI has better outcomes than Open repair
- Need to consider that most TEVAR devices are designed for aneurysmal disease
 - Generally require a larger diameter and their compliance is not ideal for these smaller/normal caliber aortic arch sizes
 - Average size of aorta proximal to injury site is 19mm (smallest devices are at 22mm)
 - Previously, iliac devices used to treat aortic injuries
- There are new advanced TEVAR devices that are available for non-aneurysmal disease with normal caliber aortic sizes.
 - ex: GORE TAG Conformable Thoracic Stent Graft with ACTIVE CONTROL System
- Recommend heparinization if not contraindicated
 - i.e. TBI, femoral shaft fracture with hematoma, solid organ injury, etc.
 - TEVAR can still be performed safely without heparin
 - Higher threshold for heparinization in COVID+ patients due to their hypercoagulable state
 - Goal Activated Clotting Time (ACT) > 250 Ensure that ACT machine is available prior to start of case

Methods of Endograft Size Measurement

- CTA imaging
 - Used to help measure aortic size
 - Need to consider that patient is usually hypovolemic, constricted, and young (so aorta more compliant/elastic) true aortic size may be under-represented, 40% discrepancy between pre- and post-resuscitation, on imaging when choosing the appropriate device for repair
 - Recommend over-sizing stent graft by 20-25% from original measurement
- Intravascular Ultrasound (IVUS)
 - During angiogram, can confirm or redirect sizing in real-time during systole(Azizzadeh et al. 2011; Shi et al. 2015)

Access

- If iliac size is >7-8 mm, can use transfermoral access
- Small iliacs (< 6 mm) may need conduits
 - Recommend retroperitoneal approach for exposing common femoral and/or external iliac and using a Dacron conduit(F. J. Criado 2007)
- If iliacs are diseased (i.e. calcium burden), use of conduits would be difficult
 - Can consider putting mineral oil over sheath to help pass and deliver the endograft
- Femoral cutdowns considered to ensure safe delivery and removal of device

Positioning of Device

- Most important aspect is to cover the proximal defect and seal the entry point
- Average distance between the injury and the LSA is approximately 5.8mm (located at aortic isthmus)
- Recommend 2cm coverage proximally of healthy vessel and 2cm coverage distally
 - With these recommendations, approximately 40% of TEVAR repairs cover the LSA
 - Preemptive revascularization of LSA is not necessarily required, most patients do fine without arm claudication
 - Complications to consider when covering LSA
 - Stroke
 - If there is concern for vertebral perfusion, LSA must be revascularized prior to covering
 - Left vertebral artery may be dominant or significant disease in right vertebral artery
 - Arm claudication and ischemia
 - Spinal cord ischemia
 - Rare
 - Collateral pathways between LSA and IMA (internal mammary artery) can be compromised
 - Spinal cord ischemia can present at end of case or delayed 24-72 hours post-operatively

- Spinal cord drainage not routinely done to prevent paraplegia unless there is concern if abnormal neuro-exam.
 - If abnormal neuro exam at the end of the case, elevate blood pressure to improve perfusion and consider placing a drain at that time
 - There is high risk of placing and monitoring drains if not needed
- Relative contraindications to covering LSA without revascularization
 - Dominant Left Vertebral Artery perfusion to brain
 - Right Vertebral Artery atherosclerotic disease
 - Hemodialysis access on ipsilateral arm
 - Prior CABG with LIMA (left internal mammary) to LAD (left anterior descending artery)
- Some devices (i.e. GORE conformable as above) allow control of first stent ring to better conform/lay down the stent within the arch

Dealing with Malpositioned Devices

- Can deploy too proximal or too distal
- "Bird Beaking" Where the lesser curve of the graft is not well approximated to the lesser curve of the aorta (i.e. the lifting of the inferior aspect of the most proximal endograft with the first covered stent ring now lifted off into the inner curve of the aortic arch)(Frohlich et al. 2020) This can cause endoleaks. Methods to avoid "bird beaking"
 - Use stiff wire to get into the aortic root (i.e. Amplatz or Lunderquist wires) with forward pressure to have a constant tension
 - Wire should be snug against the greater curvature of the aorta
 - Can consider taking the graft over the origin of the LSA
 - Allows more room in the flatter inner arch to lay the graft down

Timing of Repair

- SVS Grade 3 or Harborview Classification of Moderate Injury
 - Delayed repair 24-48 hours after injury is acceptable and there is a survivable benefit(Alarhayem et al. 2021)
 - Some delays can be up to 5-7 days with medical management
 - Allows time for operative planning and optimization with anti-impulse therapy/resuscitation

- Also enables evaluation of any underlying infectious process that may compromise and infect the endograft
- SVS Grade 4 or Evolving SVS Grade 3/Harborview Moderate injuries
 - More urgent repair (within 24 hours)

9.2 Blunt Abdominal Aortic Injuries

9.2.1 Demographics

9.2.1.1 Epidemiology

- Very Rare
- Commonly associated with polytrauma patients, especially with concomitant blunt thoracic aortic injuries
- Similar mechanisms of injury as BTAI

9.2.1.2 Anatomy

- Zone 1: Diaphragm to SMA (superior mesenteric artery)
- Zone 2: SMA to renal arteries
- Zone 3: Inferior renal artery to aortic bifurcation

9.2.2 Evaluation

- CTA
- IVUS Also used for operative planning and evaluating extent of defects into branches of the mesentery

9.2.3 Management

Overview

- Zone 1 and 3 can be repaired endovascularly (above the SMA and below the renal arteries)
- Can cover the celiac artery safely if there is no obvious compromise of the other mesenteric arteries (i.e. inferior mesenteric artery, internal iliac arteries, prior gastrectomy or other complex intra-abdominal operation)(Banno et al. 2020)
- Zone 2 injuries typically repaired with open approach
- Endovascular approach can minimize risk of contamination if there is bowel spillage etc.
- Want to use minimum amount of endograft that will appropriately cover injury 2cm coverage proximally and 2 cm coverage distally
- Injury can sometimes result in occlusion, which can also be treated endovascularly.

Types of Endografts Used - Can use combination of all grafts types to get the appropriate coverage

- TEVARs
- Covered stents
- Aortic cuffs

9.3 Blunt Axillo-Subclavian Injuries

9.3.1 Demographics

Epidemiology

- Very rare, 9% of all vascular trauma
- Low incidence due to protection of vessels from surrounding bony structures and high degree of exsanguination with devastating injuries
- Endovascular repair preferred due to location of these injuries and highly morbid open repair due to central hemorrhage and risk of brachial plexus injury.(Branco et al. 2016; DuBose et al. 2012)
- Difficult to access via open methods with proximal/distal control due to protective anatomic nature of overlying skeleton

Anatomy

- Thoracic Outlet
 - Axillo-subclavian vein located anterior to anterior scalene and runs between the first rib and clavicle through the subclavius muscle/tendon
 - Axillo-subclavian artery is located in the arterial triangle
 - More lateral than the axillo-subclavian vein
 - Courses posteriorly to anterior scalene

9.3.2 Management

Access

- Transbrachial or transradial access preferred
- Can combine transbrachial and transfemoral
 - Provides more options for delivering stent grafts with difficult aortic arches (anterograde and retrograde approach) via a through wire
- With transbrachial access, brachial cutdown can be more reliable in trauma setting
- Can also consider HYBRID approach
 - Gain proximal and distal control via through wires from transbrachial/transradial and then perform the open repair
 - Can perform coil embolization of subclavian arteries to gain control
 - Caution to vertebral perfusion
 - Stent graft to temporize injury and maintain in-line flow, then have interval axillary artery repair with a bypass (within a week to a month later)

Types of Endografts used

- Self-expanding stents (i.e. Viabahn)
 - Traditional approach
 - May provide greater flexibility important for mid-subclavian penetrating trauam due to mobility of this region. Stents in the axilla have particularly poor patency for this reason.(Chopra et al. 2016; Gray et al. 2017; Shalhub et al. 2011)
 - Helpful for more distal injuries
- Balloon-expandable stents

- Newer devices, also flexible
- Has more precise landing (especially if injury is near branch points)
Part IV

Abdomen

10 AAA

Authors: Mia Miller and Julie Duke

10.1 Demographics

What is an abdominal aortic aneurysm (AAA)? (Moore, Lawrence, and Oderich 2019)

- Defined as a localized dilation of an artery to a diameter greater than 50% (1.5x) of its normal diameter. It is generally accepted that >3cm in adults is considered aneurysmal for the abdominal aorta.
- AAAs can be described as:
 - Infrarenal distal to the renal arteries with normal aorta between the renal arteries and the aneurysm origin.
 - Juxtarenal aneurysm extends to the renal arteries but does not involve them
 - Pararenal aneurysm involving the origin of at least one of the renal arteries
- Estimated 1.1 million Americans have AAAs, which equates to a prevalence of 1.4% in 50-84 year old general population.
 - Men 65-75y with a history of smoking have a prevalence of 6-7% hence the need for screening.(LeFevre and Force 2014; Chaikof, Dalman, Eskandari, Jackson, Lee, Mansour, Mastracci, Mell, Murad, Nguyen, and al. 2018)
 - Men without a history of smoking have a prevalence of 2%.
 - Women who are current smokers 2%, and women with a history of smoking 0.8%.(Wanhainen et al. 2006)
- AAAs are 3-7x more prevalent than thoracic aortic aneurysms and can co-exist with other aneurysms throughout the arterial vascular system like popliteal artery aneurysms.
 - In a 10-year review originating from Ireland, 50% of patients that presented with unilateral popliteal artery aneurysms had associated AAA. In patients with bilateral popliteal aneurysms, 63% of those had associated AAA. (Duffy et al. 1998)

- Patients with femoral aneurysms have an 85% chance of an associated AAA. (Diwan et al. 2000a)
- Conversely, if a patient is first found to have a AAA, there is an 11% chance of having associated popliteal artery aneurysms >15mm. (Tuveson, Löfdahl, and Hultgren 2016)
- Another study showed a rate of femoral-popliteal aneurysms in AAA patients is approximately 14%. (Diwan et al. 2000a)
- This association stresses the importance of a good physical exam when evaluating a patient with a AAA and is commonly tested on exams.(Chaikof, Dalman, Eskandari, Jackson, Lee, Mansour, Mastracci, Mell, Murad, Nguyen, and al. 2018; Diwan et al. 2000b)

10.1.1 Etiology

What is the pathogenesis of an abdominal aortic aneurysm? (Moore, Lawrence, and Oderich 2019)

- More than 90% of AAAs are associated with a therosclerosis.
- Other causes include cystic medial necrosis, dissection, Marfan's syndrome, Ehlers-Danlos syndrome, Loeys-Dietz, HIV and syphilis.
- Elastin and collagen are the major structural proteins responsible for the integrity of the aortic wall and defects in these cause degeneration and further aneurysmal change.
- For example, a mutation in fibrillin in Marfan's syndrome causes elastin fragmentation and pathological remodeling of the wall of the artery to form cystic medial degeneration.
- Several investigations have also shown that upregulation of metalloproteinase activity, specifically MMP-2 and MMP-9, have an essential role in aneurysm formation. Imbalances between aortic wall proteases and antiproteases cause degradation of the extracellular matrix and loss of structural integrity of the aortic wall.
 - Nicotine increases MMP expression in vascular smooth muscle cells.(Z.-Z. Li and Dai 2012; Rabkin 2016)
- Increased thrombus burden is associated with wall thinning, medial loss of smooth muscle cells, elastin degradation, adventitial inflammation and aortic wall hypoxia which all increase the rate of AAA growth.
 - Of note, intraluminal thrombus rarely embolizes and is not an indication for anticoagulation.(Singh et al. 2019; Cameron, Russell, and Owens 2018)

- Infective aortitis is a rare but highly morbid cause for aortic aneurysm formation. Treatment in this setting should be more aggressive with the addition of antibiotic therapy and open surgical repair with rifampin soaked graft whenever feasible.(Gornik and Creager 2008; Paravastu et al. 2009; T. E. Rasmussen and Hallett 1997)
- For more about a rtitis, see Section 3.3.5

What are the risk factors for AAA occurrence and growth? (Moore, Lawrence, and Oderich 2019)

- Risk factors for AAAs are similar to the risk factors for occlusive atherosclerosis and include age, tobacco use, hypertension, male gender and hyperlipidemia.
- Cigarette smoking is the single most important modifiable risk factor to prevent occurrence and growth of AAAs. Smoking increases the rate of growth by 35% for abdominal aortic aneurysms.
 - Smoking has a dose dependent relationship with AAA, smoking 1/2 ppd has one third the risk of smoking 1 ppd.(Kent et al. 2010)
- It has been found that diabetes is protective for AAA progression and rupture.
- Men more likely to have AAA, but women develop at a later age with more aggressive growth and risk for rupture. (Lo and Schermerhorn 2016a)
 - Stronger association with smoking in women.
 - Women have same outcomes from endo repair, higher complications from open repair due to poorly managed CV risk factors.
- Fluoroquinolones(C.-C. Lee et al. 2015, 2018; LeMaire et al. 2018; Pasternak, Inghammar, and Svanström 2018) - In a recent study just published in JAMA Surgery in January 2021, the group at UNC showed an increased short-term risk of developing an aortic aneurysm with fluoroquinolone use. (Newton et al. 2021)
 - They reviewed all prescription fills for fluoroquinolones or comparative antibiotics from 2005-2017.
 - This included >27 million US Adults aged 18-64 years old with no history of aneurysms.
 - 18% of the prescriptions were fluoroquinolones.
 - Fluoroquinolones were associated with increased incidence of aortic aneurysms. Compared to the other antibiotics, fluoroquinolones were associated with a higher 90-day incidence of AAA and iliac aneurysms as well as more likely to undergo aneurysm repair.

- They recommended that fluoroquinolone use should be pursued with caution in all adults, not just high risk individuals, and they recommended broadening of the warnings from the FDA.
- Fluoroquinolones playing a role in dissections and aneurysm formation is often a highly tested question
- Medical therapy has been studied with disappointing results. Beta-blockers and ACE/ARB inhibitors have been studied but have not shown any effect on growth of AAAs.

What is the dreaded complication of AAA?

- Aneurysm rupture is the fear with a diagnosis of AAA. The risk of rupture increases yearly as the aneurysm expands. Once an aneurysm develops, it tends to enlarge gradually yet progressively. This is an important concept to grasp for testing.(Lindsay 2019a; Tracci, Roy, and Upchurch, n.d.)
- Growth rate
 - \bullet For smaller an eurysms (3-5cm in size), the growth rate is approximately 2-3mm/year
 - For larger aneurysms (>5cm), the growth rate is higher at 3-5mm/year.
- Rupture risk, historically, has been described as:

Table 10.1: Historical reporting of AAA rupture risk(Tracci, Roy, and Upchurch, n.d.)

AAA Size	Cumulative Annual Rupture Risk
3-3.9 cm	0.3%
$4\text{-}4.9~\mathrm{cm}$	$0.5 ext{-} 1.5\%$
$5\text{-}5.9~\mathrm{cm}$	1-11%
$6\text{-}6.9~\mathrm{cm}$	$11 extsf{-}22\%$
$>7 \mathrm{~cm}$	>30%

- Newer data suggests the true rupture risk per year is decreasing with time. Rupture risk has recently been reported to be much lower.
 - A systematic review from the UK published in JVS in 2015, the rupture risks were far lower than previously reported and what is documented in most textbooks. Data included 11 studies, reviewing 1514 patients, published between 1995 and 2014.(Parkinson et al. 2015)

AAA Size	Cumulative Annual Rupture Risk
< 5.5 cm	<1%(Oliver-Williams et al. 2019)
$5.5\text{-}6.0~\mathrm{cm}$	3.5% [0-9%](Parkinson et al. 2015)
$6.1\text{-}7.0~\mathrm{cm}$	4.1% [0-9%](Parkinson et al. 2015)
${>}7.0~{\rm cm}$	6.3% [0-14%] (Parkinson et al. 2015)

Table 10.2: AAA rupture risk based on most recent publications

• Factors that increase the risk of rupture other than the size of the aneurysm are smoking, COPD, hypertension, transplant recipient, and rapid enlargement (defined as 1 cm/year or more).

10.2 Evaluation

What is the work up for a AAA? (Moore, Lawrence, and Oderich 2019)

- 75% of all infrarenal AAAs are asymptomatic when first detected and often incidentally discovered on unrelated imaging.
- Symptoms Some patients may report symptoms such as abdominal, flank or back pain from pressure on adjacent somatic sensory nerves or overlying peritoneum. Tenderness by itself is not a reliable indicator of impending rupture. Other symptoms include thrombosis and distal embolization.

10.2.1 Imaging

- Ultrasound, when feasible, is the preferred imaging modality for aneurysm screening and surveillance.
 - The Society for Vascular Surgery (SVS) recommends a one-time ultrasound screening in men and women ages 65 to 75 years with either a history of smoking or a family history of AAA, as well as men and women over the age of 75 with a smoking history in otherwise good health who have not previously undergone screening. (Chaikof, Dalman, Eskandari, Jackson, Lee, Mansour, Mastracci, Mell, Murad, Nguyen, Oderich, et al. 2018)
 - For more technical details on duplex screening see Section 20.6.1.

Table 10.3: Recommended Intervals for AAA Surveillance(Thompson et al. 2013; Chaikof, Dalman, Eskandari, Jackson, Lee, Mansour, Mastracci, Mell, Murad, Nguyen, Oderich, et al. 2018)

Size on Imaging	Surveillance Interval
$2.5-2.9~\mathrm{cm}$	10 years
3 - 3.9 cm	3 years
4 - 4.9 cm	1 year
5 - 5.4cm	6 months

- It is important to note that these screening guidelines are Level 2, Grade C evidence from the SVS.
- Traditionally, once duplex reveals an aneurysm 5cm in size, an initial CTA is performed and patients are followed with additional CT scans to assist with operative planning.
- CT angiograms are helpful in operative planning and determining candidacy for EVAR. You can assess the relationship of the aneurysm to the renal arteries, assess the access vessels, and measure seal zones
 - The maximum aneurysm diameter derived from the CTA should be based on outer wall to outer wall measurement perpendicular to the path of the aorta (the center-line of the aneurysm).
- MRA is recommended for patients with renal insufficiency who cannot tolerate iodinated contrast.

10.3 Management

What are the indications for repair? (Moore, Lawrence, and Oderich 2019)

- The current recommendation to repair a fusiform aneurysm is 5.5cm for men (*Level 1, Grade A evidence*), 5.0cm for women as they have a higher risk for rupture, and rapid growth (>5mm over 6 months). (Chaikof, Dalman, Eskandari, Jackson, Lee, Mansour, Mastracci, Mell, Murad, Nguyen, Oderich, et al. 2018)
- For saccular aneurysms, the SVS practice guidelines recommend elective repair (*Level* 2, Grade C evidence). (Chaikof, Dalman, Eskandari, Jackson, Lee, Mansour, Mastracci, Mell, Murad, Nguyen, Oderich, et al. 2018)
 - Studies show equivalent wall stress in saccular aneurysms at much smaller sizes when compared to fusiform aneurysms. This has led to the notion that they have a higher rupture risk at smaller sizes.

- A study published in Annals of Vascular Surgery in 2016 showed a significant portion of ruptures <55mm in size were saccular in nature. (Kristmundsson et al. 2016)
 - Specific size guidelines for repair are currently lacking because of their infrequent presentation.

What are the options for repair, and how do you choose? (Moore, Lawrence, and Oderich 2019; Fairman 2019)

- Two options: open repair and endovascular aortic aneurysm repair (EVAR).
 - When attempting to decide between the two, one must consider the patient's perioperative risk as well as the patient's anatomy, which will be reviewed further here.
- When reviewing the patient's risk for surgery, there are many tools to assist, which are outlined in the Society for Vascular Surgery's practice guidelines.
- The VSGNE or Vascular Study Group of New England developed a risk prediction model for mortality which can assist in your decision making. This is endorsed by both SVS and the Vascular Quality Initiative. (Chaikof, Dalman, Eskandari, Jackson, Lee, Mansour, Mastracci, Mell, Murad, Nguyen, and al. 2018; Eslami et al. 2015)
 - This risk model looks at open vs endovascular repair and further delineates infrarenal vs suprarenal clamps
 - It includes aneurysm sizes with 6.5cm as the cut off.
 - It includes age above or below 75yo.
 - Gender and comorbidities are included like heart disease, cerebrovascular disease and COPD.
 - An important risk factor is also renal function which is delineated by creatinine at 1.5-2 or >2.
 - Each of these risk factors is assigned a point grading.
 - These points are added together and they place the patient on a spectrum of mortality risk. Depending on the amount of points accumulated, the risk is divided into low, medium, high or prohibitively high-risk groups
 - This is something that can help both the patient and physician in deciding on surgery and how to proceed.
- Recent studies have shown that decreased aerobic fitness and high frailty score predicted increased morbidity and mortality after open aneurysm repair.
- High-risk patients are defined by the following in the SVS guidelines:

- Unstable angina or angina at rest
- Congestive heart failure with EF < 25--30%
- Serum creatinine level > 3 mg/dL
- Pulmonary disease manifested by room air ${\rm PaO2} < 50$ mmHg, elevated PCO2, or both.
- To help delineate a patient's risk, a preoperative workup is necessary. The SVS practice guidelines recommend the following: (Chaikof, Dalman, Eskandari, Jackson, Lee, Mansour, Mastracci, Mell, Murad, Nguyen, Oderich, et al. 2018)
 - Determine if the patient has an active cardiovascular condition. Coronary artery disease is responsible for at least 50% to 60% of perioperative and late deaths after operations on the abdominal aorta, therefore, it is important for patients to undergo cardiac evaluation prior to surgery.
 - Unstable angina, decompensated heart failure, severe valvular disease, significant arrhythmia -> Cardiology consultation (Level 1, Grade B)
 - Significant clinical risk factors such as coronary artery disease, congestive heart failure, stroke, diabetes mellitus, and chronic kidney disease -> Stress test (Level 2, Grade B)
 - Worsening dyspnea -> Echocardiogram (Level 1, Grade A)
 - All patients undergoing EVAR or open repair require EKG
 - In patients capable of moderate physical activities, such as climbing two flights of stairs or running a short distance (MET >= 4), there is no benefit in further testing.
 - If coronary intervention is required, this takes precedence over aneurysm repair.
 - History of COPD
 - Pulmonary function test with ABG (Level 2, Grade C)
 - Smoking cessation for at least 2 weeks prior (Level 1, Grade C)
 - Pulmonary bronchodilators at least 2 weeks before aneurysm repair (*Level 2*, Grade C)
- In patients who are deemed high risk, EVAR is the most attractive option in anatomically suitable patients
- Morbidity and mortality rates are lower for EVAR than open repair in the short term. This is illustrated in multiple studies.

- The EVAR-1 trial, a randomized prospective UK study including 1082 patients, compared EVAR with open AAA repair in patients who were fit enough to undergo open surgical repair from 1999-2003. The 30-day mortality rate was reduced in the EVAR group (1.7% vs 4.7%), although secondary interventions were more common in the EVAR group (9.8% vs. 5.8%). (R. Greenhalgh 2004)
- The DREAM trial, a multicenter randomized trial from 2000-2003, compared EVAR vs Open repair in 345 patients with a reduction in operative mortality (4.7% vs 9.8%) with the majority of complications accounted for by pulmonary issues. (Prinssen, Cuypers, and Buskens 2004)
- This early survival benefit with EVAR over open repair disappears by the third postoperative year.
 - The Open vs Endovascular Repair (OVER) trial included 881 patients from 42 VA centers randomized to either EVAR or open repair. This demonstrated that perioperative mortality was improved in the EVAR group (0.5% vs 3.0%), yet no statistically significant difference was seen in mortality at 2 years (7.0% vs 9.8%). (Lederle et al. 2019)

Take a Listen

Check out our episode with Dr. Julie Freischlag where we discuss her involvement with the OVER trial.

• Late mortality seems to be higher in EVAR due to ruptures from endoleaks that do not occur in open repair. (Rajendran and May 2017)

Reviewing the anatomic criteria for traditional infrarenal EVAR may rule out EVAR as an option in some patients. These criteria vary slightly depending on the particular device being used.

• Neck

- A neck length of at least 10-15mm from the renal arteries to the aneurysm start with a diameter of 18-32mm.
- It is important that the neck is relatively free of thrombus or calcification to decrease the risk of endoleaks.
- Infrarenal necks 4-15mm would require fenestrated endovascular repair. The only FDA approved device for fenestrated repairs is the Cook Z-Fen device.(Gustavo S. Oderich et al. 2014)

- Z-fen requires diameter 19-31mm, angle <45deg angle to long axis of aneurysm, ipsilateral iliac fixation >30mm and 9-21mm in diam, contralateral iliac >30mm and 7-21mm diam. Multiple or early branching renal arteries can limit this reapir as well. (Gustavo S. Oderich, Correa, and Mendes 2014)
- Parallel grafts for short necks (i.e. snorkel or chimney) require 20mm seal for good outcomes. (Donas et al. 2015)

💡 Take a Look

Check out our "How I Do It" episode where Dr. Wes Jones and Murray Shames review their technique for performing a Zfen endovascular aortic anuerysm repair. This technique is beyond the scope of this review.

- Angulation
 - Neck angulation should be < 60 degrees for current devices
- Access vessels
 - Access vessels must be adequate for delivery of the device depending on the sheath size required (6-8mm)
- Aortic bifurcation
 - The aortic bifurcation must be >20mm in size to accommodate the graft opening to full caliber
- Iliac landing zone
 - Adequate seal zone in the distal common iliac arteries of 10-15mm in length and diameter of 7.5-25mm.
 - If covering the hypogastric arteries is necessary unilaterally to obtain a seal, you can embolize the hypogastric artery (to prevent retrograde flow) and extend the graft into the external iliac artery.
 - If this is an issue bilaterally, an iliac branch device can assist in maintaining perfusion into the hypogastric arteries.

Special considerations

Aortocaval fistula - sometimes aneurysms can be complicated by fistula with the IVC. This can occur with primary fistula, rupture and after repair. This is rare, 1% in elective AAA and 6% in rupture.(Brightwell, Pegna, and Boyne 2013; R. Schmidt et al. 1994) High risk of persistent endoleak due to ongoing flow into IVC. Open repair often best.(Orion, Beaulieu, and Black 2016)

10.3.1 EVAR

Can you briefly go over the steps of an EVAR? (Moore, Lawrence, and Oderich 2019)

EVAR now accounts for approximately 70-80% of elective abdominal aortic aneurysm repairs and 65% of iliac aneurysm repairs in the United States and many other countries.

- 1. Performed in the operating room or IR suite with a fixed or portable C-arm
- 2. Anesthesia
 - Regional block, local anesthesia or general anesthesia depending on surgeon preference and patient risk
- 3. Groin access and short sheath placement
 - Percutaneous Closure devices are introduced prior to inserting the large sheaths containing the stent-grafts
 - Cut-down
- 4. Pigtail catheter is used to perform an aortogram of the abdominal aorta and iliac arteries
- 5. The renal artery orifices are marked. If there is any concern about good visualization, IVUS (intravascular ultrasound) can be used to assist.
- 6. Systemic heparin is given
- 7. Bilateral femoral sheaths are exchanged over a stiff wire for the necessary sheaths required for the device size chosen.
 - Main trunk and ipsilateral limb sheath on one side
 - Contralateral limb sheath on the other side
- 8. The main body is positioned in the proximal neck and a repeat angiogram is commonly performed to confirm the positioning of the device at the desired level just below the lowest renal artery. It is best to position the main body so that the gate is directed at the simplest angle to cannulate.
- 9. The main body is deployed to the point where the gate is opened
- 10. Contralateral limb gate cannulation is performed using a wire and directional catheter.
- 11. Once in the gate, a pigtail catheter is formed within the main body and must be able to spin freely 360 degrees to confirm placement within the endograft. There are multiple techniques to confirm that the wire is in the endograft, but this MUST be confirmed before proceeding to deploy the contralateral limb.

- 12. The contralateral limb is introduced and deployed taking care to preserve flow to the internal iliac artery.
- 13. The remainder of the main body is deployed and iliac extensions deployed if required.
- 14. The stent graft is ballooned at the neck, within the gate, at the bifurcation, and distal iliac seal zones.
- 15. An aortogram, usually multiple in different views, is performed to exclude any endoleaks.
- 16. The sheaths are removed, and the groin sites are closed using Perclose devices if performed percutaneously, or primary repair if open cutdown performed.
- 17. Check pedal signals at the end of the case to ensure no thromboembolic events or femoral artery access injuries have occurred. If there is concern, an ultrasound duplex can be performed intraoperatively.

🛿 Take a Listen

Check out our episode where we review the basics of the EVAR procedure in more detail.

You mentioned endoleaks, can you discuss the complications specific to EVAR and the management? (Moore, Lawrence, and Oderich 2019)

- 15 year follow up from EVAR-1 shows inferior late survival and higher re-intervention in EVAR patients when compared with open repair. (R. Patel et al. 2016)
- Many of the cardiopulmonary complications inherent with open repair do not occur with EVAR as there is no aortic cross clamping.
 - In a study from Mayo clinic evaluating elective infrarenal AAA repairs from 1999 to 2001, Elkouri et al found that cardiac and pulmonary morbidity after EVAR was drastically reduced compared to open repair (11% vs 22% and 3% vs 16%, respectively). (Elkouri et al. 2004)
- Risk of ischemic colitis remains as the IMA is covered with EVAR. It is lower than with open repair but remains 1-2%.
- Renal insufficiency may occur secondary to contrast administration in a patient with underlying chronic kidney disease. Thromboembolic events may occur from thrombusladen aortic necks with wire and device manipulation to the renal arteries as well.
- Iliac limb occlusion can occur within hours of EVAR (3-5%). Risk factors included iliac artery tortuosity, calcified plaque burden or occlusive disease, external iliac artery dissection(Vacirca et al. 2019), inadequate outflow due to access site complication. Post operative CFA duplex will show dampened waveforms.(R. M. Greenhalgh et al. 2010; Taudorf et al. 2014)

• Emergent conversion from EVAR to open are associated with significant increase in morbidity and mortality. Risk factors include aneurysm diameter, young age, female gender and nonwhite race.(Ultee et al. 2016a)

10.3.1.1 Endoleak

Defined as persistent blood flow within the aneurysm sac following EVAR.

- Type I
 - A leak at the graft ends secondary to inadequate seal proximally (1a) or distally (1b)
 - If identified intraoperatively, Type I endoleaks require attention with further balloon angioplasty, proximal or distal extension, or endoanchors.
 - Use of endoanchors in thrombus and calcium-laden aorta is not recommended.(Jordan et al. 2014) Outcomes are similar with or without suprarenal fixation.
 - If seen in follow up surveillance, intervention is necessary.
- Type II
 - Sac filling secondary to retrograde filling via a branch vessel off of the aneurysm sac such as a lumbar artery or the IMA
 - If identified intraoperatively, this typically does not need to be addressed in the OR. Typically, type II endoleaks spontaneously thrombose and therefore can be observed.
 - It is common to continue monitoring even if there is persistent flow as long as there is no aneurysm sac growth. Type II endoleaks persist in 5-25% of patients. Risk factors for persistent endoleak include anticoagulation(Bobadilla et al. 2010), patency/number/diameter of IMA/lumbar arteries(T. J. Ward et al. 2014), hypogastric coil embolization, distal graft extension, absence of COPD, age >80yrs, and graft type. Risk of rupture is 1.5-3%.(Lo and Schermerhorn 2016b; Abularrage et al. 2010; Timaran et al. 2004; Shalaby et al. 2016; Sarac et al. 2012; J. E. Jones et al. 2007)
 - If the leak persists for > 6 months with sac enlargement >5mm, intervention is recommended. Several techniques exist to eliminate type II endoleaks, most frequently embolization. After multiple attempts at repair and if continued sac growth with impending Type I leak, then consider open conversion. (Kelso et al. 2009; Mohapatra et al. 2019)

💡 Take a Listen

Check out our conversation with Drs Gregorio Sicard and Frank Caputo where we discuss management of endograft failures and some techniques for open conversion that are a little beyond the details of this review.

- Type III
 - Separation of graft components. Treated by relining the EVAR graft.
 - Usually identified in follow-up surveillance and necessitates intervention.
- Type IV
 - Secondary to a porous graft which typically does not occur any longer as endograft material and devices have improved. If seen, no intervention is needed at the time, and they usually thrombose on their own.
- Type V
 - Increasing an eurysm sac size with no identifiable endoleak. Commonly referred to as endotension.
 - Usually necessitates graft explantation and open repair or re-lining of the graft.

10.3.2 Open Repair

Now we can move onto open repair. Describe an open infrarenal aneurysm repair. (Moore, Lawrence, and Oderich 2019; Gary G Wind and R. James Valentine 2013)

After thorough preoperative evaluation and clearance, the patient is taken back to the operating room. An epidural may be placed preoperatively depending on institutional preference. The patient is intubated, and arterial and central venous catheters are placed. The abdomen is prepped from chest to bilateral thighs.

A cell-saver should be available to optimize resuscitation during the procedure due to expected large amounts of blood loss. Balanced resuscitation to prevent coagulopathy is important with significant blood loss.

Exposure

• Trans-peritoneal or retro-peritoneal. First we will describe the most common approach: trans-peritoneal.

Surgical steps

- 1. Mid-line laparotomy, transverse or chevron-style incision
- 2. A retractor system such as an Omni, Bookwalter or Balfour retractor is used to assist in exposure depending on physician preference.
- 3. The transverse colon is retracted cephalad, and the small bowel is retracted to the patient's right to expose the aorta. The duodenum is mobilized and the ligament of Treitz is divided. The posterior peritoneum is opened along the anterior wall of the aorta.
- 4. The aneurysm sac is now in view and careful dissection proximally for clamp site is achieved. Identification of the left renal vein crossing the aorta is key and can be divided if necessary.
- 5. Identification of the renal arteries proximally is required if there is a plan for suprarenal clamping.
- 6. Isolate bilateral common iliac arteries for distal clamp site. Use caution when dissecting the fibro-areolar tissue overlying the left common iliac artery as it contains nerves that control sexual function. Damage can result in retrograde ejaculation.
 - You can avoid nerve injury with mobilization of the sigmoid colon medially and identifying the iliac bifurcation distally, thus avoiding transecting the tissue overlying the left common iliac artery.
 - If the iliac arteries are severely calcified and pose risk for injury with clamping, intraluminal balloon catheters can be inserted for distal control instead.
 - Also, you must be cognizant of the location of the ureters crossing over the iliac bifurcation to prevent injury.
- 7. After proximal and distal clamp sites have been identified, systemic heparin is administered by anesthesia.
- 8. Clamp the distal vessels first to prevent distal embolization.
- 9. Open the aneurysm sac in a longitudinal fashion toward the patient's right to avoid the IMA and clear the sac of thrombus. Extend proximally to normal aorta and then t off the incision on the aortic wall.
 - Some physicians prefer to transect the aortic wall as opposed to leaving the posterior wall intact for the anastomosis.
- 10. Lumbar arteries on the posterior wall are ligated using figure-of-eight sutures.
 - Back-bleeding lumbar vessels can be the source of significant blood loss.
- 11. Graft

- A tube graft or bifurcated graft depending on the patient's anatomy and aortic diameter is chosen. Dacron or PTFE grafts are most common, and the choice depends on physician preference. This is anastomosed proximally in a continuous fashion.
- Once complete, the graft is flushed forward to flush out any thrombus. The graft is then clamped and the aortic clamp removed to test the anastomosis. Repair if needed.
- The distal anastomosis is completed to the aorta or bilateral iliac arteries depending on extent of the aneurysm.
- The graft is flushed forward prior to completion to remove any thrombus within the graft. The anastomosis is completed and clamps removed.
- 12. Hypotension may occur at this point from re-perfusion of the lower extremities and pelvis. Anesthesia should be notified that unclamping will occur soon prior to completion of the distal anastomosis to allow for fluid resuscitation in preparation.
 - The graft can be slowly unclamped or partially clamped to assist with blood pressure management during this time. You can also place manual pressure on the iliac arteries or femoral arteries to slowly release flow and avoid significant hypotension.
- 13. Next, the IMA must be addressed. The IMA orifice is identified within the aneurysm sac.
 - Chronically occluded or pulsatile back bleeding -> ligate.
 - Anything between occlusion and strong pulsatile back bleeding requires further evaluation. This should be performed at the end of the case after the internal iliacs have been reperfused. Methods to measure perfusion:
 - Place vessel loops or micro bulldog on IMA and assess the sigmoid colon. If there is a poor doppler signal on the antimesenteric border of the sigmoid colon, the IMA should be reimplanted.
 - Insert blunt-tip needle through the IMA orifice and pull vessel loop around needle to secure and connect to a transducer. Pressure less than 35 mmHg requires reimplantation. (Hoballah 2021)
 - The Carrel patch technique involves excising a circular button of the aortic wall around the IMA and anastomosing it to the graft wall.
 - Newer studies have shown that IMA reimplantation does not eliminate the risk of ischemic colitis after open AAA repair. In a study out of George Washington University in DC published in JVS in 2019, there was still significant risk of ischemic colitis rates with IMA reimplantation. (K. B. Lee et al. 2019)

- Using NSQIP data collected prospectively and studied retrospectively
- Out of 2397 patients undergoing AAA from 2012-2015, 135 patients (5.6%) had ischemic colitis.
- 672 patients were evaluated further after exclusion criteria applied (suprarenal clamp, emergent or ruptured, occluded mesenteric vessels)
- Of these, 637 patients had IMA ligation, 35 had IMA reimplantation
- Reimplantation was associated with More frequent return to the OR (20% vs 7.2%), Higher rates of wound complications (17.1% vs 3%), Higher rates of ischemic colitis (8.6% vs 2.4%)
- Difficult to interpret impact of revascularization of IMA on ischemic colitis rates, due to selection bias, but should be noted that patients who require revascularization still may experience colon ischemia.
- 14. To finish, the aneurysm sac is then closed over the graft to protect the viscera, and the retro-peritoneum is reapproximated. Occasionally, a vascularized omental pedicle flap may be used to separate the graft from the duodenum to prevent an aorto-enteric fistula if the peritoneum cannot be closed securely.

💡 Take a Look

Check out our "How I Do It" episode where Dr. Ashlee Vinyard and Dr. John Eidt review their technique for performing a open AAA repair.

Steps for the retro-peritoneal approach:

- Positioned semi-lateral with the left side up with bilateral groins exposed for femoral artery access. This is done in a lazy lateral position where the patients upper body is near complete lateral but the hips are rotated to the patient's left in attempt to keep both groins in the field in case they need to be accessed.
- An oblique incision extends from the left 11th intercostal space or tip of the 12th rib to the edge of the rectus abdominus muscle, through the external and internal oblique muscles, transversalis fascia until you are just superficial to the peritoneum. Using blunt finger dissection, the peritoneum is dissected from the abdominal wall posteriorly over the psoas muscle until the aorta is reached.
 - Benefits include less postoperative ileus, less intraoperative hypothermia, lower IV fluid requirements, and less post-op respiratory compromise.
 - A disadvantage is the difficulty addressing the right iliac artery from this approach. Patients can also develop denervation of the lateral abdominal wall and develop a bulge there.

🖡 Take a Listen

Check out our Origin Stories episode reviewing the history of aortic aneurysm repair.

10.3.2.1 Complications

What are some of the complications with open aortic aneurysm repair? (Moore, Lawrence, and Oderich 2019)

- Recent studies have demonstrated equivalent outcomes in younger healthy patients undergoing endovascular and open repair. (Liang et al. 2018)
- Myocardial dysfunction which is usually secondary to cardiac ischemia or hemorrhage.
 - Acute conditions associated with poor post-operative cardiac outcomes include recent MI (30d), unstable angina, decompensated heart failure, high-grade arrhythmias, severe valvular disease.(Devereaux and Sessler 2015; Fleisher et al. 2014; Kristensen et al. 2014; Tashiro et al. 2014)
- Abdominal compartment syndrome secondary to coagulopathic bleeding postoperatively or third spacing of fluids can cause abdominal compartment syndrome requiring emergent laparotomy.(Cheatham et al. 2007; Mehta et al. 2005; T. E. Rasmussen and Hallett 1997)
 - Signs include unexplained oliguria, difficulty maintaining adequate ventilation– elevated peak inspiratory pressure, and hypotension with significant abdominal distension is concerning for abdominal compartment syndrome. A sustained bladder pressure > 25 mmHg with associated organ dysfunction (elevated peak airway pressures, new onset acute renal failure) is indicative of abdominal compartment syndrome.
 - Abdominal compartment syndrome can still occur after EVAR during an aortic rupture in up to 10-20% of patients, therefore, one must keep a heightened suspicion for this in the post-operative period. (Veith et al. 2009)
 - It is important to note that a patient with a soft abdominal exam can still have abdominal compartment syndrome particularly with an enlarged body habitus.
 - Laparotomy may initially worsen hypotension and hemodynamic instability due to reperfusion injury.
- Renal failure can occur due to suprarenal aortic clamping, atheromatous embolization or hypotension causing acute tubular necrosis (ATN).
 - A suprarenal clamp time of less then 30 minutes is ideal to reduce the risk of renal failure.

- Postoperative ileus is common. Duodenal obstruction from dissection of the ligament of Treitz can mimic a gastric outlet obstruction.
- Ischemic colitis of the left colon and rectum is the most serious gastrointestinal complication.(Moghadamyeghaneh et al. 2016)
 - Incidence ranges from 0.2 10%, 3-4x more common after operations for occlusive disease than aneurysmal disease.
 - It is important to study the collateral pathways on the preoperative CT scan and the patient's history to assist in surgical decisions regarding IMA reimplantation including:
 - Stenosis/occlusion of the SMA
 - Previous colectomy
 - IMA or Hypogastric artery occlusion
 - Earliest manifestation is postoperative diarrhea, especially bloody diarrhea.
 - Sigmoidoscopy is needed for diagnosis.
 - Mild colon ischemia with patchy mucosal involvement should be treated with bowel rest, fluid resuscitation and antibiotics. Transmural necrosis requires emergent operation with colon resection. Patients can be left in discontinuity or an end colostomy performed depending on stability.
 - The mortality rate with colon ischemia after aneurysm surgery is about 25% but reaches over 50% if bowel resection is required.(Brewster et al. 1991) This is a very heavily tested topic for both general surgery and vascular surgery boards.
- Distal ischemia from embolization downstream can lodge in larger vessels or cause microembolization, colloquially known as "trash foot".
- Infection is rare but can be associated with graft-enteric fistula which is another highly tested topic. (Fatima et al. 2013; Gustavo S. Oderich et al. 2011; Smeds et al. 2016)

🛿 Take a Listen

Check out our conversation with Miss Rachel Bell where we discuss management of infected aortic grafts and mycotic aneurysms that is a little beyond the scope of this review.

10.3.2.2 Postoperative Surveillance

What is the post-operative surveillance required for open and endovascular approach, and how do they differ? (Moore, Lawrence, and Oderich 2019)

- That is a great question because it highlights why open repair has continued to be so important, especially for young, healthy patients.
- Post-operative surveillance is necessary in the immediate post-operative period for open repair to evaluate incisions. Follow-up is only needed every 5-10 years, unless the patient becomes symptomatic.
 - Perianastomotic degeneration occurs in 0.5-10% of cases. Can be diagnosed with CT scan or ultrasound.(Curl et al. 1992)
- In contrast, EVAR patients require a strict postoperative surveillance regimen to allow for detection of endoleaks, aneurysm sac expansion, stent fracture, limb kinking and material fatigue.
 - CT scans at 1-, 6- and 12-month intervals initially then annually are recommended which raises concerns related to cost, cumulative radiation exposure, and contrast administration.
 - Some physicians may elect to use ultrasound for surveillance with CTA prompted if an endoleak is identified or the sac is enlarged, particularly in patients with stable aneurysms.
 - The long-term follow-up is often inconsistent and a study of 19,962 Medicare beneficiaries undergoing EVAR from 2001 to 2008 showed that 50% of patients were lost to annual imaging follow-up at 5 years after surgery.(A. Schanzer et al. 2015)
- Some patients will elect for open repair to avoid frequent surveillance if they are a candidate for both, while other patients will select endovascular management to avoid the short-term effects like longer hospitalizations, post-operative pain, and longer recovery time to baseline functioning in open surgery.

10.3.3 Ruptured Aneurysms

Although elective repair is important, can you touch on the management of a ruptured AAA (RAAA) as our last topic of the session? (Moore, Lawrence, and Oderich 2019; Lindsay 2019b)

Ruptured AAAs have declined secondary to improved medical management, decreased rates of smoking and superior diagnostic imaging and surveillance.

Traditionally, it has been taught that 50% of ruptured AAAs die in the field and of those remaining, 50% will die in the hospital. With time, the in-hospital mortality rate has decreased.

• In one study out of Finland, of 712 patients with ruptured AAAs from 2003-2013, 52% died prior to arrival to the hospital. Of those that were offered surgery, 67% of patients were alive at 30 days indicating a mortality rate of 33%. (M. T. Laine et al. 2016)

Diagnostic triad on presentation:

• Pain, syncope and known or palpable AAA.

When a ruptured AAA is suspected or diagnosed, permissive hypotension is key in the initial management before surgery.

- Allowing systolic arterial pressures of 50-70 mmHg as long as the patient is mentating appropriately.
- Limits internal bleeding which further limits loss of platelets and clotting factors.

Initial management involves many considerations like patient stability, patient's anatomy and the surgeon's experience with either open or endovascular repair.

Due to the developments of endovascular techniques, it is ideal to have a CTA prior to the operating room to determine if the patient is a candidate for an EVAR.

There are two options for expedient aortic control in an unstable patient with a ruptured aneurysm.

- Open supraceliac aortic clamping
 - Achieved by retracting the stomach caudally, entering and dividing a portion of the gastrohepatic ligament, reaching under and medial to the caudate lobe, dividing the pars flaccida, and identifying the spine. The aorta lies to the patient's left of the spine and is bluntly dissected anteriorly and laterally for aortic clamp placement.
 - Another method of supraceliac exposure and control is to mobilize and reflect the left lobe of the liver, sweep the esophagus to the patient's left, divide the right crus of the diaphragm and bluntly dissect both sides of the aorta then apply the clamp.
 - A nasogastric tube can help identify the esophagus when placing this clamp to ensure the esophagus has been swept to the patient's left and protected.
 - The clamp should be moved down to the desired position for repair (supra or infrarenal depending on anatomy of the aneurysm neck) to decrease ischemia time to visceral vessels as soon as possible.
- Percutaneous occlusive aortic balloon

• Gain percutaneous access and place an occlusive aortic balloon for stabilization in the distal thoracic aorta. This will require a long support sheath, usually 12fr in size, to prevent distal migration of the occlusive balloon.

EVAR has been used increasingly to treat ruptured AAAs and offers many theoretical advantages over open repair.

- Less invasive, eliminates risk of damage to periaortic and abdominal structures, decreases bleeding from surgical dissection, minimizes hypothermia and third space losses, and lessens the requirement for deep anesthesia.
- EVAR has been deemed superior to open repair for the treatment of RAAA in many studies.
 - In a study out of UVA published in JVS in August 2020, they looked at ruptures in the VQI database from 2003-2018. This resulted in 724 pairs of open and endovascular pairs after propensity matching. (L. J. Wang et al. 2020)
 - There was a clear advantage of endovascular compared to open repair in patient's with suitable anatomy.
 - Length of stay was decreased with 5 vs 10 days in open. 30 day mortality was much lower at 18% vs 32%. Major adverse events like MI, Renal failure, leg ischemia, mesenteric ischemia, respiratory complications were much lower in the EVAR group at 35% vs 68% in the open group.
 - All cause 1 year survival was much higher with EVAR at 73% vs 59% in the open group.

💡 Take a Listen

Check out our conversation with Professor Janet Powell where we discuss her landmark IMPROVE trial comparing open and endovascular management of ruptured aortic aneurysms.

- Despite improved RAAA results with EVAR, conversion from EVAR to open AAA repair appears to have the most unfavorable outcomes in terms of mortality.
 - Conversions can be early or late and are due to access-related problems, errors in endograft deployment, graft migration, persistent endoleak, graft thrombosis, or infection.
 - In a study evaluating 32,164 patients from NSQIP with 300 conversions (7,188 standard open repairs and 24,676 EVARs), conversion to open repair was associated with a significantly higher 30-day mortality than standard open repair (10% vs 4.2%) and EVAR (10% vs 1.7%). In addition, conversion patients compared to standard open

patients were more likely to undergo new dialysis (6.0% vs. 3.5%), cardiopulmonary resuscitation (5.3% vs. 1.9%), postoperative blood transfusion (42.3% vs. 31.6%), and have a myocardial infarction (5.0% vs. 2.2%). (Ultee et al. 2016b)

• A more recent systematic review and meta-analysis found the elective late open conversion has comparable outcomes to primary open aneurysm repair. 36% are able to complete with infrarenal clamp alone. Highest complication rate in urgent setting (10x higher mortality). (Goudeketting et al. 2019)

10.4 Iliac Artery Aneurysms

10.4.1 Demographics

CIAA aneurysm defined as >1.5cm.

10.4.2 Presentation

Often found incidentally, but carries risk of rupture and distal embolization.

• Distal embolization is rare, but when it occurs patient should undergo embolectomy and treatment of aneurysm thought to be the embolic source.(Bacharach and Slovut 2008; Ferreira et al. 2010; Nachbur, Inderbitzi, and Bär 1991)

10.4.3 Management

Treatment threshold for CIAA is often accepted at 3.5cm.(Dix, Titi, and Al-Khaffaf 2005; Matti T. Laine et al. 2017)

Treatment options include

- Open repair
 - If open repair involves aneurysm or occlusive disease of internal iliac arteries, should attempt to revascularize at least one IIA via a jump graft.(Krupski et al. 1998; Huang et al. 2008)
- Iliac branched device to preserve hypogastric. Minimum size requirements include:
 - CIA 17mm in diameter
 - EIA 6.5-25mm in diameter, 10mm seal zone
 - IIA 6.5-13.5mm in diameter, 10mm seal zone(Schneider et al. 2017)

💡 Take a Look

Check out our "How I Do It" episode where Dr. Lily Johnston and Dr. Gustavo Oderich review their technique for performing an iliac branch device repair with is a little beyond the scope of this review.

- Extension of EVAR into external iliac and coil of internal iliac artery
 - If internal iliac arteries also aneurysmal, ensure that outflow vessels are all coiled. Posterior branch gives rise to the superior gluteal artery, which is high risk for retrograde filling.(Ryer et al. 2012)
 - Coil embolization can result in ipsilateral buttock claudication in 12-22% of patients. Treatment is with medical therapy and walking regimen, resolution usually occurs at 6 months.(Papazoglou et al. 2012; Stokmans et al. 2013)
 - If the IMA is chronically occluded, then coverage of the IIA may compromise colonic blood flow and increase risk of ischemia.(Angiletta et al. 2011; Karch et al. 2000)

11 Mesenteric Disease

Authors: Matt Chia, MD and Nick Mouawad, MD

Mesenteric vascular disease can be broken down into three disease states that we'll cover today. There's the arterial disease, which is clearly separated into acute mesenteric ischemia and chronic mesenteric ischemia. Then there is venous disease, which we'll touch on briefly. There are also a handful of somewhat related diseases that we'll also sprinkle into these discussions, like median arcuate ligament syndrome and SMA syndrome, but that's overall where we're headed.

Mesenteric collatorals are important for protecting against mesenteric ischemia. The gastroduodenal and pacreaticoduodenal arteries connect the celiac and SMA. The meandering artery (Arc of Riolan) and marginal artery of Drummond connect IMA and SMA. Hemorrhoidal branches connect internal iliacs and IMA. (A. Chandra and Quinones-Baldrich 2010; van Gulik and Schoots 2005)

11.1 Acute Mesenteric Ischemia

11.1.1 Presentation and Evaluation

Can you tell me about the classic presentation and approach to patients presenting with acute mesenteric ischemia?

These patients present with the sudden onset of abdominal pain. Nausea, vomiting, distention, and diarrhea (possibly bloody, described as "sudden and forceful evacuation") are the common symptoms. Pain out of proportion is the classic buzzword, and can be hidden on multiple choice tests with a pain score of 10/10 with only mild abdominal tenderness on physical exam.

Etiology most often from acute embolism (40-50%), which is often secondary to cardiac source (atrial fibrillation or recent MI).(Wyers, Mark C and Martin 2019)

Vitals typically normal, possible tachycardia.

Lab evaluation typically unremarkable. Leukocytosis, hemoconcentration, and acidosis (high anion gap) all are frequently found, but the absence of these definitely does **not** rule out acute mesenteric ischemia.

D-dimer has been proposed as a reasonable rule-out test for acute mesenteric ischemia.

How about radiology studies?

Plain film:

Frequently normal, may show ileus in the early stages. In late stage acute mesenteric ischemia, findings on plain film can include bowel wall edema (thumbprinting) or pneumatosis.

CT angio:

Probably represents the most common diagnostic modality to diagnose acute mesenteric ischemia. CTA has the advantages of speed, availability, and noninvasiveness when compared to conventional angiography, and also allows for some assessment of the degree of bowel involvement.

Preferentially, a "negative" oral contrast agent would help prevent oral contrast from causing artifact or obscuring evaluation of the vessels, although the availability of these agents may limit their use.

How about mesenteric duplex (vascular lab studies)?

Mesenteric duplex has the advantage of being able to see the velocity of flow across a stenosis, giving you a good method of quantifying the significance of a stenosis. However, bowel gas often limits the acoustic windows for visualizing the mesenteric arteries, and so we usually will have patients fast for several hours before a study. Also, duplex is more sensitive for proximal disease rather than distal mesenteric involvement. For these reasons, mesenteric duplex is considered the gold standard for evaluating chronic mesenteric ischemia, but has no real role in the evaluation of acute mesenteric ischemia. Can you imagine, having a tech mash a transducer into a patient with acute abdominal pain?!

11.1.2 Management

What's your approach to the initial management of the patient?

- 1. <u>Resuscitation</u>. Fluids, fluids. These patients are really volume down, and are headed towards a profound distributive shock that will be worsened by your eventual plan for revascularization (think ischemia-reperfusion injury). Also look for electrolyte imbalances and correct those early.
- 2. <u>Antibiotics</u>. These patients are not usually septic on initial presentation, but are at highrisk, so broad-spectrum antibiotics with gut coverage (Gram negative and anaerobes) are standard of care.
- 3. <u>Heparin</u>. In the absence of a clinical contraindication, these patients should be systemically heparinized, with a bolus, as soon as the diagnosis is made.

Which blood vessel does acute mesenteric ischemia typically involve?

The SMA. This makes sense if you think of each mesenteric distribution. The celiac distribution has organs that have redundant blood supply (like liver and stomach). The IMA is frequently occluded in patients with AAA, but the patients are rarely ever symptomatic due to collateral flow. The hypogastrics (which ARE mesenteric vessels, especially in the situation of an occluded IMA) principally supply the rectum from a mesenteric standpoint.

For embolic pathology, the acute angle of the SMA seems to predispose it to capture emboli from above, but this is more theoretical than proven. The embolism is often lodged distal to middle colic vessel.(Wyers, Mark C and Martin 2019; Kazmers 1998)

Tell me about the two main pathologies, and how they would differ in terms of the anatomy and operative findings.

So the two most common etiologies of acute mesenteric ischemia are embolism and thrombosis.

Embolism is the more common, where preexisting thrombus (think atrial fibrillation, mural thrombus from thoracic aneurysm, etc.) or plaques from atherosclerotic disease break off and lodge in the SMA. The classic operative finding is that an embolism lodges just distal to the middle colic artery, where there is a significant caliber change in the SMA. This is distal to the first few jejunal branches off of the SMA, leading to the classic sparing of the proximal jejunum and transverse colon. In other words, the mid to distal jejunum and all of the ileum will be ischemic, but other areas of the SMA territory are spared. Atherosclerotic debris is typically smaller, and results in smaller, more patchy areas of ischemia.

Thrombosis occurs primarily as a plaque rupture of preexisting atherosclerotic disease, resulting in acute thrombosis at the site of the disease. Thus the patients will often present with an acute-on-chronic symptomatology, having classic symptoms of postprandial abdominal pain, food fear, and weight loss, but with a sudden onset of severe symptoms. This frequently allows for the development of mesenteric collaterals, which may make the onset of acute symptoms more insidious than for embolic pathology. In the majority of situations, the atherosclerotic disease is most severe right at the origin (consistent with what we know about shear stress and branch points in blood vessels). This means that when the plaque ruptures, the entirety of the SMA occludes, leading to ischemia of the entire territory, as opposed to the jejunal-sparing distribution seen in embolic disease.

Describe the operative steps to getting exposure of the supraceliac aorta (or the celiac artery).

- 1. Divide triangular ligament to mobilize left lobe of liver
- 2. Divide gastrohepatic ligament to enter the lesser sac
- 3. Retract liver to right with a self-retaining retractor
- 4. Push esophagus left (use NGT to assist with identification)

- 5. Divide peritoneum overlying crura to identify celiac vessels
- 6. Typically trace common hepatic artery backwards to identify celiac artery
 - Watch out for the left gastric vein as it crosses the celiac artery as it drains the lesser curve of the stomach into the portal vein.
 - About half of the time, the phrenic artery takes an origin from the celiac artery and must be controlled during exposure.
- 7. To expose the supraceliac aorta, divide the median arcuate ligament and separate the left and right crura from each other.

So through this kind of exposure, what mesenteric vessels do you get access to?

You can trace most of the proximal celiac distribution right at the origin, and through this exposure you get access to the origin of the SMA if you mobilize the superior border of the pancreas. The neck of the pancreas and the splenic vein cross the anterior of the SMA, obscuring the rest of the mid and distal SMA from the superior approach.

How about getting to the rest of the SMA?

There are a couple of places you can get exposure to the SMA.

Most commonly you'll hear it described at "root of the mesentery." Specifically, lifting up the transverse colon will stretch out its mesentery (i.e. transverse mesocolon). At the bottom, or "root" of the transverse mesocolon, a transverse incision is made. If the middle colic artery is palpable in the mesocolon, the incision can be made around it, and you can trace the middle colic backwards to the SMA. Usually you'll find the SMV first, and the SMA will be just to the left of it. Be sure to identify and preserve small jejunal branches during the dissection. If needed, careful dissection superiorly, going behind the inferior border of the pancreas can get a little more proximal exposure. Embolectomy through a transverse arteriotomy in the SMA is the best approach.(Wyers and Martin 2019; Kazmers 1998)

Alternatively, you can get to the SMA from a lateral approach, specifically from the left side. Begin by dividing the ligament of Treitz and mobilizing the 4th portion of the duodenum. The SMA is found in the tissues just cephalad to the duodenum. You can also improve your proximal exposure if needed by retracting the inferior border of the pancreas cephalad to the level of the left renal vein.

Other options include a retroperitoneal exposure, like you were preparing to treat a thoracoabdominal aneurysm. Additionally, the more distal SMA can just be identified in the small bowel mesentery.

So for the operative strategy for acute mesenteric ischemia, tell me about the general approach to the patient.

- 1. <u>Resect frankly necrotic bowel and contain gross spillage</u>. Once you revascularize the bowel, compromised-appearing bowel may improve and not need immediate resection. Thus the first step is only damage control, to remove anything completely unsalvageable that is making the patient sick, or anything causing gross contamination of the operative field. The key is you're not doing anything definitive with the bowel as your first step.
- 2. <u>Revascularization</u>. SMA embolectomy is the initial management of choice for embolic disease. Thrombotic disease, on the other hand, may be more challenging to treat by embolectomy alone, and frequently are treated with a bypass. (More to come on these procedures).
- 3. <u>Re-assess bowel viability</u>. Clinical status permitting, 20-30 minutes should be taken to fully assess the results of the revascularization before proceeding with resection. Perfusion can be assessed by many methods, including clinically, by Doppler, pulse oximetry (a.k.a. photoplethysmography), fluorescein fluorescence, etc. The take home is to give the bowel enough time to be perfused before going ahead with resection.
- 4. <u>Proceed with temporizing or definitive bowel repair</u>. Resection, leaving in discontinuity, primary anastomosis, diversion, etc. All of these are options on the table, but the key here is that all of the previous steps occur before addressing the bowel.
- 5. <u>Consider second-look laparotomy.</u> Many times, bowel may look questionable even after revascularization and thorough re-assessment. To preserve the most bowel length, it may be reasonable to leave borderline bowel alone at the index operation and do a "second look" to fully reassess the bowel, especially after the patient has benefited from aggressive resuscitation in the ICU.

Endovascular approaches to acute mesenteric ischemia have been described but data is limited. Exploratory laparotomy and embolectomy should be preferred management.(Wyers and Martin 2019)

Let's talk a little about the steps for an SMA embolectomy.

Typically, you'll expose at the root of the transverse mesocolon. After obtaining proximal and distal control, an arteriotomy is made, and embolectomy can be performed by passing Fogarty catheters in both a retrograde and antegrade fashion. The arteriotomy can be made transversely for an embolectomy, and thus could be closed primarily. A longitudinal arteriotomy may be advisable if you have a high suspicion that you'll need to do a bypass, and if not, may be closed with vein patch angioplasty (remember that the field is contaminated or dirty in many situations).

So if I'm gonna do a bypass, what are my options for conduit?

Yeah, so again, because the field is frequently contaminated or dirty, a good conduit is saphenous, followed by femoral vein. Thus every patient undergoing surgery for acute mesenteric ischemia should have both legs prepped out in the field. Prosthetic conduit has the advantages of being more resistant to kinking (externally reinforced), likely better patency than vein (although data are a little mixed), but in the situation of gross contamination may be less preferred than vein. Other less common options include cryo-preserved cadaveric homograft, or rifampin-soaked prosthetic.

And what are some of my options for constructing a bypass?

Short retrograde aorto-SMA bypass:

This bypass takes is origin off of the aorta just below the SMA, anastomosing typically end-side onto the SMA just below its origin in order to bypass ostial or very proximal disease. This is a relatively quick bypass, with only one field of dissection directly from the aorta below the SMA onto the proximal-mid SMA. The length of the bypass is very short, limiting concerns with kinking or twisting of the bypass. However, this may not always be feasible, as SMA disease often coexists with significant aortic disease. Additionally, the other bypasses described have better reported patency.(Scali et al. 2019)

Long retrograde R iliac-SMA bypass ("C-loop"):

This bypass originates from the right common iliac artery, which presents a number of distinct advantages over an aorto-mesenteric bypass. First, using the iliacs avoids the hemodynamic consequences of an aortic cross-clamp, which may be contraindicated depending on your patient's medical condition. Second, you can avoid showering, causing dissection, or otherwise injuring your clamp sites if you have significant disease in the mesenteric segment of the aorta, which is common in patients with chronic mesenteric ischemia. The graft should be tunneled in a gentle C-loop towards the SMA to avoid kinking or twisting. The proximal anastomosis is performed end-side on the iliac artery, and the distal can be performed either end-end or end-side depending on the anatomy of the disease. Especially when using prosthetic in a contaminated field, you can consider taking an omental flap to wrap or cover the prosthetic.

Antegrade supraceliac aorta-SMA bypass:

This bypass originates from the supraceliac aorta. If revascularization of both celiac and SMA is planned, a bifurcated graft can be selected. A side-biting aortic clamp can be used to mitigate the hemodynamic effect of an aortic cross-clamp. The tunnel to the SMA is created with gentle finger dissection in a retropancreatic plane, taking care to avoid injury to the SMV.

What other options have been described for treatment of acute mesenteric ischemia?

Retrograde open mesenteric stenting (ROMS)

So ROMS is a hybrid procedure involving an upper mid-line laparotomy that is used to evaluate the bowel. Through this incision, SMA exposure is obtained just as in a traditional open fashion at the root of the mesentery. The mid-SMA is then punctured under direct vision, and the area of disease is attempted to be treated from a retrograde approach back into the aorta. If bowel ischemia is found, the upper mid-line is easily lengthened into a traditional vertical laparotomy incision.(Milner, Woo, and Carpenter 2004; Blauw et al. 2014; Gustavo S. Oderich et al. 2018)

Endovascular treatment (percutaneous thrombectomy / thrombolysis / pharmacomechanical thrombectomy)

Some authors have described completely endovascular approaches to treatment of acute mesenteric ischemia. However, the major limitation is the inability to assess the bowel. These patients are frequently those who are deemed to be lower-risk for frank bowel ischemia or perforation, but the rates of laparotomy and bowel resection after these treatments have been described very high rates. Probably not your first answer for oral boards. (Lim et al. 2019)

11.1.3 Other Etiologies

What are some other, more rare etiologies of acute mesenteric ischemia?

Embolism is the most common etiology of acute mesenteric ischemia (40-50%), and thrombotic etiology composes another quarter to a third of these populations. The other two etiologies to consider are non-occlusive mesenteric ischemia and mesenteric venous thrombosis.

Perfect, so how does a non-occlusive mesenteric ischemia (or NOMI) patient differ from what we've been talking about?

So NOMI patients typically do not have a focal lesion like you see with embolism or thrombosis. What happens to these patients is that classically they're pretty sick patients with some predisposing factors, most commonly ESRD. On top of that, there was some clear inciting hemodynamic event causing sustained hypotension, such as recently getting a session of hemodialysis or undergoing cardiopulmonary bypass. The presentation is more indolent and less obvious than embolic or thrombotic acute mesenteric ischemia, and the imaging findings are more consistent with a diffuse vasospasm and hypovolemia picture.

Treatment is primarily conservative, with the emphasis on resuscitation and addressing whatever the underlying etiology is. Adjuncts to this include placement of infusion catheters into the affected vessel with infusions of vasodilators (most commonly papaverine at 30-60mg/hr) or prostaglandin (Wyers, Mark C and Martin 2019; Trompeter et al. 2002) NOMI with portal venous gas, pneumatosis and free air often indicate need for laparotomy.

How about mesenteric venous thrombosis?

This is the most rare and most difficult to diagnose. They have a very slow course, frequently with a lot of other workup already done. There's a wide variety of causes that have been reported, and any of the things that contribute to Virchow's triad have been reported (thrombophilia from coagulopathy or malignancy, venous stasis from abdominal hypertension or obesity, direct injury from trauma, surgery, or inflammation). The diagnostic test of choice is a CT with portal vein contrast, which most commonly identifies thrombosis in the superior mesenteric vein (but can also involve the IMV, portal vein, or splenic vein). The treatment of choice is therapeutic anticoagulation.(Acosta and Björck 2019)

11.2 Chronic Mesenteric Ischemia

11.2.1 Presentation and Evaluation

How does chronic mesenteric ischemia differ in its presentation?

Chronic mesenteric ischemia is characterized by post-prandial abdominal pain, typically 30-60 minutes after eating (think after gastric emptying time). This pain is usually severe, crampy, and resolves after minutes to hours of time. The pain leads to food fear and eventual <u>unintentional</u> weight loss. The next step in evaluation in these patients is a mesenteric duplex to evaluate for elevated velocities.

However, the clinical presentation here is key. Chronic mesenteric ischemia is pretty unlikely in patients who do not have this constellation of clinical symptoms, and it's really common for patients to be referred with imaging findings of elevated mesenteric velocities on duplex who have none of the clinical findings, and thus do not benefit from any intervention.

What are the duplex criteria for chronic mesenteric ischemia?

Yeah, so really the two vessels we're concerned with here are the celiac artery and the SMA. It's pretty rare that stenosis of the IMA results in clinically significant ischemia, and also pretty rare that isolated disease of either the celiac or the SMA would be enough to cause significant symptoms. Typically you see the classic symptomatology of chronic mesenteric ischemia in patients with occlusive disease in both the celiac and the SMA.

The key numbers to remember here are peak systolic velocities (that's PSV) of 200 cm/s for the celiac artery, and 275 cm/s for the SMA. These correspond to a stenosis of at least 70%. Remember, the **superior** mesenteric artery has **higher** velocity criteria, in case you forget. These classic numbers come from a study done in 1993, and many other studies have demonstrated other thresholds. Make sure to check with your friendly neighborhood vascular technician to see what the thresholds in your local lab are!

In addition to the PSV, you can also use the end diastolic velocity (EDV) to predict stenosis. For the EDV values in the celiac and the SMA, think approximately 50 cm/s predicts a stenosis of 50% (easy to remember, **50-50**). This is known as Bowersox criteria, and is actually >45, but 50 is easier to remember. (Bowersox et al. 1991)

You can also predict a higher degree of stenosis with EDV values. This is easy for the SMA, where an EDV of 70 cm/s predicts a stenosis of 70% (**70-70**). For the celiac, it's a bit higher, where an EDV of 100 cm/s predicts that same 70% stenosis.

So to review the numbers: (Moneta et al. 1991, 1993; Zwolak et al. 1998; AbuRahma et al. 2012)

- PSVs: 200 in the celiac or 275 in the SMA gives you 70% stenosis
- EDVs: 50 in the celiac or 50 in the SMA gives you 50% stenosis
- EDV #2: 100 in the celiac, or 70 in the SMA gives you 70% stenosis.

Final thought. When the celiac artery is severely narrowed or occluded, often you'll have retrograde flow coming from the SMA via collaterals (these are your pancreaticoduodenal arcades via the GDA). You can see this as backwards flow in the common hepatic artery, and this has 100% specificity for disease in the celiac artery.(Warncke et al. 2019)

11.2.2 Management

What's next in the approach to these patients?

Frequently they're going to get some kind of axial imaging, either CTA or MRA, to help plan for their revascularization, although this isn't strictly necessary. Often endovascular approaches are the mainstay of treatment. Remember that the angle of the SMA frequently favors a brachial approach, although you can also engage the SMA from a femoral approach with preformed catheters (think Sos, Cobra, etc) or steerable sheaths. Remember to watch out for median nerve compression from brachial sheath hematomas if you've opted for a percutaneous brachial approach, with a low threshold to evacuate the hematoma and perform median nerve decompression if the patient develops and symptoms in the median nerve distribution postoperatively. Also remember that a brachial sheath hematoma will not be easily palpable or visible at the skin level, so the neurologic exam postoperatively is key.

Single vessel SMA revascularization is widely accepted for chronic mesenteric ischemia without necessarily needing to revascularize the celiac axis. (Gustavo S. Oderich and Ribeiro 2019a; Foley et al. 2000)

Any other technical tips for endo intervention on the mesenteric arteries?

The best view for diagnostic angiography in mesenteric ischemia is the lateral view.

Most often balloon-expandable stents are selected for the mesenteric circulation. The precision of deployment and the increased hoop strength of a balloon expandable stent make them more favorable here over the radial force of a self-expanding stent. You can also flare the proximal end of a balloon-expandable stent into the aorta, making sure that you've really treated any ostial disease. Some authors advocate for use of embolic protection devices to prevent distal embolism, and other authors also advocate for the use of covered stents for the mesenteric circulation due to better patency.(Gustavo S. Oderich et al. 2013; Gustavo S. Oderich and Ribeiro 2019a)

What is the surveillance protocol for an SMA stent?

At most institutions it is 1mo, 6mo and annually thereafter. (Mohler et al. 2012; Zierler et al. 2018)

How about open revascularization?

In most centers, up to 80% of CMI is treated endovascularly. Open is often reserved for those that fail endovascular therapy.(Gustavo S. Oderich, Gloviczki, and Bower 2010; Gustavo S. Oderich and Ribeiro 2019a)

So we talked a lot about open revascularization earlier in the acute mesenteric ischemia segment. The techniques are broadly similar, with the exception that often for chronic mesenteric ischemia, frequently you'll be planning to revascularize both the celiac and the SMA. The approach for this is with a bifurcated graft from the supraceliac aorta, taking one limb down to the celiac artery, and the other limb tunneled in a retropancreatic fashion to the mid-SMA. Remember that this approach requires a supraceliac aortic cross clamp, which your patient may not be able to tolerate. But single revascularization of SMA is often acceptable with appropriate collaterals. Taking a retrograde bypass off of the common iliac artery as we previously described may be a better option for patients with significant comorbidities. A further option that completely avoids a bypass is to go right for a trapdoor endarterectomy of the celiac artery and the SMA, allowing you to address coral reef or proximal/ostial disease (such as flush occlusions) that are difficult to treat endovascularly.

11.3 Visceral Aneurysms

Aneurysms can occur in a diverse range of vascular beds in the visceral vessels. Size criteria for repair is dependent on the location and the etiology, with some aneurysms mandating repair at any size (gastric, gastroepiploic, gastroduodenal, pancreaticoduodenal, SMA, colic), and others considered appropriate to observe in asymptomatic patients until certain size thresholds are met. The SVS clinical practice guidelines are the optimal resource to review these criteria.(Chaer et al. 2020)



11.3.1 SMA Aneurysm

If symptomatic, most SMA aneurysms are mycotic in nature and require treatment. Treat with resection and saphenous vein interposition and 6 weeks antibiotic therapy based on in-traoperative cultures.(W.-K. Lee et al. 2008 Nov-Dec)

11.3.2 Splenic Aneurysm

Repair all pseudoaneurysms, or if size > 3cm, or at any size if of childbearing age. Consider splenectomy vaccines pre-treatment (covers encapsulated organisms Shigella, Haemophilus influenzae, and Neisseria).

Post splenic aneurysm embolization, L flank pain in stable patients likely represents splenic infarction. Treat with hydration and analgesia. Rates are lower in stented patients.(Kwong, Rockman, and Kashyap, n.d.; Piffaretti et al. 2007; C. Zhu et al. 2019)

Splenic artery embolization also an effective way to treat thrombocytopenia, particularly chemo induced, when splenectomy is too high risk.(Bhatia et al. 2015)

11.4 Median Arcuate Ligament Syndrome (MALS)

Can you tell me about median arcuate ligament syndrome and how that differs from chronic mesenteric ischemia?

Median arcuate ligament syndrome (MALS), which has many names (Dunbar syndrome, celiac axis compression syndrome, etc.) is a somewhat controversial entity that occurs when repeated compression of the celiac artery occurs against the median arcuate ligament during respiratory variation. The thing you want to visualize here is that during full exhalation, the lungs are completely emptied, and the diaphragm moves up at a sharper angle. This angulation kinks off the celiac artery more severely, so velocities in **exhalation** are **higher** in MALS. However, these findings are common in asymptomatic patients, and so just like chronic mesenteric ischemia, the clinical presentation is key. They'll have a similar presentation as chronic mesenteric ischemia patients, with post-prandial pain, food fear, and weight loss, but often the symptomatology is a little more indolent in these patients. Because of this, these patients have frequently gotten the million dollar workup for nonspecific GI pain. Duplex ultrasound of shows celiac PSV >200 and EDV >50, normalizes during deep inspiration, and post-stenotic spectral broadening.(Zwolak et al. 1998)

The treatment of choice for suspected MALS is a laparoscopic median arcuate ligament release, frequently performed by a MIS/foregut surgeon. A key point here is that you don't want to be fooled into putting a stent in these patients before they've gotten their median arcuate ligament release, because the dynamic motion of the diaphragm is likely to crimp or bend the stent if that hasn't been treated yet. It may be that some patients benefit from endovascular
treatment after release, though, as some think that the chronic damage from MALS can result in intimal damage/scarring that persists even after the extrinsic compression is treated by the median arcuate ligament release.

11.5 SMA Syndrome

What is SMA syndrome?

SMA syndrome, also called Wilkie's Syndrome is a rare entity where the 3rd portion of the duodenum gets compressed between the SMA and the aorta, causing a functional gastric outlet obstruction. Patients are typically emaciated, having lost a significant amount of weight before their symptom onset. What's happened is they've lost the retroperitoneal fat pat that normally surrounds the SMA, and so the angle between the SMA and the aorta becomes more acute, pinching off the duodenum.

Classic findings in fluoroscopic swallow in SMA syndrome are delayed gastric emptying, a dilated duodenum, to and fro peristalsis in the proximal duodenum and cutoff in the third portion. EGD can show pulsation in the 3rd portion of the duodenum.(Warncke et al. 2019)

The treatment of choice here is enteral feeding with a nasojejunal tube or other surgically placed tube and TPN, because what really will help them here is weight gain. GJ bypass or duodenal mobilization is sometimes needed for treatment, but is controversial. It's not really a mesenteric vascular disease, but it sometimes shows up on exams as a related entity.(Welsch, Büchler, and Kienle 2007; Merrett et al. 2009)

11.6 Uterine Fibroids

Symptomatic uterine fibroids are now often treated with uterine artery embolization. Many patients require bilateral uterine artery embolization (UAE), but may only need unilateral if blood supply is unilateral. Infarcted tissue can be expelled or become infected even years after initial embolization. (Stępniak 2018)

12 Renal

Authors: Cullen McCarthy and Matthew Edwards

12.1 Renovascular Ischemic Disease

12.1.1 Pathophysiology

What is renovascular hypertension?

Hypertension as a result of progressive renal artery stenosis. While renal artery stenosis is a relatively common finding in older patients with hypertension, it's relatively uncommon as the primary cause of hypertension.

What is ischemic nephropathy?

Decreased renal function and/or chronic kidney disease that results from atherosclerotic renal artery stenosis due to a reduction in glomerular filtration rate (GFR) and rise in creatinine produced by any cause of diminished renal blood flow. Primary mechanism is chronic hypoperfusion. (Rickey and Geary 2019a)

Who is at risk for ischemic nephropathy?

Renal injury can develop in anyone with a kidney or kidney region beyond a critically stenotic artery. Any of the following flow limiting lesions can cause ischemic neuropathy OR renovascular hypertension.

- Atherosclerotic disease of the aorta or renal arteries (most common) you'll generally see this in patients over 45 years old, likely with known PVD/CAD—though it can occur as an isolated renal lesion—usually involving the aortic orifice or the proximal main renal artery.
- Fibromuscular dyplasia These patients are most often women under the age of 50 years and typically involves the mid- or distal main renal artery or the intrarenal branches.
- Coarctation of the aorta or mid-aortic syndrome

- Large vessel arteritis (rare) although a rare cause of renovascular hypertension, this is the most common presentation of Takayasu's arteritis (60%). First line therapy is open revascularization. (Ehlert and Abularrage 2019b; F. A. Weaver et al. 2004; F. P. Zhu et al. 2012)
- Iatrogenic causes Rarely iatrogenic from malposition or migration of endovascular aortic stent grafts over the renal orifices.

The prevalence of renovascular hypertension in patients with mild hypertension is probably less than 1%. However in patients with acute progression, severe or refractory hypertension the prevalence may be as high as 10 to 40%. However, disease may not entirely be attributable to a flow limiting lesion. There may be a component of cholesterol embolization or small vessel atherosclerotic disease that may contribute to progression of renovascular disease.

The effect of the stenosis may not be clinically apparent due to compensatory function of the unaffected contralateral kidney. We know 5-22 % of patients 50 years or older who have advanced CKD have some degree of renal artery stenosis and 23-54% of those patients with stenosis have bilateral renal artery disease.

- Most cases of renal artery stenosis affect one side much more than the other; one kidney is affected with the second kidney being essentially normal, hence the designation "unilateral" disease.
 - In patients with unilateral renal artery stenosis, contralateral kidney is damaged by uncontrolled hypertension and circulating angiotensin II and aldosterone. (Rickey and Geary 2019a)
- Patients who are diagnosed with ischemic nephropathy usually have high-grade stenosis of both renal arteries or stenosis to a solitary functioning kidney.
 - It is this subpopulation of the disease that merits specific consideration because of its additional contribution to fluid retention, loss of kidney function, and congestive heart failure.
- Bilateral renal arterial stenosis is associated with more widespread atherosclerotic disease, higher serum creatinine levels, and higher mortality than unilateral disease

However, flow limiting lesions may be an "incidental" finding in patients who have CKD or hypertension that is caused by a separate disorder (i.e. diabetic nephropathy and essential/primary hypertension). Because of this, it can be very difficult to distinguish between patient whose disease is induced by renal artery stenosis and those who have alternative causes of CKD or renovascular hypertension.

Renal (especially bilateral) hypoperfusion induces activation of the renin-angiotensinaldosterone system which increases vascular tone and impairs sodium excretion resulting in expansion of the extracellular fluid volume.

The clinical findings that suggest that ischemic renovascular disease is an underlying cause of hypertension, include:

- Recent or rapid development of severe hypertension.
 - Relatively specific for renovascular hypertension and is the strongest predictor of anti-hypertensive benefit from revascularization.
- Severe hypertension that may be treatment resistant.
 - Some patients with ischemic nephropathy are normotensive, which may be due in part to a reduced cardiac output
- Recent reduction in kidney function (over last 6-12 months) with or without the administration of angiotensin-converting enzyme (ACE) inhibitors or angiotensin receptor blockers (ARBs).
 - Rise in serum creatinine is more common with agents that block the reninangiotensin system than with other antihypertensive drugs because glomerular filtration rate (GFR) often depends upon the efferent arteriolar actions of angiotensin II in this setting. RAS can worsen ACE-I and ARB induced renal dysfunction due to systemic hypotension, efferent arterial vasodilation, and reduced glomerular hydrostatic pressure, in turn lowering GFR. (Rickey and Geary 2019a; Schoolwerth et al. 2001)
 - More common with bilateral as compared with unilateral disease because there is hemodynamic compromise to the entire renal functional mass
 - This usually resolves after withdrawal of the drug.
 - Restoring the renal blood supply in such cases can recover the ability to use these drugs for blood pressure control.
 - Significant variability of serum creatinine concentration that may be due to changes in volume status
- A rapid rise in arterial pressure associated with sudden development of left ventricular failure ("flash pulmonary edema").
 - This finding is more common with renal artery stenosis because bilateral disease is also associated with diuretic resistance and sympathetic adrenergic activation
- Deterioration of kidney function after placement of an endovascular aortic stent graft IATROGENIC

• EVAR with isolated L renal coverage -> splenal renal bypass, ligate splenic artery distally and then perform end-to-end anastomosis to renal artery. (Benjamin and Hansen 2019)

12.1.2 Evaluation

So how can we determine who has CKD or hypertension due to renovascular stenosis that we can actually help?

This is probably the most important question since in this whole disease process.

To start, if a patient has the clinical manifestations of ischemic nephropathy or renovascular hypertension as we discussed above, a presumptive diagnosis of ischemic nephropathy can be made if there is radiologic documentation of significant stenosis (usually more than 70 % luminal occlusion) of both renal arteries or of one renal artery to a solitary functioning kidney.

But how do we know the vascular occlusive disease posing critical hemodynamic limitation to kidney function?

- Doppler ultrasound criteria (Hoffmann et al. 1991; Zierler and Dawson 2016)
 - PSV >200 cm/sec or renal aortic ratio >3.5 and post stenotic turbulence, suggests >60% stenosis, but some labs have proposed higher velocities (250-285 cm/s).(Schäberle et al. 2016; Rickey and Geary 2019b)
 - Elevated velocities can be seen with tortuosity, but this should be able to be confirmed with B-mode. For more see Section 20.6.4
- Generally, luminal occlusion of at least 60 to 75 % is required to limit blood flow and reduce perfusion pressure. CTA or angiogram with CO2 are helpful to delineate the anatomy. MRA can be used for evaluation:
 - Pro no radiation, good imaging of distal renal arteries, no degradation from ostial calcium.
 - Cons requires GAD -> interstitial fibrosis in CKD (Galan, Cowper, and Bucala 2006) and degrades with motion and respiration (Nelson et al. 1999)
- This degree of stenosis is usually associated with a measurable translessional "pull-back" pressure gradient of 10 to 15 mmHg, which can be measured during conventional angiography.

Can we tell who has cortical hypoxia through diagnostic tests?

To some degree. Cortical perfusion can be measured by blood oxygen level dependent magnetic resonance (BOLD-MR). Additionally, inflammatory markers sampled from renal veins of stenotic kidneys correlated strongly with the degree of hypoxia (as measured by BOLD-MR), particularly after correction of the stenosis with angioplasty. Blood Oxygen Level Dependent Magnetic Resonance (BOLD-MR) can identify levels of cortical hypoxia which may be associated PSV > 385 cm/sec or reduced GFR (20 to 25 ml/min).

So we have a patient with evidence of malperfused kidneys, either through worsening renal function or uncontrolled hypertension, with known discrete stenoses, and we even got a BOLD-MRI which confirms it. Let's just revascularize them and be done with it?

Although vascular stenosis or occlusion can initiate these processes, long-standing ischemia causes parenchymal injury characterized by inflammation and fibrosis which eventually becomes an irreversible process. At some point, restoring renal blood flow provides no recovery of kidney function or clinical benefit. We can at least have some idea by considering the renal resistive index, the six-month trajectory of kidney function, and the size of the kidneys or by performing a kidney biopsy (which is not usually done).

- None of these factors predict the outcome of revascularization with certainty.
- Improved and validated methods to evaluate the salvageability of kidney function in this disorder are greatly needed and are the holy grail of this disease process.

Here are some additional criteria that can be useful adjucts to duplex velocities and cross sectional imaging.

- Renal Resistive index Some studies indicate that elevated resistive indices in segmental vessels (above 0.80) measured by duplex ultrasound denote poor prognosis for renal recovery while a low resistive index is a favorable sign. Initially, many were hopeful that this would be a very useful indicator, but not many have been able to reproduce the initial studies describing this association.(Radermacher et al. 2001)
- **Kidney size** Very small kidneys (less than 8 cm in longest diameter) are usually considered unlikely to recover after revascularization.
- **Kidney biopsy** (not usually performed) Previous studies suggest that biopsy demonstrating preexisting atheroembolic changes and interstitial fibrosis indicate a limited potential for recovery.
- Comparison of kidney morphology with kidney function Some investigators have recommended assessing morphological parameters, such as renal parenchymal volume and cortical thickness with MRI, and comparing these parameters with kidney function measured by radionuclide scanning. A kidney with normal morphology in the

setting of renal artery stenosis and reduced function may indicate a "hibernating kidney" which may respond to revascularization.

So how do we get a definitive diagnosis?

A definitive diagnosis is not usually made before revascularization. In practice, confirmation of the diagnosis is based upon stabilization or improvement of the GFR after successful revascularization.

12.1.3 Management

12.1.3.1 Medical Management

Now we think our patient's renal artery stenosis maybe is causing hypertension or decline in renal function and we can possibly reverse it... can patients with renovascular hypertension just be treated with anti-hypertensives, such as ACE inhibitors?

Many of these patients can be treated with medical therapy alone without loss of function or irreversible fibrosis, sometimes for many years.

Some in-human studies demonstrate that even with moderate reduction in renal perfusion pressure (40%) and in renal blood flow (30%), GFR may be reduced but renal cortex and medulla tissue oxygenation may be able to adapt and avoid developing severe hypoxia. However, there are limits to this adaptation.

As the hypertension is treated, we're lowering the pressure gradient across the stenosis and can actually increase the degree of renal malperfusion and worsen the renal function. Oftentimes this loss of kidney function is a reversible consequence of antihypertensive therapy but it limits our ability to control the hypertension medically without causing further damage to the kidney.

Further reduction in renal function can also reflect progressive narrowing of the renal arteries and/or progressive intrinsic kidney disease as more advanced vascular occlusion, corresponding to a 70 to 80% narrowing of the renal artery, leads to demonstrable cortical hypoxia.

In addition, all of these patients should receive secondary prevention of cardiovascular morbidity with aspirin, statins, smoking cessation, and, in patients with diabetes, glycemic control. We're addressing or prevention progression of stenosis in those with atherosclerotic disease.

For more on the medical management of vascular disease, see Section 15.2.2.1

Who should we fix operatively?

Some but not all patients should undergo revascularization, patient selection single most important factor depends upon the hemodynamic severity and likely recoverability of kidney function. *Again, the factors most closely associated with recoverability include:*

- A recent onset of hypertension prior to the diagnosis of renal artery stenosis (strongest predictor of response to treatment)
- Failure of optimal medical therapy to control the blood pressure or intolerance to optimal medical therapy (i.e. deterioration of renal function during antihypertensive drug therapy, such as ACE)
- Recurrent flash pulmonary edema and/or refractory heart failure
- Otherwise unexplained progressive renal insufficiency, particularly if proteinuria is absent
 - CKD stage 3a and 3b most likely to benefit from revascularization. (Singer et al. 2009) Lower GFR likely to progress to ESRD.
- Degree of stenosis, age, pre-procedure BP control and meds are not associated with improvement. (Textor, Misra, and Oderich 2013)

12.1.3.2 Endovascular

Before undergoing diagnostic angiography or potential intervention, the patient's ACE inhibitor and metformin should be stopped. Continuing or even initiating anti-platelet therapy and statins are important to prevent atheroembolization during wire manipulation of the renal arteries and stent placement.(Hiramoto et al. 2005)

But do we have any good data proving our interventions help?

Early on, observational studies demonstrated a high rate of procedural success with percutaneous transluminal renal angioplasty (PTRA) and stent placement (\sim 85%) in patients with ostial atherosclerotic disease, as well as a high rate of clinical success measured by improvements in blood pressure and kidney function in 50 to 75 % of subjects.

Unfortunately, randomized trials showed no additional benefit from stenting when added to medical therapy with respect to blood pressure control, renal function, cardiovascular events, and mortality. But these studies have their own limitations.

The one we keep hearing about is the CORAL trial.(Cooper et al. 2014) Tell me about that.

- Cardiovascular Outcomes in Renal Atherosclerotic Lesions (CORAL) trial
- 947 patients (80 % had unilateral disease) who met the following two criteria:
 - Unilateral or bilateral atherosclerotic renal artery stenosis >60 % if diagnosed with conventional angiography, peak systolic velocity >300 cm/second if diagnosed by duplex Doppler ultrasonography, Luminal narrowing >80 % if diagnosed with magnetic resonance angiography or computerized tomography angiography (or >70 % with additional evidence of renal ischemia)

- Systolic hypertension despite two or more antihypertensive medications and/or an estimated glomerular filtration rate (eGFR) ${<}60~{\rm ml/min}/{1.73}~{\rm m2}$ that was presumably due to the stenosis.
- All patients received antiplatelet therapy plus best medical therapy including ARB
- Revascularization had no additional effect on the primary outcome (a composite of cardiovascular or renal death, stroke, myocardial infarction, hospitalization for heart failure, a reduction in eGFR by more than 30 %, or end-stage renal disease) as compared with medical therapy alone (35.1 versus 35.8 %).
- No effect on any of the individual components of the primary outcome.
- Low procedural complication rate $\sim 2\%$
- Similar findings in the ASTRAL trial (A. Investigators 2009)

Well, that sounds pretty convincing, however there are limitations on existing treatment data:

<u>Considerable selection bias</u> – For the most part, the patients enrolled in these trials did **not** meet the criteria for selecting patients likely to benefit from intervention (i.e. short duration of blood pressure elevation, hypertension resistant to medical therapy, recurrent flash pulmonary edema):

- CORAL (Cooper et al. 2014)
 - Patients hospitalized for heart failure within 30 days of screening for the trial were excluded, thereby limiting the number of trial participants with recurrent flash pulmonary edema.
 - Mean number of antihypertensive medications used by CORAL participants at baseline was 2.1- many had not failed optimal medical therapy
 - More than 25 % had controlled blood pressure upon entry into the trial.
 - Mortality and event rates lower than in most previous registries, suggesting that many high-risk patients were not enrolled.
- ASTRAL (A. Investigators 2009)
 - Large number of patients had stenoses that were probably not clinically significant (50 to 70 %), and patients were excluded if their primary doctors felt that they "definitely" needed revascularization.

Results of the trials differ substantially from observational reports of "high-risk" subsets

• For the most part, patients selected by their treating clinicians to undergo revascularization have derived greater benefit from revascularization than did patients enrolled in the trials who were randomly assigned to revascularization

We've determined our patient is an appropriate candidate for intervention, and we don't fully buy into CORAL, what can we do?

Percutaneous renal angioplasty/stenting in addition to medical therapy are most commonly employed if technically feasible. Most amenable lesions to angioplasty are those producing incomplete occlusion in the main renal artery.

• Total occlusions and ostial lesions extending into a respond well to angioplasty alone due to elastic recoil.

In general, the effects of revascularization on blood pressure were greater in bilateral disease, but effects on renal function and mortality did not differ in those with bilateral as compared with unilateral stenosis .

Most atherosclerotic lesions are now treated with primary stenting to avoid rapid development of restenosis.

- Quick results: maximum antihypertensive response is generally observed at 48 hours after the procedure, but BP levels and antihypertensive drug requirements often change over subsequent weeks
- A higher initial primary success rate, defined as less than 50 % stenosis (88 versus 57 %).
- At six months, a higher patency rate (75 versus 29 %) and a lower restenosis rate (14 versus 48 %).
- Twelve patients assigned to PTRA alone underwent stenting because of treatment failure within six months. These patients had a similar blood pressure response as those initially treated with stenting.

Performing a renal angiogram and intervention (Edwards and Cooper 2019)

- 1. Supine with arms overhead or straight out to the sides
- 2. LAO 15-20deg
- 3. Femoral access
- 4. Flush catheter placed just above the renals. Often start with CO2 and selective contrast runs.
- 5. Breath hold, high frame rate and non-DSA due unavoidable patient movement
- 6. Often utilize an 0.014 to cross the lesion, and advance a guide catheter to secure access.

- 7. Many advocate for primary stenting in atherosclerotic disease with uncovered stents to minimize manipulation. Some believe that covered stents carry benefit, but they are more expensive, and the evidence does not directly support this added cost.
 - 2005 ACC/AHA guidelines on peripheral artery disease recommends that a stent be placed in patients undergoing percutaneous transrenal angioplasty (PTRA) for treatment of atherosclerotic renal artery stenosis (Hirsch et al. 2006)
 - PTRA without stent placement is rarely performed unless the anatomy precludes stenting.
 - POBA without stenting is generally less successful and associated with more complications (i.e. atheroemboli)

12.1.3.2.1 Outcomes

Unilateral disease

- PTRA alone results in normalization of blood pressure (removal of antihypertensive drug therapy) ~8-20%. Some improvement 50-60%.
 - Failure rate $\sim 20-30\%$
 - Restenosis rate of 8 to 30 % at two years (without stent)
 - Better results with unilateral fibromuscular disease.
 - Less consistent for patients with chronic hypertension compared with patients who have an acute elevation in blood pressure
- In-stent restenosis rates are closer to ${\sim}11{\text{-}}17\%,$ with $11{\text{-}}39\%$ during the first one to two years
 - Detected as a rise in blood pressure requiring more intensive therapy
 - Angioplasty/stenting injures the vascular endothelium, which may result in restenosis.
 - Symptomatic stenosis leading to a rise in blood pressure or a fall in GFR are less common and are reported in 10 to 20 % of patients

Bilateral disease

- 25-30% will recover kidney function to a meaningful degree, sometimes avoiding progression to end-stage kidney disease (ESKD) and/or the need for renal replacement therapy.
- $\sim 50\%$ will have little immediate change in kidney function but will "stabilize"

- ${\sim}20\%$ will have a progressive deterioration of kidney function, sometimes related to the procedure

12.1.3.2.2 Complications

Complication rate with percutaneous transluminal renal angioplasty with or without stenting is between 5 and 15 %

- Mostly minor: puncture site hematoma and renal artery dissection.
- Serious complications more rare: renal artery thrombosis or perforation, AKI 2/2 atheroembolic disease (~1%)or radiocontrast agent injury.
- Mortality exceedingly rare

12.1.3.2.3 Surveillance

Follow-up of patients who have had a renal artery stent should include serial measurements of blood pressure and estimation of GFR.

- Post-stent duplex ultrasound at 2-4weeks with
- Repeated examinations on a quarterly basis (not much data)
- Patients who develop an increase in pressure or reduced GFR after stenting should undergo duplex ultrasonography to identify restenosis
- Re-treatment with angioplasty with or without repeat stenting can be attempted, but the restenosis rate after repeat angioplasty is increased.
 - Surgical reconstruction may be pursued in patients with recurrent episodes of restenosis and loss of kidney function.

12.1.3.3 Open Surgery

What about an open operation?

Surgical revascularization used in addition to medical therapy is less common since the widespread application of effective antihypertensive drug therapy and endovascular stents in the mid 1990s.

So who still gets open repair? (Benjamin and Hansen 2019)

- Younger patients
- Unfavorable anatomy (i.e. occlusion or branch disease)
- Failures of endovascular therapy (i.e. in stent restenosis)

• Need for concomittent a ortic revascularization.

How do we do it?

Involves by passing the stenotic segment or of removing a small atrophic kidney with nearly complete arterial occlusion.

- From the aorta or hepatorenal or splenorenal bypass to avoid diseased aorta.
- Bilateral: either bilateral repair or unilateral repair with contralateral nephrectomy of a nonfunctional, atrophic kidney.

Bilateral ostial disease in a young patient can be treated with transverse arteriotomy and bilateral renal endarterectomy. Close primarily or with PTFE/polyester patch. Second line is bilateral bypass with GVS. (Benjamin and Hansen 2019)

How do outcomes compare to PTA/stenting?

Equally or more effective than PTRA in the treatment of a therosclerotic disease, with cure of or improvement in the hypertension occurring in 80 to 95 % of patients. Cure of hypertension after surgery is most likely in patients who have been hypertensive for less than five years

Lack of complete response was usually associated with one of two factors:

- Presence of underlying primary/essential hypertension
- Development of intrarenal vascular disease due to exposure of the contralateral kidney to the elevated blood pressure.

Guidelines recommendations

2005 American College of Cardiology/American Heart Association (ACC/AHA) guidelines (Hirsch et al. 2006) Open surgery is reserved primarily for patients with:

- Multiple small renal arteries
- Early primary branching of the main renal artery
- Require a rtc reconstruction near the renal arteries for other indications (i.e. aneurysm repair or severe a ortoiliac occlusive disease)
- In order to avoid manipulation of a highly diseased aorta or failed endovascular stents, using extra-anatomic bypass (i.e. splenorenal, ileorenal, or hepatorenal bypass).

12.1.3.3.1 Complications

In-hospital mortality: ${\sim}3\text{-}10~\%$ in high volume centers

- Risk factors diffuse atherosclerosis, advanced age, chronic kidney disease, heart failure, or chronic lung disease.
- No deaths in 105 procedures for fibromuscular dysplasia (FMD).

12.2 Fibromuscular Dysplasia (FMD)

12.2.1 Demographics

You mentioned FMD as a cause of renovascular hypertension, tell me more about that...

Fibromuscular dysplasia (FMD) is a non-atherosclerotic, non-inflammatory disorder that results in progressive arterial stenosis, occlusion, aneurysm, dissection, or tortuosity.(Jeffrey W. Olin and Sealove 2011)

• Virtually always diagnosed radiographically – formerly pathologically, but rarely sent for specimen in modern diagnosis or treatment

How do we classify it?

Most commonly classified by angiographic appearance:

- Multifocal FMD (more common)
 - Angiographic appearance of a "string of beads."
 - Corresponds pathologically to medial fibroplasia, the most common histologic type, and to perimedial fibroplasia, which is less common.
- Focal FMD (less common)
 - Angiographic appearance of a "circumferential or tubular stenosis"
 - Corresponds pathologically to intimal fibroplasia but medial hyperplasia and periarterial hyperplasia may also have a focal appearance.
- These two different angiographic subtypes of FMD (multifocal and focal) have different phenotypic presentations and natural history. If FMD is, in fact, a single disease?

Where does it occur?

FMD has been observed in nearly every arterial bed. Two thirds of patients have multiple arteries involved.

- Involvement of the renal arteries $\sim 75-80\%$
- Involvement of the extra cranial cerebrovascular arteries (i.e. carotid and vertebral arteries) ${\sim}75\%$

Who has FMD?

• ~90% of cases in a dults are in women.

- No female predominance among children with FMD.
- Mean age at diagnosis was 52 years, with a range of 5 to 86 years
 - In the past, it was believed that FMD was a disease of young women. However, older now know to make up a large proportion of affected
- 35-50% of cases in children and 5-10% of cases in a dults under the age of 60 years with renovascular hypertension
- Often an incidental finding:
 - 4.4% of potential kidney donors had evidence of FMD.
 - CORAL trial:
 - FMD was discovered in 5.7% of the total study population (8.8% of enrolled females)

12.2.1.1 Etiology

What causes FMD?

The exact etiology of FMD remains unknown, but some mechanisms have been proposed

- Most often results from medial fibroplasia (60-90% of cases). Collagen deposits in the media result in elastic fibrils and fibromuscular ridges. (Jeffrey W. Olin et al. 2014)
- 'Genetics may play an important role in development
 - Some studies report autosomal mode of inheritance with variable penetrance
 - Potential association with a single nucleotide variant in the phosphatase and actin regulator 1 gene (PHACTR1)
 - Variant rs9349379 is also a risk locus for coronary artery disease, migraine headache, and cervical artery dissection.
- Predominance of young/childbearing age women hormonal influences are thought to play a role
 - Remains unproven.
- Mechanical factors such as stretch and trauma unproven.

12.2.1.2 Presentation

Does FMD present differently that the atherosclerotic renovascular disease we talked about?

- Varies widely depending on artery affected and as it results from:
- Ischemia related to stenosis
- Dissection and occlusion of major arteries (renal infarction, stroke, myocardial infarction)
- Rupture of aneurysms
- Embolization of intravascular thrombi from dissection or aneurysms

What are the common presenting symptoms and signs

Manifestations of renal FMD (i.e. hypertension, flank pain) are more likely to occur in men, as are arterial dissections and aneurysms.

- Most common presenting signs:
- Hypertension -67% (66% of women and 74 % of men)

But overall hypertension is the most common manifestation of renal artery FMD in both genders

• Flank pain and abdominal pain can result from ischemia, aneurysm rupture, or dissection of renal and mesenteric arteries, respectively.

Are these dissections common?

- High prevalence of aneurysm and/or dissection
 - Aneurysm (22%) and dissection (26%).
 - 34% of aneurysms were renal
 - 11% of dissections were renal
 - 42% had an aneurysm and/or dissection.

So should we screen for these dissections in a patient with known FMD?

- Every patient diagnosed with FMD should have one-time, head-to-pelvic CTA (or MRA) is an alternative.
 - CTA of the neck and head on one day followed one week later by CTA of the chest, abdomen, and pelvis

When should we suspect FMD?

- Hypertension (particularly in a woman under the age of 60 years) with findings that would prompt an evaluation for secondary hypertension:
 - Severe or resistant hypertension.
 - Onset of hypertension before the age of 35 years.
 - A sudden rise in blood pressure over a previously stable baseline.
 - A significant increase in the serum creatinine concentration after the institution of therapy with an angiotensin-converting enzyme (ACE) inhibitor or angiotensin receptor blocker (ARB) in the absence of an excessive reduction in blood pressure.
 - An epigastric/abdominal bruit.
- Renal artery dissection (or carotid, vertebral, coronary)
- Aneurysm in a visceral, carotid, vertebral, or intracranial vessel.
- Renal infarction.

12.2.2 Evaluation

How do we diagnose this and/or distinguish it from renovascular atherosclerotic disease?

- Confirmed by diagnostic imaging that reveals consistent findings
- Noninvasive imaging test is usually performed first. This includes CTA, MRA, and Duplex ultrasound.

Let talk about CTA...

- CTA is preferable due to higher spatial resolution than MRA, less dependence upon technical expertise, and a shorter scan time
- Excellent diagnostic accuracy for FMD of the main renal arteries, although the sensitivity decreases when FMD is only present in the smaller branch renal arteries.
- Multi-row detector CT scanners may provide additional advantages in diagnosis and surveillance because they offer more rapid image acquisition, variable section thickness, three-dimensional rendering, diminished helical artifacts, and smaller contrast requirements.

What about MRA?

• Inconsistent detection of FMD and is performed if CTA is contraindicated

- The spatial resolution in the branch vessels is not adequate, and artifact may occur, suggesting "beading" when none is present.
- May miss mild FMD.
- Can be useful for detecting aneurysms and dissections can be difficult to diagnose on duplex ultrasonography.(Renaud et al. 2012)

And finally, Duplex ultrasonography

- Detects elevated blood flow velocities in the mid and distal portions of the renal artery, most common locations for FMD.
- Increased peak systolic velocity, turbulent blood flow, and tortuosity of the mid and distal artery.
 - % diameter stenosis reports less helpful and usually inaccurate
- lowest spatial resolution of all of the cross-sectional imaging modalities
- most operator dependence
- first choice only in high-volume centers with extensive expertise in this technique

What about non invasive testing?

- DSA is performed in patients if there is a high clinical suspicion of FMD, and treatment with revascularization is planned if a stenosis is found.
 - Can improve visualization of the arteries by eliminating background soft tissue and bone and has higher spatial resolution than any of the other imaging modalities
 - Can measure the pressure gradient across the stenosis
 - Pressure decrease threshold of 10 % or more of the mean pressure should be used to decide whether a lesion is hemodynamically significant
 - IVUS and optical coherence tomography (OCT) can help identify dissection or intramural hematoma or determine response to angioplasty.
 - Negative DSA excludes a diagnosis of FMD in the vascular bed that was imaged.

Any place for pathological diagnosis in modern therapy?

- Histopathology (and histologic classification) is **no longer** part of the diagnosis.
 - Only in the rare patient who requires surgical revascularization or resection of an aneurysm.

12.2.3 Management

12.2.3.1 Medical Management

So how do we treat this, what warrants intervention and are those interventions different than what we offer atherosclerotic disease of the renal arteries?

- All patients with FMD should be placed on antiplatelet therapy (ASA) unless otherwise contraindicated
- Antihypertensive therapy
 - Most patients will require antihypertensive therapy, even if they undergo revascularization.
 - Majority of patients with **focal** FMD have their blood pressure cured with angioplasty

12.2.3.2 Endovascular

But what about revascularization?

- Revascularization Goal: control of hypertension
 - BP can be controlled in most adults with multifocal FMD with a mean of two antihypertensive medications
 - Weigh risks and benefits in well controlled hypertension.
- No randomized trials comparing revascularization with medical therapy

Well then who do we treat?

- Recent-onset hypertension, with goal to cure hypertension.
- Resistant hypertension despite compliance with an appropriate three-drug regimen.
- Patients unable to tolerate antihypertensive medications or who are non-compliant with their medication regimen.
- Adults with bilateral renal FMD, or unilateral renal FMD to a single functioning kidney, and unexplained progressive renal insufficiency thought to result from renal artery stenosis
- Hypertensive children.

• may be at higher risk than adults for progressive renal parenchymal loss, and therefore could benefit from revascularization even if their hypertension can be well-controlled with one or two antihypertensive medications.

And what kind of results do we get with revascularization?

Hypertension is cured or improved following revascularization in a large proportion of patients with FMD.

- Much better than 2/2 atherosclerosis
- Varies considerably from study to study, although hypertension control improves in most patients and depends in large part upon the definition of cure.
- Not good data on stabilization of either GFR or renal size in patients with FMD.

What options do we have in terms of revascularization and do we have good results treating FMD with endovascular or open interventions?

- Patients most often treated with angioplasty alone with good success. (Davies et al. 2008; T. L. Jenkins et al. 2015)
- Improvement in blood pressure (including those with and without cure) was similar with PTA as compared with surgery (86 versus 88 %).
- Older age and longer duration of hypertension prior to revascularization were significantly associated with a lower cure rate.

How do these compare?

- PTA achieves similar technical success and is associated with a lower risk of adverse events in observational studies
- Most patients with FMD who are selected for renal revascularization have PTA rather than surgery
- Major adverse events were more frequent with surgery (15 versus 6 %).

So why choose open surgery?

- Cure rates were higher with surgery (54 versus 36 %).
- Surgery rather than PTA if PTA fails or if the arterial anatomy is not amenable to PTA
 - Patients with small renal arteries (<4 mm), with branch renal artery disease, or with extensive intimal fibroplasia.

So how do we perform Percutaneous transluminal angioplasty for FMD?

• Angioplasty is often performed through a transfemoral approach, but concomitant stenting is rare, unlike PTA for atherosclerotic RAS

Why not place a stent?

- Patients do very well with angioplasty alone, no reason to place a stent.
 - Lesion is so fibrotic that the pressure gradient cannot be obliterated with an angioplasty, a stent will not correct this problem
 - Such patients should be referred for surgery.
- Usually have stenoses in the mid and distal portions of the artery rather than at the ostium or proximal portion (as occurs with atherosclerosis).
 - Should surgical revascularization become necessary due, for example, to in-stent restenosis, patients may require more complex branch repair to bypass the occluded stent since the stent often covers the renal artery up to the point of the first intrarenal branch.

Do we ever place stents?

Two situations that may require stenting include dissection or perforation (rare) after angio-plasty.

And we're getting good outcomes with PTA alone?

- Technical (angiographic) success rates for PTA 83-100
- Rate of restenosis 12-34% over follow-up intervals of six months to two years
 - Difficult to determine if patients with FMD develop restenosis, or if the lesion was not completely treated correctly the first time.
 - Not necessarily associated with recurrent hypertension.

But generally we can achieve significant and sustained reductions in systolic blood pressure, diastolic blood pressure, serum creatinine, and number of antihypertensive agents.

• Systolic blood pressure response was better in patients with FMD affecting the main renal artery than in patients with branch vessel involvement.

Any specific technical tips?

- Cutting balloon angioplasty should be avoided because there is an increased risk of rupture
- Post angioplasty visual inspection alone is **not** accurate.

- Measure pressure differential using a pressure guidewire, with a mean gradient of <5 mmHg across the treated segment suggesting a satisfactory result
 - Measure before and after angioplasty
- Post-procedure renal duplex scanning
 - Degree of turbulence is less prominent, and velocity elevation in the mid-distal renal artery returns to normal.
- Intravascular ultrasound (IVUS) or optical coherence tomography (OCT) is occasionally used to evaluate the elimination or reduction of various endoluminal defects.

What should we do if it doesn't work?

If either has no improvement in blood pressure or an initial improvement followed by recurrence, repeat angiogram and PTA.

• Restenosis may actually represent inadequate angioplasty during the first procedure

Persistent HTN despite technically successful PTA suggests that the cause of hypertension is unrelated to fibromuscular disease or is related to small vessel disease within the kidney (nephrosclerosis) due to longstanding hypertension.

What kind of complications do we see after this?

- Mostly related to vascular access
- Rarely: renal artery perforation, dissection, or segmental renal infarction may occur.
- Decreasing over time- 16 % in 1998 to 3 % in 2001

12.2.3.3 Surgery

Lets switch gears to open revascularization?

Aortorenal bypass with a saphenous vein graft is the most common technique. Artificial graft material used occasionally.

For everyone? What about for pediatric patients?

Pediatric patients: hypogastric artery grafts are used or else aortic reimplantation of the renal artery is performed because vein grafts become aneurysmal

How does this compare again to PTA?

Similar success rates compared to PTA (82-89% patency) but with higher morbidity.

• Perioperative mortality appears to be very low (~1.2%)

• Usually limited to complex cases so success and complication would probably be higher if simpler cases were included.

What does monitoring and follow-up look like for these patients?

Medical management only:

- Renal artery stenosis and kidney dysfunction may progress despite good blood pressure control
 - Mostly in patients with focal FMD and intimal fibroplasia
- Every patient with FMD should have measurement of serum creatinine and renal artery duplex ultrasound every 12 months.

After revascularization:

- Duplex ultrasonography and serum creatinine measurements performed on the first office visit post procedure, then every six months for two years, and then yearly, if stable.
- With worsening on new hypertension, or unexplained increase in the serum creatinine, he or she should be imaged at that time with duplex ultrasound (or CTA if the ultrasound is equivocal or poor quality).

12.3 Renal Artery Aneurysms

12.3.1 Demographics

- Renal artery an eurysms are rare - Autopsy studies have revealed an incidence of 0.01% to 0.09%.
- Females > males, although females = males with FMD excluded
- Although atherosclerotic changes have been identified in most aneurysms in patients with multiple lesions, this is not a uniform finding, suggesting that arteriosclerosis may not be the most important factor in the genesis of renal artery aneurysms.
- More likely due to a congenital medial degenerative process with weakness of the elastic lamina.
- Fibromuscular dysplasia (FMD) is often a direct contributor to the development of an aneurysm.
 - Medial fibroplasia is typically associated with multiple stenoses and post-stenotic dilatation of the distal two thirds of the renal artery.

- Renal artery aneurysms in association with FMD are generally only a few millimeters in diameter.
- The typical angiographic appearance of a renal artery involved with medial fibroplasia is a "string of beads."
- A rare cause of renal artery aneurysms is Ehlers-Danlos' syndrome.
 - This disorder is associated with extreme arterial fragility and spontaneous rupture.

12.3.2 Anatomy

- Most frequent site of involvement is primary bifurcation, intraparenchymal (<10%)
- Most are saccular
- Right slightly more common than left, bilateral 10%
- 90% are extraparenchymal

12.3.3 Presentation

- Majority are associated with hypertension (70%) (Coleman and Stanley 2015)
- 10% mortality
- 90% risk of kidney loss
- Less than 3% rupture.

12.3.4 Management

Size criteria currently controversial

- Uncontrolled hypertension is an indication for repair when smaller than 2.5cm. (Coleman and Stanley 2015)
- VLFDC recently proposed a 3cm threshold for asymptomatic renal artery aneurysm. (Klausner et al. 2015)
- Many aneurysms with circumferential calcification which could offer protection against rupture, however anecdotally calcification on imaging does not correlate to appearance of intraoperatively

In an elderly patient, observation of this aneurysm with Duplex surveillance is the appropriate treatment.

For larger aneurysms in younger patients, aneurysmorrhaphy with primary repair or patching, interposition grafting, or bypass can be performed with low mortality. (Coleman and Stanley 2015)

• Comparison of ex vivo or insitu renal artery reconstruction have shown no difference in mortality, morbidity, LOS or reoperation.

Endovascular techniques such as coiling have been reported to be successful in treating these saccular aneurysms; however, most aneurysms occur at branch points making covered stent placement difficult. (Coleman and Stanley 2015)

• Renal artery dissection caused by guide wires or catheters can occur, but is rare.

12.4 Renal Artery Dissection

12.4.1 Demographics

12.4.1.1 Presentation

Renal ischemia from renal artery dissection should be considered if a patient presents with acute onset of hypertension, flank pain, hematuria and proteinuria. (B. T. Müller et al. 2003)

Renal artery dissection can sometimes be found incidentally on imaging performed for other reasons.

12.4.2 Evaluation

CTA is the best modality of imaging because it is rapid and accurate at diagnosis.

12.4.3 Management

If the patient has normal renal function, normal blood pressure or stable hypertension, is asymptomatic or the dissection appears chronic, then there likely is no role for intervention.

However, if emergent therapy is required if there is concern for renal perfusion and viability.(Calligaro and Dougherty 2019) There is a role for endovascular stenting, however endovascular therapy should be avoided in the setting branch involvement. Bypass, in situ repair, auto-transplant if renal branch involvement and possibility of renal salvage. Nephrectomy may be required in uncontrolled hypertension, extensive dissection and irreversable ischemia.

12.5 Nutcracker Syndrome

12.5.1 Demographics

12.5.1.1 Etiology

Renal vein compression between the SMA and the aorta, often seen when the SMA angle is <40 deg.(Yun et al. 2016)

12.5.1.2 Presentation

Classically presents with left flank pain and hematuria (microscopic and macroscopic), and sometimes pelvic or vulvar varicosities. Symptoms are often non-specific and other abdominal causes of pain should be evaluated for simultaneously.

12.5.2 Evaluation

Often first identified on axial imaging with a narrowed renal vein and a prominent ovarian vein. Definitive diagnosis often requires venogram and absolute pressures.(K. W. Kim et al. 2011; Yun et al. 2016)

12.5.3 Management

Open surgery has been the mainstay of treatment, which usually requires reconstruction of the renal vein to avoid compression. This can be done with a transposition of the renal vein with or without a vein cuff or transposition of the ovarian vein over the IVC. (Erben et al. 2015; Kurklinsky and Rooke 2010; N. R. Reed et al. 2009)

Endovascular therapy with renal vein stenting has shown some promising results but can be complicated by stent migration and fracture. Antiplatelet therapy is recommended and there has been good patency out to one year. (Erben et al. 2015; Ananthan, Onida, and Davies 2017; Quevedo, Arain, and Abi Rafeh 2014; X. Wang et al. 2012)

12.6 Renal Vein Thrombosis

12.6.1 Evaluation

CT scan is best. Difficult to visualize native renal vein on duplex imaging. (Asghar et al. 2007; Velazquez-ramirez and Corriere 2019)

12.6.2 Management

- Renal vein thrombosis initially managed with heparin, then warfarin for 6mo. [(Asghar et al. 2007; Velazquez-ramirez and Corriere 2019)]
- Thrombectomy or thrombolysis reserved for acutely threatened kidney in young patient, complication of AC or thrombosis of solitary kidney with renal failure.
- Nephrectomy for post-infarct hemorrhage.
- Thrombolysis requires areterial and venous access venous access to debulk and arterial access to drip and clear small intra-paranchymal veins.

12.7 Renal Arteriovenous Fistula

Relatively common complication of renal biopsy (9-18%). (Schwarz et al. 2008)

12.7.1 Presentation

Bruit over kidney, renal impairment, varicocele, hematuria and abdominal pain. (Hunter, Berman, and Walser 2019)

12.7.2 Evaluation

Diagnose with duplex, CTA or MRA. Duplex will show marked turbulence, elevated PSV and high end diastolic flow, low resistive index. (Ozbek et al. 1995)

12.7.3 Management

Indications for treatment (Merkus et al. 2005; Morimoto et al. 1995)

- Gross hematuria requiring blood transfusion
- High output cardiac failure
- Worsening hypertension or renal failure
- Persistence at >1yr

Treated most often with angiogram, covered stent or highly selective micro-coil embolization. (Ginat, Saad, and Turba 2009; Saliou et al. 1998)

13 Trauma - Arterial

Authors: Kevin Kniery, Adham Elmously, Nakia Sarad, and Todd Rasmussen

These trauma episodes were developed in collaboration with Behind the Knife: The Premier Surgery Podcast.

Vascular trauma requires close collaboration between vascular surgeons and trauma surgeons. The decision of which specialty should manage which injuries varies across different centers. Through these trauma chapters we will discuss what management decisions the majority of vascular surgeons should be comfortable managing as a part of a multidisciplinary trauma team.

💡 Take a Listen

Check our debate between leading trauma and vascular surgeons about how to best develop a collaborative team to manage vascular trauma.

For relevant images and a more in depth discussion of this topic, please review Chapter 18: Abdominal Aortic Trauma, Iliac and Visceral Vessel Injuries(Aylwin and Jenkins 2022) in Dr. Rasmussen's 4th Edition of *Rich's Vascular Trauma*.

13.1 General Principles

As in all traumatic injuries, ensure that the patient is appropriately resuscitated with supportive care.

It is important to consider the broad principles and approach:

- 1. Urgency of situation
- 2. Resuscitation Efforts
- 3. Communication with Team
- 4. Optimal Exposure
- 5. Appropriate Equipment

Decision to maintain patency of a vessel vs. ligation is dependent upon the situation. During damage-control procedures where bleeding is not controlled and location of hemorrhage not known, the inclination is to ligate. The risk is organ ischemia, but the benefit would be patient survival.

Angiography can be utilized in conjunction to open operative explorations in areas of injury that are difficult to access/control (i.e. Zone 3 retroperitoneal hematomas)

Take a Look

Here is an image of the zones of the retroperitoneum, hosted by basicmedicalkey.com.

General Guidelines of Penetrating vs. Blunt Retroperitoneal Exploration

- Zone I hematomas: explore whether penetrating or blunt mechanism.
- **Zone II hematomas:** always explore if penetrating, do not explore during blunt unless expanding. Kidneys are in Zone II and are in the lateral gutters.
- **Zone III hematomas:** always explore if penetrating, do not explore during blunt unless expanding.

13.2 Basic Exposure Techniques

Procedural Sequence in Operative Abdominal Trauma

- Expose
- Achieve Proximal Control
- Explore and Evaluate Injury
- Restore Flow

Supraceliac aortic and thoracic cross-clamping are important surgical maneuvers to control massive intra-abdominal bleeding in hemodynamically unstable patients.

Tips and Tricks

- In order to get umbilical tape or vessel loop around aorta, use combination of Metzenbaum scissors and controlled pressure.
 - As you try to free the aorta from the connective tissue around it to get back to the spine, can use a metallic pediatric yankauer (small profile and blunt tip) as a dissection tool and push with the Metzenbaum scissors along with direct feel to get the aorta freed.

- Can then use a right angle to get underneath the aorta.
- Ensure that lumbar vessels are identified when trying to get around the aorta
 - "find the outside before you find the inside"
 - Can clip lumbar vessels with medium/large Weck clips
- Use blue operative towels to cover and bring up the viscera to be held with the large retractors (i.e. Omni retractor) so your hands are free to operate.
- Spine exposures with orthopedic and neurosurgical colleagues and transplant procurements are also a great opportunity to learn the tenets of rotating the viscera either right or left and exposing the retroperitoneum

Take a Listen

Check out our conversation with spine surgeons at HSS in New York City on collaborative work between vascular and spine surgeons to improve the safety of spine surgery.

13.2.1 Supraceliac Control

Supraceliac control is recommended prior to any medial visceral rotations if it is a Zone I hematoma.

Exposure

- Basics
 - Exposure starts with midline laparotomy incision from xiphoid to publis that may be extended to the first 2-3 cm of a sternotomy.
 - This releases the inferior costal margins of the thoracic area under the diaphragm to pull the costal margins laterally and up – which suspends the patient's rib cage up and away from the aorta and the abdominal contents
 - Recommend use of Omni retractor so that retractors can be placed under costal margin
 - Ensure that there is nasogastric tube placed by an esthesia
 - Allows circumferential control of esophagus with either umbilical tape or penrose drain for appropriate left-sided retraction of the esophagus, away from the crus that is surrounding the aorta
- Steps

- Midline laparotomy incision supraxiphoid to pubis with release of costal margin (suspended away from abdominal contents)
- Clear left lateral segment of liver by mobilizing left lateral segment either cephalad or tucked inferiorly to the right side.
- For better exposure, place patient in reverse Trendelenburg to let the stomach down and pull it inferiorly to create a better window
 - Esophagus is retracted to the patient's left and the stomach pulled down (further retraction can be performed by a resident or operative assistant by putting their hand on the stomach and retracting it towards the pelvis)
- Next, divide the crus overlying the aorta, using a large right angle and a bovie extender
 - Maintain adequate control during muscle dissection by feeling the aorta and its pulse (even a weak pulse)
 - Divide the crus over the aorta with electrocautery and with blunt dissection on either side.
 - Note: do not need to spend time to get circumferential control of the aorta like in an elective aneurysm repair.
- Supraceliac aorta should now be exposed enough on either side to have a clamp appropriately placed to gain control

For more on exposure of the visceral aorta, check out Chapter 7

13.2.2 Visceral Rotation

Left medial visceral rotation (Mattox Maneuver)

- Maneuver performed when a rtic injury or a close branch off the a rta is suspected
- Designed to give optimal exposure of the para-visceral segment of the aorta
- The para-visceral segment of the aorta includes the supraceliac segment all the way down to the aortic bifurcation, primarily the left common iliac artery
- Steps
 - Divide the white line of Toldt along the sigmoid colon and the left colon.
 - Establish plane in the retroperitonium
 - Achieved through steady blunt dissection facilitated with Bovie electrocautery

- The dissection plane is taken from the sigmoid/left colon up to divide the attachments ingrained in the spleen and the left kidney, then to the mid-line and then ultimately to the patient's right.
- Left kidney is almost always brought up
 - It is generally easier and faster, because you do not have to take the time to find the plane above the left kidney

Take a Look

Check out this video for an example of a left medial visceral rotation by Dr. Alan Lumsden at Houston Methodist.

Right medial visceral rotation (Cattell-Braasch Maneuver)

- Important when concerned about an IVC injury
- Steps
 - Divide the white line of Toldt starting at the cecum and coming along the right colon.
 - Bring the cecum and right colon cephalad and then to the patient's left.
 - Note: it is actually not so much of a rotation to the midline and then the left, but bringing everything up cephalad and then left to the patient's left costal margin almost bringing the cecum and right colon up onto the patient's left chest.
 - This plane is similar to the left medial visceral rotation as it is largely developed with blunt dissection in combination with Bovie electrocautery to open the retroperitoneum.
 - At this point, you will almost immediately see the left iliac vein and then the vena cava.
 - Continue to bring the left colon up until you see the right renal vein and kidney.
 - At the superior extent, this will expose the head of the pancreas and duodenum.

💡 Take a Look

Check out this video for an example of a right medial visceral rotation by Dr. Alan Lumsden at Houston Methodist.

13.2.3 Exposure of Infrarenal Aorta

The exposure of the infrarenal aorta is also called the transperitoneal inframesocolic exposure

Inframesocolic means that the transverse colon is reflected cephalad.

Steps

- Hold up the transverse colon and drape it up cephalad to patient's costal margin (can be held up with a blue towel and retractor).
- Divide the ligament of Treitz (by the surgeon on the patient's right) and get down to the fourth portion of the duodenum which should be to the left of patient's midline.
- Once divided, the left colon remains on the patient's left and everything else goes to patient's right.
 - Note: this is not a medial visceral rotation because only the fourth portion of the duodenum and all of the small bowel is partitioned to patient's right.
- At this point, one should see the infrarenal aorta. The left renal vein is the most cephalad extent and the aortic bifurcation is the most caudal extent (located at root of colon mesentery).
- The left renal vein can be ligated if necessary to facilitate aortic control
 - Note: try to preserve the branches of the left renal vein that drain the left kidney (gonadal and lumbar) so that dividing the left renal vein is not as much of a detriment to the kidney

13.3 Management

13.3.1 Celiac Artery

Exposure

- Origin of the celiac artery can be seen with left medial visceral rotation.
 - This maneuver will expose the entire paravisceral segment of the aorta
- Can also use anterior approach through the lesser sac (very similar to exposure needed for the supraceliac aortic exposure)
 - Stomach is retracted caudally towards the pelvis, lesser sac is opened, and then the stomach is pulled down further.

• Pancreas is pulled down as well and the origin of the first centimeter of the celiac artery is now exposed.

Repair

• Small and proximal injuries to celiac - Primary repair may be an option use 4-0 or 5-0 prolene sutures.

Ligation

- Large injuries with destruction of vessel wall Primary repair may not be an option
- Considered as damage-control situation to control bleeding and ligation considered as an initial step to save a patient's life
- Ligation may be tolerated if there is robust GDA collateral circulation. However, once bleeding is controlled, this should be assessed with doppler.
- If the patient is stable or improving and particularly if there is clear signs of liver, pancreas or duodenal ischemia, then this should be reconstructed.

13.3.2 Common Hepatic Artery (CHA)

Exposure

- Best visualized through that similar transperitoneal anterior approach for exposure of the celiac artery through the lesser sac.
- Prior to identifying the common hepatic, you have to find the celiac and follow the branches
- Recommend positioning in reverse Trendelenberg, the stomach inferior, and costal margins flared up to have the appropriate space
 - This will expose the second and third portions of the celiac artery, and its major branches including the common hepatic.
- The common hepatic artery can not be visualized using left medial visceral rotation.

Repair

- Primary repair is ideal
- Ligation may be only option in damage control surgery
 - If possible, ligation of the common hepatic proximal to the GDA is preferable as there will be retrograde flow from the GDA into the proper hepatic to maintain the viability of the duodenum, pancreas, and liver.

13.3.3 Gastroduodenal Artery (GDA)

Exposure

The gastroduodenal artery is the first branch off of the common hepatic artery and defines the junction between the common and proper hepatic arteries.

Repair

Primary repair should be sought if possible. If it is a damage control situation, then the GDA must be ligated to control hemorrhage.

Ligation

- Assess patient's condition, resuscitate, utilize the doppler, and assess the area distal to ligation to evaluate the consequences
- If ligation showed appropriate collaterals, then the vessel can remain ligated
- If there is clear ischemia, then there is a difficult decision between repair vs. shunt vs. permanent ligation
- Consider metabolic consequences of ligation as well.
- Decision is situationally dependent the principles of damage control resuscitation and damage control surgery should always be maintained.

13.3.4 Splenic Artery

One of the key tenets of splenic injury is exposure. Ensure good retraction with incision up to xiphoid and alongside the left costal margin.

Low threshold for splenectomy for control of hemorrhage from the splenic artery or splenic parenchyma.

13.3.5 Superior Mesenteric Artery (SMA)

It is important to remember that a majority of these axial injuries are typically unsurvivable and proximal SMA repair is uncommon in the trauma setting. Even if the SMA is repaired and the bowel appears viable, it is preferred to delay abdominal closure and perform a second look 24-36 hours later or sooner depending on the patient's condition.

SMA is divided into 3 segments based on its course

- Segment 1: Aorta to Pancreas
- Segment 2: Pancreas to Root of Mesentery
• Segment 3: Mesentery to Small Bowel

Exposure

- Visualization is dependent on the segment that needs to be exposed
- Segment 1 (most proximal) left medial visceral rotation.
- Segment 2 (near head of pancreas)
 - Head of pancreas may need to be divided to get appropriate exposure
 - If distal to head of pancreas, can use inframesocolic approach
- Segment 3 (most distal) anterior transperitoneal approach

Repair

- All efforts should be made to maintain flow through the SMA
- Primary repair performed on small injuries with 4-0 or 5-0 prolene suturing
- Patchy angioplasty of interposition graft
 - For larger injuries
 - Conduit types: PTFE, Dacron, or autologous tissue such as proximal greater saphenous vein or deep femoral vein
 - Autologous vein is preferred in contaminated field, but can also use rifampin-soaked Dacron.
 - Tunneling is dependent on location, proximal segments would need 6-8 mm of graft while distal segments would need 3-4 mm.
- Again, repair in a strictly traumatic setting is rare, because injuries that include the proximal mesenteric arteries are unlikely to be survivable. However, these techniques can be applied to other situations when assisting other surgeons who have to resect the SMA as a part of an oncologic resection or have induced an iatrogenic injury.

Take a Listen

Check our episode where we discuss the role of a vascular surgeon in assisting with oncologic resections.

Ligation

• Distal SMA segments may be more tolerant of ligation.

13.3.6 Inferior Mesenteric Artery (IMA)

IMA can be ligated in majority of cases.

Prior to ligation, assess the patency of the iliac vessels — specifically the internal iliac vessels.

If both celiac and SMA are patent, then the IMA ligation can be tolerated.

If there is concern for atherosclerotic disease or occlusion in the internal iliacs, then ligation may not be the preferred option and re-implantation may be considered.

For more discussion of evaluation and re-implantation of the IMA, see Section 10.3.2

13.3.7 Renal Arteries

Renal artery is unlikely to be reconstructed in a trauma setting (similar to SMA)

Most commonly, if there is an expanding hematoma in Zone II and the renal artery/parenchyma is damaged and confirmed as source of hemorrhage, a nephrectomy is warranted and there is no indication for repair.

In renal trauma, the renal artery is either not injured (with only injury to renal parenchyma) or renal artery is injured and thrombosed.

• If renal artery thrombosed, salvageability is unlikely due to prolonged warm ischemic time of the kidney.

Repair

- Occurs if source of hemorrhage is from an injured renal artery that still has flow
 - Seen during exploration of expanding hematoma in Zone II
- Most commonly, if renal artery is injured it is ligated for damage-control
- If patient hemodynamically stable, primary repair recommended
 - 4-0, 5-0, or even 6-0 monofilament prolene suture
 - If a primary repair would compromise the lumen, then consider patch angioplasty
- Distal branch artery injuries can be managed with endovascular coil embolization.
 - Again, the decision to take the time to coil embolize is dependent on the patient's current hemodynamics

Incidental CT findings

• Types: intimal flap, fistula, and/or possible pseudoaneurysm of the renal artery

• Management

- The injury can be observed if there is continued renal perfusion and no flow limitation.
- If it is a flow limiting injury with more than 50% stenosis caused by the intimal flap, it may be amenable to endovascular placement of bare metal stent to tack down the flap.

14 Trauma - Venous

Authors: Kevin Kniery, Adham Elmously, Nakia Sarad, and Todd Rasmussen

These trauma episodes were developed in collaboration with Behind the Knife: The Premier Surgery Podcast.

For relevant images and a more in depth discussion of this topic, please review Chapter 19: Inferior Vena Cava, Portal and Mesenteric Venous System in Dr. Rasmussen's 4th Edition of *Rich's Vascular Trauma*.(T. Fabian and Savage 2021)

14.1 General Principles

14.1.1 Anatomy

For the discussion today we will divide the IVC into four segments:

- Infrarenal
- Renal/suprarenal
- Retrohepatic
- Suprahepatic

We will then discuss management of portal vein and SMV separately.

14.1.2 Non-operative management

- Generally well-tolerated due to low pressure system and difficulty of exposure increasing risk of more bleeding
 - Especially for suprarenal and retrohepatic segments
- Option for blunt mechanism of injuries
- If patient is hemodynamically stable and there are no other associated injuries, can consider non-operative management with serial imaging (CT with contrast)

14.1.3 Hemorrhage Control

- Pre-operative planning with appropriate resuscitation and blood products available
- Keep in mind coordination with other surgical teams and anesthesia
- Patients with abdominal venous injuries likely to be polytrauma patients, approach to assess and repair injuries should be time efficient
- IVC injuries considered Zone I retroperitoneal hematomas
- Follow general principles of gaining proximal and distal control
 - Can use low profile sponge sticks (Kittner sponge stick) or manual pressure
 - Be prepared to refine where you control as the dissection progresses
- Immediate control of hemorrhage may require ligation. However, once the field is cleared, coordinate with anesthesia to confirm they are making progress with resuscitation and then evaluate whether the patient can tolerate reconstruction at that time or whether you need to continue to follow damage control principles.
- Iliac vein injury often able to be accessed by ligating the internal iliac artery and mobilizing the common and external iliac artery. Transection of iliacs should only be used as a last resort if this fails. (J. T. Lee and Bongard 2002)

14.1.4 Primary repair

- Option for partial tears without risk of narrowing lumen
 - Narrowing lumen increases risk of thrombosis and thromboembolism
- Recommend figure of 8 of 16 to repair with 4-0 prolene suture on SH (small half circle) needle
 - Larger needles are easier to visualize in cases with high amounts of bleeding
 - When first suture is placed, recommend lifting on either side of suture with both strands to elevate the defect and maintain tension » improves visualization and placement of subsequent sutures to achieve full thickness bites

14.1.5 Ligation

- Well-tolerated really only for infrarenal segment IVC
- Not a viable option renal/suprarenal, retrohepatic, and suprahepatic IVC due to compromised outflow of vital organ structures (i.e. kidneys and liver)
- Interposition grafts are not readily performed due to inefficiency of time.

14.2 Infrarenal IVC

14.2.1 Anatomy

IVC from inferior portions of renal veins to iliac bifurcation

14.2.2 Non-operative Management

This is a low pressure system and therefore non-operative management can be observed in patients with blunt injuries and hemodynamically stable. However if there is penetrating trauma, or a laparotomy is performed for other reasons, the infrarenal IVC should be explored and repaired.

14.2.3 Exposure

- Right medial visceral rotation (Cattell-Braasch maneuver)
- Exposes in the proximal portion the IVC, inferior renal veins, 4th portion of duodenum, and pancreatic head
- Be aware of lumbar vessels posteriorly and control with vessel loop or clamp

14.2.4 Hemorrhage Control

• Proximal and distal control may not be sufficient as there can be significant bleeding from the many lumbar veins that feed into this area of the vena cava.

14.2.5 Primary Repair

- SH needle on 4-0 prolene
- Avoid luminal narrowing of IVC to prevent thrombosis/thromboembolism

14.2.6 Ligation

- Ligation can be tolerated and is appropriate when primary repair narrows lumen or if primary repair cannot be performed due to damage-control nature of operation
- Empirical 4-compartment fasciotomies of lower extremities is dependent on patient's resuscitation status
- Recommend fasciotomy if patient received large volumes of fluids/products, continues to be hemodynamically tenuous, anticipate long recovery

14.3 Renal/Suprarenal IVC

14.3.1 Anatomy

- 2-3 cm of IVC, from renal vein to inferior portion of hepatic veins
- Includes renal veins and gonadal veins

14.3.2 Non-operative Management

- Preferred due to location of injury » similar to retrohepatic and suprahepatic segments
- Exposure of segment may increase risk of bleeding, increase size of defect, or unroof hemostasis formed

14.3.3 Exposure

- Right medial visceral rotation (Cattell-Braasch) and mobilizing liver cephalad
- Liver mobilization may be required to adequately expose this portion of the IVC as it is below the hepatic veins but posterior to the liver parenchyma.
 - Divide the attachments of the liver to the diaphragm to be able to move liver cephalad
 - Caution as there are posterior lumbar veins and other branches that can cause more significant bleeding

14.3.4 Primary Repair

- Option if liver mobilization does not increase size of defect/tear
- Follow principles as described above

14.3.5 Ligation

• Not as well tolerated as infrarenal segment due to compromised of outflow to renal veins

14.4 Retrohepatic IVC

14.4.1 Anatomy

This region is very difficult to access and includes the IVC at the confluence of the hepatic veins.

14.4.2 Management

Non-operative management is preferred due to the location of injury. The liver provides hemostasis due to overlying pressure and if liver mobilized, can cause more significant bleeding.

14.4.3 Exposure

• Right medial visceral rotation (Cattell-Braasch) and mobilizing liver cephalad. Mobilizing the liver completely off the diaphragm can allow for better visualization.

14.4.4 Hemorrhage Control

- As in the more inferior segments of the IVC, it is important to control venous inflow from the more inferior IVC and lumbar veins.
- Direct control of the hepatic veins is difficult, so ongoing bleeding may may require Total Hepatic Isolation to locate bleeding behind the liver
 - Total Hepatic Isolation is accomplished with vascular control of the porta hepatis with a Rommel tourniquet, also known as the Pringle maneuver.
 - Portal Vein Accounts for 70-80% of hepatic inflow

- Hepatic Artery
- Finally, you need to control the suprahepatic segment of the IVC under the diaphragm to prevent back bleeding.
- Other options for bleeding control if total hepatic isolation not successful
 - Supraceliac aortic clamp
 - Can be considered if bleeding continues to help identify injury
 - Last option, usually unsurvivable
- If vascular control is required for an extended period, then an Atriocaval Shunt (AKA Schrock shunt) may be required to maintain venous return to the right atrium and prevent cardiovascular collapse.(Schrock 1968)
 - Can be accomplished with large chest tube with extra hole cut near base
 - Chest tube clamped at base and inserted via purse string suture at the right atrial appendage > advanced to infrarenal IVC
 - Place Rommel tourniquet below area of suprarenal IVC (above the renal veins and last hole in tube)
 - Second Rommel tourniquet placed above hepatic veins in intrapericardial IVC
 - Click here for an image

14.4.5 Primary Repair

- Careful liver mobilization due to multiple branch points and high risk of tear > will likely require total hepatic isolation for bleeding control
- Be careful of timing and patient status. May require packing and resuscitation or atriocaval shunt during repair.

14.4.6 Ligation

Not as well tolerated due to compromised outflow of both renal and hepatic veins, can cause both kidney and liver ischemia

14.5 Suprahepatic IVC

14.5.1 Anatomy

Includes phrenic veins and pericardium to right atrium

14.5.2 Non-operative Management

- Preferred due to location of injury, difficult to access
- Liver provides hemostasis due to overlying pressure
- If liver mobilized, can cause more significant bleeding

14.5.3 Exposure

Divide xiphoid to open costal margins (approximately 2-3 cm cephalad). Wiley vein retractor may be used to fully mobilize the diaphragm and access the IVC in the chest through the diaphragm from the abdomen.

14.5.4 Hemorrhage Control

Likely will require total hepatic isolation and atriocaval shunt to fully control bleeding and maintain venous return to the heart.

14.5.5 Primary Repair

Can consider supraceliac aortic clamp and/or atriocaval shunt if extensive repair is required.

14.5.6 Ligation

Not as well tolerated due to compromised outflow of both renal and hepatic veins, can cause both kidney and liver ischemia.

14.6 Portal Vein and Superior Mesenteric Vein Injuries

Injuries to portal vein and SMV can be very catastrophic. Non-isolated injury — usually associated with other injuries (ie pancreatic + celiac vessels).

14.6.1 Exposure

- Pringle maneuver to gain control of porta hepatis
- Division of pancreatic head for better visualization

14.6.2 Management

- Resuscitation is key
 - 50% of blood volume is sequestered by splanchnic circulation
 - Coordination with surgical team and anesthesia to resuscitate appropriately
 - Consider insensible losses with open abdomen during OR time
- Repair these injuries if possible
 - Primary repair
 - Interposition graft conduits
 - Great saphenous vein
 - PTFE
 - Bovine pericardium
 - Ligation an option in damage-control (high mortality)
 - Temporizing measure to prepare for reconstruction once bleeding is controlled

Part V

Lower Extremity

15 Medical Management and Claudication

Authors: Nedal Katib and Danielle Bajakian

Contributors: Eilidh Gunn

15.1 Pathogenesis of peripheral arterial disease

What is Peripheral Artery Occlusive Disease (PAD)?

Peripheral Arterial Disease encompasses extremity arterial disease but generally is used to describe lower limb arterial occlusive disease. Peripheral Artery Occlusive Disease (PAD) is a more specific term and this encompasses atherosclerotic disease of the lower limb arteries.

The disease has a step-wise spectrum of presentations from asymptomatic disease to intermittent claudication and finally chronic limb threatening ischemia (CLTI) formerly known as Critical Limb Ischemia (CLI) (Aboyans et al. 2018)

What is the underlying pathophysiology of PAD?

The understanding of the underlying pathophysiology which results in occlusive arterial disease of the lower limb has evolved over the last 50 years. Atherosclerosis remains the main pathological process resulting in occlusive disease with smoking being the primary modifiable risk factor.

Although with the rising incidence of diabetes a second population of patients with a significant PAD burden has developed.

Atherosclerosis begins with an injury to the intimal lining of the arterial wall, which, for example, can result from smoking, hypertension, hyperglycemia, or advanced age. Repeated injury leads to a chronic inflammatory process resulting in the uptake and oxidation of LDL, resulting in foam cells.(Hiltunen et al. 1998; Tsimikas et al. 2005; Witztum 1994) Foam cells result when unregulated scavenger receptors promote uptake of oxidized LDL by macrophages and smooth muscle cells. These foam cells become necrotic and then create the central core of the atheroma.(Owens 2019) This oxidized LDL form into intimal plaque, build up and calcification that may result in progressive vessel stenosis and subsequent occlusion, or plaque rupture with acute occlusion.

Atherosclerosis is particularly prone to developing at sites of arterial bifurcation (i.e. carotid bifurcation, femoral bifurcation) because turbulent flow and low laminar shear stress results in the down regulation of eNOS and NO production. This in turn increases V-CAM-1 and reduces inhibition of KF-kB (resulting in it's elevation.(Owens 2019)

💡 Take a Listen

Check out our episode with Dr. Melina Kibbe, who is actively researching the role of nitric oxide in vascular pathology.

Table 15.1: **AHA lesion classification** - defines plaque risk for rupture/thrombosis. Extrapolated from the coronary artery disease literature.(McGill et al. 2000)

Classification	Description
0	Normal artery
1	Initial atherosclerotic lesion
2	Fatty streak
3	Fatty plaque/raised fatty streak
4	Atheroma
5	Fibroatheroma - development of fibrotic cap, vascularization or calcium
6	Complicated lesion

Since the publication of WIII in 2014 a lot has changed in the way we view PAD leading up to last years new Global Vascular Guidelines on CLTI, which as a term has replaced CLI (see Chapter 16). (Mills et al. 2014a; Conte et al. 2019)

15.1.1 Risk Factors

What are the risk factors for atherosclerosis?

Modifiable:

- 1. Smoking
 - The most significant modifiable risk factor for developing peripheral arterial disease.
 - Causes endothelial dysfunction by reducing nitric oxide and triggering reactiveoxygen species, platelet adhesion, and permeability of endothelial surface to fibrinogen.(United States Surgeon General 2014; Hackam and Anand 2003).
 - Causes a prothrombotic environment by causing an increase in thromboxane A2 and decreasing prostacyclin thus overall resulting in increased prothrombotic environment for platelets.

• Smoking has a stronger association with Intermittent claudication than with Coronary Artery Disease(Gordon and Kannel 1972)

2. Diabetes, Metabolic Syndrome, and Insulin Resistance

- Diabetes Mellitus, after smoking, is the most significant modifiable risk factor for developing peripheral arterial disease. Both insulin resistance and hyperinsulinemia are independent risk factors for developing peripheral arterial disease. Increasing Incidence of Diabetes world-wide.(Boulton et al. 2005)
 - 2.8% of the global population affected by diabetes in 2000, estimated to be 4.4% by 2030.
 - 25% of patients with diabetes develop a diabetic foot ulcer (DFU), at some stage in their lives.
 - Diabetic foot ulcers have a very high rate of recurrence with 40% of patients developing a new ulcer within 12 months of healing a previous ulcer. (D. G. Armstrong, Boulton, and Bus 2017)
 - Limb Loss occurs every 20 seconds world-wide to due to Diabetes related PAD or infection.
- The Odds Risk for developing PAD in patients with DM ranges from 1.89 to 4.05.
- Factors associated with developing PAD in diabetics include female gender, chronicity/severity of DM, insulin dependence, African American race or Hispanic ethnicity.(Regensteiner et al. 2015; Selvin et al. 2006; Wattanakit et al. 2005) An increase in HbA1C by 1% correlates with a 28% increase risk of developing PAD. (Amanda I. Adler et al. 2002)
- The following mechanisms may lead to PAD in diabetics
 - Hyperglycemia reduces bioavailability of NO, which is a vasodilator, enhances endothelial function, and inhibits both platelet activation and smooth muscle migration.(Higashi et al. 2009)
 - Hyperglycemia activates protein kinase C and thus accumulation of mitochondrial ROS and prostanoids.(Geraldes and King 2010)
 - Endothelial dysfunction leads to increased endothelin-1 and vasoconstriction.(Ahlborg et al. 2007)
- Given the pathological effect of diabetes upon multiple structures and systems the normal progressive history of diabetes related PAD is somewhat different to smoking related PAD. Clinically due to the combined effect of peripheral neuropathy and microvascular arterial disease, foot ulceration for many patients tends to be the initial presentation. This results in a lack of a 'safety net' where presenting with

progressive claudication allows for a period of detection, management and and risk factor modification before they develop tissue loss and are at risk of amputation, leading to poor outcomes and greater morbidity and mortality. (D. L. Armstrong et al. 2012)

- Neuropathy. The loss of the basic nociceptive mechanisms in the foot among diabetics, presents as a loss of protective sensation (LOPS). Neuropathy can be divided into three types:
 - Sensory: "stocking-glove" distribution
 - Motor: Intrinsic muscle wasting resulting in deformities
 - Autonomic: Sympathetic nervous system pathology
- Structural deformities and gait disturbances
- Arterial disease of large and small vessels.
 - Large Vessel
 - Small Vessel
 - Both
- Foot infection(Dennis F. Bandyk 2018; KAYSSI, ROGERS, and NEVILLE 2019; Lepäntalo et al. 2011)

3. Hypertension

- The most common cardiovascular risk factor worldwide.
 - USPSTF Grade A recommendation for HTN screening confirm readings outside clinical setting or at home before initiating treatment.(Piper et al. 2014)
- The Incidence of PAD increases to 2.5-fold in patients with Hypertension. (Kannel and McGee 1985)
- Each 20mg increase in BP, results in a 35-63% increase risk in PAD. Recommend keeping BP <140/90 or <130/80 in diabetics or CKD.(A. I. Adler et al. 2000; Bavry et al. 2010; Emdin et al. 2015; Itoga et al. 2018)

4. Dyslipidemia

- A strong association has long been identified as a risk factor for cardiovascular disease.(Langlois and Nordestgaard 2018; Diehm et al. 2004)
- 25% cardiovascular event reduction for each 39 mg/dL (1mmol/L) reduction in LDL. (H. P. S. C. Group 2007)

- JUPITER Study Randomized rosuvastatin 20mg to placebo in healthy patients. Found significant decrease in unstable angina, myocardial infarction and stroke. However, patients had a higher incidence of physician-reported diabetes, but no difference in myopathy.(Ridker et al. 2008)
- Fish oil is also indicated for secondary prevention among patients with known cardiovascular disease.(Grenon et al. 2012; Kris-Etherton et al. 2002; Siscovick et al. 2017)

Non-Modifiable:

1. Age

- Age is identified as a risk factor for PAD regardless of gender.
- Prevalence of PAD increases with age: 15% > 70 years of age.

2. Gender

- The Framingham Study has found that the risk of developing PAD is doubled in men.
- 3. Race and Ethnicity
 - The MultiEnthic Study on Atherosclerosis (MESA) study showed a higher prevalence of PAD (ABPI <0.9) in African Americans compared to Whites. 7.2% versus 3.6%. (Bild 2002)
 - Cross Sectional analysis 6653 subjects all with ABPI assessment revealed a prevalence of PAD (<0.9) of 4%. Non-Hispanic Whites: 3.6%, Asian: 2%, African American: 7.2% and Hispanic: 2.4%. (p<0.01) (Allison et al. 2006)
- 4. Family History
 - Family history of atherosclerotic disease increases an individuals risk of atherosclerotic disease themselves.

15.1.2 Landmark Studies

What are the some of the major population-based trials looking at the natural history of atherosclerotic disease?

1. **The Framingham Heart Study**: The original Cohort from the town of Framingham, n=5183 patients, followed over time for over 30 years. There have been multiple subsequent recruited populations since. The majority of information we have about risk factors related to cardiovascular health comes from this study. (Mahmood et al. 2014)

- The Rotterdam Study: 1990, Longitudinal Study, >7000 participants.(Ikram et al. 2017)
- 3. CVHS: 1989-1999 Longitudinal Study : n>5000 Multicenter Study.
- 4. **MESA**: Cross Sectional analysis 6653 subjects all with ABPI assessment revealed a prevalence of PAD (<0.9) of 4%. Non-Hispanic Whites: 3.6%, Asian: 2%, African American: 7.2% and Hispanic: 2.4%. (p<0.01) (Allison et al. 2006)
- 5. **The Edinburgh Study**: The EAS began as a cross sectional study of 1592 men and women in Edinburgh with the goal of examining the frequency of risk factors for peripheral arterial disease. The subjects were followed over 20 years. (F. G. R. Fowkes et al. 1991)

15.2 Intermittent Claudication

What is Intermittent Claudication and the classic patient presentation?

The original population studies we mentioned determined the epidemiology and natural history of Intermittent Claudication based on historically validated and widely accepted questionnaires, namely the **Rose** (Rose 1962) (which later was adopted by the WHO) and subsequently the **Edinburgh questionnaire**. (Lend and Fowkes 1992).

All questionnaires are based on a number of key diagnostic clinical factors that define claudication, they are:

- Onset
- Calf Involvement
- Reproducibility
- Relief with Rest
- Not Occurring at Rest

The progression of intermittent claudication was historically graded by Fontaine (1954) (Fontaine, Kim, and Kieny 1954) followed by the Rutherford Grading System (1986, Revised 1997) (Rutherford et al. 1997)

Rutherford et al. Ad Hoc Committee on Reporting Standards, SVS/North American Chapter ISCVS:

Grade	Category	Clinical Description
0	0	A symptomatic - no hemodynamic significant occlusive disease
Ι	1	Mild Claudication

Grade	Category	Clinical Description
Ι	2	Moderate Claudication
Ι	3	Severe Claudication
II	4	Ischemic Rest Pain
III	5	Minor Tissue Loss
III	6	Major Tissue Loss

Leriche Syndrome - Is a clinical triad of: claudication (buttock, thigh, and calf), impotence, and decreased lower limb pulses, which signifies aortoiliac occlusive disease. (Frederick, Newman, and Kohlwes 2010; Leriche and Morel 1948; Setacci et al. 2012)

Neurogenic Claudication - Differentiating neurogenic claudication from vasculogenic claudication can be difficult. Vasculogenic claudication is relieved by cessation of ambulation and mostly affects a unilateral calf. Neurogenic improves with postural changes, more often bilateral and affects the thighs. (Nadeau et al. 2013)

15.2.1 Evaluation

What is involved in the work up of patients with PAD/Intermittent Claudication?

SVS Guidelines:

"We recommend using ABPI as the first-line non-invasive test to establish a diagnosis of PAD in individuals with symptoms or signs suggestive of disease. When the ABI is borderline or normal (>0.9) and symptoms of claudication are suggestive, we recommend an exercise ABPI."(Conte, Pomposelli, Clair, Geraghty, McKinsey, Mills, Moneta, Murad, Powell, Reed, and al. 2015)

Grade 1 Level of Evidence A

- ABPI
- Exercise ABPI
- Ultrasound

What is an ABPI and how is it measured?

The AHA released guidelines on how to perform an ABPI in an attempt to standardize the method to allow for more comparable results from studies:

Divide the higher of the PT or DP pressure by the higher of the right or left Brachial SBP (Class 1 Level of Evidence A) (Aboyans et al. 2012)

Sensitivity and Specificity both >95% (when ABPI cut off </=0.9 – in detecting >/= 50% stenosis) (Yao, Hobbs, and Irivne 2005; K. Ouriel et al. 1982)

ABPI	Interpretation
>1.4	Non-compressible
>0.9-1.39	Normal
0.5 - 0.9	Mild to Mod PAD
0.0 - 0.5	Severe PAD

Table 15.3: Interpretation of ABPIs.

Patients with diabetes may have non-compressible tibial vessels and therefore toe pressures can be particularly useful in this population.

Segmental pulse volume recordings (PVR) can provide information about the quality of perfusion at the level of the thigh or calf helping to localize the likely level of occlusive disease.

What are Exercise ABPI studies?

Constant Load Testing – (unlike the Graded Test – Bruce Protocol used in testing for coronary artery disease).

The physiological changes in ABPI related to exercise are due to the fact that exercise increases SBP and decreases SVR and, therefore, will affect flow through a stenosis making it more physiologically relevant during exercise and result in a decreased pressure in the lower extremity. (Nicolaï et al. 2009; Alqahtani et al. 2018)

Walking distance has been shown to correlate with level and severity of POAD. (Strandness and Sumner 1975)

Furthermore patients with single level iliac disease may present with buttock/thigh claudication with preserved pulses. Exercise treadmill ABPI is particularly useful in determining the severity of disease in these patients. (Aboyans et al. 2010) For more on exercise testing see Section 20.5

Stenosis		Velocity	Distal Artery Spectral
Category	Peak Systolic Velocity	Ratio	Waveform
Normal	<150	<1.5	Triphasic, Normal PSV
30-49%	150-200	1.5-2	Triphasic, Normal PSV
50-75%	200-400	2-4	Monophasic, reduced PSV
>75%	>400	>4	Damped, monophasic,
			reduced PSV

What is the ultrasound duplex criteria for defining PAD?

Stenosis	Peak Systolic Velocity	Velocity	Distal Artery Spectral
Category		Ratio	Waveform
Occlusion	No Flow – B -mode, Terminal Thump	NA	NA

Adapted from Stone and Hass. Vascular Laboratory: Arterial Duplex Scanning. Rutherford's Vascular Surgery and Endovascular Therapy. 2019.(Patrick A. Stone and Hass 2019)

What Guidelines are there pertaining to PAD Management.

SVS Guidelines (2015)

The SVS published the SVS practice guidelines for atherosclerotic occlusive disease of the lower extremities: Management of asymptomatic disease and claudication. *Conte and Pomposelli et al. JVS 2015* (Conte, Pomposelli, Clair, Geraghty, McKinsey, Mills, Moneta, Murad, Powell, Reed, Schanzer, et al. 2015)

Additional Guidelines

- TASC 1 and 2(Norgren et al. 2007)
- European Society of Vascular Surgery(Aboyans et al. 2018)
- AHA/ACC Guidelines(Gerhard-Herman et al. 2017) multidisciplinary, involving vascular surgeons

Take a Listen

Check out our episode with Dr. Leila Mureebe, where we briefly discuss her role in the multidisciplinary development of the AHA/ACC guidelines.

15.2.2 Management

15.2.2.1 Medical Management

What is the initial management of asymptomatic patients with PAD?

- 1. Smoking Cessation Multidisciplinary comprehensive smoking cessation interventions repeatedly until tobacco use has stopped (Grade A 1) First line therapy is behavioral therapy + varenicline/bupropion or nicotine replacement therapy (NRT) per EAGLES trial.(Anthenelli et al. 2016; Vogeler, McClain, and Evoy 2016)
 - Preloading with varenicline (4 weeks) or bupropion (5-7d) prior to quitting has demonstrated higher abstinence at 12w.(Ratchford and Black 2011)

- Varenicline is a partial nicotine receptor agonist.(Cahill et al. 2016)
- Bu propion is a selective dopamine and norepinephrine uptake inhibitor. (J. R. Hughes et al. 2014)
- NRT is preferred if rapid onset of action is needed.(Thomsen, Villebro, and Møller 2014) However, contraindicated in setting of recent MI, severe angina, or life threatening arrhythmias.
- 2. Intervention is not only not recommended, but invasive treatment is recommended against, in the absence of symptoms (Grade A 1)

How can we medically manage $\underline{asymptomatic PAD}$ based on the SVS and AHA Guidelines?

- Anti-Platelet Therapy
 - The Aspirin for Asymptomatic Atherosclerosis Trial n=3350, aspirin versus placebo. 8 years follow up no difference in events (F. Gerald R. Fowkes 2010) therefore benefit unknown
- Statin Therapy
 - The Heart Protection Study (H. P. S. C. Group 2007) this study looked at statins in patients with PAD but not completely asymptomatic, they had other risk factors such as diabetes, IHD, cerebrovascular disease, or hypertension. Without these risk factors the benefit of Statin therapy remains unclear.
 - However, the AHA from the Framingham Study does recommend using statins if 10-year risk based on risk calculators >7.5% (which would be positive if PAD present).
 - Statins are particularly useful for risk reduction of cardiovascular events in diabetics.(Collaborators et al. 2008; Colhoun et al. 2004; Stamler et al. 1993)
 - Statin-associated muscle symptoms (SAMS) is the most common side effect and occurrence has been reported as high as 30% in some cohorts.(Keen et al. 2014) Other side effects are rare (1.5-5% of patients) and include DM2 new onset, neurocognitive effects, hepatotoxicity, renal toxicity and pancreatitis.(Bitzur et al. 2013; Saxon and Eckel 2016; N. C. Ward, Watts, and Eckel 2019)

• Exercise and Limb Function

- No clear evidence that physical therapy improves the patient outcomes or quality of life.
- Surveillance
 - No benefit from US surveillance, unclear benefit of ABPI surveillance.

How can we medically manage Intermittent Claudication based on the SVS Guidelines?

- Smoking Cessation: As in with asymptomatic disease above, multidisciplinary comprehensive smoking cessation interventions repeatedly until tobacco use has stopped (Grade A 1)
- **Dyslipidaemia:** Statin Therapy Recommended most recent evidence on lipid therapy has suggested focusing on reducing 10-year cardiovascular event risk rather than specifically reducing lipid levels. (Grade 1-A)
- Anti-Platelet Therapy: Aspirin therapy (75-325mg daily) is recommended to reduce cardiovascular events in patients with PAD (Grade 1-Level A) (Antithrombotic Trialists'Collaboration 2002; Antithrombotic Trialists'(ATT) Collaboration et al. 2009)
 - Literature reviews find benefits in secondary prevention with antiplatelet monotherapy (aspirin or clopidogrel) in symptomatic PAD.(Banerjee et al. 2015) Some initial studies did demonstrate benefit from vorapaxar as well.(Rooke et al. 2013)
 - There is evidence that Clopidogrel 75mg compared to Aspirin is better in event reduction. (CAPRIE)(Committee 1996a)– replacing Aspirin with Clopidogrel (Grade 1-Level B)
 - Recent evidence from the COMPASS trial has suggested that the addition of low dose rivaroxaban (2.5mg BID) to aspirin (100mg daily) has lower MALE, composite endpoint of cardiovascular death, stroke and MI. Subgroup analysis suggested clopidogrel may be better with reduced risk of GI bleed and MACE.(Anand et al. 2018; Eikelboom et al. 2017)
- Diabetes Mellitus: Optimization of HbA1C < 7% (Grade 1-Level B)(Lajoie 2019)
- Hypertension: Indicated B-Blockers for hypertension. (Grade 1-Level B)
 - There's no evidence that beta blockers worsen the symptoms of intermittent claudication, but the patients should be monitored for episodes of hypotension
 - Initiation of beta-blockers in the immediate preoperative period is associated with worse outcomes.(Blessberger et al. 2018; P. S. Group et al. 2008)
 - ACE Inhibitors are first line in patients with heart failure, diabetes or kidney disease but should be avoided in patients with renal artery stenosis (RAS).(Barrons and Woods 2016)
- Homocysteine: While higher homocysteine levels are associated with PAD and risk of venous and arterial thrombosis. (D'Angelo and Selhub 1997; Khandanpour et al. 2009; Liebman 2019) Recommendation against Folic Acid and Vit B12(Grade 2 C)

• Patients are at higher risk of depression (almost 20%), therefore PHQ-9 and -2 are good screening tools and should be use in these patients.(Jha et al. 2019; McDermott et al. 2016; J. L. Ramirez, Drudi, and Grenon 2018)

To improve Limb Function in patients with Claudication:

- Cilostozol use Claudication without CHF 3-month Trial (Grade 2 A)
 - If unable to tolerate Cilostozol Pentoxifylline (400mg TDS) (Grade 2 B)
 - Based on Meta-analysis 26 trials (Stevens et al. 2012)

• Supervised Exercise Therapy

- First Line Therapy recommended SEP: minimum three times per week (30-60 min/session) for at least 12 weeks (*Grade 1 Level A*)
- More durable functional improvements than angioplasty alone, however no current data demonstrating impact on need for revascularization or amputation rate.(F. Fakhry et al. 2013; Farzin Fakhry et al. 2015; T. P. Murphy et al. 2012, 2015)
- Meta-analysis of 32 RCTs: Placebo versus exercise: Walking Time, Walking Ability, Pain Free Walking and maximum walking distance improves. BUT no difference in ABPI, Mortality or amputation. (Lane et al. 2017)
- Meta-analysis of 14 RCTs: Supervised better than Non-Supervised Programs. (Hageman et al. 2018)

15.2.2.2 Open and Endovascular Management

What is the surgical management for patients with Intermittent Claudication?

- With appropriate medical management and risk modification, 25% of patients will improve, 50% will remain stable and 25% will progress and requiring intervention. (Conte, Pomposelli, Clair, Geraghty, McKinsey, Mills, Moneta, Murad, Powell, Reed, and al. 2015; Norgren et al. 2007)
- Patient Selection for Intervention:
 - As above, 20-30% of patients with IC who adhere to risk factor modification will have progressive symptoms that will eventually be treated with intervention.
 - Patient selection should be based on QoL and functional impairment in an active person (loss of ability to perform occupation or that limits basic activities of daily living) rather than hemodynamic (ABPI or US) or anatomical disease progression/severity.

- Always remember multifactorial causes of immobility particularly in the elderly.
 - SVS recommends that invasive therapy for IC have a >50% likelihood of sustained clinical improvement for at least 2 years.
- Anatomical Selection:
 - *Aortoiliac Disease:* Previous TASC Classification has attempted to categorize anatomy of disease and subsequent recommendation of Endovascular versus open surgery. But as the authors of the SVS guidelines highlight, "improvements in technology and endovascular techniques have resulted in EVT replacing open surgical bypass as a primary treatment for both focal and advanced AIOD in many cases." The majority of evidence is non randomized and meta analyses of non-randomized series.
 - Endovascular procedures is preferred as primary thearpy over open surgery for focal aortoiliac, common Iliac and external iliac disease causing Claudication. (Grade 1 Evidence B)
 - Hybrid approaches are recommended for Iliac disease involving CFA with the addition of an open femoral endarterectomy. (Grade 1 Level B)
 - Direct Surgical reconstruction (bypass, endarterectomy) in patients with reasonable surgical risk and diffuse aortoiliac occlusive disease not amenable to endovascular approach, after one or more failed attempts at EVT, or combined occlusive and aneurysmal disease. (Grade 1 Evidence B)
 - Bilateral external iliac occlusion may be best treated with end to side aortobifemoral bypass to allow for continued perfusion of the pelvis. (Jaquinandi et al. 2008; Akker et al. 1992; Brewster and Darling 1978)
 - *Infrainguinal Disease:* Reviewing historical data comparing all EVT compared with surgical bypass EVTs are less durable, especially when there's diffuse or long segments of occlusion/multilevel infrainguinal disease. Most recommendations are based on low level evidence when comparing EVT versus Open Surgery
 - Endovascular therapy has become first line therapy for focal SFA disease not involving the vessel origin. (Grade 1 Level C)
 - Endovascular therapy with self-expanding stent (with or without paclitaxel) is recommended for SFA disease 5-15cm (Grade 1 Level B)
 - After SFA stenting, SVS and ACC recommend at least 30d dual antiplatelet, then single anti-platelet or anti-platelet and rivaroxaban.(Hussain et al. 2018; Vascular Surgery Lower Extremity Guidelines Writing Group et al. 2015; Strobl et al. 2013; Tepe et al. 2012)

- This was in 2015 prior to the Katsanos meta-analysis suggesting there may be a mortality risk with paclitaxel drug delivery(Katsanos et al. 2018)
- Recommend against treatment of isolated infrapopliteal disease for claudication (Grade 1 Level C)
- Initial Surgical Bypass (with vein: Grade 1 Level A) is recommended in the following clinical scenarios.
 - Diffuse femoropopliteal disease
 - Small caliber vessels <5mm
 - Extensive diffuse calcification in SFA
 - Average or Low Operative Risk (Grade 1 Level B)

💡 Take a Listen

Check out our episode where we discuss overuse of early peripheral interventions in patients with claudication with Dr. Caitlin Hicks.

16 CLTI

Authors: Nedal Katib and Danielle Bajakian Contributor: Eilidh Gunn

16.1 Guidelines

Much of the content here is adapted from the most recent Global vascular guidelines on the management of chronic limb-threatening ischemia(Conte et al. 2019).

In 2019 the SVS, the ESVS and the World federation of Vascular Societies (WFVS) joined forces to put together the structure and funding of the *Global Vascular Guidelines Initiative (GVG)*. Importantly all sponsorship was directly from the societies and any direct industry sponsorship or external sources were excluded. They put together a steering committee responsible for recruiting a large and diverse writing group and outlined the scope and developed the section briefs of the guideline.

They determined that:

- "The term"critical limb ischemia" (CLI) is outdated and fails to encompass the full spectrum of patients who are evaluated and treated for limb-threatening ischemia in modern practice."
- Chronic Limb Threatening Ischemai (CLTI) was promoted as the term of choice and was defined by the target population.
- The target population were:
 - 1. Ischemic Rest Pain with confirmatory hemodynamic studies.
 - 2. Diabetic Foot Ulcer or any lower limb ulceration present for at least 2 weeks.
 - 3. *Gangrene* involving any portion of the lower limb or foot.
- Exclusion from the population:
 - 1. Purely Venous Ulcers
 - 2. Acute Limb Ischemia/acute trash foot/ischemia due to emboli

- 3. Acute Trauma or mangled extremity
- 4. Wounds secondary to non-atherosclerotic conditions

Methodology of the guidelines utilized the structure of the ${\it GRADE}$ certainty of evidence system.

They highlighted particular important sections in the evaluation and management of patients with CLTI: *Patient Risk stratification*, *Limb Assessment and Severity of Limb Threat* and the development of a specific *evidence-based revascularisation guideline in CLTI*.

It is important to note that compared to most guidelines, unfortunately in the management of CLTI, particularly when it comes to revascularisation, the level of evidence is generally **LOW**. Again, it is important to highlight that the significance of these guidelines in developing a standard approach and appropriately stratifying patients in not only management but ongoing research.

16.2 Demographics

16.2.1 Etiology and Presentation

How does PAD pathologically progress clinically into Chronic Limb Threatening Ischemia (CLTI)?

The latter stages of both the Rutherford and Fontaine Classification systems highlight this progression, with the Rutherford classification of Stage 5 being specifically minor tissue loss with focal gangrene, and stage 6 as major tissue loss identified by spreading of gangrene beyond the Trans metatarsal level.

Rutherford et al. Ad Hoc Committee on Reporting Standards, SVS/North American Chapter ISCVS:

Grade	Category	Clinical Description
0	0	Asymptomatic -no hemodynamic significant occlusive disease
Ι	1	Mild Claudication
Ι	2	Moderate Claudication
Ι	3	Severe Claudication
II	4	Ischemic Rest Pain
III	5	Minor Tissue Loss
III	6	Major Tissue Loss

What is Chronic Limb Threatening Ischemia (CLTI), sometimes previously known as Critical Limb Ischemia (CLI)?

In the last decade leading up to the 2021 Global Vascular Guidelines (GVG), the term (Chronic Limb threatening Ischemia) CLTI has been gradually replacing CLI. The GVG mentions that their "promotion" of the term CLTI is partly due to terms such as "critical or severe limb ischemia" failing to "recognize the full spectrum and inter-relatedness of components beyond ischemia that contribute to major limb amputation..."

What was the original definition and threshold for CLI, and how can we make sure we elicit the right symptoms from the patient?

John Cranley back in his publication in 1969 defined Ischemic Rest Pain as,

"...pain that occurs in the toes or in the area of the metatarsal heads. Occasionally...in the foot proximal to the metatarsal heads. Elevation of the limb above or at the horizontal position aggravates the pain and pendency...brings relief..." (Cranley 1969)

Nocturnal Rest Pain: Worse due to horizontal positioning (no gravitational assistance) and systolic BP drop during sleep.

16.2.2 Natural History

What is the natural history of CLTI and what do we know about its prognosis?

Fortunately, only a small portion of patients with Intermittent Claudication will go on to develop rest pain or tissue loss. Its estimated that anywhere between 5% -29% of patients with PAD or IC go on to develop CLTI over 5 years.

However, those that do develop CLTI, have a high risk of limb loss (greater than 20% annual risk).

Patients with CLTI have a high mortality risk (10-15% annual risk), the majority of terminal events being related to cardiovascular events. Limb loss or mortality may reach as high as 50% in 1 year. (Adam et al. 2005; Norgren et al. 2007)

What has changed in the last few decades?

Etiologically the prevalence of smokers (Ex and Current) in the population has decreased and the prevalence of diabetes has increased. For more details on the relationship between diabetes and pad, see Section 15.1.1

16.2.3 WIFi Classification

What is the WIFI Classification?

Interestingly in the original article by Bob Rutherford regarding Diabetes and PAD:

- "It was generally agreed that diabetic patients who have a varied clinical picture of neuropathy, ischemia and sepsis make the definition even more difficult and it is desirable that these patients be excluded...diabetic patients should be clearly defined as a separate category or should be clearly defined as a separate category."

Since then, the SVS, while acknowledging that we can no longer exclude these patients and treat them separately given the overlap, have decided that a new classification system is necessary, as one of the key authors (Joseph Mills) states:

• "We classify things into groups to differentiate, remember and compare, observe and predict their behavior over time." –Joseph Mills

WIfI stands for: Wound, Ischemia and foot Infection. Most of the existing Vascular and non-Vascular classification systems don't include all three components or fail to stratify the degree of ischemia and presence of gangrene.(Mills et al. 2014a)

Principles of WIfI:

- 1. *Grades, Classes and Stages* Each of the three categories (WIfI) have Grades 0,1,2,3: Resulting in 64 Classes.
- 2. **Delphi Consensus** Clinical Stages 1 (Very Low), 2 (Low), 3 (Moderate), 4 (High Risk/Benefit).
 - 1. What is the one-year risk of amputation with medical therapy alone?
 - 2. What is the potential benefit from successful revascularization?
 - 3. Analogous to TNM Staging

"It is intended to be an iterative process with the goal of more precisely stratifying patients according to their initial disease burden, analogous to TNM cancer staging, but not to dictate therapy."

Get the App!

Check out the SVS iPG App for help using the WIfi classification during your day-to-day care of CLTI patients. The best way to understand the WIfi score is to use it on a few patients and see what recommendations are provided.

16.2.4 Differential Diagnosis

What about a differential diagnosis or other causes of similar pain as rest pain?

Acute lower limb ischemia has a different clinical presentation, but there may be some overlap with Acute on Chronic disease such as in the case of in situ thrombosis in the lower limb arterial system.

Other causes of ischemic pain include:

- Buergers Disease, or Thromboangiitis Obliterans for more see Section 3.3.4
- Scleroderma
- Fibromuscular Dysplasia for more see Section 12.2
- Popliteal Artery Entrapment for more see Section 17.4.1
- Cystic Adventitial Disease for more see Section 17.4.2
- Persistent Sciatic Artery Disease
- Neurogenic pain

What is the Rutherford Acute Ischemia Grading System?

Although acute ischemia is very different from chronic ischemia, patients with progressive chronic PAD can develop an acute picture whether from embolism or in-situ thrombosis secondary to plaque rupture. See Chapter 17 for more.

16.3 Evaluation

What aspects of the clinical assessment is important?

Clinical Assessment involves a full history (the differential mentioned in Intermittent Claudication) and examination.

- Clinical Examination:
 - Buergers Test (Buerger 1908) / AKA Ratschows Test (Max-Ratschow-Klinic) identifies when there is critical ischemia without necrosis yet or gangrene, and is characterized by pallor when the leg is elevated above the level of the heart, which then turns red when hanging down over the edge of the bed. This redness is referred to as "Sunset appearance" and its due to abnormal autoregulation. Its been described that normally only a third of the capillary bed is open at any time but in a state of critical ischemia because of the autoregulation being paralyzed a significantly higher portion of the capillaries open up.

- **The ischemic Angle:** A refinement to Berger's Test: The angle of elevation from the horizontal at which the Doppler Signal of the PT or DP disappears. This is also referred to as the 'pole test', whereby the foot is raised alongside a calibrated pole marked in mmHg.
- Tissue Loss:
 - Gangrene Dry or Wet (infection)
 - Level of tissue Loss
 - Probing To Bone/ Exposed structures: Tendons, Soft Tissue, bone, Joint Capsule.
 - Examination of an Ulcer (may have many etiologies) important not only to identify extent of disease but also to exclude other etiologies:
 - Such as venous, mixed, infective, autoimmune, inflammatory, malignancy or trauma.
- Foot Infection Signs of infection, erythema, rubor, cellulitis, tenderness or unexplained hyperglycemia in diabetic patients should prompt urgent referral.(Kalish and Hamdan 2010)

16.3.1 Physiologic Testing

What other clinical evaluation is necessary for the patient with CLTI?

In addition to history, examination, and WIII assessment mentioned above, for patients with diabetes and an ulcer a full assessment of neuropathy and a "probe to bone" test for any open ulcers is recommended as part of good practice.

In the PAD and intermittent claudication sub chapters the non-invasive methods of assessment for these patients have been discussed. In addition it is important to emphasize the role of Toe Pressures (TP) and Toe Pressure Index (TPI) in this cohort of patients.

CLTI measurements most predictive of non-healing are ankle pressure <50mmHg, ABI <0.4, TcPO2 <20mmHg, and TP <20mmHg.(Gerhard-Herman et al. 2017; Wickström et al. 2017). Healing of an ulcer or tissue loss is unlikely if a patient's toe pressures are less than 55mmHg. And Toe Pressures have been validated in multiple studies to correlate with Amputation free survival and wound healing: Amputation Free Survival TP <30mmHg 2.13 HR (1.52-2.98). (Wickström et al. 2017; Hicks et al. 2018)

Patients with ESRD or DM develop medial calcification and often have elevated ABPI (>1.3) - which is associated with an elevated risk of cardiovascular mortality. Outcomes in patients with ESRD are worse in relation to amputation free survival and amputation rates, regardless of revascularization strategy.(Meyer et al. 2018) Toe pressures are particularly important in this scenario. (Resnick et al. 2004; Vitti et al. 1994) Regarding non-invasive assessment for wound healing, a TcPO2 greater than 40mmHg has the greatest correlation with amputation stump healing.(Malone et al. 1987)

16.3.2 Imaging

What imaging assessment is required?

The CLTI Guidelines outline an algorithm of attaining Arterial Anatomical Imaging. Starting with US and then depending on the information required, CTA, MRA, or eventually digital subtraction angiography. They emphasize the importance of obtaining good quality imaging to appropriately stage and be able to compare the level and degree of disease.

For wounds with concern for underlying osteomyelitis - initial workup is with plain radiography, which can identify soft tissue emphysema, evidence of osteomyelitis, or presence of a foreign body. High suspicion of early osteomyelitis with negative x-ray may warrant an MRI.(Giurato et al. 2017)

The most recently published RCT in CLTI, BEST-CLI, highlights the good outcomes of surgical bypass with saphenous vein, suggesting that vein mapping also be considered a routine imaging study during the workup of patients. This will help to clearly understand revascularization options before attempting endovascular diagnostic or therapeutic interventions.(Farber et al. 2022)

Take a Listen

Check out our episode with BEST-CLI Principle Investigators Dr. Alik Farber and Dr. Matthew Menard as they discuss the development of this trial. This episode was created as they were completing enrollment and prior to the release of the results of their study.

What is the Global Limb Anatomic Staging System (GLASS)? (Conte et al. 2019)

Because the existing arterial anatomical staging of disease is vague, "lesion focused", and not all encompassing (beyond the concept of 'in-line pulsatile flow to the foot'), **GLASS** focuses on infrainguinal disease, and attempts to incorporate all aspects in its staging to improve vascular care and evidence-based revascularisation (EBR) outcomes.

GLASS is a grading system based on anatomical and subjective assessment of calcification and incorporates two novel and important concepts:

- 1. The Target Arterial Path (TAP)
- 2. Estimated Limb-Based Patency (LBP)

As GLASS focuses on Infrainguinal disease, with the aortoiliac (AI) segment considered the inflow disease which includes the Common Femoral Artery and the Profunda Artery. Therefore, the GLASS grades assume the inflow vessels are treated and adequately 'dealt with'.

Infrainguinal disease assessment for Femoropoliteal (FP) and Infrapopliteal (IP) is based on length of disease and the extent of CTOs. The FP and IP GLASS Grades are then combined into Stages 1-3.

The calcification scale is a dichotomous subjective assessment of the degree of calcification and if there is >50% circumference of calcification, diffuse or bulky calcification or "coral reef" plaques, then there is an increase in the within-segment grade by one numerical value.

There is also mention of the Inframaleolar (IM) degree of disease (PO, P1-absent arch, P2no target artery crossing into foot) which is not included in the GLASS staging given little evidence on the outcomes this difference makes on overall patency and limb salvage.

Once the GRADES (0-4) of FP and IP disease are determined then staging (1-3) can be performed based on the matrix or grid that is provided. Staging then allows for est imated **Peripheral endoVascular Intervention outcomes** (PVI) to be predicted, Immediate Technical Failure (ITF - <10% or <20% or >20%)) and 1-year Limb Based Patency (LBP - >70%, 50-75% or <50%).

What is the Target Arterial Path (TAP)?

"The selected continuous route of in-line flow from groin to ankle. The TAP typically involves the least diseased IP artery but may be angiosome based."

Again, Get the App!

Check out the SVS iPG App for help with also the GLASS Criteria during your day-today care of CLTI patients. The best way to understand the GLASS Criteria is to use it on a few patients and see what recommendations are provided.

16.4 Management

16.4.1 Medical Managment

What are the recommendations for patients with CLTI when it comes to Medical Therapy and Risk Factor Modification?

• Treat all patients with CLTI with an **antiplatelet** agent (Grade 1 Level A). Consider **Clopidogrel** as the single agent (Grade 2 Level B) – CAPRIE(Committee 1996b)

- **High-intensity statin therapy** to reduce all-cause and cardiovascular mortality Atorvastatin 80mg or Rosuvastatin 40mg. These can be titrated down to atorvastatin 40mg or rosuvastatin 20mg if unable to tolerate. (*Grade 1 Level A*) (Arya et al. 2018; Grundy et al. 2019)
- Control **Hypertension** to BP target <140mm Hg systolic and <90mm Hg diastolic in patients with CLTI (*Grade 1 Level B*)
- Offer **Smoking Cessation** interventions and ask all smokers or former smokers about status of tobacco use every visit (*Grade 1 Level A*)
- **Diabetic foot wounds** with signs of infection, erythema, swelling, pain and foul smelling drainage should be investigated with plain radiography, ESR, CRP, cultures and, managed with iv antibiotics. Signs of systemic sepsis, such as fevers, tachycardia or shock, such as hypotension, should warrant urgent debridement and drainage, regardless of vascular status.
- For more details, see Section 15.2.2.1

💡 Take a Listen

Check out our episode with Dr. Venita Chandra, where we discuss the best practices for multidisciplinary limb salvage care.

16.4.2 Endovascular and Surgical Management

Which revascularisation management strategies exist for CLTI?

The mainstay of management for patients with CLTI have always been based on the fundamental principle of limb salvage. Given the high risk of limb loss in these patients there is a low threshold to revascularize these patients if they have occlusive disease that is treatable. But strategy has varied significantly.(Conte et al. 2019)

The CLTI Guidelines provide an approach to dealing with this complex condition on planning three aspects to each case:

1. Patient Risk Estimation

- 2. Limb Staging
- 3. Anatomic Pattern of Disease

What is involved with the Patient Risk Estimation?

Good Practice Statements (Recommendations section 6)

- "Refer all patients with suspected CLTI to a vascular specialist for consideration of limb salvage, unless major amputation is considered medically urgent."
- "Offer primary amputation or palliation to patients with limited life expectancy, poor functional status (e.g. non ambulatory), or an unsalvageable limb after shared decision-making."

Recommendation 6.3:

- Estimate periprocedural risk and life expectancy in patients with CLTI who are candidates for revascularization. Grade 1 (Strong) Level of Evidence C (Low)
- Average Surgical Risk: <5% operative mortality and 2-year survival >50%
- Severe Surgical Risk: > = 5% operative mortality and 2-year survival < = 50%

Understanding disparities is important when interpreting risk for amputation. Observational studies have found certain populations to be at higher risk of amputation regardless of disease severity. These groups include African Americans, lowest median income, medicaid insurance, uninsured, or those from regions with less access to vascular surgeons.(K. Hughes et al. 2019; Ho et al. 2005; Eslami, Zayaruzny, and Fitzgerald 2007)

What is involved and recommended with the Limb Staging and recommendation for Management?

- Use an integrated threatened limb classification system (such as WIII) to stage all CLTI patients who are candidates for limb salvage. Grade 1 (Strong) Level of Evidence C (Low)
- Perform urgent surgical drainage and debridement (including minor amputation if needed) and commence antibiotic treatment in all patients with suspected CLTI who present with deep space foot infection or wet gangrene. (Good Practice Statement)
- Offer Revascularisation to all "average surgical risk patients" (<5% operative mortality and 2-year survival >50%) with advanced limb-threatening conditions (e.g. WIfI stage 4) and significant perfusion deficits (e.g. ischemia grades 2 and 3). Particularly if they have good saphenous vein. *Grade 1 (Strong) Level of Evidence C (Low)* (Adam et al. 2005; Norgren et al. 2007)

What is involved in the Planning of the Anatomic pattern of disease and its effects of revascularisation strategy?

The overall pattern of arterial occlusive disease is a dominant factor in guiding type of revascularisation and timing of such.

Do all patients require direct in-line flow to the foot as a primary technical outcome with revascularisation?
Patients with rest pain do not necessarily require direct in line flow are those with rest pain "for which correction of inflow disease alone or treatment of FP disease even without continuous tibial runoff to the foot may provide relief of symptoms. This may also be the case in patients presenting with minor degrees of tissue loss."

Profunda-popliteal collateral index (Segmental pressures AK-BK/AK) of less than 0.25 may suggest that there is sufficient collateral network between profunda and popliteal that SFA treatment may not be necessary. (Boren et al. 1980; Mawatari et al. 2000)

What are some essential Key Factors to consider before deciding Open versus Endovascular according to the CLTI guidelines?

1. The "availability of and quality of autogenous vein conduit"

- Single segment GSV best conduit for infrageniculate bypass. (Arvela et al. 2010; Avgerinos et al. 2015; Moreira et al. 2016).
- Fem-AK pop bypass with prosthetic may be preferred to contralateral GSV. (Moreira et al. 2016)
- If the target is below the knee, then all autogenous conduits-contralateral GSV, SSV, and spliced arm vein-are preferred over prosthetic grafts.(Brochado Neto et al. 2014; Faries et al. 2000; Taylor et al. 1987)

2. Patient overall risk (as mentioned above) and Limb Staging

- Planning distal bypasses should take into account the angiosome of the wound and the most distal healthy inflow vessel to result in the shortest bypass possible.(Hingorani et al. 2016; Jongsma et al. 2017)
- Intensified anti-thrombotic therapy may be needed in patients with "high risk" infrainguinal bypasses-prosthetic conduit, below the knee target, suboptimal conduit, poor arterial runoff, extensive lesions or tissue loss. Single anti-platelet may not be sufficient, and should be intensified to asa/rivaroxaban, dual antiplatelet, or anticoagulation with Vit K antagonist.(Conte, Pomposelli, Clair, Geraghty, McKinsey, Mills, Moneta, Murad, Powell, Reed, and al. 2015; Hussain et al. 2018; Strobl et al. 2013; Tepe et al. 2012)

3. The Target EndoVascular Intervention (TVI) outcomes

What evidence do we have for deciding between Endo and Open management? What is the BASIL trial?

In November of 2022, the first results of the **BEST-CLI** were published in the NEJM. BEST-CLI is the first international, open-label, multicenter, superiority trial of its scope and scale. The key results found that in patients with adequate great saphenous vein conduit, the surgical group demonstrated a lower incidence of major adverse limb events or death (the study's

primary endpoint) than the endovascular group. They found relatively equivalent outcomes in patients with inadequate autogenous conduit.(Farber et al. 2022) The full impact of the results of this study is yet to be determined, and there will likely be several additional analyses published from this study's data. It is unlikely that this data will impact what is tested on this year's exams. However, you should keep an eye out for future results!

Take a Listen

Check out our episode with BEST-CLI Principle Investigators Dr. Alik Farber and Dr. Matthew Menard as they discuss the development of this trial. This episode was created as they were completing enrollment and prior to the release of the results of their study.

Other than this most recent study, the evidence is largely of poor quality, and is retrospective, non-controlled, or industry sponsored. **BASIL** (Bypass versus Angioplasty in Severe Ischemia of the Leg) previously had been the only multicenter RCT (BASIL -2 and 3 underway) directly comparing an endo versus open management strategy for CLTI and Infra-inguinal occlusive disease.

BASIL compared POBA and Bypass across multiple centers (27 centers, n=452, 1999-2004) in the UK. Primary endpoint was Amputation-Free Survival (AFS). (Bradbury et al. 2005)

Major Findings:

- 1. At 6-months follow up: no difference in AFS.
- 2. Intention-To-Treat Analysis of overall follow up showed no significant difference in AFS and overall survival.
- 3. Among patients who survived >2 years, overall survival was better for those treated with Bypass as a first approach
- 4. Analysis to treat:
 - Prosthetic Bypass Patients did very poorly (even compared to POBA).
 - Patients who had bypass after failed POBA had significantly worse AFS compared to those treated with a bypass as initial treatment.

Criticism of BASIL:

- 1. Majority had POBA alone (not currently best endovascular option)
- 2. 25 % of Open Bypass were Prosthetic
- 3. The technology and technical Skill with growing operator experience in Endovascular has improved.

For isolated tibial disease, first line endovascular treatment of choice is transluminal, or subintimal angioplasty. Atherectomy, stenting, and drug coated balloons are often used, but should currently not be considered first line, but this data is rapidly evolving. Patency is often poor but limb salvage is reasonable.(Popplewell and Bradbury 2019; Kayssi et al. 2016; Mustapha et al. 2016)

Optimal follow up for endovascular interventions of the lower extremity has not yet been established, but should at least include a pulse exam, ABPI and duplex to establish a new baseline after intervention.(Zierler et al. 2018; Mohler et al. 2012)

What are the BASIL 2, BASIL 3, and BEST-CLI Trials?

BASIL 2: Infrapopliteal Disease: Vein Bypass First vs. Best Endovascular Treatment first.

BASIL 3: PBA +/- BMS vs. DCB +/- BMS vs. DES

(Both Follow up 24-60 months, Primary Endpoint AFS)

BEST-CLI: Open Bypass versus Endovascular Intervention, Primary Endpoint: MALE-Free Survival. Major Above-the-Ankle Amputation, Major Bypass or Jump/interposition graft revision or the need for thrombectomy or thrombolysis (MALE).

16.4.3 Complications

What are some complications of lower extremity revascularization procedures?

For a comprehensive list of access complications after endovascular therapy, see Section 22.1

Femoral exploration carries high risk of infection or lymphatic leak. Lymphatic leaks often resolve spontaneously. Infected lymphatic leaks, particularly in the setting of prosthetic bypass require exploration. Should a lymph leak not resolve with conservative management, definitive therapy requires alcohol ablation or muscle flap coverage. The most important aspect of lymph leak is prevention with careful dissection and tissue management during femoral exploration.(A. Obara et al. 2014; M. V. Weaver et al. 2014)

17 Acute Limb Ischemia

Authors: Alex Forsyth and Sarah Carlson Contributors: Eilidh Gunn

17.1 Overview

17.1.1 Demographics

17.1.1.1 Etiology

What is Acute Limb Ischemia and what does it encompass?

Acute Limb Ischemia (ALI) is defined as any process that leads to an abrupt cessation of blood flow to a limb resulting in ischemia. There are several causes, but the most common two are embolic and thrombotic.

- Embolism
 - Cardiac
 - Typically due to atrial fibrillation.
 - Acute upper limb ischemia is most commonly due to cardiac embolism.
 - Endocarditis as seen in IV drug users or patients with bacteremia from other causes.
 - Cardiac tumors such as atrial myxoma.
 - Atherosclerosis e.g. iliac disease embolizing downstream to the lower leg.
 - Paradoxical Embolism Thromboembolic venous system with PFO.
 - Aneurysm e.g. thrombus from within an aortic aneurysm embolizing downstream to the leg.
- Thrombosis

- **Aneurysm** Small aneurysms (such as popliteal) are less likely to rupture, but more likely to thrombose or embolize distally, causing an acute limb ischemic event.
- Acute on Chronic Progression of Atherosclerosis Acute on Chronic Progression of Atherosclerosis: Once a chronic stenosis becomes critically tight, platelet thrombus can develop leading to an acute occlusion; or unstable plaque can "rupture" leading to an acute occlusion of a chronic lesion.
- Bypass Graft
- *Dissection* for more, see Chapter 6
- Thoracic Outlet Syndrome in the upper extremity fore more, see Section 4.3
- Vasospasm (severe) for more, see Section 3.3

17.1.1.2 Presentation

What is the patient presentation of ALI? Are there any differences between upper and lower extremity presentations?

Classically remembered by the 5 or 6 Ps (depending on who you ask)

- **Pain**: Usually located distal to the occlusion and gradually increases in severity as the ischemic time increases until neuroischemic sensory loss occurs.
- *Pallor*: the limb appears pale compared to the non ischemic limb. There is delayed or absent capillary refill.
- **Poikiolothermia**: (just a way to make "cold" into a "P" really means cold limb) Literally the inability to regulate one's body temperature, or dependent on ambient temperature as cold blooded animals are. If there is no limb perfusion, it acclimates to the ambient temperature and feels cold to the touch.
- **Pulseless**: Self explanatory. It is worth examining the contralateral limb, if it has normal pulses, it suggests the absence of PAOD or CLTI making thrombosis in situ less likely to be the cause of ALI.
- **Paresthesia and Paralysis** are the last two Ps. Paresthesia is an earlier sign of ischemic nerve dysfunction, and paralysis is a later sign. In the lower leg, ischemic changes often affects the anterior compartment first, and sensory loss over the dorsum of the foot is, therefore, one of the earlier neurologic deficits in ALI.

This is why a thorough physical exam is key; comparing both limbs and including handheld doppler exam. It can be difficult to tell whether a limb is acutely threatened, especially in patients with chronic disease where the presentation of an acute change can be more subtle. This is why it is also essential to assess sensory and motor function in addition to the pulse exam. This helps gauge the urgency of intervention.

Acute paralysis, mottling of bilateral lower extremities and absent femoral pulses should raise concern for aortic occlusion.(J. C. Wang, Kim, and Kashyap 2016)

How is ALI classified?

From Rutherford RB, Baker JD, Ernst C, et al. Recommended standards for reports dealing with lower extremity ischemia: revised version. J Vasc Surg. 1997;26:517–538. (Rutherford et al. 1997)

- 1. Viable Not immediately threatened
 - Sensory Loss: None; Motor Weakness: None
 - Arterial Signal: Present; Venous Signal; Present
- 2. Threatened
 - a. Marginally Salvageable if promptly treated (<24hrs)
 - Sensory Loss: Minimal (toes) or none; Motor Weakness: None
 - Arterial Signal: Weak/absent; Venous Signal: Present
 - b. Immediately Salvageable if immediately treated (<6hrs)
 - Sensory Loss: More than toes; Motor Weakness: Mild/Moderate
 - Arterial Signal: Absent; Venous Signal: Absent
- 3. Irreversible Major tissue loss or permanent nerve damage inevitable; Minimal benefit from revasc
 - Sensory Loss: Profound anesthesia; Motor Weakness: Dense paralysis (rigor)
 - Arterial Signal: Absent; Venous Signal: Absent

17.1.2 Evaluation

What does the work up for ALI entail? How is the diagnosis made?

- The diagnosis can often be made on history, physical exam, and bilateral ABPIs. Imaging can be done in patients in who the diagnosis is uncertain.
- As with most urgent cases, the type of imaging done depends on the availability at your institution, but generally CT angiography or arteriography should be done on viable and marginally threatened limbs. Arteriography often can distinguish between embolic vs arterial thrombosis which may help to direct management.
- Bedside ultrasound can also be very helpful, especially color flow doppler US.

The decision to perform cross-sectional imaging before angiography varies depending on how severe the presentation is and how quickly you can obtain imaging. The patient's renal function may also influence the decision to administer two contrast loads (CT followed by endovascular intervention). If a femoral pulse is present it could be feasible to proceed with on-table angiogram without a CT scan. If femoral pulses are absent there is increased suspicion of aortoiliac disease and therefore CT angiogram will be beneficial for both diagnosis and operative planning.

In patients with severe renal insufficiency, MRA or MR time-of-flight can be helpful, but these studies usually take longer to obtain and may not be quickly available in an acute threatened limb situation. Therefore going straight to angiography should be considered, as to get the most immediate and best imaging of the tibial vessels. (Creager, Kaufman, and Conte 2012; Earnshaw 2019)

17.1.3 Management

What is normally done in the initial management of ALI?

- Anticoagulation: IV unfractionated heparin should be immediately administered to prevent proximal and distal progression of secondary thrombus as long as heparin is not contraindicated. The dose should be titrated to maintain activated partial thromboplastin time between 50 and 80 seconds (2-3 times normal values). A good rule of thumb for IV heparin is to start with a bolus of 80-100 units/kg, and then drop at 18units/kg/hr titrating to PTT at 2-3x normal or follow local protocol.
- Supportive Care: Analgesia, iv fluids.
- *Labs*: A full panel including serum chemistry with BUN, Cr, CBC, and coagulation studies should be obtained. Baseline plasma CPK can be helpful to monitor for evidence of rhabdomyolysis after reperfusion.

17.1.3.1 Medical Management

Primarily with anticoagulation using heparin or a direct Xa inhibitor.

17.1.3.2 Surgical Management

- *Thrombectomy*: Balloon catheter based (Fogarty embolectomy balloon Dr. Fogarty invented this while he was a medical student)
 - Patients with acute limb ischemia, neuro deficit and distal pop embolism may be best treated with popliteal exposure and open thrombectomy. (Darwood et al. 2018; Kempe et al. 2014)
- Bypass
 - Patients with a ortic occlusion, multiple life limiting comorbidities, and significant a ortoiliac baseline disease may benefit from primary axillo-bifemoral bypass. (Mohapatra et al. 2018)
- **Endarterectomy** not necessarily required but might be used for common femoral exposure and distal access depending on vessel disease burden.

17.1.3.3 Endovascular Management

- Pharmacological Catheter Directed Thrombolysis
 - Absolute contraindications include active bleeding disorder, CVA (<6mo), CNS injury or head injury (<3mo), or GI bleed (<10d).
 - Relative contraindications include recent major surgery, uncontrolled hypertension, intracranial tumor, pregnancy, recent eye surgery, hepatic failure, CPR (<10d), or bacterial endocarditis.
 - 1-2% risk of hemorrhagic stroke. If neurologic deficit develops during thrombolysis, stop lysis and perform immediate CT head. (Kenneth Ouriel, Veith, and Sasahara 1998)
 - If fibrinogen drops below 100mg/dl, recommend immediate cessation of lytic agent and consider cyroprecipitate (includes fibrinogen, factor VIII, von Willebrand's factor, factor XIII).(Mann 2019)
 - Pharmacological thrombolysis may be delivered over a short duration in the operation room, or a catheter may be left and tPA administered over a longer period. Two catheters for delivery include an Cragg-McNamara catheter with multiple side holes or an EKOS catheter that uses ultrasonic waves to aid with thrombus lysis.

- **Percutaneous Thrombus Aspiration**: Useful for small fresh thrombi such as after angioplasty, as distal diameter of the catheter tip limits the size of the thrombus that can be removed.
- *Mechanical Thrombolysis and Aspiration*: Are also useful for patients with contraindications for thrombolytic therapy, and also may allow for a lower dose of a thrombolytic agent, but risk damage to the arterial wall.

17.1.3.4 Relevant Literature

STILE Trial: One of the first large RCTs comparing catheter thrombolysis with open surgery.

- Overall the study showed some short term benefit to open surgery however this can probably be attributed to:
 - 1. In 28% of patients randomized to CDT they weren't able to get a catheter in place so these patients were considered treatment failures and crossed over to the surgery arm.
 - 2. Patients with very long duration of ischemia, up to 6 months, were also included in the study.
- When outcomes for patients who had been symptomatic for less than 2 weeks were analysed, the thrombolysis patients actually did better. (T. S. Investigators 1994)

TOPAS Trial: Larger RCT which enrolled patients who had an acute arterial occlusion of less than 14 days. This showed no difference in mortality or amputation-free survival between the open surgery and CDT groups but higher major bleeding in the CDT group. (Kenneth Ouriel, Veith, and Sasahara 1998)

Meta Analysis: Originally published in 2002 but updated in 2013 and 2018. Demonstrated no difference in mortality or limb salvage between surgical and thrombolytic therapy, but endovascular demonstrated higher rates of complications including ongoing limb ischemia and bleeding within 30 days of treatment. Previously they had reported higher rates of stroke in the thrombolysis category but the most recent update is unable to support this finding. (Darwood et al. 2018)

• Evidence considered, it is very reasonable to think about a catheter-directed therapy especially if the presentation is acute, less than 2 weeks or so. There are, however, certain anatomical locations, such as common femoral or brachial arteries, that most surgeons would favor a simple open procedure (i.e. balloon thrombectomy).

17.1.3.5 Decision Making

Who gets which kinds of treatment? Who needs emergent treatment?

- **Class I** patients might just need medical therapy like anticoagulation and revascularization can be elective.
- Class IIb patients do not need immediate revascularization
 - If symptoms have been present for less than 2 weeks endovascular therapy is preferred
 - If more than 2 weeks or lytic therapy has failed then surgical intervention is preferred
- **Class IIb** need immediate revascularization. Historically surgical revascularization has been preferred because of its immediacy, but catheter directed thrombolysis and percutaneous mechanical thrombectomy have shortened time to revascularization.
- **Class III** ALI is usually treated with primary amputation because revascularization is unlikely to restore function to the limb and restoring blood flow can cause the patient serious harm.
 - What are the risks of revascularization for a class III or prolonged ischemia? Myonephropathic metabolic syndrome: muscle cells undergo liquefaction necrosis due to ischemia. Potassium, myoglobin, lactic acid, and superoxide accumulate and can perfuse through the body or can have a sudden increase in the event of revascularization which leads to hyperkalemia, arrhythmias, pulmonary edema, metabolic acidosis, myoglobinuria, and can even cause sudden death from heart and/or renal failure. (H. Obara, Matsubara, and Kitagawa 2018) Treatment of this is largely supportive with fluids.

17.2 Compartment Syndrome

17.2.1 Etiology

What is the pathophysiology and presentation of compartment syndrome?

• Increased intramuscular compartment pressure results from increases in capillary permeability due to ischemic reperfusion. The increase in pressure leads to neuromuscular dysfunction and interferes with circulation. Irreversible damage occurs when pressures exceed 30mmHg in each compartment.

- Ischemia reperfusion causes increased capillary permeability due to free oxygen radicals, neutrophils and endothelial factors that collect during ischemia. Severity depends on time to reperfusion, muscle mass, and flow pattern (i.e. direct or collateral). Symptoms often develop within 6 hours of reperfusion. (Keudell et al. 2015)
- The pathophysiology underlying acute extremity compartment syndrome is related to the arteriovenous pressure gradient theory. Increased compartment pressure reduces the gradient, reducing arterial pressure and increasing venous pressure.(Frink et al. 2010; McQueen and Court-Brown 1996; Elliott and Johnstone 2003; Papalambros et al. 1989)
- Clinically patients can have neurological dysfunction with sensory motor deficits, but the most common presentation is a tense extremity with severe pain on passive movement of the muscles in the compartment. If the lower leg is affected this is often on dorsiflexion/plantar flexion of the ankle. A sensitive indicator is loss of two point discrimination (proprioception).

17.2.2 Evaluation

How do you diagnose compartment syndrome?

- **Physical Exam**: Tenderness, (especially over anterior compartment) and paresthesias (especially between first and second toes).
 - *Anterior Compartment*: Sensory distribution and muscular innervation from the deep peroneal nerve (this is a VSITE favorite).
 - **Deep Posterior Compartment**: Sensory distribution and muscular innervation from the tibial nerve(Velmahos and Toutouzas 2002).
- Compartment Pressures:
 - A needle is required to access the compartment and a pressure monitoring system (this can be handheld Stryker kit, or just a hollow bore needle connected to an arterial pressure bag).
 - Normal compartment pressure is <10-20mmHg; greater than 30 is highly concerning. More accurate than an absolute number is comparing the compartment pressure to the mean arterial pressure or diastolic pressure. If the compartment pressure is within 40mmHg of the MAP (for example, MAP is 60 and compartment pressure is 25 this is concerning) OR if the difference between compartment pressure and diastolic pressure is less than 10 (for example, diastolic pressure is low at 30, and compartment pressure is 22).

If the clinical suspicion of compartment syndrome is high, do not be reassured by "normal" compartment pressures. It is relatively low risk to do fasciotomies, but the risk of limb loss is high for a missed compartment syndrome.

17.2.3 Management

Management of compartment syndrome is a fasciotomy, what are the types of fasciotomies?

- Forearm and Upper Arm Fasciotomies: Often performed by orthopedic or hand surgeons. The forearm fasciotomy includes dorsal and volar incisions to release the dorsal and volar compartment, and mobile wad, while avoiding numerous superficial cutaneous nerves. The arm fasciotomy releases the medial, lateral, and deltoid compartments through medial lateral incisions.
- **Thigh**: Performed through medial and lateral incisions to release the lateral, medial and posterior compartment.
- Lower Leg: The most common type is a lower leg 4 compartment fasciotomy and is explained in more detail below.

How is a lower leg 4 compartment fasciotomy for the lower extremity performed?

- 1. A longitudinal incision is created between the fibular shaft and the crest of the tibia over the intermuscular septum and the anterior and lateral compartments are opened. If tissues are swollen occluding the view of the intermuscular septum, the perforating vessels can be followed down to it. Nerves including the peroneal nerve are most at risk near the fibular head.
- 2. A second incision is created on the medial surface of the lower leg approx. 1cm posterior to the edge of the tibia to avoid the greater saphenous vein. The superficial posterior compartment is incised. The gastrocnemius-soleus complex is taken down from its attachments to the tibia in order to access the deep posterior compartment.
- 3. The incisions should be made generously. Sometimes the skin incision can be a little short of the fascial incision, but they should be long in order to fully release the compartments.
- 4. After hemostasis loose dressings should be applied. The leg should be elevated to reduce edema that can complicate closure. Closure can be done in 48-72 hours, but may be delayed and dressed with wound vacs to attempt primary closure. If primary closure is not possible, a split thickness skin graft can be used.

Who should a prophylactic fasciotomy be performed on?

- Patients with high occlusion and extensive ischemia.
- Acute ischemia of greater than 6 hours with few collaterals.
- Patients with combined arterial and venous injury.
- Patients who are obtunded making serial examination difficult.

What is the prognosis for a patient with ALI? What are some patient factors that lead to a poor prognosis?

Amputation rates after acute limb ischemia are typically described in the 10-20% range, and mortality is also in the 10-25% range whether you're talking about surgery or catheter-directed procedures (that's excluding the patients who present with Rutherford class III and by definition have an unsalvageable limb). Many factors determine likelihood of amputation; typically, patients with more medical comorbid conditions tend to do worse as you might expect: baseline CAD, kidney disease and smoking are predictive of worse outcomes. There is a trend toward improved limb salvage rates (decreased amputation rate) over time, and I think this speaks to wider availability of different limb salvage techniques among vascular surgeons across the globe.

17.3 Blue Toe Syndrome

17.3.1 Etiology and Presentation

Blue toe syndrome is characterized as a painful discolored toe in the setting of a normal vascular examination or palpable pulses.

Acute onset of blue toe after MI - consider cholesterol embolization from wire manipulation in the aorta. (Saric and Kronzon 2012)

17.3.2 Evaluation

Patients should have a work up for investigation of an embolic source. Patients without a clear recent source of embolization on history or physical examination, should undergo:

- CTA of chest/abdomen/pelvis to look for an arterial lesion.
- Echocardiogram to identify a cardiac source.
- Holter monitor to evaluate for an underlying arrhythmias.

If no obvious source of embolism can be identified, further investigation for underlying cancer may be required due to malignancy leading to prothombotic state.

17.3.3 Management

Cholesterol Embolization: Medical management with a single antiplatelet agent and statin. Anticoagulation may precipitate further embolization. (Ghahramani, Seline, and Wanat 2016; Quinones and Saric 2013)

Specific Thrombogenic Arterial Lesion: Medical management with anticoagulation. Recurrent symptoms on anticoagulation may require stent coverage or, in a young patient, thrombectomy. (Reyes Valdivia et al. 2017; Jeyabalan et al. 2014; Verma et al. 2014)

17.4 Non-atheromatous Popliteal Artery Disease

17.4.1 Popliteal Artery Entrapment Syndrome (PAES)

17.4.1.1 Etiology and Presentation

Most often seen in young healthy patients (often athletes) who present with claudication.

Etiologically due to embryologically abnormal lateral attachment of the medial head of the gastrocnemius muscle. (Gokkus et al. 2014; Lejay et al. 2014) Type IV PAES caused by compression with popliteus muscle.

17.4.1.2 Evaluation

Physical examination findings will be loss of pedal pulse on active plantar flexion or passive dorsiflexion of the foot.(Gokkus et al. 2014; Lejay et al. 2014)

Best diagnosed with axial cross-sectional imaging from the MRA to identify the abnormal muscle course and insertion. (Sinha et al. 2012; Lejay et al. 2014)

17.4.1.3 Management

Symptomatic PAES should be treated, even if presenting with mild or moderate claudication, due to risk of progression and thrombosis from scaring.(Forbes and Kayssi 2019)

17.4.2 Cystic Adventitial Disease

17.4.2.1 Etiology and Presentation

Controversial etiology, often considered related to repeated popliteal trauma due to knee flexion, causing fluid to collect between the adventia and media layers of the artery wall. Presentation overlaps with atherosclerotic lesions of the popliteal segment, but in patients without traditional risk factors.

17.4.2.2 Evaluation

Often diagnosed on duplex ultrasound or CTA. For a representative image, see Section 20.5.1.4

17.4.2.3 Management

Compression can sometimes be relieved by percutaneous drainage of large cystic portion that is causing luminal compression. PTA and stenting should be avoided as often poor durability. Operative resection and reconstruction with an interposition graft can be effective but should be considered as a final management strategy. (S. Li et al. 2017)

17.5 Amputations

The majority of questions regarding amputations stem from decision making around peripheral arterial disease. You may not see many questions asking uniquely about amputation technique or management, however this topic is VERY important for the successful management of peripheral arterial disease. Rutherford's has two fantastic chapters on decision making(Bianchi and Jr 2018) and techniques(Eidt and Kalapatapu 2019).

The most commonly performed lower extremity amputations include:

17.5.1 Above-knee, trans-femoral amputation (AKA)

Above the knee amputations are most often performed with a fish mouth incision. Suture ligation of the neurovascular bundle at the level of the mid SFA and transection of the femur.

17.5.2 Below-knee, trans-tibial amputation (BKA)

Below the knee amputation often performed with a long posterior, or Burgess, flap. Each tibial neuro vascular bundle is ligated individually. The fibula is transected 2-3cm more proximally than the tibia to avoid weight bearing from the prosthetic.

Although hotly debated, a Cochrane review demonstrated no difference in outcomes based on flap type. However, the review does recommend performing amputations in two stages in the setting of wet gangrene. (Tisi and Than 2014)

17.5.3 Transmetatarsal amputation (TMA)

Transmetatersal amputation is performed with transection of the metatarsal bones just distal to the heads. Best to transect in a graduated parabola with the 5th metatarsal most proximal. Patients may benefit from concomitant Achilles tendon lengthening to prevent equinous deformity that can result in ulceration of the medial aspect of the TMA. (Eidt and Kalapatapu 2019)

18 Venous Disease

Authors: Leanna Erete, Andrew Nickinson, Aminder Singh and Manj Gohel

18.1 Chronic Venous Insufficiency

18.1.1 Terminology and Presentation

What is chronic venous insufficiency and how common is it?

I thought the first question was going to be nice, easy and uncontroversial, but you've started with CVI. What you've highlighted is one of the main problems with venous disease, which is terminology. So if you type in chronic venous insufficiency into Google, you end up with, this enormous range of descriptions and how its used in lots of different ways, by lots of different people.

One of the problems is that this led to a lot of confusion among people within the venous space and everywhere else. A really important documents to guide people is the VEIN-TERM consensus document published in 2009. (Eklof et al. 2009) The lead author was Dr. Bo Eklof and this was a document where a group of venous experts around the world got together and said, okay, we need to be clear and define what all these different terms mean.

And one of the terms that was described was chronic venous insufficiency. The official definition is it is a venous disease between C3 and C6 on the CEAP classification. We'll talk about that a bit later, I'm sure. But between C3 and C6. In practice, it is used to describe the entire spectrum of venous disease ranging from thread veins C1 all the way up to venous ulceration. But technically it is C3 to C6. And I would urge anybody who is learning about the terminology to look at the vein term documents.(Eklöf et al. 2004; Eklof et al. 2009)

And silly little things like great saphenous vein and small saphenous vein–I think UK is the only place where long saphenous is ever used and the real problem was LSV in lots of other countries will be the lesser saphenous vein, which is the small saphenous vein. So again, I'd urge anyone to look at that document.

How common is chronic venous insufficiency?

Chronic venous disease is very common, with the weather getting a bit nicer, people wearing shorts, you're spotting, venous disease all over the place. The studies that have been done

indicate that at least a quarter, probably up to a third of people have C2 to C6 disease. So a very common problem, with lots of people that need treatment.

The Edinburgh Vein Study found that half of the general population with chronic venous disease deteriorated over the 13 year study. Factors associated with disease progression included family/personal history of DVT. Of note, female gender, obesity, number of pregnancies, and smoking status were not associated with progression.(C.-C. Lee et al. 2015)

Now we would like to discuss the classification and scoring systems, you mentioned CAEP, could you expand on that?

So CEAP-clinical, etiological (with an E because it's American), anatomical, pathophysiologicalclassification system was introduced, I think in the mid 1990s, by the American Venous Forum. And it's a classification system.

So if you have a patient that comes into your clinic with a venous problem, it's a system to allow you to describe that patient's current status in an objective, clear way. What it does not do is give you any information about prognostication and it isn't useful for measuring responses to treatment.

So it is purely a descriptive classification system. And the most commonly used part is the clinical bit, which is C1 to C6. CEAP was just recently revised last year, a 25 year revision, which again, I would urge trainees to have a look at, Dr. Fedor Lurie was the lead author, and th is was a useful revision of CEAP.(Lurie et al. 2020)



Now in response to the limitations of CEAP, the Venous Clinical Severity Score (VCSS) was created. So this is a 10 item scoring system, each item scores between zero and three. So you get a maximum score of 30. Now this is designed to be responsive to treatment and complimentary to CEAP.(Vasquez et al. 2010)

There are lots of other things that have been described, quality of life tools or VDS, but I would say that those two (CEAP and VCSS) are the most relevant and important ones to know about. Many of the QOL tools are too cumbersome to use in day to day practice. We are in dire need of a easy to use clinical tool that is validated for assessment of patient reported

outcomes. And again, the number of papers I've reviewed where people have reported CEAP scores as one of the outcome measures in response to treatments. That is not what it's designed for, for example, you can never get lower than C5. You can never get better than a healed ulcer. So again, that's the important distinction.

18.1.2 Evaluation

When we do see patients in clinic with chronic venous insufficiency, which type of imaging modalities should we use to investigate the function and the anatomy of the lower limbs?

If a patient comes to clinic, we can't underestimate the importance of the clinical assessment. And when you assess these patients, you are really looking for the impact of their venous disease on their quality of life and the impact of the venous disease has on their normal function.

But also you're looking for complications of venous disease. This is a clinical assessment. But then when ordering investigations, you really want to identify underlying treatable causes of their venous disease. And that's the goal of any investigation.

Venous duplex imaging is obviously the first line, gold standard that pretty much anybody will go to around the world. And what that does is essentially gives you information about the venous flow, antegrade or retrograde or refluxing flow in superficial and deep veins.

It'll give you information about the anatomy of the veins, which will help you to plan treatment, but it also gives you lots of other information as well. It gives you information in the common femoral vein, for example, if there's good phasicity of flow or if there's any scarring. It helps you to evaluate whether there might be some outflow obstruction, so duplex imaging is the absolute first line, gold standard. For more on venous duplex scanning, see Section 20.7.

Now, once you've done the duplex imaging, if there's clinical suspicion, then of course we can move on to other investigations. So proximal CTV, MRV, venography, IVUS. But clinical assessments and duplex assessment are the bedrock of every assessment of every venous patient. So that's the foundation.

18.1.3 Management

Moving on to the treatment of superficial disease. When we talk about treatment of superficial venous incompetence, I don't think we can really get away from talking about the two major studies, the ESCHAR(Manjit S. Gohel et al. 2007) and the EVRA(Manjit S. Gohel et al. 2018) studies, which I know you've been involved with. Can you review what these studies showed and their importance in the care of our patients. Before I go into the trials, I'm just going to go back a step and just talk about the general mindset that's important for these patients. So what we're treating with these patients is not superficial venous reflux. What we're treating is chronic venous hypertension. So we're treating a pathophysiological entity of which the superficial reflux is one correctable factor. And the reason that's important is that there may be other correctable factors, such as venous outflow obstruction, which may be addressed. Decision making also takes into account patient presentation. For example a patient with asymptomatic varicose veins, even with proximal obstruction and superficial reflux, should still only be treated with compression. (Boezem et al. 2011; Wittens et al. 2015)

But there are also going to be lots of uncorrectable factors, such as poor mobility, heart failure, dependency, ankle stiffness. If your ankle is stiff, you can't use your calf muscle to pump effectively. You don't have that mechanism for reducing the venous pressure in your legs. So the success of these treatments, particularly for the ulcer population, has to be taken in the context of treating venous hypertension as well.

Having said that, the aim of these trials was really very simple. Most patients with venous ulceration have superficial reflux. So what is the role of treating the superficial reflux? That's the simple question that was asked by these trials.

The ESCHAR trial recruited between 1999 and 2002, and included 500 patients with healed and open leg ulcers. They were either randomized to compression or compression and traditional, superficial venous surgery (stripping or ligation). And that trial was very clear. It showed that with venous stripping there wasn't a benefit for healing, but there was a significant and sizable reduction in the risk of ulcer recurrence.(Manjit S. Gohel et al. 2007)

So one of the criticisms of ESCHAR was lots of people didn't have the same surgery. Lots of people had just ligation alone, et cetera. We don't really do stripping anymore, it's all endovenous. So EVRA really aimed to bring the ESCHAR trial up to modern practice. We randomized 450 patients with open leg ulcers to early endovenous treatments, delivered within two weeks versus a more delayed approach. What EVRA showed, again is common sense, if you deal with the underlying problem (usually superficial reflux), the venous ulceration healed significantly quicker.(Manjit S. Gohel et al. 2018)

So putting them together, there is a clear, unequivocal argument for treating the superficial reflux in these patients as quickly as possible.

18.1.3.1 Endovenous

Can you broadly outline the different treatment options that can be offered to patients with superficial venous reflex?

I have counted about 42 different ways of destroying a saphenous vein. There are lots of different treatment options. Do something that you're trained in and proficient with and something that's appropriate. Thermal ablation is in the NICE guidance and in lots of other

international guideline documents has been combined, whether it's radiofrequency or laser. Although foam sclerotherapy therapy, when studied in randomized trials, has lower technical success rates and lower vein closure rates, the clinical success rates in terms of healing are just as good. This reduced effectiveness in closing the vein, doesn't really seem to translate into a significant drop in clinical effectiveness after foam sclerotherapy.

So I think the most important thing is not necessarily what you use, it's doing it quickly and delivering it to the people that need it. However, there are a few relative contraindications to GSV ablation.(Atasoy 2015; Gloviczki et al. 2011; Lowell S. Kabnick and Sadek 2019) These include:

- GSV >12mm width raises concern for incomplete obliteration
- Superficial GSV unlikely to be pushed down with tumescence could cause staining or thermal injury to the skin
- GSV tortuosity may limit the ability to pass a catheter and sufficiently ablate the vein
- Acute superficial vein thrombosis.

💡 Take a Listen

Check out our episode with Dr. Andrew Meltzer as he discusses some controversies in variations in practice regarding the use of endovenous ablation procedures in the United States.

When we are consenting patients for endovenous intervention, what risks should we discuss with the patient and that they need to be aware of?

You can spend an hour doing this, if you're being very diligent. But what I say to patients are that there are generic risks of venous thromboembolism, as is the risk of any intervention. And I think that's worthy of specific discussion after endovenous procedures, but I quote a risk of around one in 200 for most endovenous interventions. *Extension of proximal thrombus into the deep system is termed endothermal heat-induced thrombosis (EHIT) and is evaluated and treated aggressively in the United States.* (Sadek et al. 2013; Lowell S. Kabnick et al. 2021) EHIT is defined as:

EHIT Grade	Description	Treatment
1	Extension to SFJ	None
2	Extension into deep system ${<}50\%$ of	Close observation vs
	lumen diameter	anticoagulation
3	Extension into the deep system $>50\%$ of luman diameter	Anticoagulation
4	Occlusion of the femoral or popliteal	Anticoagulation
	vein	ThreedSalarion

There's always a risk of bruising. And if you do phlebectomies, what I often tell people is that bruising sometimes looks worse than it feels. And with the thermal interventions in particular, I think the risk of nerve related complications, so some numbness or some nerve pain, are worthy of mentioned, and of course the risk of recurrence. Recurrence of varicose veins in the lateral thigh may be related to untreated anterior accessory greater saphenous vein disease.(Laredo, Lee, and Neville 2010)

Foam sclerotherapy has a different mechanism of action. It's a detergent that's injected into the vein that destroys the endothelium. And there are some people that have associated foam sclerotherapy with some neurological events. So a very large registry that was organized by the manufacturers of STS recently looked at 10,000 patients in the UK and found very few events-around one in 4-5,000.

There are a number of complications related to sclerotherapy that should be understood. Hyperpigmentation is probably the most common complication (11-80%), but only a small percentage persist up to a year (1-2%). Incision and drainage of the thrombus 2-4 weeks after treatment may reduce hyperpigmentation. New appearance of fine red telangiectasias (AKA telangiectatic matting) is related to underlying vessel injury in 5-57% of patients and can persist up to 1 year (1%). Cutaneous necrosis is rare (<1%) and results from extravasation of sclerosing agent, injection into dermal arteriole, reactive vasospasm or excessive cutaneous pressure. DVT and cutaneous nerve injury are also rare.(Bergan, Pascarella, and Mekenas 2006; Munavalli and Weiss 2007)

Some people say that if there is a history of migraine, then the risk of these neurological events might be a little bit higher. So, a little bit of caution, although in practice, I've treated many people with migraines. But if you believe the literature, one in four or five, people have got a little ASD and obviously we treated lots of those people, so the risks are very low but it's just important to have that discussion beforehand.

Are there any techniques that we can use to mitigate some of these risks?

So I think VTE is something that is really worth specifically discussing. It's a very rare event after superficial venous interventions, but it's potentially catastrophic, with sometimes national press coverage for bad outcomes. What we've got to realize is that the rate of VTE events is very low, there are almost certainly some people that have a higher risk. And so what we've got to get slightly smarter about is identifying the people that have a higher risk of VTE and maybe giving them prolonged course of thromboprophylaxis, as well as appropriate counseling and mitigation as well.

So it's about getting smarter about risk assessment. The current risk assessment tools are not really fit for purpose when it comes to superficial venous intervention. So the department of health or the Caprini tools don't really take into account some of the really important factors that I think do impact on VTE risk after superficial venous interventions.

There's nothing validated, there's nothing widely available, but we have a local, specific risk assessment for patients having superficial venous interventions. There are major risk factors we

look out for, such as previous ipsilateral VTE, active malignancy, but also significant chronic inflammatory conditions (i.e. inflammatory bowel disease), these patients have a much higher risk of VTE than we previously recognized. There are also minor factors, superficial vein thrombosis, obesity, comorbidities. All of these things are put together and we come out with a score. For patients above a certain threshold, they get offered extended rivaroxaban or low molecular weight heparin.

In addition, it is important to implement mitigation. So traditionally things like a contraceptive pill, HRT and Tamoxifen, we wouldn't usually stop them for a local anesthetic procedure, but being on an estrogen containing pill doubles or triples your risk of VTE. It's an easily reversible thing in the short term that can be done. So for my patients, I would normally recommend that we stop the pill or HRT a full month before and after the procedure.

For preventing EHIT specifically, the radiofrequency ablation (RFA) catheter should be placed 2cm from the SFJ. EHIT is a primarily technical complication. (Joh et al. 2014)

You mentioned it an extended course of anticoagulation for those patients with risk factors. Is there a specific timescale locally that you use for this?

The venous forum produced some guidance the year before last in exactly this area. We often only do about a week or 10 days for a lot of these patients. However, if there's an ongoing, persistent risk factor, then actually the VTE risk persists for four to six weeks, and thus there may be an argument for up to six weeks of therapy in particularly high risk patients.

However, if they are really that high risk, you've really got to question whether a superficial venous intervention is needed, balancing the risks and the benefits. But since for some patients the high risk period persists for at least another six weeks afterwards, be careful not to stop prophylaxis too soon.

Moving on to a more specific question about treatment, what is the role for treating and incompetent perforator veins in these patients?

In general, the more advanced the venous disease, the more likely it is that the disease is recurrent, the more likely there is to be deep venous disease or posts thrombotic disease, and the more likely you are to finding incompetent perforators. If you imagine these patients with venous ulcer, it is very common to find perforators, and if you take a sort of super aggressive role saying you've got to obliterate every single bit of reflux you can find, then you'd be doing a couple of perforators in every single patient, and that is the approach of a number of centers around the world.

The pragmatic reality is, number one, if you ablate the superficial reflux, we know that a lot of these perforators actually become competent because we've got rid of the outflow or you've changed the dynamics in some other way, and they've become competent. Number two, even if they've stayed incompetent, the clinical benefit is still there and is usually still pretty durable. The clinical criteria most often used to as an indication for treatment is >350ms of deep to superficial reflux, diameter greater than 3.5mm (associated with reflux)

in 90% of patients)(Sandri et al. 1999) and near a healed or active ulcer.(O'Donnell et al. 2014; Min, Khilnani, and Golia 2003; Rueda et al. 2013)

Now, of course, there may be some people that develop recurrent problems and then there can be a more targeted approach to the perforator. The ESCHAR and EVRA studies did not target perforators at all and the outcomes in the EVRA study was the best healing rate of any published prospective leg ulcer study. So, putting all that together, it's difficult to make a case for aggressive treatment of perforators first up. Having said that for some people with recurrent disease, it's not uncommon to have a big mid thigh Huntarian perforator or Cocker's perforator lower down on the medial calf. So, I do treat perforators, but almost always it's in recurrent disease with recurrence or deteriorating symptoms.

18.1.3.2 Open Surgery

So we've discussed quite a bit about endovenous intervention. Is there a role for open surgery in some patients?

Open surgery remains the most commonly used superficial venous intervention around the world, and there are still lots of centers in the UK that primarily offer open surgery. Being balanced, it's a very effective treatment if you do it well and if you use modern approaches, such as tumescent anesthesia and ultrasound guidance, then actually some of the traditional issues with open surgery, which are often technical and complication related, don't really apply.

Having said that, all the randomized trials have shown the same thing while the effectiveness may be as good in open surgery done well, the recovery is much better after endovenous ablation. So it's difficult to make the case for open surgery when you've got something that is so well established and the complications just so low, and it's cheaper when you look at their theater capacity, et cetera.

So, I think there has to be a move towards endovenous, but there are lots of things stopping this. In Germany, for example, the reimbursement is greatest for open surgery. So surprise, surprise, there are still lots and lots of open varicose vein operations. For trainees, it is important to recognize that if you want to change practice, the single biggest driver for change is reimbursement. You can change whatever you like. So, I've treated maybe three patients with open surgery in the last five years. One was a GP, who I had treated the other leg with open surgery a few years earlier, and she said, "I want the same operation, please." So I did struggle to argue with that. The other two were big three, four, five centimeters saphenovarices in the groin. So specific indications, but not really for most patients.

For deep venous reflux, surgical management follows a relatively complex algorithm. (O'Donnell et al. 2014) Valve prolapse due to vein wall dilation may be treated with external banding. Isolated valve prolapse may be treated with external valvuloplasty, and in severe cases an internal valvuloplasty. When no adequate valve exists, then valve transplant from a distant

source, such as the axilla, may be necessary. (Kabbani et al. 2011) These are not commonly performed procedures and should be referred to centers with significant experience.

18.1.3.3 Medical Management

So we've talked about endovenous and surgical treatment, but are there any pharmacological therapies that can be used in patients with chronic venous insufficiency?

So we're pretty skeptical in the UK, when it comes to venoactive or pharmacological treatments. If you go to Europe, they love them. They're over the counter in a lot of pharmacies. The drug with probably the most evidence is something called Daflon, micronized purified flavonoid fraction.

So flavonoids are naturally occurring, venoactive compounds. There are a whole variety of effects, only some of which we understand. But a lot of studies have shown improvements in edema and heaviness. Daflon is marketed by a French company and it's got good evidence, but they do not see places like the UK as a big enough market because of our inherent skepticism over these sort of things, to make it worth their while to go through the marketing and the regulatory processes. So it's not available. But patients that have gone to France and picked it up over the counter and have said it works well. So there may be a roll for some venoactive medications such as pentoxyphyllin —there's some good evidence that it accelerates healing of leg ulcers.

But I don't think we're very receptive in the UK in general, not in the vascular surgery community to prescribe pharmacological or herbal type remedies.

Wound management in patients with venous insufficiency however has many medical adjuncts. Topical agents include excimer (cadexomer) iodine, silver sulfadiazine, silvercontaining dressings, growth factors or cell-based therapy.

What are the healthcare costs associated with the lower extremity venous disease?

The educated guesstimates in the UK, NHS, is around 2 billion pounds a year. The UK NHS budget is about a 120 billion. So, up to about 2% of the budget, and that's just C6 disease. We're not even going into all of the other diseases, the associated cellulitis, but also not only the healthcare costs, but the societal costs, the time off work, the other things. And then of course the superficial vein thrombosis, VTE and other associated diseases, it's a massive underestimate.

18.1.4 Other Complications of Superficial Venous Disease

18.1.4.1 Superficial Vein Thrombosis

You mentioned superficial vein thrombosis, also known as superficial thrombophlebitis. What is it and how do you manage it?

I'm very pleased you used both terms because I'm going to take the opportunity to encourage robustly the use of **superficial vein thrombosis** rather than thrombophlebitis. The problem with thrombophlebitis is it is too closely associated with an infectious etiology, which can lead to useless courses of antibiotics. Of course, there's no infection, it's a thrombus in a superficial vein. Traditionally, it's just been seen as just a bit of a nuisance and let's not worry about it, treat with analgesia and anti-inflammatories, and you'll be fine.

It can be a really sinister pathology and associated with VTE. If you think of how many thousands, millions of people that have varicose veins. They don't all get clots in their varicose veins. They've all got stasis to a certain extent but they don't get clots. So it's more than just the flow dynamics going on here. These people likely have an thrombogenic innate tendency that leads to this. That's shown out in the studies, some really good work in some French studies, such as the POST study. They identified patients who were scanned with superficial vein thrombosis, and a quarter of them on their first duplex scan had a DVT. A lot of these DVTs were remote to the superficial vein thrombosis.(Decousus et al. 2010) So they're not all extending into the deep vein—they were remote—confirming this idea that actually a thrombogenic problem is going on here. A patient with superficial vein thrombus and a previous history of DVT could suggest a subsequent risk of VTE up to 20%.

So in the latest ESVS guidelines, there is quite a lot of content about SVT treatment with algorithms. I would urge people to have a look at that, and it's important to risk stratify people. Limited disease can be followed with ultrasound at 1-2 weeks, but the closer you are to the junction, the higher VTE risk, and actually for anyone other than just a bit of clot in a varicosity, anticoagulation is the treatment of choice to reduce progression to VTE.(Kakkos et al. 2021; Di Nisio, Wichers, and Middeldorp 2018; Scovell, Ergul, and Conrad 2018)

NSAIDs and Warm Compresses	Anticoagulation	
Low risk Patients	High risk patients	
Segment less than 5cm	Segment greater than 5cm	
Remote from SFJ	Less than 5cm from SFJ	

CALISTO is the largest RCT and used treatment with 2.5mg SC Fondaparinux.(C. S. Group et al. 2010; Blondon et al. 2012) Rivaroxaban did subsequently demonstrate non-inferiority.(Werth et al. 2016) So look at the guidelines, it's very clear what we should be doing with these people, evidence-based, Once the anticoagulation is done, they need to be re-scanned and that residual incompetent, scarred thrombogenic, saphenous vein needs

to be ablated or excised, otherwise they'll recur. So a real change in mindset and level of aggression.(Kakkos et al. 2021; Di Nisio, Wichers, and Middeldorp 2018; Scovell, Ergul, and Conrad 2018)

Pregnant patients are often high risk for lower extremity venous complications and pose difficult clinical challenges. Extensive superficial vein thrombus in pregnant patients should be treated with LMWH and paused at delivery. (Kupelian and Huda 2007; Litzendorf and Satiani 2011) If symptoms persist 3-6mo after delivery, the further workup for pelvic congestive syndrome may be required with a transvaginal ultrasound or cross sectional imaging. (Labropoulos et al. 2017)

18.1.4.2 Bleeding

What's your approach to managing patients with bleeding varicose veins?

Really important question. It is a vascular emergency. There are several depressing reports of patients having died from bleeding from a varicosity. These are often elderly patients who are frail, with limited mobility, and therefore can't bend down to press on the bleeding varicosity down by the ankle ulcer. There's some very grim photos in various case reports and it's really sad. There's two or three photos that I've seen where an elderly patient has not wanted to wake their partner and so has gone and laid down in the bath with this bleeding vein that they couldn't control and basically exanguinated.

So, it's a vascular emergency and these patients should be seen and assessed and their superficial venous disease should be dealt with as soon as possible. We have an emergency clinic running and anyone with bleeding veins have the same urgency as the CLI patients and they were treated there and then.

In terms of the treatment, there's two aspects: you want to decompress the venous hypertension by ablating the saphenous reflux, but for the specific bleeding area, I think there's a real case for some local foam sclerotherapy. You want to block off that vulnerable bleeding vein to ensure that this doesn't happen again. But again, if anyone receives a referral, these are people to see straight away.

18.2 Acute Deep Venous Thrombosis (DVT)

So let's move on to discuss deep venous thrombosis. You're recently involved in the European society of vascular surgery, venous thrombosis guidelines. In the most part, calf DVTs are managed by medical teams or dedicated DVT services within the hospital. So in this section, we'll be focusing mostly on proximal DVT.

18.2.1 Demographics

18.2.1.1 Physiology

In medical school, we've learned a lot about Virchow's triad and the etiology of venous thrombosis. Does that simple concept still hold true today?

I think it does. There's a few caveats and a few nuances, but I think the principle that the flow, the vessel wall, and the blood constituents are the main factors that affect whether or not you get thrombosis in a vein is still pretty solid. However, there is also a real change in mindset about the whole idea of provoked and unprovoked DVT. Again, I would urge people to look at some of the new areas that we've covered in these guidelines.

18.2.1.2 Prevalence

Incidence of DVT increases with age by a factor of 1.9 per 10 years. Of patients who develop DVT, 1% will have phlegmasia and 10% may develop a pulmonary embolism.

For post-thrombotic syndrome, 17% will develop at 1 year and 29% by 8 years after the initial DVT. For recurrence, 30% will develop over 10 years but the highest risk for recurrence is in the first year.(Beckman et al. 2010; C. M. Bulger, Jacobs, and Patel 2004; Meissner et al. 2007)

18.2.1.3 Etiology

You mentioned there about potential risk factors for DVT, and that can be classified into these provoked, which can be transient or persistent, and then unprovoked risk factors. Can you talk about some of the specific factors and causes?

In general, what's been happening for several years is a recognition that there are some DVTs that happened with a clear provoking factor—a common major, transient, provoking factor is major surgery, for example. So a lot of DVT are caused with that clear provoking factor, but there's an enormous population of people that don't have a clear provoking factor at all, or have a minor factor that may or may not have contributed.

So traditionally there's been a dichotomy between provoked and unprovoked. If it's provoked, they get a limited period of anticoagulation is unprovoked it's long-term anticoagulation. And certainly the last big thrombosis guidelines was the ACCP guidelines, the American CHEST guidelines, and that's what they suggested, but it's much more nuanced than that.

What we've got to remember is that there are some provoked DVTs where there's much higher risk, and those are there's some unprovoked DVTs, which actually there may be other factors that you need to take into account. So I think for those people that are interested in this area, it's really important to get to the nitty-gritty. So, long-winded way of answering your question. But here are some risk factors:

- Major provokes
 - Surgery
 - Major trauma
- Minor provoked
 - Shorter periods of immobility a few days sick in bed with the flu
 - Contraceptive pill depending on whether it was just started or the patient has been taking for two to three years

In the guideline document is a big, long list of other things where for people to think about it and consider.

In this review we are discussing mostly lower extremity DVTs, however catheter-associated DVTs are sometimes encountered on examinations. In general the DVT is considered to be a result of the line. However, management is treatment of the DVT with or without line removal. If the line is functioning well and the patient requires the line for other therapy, then simply treating the DVT with anticoagulation should be sufficient.(Kearon, Akl, Ornelas, Blaivas, Jimenez, Bounameaux, Huisman, King, Morris, Sood, and al. 2016; Kovacs et al. 2007) Treatment is similar in patients with cancer.(Debourdeau et al. 2009)

Prevention is a key component to mitigating the impacts of DVT. One of the more common risk stratification tools in the United States is the Caprini risk assessment model.(Bahl et al. 2010) Patients identified to be high risk according to the Caprini risk index require mechanical prophylaxis, pharmacological prophylaxis and early ambulation.(Gould et al. 2012; Laryea and Champagne 2013)

In those patients where, after you've taken a thorough history, there's no obvious provoking factors. Is there a role for screening for occult malignancies and thrombophilias?

For thrombophilia testing, the history of it is really interesting. When they were first identified and our ability to screen them first became available, there's a lot of excitement because people thought they were going to be able to identify a clear cause for all of these unprovoked DVTs. The reality is that at least 50% of them, there's no identifiable thrombophilia. So what it almost certainly means is there's lots of thrombophilias that we haven't yet identified, we can't test for. So it hasn't really been the panacea in that regard. The other pragmatic reality for the thrombophilias is that if we're going to be putting people on long-term anticoagulation anyway, the additional value of testing for these things is really very minimal. And the ones that are really important—the thrombosis history is so stark that they normally smack you in the face. So I think it's generally gone out of fashion. And again, there's only very specific circumstances where we would suggest testing them:

- 1. If you're going to be trying to stop anticoagulation, for some reason, the patients don't want to be on it, or if there's bleeding risks
- 2. There's a clear family history. Then it might be worth excluding some of the more sinister thrombophilias.

Otherwise, certainly not a role for routine testing and similarly for malignancy—for unprovoked DVT, the prevalence of an underlying malignancy is somewhere between 6-10%. The studies that have looked at an aggressive, thorough assessment process rather than a more selective process, have picked up a few more cancers, but the additional cost in terms of anxiety, additional investigations, et cetera, and no demonstrable improvement in outcome has meant that the guidelines have not recommended routine cancer screening beyond the appropriate, symptom driven age and sex specific screening—PSA, breast assessments, colonoscopy, etc.

One unique situation is recurrent or migratory superficial vein thrombosis, particularly in older patients, can be associated with an underlying malignancy.(Litzendorf and Satiani 2011)

For patients with established diagnosis of thrombophilia (Factor V Leiden homozygous mutation, antithrombin deficiency, prothrombin G20201A homozygous) who become pregnant, recommendations state they should be on LMWH prophylaxis during pregnancy and 6w postpartum regardless of personal history of DVT.(Heit et al. 2005)

18.2.2 Evaluation

What is the algorithm for investigation?

The most important part is the first component of the algorithm, which is an assessment of the pre-test probability of DVT. So anyone who comes with a DVT, you can go through a validated tool. The most common is the Wells probability score, which is a series of questions and give you an indication as to whether a DVT is likely or unlikely. (Tritschler et al. 2018)

If a DVT is likely, then the algorithm is to do a scan, very straightforward. If a DVT is unlikely, then the algorithm is to do a D-dimer. Because if that is negative, DVT is effectively excluded. The guidelines are very clear on the assessment process, but a lot of this won't reach our radars as vascular specialists at all until the DVT has been diagnosed, so we don't appreciate the importance of the early stages. So again, a quick plug for the guidelines, there's a very nice algorithm, beautiful colors—I definitely recommend having a look.

On ultrasound, acute DVT is characterized by an enlarged vein with a non-compressible lumen and hypoechoic thrombus. Chronic post-thrombotic scarring is often contracted with echogenic appearance. (Yusof et al. 2019) For more details, see Section 20.7.

18.2.3 Management

18.2.3.1 Medical

So the exact choice and duration of anticoagulation in the management of a DVT is probably beyond the scope of this podcast, but on the whole are DOACs preferred over vitamin K antagonists, such as Warfarin and why?

In a word, yes, lots and lots of studies, funded by big companies that make DOACs, have been done and they've all shown fairly consistent things. They've shown that the DOACs are as effective as the vitamin K antagonists at preventing DVT. They have almost always a significantly lower bleeding risk, but they've got this enormous added convenience of not needing regular blood tests, not having this dietary limitations, and the erratic control, et cetera. So the general direction of travel is inexorably towards more and more use of DOACs.

A couple of other areas where DOACs have really shined: The tendency is to recommend longer courses of anticoagulation and often indefinite anticoagulation. A number of studies, both rivaroxaban and Apixaban have looked at the effect of a lower dose of DOAC in these patients. So if you need to extend anticoagulation, can you use a lower dose, a prophylactic dose, so 10mg of rivaroxaban or 2.5mg BID of Apixaban. The results have been impressive because the prevention of recurrent VTE is very good, the same as the therapeutic dose, but the bleeding risk is much lower. So you've got this additional benefit of being able to give an even load those with even lower bleeding risk.

And the real icing on the cake for the DOAC has been a few recent studies that have been looking at cancer patients. Traditionally cancer patients where low-molecular weight heparin only, not for DOACs.(Kearon, Akl, Ornelas, Blaivas, Jimenez, Bounameaux, Huisman, King, Morris, Sood, and al. 2016) But the Hokusai study(Raskob et al. 2018) and the Caravaggio study(Agnelli et al. 2020) more recently have both looked at Edoxaban and Apixaban in the context of cancer and have shown that actually they're very effective. In addition, a recent meta-analysis found that recurrence rates were lower with DOACs, but bleeding rates were no different.(Dong et al. 2019; Rossel et al. 2019)

In almost all of the areas, DOACs are demonstrating their superiority. The one very notable exception is antiphospholipid syndrome, particularly triple positive antiphospholipid syndrome. This is a pretty nasty acquired thrombophilia where Warfarin is still the gold standard.(Pengo et al. 2018)

To step back a little, trainees should make sure they understand some basics behind the coagulation pathway and anticoagulants. Warfarin is one of the most studied and longest used anticoagulants and is a Vitamin K antagonist. Vit K is essential for activation of factors IX, X, VII, prothrombin (II), C and S, which are integral to the extrinsic pathway of coagulation.(Fair, Marlar, and Levin 1986; Wu, Morris, and Stafford 1991) The extrinsic pathway is activated by vessel wall damage and exposure of tissue thromboplastin or tissue factor (TF), which binds Factor VIIa activating Factor X and Xa.(Morrissey 2001)

Warfarin induced skin necrosis is important to understand, although very rare (0.01-0.1%). This occurs shortly after initiating warfarin and presents with purple, cool, painful toes and punctate areas of necrosis and petechiae. Treatment includes immediate cessation of warfarin and transition to LMWH, normalization of INR with Vit K and FFP.(Pourdeyhimi and Bullard 2014; H. A. Tran et al. 2013)

Reversal agents are also another tested topic in the medical management of vascular disease. Other anticoagulants and their reversal agents include:

- Dabigitran Idarucizumab.(Glund et al. 2019)
- Edoxaban, apixiban and rivaroxaban and exanet alfa. (Connolly et al. 2019)
- Nonspecific with some effect on other DOACs PPC, Factor 7, and activated PPC.(Cuker et al. 2019; Kearon, Akl, Ornelas, Blaivas, Jimenez, Bounameaux, Huisman, King, Morris, Sood, and al. 2016)

Finally it is also good to be aware of a few additional coagulation factors and their function:

- Integrin glycoprotein GPIa/IIa platelet aggregation
- Protease-activated receptors (PARs) platelet aggregation
- Thrombomodulin activation of protein C
- Plasminogen activator inhibitor-1 suppression of plasmin formation/fibrinolysis.

Patients with a proximal DVT, in the iliofemoral or the femoropopliteal segment, is there a role for elastic compression in the acute phase?

Traditionally people have been a bit nervous about putting compression on patients with acute DVT, concerns about causing propagation and causing PEs. A very important study was the ideal DVT study(Ten Cate-Hoek et al. 2018), run by Dutch colleagues, and a number of other additional studies more recently that have built on the original IDEAL DVT study.(Schreurs et al. 2022)

The short answer to your question is that early compression is really important. It's a really positive thing to do within 24 hours and it's proper compression. It's not just the Ted stocking and see what happens. It's 30 to 40 mmHg, so that in practice is a class two stocking with a bandage on top. The benefits are that this reduces the early pain and swelling of the DVT, but also interestingly, it reduces the risk of post-thrombotic syndrome and post-thrombotic venous obstruction. So if you put compression on early, it reduces the risk of them developing post-thrombotic syndrome. So a lot of focus has been on clearing the vein and early thrombus removal and thrombolysis and thrombectomy and all this sort of aggressive stuff. But just by doing the compression, we can have at least part of the same effect and we've just not been doing it. Very rarely do these patients get proper compression.

So there is a big section in the ESVS guidelines about the role of compression and algorithm as to when, how to use it, and how long do you use it for. What we want to see now is the DVT pathways around the country, around the world, updated and amended with this latest guidance.(Kakkos et al. 2021)

18.2.3.1.1 Complications - Pulmonary Embolism

More and more vascular surgeons are becoming involved with *Pulmonary Embolism Re*sponse Teams (PERT) and so surgeons should be familiar with the acute management of DVTs that progress to pulmonary embolism. The majority of pulmonary embolisms simply require anticoagulation.

However, if a patient becomes hemodynamically unstable, such as develops hypotension, then the patient should undergo systemic thrombolysis.(Kearon, Akl, Ornelas, Blaivas, Jimenez, Bounameaux, Huisman, King, Morris, Sood, and al. 2016) Contraindications to systemic thrombolysis include:

- Age >75
- Intracranial neoplasm
- Intracranial/spinal surgery or trauma with the last 2 months
- History of hemorrhagic stroke
- Active bleeding or known bleeding disorder
- Non-hemorrhagic stroke within the past 3 months.

Patients who do not immediately respond to systemic thrombolysis, are a high risk of bleeding, or in shock that may cause death before thrombolytics take effect, then catheter based thrombus removal may be considered. This complex decision making is best made through the previously mentioned PERT.(Eleftherios S. Xenos et al. 2019) Endovascular techniques include:

- Thrombus fragmentation
- Suction thrombectomy
- Rotational embolectomy
- Ultrasound assisted thrombolysis

There are very high risk procedures and complications include pulmonary artery perforation, pericardial tamponade, cardiogenic shock, hemaglobinuria or other hemorrhagic complications.

18.2.3.1.2 Complications - Post Thrombotic Syndrome

One of the concerning complications of a DVT, particularly a proximal DVT is post-thrombotic syndrome (PTS). Can you recap the symptoms of PTS and how it's diagnosed?

PTS is a fairly amorphous entity that includes a series of patient symptoms and clinical signs that develop after deep vein thrombosis. That's deliberately very vague, because the condition is so variable. In terms of diagnosing PTS, the official diagnostic tool is a score of five or more on the Villalta scale. The Villalta score is another one of these tools that have been developed for both making a diagnosis, but also assessing progression of PTS.(Kahn et al. 2014; Villalta et al. 1994)

But in terms of sort of symptoms—people get swelling, heaviness, pain. There is a very unique symptom, which is venous claudication, which is almost pathonogmonic of post-thrombotic syndrome with venous outflow obstruction. It's this pain which has a bursting tense quality in the calf on exertion, which does settle on stopping, but it takes a lot longer than arterial claudication. So sometimes people will take half an hour to get better and the leg has to be elevated. So that's venous claudication and that's often the most disabling symptom. But of course, people can develop the same clinical complications than you can with any venous disease—skin changes, lipodermatosclerosis, venous leg ulceration, et cetera. Those are the summery of symptoms for PTS, seen in up to 50% but probably closer to 25% of proximal DVT.

18.2.3.1.3 Complications - IVC Filters

A quick question about IVC filters. As vascular surgeons, we sometimes get asked about the appropriateness of inserting an IVC filter. When would you consider one and how long can one stay in situ for?

So short answer is I virtually never consider one, and if already in situ, remove as soon as possible. Let's expand on this a little bit. So the history of IVC filters is a painful story, particularly in the US, super aggressive use of IVC filters, very few of them retrieved, lots of IVC occlusions, and major lawsuits against big companies. The most important factor in successful IVC filter retrieval is dwell time less than 7 months. Type of filter, tilting or protruding struts are not associated with failure. Adverse events have found to be as high as 2%.(Desai et al. 2017) That's led to a worldwide, reluctance to use IVC filter, so now we may not be using them in some cases when we should be using them. The traditional indications and reasons for using IVC filters included weird and wonderful things like floating thrombus and other factors.

The reality is the only real indication that we could come up with was when there's a proximal DVT, and there's a clear contraindication to anticoagulation. So if there's no anticoagulation, then the risk of propagation and PE is reasonably high, so there is a strong case for a temporary IVC filter. If a patient is anticoagulated, very rarely can the case be made for an IVC filter.

I don't want to go into nuanced details—such as where somebody already had a PE there's right heart strain, there may be some anatomical features that you're worried about, their anticoagulant control is borderline—there might be specific cases, but this is really very rare. Most people anticoagulated do not need an IVC filter.

A rather challenging and unique patient would be a pregnant patient with a DVT and placenta previa. Anticoagulation is contraindicated in placenta previa and in addition, an infrarenal IVC filter is also contraindicated, so she would require a suprarenal IVC filter.(Bates et al. 2018; DeYoung and Minocha 2016; Harris, Velineni, and Davies 2016)

In the setting of a thrombosed filter, a patient may require urgent intervention if they develop phlegmasia. Suprarenal filter placement may be useful during the procedure to prevent embolization. Removal of the IVC filter is contraindicated unless all clot is cleared. (Sheth et al. 2015; Teter et al. 2019)

18.2.3.2 Surgery

So clot burden reduction techniques are a hot topic at the moment. Before we talk about endovenous techniques, is there a role for surgical venous thrombectomy? And when would you consider this?

Open thrombectomy is part of the spectrum of early thrombus removal techniques. It used to be very popular, a very satisfying technique—I think I've done it three times. They've all been for the same sort of indication, which is phlegmasia, severe limb threatening post DVT change, and in a patient where they weren't candidates for thrombolysis. However, if you're with a patient where the leg is threatened, you can't anticoagulate them, you can't thrombolize them. Then in that case this is a sort of last ditch approach. But if that's not the case, then almost always a catheter based approach is less invasive and more acceptable.

In addition, most people have got more training. Trainees should read some of the descriptions of open thrombectomy— it's a very satisfying technique, but it's quite involved. It involves finding and exposing the common femoral, deep femoral and femoral veins, applying tight compression around the lower leg to milk the clots out distally to proximally. So the original descriptions are really interesting, but really very rarely would we need to do that.(Comerota and Aziz 2019)

18.2.3.3 Endovascular

So you've mentioned catheter directed techniques for clot burden reduction. Some recent trials have challenged our thinking on the appropriateness of these techniques, such as the ATTRACT and the CaVent trial. What's your take on the current evidence? This is very controversial. The fundamental issue here is that the trials that have been done almost certainly would not have included a large number of people who would benefit from interventions in the opinion of the investigators. So if somebody has got severe DVT, it is unlikely, particularly in the US, where the ATTRACT trial was done, but these people would have been randomized into the trial.

So the immediate criticism of the trials is that this is not reflective of the population. You've excluded the people that benefit, so it's an underestimate of the benefit of the technique. But if we look at trial results ourselves, there is a modest benefit. In the CaVent trial, PTS rate was reduced. The quality of life was less impressive, but the PTS rate was reduced moderately.(Haig et al. 2016) And in the ATTRACT trial, when they subgroup analyzed the iliofemoral DVT group, then there was a reduction in the number of people who develop moderate and severe PTS. So I think the numbers are something along the lines of, you'd have to treat 10 people with thrombus removal to prevent one person developing moderate to severe PTS.(Vedantham et al. 2017)

So then the discussion is are those numbers needed to treat appropriate for intervention? Particularly when you then start to consider the additional need for stenting and long-term surveillance, et cetera. So I think the jury is very much out. You've got really stark, strong, firmly held unshiftable views on both sides of the argument and what we need is a bit of nuance. What we need is everyone accepts that some people benefit let's all work together, try and work out, which people benefit. Let's work on case selection.

So we'll put you on the spot then, when would you offer thrombolysis? Which patients would you offer it in and which circumstances would you also stent?

There's no role for thrombolysis in anything other than the iliofemoral segment. So if it's infrainguinal, proximal DVT, even if its severely symptomatic, the studies that actually you cause harm by attempting early thrombus removal. That's the first important point.

Even if they have got severe symptoms, there's always a role for initial conservative management, compression with elevation and a lot of these people have a dramatic improvement in their symptoms. If they don't, and there's an honest discussion, and the bleeding risk is low then potentially there is a role. But when there's a treatment with a borderline benefit or the benefit is unclear, then I think what we've gotta do pick low risk patients, so low bleeding risk. We've got to choose teams and techniques with the highest risk of technical success. And we've got to pick people where we think the benefit is greatest, i.e. those that have the highest risk of PTS.

Whether or not you stent is very difficult. We had a case not so long ago where we had a very nice result from early thrombus removal, we stented, and the patients thrombosed again early and subsequently was identified to have triple positive antiphospholipid syndrome. Obviously there wasn't the opportunity in the time to make that diagnosis prior to the decisions around the thrombus removal, so you end up stung and now this patient has a stent and can be very difficult to open up again. The advocates and the aggressive stenters would say, you can't leave a significant stenosis after thrombus removal, otherwise they will rethrombose. Whereas
others will say, if we go back to the early studies, like CaVent, very few, if any, people actually had stents, but actually their results were still reasonable.

I think once you get on the roller coaster of thrombus removal, it's very difficult not to then proceed also to stenting, if you see a significant lesion, for example, at the May-Thurner point.(W. Wang et al. 2018; Meissner et al. 2012; Rollo et al. 2017) Final comment, I'll make is that the use of recanalization and stenting for chronic PTS is a very effective technique. So another approach, if you're not sure, is to manage them conservatively, but always have that up your sleeve, if they get significant PTS at six months, no earlier, then we could potentially consider recanalizing and stenting at that point.

Following thrombolysis, what duration of anticoagulation is required?

So the whole point of anticoagulation after DVT is to reduce the risk of propagation and pulmonary embolus, firstly, but also then to reduce the risk of recurrent VTE. The aims don't really change if you thrombolyzed or if you put a stent in. Some people used to say, I put a stent in, we've treated the May-Thurner and that was the main cause of DVT, so we can stop the anticoagulation. That's not the case. So it shouldn't change and it shouldn't be any shorter if you've thrombolize and put a stent in.

Now, whether it should be longer because you put a stent in that's debatable. Some people have very aggressive anticoagulation protocols after stenting, but that's to do with the stent rather than the DVT. So remember the aims, which are to reduce the risk of recurrent DVT and PE and to stop propagation of the current event.

18.3 Phlegmasia

Some patients may present with very profound, lower limb swelling. Pain and cyanosis with a threat to viability of the limb. Can you explain for us what phlegmasia is and an approach to treatment?

There's two sorts phlegmasia that are described, but to be honest, I don't think anybody has a detailed and robust understanding of what these actually mean. Phelgmasia alba dolens is the white leg, cerulea dolens is the blue leg and the severe, form a phlegmasia. Theoretically, cerulea dolens has a limb loss rate of 30-40%–severe limb threat, compartment syndrome, skin blistering, swelling, etc. The reality is that it's often over-diagnosed, so you'll see a lot of people saying I did venous thrombolysis for phlegmasia and it's just a big swollen engorged leg.

The reason it happens is almost certainly a profound obstruction of venous outflow. So the venous blood in the leg cannot escape. You get this cycle of worsening pressure, leaking of fluid from the capillaries, increasing pressure on the soft tissues. Then that starts to threaten tissue perfusion and potentially if it's bad enough this will threaten the actual arterial profusion of the leg as well. Presentation often includes iliofemoral DVT with non-palpable pulses and

loss of sensation with out improvement on anticoagulation and elevation. Endovascular early thrombus removal is the first line with fasciotomies, with open thrombectomy as back up for salvage.(Erdoes et al. 2011; Thomas, Hollingsworth, and Mofidi 2019)

So that's the sort of most severe and extreme form. I don't think anybody has confidently linked the anatomical findings and features of a DVT to what the leg looks like. There was a patient I saw who literally had every deep vein acutely thrombosed from the IVC down to both lower legs and was almost asymptomatic. So we've got this very uncomfortable lack of understanding between the symptoms and the clinical and anatomical pattern of DVT.

18.4 Deep Venous Obstruction

18.4.1 Etiology

So finally, let's move on to talk about the treatment of proximal deep venous insufficiency, both obstruction and incompetence. What are the common causes for deep venous obstruction? Thinking about post-thrombotic and non-thrombotic obstructions.

So again, the terminology is important. So it's venous outflow obstruction. Obstruction implies that there's a physiological issue causing the consequences. As opposed to occlusion, which is an anatomical term. So venous outflow obstruction, is essentially anything that reduces the ability of venous blood to escape the leg via deep venous channels will cause an obstruction.

And again, going back to a bit of physiology. Poiseuille's Law talks about the flow rate being proportional to the fourth power of the radius. So it's really interesting. A question to ask is that if you've got a two centimeter vein, how many, one centimeter veins would it take to carry the same potential flow as a single two centimeter vein?

The long story short is that it takes 16 one centimeter veins to carry the same amounts of potential flow as a single two centimeter vein. So it doesn't take much of a narrowing to have a great potential impact on flow. Of course, most of the time, you don't need that flow potential, it's only an absolute extremes of circulatory stress that you'll need that sort of potential, but it is interesting. And again, it also challenges when people say there's great collaterals and that will be enough. Well unless they're enormous, you need 16 of them to have the same potential blood carrying capacity.

So in terms of the causes, the most common by far is post-thrombotic disease. So when you've had a DVT, one of three things happens. If you're lucky, the vein will recanalize completely. If you're unlucky, it will obstruct completely. And if you're somewhere in between, you'll end up with partial stenosis and obstruction. What's important is even if the vein is open, what you almost always lose in the post-thrombotic vein is venous compliance. It's not the same compliant vein that can respond to demands and physiological need that you have, in a primary non-scarred vein.

Other potential causes are a non-thrombotic iliac vein lesion, or *May-Thurner lesion*. So the original paper, 1957, is very interesting reading.(May and Thurner 1957) There were a couple of Austrian pathologists who identified that a very high proportion of people, particularly those had left-sided DVT had, not only the overriding iliac artery causing compression, but they had a secondary tight fibrosis in the iliac vein.(Nazzal et al. 2015) So it's not just the compression because any one of us in a certain position we'll have a squashed vein. It's the fact that there's a secondary inflammatory fibrotic change, which is what's causing the obstruction. So that's a true May-Thurner or non-thrombotic lesion.(Liddell and Evans 2018; Rollo et al. 2017)

The other big group of patients of course, is malignant lymph node or retroperitoneal fibrosis related venous outflow obstruction. For example, a patient with a large retroperitoneal leiomyosarcoma may be involving the vena cava. In this case, these patients may require en bloc resection of the IVC and reconstruction with ringed PTFE, sized off the preoperative imaging.(Quinones-Baldrich et al. 2012; Fiore et al. 2012) But in the end, post-thrombotic by far was the most commonly diagnosed pathology.

18.4.2 Evaluation

We may see patients who come in with severe venous disease with skin changes, for example. When would you consider imaging the proximal deep veins in addition to infrainguinal superficial and deep veins of the leg and what are the best modalities to do this?

So we are really getting down to the importance of a detailed assessment of the entire superficial and deep venous system. Every patient with venous disease should have a whole leg infrainguinal duplex to start with—that's deep and superficial veins. There's no real role for just looking at superficial veins. So you need to understand the context. In terms of if their obstructed or if they're refluxing. Again, remember that, reflux on a duplex, yes, all we see is arrows going in certain directions. But all reflux means is that after a calf augmentation maneuver, there is some retrograde flow—in a superficial vein, more than half a second and in a deep vein, more than one second. That's all it means. So for example, you can have retrograde flow for 0.6 seconds in a saphenous vein, and that will be labeled as reflux for all the rest of the time that superficial vein is doing the right job, it's returning the blood to the heart. So just be wary of interpreting what these arrows mean. For more, see Section 20.7.

In terms of how that affects your decision-making—one approach is to consider the overall burden of venous disease, such as those with venous hypertension. And what I try and do is draw a pie chart and say, you got venous hypertension, here are the different causes of your venous disease—that might include superficial reflux, deep venous reflux, deep venous obstruction, and other factors. You try and apply some clinical judgment as to what the different factors are. Now, even if they've got significant, deep venous reflux and superficial venous reflux, there is still a role for treating superficial reflux in most of these cases. I still come across colleagues and other people who say, look, you shouldn't treat the superficial veins if there's deep reflux. But all the studies—EVRA, ESCHAR-have shown significant benefits, even if there's deep flux.(Manjit S. Gohel et al. 2007; Manjit S. Gohel et al. 2018) So again, the key message is treat the superficial reflux if it's there. However, if the patient has isolated deep reflux then management is limited to wound care and compression, often as high as 40-50mmHg. Some series have found that 70-80% will heal.(O'Donnell et al. 2014; Maleti and Perrin 2011)

Now, deep venous obstruction and looking for it is a different matter. There are again, lots of approaches, some people are very aggressive about evaluating the iliac veins—CTVs and IVUS on everybody. The problem is if you get super aggressive, then you start to over-diagnose and if you start to over-diagnose, you start to over treat and this all gets very messy.

I think the sort of people that have clear signs and symptoms of major venous outflow obstruction, particularly venous claudication, history of previous DVT, really disproportionate venous changes to the superficial and the infrainguinal imaging. Proximal obstruction can present with atypical proximal varicose veins in the mons public, perineum, buttock, high posterior/lateral thigh, abdominal wall. These findings are 93% predictive of deep venous obstruction proximal to the inguinal ligament.(Kurstjens et al. 2016) Those are the people that I would image.

And on the duplex imaging of the leg, if there are changes in the common femoral vein that make you think there is a proximal problem, particularly a loss of phasicity of flow then that's pretty concerning for a significant venous outflow problem. In an unobstructed venous system, the common femoral vein should really reflect the right hearts, which has some phasicity. There should be transmission of that respiratory phasicity. So if that isn't there, then you're thinking, hang on, there's something interrupting that transmission, which is almost always an obstruction somewhere. So those are the people I'd image, but I wouldn't over image—because then you end up with a headache and a problem—Should I treat or should I not treat?

So you mentioned IVUS, what is intravascular ultrasound and what are the particular benefits of using this technique over venography alone?

IVUS is a 9F probe that is inserted into the vein and it gives you a 360 degree view in Bmode ultrasound—traditionally without color flow, just B mode—of the anatomical structures. What it's very useful for is identifying venous anatomy, confluences, identifying the size of veins, so it allows you to plan sizes of stents. It really is an important adjunct to venography and other imaging modalities. We did our first 30 or so cases of deep venous stenting without IVUS and thought we were brilliant and IVUS was a waste of money. And then we started using IVUS and realized I can't believe we were doing this without. Once you start using it, you suddenly realize that it's a really useful adjunct. *Recommended oversizing for vein stenting is 10-15% of largest diameter of adjacent vein.* (Gloviczki et al. 2011; O'Donnell et al. 2014; Raju, Owen, and Neglen 2002; Raju, Darcey, and Neglén 2010)

In terms of when you use it, in some parts of the world it's commonly used for diagnostic purposes, but beware of the risk of over-diagnosis. More commonly, almost routinely, IVUS

is used for therapeutic, particularly when we're stenting deep veins to plan our landing zones. When CTA/MRA shows compression, it must be confirmed with IVUS prior to treatment. IVUS has far greater sensitivity and specificity and must show a cross-sectional surface area reduction of greater than 50%.(Gagne et al. 2018; Forauer et al. 2002)

You talked about over-diagnosis. Can you talk us through some of the decisionmaking processes that you go through when you're assessing a patient who has a deep venous obstruction?

There are lots of patients that have this anatomical change, the number of referrals I get from people who have done a CT and they've picked up a problem. The most important thing is the clinical status and the clinical picture. So have they got symptoms—go about right at the beginning of the podcast—have they got symptoms? Have they got complications? If they haven't got significant symptoms and they haven't got significant complications then there's not really an indication for treatments.

So leg swelling was initially, frequently pushed as, oh you're swelling will get better, if we get rid of your obstruction. That's only the case in probably less than 50% of patients. The reason for that is that, yes the original etiology for the swelling may well have been venous, but you almost always get a secondary lymphatic dysfunction when you've had swelling, for whatever reason—whether it's heart failure, whether it's venous, the lymphatic system gets damaged. And so even if you get rid of the original cause, the lymphatic damage persists, of course, and the swelling persists.

The symptoms are very important and in terms of complications, if somebody has intractable ulceration, et cetera, these are all important indications. But a bit of skin change, a bit of mild swelling, careful case selection is very important for these procedures.

18.4.3 Management

18.4.3.1 Endovascular

There is variability in the availability of stents for venous obstruction. However, closed cell braided stents (i.e. the Wallstent) have been around for many years and has the most data related to long term durability.(Gagne et al. 2019) These stents have the least radial force at the ends, so it is recommended to extent 1-2cm beyond the venous obstruction. In cases of proximal iliac vein obstruction, this may require extending into the IVC 1-2cm which may cross, or "jail" the contralateral iliac vein. When jailed, the rate of DVT on that side has been reported as high as 10%.(E. H. Murphy et al. 2017; Le et al. 2018)

So following a deep venous recanalization and stenting, what surveillance do you offer for these patients?

Surveillance has been evolving. The first availability CE (UK version of FDA) marked deep venous stents in the UK was around 2012. A few services across the country really got started

at a few centers in 2012, 2013. Initially there was an early scan and then a scan maybe at six weeks and three or six months down the line.

I think we realized that in general, the people that developed problems, often develop them very early. So we are scanning these people the day after the intervention—obviously there's intraoperative quality control and venograms and scanning—but the day after the intervention to ensure that there's no early thrombus problems. Then we scan a couple of weeks after to make sure that there's nothing that's happened early, because there's an opportunity to re-intervene, to balloon, to correct any technical issues, at that point. After that point, it's usually six weeks, three months, six months, and then annually thereafter.

So that's a sort of protocol that we have and in general, what we've seen is that people that develop problems tend to develop them relatively early. So once you get to a year, then in general, the patency rates are very good. In our center, the the chronic PTS stenting primary patency rates at around three years are over 80%-85%. Of course, there are some that have problems, but it's important to remember that, even those that have thrombosed, they generally go back to the level of symptoms they have prior to intervention. Which is very reassuring. The last thing you want to do is to make people worse with a novel intervention.

And so what about your anticoagulation and antiplatelet regime for these patients?

There was a systematic review done a few years ago where all of the published literature for deep venous stenting was reviewed and I think the authors identified 28 different regimens for anticoagulation of the stenting.(Notten, Ten Cate, and Ten Cate-Hoek 2021) The honest answer is nobody knows, our approach is to have low molecular weight heparin as the primary anticoagulant for the initial two week period. And then at the two week point, they undergo a stent. If there's no issues, they get transitioned to a DOAC.

In addition, the patients get six weeks of clopidogrel as an anti-platelet. And again, the reason for that is that it's a foreign material into the vein. Yes, the role of anti-platelets in preventing venous thrombosis is very limited but there has to be some platelet related activity and platelet activation involved there. So it makes logical sense to have some anti-platelet activity. Of course the downside is that there's an increasing bleeding risk. These are the constant tightropes that you're walking with these patients.

💡 Take a Listen

Check out our episode with Drs. David Gordon and David Williams on a recent publication looking at the evolution of in stent restenosis and the classification of thrombus forming in venous stents.

18.4.3.2 Surgery

We've talked about endovascular and endovenous treatments, but is there a role for open surgery in these patients?

So in the iliac segment, very rarely. In some patients, you can do PTFE reconstructions and bypasses. And of course the traditional operations of Palma-Dale(Palma and Esperon 1960) bypass and May-Husni(Shaydakov, Porembskaya, and Geroulakos 2015) type procedures. They all sounded very elegant in the books, but the reality is a great saphenous vein going across the lower abdomen to the other side is not really going to sort the problem out in a durable way for all sorts of obvious reasons—the size of the vein, the pressure and all that sort of stuff. So there's really very limited role for those sorts of procedures.

Now, the one area where there may be a role for open reconstruction is in the common femoral vein. The consistent observation is that good inflow is absolutely imperative to get good outcomes in these patients. And sometimes if the common femoral vein has got a lot of scarring, then it's very difficult to reconstruct that inflow endovascularly. So a number of authors have advocated the use of endophlebectomy, opening up the common femoral vein, and cutting out the scarring and the tribulations, putting a patch on it. Sometimes even putting an AV fistula—a little six millimeter graph from the common femoral artery onto the vein to really drive the flow through the stent. It's a pretty complicated operation and you can imagine you're dissecting down and there's lots of little veins everywhere and lots of lymphatics. The wound complication rate is very high.

So really very limited role only—the endovascular approach is really the way forward for these patients.

18.5 Pelvic Congestive Syndrome

18.5.1 Presentation

So finally, let's talk about pelvic congestion syndrome. What is pelvic congestion syndrome and how does it present?

So it often doesn't present anywhere near a vascular surgeon. There are a number of women who present with these chronic heaviness and dragging symptoms in the lower abdomen. There may be some association with menses, there's often dyspareunia and a whole variety of other symptoms.

There may be associated vulvar varices or posterior buttock and thigh varices all as a result of pelvic venous incompetence. Knowing the proximal anatomy of the saphenous vein tributaries is important, as the pudendal vein tributary connects vulvar veins and inferior gluteal vein connects to the posterior buttock. (Iafrati and Donnell 2019) But it's a complicated field—there are some people that have pelvic symptoms, there are some people that have the varicose veins

or some will have a combination of the two. Almost always they will be seen and initially assessed by the gynecology teams rather than coming to the vascular surgeons.

18.5.2 Evaluation and Management

When we do see these patients, are there any treatment options available for them? And when would you consider treating?

In a highly selective way—this is venous disease in a nutshell—the association between anatomical changes and clinical symptoms is ropey at best. In terms of postpartum women, an enormous proportion of these women have got venous reflux in the pelvis—if you look hard enough. so the problem is similar to that of May-Thurner syndrome, if you start looking hard enough, you start identifying these symptoms and then you lose the ability to really discern who's going to benefit most.

So again, I would go back to clinical symptoms as the single most important thing. If they've got severe clinical symptoms majorly affecting their quality of life and associated significant venous changes, then those people to try to intervene on. It's not necessarily a benign procedure. Pelvic vein embolization involves implantation of coils. There's concerns about non-targeted embolization as coils have ended up in lungs. There are a number of people with unknown nickel allergies that have problems with these coils. So again, it's not an area to be over aggressive. So careful case selection is important. So again, a quick plug for the ESVS guidelines. There is a very good section on pelvic venous disease assessment. (De Maeseneer et al. 2022)

What's very important is again, to differentiate between pelvic congestion syndrome and pelvic origin varicosities. One thing that is certainly being advocated is that if the varicose veins are coming from the pelvis, but there's no pelvic congestion syndrome, it's perfectly reasonable just to deal with the visible varicosities from below—with some foam sclerotherapy or other treatment from below, rather than being super aggressive embolizing from above. But I think anybody who's involved in managing this really needs to work closely with the gynecologists. There are gynecological diagnoses that need excluding before we start to address the venous disease. And there are a number of validated questionnaires and other assessment tools that are probably quite important. So not for the faint-hearted.

18.6 Superior Vena Cava (SVC) Syndrome

SVC syndrome is an acute presentation of head and neck venous congestion often due to SVC obstruction. This can be intrinsic due to indwelling devices or extrinsic due to malignancy or mediastinal fibrosis (half of benign cases).(Kalra, Bjarnason, and Gloviczki 2019; Rice, Rodriguez, and Light 2006; Parish et al. 1981)

18.6.1 Management

Angioplasty and stenting are first line treatment for benign etiology. (Rizvi et al. 2008; Sheikh et al. 2005) Angioplasty alone may be sufficient, but stenosis refractory to angioplasty may require stenting. SVC syndrome with failed endovascular management in reasonable operative candidates can undergo open reconstruction with large diameter prosthetic graft or spiral vein graft. (Doty, Flores, and Doty 1999) However, restenosis in open repair can be as high as 15%. (Kalra, Sen, and Gloviczki 2018; Sfyroeras et al. 2017)

SVC syndrome secondary to malignancy, bronchogenic or lymphoma, is often most effectively treated with palliative radiotherapy with good response. (Talapatra et al. 2016) Endovascular or open recannalization is often reserved only for those with severe symptoms refractory to radiation or chemotherapy. (Higdon and Higdon 2006) Lymphedema pumps can help with symptom management as well.

18.7 Lymphedema

Here we will discuss lymphedema of the lower extremity. This is a broad topic, but the management decisions are relatively straightforward and knowledge is required for examinations. For upper extremity lymphedema, see Section 3.8

18.7.1 Demographics

18.7.1.1 Etiology

Lymphedema can be thought of as a primary dysfunction of the lymphatic channels or the secondary result of a disease that results in the destruction of lymphatic flow. Causes of secondary lymphedema include filariasis, which is endemic to parts of South America.(Pfarr et al. 2009; Shenoy 2008)

18.7.1.2 Presentation

The management of lymphedema is determined primarily on its clinical stage at presentation.(Grada and Phillips 2017) The staging for lymphedema is as follows:

Lymphedema Stage	Description	Symptoms
$\begin{array}{c} 0 \\ 1 \end{array}$	Sub-clinical Spontaneous Reversible	No swelling, heaviness or discomfort Swelling improved with limb elevation

Lymphedema Stage	Description	Symptoms
2	Spontaneous Irreversible	Pitting and swelling, not improved with elevation
3	Lymphostatic elephantiasis	Skin hardening, non-pitting edema, verrucas changes, recurrent infection

18.7.2 Evaluation

On physical exam, a classic finding is Stemmer's sign, which is a thickened skin fold at the based of the second toe/finger. The sign is positive when the tissue is hardened and cannot be lifted.(S. G. Rockson et al. 1998; Stanley G. Rockson 2019)

Lymphedema is most often diagnosed based on clinical presentation and physical exam. Lymphoscintography can be used to confirm the diagnosis in complex presentations and would demonstrate dermal back flow, absent or delayed transport, crossover filling, absent or delayed visualization of lymph beds.(Stanley G. Rockson 2019; O'Donnell, Rasmussen, and Sevick-Muraca 2017)

18.7.3 Management

Treatment is best initially managed with combined decongestive therapy (CDT) which includes an intensive reductive phase (4-8wks), followed by life long maintenance, often with specific compression garments.(Grada and Phillips 2017) CDT in the setting of heart failure has been reported to lead to volume overload.(Lawenda, Mondry, and Johnstone 2009)

In extreme cases, after CDT therapy has been instituted for at least 6 months and the patient has persistent refractory symptoms, surgery may be used for reconstruction or debulking.(Grada and Phillips 2017; Baumeister et al. 2016)

19 Trauma - Peripheral

Authors: Kevin Kniery, Jason Bingham, Nakia Sarad, and Todd Rasmussen

These trauma episodes were developed in collaboration with Behind the Knife: The Premier Surgery Podcast.

DO NOT GET DISTRACTED

- These patients are often poly-trauma patients with likely other injuries
- Follow ABCDE of the primary survey
 - Identify bleed and hold pressure to stop the bleeding
 - Pressure can be manual pressure, such as placing a finger in the wound, or applying a tourniquet
 - Once bleeding is effectively controlled, RESUSCITATE and continue the primary survey to address other life threatening injuries

Vascular trauma requires close collaboration between vascular surgeons and trauma surgeons. The decision of which specialty should manage which injuries varies across different centers. Through these trauma chapters we will discuss what management decisions the majority of vascular surgeons should be comfortable managing as a part of a multidisciplinary trauma team.

💡 Take a Listen

Check our debate between leading trauma and vascular surgeons about how to best develop a collaborative team to manage vascular trauma.

For relevant images and a more in depth discussion of this topic, please review Chapter 21: Upper Extremity and Junctional Zone Injuries (Vuoncino, White, and Clouse 2022) and Chapter 22: Lower Extremity Injuries (David S. Kauvar and Propper 2022) in Dr. Rasmussen's 4th Edition of *Rich's Vascular Trauma*.

19.1 Demographics

19.1.1 Epidemiology

Penetrating extremity trauma is leading cause of peripheral vascular injuries (75-80%)

Most common arterial injury found in femoral or poplite al arteries (50-60%), then brachial artery (30%)

19.1.2 Types of Injuries

Can occur with either blunt or penetrating trauma

- 1. Complete Wall Defect (hemorrhage or pseudoaneurysm)
- 2. Intimal Injury (subintimal/intramural hematomas, flaps, disruptions)
- 3. Arteriovenous fistulas
- 4. Complete Transection (with hemorrhage or occlusion)
- 5. Spasm

19.1.3 Presentation

Hard Signs of Peripheral Vascular Injury

- Active Hemorrhage
 - Obvious bleeding (arterial or venous)
 - Can be pulsatile or history of bleeding in field
- Profound Ischemia of the extremity
 - Determined by absence of palpable pulse or dopplerable signal beyond the area of injury
- Expanding hematoma
 - Can occur underneath a closed wound/injury
- Audible Bruit or Palpable thrill near site of injury
 - Concern for an arterial-venous fistula, often high flow

Soft Signs of Peripheral Vascular Injury

• Subjective reduced or unequal pulses

- Diminished blood flow to extremity
- Weak, but audible arterial doppler signal
- Large non-pulsatile hematoma
- Orthopedic injury patterns carrying a high index of suspicion of vascular injury
 - Examples posterior knee dislocation, displaced long bone fractures
- Penetrating wound in proximity to a major axial vessel
- Peripheral Nerve Injuries
 - Femoral nerve injury
 - Paresthesia to anterior thigh
 - Decreased hip flexion and knee extension
 - Sciatic nerve injury
 - Paresthesia to lateral leg and dorsal/lateral/plantar foot
 - Weakness of foot plantar flexion
 - Tibial nerve injury
 - Paresthesia over heel
 - Weakness of plantar flexion
 - Deep peroneal nerve injury
 - Paresthesia of first digital interspace
 - Foot drop (E. M. Bulger et al. 2014)

19.2 Evaluation

Continuous Wave Doppler

- Handheld device that can amplify arterial signal to measure quality of pulse
 - Triphasic, biphasic, monophasic, or absent
- Injured Extremity Index(Vuoncino, White, and Clouse 2022)
 - Utilizes continuous wave doppler with manual blood pressure cuff to obtain objective measurements of perfusion

- On injured extremity, inflate cuff proximal to arterial signal and measure pressure at which arterial signal diminishes. Compare to a normal/non-injured extremity. Injury extremity index is the ratio between these two values.
 - Injured Extremity Index > 0.9, no flow limiting vascular injury Recommend observation
 - Injured Extremity Index <0.9, flow limiting vascular injury likely Recommend further imaging with CTA
- If patient in shock and cold, repeat injury extremity index as the patient is resuscitated and warmed

CT angiography (CTA)

- Gold standard for evaluation of extremity vascular injury
- Ensure appropriate timing of contrast to follow-through to distal extremity
 - Can occur in conjunction with CT imaging of head/chest/abdomen/other extremities to evaluate for traumatic injury

Invasive angiography

• Has been readily replaced by CTA for evaluation of extremity vascular injury

19.3 Management

If patient has hard sign of vascular injury, OR for exploration and vascular repair/bypass

If patient has suspicion of vascular injury (soft signs), recommend bedside injury extremity index.

If injured extremity index > 0.9, recommend observation.

If injured extremity index < 0.9, recommend further imaging with CTA or angiography (if patient stable). If CTA positive for occlusion or extravasation, OR for operative repair.

19.3.1 General Considerations

Exposure Basics

- Prep patient widely
 - Any lower extremity injury, prep patient from umbilicus down to both legs
 - Anticipate need for proximal control, intraoperative angiograms at the femoral level, and potential harvesting of saphenous vein for conduit
- Recommend leaving tourniquet on and prep into field
 - Removing tourniquet prematurely can result in arterial bleeding, leading to hypotension, hemorrhagic shock, and multi-organ system failure
 - Communicate with an esthesia and OR team when to urniquet will be put down and have blood ready for transfusion
- Incision
 - Recommend making incision twice as long as what you would initially anticipate
 - Important to have long incision to get appropriate control and visualization of injury
- Junctional Hemorrhages
 - Junctional area between the torso and extremity, most difficult to control
 - Lower extremity vessels include distal external iliac and common femoral
 - Commonly presents as a groin hematoma
 - Upper extremity vessels include subclavian and axillary

💡 Take a Look

You can review surgical exposure of lower extremity arteries here. You can review surgical exposure of upper extremity arteries here.

Repair

- Gain proximal and distal control
- Debride injury back to healthy tissue
- Perform thrombectomy
 - Both proximal and distal to area of injury using Fogarty catheter to completely clear the inflow/outflow thrombus

- Once thrombus burden removed, apply 40-60cc of regional heparinized saline down the outflow before applying the clamp. Perform again on the inflow.
- Types
 - Primary Repair
 - Good for focal injuries (grazing wounds or stab wounds from a knife) where there is minimal tension and arterial lumen maintains its integrity after debridement to healthy tissue
 - Can consider with venous injuries due to higher compliance or when vessels are redundant (ie brachial artery)
 - Patch Repair
 - Also good for focal injuries not amenable to primary repair
 - Again, can perform if there is no concern for tension or compromise of arterial lumen
 - Interposition graft
 - Most common repair type, avoids issue of tension and compromise of arterial lumen
 - Conduit types
 - Autologous vein (great saphenous vein)
 - Preferred choice with concern for contamination and for more distal extremity injuries (i.e. brachial, radial, SFA, and popliteal)
 - Saphenous vein from contralateral extremity (non-injured) routinely recommended
 - Ipsilateral saphenous vein CAN be used if there is no deep vein injury and it is part of the incision/exposure
 - Prosthetic ePTFE or Dacron
 - Suitable if vein is not available, wound with minimal contamination or injury is more proximal (axillary, subclavian, proximal femoral, iliac)
 - Ligation
 - Dependent on location of injury and overall hemodynamic status of patient
 - If patient unstable, ligate as damage control maneuver (Liang et al. 2016; Fox et al. 2012)

💡 Take a Listen

Check out our Vascular Origins Story outlining the history of arterial injury management in military medicine and the transition from ligation to arterial repair during the Korean war.

Heparinization

- Recommend selective heparinization
- If patient has isolated extremity injury, recommend systemic heparin during revascularization
- If patient has concomitant injuries (i.e. head trauma, torso bleeding, or extensive soft tissue injury), recommend regional heparin flushed up and down injured vessel (Liang et al. 2016; Fox et al. 2012)

Shunts

- Can be utilized in damage control maneuvers to prevent need for ligation and achieve temporary perfusion
- Larger caliber vessels with high flow can be shunted without need for heparinization for 4-6 hours
- Smaller caliber vessels (in distal extremities) usually clot off but do not cause harm when clot occurs intra-operatively as the vessel needs to be re-assessed for repair
 - However, recommend that shunts are used in patients that do not have contraindications to systemic heparin use
- Types
 - Javid
 - Sundt
 - Argyle
 - Has advantage of multiple sizes
 - Pruitt-Inahara
 - Makeshift shunts small caliber chest tubes
 - Can place in injured vessel proximally and controlled with either ligature around vessel to lock shunt in place or rubber vessel loop
- Placement

- Insert in injured vessel proximally first, then allow some pulsatile bleeding to allow any thrombus to exit before inserting it distally
- Shunts can be secured either with heavy silk tie or plastic vessel loops
- Can use doppler with water or acoustic gel over shunt to verify arterial flow
- More information about the types and use of intravascular shunts can be found here.

Fasciotomies

- Avoid hesitation to perform 4-compartment fasciotomies in threatened limb ischemia
- Recommend prophylactic fasciotomy to avoid compartment syndrome
- Especially important when transporting patients to higher level of care facilities
- Two-incision Four Compartment Lower Extremity Fasciotomy can reviewed here (Bowyer 2015)
 - Reference points for lateral incision here.
 - Reference points for medial incision here.
- Comprehensive review of fasciotomy techniques in both upper and lower extremities can be reviewed here.

19.3.2 Iliac Vessels

Exposure

- Retroperitoneal exposure preferred oblique incision in lower quadrant of abdomen (i.e. transplant incision)
- Midline laparotomy can also be considered
- Groin hematomas can either be iliac or femoral etiology, recommend extensive prep with either transplant incision or midline laparotomy incision to be able to gain common iliac artery control if needed

Repair

- Internal iliac injury
 - Prefer ligation
 - Difficult access for exposure and repair

- Patients usually in hemorrhagic shock and control is time-consuming and difficult to achieve
- Can be safely achieved, pelvic ischemia rare due to collateral blood supply from contralateral iliac vessel
- If repair can be performed, recommend larger needle (SH blunt tip needle on 3-0 or 4-0) for repair due to low visibility in pelvis
- External iliac injury
 - Repair at all costs
 - Gain control first then decide on method of repair depending on patient stability (immediate repair or temporary repair with vascular shunt)
 - 14 French Argyle vascular shunts can be used on iliac vessels
 - May need to use prosthetic interposition graft due to larger caliber of vessels
 - Ligation has high morbidity and mortality, should be avoided
 - If external iliac artery is ligated, patient will experience proximal limb ischemia and will either need an above knee amputation (AKA) or hip disarticulation
 - In life or death situations, ligation and subsequent amputation is inevitable to save a patient's life
 - Can consider temporary ligation with return to OR in 2-3 hours to re-establish inline flow (consider cross femoral graft) when patient is more resuscitated, however, not common in trauma situation

19.3.3 Femoral Vessels

Includes the Common Femoral Artery (CFA), Superficial Femoral Artery (SFA), and the Deep Femoral (Profunda femoris)

Exposure

- Proximal CFA exposure similar to exposure of external iliac (midline laparotomy or transplant incision)
- Distal CFA, SFA, and Deep Femoral can be exposed with vertical groin incision (below inguinal ligament midway between publis and ASIS over anterior proximal thigh) or oblique groin incision (directly beneath inguinal ligament or few centimeters distal)

Repair

• Gain control

- May be difficult to only use vessel clamps for more proximal injuries
- At common femoral junction, can use Fogarty catheter into orifice of deep femoral with balloon inflated to control back bleeding
- If SFA injured, can clamp distal CFA for control
- Recommend shunt placement
- Repair with primary, patch, or interposition graft (autologous vein preferred)
- Ligation also has high morbidity and mortality, should be avoided (see iliac section for consequences of ligation and options of shunting) (Abou Ali et al. 2017; Subramanian et al. 2008)

19.3.4 Popliteal Artery

Exposure

- Need to gain control of popliteal artery above and below knee
 - Above knee popliteal artery is a continuation of the distal SFA through the adductor magnus (Hunter's canal)
- Position patient in "frog-leg" with bump of rolled towels placed underneath calf
- Make medial incision from above the knee to mid-thigh
 - Locate the distal SFA just at or beyond Hunter's canal in order to locate the above knee popliteal artery
- Move bump to above the knee and make a medial incision from below knee to mid calf
 - Expose the gastrocnemius and soleus muscles and bring down to the inferior edge of the tibia to open the below the knee popliteal space
- Retraction and good lighting is key
 - Can use Weitlaner retractor, Pilling retractor, or Henly popliteal retractor to retract muscles and better visualize space
 - Recommend narrow handheld retractor (appendiceal or Wylie renal vein retractor) on either the superior or inferior aspect of below or above knee incision, respectively, to gain more exposure of the popliteal space

Repair

• Recommend bypass repair (autologous vein preferred if available)

- Inflow from distal SFA to distal below knee popliteal target
- Always check repair with intraoperative doppler and check distal flow to foot
 - If signal goes down and you are concerned about the technical repair, be prepared to re-open and pass thrombectomy catheters proximally and distally to evaluate issue
- Even with good technical repair, spasm may occur
 - Common among young patients
 - Signal usually improves after re-warming and resuscitation
 - Ensure that there is audible flow in the bypass and patient well resuscitated prior to re-opening injury to assess repair
 - On-table angiogram also an option, however, spasm may appear as occlusion and repair is unnecessarily re-opened
 - Important to note: Water hammer signal is not appropriate audible flow as it indicates diastolic flow in the arterial signal

Take a Listen

Check out our discussion with Mr. Paul Blair where he discusses management of peripheral arterial trauma during the troubles in Northern Ireland. This episode was developed in collaboration with the Rouleaux Club.

19.3.5 Tibial Vessels

Repair is selective on extent of injury

Recommend repair if all three tibial vessels are injured and there is no flow to the foot (anterior tibial, posterior tibial, and peroneal arteries) (Burkhardt et al. 2010)

• Saphenous vein graft preferred

Recommend ligation if one or two tibial vessels injured, as there is redundant arterial flow to the lower extremity

Take a Look

Find additional resources for lower artery exposure here.)

19.3.6 Subclavian and Proximal Axillary Arteries

Important to keep in mind, upper extremity disability can have a significant impact on quality of life.

Long term functional impairment is measured by the Disabilities of Arm, Should, Hand (DASH) questionnaire. Impairment most closely associated with duration and severity of ischemia and concomitant nerve injury (36%). Open reconstruction is preferred and patency rates are high.(Frech et al. 2016)

Due to protected anatomic location of subclavian vessels, most subclavian artery trauma is usually cause by penetrating mechanism.

Considered a junctional region for hemorrhage with difficult access for exposure

Endovascular approach with covered stents may be used in traumatic axillo-subclavian injuries that are stable enough for intervention (i.e. do not require a resuscitative thoracotomy) (Eleftherios S. Xenos et al. 2003; Branco et al. 2016)

Exposure

- Requires thoracotomy or sternotomy associated with clavicular incision for access
- Right subclavian/axillary arteries
 - Median sternotomy with extension to supraclavicular incision
 - Divide sternocleidomastoid muscle 1 cm above clavicle to expose subclavian vein
 - Retract vein and divide anterior scalene muscle to expose subclavian artery
 - Careful to locate and avoid phrenic nerve (traverse laterally over anterior scalene)
 - Aberrant right subclavian artery directly arises from thoracic aorta (0.5-2%) of population
- Left subclavian/axillary arteries
 - Left posterolateral thoracotomy is most optimal, however, requires lateral recumbent positioning
 - If patient is requiring midline laparotomy for operative management of abdominal trauma, recommend anterolateral thoracotomy in 3rd or 4th intercostal space
 - Extension with supraclavicular incision or median sternotomy to better expose proximal subclavian
 - Extension with infraclavicular incision to deltopectoral groove to gain access to proximal axillary artery

Repair

• Primary, patch repair, interposition graft, or bypass dependent on extent of injury after debridement to healthy tissue and follows tenets of avoiding tension and preserving arterial lumen integrity

19.3.7 Distal Axillary and Brachial Arteries

Brachial artery is continuation of distal axillary artery

Repair

- Primary Repair can be considered in brachial artery if there is redundancy, lack of tension, injury is focal, and no compromise to arterial lumen.
- Great saphenous vein preferred over prosthetic for conduit.
- Ligation should be avoided as upper extremity limb ischemia is highly disabling and morbid

19.3.8 Radial and Ulnar Artery

Repair

- Ligation can be considered if there is appropriate collateralization from either the radial or ulnar artery into the distal palmar arch
- If either radial or ulnar injury causes distal hand ischemia, should be repaired

19.4 Venous Injury

Venous injuries can be selectively repaired. More proximal venous injuries should be repaired, while distal injuries can be ligated.

Primary repair with lateral venorrhaphy can be performed due to higher intrinsic compliance of veins.

19.4.1 Lower Extremity Venous Injury

Proximal Vessels (popliteal and femoral veins)

- Ligate if patient unstable, acceptable as damage-control maneuver
- Repair if patient is stable and not in profound shock
 - Can consider venous shunts as well
- Studies confirm that venous repair improves arterial flow in the arterial repair since venous outflow is not impeded (Quan et al. 2008; Clouse et al. 2007; Todd E. Rasmussen and Tai 2022)
- Even if repair of injured venous vessels thrombose over time, thrombosis occurs slowly and does not lead to pulmonary emboli

19.4.2 Upper Extremity Venous Injury

All EXCEPT proximal axillary and subclavian vein injuries can be ligated in the upper extremity (ie distal axillary, basilic, cephalic, brachial, veins).

Repair can be performed primarily or with graft (autologous or synthetic).

Part VI

General

20 Vascular Lab

Authors: Alaska Pendleton and Anahita Dua

Acknowledgements: A special thanks to Drena Root, Technical Director at the Massachusetts General Hospital Vascular Center, without whom this imaging would not have been possible.

So let's start with an overview of ultrasound modalities, waveforms, and changes proximal and distal to flow-limiting stenosis.

20.1 Overview

Brief overview of ultrasound: Ultrasonography uses sound waves with frequencies higher than those audible to humans. Ultrasound images are created by sending pulses of ultrasound into tissue using a probe. Reflected pulses are recorded and displayed as an image. There are many types of ultrasound images, but two modes commonly seen in vascular ultrasonography are B-mode and doppler:

B-mode (Brightness) imaging is a 2-D, black and white display of tissue acoustic impedances.

Doppler mode uses the Doppler effect to measure and visualize blood flow.

Spectral Doppler converts frequency shifts from moving blood to velocities using the Doppler equation and displays a "spectrum" of these frequencies as Doppler waveforms.

Color Doppler presents velocity information as a color-coded overlay on top of a B-mode image. Duplex ultrasonography is a term commonly used for the simultaneous presentation of B-mode and doppler data. Of note, when looking at a still image of a color flow duplex, the color may be misleading for direction because it may not be noted at which part of the cardiac cycle the picture was taken. It is far more reliable to look at the spectral doppler waveform.

Power Doppler is less frequently used, but is based solely on amplitude of the doppler signal without giving information on direction. This is particularly useful to detect certain low or abnormal flow states, such as testicular/ovarian torsion, carotid string sign, slow intrarenal flow, supporting occlusion seen on B-mode or color.(Pellerito and Polak 2019)

20.1.1 Waveforms

What do normal spectral waveforms look like?

<u>Normal spectral waveforms</u> have a brisk upstroke, sharp peak, rapid downstroke. A "spectral window" under the waveform, that is the black space between the spectral waveform and the 0 velocity axis, represents the absence of lower velocities - indicative of laminar flow within the vessel (Image 1: normal spectrum waveform)



How can we differentiate low vs high resistance waveforms?

Waveform profiles change depending upon the nature of the distal vascular bed being supplied. Organs like the brain, kidneys, liver, spleen, peripheral muscle during exercise, and postprandial SMA have constant high metabolic demand, and are therefore low resistance vascular beds. Low resistance waveforms for arteries supplying these organs demonstrate constant forward flow throughout the cardiac cycle because the distal bed being supplied has low resistance leading to high end-diastolic flow.



In contrast, <u>high resistance waveforms</u> are seen for arteries supplying resting peripheral muscles, fasted mesenteric beds (such as the fasting SMA), and the external carotid artery. High resistance waveforms are characterized by triphasic morphology, with a sharp peak, early diastolic flow reversal, brief forward flow (elastic recoil of the artery, and then no flow during the remainder of the diastolic phase.



How does flow-limiting stenosis change the waveform?

First let's define <u>Stenosis</u>: A hemodynamically significant stenosis (area reduction >50%) will result in a doubling of velocity from the inflow segment to the area of maximal stenosis (velocity ratio >2).

Now what do waveforms look like AFTER a flow-limiting stenosis?

<u>Tardus et parvus</u> refers to a pattern of Doppler ultrasound spectral waveform resulting from arterial stenosis. The tardus et parvus waveform is delayed with prolonged systolic acceleration (tardus) and diminished with a small systolic amplitude and rounded systolic peak (parvus). This phenomenon is observed downstream from the site of stenosis. Tardus parvus in the CFA represents upstream (iliac) stenosis. Tardus parvus in the brachial artery likely represents upstream (subclavian or axillary) stenosis.(Hwang 2017; Pellerito and Polak 2019)



So what will the waveform look like BEFORE a flow-limiting stenosis?

Distal stenosis: Distal occlusive disease will result in a high resistance waveform, with absent diastolic flow. (Image 5: Distal stenosis with absent diastolic flow)



Are there any considerations we need to understand when assessing a stented vessel with ultrasound?

Stenting decreases vessel compliance and increases observed velocities. There is no defined consensus, but in-stent restenosis is classified by some as follows:

• >50%

- SFA Vr > 1.5 or PSV >200 (Baril et al. 2009; Kawarada et al. 2013)
- ICA ICA/CCA >2.7 and PSV >220.(Lal et al. 2008)
- >80%
 - SFA Vr > 3.5 or PSV >275 (Baril et al. 2009; Kawarada et al. 2013)
 - ICA ICA/CCA > 4.15 and PSV > 340.(Lal et al. 2008)

It should be noted that restenosis of stents vs vein grafts in the lower extremity will have higher velocity for the same percentage of stenosis. Tardus et parvus waveforms distal to a stent should raise concern that stenosis is hemodynamically significant. (Baril et al. 2009; Kawarada et al. 2013)

Ultrasound compared to other diagnostic modalities

Duplex is often adequate to visualize any peripheral arterial bed, even the aorta and iliacs.(Muela Méndez et al. 2018)

- CT limitations include ionizing radiation, iodinated contrast and is not appropriate in patients with CKD.
- MRA is performed with gadolinium-based contrast and is not appropriate in patients with ESRD.

Now that we've covered the basics, let's move through by organ-system high-yield vascular lab studies, findings, and pathologies. We'll start with the extracranial evaluation, a highly-tested area of vascular ultrasonography.

20.2 Extracranial

What does a typical extracranial evaluation involve?

Examine CCA (2 views), ICA (2 views), ECA, vertebral arteries

What are normal ICA and ECA waveforms?

Normal ECA vs ICA waveform: The external carotid artery waveform reflects a high resistance vascular bed. This means minimal diastolic flow. Conversely, the ICA waveform reflects a low resistance vascular bed with antegrade diastolic flow.



This makes sense, as the ICA is supplying the brain while the ECA is supplying the face. Intuitively, the common carotid artery is a mixture of the ICA and ECA waveform morphologies - like the ICA with forward flow throughout diastole, but less as compared to the ICA due to the high-resistance influence of the ECA.

Another way of differentiating the external and internal carotid arteries is the "temporal tap". Tapping on the superficial temporal artery (a branch of the ECA) will be transmitted as small pulsations in the diastolic component of the external carotid artery. (Pellerito and Polak 2019; Size et al. 2013)



Can you talk about diagnostic criteria for ICA stenosis?

Parameters for ICA stenosis: These are a few numbers that are (unfortunately) essential to memorize for the VSITE and RPVI.

Although criteria differ between guidelines, the Carotid Consensus Criteria, define ICA stenosis >= 70% as a peak systolic velocity >= 230 cm/sec, EDV > 100 cm/sec, and ICA/CCA ratio > 4.0. Of note, post-stenting criteria vary from pre-stenting criteria. Stenosis criteria are not clearly defined for the CCA or ECA.



When is surgery indicated for ICA stenosis?

Parameters for when surgery indicated:

Asymptomatic Carotid Atherosclerosis Study (ACAS) ${>}60\%$ stenosis asymptomatic, NASCET ${>}50\%$ stenosis symptomatic

For more see Chapter 1.

Let's talk about other pathologies that can be visualized on extracranial ultrasound:

Pathologies: Stenosis (plaque), dissection (flap), aneurysms (rare), occlusion (no flow, do not operate), carotid body tumor (splaying of ECA/ICA, fed by ECA branches), FMD.

FMD is frequently encountered on the VSITE/RPVI. How would this appear on the exams?

Fibromuscular dysplasia of the internal carotid arteries affects women more commonly than men. Duplex findings show a "chain of lakes" appearance, demonstrative of multiple septa and small aneurysms. Velocity elevations and increased turbulence in the waveform patterns is typically found on Doppler interrogation. (Jeffrey W. Olin et al. 2012)



Aortic stenosis: Tardus parvus waveforms in both common carotid arteries can be a sign of more proximal disease of the aortic valve or global myocardial dysfunction.

How to treat?

Aspirin if asymptomatic, POBA if symptomatic.

Are there other frequently tested pathologies demonstrated on extracranial exam?

<u>Subclavian steal</u>: Subclavian steal occurs when a proximal subclavian stenosis or occlusion leads to reversal of vertebral artery flow. This causes "stealing" of blood from the posterior cerebral circulation, and presents as vertebrobasilar insufficiency.

How does this look on duplex?

Normal vertebral flow looks very similar to ICA: antegrade low resistance waveforms with constant forward flow throughout the cardiac cycle. As subclavian stenosis progresses, one can see mid-systolic velocity deceleration ('bunny ears"), with mild stenosis bidirectional flow can signify "pre-steal" phenomena, and with severe steal, there is a complete reversal of flow in the vertebral artery towards the arm rather than towards the brain.(Kalaria et al. 2005; Mousa et al. 2017)



<u>Innominate stenosis</u>: A phenomenon that is related to this, is innominate stenosis. Here again the patient will present with vertebrobasilar insufficiency, indicative of diminished vertebral antegrade flow, but additionally will experience right hemispheric insufficiency secondary to diminished R ICA antegrade flow. The right-side duplex will demonstrate flow reversal in the vertebral artery, abnormal waveforms in the subclavian, as well as steal pattern waveforms in the common and internal carotid arteries. The common denominator for all of these findings is significant disease in the innominate artery.

What other extracranial arteries of the head and neck can be evaluated separately?

<u>Temporal arteritis</u> - well visualized with high frequency transducer (10 MHz or greater). Pathognomonic halo effect of anechoic/hypoechoic edematous tissue surrounding the lumen due to concentric inflammation.(Ball et al. 2010; W. A. Schmidt 2014)

20.3 Intracranial

So that covers extracranial vascular lab evaluation. What about intracranial?

This is less frequently tested, so we will discuss just a brief overview of views and some of the more commonly tested pathologies related to transcranial doppler (TCD).

Intracranial - Arteries differentiated by depth. MCA 3-6 cm, everything else deeper. Three primary views:
- **Temporal view:** Used to interrogate PCA, ACA, MCA, and ICA. The MCA, ICA and PCA flow direction is towards the probe, the ACA flow direction is away.
- Occipital (Foraminal) view: Basilar and vertebral arteries (both away).
- Orbital view: Ophthalmic and ICA

What are some frequently tested pathologies that are identified on TCD?

Indications often tested:

- MCA spasm (severe PSV>200): Can be seen in sickle cell disease with studies indicating a strong correlation between mean velocities of >200cm/s and the rate of stroke in children with sickle cell disease. With blood transfusions, stroke risk can be reduced from >10% to <1% per year. (Bulas et al. 2000)
 - Lindegrad ratio is the MCA velocity divided by the distal ICA velocity and can also be used to assess MCA spasm.
 - Ratio of 3-6 is indicative of mild-mod vasospasm.
 - >6 is severe vasospasm.
 - <3 is a sign of hyperemia.(Kirsch et al. 2013)
- <u>Cerebral ischemia during CEA</u>: Comparing transcranial Doppler sonography, nearinfrared spectroscopy, stump pressure measurement, and somatosensory evoked potentials, cerebral ischemia was most accurately predicted by the percent change in transcranial Doppler detected middle cerebral artery velocity. Detection of a greater than 50% drop in middle cerebral artery velocity using transcranial Doppler is 100% sensitive for detecting cerebral ischemia.(Moritz et al. 2007a)
 - TCD can also demonstrate microemboli (high spikes of white vertical lines on spectral doppler) during CEA.
- A shower of bubbles in both cerebral hemispheres upon injection with agitated saline is consistent with a R to L cardiac shunt and patent foramen ovale.(Zito et al. 2009)
- Elevated contralateral ICA velocities can demonstrate compensatory flow and collateralization and can be a sign of persistent MCA occlusion.
- Mechanical Compression of the vertebral arteries: TCD will show normal posterior circulation velocities at rest, disappearance of waveform on provocative maneuvers, and increase in velocity once back in a neutral position representing a reactive hyperemia. (Vilela et al. 2005)

- Vasomotor activity can be evaluated as well with TCD. Hypercapnia or breath-holding should result in vasodilation and increased flow. Hypocapnia will result in decreased flow. No change in response to hyper or hypocapnia can show poor cerebral autoregulation.(M. Müller et al. 1995)
- Reversal of flow in the ophthalmic artery can be a sign of ipsilateral ICA occlusion.(Guan et al. 2013)
- Brain death determination In patients with cerebral arrest (brain death) TCD will demonstrate a low amplitude, high-resistance waveform with to and fro flow with zero net flow.

20.4 Upper Extremity

Having covered head and neck vasculature, let's move on to peripheral vasculature. This is a huge area both on the VSITE/RPVI and in practice. In this section we'll cover first the upper, then the lower extremity vasculature.

So first, what are characteristics of waveforms in the peripheral vasculature?

<u>Peripheral</u>: Normal waveforms are indicative of high resistance distal beds, so we would expect triphasic waveforms

What are normal arterial parameters in the upper extremities?

Normal pressure gradient between the right and left brachial pressures is <20 mmHg. Normal finger pressure is >80% of the ipsilateral brachial systolic pressure. Digital brachial index <0.8 is abnormal and <0.5 is diagnostic of ischemia. Absolute digital pressure <70mmHg is abnormal and <60mmHg is diagnostic for ischemia. A gradient between digits of >15 mmHg is considered abnormal. These criteria are used in occlusive disease and steal syndrome.(Chloros et al. 2008; Sen and Tripathi 2016)

Let's talk about some of the most frequently tested pathologies, starting with arterial TOS.

<u>Arterial TOS</u>: Results from compression of the subclavian artery at the level of the first rib within the scalene triangle. Arterial TOS testing is done by placing a sensor, most often photoplethysmography (PPG), on one finger of each hand, recording the resting waveforms and then recording while during maneuvers to evoke arterial compression in the thoracic outlet. (Size et al. 2013)

Can you tell us a little more about PPG testing?

<u>Photoplethysmography (PPG)</u> uses an infrared light to illuminate superficial tissue. The reflection is received by a photosensor, and amplitude of the reflected light is proportional to the volume of red blood cells in the sample area. A normal digital arterial PPG has a

brisk upstroke with a narrow systolic peak, and a dicrotic notch on the downslope during diastole.(Carter and Tate 2001)



Digital PPGs change with progression of peripheral vascular disease. The first changes are a loss of amplitude and loss of the dicrotic notch. More advanced disease findings include a flattened systolic peak and a prolonged upstroke. Significant arterial TOS is suggested when there is a loss or persistent flattening of the digit waveforms during any of the positional changes that can compress the subclavian artery (either with the clavicle, first rib and scalene muscle). However, it should be noted that up to one-third of patients without arterial TOS may have some degree of subclavian artery compression with positional maneuvers.

What are other diseases affecting the upper extremity?

<u>Raynaud's</u>: Vasospastic disorder characterized by temporary vasospasm. Diagnosis may be assisted by decrease in digital waveforms with immersion of the hand in cold water.

Thromboangiitis obliterans (Buerger's Disease): Is a segmental non-atherosclerotic inflammatory disorder characterized by microthrombosis that primarily involves the small- and mediumsized arteries. Ultrasonography may demonstrate the classical "corkscrew" collateral development at the level of occlusion. TA has a male predominance and first-line treatment is smoking cessation.

20.4.1 Hemodialysis Access

Before we segue to the lower extremities, this is a good time to discuss an entity frequently tested on the VSITE that constitutes for many vascular surgeons a notable portion of their practice: hemodialysis access, and specifically fistulas.

<u>Fistulas</u>: Ultrasound is one of the key modalities used in identifying suitable anatomy for fistula placement, suitability of a fistula for dialysis, and finally complications of fistulas.

So first, assessment for fistula placement:

The optimal configuration for an AVF is determined on the basis of vein mapping and non-invasive studies. Veins should measure >3 mm in diameter (>2.5 mm may be acceptable, as veins are likely to dilate under anesthesia), and there should be no venous outflow stenosis or arterial inflow stenosis.

Loss of phasic variation in the upper extremity veins may indicate central stenosis and that extremity may not be appropriate for access creation.(Y. C. Kim et al. 2009; Lok et al. 2020) Duplex ultrasound arterial imaging can be performed at the same time as vein mapping and can provide important predictors of fistula maturation, such as arterial diameter and flow. The minimal arterial lumen diameter is 2 mm.

How can we tell if a fistula is ready to be used for hemodialysis access?

Rule of 6's: At six weeks post-creation the diameter of the fistula should be at least 6 mm and the depth no more than 0.6 cm. The flow rate should be at least 600ml/min, and the length of the fistula should be 6 cm to allow for a successful two-needle dialysis.

Brachial artery volume flow rate (VFR) is one of the best measurements to identify poor fistula maturation. >800 ml/min is ideal, but >600 is often sufficient.(Ko et al. 2015) To determine the volume flow (VF) the sample volume (SV) should be opened wall to wall in a segment of fistula that is straight with a constant diameter and include all velocities present within the fistula. Angle correction should be less than 60 degrees.(Gerrickens et al. 2018; Ko et al. 2015)

What do normal fistula spectral waveforms look like?

Waveform: The arterial waveform should demonstrate very low resistance throughout diastole. End diastolic velocity should be one half to two thirds of peak systolic velocity in a well-functioning fistula. As a side note, this is also what one would see in an iatrogenic arteriovenous fistula, as between the femoral artery and vein.(Teodorescu, Gustavson, and Schanzer 2012)

Let's discuss commonly encountered complications and pathologies identified in association with fistulas:

<u>Pseudoaneurysms</u>: Pseudoaneurysms commonly occur when a puncture fails to seal and the blood is contained by the surrounding soft tissue. As in other locations, pseudoaneurysms

are defined on imaging by a communicating neck between the arterial vessel and pseudoaneurysmal sac with "to-and-fro" waveform at duplex.(Mahmoud et al. 2015) While small pseudoaneurysms can be managed without intervention or surgery, larger pseudoaneurysms, pseudoaneurysms associated with infection or overlying skin changes or bleeding may require excision and repair.(Dennis F. Bandyk 2013)

<u>Steal syndrome</u>: Hemodialysis-related steal, also known as access-related hand ischemia, which may occur in over half of all patients undergoing access creation. Steal is characterized by retrograde diastolic flow distal to the donor artery. Of note, reversal of flow in and of itself is not sufficient to cause distal ischemia with an intact palmar arch. This is commonly seen after access creation and represents physiologic steal phenomenon, rather than symptomatic steal syndrome. Digital pressures <60 mm Hg are highly sensitive and specific for predicting steal. Patients who have no symptoms (Grade 1 access-related hand ischemia), may be closely monitored without any intervention.

How do we treat more severe steal?

Flow rate measurements of the fistula can help determine the optimal treatment (banding, revision using distal inflow, distal revascularization with interval ligation, proximalization of arterial inflow or ligation of the fistula). For more on steal, see Section 5.3.3

So steal can occur in the context of high fistula output, can we talk a bit about low fistula flow, as from stenosis?

<u>Central venous stenosis</u>: Venous outflow stenosis is the most common reason for arteriovenous graft failure. A low flow rate results in recirculation during the dialysis session. Venous obstruction manifests as arm swelling, and with central venous stenosis may present with collateral development over the upper extremity and chest wall.

<u>Stenosis of fistula:</u> Arterial and mid-graft stenosis can also cause complications, but are less common than venous stenosis. Stenosis on imaging will be represented by narrowing of luminal diameter of 50% (which correlates to a 75% reduction in cross-sectional area) on b-mode ultrasound, PSV>400, Vr>2.5, high-resistance waveform proximal to the stenosis, and tardus parvus waveform distal to the stenosis.(Dennis F. Bandyk 2013)

20.5 Lower Extremity

Can you please tell us about some of the diagnostic modalities that are used to examine perfusion in the lower extremities?

<u>ABIs</u>: Ankle-brachial index measurement requires calculating the ratio of the highest ankle systolic pressure (posterior tibial artery or dorsalis pedis) over the highest brachial systolic pressure. Regardless of whether you're doing the R or L ABI, use the higher arm pressure for both ratios. Normal ABI >0.9, severe disease indicated by ABI<0.5, and CLI by ABI<0.3.

The ankle-brachial index in diabetic patients is frequently unreliable due to incompressibility of the tibial vessels at the level of the cuff secondary to calcification. Consequently, toe pressures are mandatory in all patients with diabetes mellitus. Normal TBI>0.7.

Of note, monophasic waveforms in the brachial artery likely represent proximal occlusive disease in the upper extremity and therefore make the ABI non-diagnostic because the upper extremity blood pressure is not likely not representative of central blood pressure.(A. Chen, Prout, and Jain 2014)

<u>TcPO2</u>: When there is significant tissue loss, preventing TBI measurement, another option is transcutaneous oximetry (TcPO2).(Mills et al. 2014b) Transcutaneous oximetry is a noninvasive method of measuring the tissue partial pressure of oxygen through a heated sensor on the skin. A TcPO2 value of 40 mmHg is the critical value below which wound healing is impaired and ischemia develops.

Are there other non-invasive ways of determining extremity perfusion?

<u>PVRs</u>: Pulse volume recordings. Normal PVR waveforms have a rapid upstroke, sharp peak, prominent dicrotic notch and downslope. This typically uses 4 cuffs. High thigh cuff should be 30% greater than brachial pressure, hence a thigh-brachial index of 1.3 is normal. However, ABI/PVRs may not demonstrate significantly abnormal values/waveforms in individuals with single level disease.



Exercise Test: During exercise testing, patients are placed on a treadmill after baseline resting ABIs are measured. The patient is then asked to walk for 5 minutes or until physical discomfort

requires test cessation. The point when the patient reports pain starts is the initial claudication distance and the point where the patient stops is defined as the absolute claudication distance. Diagnostic criteria for a positive test include a drop-in ankle pressure of greater than 20 mmHg from baseline, drop in ABI greater than 0.2 from baseline, or inability of ankle pressures to return to baseline after 3 minutes.(Strandness and Bell 1964)

So just to briefly summarize, what are parameters associated with poor wound healing?

 $\frac{\text{Parameters associated with poor wound healing: Considering these various testing modalities,}{\text{what are factors associated with poor likelihood of wound healing? An ankle pressure <50 mmHg, ABI < 0.40, TcPO2 <20 mmHg, or a toe pressure <20 mmHg are considered predictive of non-healing.}$

What is the utility of duplex in diagnosing lower extremity occlusive disease?

Duplex ultrasound is utilized to further delineate the anatomy, level and severity of occlusive disease. Increased PSV can identify the site of a hemodynamically significant stenosis. Diminished waveforms often suggest more proximal disease.

Of note, diffuse narrowing of the SFA can result in tardus et parvus waveforms in the popliteal, without necessarily demonstrating any areas of high velocity. (Aburahma and Perler 2022; Pellerito and Polak 2019)

While duplex ultrasound most easily visualizes infrainguinal arterial disease, it can also be used to evaluate for iliac artery stenosis. Patients with an iliac Vr>2.5 or monophasic/biphasic waveforms in the CFA, suggest that further imaging of the iliac system may be required.(Heinen et al. 2018)

Ultrasound is frequently used for graft surveillance. Can you talk about graft surveillance parameters?

Suggested bypass vein graft surveillance with ABI, clinical exam and duplex should begin immediately after surgery and then continues at 3, 6, and 12 months and then every 6 to 12 months thereafter. (Conte et al. 2006; Zierler et al. 2018) A velocity ratio (Vr) is often utilized in graft surveillance and is defined as the peak systolic velocity (PSV) at the site of a stenosis divided by the PSV in a normal vessel segment proximal to the stenosis.

- The highest risk for graft thrombosis, and highest cause for concern, is suggested by PSV >300 cm/s, Vr >3.5, a mid-graft flow velocity <45 cm/s or a drop in ABI >0.15. (Dennis F. Bandyk et al. 1988; Zierler et al. 2018)
- Increased risk grafts have a PSV 180-300 cm/s and a Vr 2.0-3.5.
- Spectral pulse wave doppler is more sensitive than color doppler to identify low flow states or thrombosed grafts.(Hedrick, Hykes, and Starchman 2005)
- Criteria for prosthetic grafts is higher than vein bypass. (Zierler et al. 2018)

Intervention for recurrent vein bypass graft stenosis early after angioplasty may be best treated with an open revision and patch plasty.(Jongsma et al. 2016)

Presenting with new swelling, pain and fluid around a prosthetic graft on duplex ultrasound may be a sign of underlying graft infection. PET and WBC scans may be positive up to 4 months after implantation, so do not provide much utility in the early post-operative period.(Puges et al. 2019)

20.5.1 Pathologies

Let's talk about pathologies frequently encountered in the lower extremities:

20.5.1.1 Pseudoaneurysms (particularly femoral)

Gray-scale ultrasonography demonstrates a hypoechoic cystic structure adjacent to an arterial supply. Color Doppler typically demonstrates a "yin-yang sign" within the pseudoaneurysm sac. The hallmark ultrasound sign is identification of a neck between the sac and the feeding artery with a "to-and-fro" spectral Doppler waveform measured at the neck. This represents the flow in and out of the PSA during systole and diastole.



Anastomotic pseudoaneurysms develop in around 3% of femoral anastomoses and can be associated with arterial wall weakness, endarterectomy, mismatch between artery/graft, dilation/deterioration of graft material, increased/uneven tension of the anastomosis, or underlying infection.(Pellerito and Polak 2019)

20.5.1.2 Dissections

Characteristic ultrasound findings on color Doppler include a parallel blood-flow channel that separates the true and false lumen, with diminished blood flow distally.(Ge et al. 2015; N. Kobayashi et al. 2018) Traumatic intimal flaps after knee dislocation can have a normal ABI and be asymptomatic, but they should be followed as they can progress.(Gaitini et al. 2008) Of note, if the false lumen is filled by a thrombus, it may not distinguishable from an intramural hematoma or non-calcified plaque.



20.5.1.3 Buerger's Disease

Also known as thromboangiitis obliterans, this is a rare disease, but with characteristic findings on ultrasound of serpiginous/corkscrew collaterals, occlusion of distal calf/pedal arteries, and normal proximal arteries.(K. Busch 2011; Fujii et al. 2011)

For more see Section 3.3.4

There are several disease pathologies that are frequently tested relating to the popliteal fossa. Can you talk briefly about these?

20.5.1.4 Cystic Adventitial Disease

This is a rare (but often tested) pathology. Adventitial cystic disease is a non-atherosclerotic etiology of claudication, most often affecting the popliteal artery in the lower extremity and leading to stenosis or occlusion. Duplex imaging of the popliteal artery will demonstrate an anechoic or hypoechoic *intraluminal* region with a smooth contour and stenosis documented by velocity increase on spectral doppler. Angiography will often show a classic "scimitar" sign - a smooth well-defined crescent-shaped defect.(M. Shaw et al. 2007; Winn et al. 2015)



For more on adventitial cystic disease, see Section 17.4.2. This is not to be confused with:

20.5.1.5 Bakers Cyst

A benign, cystic structure found in the popliteal fossa and arising from the joint capsule. Flexion of the knee may result in compression of the popliteal artery by the cyst. On b-mode ultrasound, a well-defined, anechoic cystic structure with a 'neck' extending into the joint space between the semimembranosus tendon and the medial head of the gastrocnemius will be identified.



<u>Popliteal entrapment syndrome:</u> decrease in ABI or loss of distal pulses with passive dorsiflexion or active plantar flexion of the foot caused by compression of the popliteal artery by the gastrocnemius.

20.6 Abdominal

Lowest frequency transducers (2-4MHz) are used for imaging abdominal vessels.

20.6.1 Aorta

Abdominal Aortic Aneurysms (AAA): Ultrasound screening of a AAA is non-invasive, accurate, and cost-effective.(Chaikof, Dalman, Eskandari, Jackson, Lee, Mansour, Mastracci, Mell, Murad, Nguyen, and al. 2018)

- Sensitivity of 98% and specificity of 99% for AAA diagnosis.(Wilmink et al. 2002)
- Aneurysm is measured in transverse view, anterior to posterior diameter, outer wall to outer wall.(IAC 2021) Size criteria for AAA repair is 5.5 cm men, 5.0-5.5 cm in women.
- Can be limited by bowel gas or obesity so recommend fasting.
- For more details, see Chapter 10

Can ultrasound be used for graft surveillance s/p EVAR?

Surveillance color duplex ultrasound is safe if CT imaging at 1 year exhibits no sac growth, graft migration, or endoleak (or stable type II endoleak). Contrast enhanced duplex, in particular, has a high sensitivity and specificity to detect endoleaks.(Abraha et al. 2017; Kapetanios et al. 2019; Chaer et al. 2009)

Type II endoleak classification is subdivided on surveillance duplex. Type IIa has a single inflow vessel and a "to-and-fro" appearance, whereas Type IIb has multiple vessels and maintains a monophasic flow. Type IIa endoleaks with velocities greater than 100cm/sec are not likely to spontaneously resolve and may be resistant to therapy.(Arko et al. 2003; Marrewijk et al. 2004)

Type II endoleaks with stable sac size can continue surveillance with color duplex. (Chaikof, Dalman, Eskandari, Jackson, Lee, Mansour, Mastracci, Mell, Murad, Nguyen, and al. 2018; Karthikesalingam et al. 2012; Chaer et al. 2009) Any findings concerning for growth, endoleak or flow in the sac should have confirmatory CTA performed. (Maleux et al. 2017)

Surveillance duplex does not reliably detect graft migration given challenge of renal arteries visualization in a long axis view of the aorta.(K. J. Busch et al. 2009)

Velocities within stents are elevated from baseline due to stent rigidity, but a limb velocity >300 cm/s with a Vr >3.5 is associated with a high risk of limb occlusion from stenosis or kinking and should undergo further investigation or intervention. (Chaikof, Dalman, Eskandari, Jackson, Lee, Mansour, Mastracci, Mell, Murad, Nguyen, and al. 2018; Blom et al. 2012)

What other pathologies can be identified on ultrasound evaluation of the abdomen?

<u>CIA aneurysms</u>: SVS defines CIA aneurysms as any permanent, localized dilatation of the iliac artery >1.5 cm in diameter (diameter 1.5x the normal diameter)

<u>Para-anastomotic pseudoaneurysms</u>: As previously discussed (reported in up to 0.5% to 10% of cases)

<u>Penetrating aortic ulcers</u>: Describes an ulcerating atherosclerotic lesion that penetrates the intima and progresses into the media. Associated with atherosclerotic plaque on ultrasound

Dissections: (as already discussed) a dissection flap is usually identified and color flow demonstrates dual channels (true and false lumens). Turbulent flow patterns are frequently encountered.

20.6.2 Mesenteric Vasculature

So far we have steered clear of the mesenteric vasculature. But this is an area frequently encountered on exams. Let's start with mesenteric vessel stenosis.

Celiac and SMA stenosis: May present as chronic mesenteric ischemia. PSV >275 cm/s in the SMA or >200 cm/s in the celiac artery indicates >=70% stenosis. Normal SMA Doppler waveforms in the fasting patient show high resistance waveform, PSV <275 cm/s and no spectral broadening. In the postprandial state, the waveform becomes low resistance, with a slightly increased PSV and little to no spectral broadening. Significant SMA stenosis may be differentiated by the presence of spectral broadening and elevated PSV (>275 cm/s) and EDV (>55cm/s).(AbuRahma et al. 2012; Gustavo S. Oderich and Ribeiro 2019b) Distal to the stenosis, one would expect a tardus parvus waveform. EDV >45 in SMA or >55 in celiac are predictive of stenosis (would expect higher diastolic flow in celiac trunk given low resistance vascular bed of liver and spleen).

Reversal of flow in the hepatic artery and mildly elevated SMA velocities may indirectly signify severe stenosis or occlusion of the celiac artery.

Note that aberrant anatomy with the hepatic artery arising from the SMA, will result in the proximal SMA demonstrating a low resistance waveform, even in the fasted state.(Pellerito and Polak 2019)

Mesenteric duplex is still valid even if the patient has not fasted. The images may be technically more difficult to acquire due to bowel gas, however if velocities can be obtained, you are likely to see elevated PSV and low resistance waveform in the SMA. The EDV will also be elevated and should be interpreted with caution.

IMA Stenosis: PSV > 200 cm/s or aortic ratio > 2.5 has a high sensitivity for hemodynamically significant stenosis. However, this is rarely clinically significant and does not require treatment.

What other mesenteric vessel pathologies may be identified on vascular ultrasound?

Dissections: Rare without concomitant aortic dissection, describe only in case reports.(Gouëffic et al. 2002; Oglesby and Sorrell 2006)

<u>Aneurysms:</u> Rare. Repair >2 cm celiac, hepatic, SMA aneurysms and >3 cm splenic and renal.

<u>Median arcuate ligament syndrome</u>: MALS can cause significantly elevated velocities at the origin of the celiac artery. Testing is for reversible mechanical compression, as opposed to a fixed lesion from atherosclerotic disease. During deep inspiration or in the upright position the MAL is elevated and celiac velocities should normalize if stenosis is secondary to MALS.(Tembey et al. 2015)

Hepatic artery thrombosis: Mesenteric duplex is often performed in liver transplant patients. Hepatic artery thrombosis most often occurs within the first 30 days after liver transplantation, with an incidence of 1.5-9% and a mortality rate of 75%.(García-Criado et al. 2009; Sanyal et al. 2012)

20.6.3 Portal Vein

So I recognize that we are jumping ahead here discussing venous circulation, but as this represents another component of a mesenteric vascular exam, let's discuss portal venous ultrasonography.

Normal portal venous flow is hepatopetal (toward the liver), whereas abnormal portal venous flow is hepatofugal (away from the liver). (The root "fugua" means to flee, or flight). Flow should be in the same direction as the hepatic artery.

Other abnormalities that may be visualized are portal vein thrombosis - often associated with portal venous hypertension in patients with chronic cirrhosis, hepatitis, or hepatocellular carcinoma. Acute portal vein thrombosis on ultrasound demonstrates dilatation of the portal vein with hypoechoic intraluminal thrombus. Chronic portal vein thrombosis is characterized by a contracted vein with heterogeneous/hyper-echoic echoes and may be associated with collateral formation.(Chawla and Bodh 2015; Nouvini and Hapani 2013)

Mesenteric duplex can also assess for functioning of TIPS (transjugular intrahepatic portosystemic shunt). A sign of a failing TIPS is portal vein peak systolic velocity less than 30cm/s.

20.6.4 Renal arteries

Awesome, and so before jumping fully into venous circulation, let's complete our discussion of abdominal ultrasound evaluation with a discussion of the renal vasculature.

Renal Pathologies: ostial (atherosclerotic) vs mid-artery (FMD)

<u>Atherosclerotic stenosis:</u> PSV > 200 cm/s and Renal-aortic ratio (RAR) >3.5, suggest >60% stenosis, however some labs use higher PSV criteria (250-285 cm/s).(Schäberle et al. 2016; Rickey and Geary 2019b) Of note, for RAR, the aortic PSV needs to be within normal range (40-100) to be appropriately interpreted.

The renal resistive index (RRI) is calculated as the RA (PSV-EDV)/PSV. The RRI (>0.8) is an indicator of intrinsic parenchymal renal disease. RRI has been used for assessment of transplant renal allograft rejection, decision making for treatment of renal artery stenosis and to evaluate progression of CKD.(Radermacher et al. 2001) Diabetic nephropathy will demonstrate elevated RRI, signifying parenchymal disease.(Viazzi et al. 2014) One study

showed it was associated with poor renal and overall outcome in critically ill patients.(Le Dorze et al. 2012)

 $\underline{\text{FMD}}$: Fibromuscular dysplasia (FMD) will demonstrate tortuous renal arteries with turbulent flow and increased PSV of the mid-distal portions. Beading may be difficult to see on ultrasound (compared to the extracranial carotid). Atherosclerosis disease is primarily ostial in nature.

<u>In-stent restenosis</u>: Higher cut off values are required to determine in stent restenosis.(Chi et al. 2009; Schäberle et al. 2016)

Renal artery aneurysms: Can see a yin-yang sign. (Gutta et al. 2008; Ham and Weaver 2014)

20.7 Venous

Venous Exam: Uses a linear transducer. Complete deep venous reflux duplex examination includes color and pulsed wave spectral doppler imaging. Spontaneous Doppler waveforms as well as provocative maneuvers are recorded in the common femoral, femoral, popliteal, and tibial deep veins. Superficial veins (GSV, SSV, and perforator veins) are evaluated with provocative maneuvers to test valve competency. Diameters are also included for superficial veins. Transverse B-mode images are used for vessel compression (as when looking for thrombus) to ensure that the vein is fully compressible under probe pressure as opposed to simply slipping out of view, as may occur in longitudinal views. Reflux exam should be performed with a patient standing, with assessment performed on the non weight-bearing leg.

We've talked extensively about normal arterial waveforms; what do normal venous waveforms look like?

<u>Normal venous waveforms</u>: Normal flow patterns of iliac and femoral veins demonstrate phasicity and should augment with distal compression. In the upper extremity central veins, doppler waveforms normally demonstrate pulsatility and phasicity.

What does this mean? Pulsatility, phasicity, and augmentation?

Pulsatility: Refers to changes in the venous waveform in accordance with the cardiac cycle. Pulsatility is normal in the upper extremity veins central veins, given their proximity to the heart.



This is an abnormal finding in the lower extremity veins, and may be a sign of pulmonary hypertension, right heart failure, or tricuspid regurgitation.(Kakish et al. 1996; McClure et al. 2000)

Phasicity: (also called respiratory phasicity) is variation in the waveform with respiration. This results from increasing and decreasing intrathoracic pressures secondary to respiration. Phasicity is an indicator of a patency proximal to the point of measurement. So if we see lack of phasicity (continuous flow) in the left femoral vein but normal phasicity in the right, we would be concerned for left iliac vein occlusion or stenosis. If we saw absence of phasicity (continuous flow) in the bilateral femoral veins, we would be concerned for IVC obstruction or stenosis.



Augmentation: Distal compression that augments forward flow. For example, if we are measuring flow at the femoral vein and we squeeze the calf and we see augmentation in the waveform, this indicates lack of occlusion in the venous system from the knee to the probe.



What are venous pathologies that are commonly tested/encountered?

20.7.1 Reflux

Venous reflux due to valvular incompetence is best assessed with duplex scanning in the upright position. Reflux in the common femoral vein and the saphenofemoral junction may be elicited with a Valsalva maneuver (which increases intra-abdominal pressure), but release of a pneumatic cuff compression is a more reproducible method. Reflux is identified as reverse flow - that is, away from the heart - following valsalva or release of the compression cuff.



Consensus guidelines suggest a cutoff value of 1 second for abnormally reversed flow (reflux) in the femoral and popliteal veins and of 0.5 seconds for the great saphenous vein, small saphenous vein, tibial, and deep femoral veins.(Gloviczki et al. 2011; Lurie et al. 2012; Labropoulos et al. 2003)

<u>Perforator veins</u>: Perforator veins connect the deep and superficial venous systems, penetrating the deep fascia overlying the muscle. Size >3.5 mm and reflux >350 ms (deep to superficial) is associated with perforator reflux.(Min, Khilnani, and Golia 2003; O'Donnell et al. 2014; Sandri et al. 1999) Pathological perforators are found in association with a healed or non-healed ulcer.

What are some other examples of reflux?

<u>Ovarian vein reflux</u>: The ultrasound evaluation of pelvic congestion syndrome is performed in a fasting patient in steep reverse Trendelenburg and standing positions with a low-frequency probe. Reflux is identified during the Valsalva maneuver. There are no validated criteria for the duration of reflux, rather, an ovarian vein diameter >6 mm is considered significant.

20.7.2 May Thurner Syndrome

May Thurner syndrome is also known as iliac vein compression syndrome, refers to a chronic compression of the left common iliac vein by the overlying right common iliac artery (CIA), with or without deep venous thrombosis. Notably, patients present with unilateral (left) lower extremity edema and pain, varicosities, DVT or venous ulcers. Intravascular ultrasound will demonstrate >50% stenosis of the iliac vein from compression.

Iliac obstruction, whether a result so May Thurner or not, would demonstrate continuous flow with a loss of phasicity and reduced augmentation with distal compression or valsalva. Additional indirect ultrasound findings include collaterals and reversal of flow. Direct ultrasound findings include intraluminal changes.(Metzger et al. 2016)

20.7.3 Nutcracker Syndrome

Nutcracker syndrome is most often diagnosed based on IVUS, venogram or renal vein pressures, but there are also diagnostic criteria for duplex ultrasonography. The left renal vein velocity ratio is the velocity at the aortomesenteric angle divided by the velocity near the left renal hilum. A ratio greater than 5 has a sensitivity of 80% and specificity of 94% for Nutcracker syndrome. (S. H. Kim et al. 1996; Seung Hyup Kim 2019)

20.7.4 Thrombosis

<u>Thrombosis</u>: Characteristics of acute thrombus are an echolucent and incompressible thrombus in a thin-walled vein with significant distension.



Acute thrombus typically causes the vein to dilate with a diameter greater than the diameter of the adjacent artery. Venous wall thickening/scarring, a contracted vein, recanalization, and collaterals are found in chronic thrombosis.

Components of a lower extremity DVT study include doppler waveforms in response to provocative maneuvers in the common femoral, femoral and popliteal veins. When performing a DVT scan, the technician should evaluate the contralateral femoral vein flow to compare flow patterns.(Needleman et al. 2018; Vascular Ultrasound 2019) Transverse views are used for compression and compressibility testing every 3-5cm is the most reliable evaluation for thrombus.(Malgor and Labropoulos 2013)

Let's discuss some examples of venous thrombosis

Venous TOS: Venous thoracic outlet syndrome is thrombosis or severe stenosis of the subclavian or axillary veins secondary to chronic extrinsic mechanical compression. Repetitive injury to the subclavian vein at the level of the costoclavicular space results in chronic injury to the veins. Venous duplex may show a dilated, non-compressible vein consistent with an acute subclavian vein DVT, or lack of pulsatility/phasicity if obstruction/stenosis is more centrally located. Duplex ultrasound is 81-100% sensitive for diagnosing upper extremity DVT.(Mustafa et al. 2002)

EHIT or Endovenous Heat Induced Thrombosis: S/p endovenous thermal ablations (RFA or laser ablation) of the GSV. 4 Grades: Grade 1 is thrombus in the GSV up to the level of the CFV. If < 50% of the CFV lumen is involved this is EHIT grade 2. EHIT grade 3 is extension into the CFV occupying >50% of the lumen and grade 4 is occlusion of the CFV.



EHIT grades 3-4 are typically treated with anticoagulation to reduce risk of PE. To minimize the risk of EHIT, the catheter should be positioned at least 2 cm from the saphenofemoral junction.

Superficial vein thrombosis: when scanning for superficial vein thrombosis, pornographers should scan superficial and deep veins of the ipsilateral leg and the contralateral central veins with doppler wave form, because the incidence of concomitant thrombosis is not insignificant. (Decousus et al. 2010; IAC 2021; Vascular Ultrasound 2019) This is discussed more in our venous chapter: Section 18.1.4.1

20.7.5 Vein Mapping

One additional venous study often performed in the lower extremity is mapping of the GSV for planning lower extremity bypasses. Mapping should identify the general course of the GSV and any major tributaries, particularly if the anterior or posterior accessory veins are of a size that might be a reasonable additional conduit. (Cohn, Caggiati, and Korver 2006) Sonographers should also make note of any significant areas of tortuosity or aneurysm that might compromise the quality of the bypass conduit. Size measurements are usually made along the GSV.

In general the ideal size for conduit is 3-4mm throughout it's length, with a size of greater than 2mm at the ankle. (Leopold et al. 1989) These size assessments are usually the same regardless of whether the vein will be fully harvested and reversed or utilized in situ. The small saphenous

vein can be utilized for a conduit if the GSV is not adequate and the diameter is 3-4mm.(B. B. Chang et al. 1992; Nierlich et al. 2019)

20.8 Artifacts

Finally, let's wrap up this episode with a discussion of imaging artifacts. These are frequently encountered and tested in vascular ultrasound, and it is important to recognize imaging artifacts in order to prevent incorrect interpretation.

Acoustic shadowing: Shadowing on an ultrasound image is characterized by a signal void behind structures that strongly absorb, reflect, or refract ultrasonic waves.



Practically speaking, this most typically occurs deep to strongly reflective surfaces such as calcified plaques, and appears as a "dark area" beneath the plaque. Subcutaneous air can result in shadowing and loss of imaging information, which can be seen in some soft tissue necrotising infections. (Thom and Warlaumont 2017)

Acoustic enhancement is essentially the inverse situation, and appears as a "bright area" deep to structures that transmit ultrasound waves exceptionally well. This can happen deep to fluid-filled structures such as cysts.

Mirroring: A mirror-image artifact is caused by reverberation of ultrasound and shows structures that exist on one side of a strong reflector as also being present on the other side of the reflector.



This is often seen around the pleura and the diaphragm, due to the strong reflection of ultrasound from the air-filled lung. These artifacts can occur in both B-mode imaging where you see the mirrored image and Doppler, in which you see the mirrored waveform. The clavicle is also a strong spectral reflector, and thus can result in a mirror effect and a duplicate subclavian artery.(Rubin et al. 2010)

Refraction: A refraction artifact is the result of ultrasound waves passing through tissues with different propagation velocities (such as air and water) and causes a structure to be improperly positioned laterally in the image. This is the phenomenon that results in a straw appearing bent when in a glass of water.

Speed artifact: Depth determination by an ultrasound machine is based on calculations using an average propagation velocity of sound in soft tissue of 1540 m/s. If the ultrasound wave passes through a medium at a different speed than predicted by the machine, an inaccurate image depth will be displayed. If the ultrasound passes less quickly through the material than soft tissue (as occurs in air or fluid), then the image will be displayed deeper than the true depth. In practical application, this is what causes a "bayonet" sign, or apparent bending of a needle when it passes from soft tissue into a cystic structure. **Inappropriate color gain:** Overly gained images will show "speckling" in areas in which no flow is present (such as in soft tissue). Increasing the gain, increases both the signal and the noise.(Kremkau 2021) The noise added to the spectral waveform can result in PSV overestimation.



Under-gaining will result in reduced sensitivity to low velocity flow.

Inappropriate angle correction: Make sure the angle correction cursor is centered in the vessel and parallel to the walls, otherwise the Doppler velocity measurement will be incorrect. Increasing the doppler angle will increase the velocity measurement (if no change to angle correction is made).



Finally, can we talk about aliasing, as I feel like this comes up a lot on exams:

Aliasing: Unlike continuous wave Doppler, pulsed wave and color flow Doppler are characterized by rapid pulses of ultrasound waves (at a rate called the pulse repetition frequency). The Nyquist limit defines the frequency at which aliasing will occur, as equal to the PRF/2. So what does this mean practically? In pulsed wave doppler, if the velocity of blood is greater than ¹/₂ the PRF, the peak velocity will be cut off, and wrapped around to the bottom of the scale.



This results in inaccurate measurement of peak velocities, and may be remedied by increasing the PRF and hence the scale. In color flow doppler, aliasing appears as red to blue hues without separation of a black region indicating no flow. This occurs in areas of high velocity (such as immediately post-stenosis). This can be remedied with an increase in the color scale. Of note, if asked to determine the direction of blood flow in a vessel demonstrating aliasing, one should assess the flow in low-velocity areas of blood flow, as seen along vessel walls.

20.9 Accreditation and Credentialing

Intersocietal Accreditation Commission (IAC) Accreditation - voluntary, not linked to reimbursement but may be required by some payers. Medical director must have a certain training/experience, including RPVI as one way; technical directors must have credentials through ARDMS, CCI or ACR with re-credentialing every three years.(IAC 2021)

• Physician accreditation with RPVI requires re-certification every 3 years with 30 hours of CME specifically in the area of vascular ultrasound.

Regulations for a completed vascular exam report(IAC 2021)

- Patient identifier
- Date of exam

- Sign/symptom as indication medical necessity should not just include "rule out"
- Reasons for any limitations to the examination
- Comparisons to any previous related studies
- Reported by medical staff within 2 working days and final report verified within 4 working days.

Troubleshooting examinations

- Any concerning acute findings, such as mobile intra-arterial material concerning for embolism, should be reported to the supervising/reading physician immediately before completing the examination.(Diagnostic Medical Sonography 2015)
- Some modifications to tests can be made if the patient is unable to tolerate, such as standing or tolerating a valsalva maneuver.(Diagnostic Medical Sonography 2015)

21 Radiation Safety

Authors: Adam P Johnson and Melissa Kirkwood

21.1 Important Terms

Absorbed Dose - energy deposited by ionizing radiation in a medium per unit mass. Dosage measured and communicated most commonly in Gray (Gy) equal to 100 rad.(A. B. Reed 2019)

Effective Dose - energy deposited by ionizing radiation in a medium, taking into account the sensitivity of the specific tissue, time and duration of exposure. These factors compose a weighting factor (W) and thus this is calculated by multiplying absorbed dose (Gy) by the weighting factor (W). Dosage measured and communicated most commonly in Sievert (Sv) equal to 100 rem.(A. B. Reed 2019)

Reference Air Karma (RAK) - Radiation output at a specific reference point along a fluoroscopic axis.(Kirkwood et al. 2013)

- Located 15cm along the beam axis toward the focal spot from the isocenter.
- Best approximation of cumulative patient dose or peak skin dose (PSD).(D. Kwon, Little, and Miller 2011)
- Includes dose from fluoroscopy and fluorography acquisitions, however does not account for gantry angulation of changes in height.

Substantial Radiation Dose Level (SRDL) - defined as 5Gy RAK and should trigger specific patient follow up. Of note, 5Gy refers to the RAK that is reported at the end of the procedure, which is an approximation of the peak skin dose (PSD).(Baiter et al. 2011; Hirshfeld et al. 2005; Kirkwood et al. 2015; Stecker et al. 2009)

21.2 Radiation effects

The effects can impact both patients and clinicians. Patients often receive a higher dose during a single procedure, where as clinician doses, even with proper shielding, accumulate over multiple procedures throughout their careers. The effects of radiation are defined as either deterministic or stochastic.

Table 21.1: Skin related deterministic effects that can be seen after a single exposure over a certain threshold (Balter et al. 2010; Hirshfeld et al. 2005; Kirkwood et al. 2014; Stecker et al. 2009; L. K. Wagner et al. 1999)

Dose	Effect
0-2Gy	No observable effects
2-5Gy	Transient skin erythema and dermatitis, full recovery
	6w-1yr(Guesnier-Dopagne et al. 2019; Kirkwood et al. 2014)
5-10Gy	Erythema and epilation (hair loss), prolonged erythema up to 1y
10-15 Gy	Permanent epilation, atrophy or inducation up to 1y
>15Gy	Dermal Necrosis

The ocular lens is the most radio-sensitive tissue and cataracts is an unfortunately common deterministic effect. (Machan 2018; Brown and Rzucidlo 2011) Annual limits for occupational exposure include 20mSv per year and total threshold of 0.5Gy, which was recently lowered due to risk of cataract development from ocular exposure. (Hamada et al. 2017)

Stochastic effects can occur after exposure of any dose, but are seen more frequently in populations with higher radiation exposure. These are based on population studies and exact dose dependent relationships have not been established. These include:

- Brain cancer(Rajaraman et al. 2016; Kirkwood et al. 2018)
- Breast and thyroid cancer(Johnson et al. 2001; Einstein 2012)

21.3 Risk factors

Certain patient, procedure and clinician factors put patients at increased risk for radiation exposure.(Killewich et al. 2011; Mitchell and Furey 2011) These should be considered when planing and executing endovascular procedures to ensure that dosages are *as low as reasonably achievable (ALARA)*

- Patient Factors
 - Obesity

- Procedure Factors
 - Case complexity
 - Need for magnification
 - Case orientation and angulation
- Clinician Factors
 - Use of multiple subtraction runs

21.4 Reduction strategies

The main source of radiation to the clinicians is scatter from the patient. There are a number of techniques that can be used to reduce exposure of clinicians to this radiation.(Heidbuchel et al. 2014; Kirkwood et al. 2013)

- Reduce the time of exposure, keeping a close eye on fluoroscopy time and dosages as a procedure progresses.
 - Preoperative planning is paramount to efficient use of radiation during a procedure. Recent ESVS guidelines recommend utilization of a 3D pre-operative planning software and image fusion (when available) for complex endovascular procedures.(Modarai et al. 2022)
 - Use last image hold to allow for procedural planning
 - Decrease the frame rate to reduce the frequency of radiation exposure per second.

💡 Take a Look

Check out Dr. Ellozy's operative planning video where he reviews how to plan for an EVAR procedure utilizing Tera-Recon 3D imaging software.

- Position the source as far away from the operator as possible to still achieve optimal imaging. Radiation dose changes according to the inverse square of distance. Therefore twice the distance results in one quarter the dose.
 - For example, left anterior oblique (LAO) will bring the source to the patient's right side, causing higher doses to clinicians on that side of the patient.(Kirkwood et al. 2013; Sailer et al. 2019)
- Appropriate shielding

- Recommendations are that clinicians wear lead aprons covering their torso and legs with a thyroid shield and leaded glasses.
- Collimation can be helpful to reduce patient and operator dose, scatter and improve image quality.(Haqqani et al. 2012)
- The location with the highest exposure to scatter is below the table.(Gonzales, Moran, and Silberzweig 2014; Miller et al. 2010) Therefore, leaded skirt and extended lower body shields reduce radiation to the operator's legs.(Kirkwood et al. 2015)
- Of note: leaded caps DO NOT reduce radiation exposure to the brain because the majority of radiation is received as scatter from the patient up through the face and neck.(Kirkwood et al. 2018)
- Lead shielding should be regularly inspected and discarded if damaged. Particularly if defect is >15mm2 on a critical organ area, >670mm2 along a seem/overlapping area, >11mm2 on a thyroid shield.(Healthcare Inspections 2014; University, Stanford, and Complaints, n.d.)

💡 Take a Look

Check out our episode with Dr. Jasmine Bhinder, discussing a recently published quality improvement project for institutional level changes that lead to reductions in radiation exposure for trainees.

21.4.1 Pregnancy considerations

Pregnancy of both patients and clinicians need to be considered in relation to radiation exposure and safety.(V. Chandra et al. 2013; Mitchell and Furey 2011; P. Shaw et al. 2011) This is a highly tested subject on vascular surgery examinations. Some important take aways include:

- CDC has released guidance for potential prenatal effects on radiation exposure(CDC 2011) The majority of effects on fetuses is extrapolated from studies of the fall out from Hiroshima, Nagasaki and Chernobyl.
 - <0.05Gy represents no measurable risk to embryo or fetus at any gestational age.
 - 0.05-0.5Gy can be dangerous in the first trimester, but has not been associated with defects later in pregnancy.(P. Shaw et al. 2011)
 - >0.5Gy can be dangerous at any point during pregnancy.
- Dose limit recommendations during the 9 months of pregnancy is 5 mSv (500mrem) or 0.5 mSv (50mrem) per month.(Dauer et al. 2015; V. Chandra et al. 2013)

Strategies to reduce exposure include:

- Avoid direct fluoroscopy to the fetus, high-gantry angulation, and femoral access
 - Use collimation to ensure fetus is excluded from imaging field.
- Use adjuncts of intravascular ultrasound and lead shielding when able.
- Limit fluoro time.
- Some recommendations state that operators who intend to get pregnant should start wearing maternity aprons (lead equivalent to 1mm) even prior to knowing they are pregnant.(P. Shaw et al. 2011)
 - Original studies demonstrated reduction in fetal exposure by 80% and is currently common practice.(Witrak and Sprawls 1984)
 - However, a recent multi-institutional review showed that fetal exposure is minimal even in regular lead, therefore the additional weight of maternal lead may be unnecessary.(V. Chandra et al. 2013)

21.4.2 Regulation

Joint Commission Oversight - sentinel radiation reporting is aimed to promote awareness of preventable events perform root cause analyses to understand the reasons for events.(Commission 2019) It is important to escalate events early, as effects may not occur until much later.(Arbique, n.d.) Sentinel events include:

- 1. The patient has a permanent cutaneous injury and the proper dose saving techniques were not used during the procedure
- 2. Cumulative dose of 15 Gy for a single field over 6mo 1yr.
- 3. Delivery of radiotherapy to the wrong body region.
- 4. Actual dose more than 25% above planned radiotherapy dose.(A. K. Jones and Pasciak 2011)

Institutional Oversight - many institutions develop their own guidelines for employee exposure and mitigation strategies. Some common policies regarding dose limits include:

- Monthly limit for dosimeter reading of 0.1 mSv (100mrem) per month.
- Recommendations for annual occupational dose is <20mSv per year averaged over 5y and no more than 50mSv in any one year. Occupational dose <100mSv per year is not thought to increase cancer risk

• Recent reduction of occupational dose limit to 50mSv is due to increasing data connecting cataracts to radiation exposure.(Hamada et al. 2017)

21.5 Contrast Reduction

The most common complications related to iodinated contrast during endovascular procedures are hypersensitivities and acute kidney injury. There are a number of mitigation strategies to limit the effect of contrast on patients undergoing endovascular procedures.

In general patients with food allergies do have an increased incidence of contrast media allergies, however no specific common allergen has been identified. A new seafood allergy should not postpone or require pre-medication if a patient has previously tolerated IV contrast.(Schabelman and Witting 2010)

21.5.1 CO2 Angiography

CO2 angiography is often utilized in place of iodinated contrast during fluoroscopy. A bolus of CO2 is injected, which then absorbs less ionizing radiation than surrounding tissue and provides a map of the arterial tree. CO2 can be used in a wide range of endovascular procedures, even a ruptured AAA.(Knipp et al. 2010) However, there are specific limitations and complications that should be understood.

21.5.1.1 Limitations

- Contraindicated in imaging above the diaphragm.(Caridi and Hawkins 1997; Sharafuddin and Marjan 2017)
- Susceptible to bolus fragmentation and often requires stacking to fully visualize the target arterial bed.(Caridi and Hawkins 1997; Sharafuddin and Marjan 2017)
- Bowel gas can limit imaging of the abdomen. Glucagon can be administered to reduce bowel gas motion artifact and improve the image.(Caridi and Hawkins 1997; Kyung Jae Cho 2015; E. Criado et al. 2012; Sharafuddin and Marjan 2017)

21.5.1.2 Complications

• Vapor lock - can occur with high volume, serial injections where contaminated air accumulates. Strategies to reduce incidence include waiting 1-3min between angiography runs. Operators should use a one way valve to reduce risk of air contamination.(Kyung J. Cho and Hawkins 2011; Kyung Jae Cho 2015)

- Cardiac/pulmonary vapor lock can occur with venography. Mimics a PE with hypoxia and hypotension. Initial management is to place patient in the left lateral decubitus/trendelenberg position.(Caridi and Hawkins 1997; Sharafuddin and Marjan 2017)
- Mesenteric vapor lock presents with significant unrelenting abdominal pain. Fluoroscopy can confirm a retained bubble. Initial management includes ongoing heparinization to prevent down stream thrombosis and maneuvers to break up the bubble by rotating the patient side-to-side or deep abdominal massage. Catheter aspiration may be needed.(Caridi and Hawkins 1997; Sharafuddin and Marjan 2017)

💡 Take a Listen

Check out our previous episode with a panel with Drs. Kirkwood, Wohlauer, and Chandra discussing occupation hazards for the vascular surgeon, of which an important hazard includes radiation safety.

22 Endovascular Access

Authors: Sammy Siada and Rafael Demarchi Malgor

Endovascular procedures are the cornerstone of any modern vascular surgery practice. Because most endovascular procedures are performed percutaneously using arterial or venous access, it is critical that vascular surgeons are facile with various techniques and devices used for endovascular access. Today we'll be discussing the various access sites, techniques for access, closure devices, and complications.

What factors play a role when choosing a site for access?

The factors to think about when thinking about which vessel to access are:

- The appropriateness of the access site the procedure performed
- Ability to obtain hemostasis at the conclusion of the procedure
- Ability to convert to open if necessary
- Effects of access on the tissues supplied by the accessed vessel and distal limb perfusion

What makes a vessel appropriate for access?

One of the most important factors when planning your access is the size of the vessel. The vessel needs to be able to accommodate the catheters and devices that will be used to perform the procedure. For instance, a brachial artery with less than 4mm diameter should not be accessed by a large bore sheaths, such as a 12Fr sheath.

The vessel also needs to be a in a location that can allow access to the target vessel of interest. Additionally, the vessel needs to have an area that is relatively healthy to access the vessel safely and minimize complications. Heavily calcified vessels especially those with anterior wall calcification might not be appropriate for access.

What about the ability obtaining hemostasis at the end of the procedure?

The ability to obtain hemostasis is critical to be able to perform endovascular procedures safely which is one reason why the common femoral artery is the most commonly accessed vessel.

Hemostasis is most commonly achieved through manual compression by compressing the artery against the femoral head. The brachial artery can also be compressed against the humerus, but because it's a more mobile vessel, compression is less effective and can lead to hematoma

or pseudoaneurysm formation which may necessitate an operation to prevent compression of the median nerve

Patients who will need to be uninterruptedly anticoagulated peri- and post-operatively pose a challenge to hemostasis. The use of closure devices is very important in these situations to prevent access bleeding.

A variety of closure devices can also be used to assist in hemostasis, each with their own inherent advantages and disadvantages. In general, closure devices are contraindicated in small diameter and heavily calcified vessels.

In any minimally invasive procedure, there is always a chance that you may need to convert to open. How does converting to open play a role in vascular access?

Conversion to open is uncommon with vessel access accounting for <5% of the cases. Sometimes a large sheath is accidentally pulled out and a cutdown becomes necessary to repair the artery. Closure devices aren't 100% effective in hemostasis and may also require a cutdown for definitive control if they fail, especially when obtaining large bore access.

This makes choosing the right vessel critical. For example, if a large sheath is accidentally pulled out of the CFA during an EVAR, the repair can be done through a straightforward groin cutdown. In contrast, the subclavian artery is rarely accessed percutaneously because converting to open would require a more challenging peri-clavicular incision or even a thoracotomy for repair.

Large diameter sheaths are often used, particularly in aortic procedures. These sheaths can be occlusive which can result in downstream tissue ischemia. What considerations should be taken when thinking about downstream tissue ischemia?

When performing diagnostic procedures using small diameter sheaths and catheters, anticoagulation may or may not be necessary depending on how diseased the access vessel is.

However, when using large devices (e.g. in EVARs), the sheaths can be partially or completely occlusive which mandates full anticoagulation to prevent thrombosis. The other thing to consider is the length of time that the sheath remains in the vessel as the leg can only tolerate ischemia for 4-6 hours. This is usually pertinent when performing complex endovascular aortic procedures.

To minimize downstream tissue ischemia, a large bore sheath should be pulled back to decrease the length of vessel obstruction by its shaft in order to unblock proximal vessel collateral branch vessels. For instance, when performing an aortic procedure through a femoral access attempt to pull the sheath back into the external iliac artery to increase distal limb perfusion through the internal to femoral artery collateral branch vessels.

The long story short is to be liberal with anticoagulation when there is reduced flow in the vessel such as the iliofemoral system during EVAR or tibial access
Do the principles that we've described also apply to veins?

The same principles apply but there are some notable differences between arterial and venous access.

Veins are a low-pressure system, so hemostasis is easier to achieve and hemorrhagic complications are much less common. However, this poses a challenge during access as there is less radial force keeping the vein open making the vein more susceptible to compression by the ultrasound probe and the needle.

If a large bore sheath is necessary to perform a venous procedure, a suture-mediated closure device can be utilized to achieve hemostasis especially in patients that will be kept fully anticoagulated

Additionally, a syringe may be needed to confirm access and can also prevent air embolism

Let's talk about accessing the common femoral artery. Why is the CFA the most common vessel used for access?

It is large caliber and can accommodate large sheaths up to 26-28 Fr. It also allows for a wide set of procedures and is ergonomically easy to work with given its location. It is relatively easy to hold manual pressure and if a conversion to open is needed, a femoral cutdown is relatively straightforward.

Where in the common femoral artery is the best spot to access?

The ideal puncture site is in the CFA in the medial third of the femoral head in between the inguinal ligament and the femoral bifurcation in the middle of the femoral head.

Accessing the vessel above the inguinal ligament makes compressing the artery very difficult which can lead to life-threatening retroperitoneal bleed.

A puncture that is too distal and into the SFA increase the risk of thrombosis or dissection causing acute limb ischemia as well as AV fistula formation between the superficial femoral and profunda femoris artery.

What are the different ways to obtain CFA access?

There are three different ways to access the CFA: manual palpation, fluoroscopic guided, and ultrasound guided.

With manual palpation, a finger is placed above and below the desired access point directly on the pulse and the needle is inserted in between the two fingers.

Fluoroscopic guidance uses bony landmarks relative to the position of the needle.

The standard of care in the modern era for obtaining CFA access is to use ultrasound guidance. Ultrasound allows visualization of the vessel and surrounding structures. PAD within the vessel can readily be identified with ultrasound, allowing safe access in a relatively disease-free part of the artery. Ultrasound also clearly shows the femoral bifurcation. Using ultrasound allows for subtle corrections in the angle of the needle and how it interacts with the surrounding tissues. It is rapid, real-time, inexpensive, and safe.(Sorrentino et al. 2020)

What anatomic considerations should be taken when accessing the CFA?

The CFA is the continuation of the external iliac artery as it courses under the inguinal ligament. It is about 5-8 cm in length and then bifurcates in to the superficial femoral and profunda femoris arteries

The inguinal ligament is a good external landmark to estimate where the CFA is. It is critical to emphasize that the inguinal ligament does not correspond to the groin crease and this is especially true in obese patients. An imaginary line is drawn from the ASIS to the pubic tubercle. The artery generally runs a third of the way from the pubic tubercle to the ASIS. A metallic instrument can be placed in this area to mark it externally and a fluoroscopic image can be obtained to identify the relation of the instrument to the medial third of the femoral head. This imaginary line also marks the superior-most extent of the access

The CFA is most often accessed in a retrograde fashion in between the inguinal ligament and femoral bifurcation. This allows for a multitude of potential diagnostic and therapeutic procedures in most parts of the body.

Can the CFA be accessed antegrade?

Yes. Sometimes antegrade CFA access is used when performing an intervention distal on the ipsilateral leg. The advantage of antegrade access is better pushability and torquability of wires, catheters, and sheaths when performing complex peripheral intervention where no other proximal procedures are needed.

Antegrade access is more challenging than retrograde access, however. This is particularly true in patients with a very short CFA, short distance between the inguinal ligament and the femoral bifurcation because the needle requires a steeper angle of entry to allow for cannulation well above the femoral bifurcation.

Obtaining antegrade access is especially difficult in obese patients and will usually require an assistant to retract the pannus to allow proper needle placement. I would say antegrade access is relatively contraindicated in morbidly obese patients with large pannus. Ultrasound guidance remains key here as well.

What are some other commonly accessed arteries for endovascular procedures?

The tibial vessels can be accessed percutaneously for retrograde recanalization for severe LE PAD. It is usually performed using micropuncture kits which we will discuss a little later. It is usually done with ultrasound guidance and uses small sheaths and wires. The PT and AT are more commonly used because they are easier to access.

The radial artery is commonly used in coronary interventions and is increasingly being used by vascular surgeons. It is easily palpable over the distal radius and can be cannulated with ease. Hemostasis is straightforward using compression. In the rare setting of radial occlusion, the

hand rarely becomes ischemic because most people are ulnar dominant. It can accommodate sheaths up to 6 French.

• The best way to reduce vasospasm and injury to the radial artery is utilizing a radial artery cocktail - calcium channel blocker, vasodilator, and anticoagulation.(Mason et al. 2018; Cauley et al. 2019)

The brachial artery can be accessed percutaneously over the olecranon process with the arm supinated. (Alvarez-Tostado et al. 2009) Ultrasound guidance allows for visualization of the brachial bifurcation. It can accommodate 6-7 Fr sheaths. Hemostasis is critically important as bleeding can result in a hematoma that results in median nerve compression, which is a surgical emergency. Cut down after brachial access has been shown to reduce complications. (Kret et al. 2016)

Let's not forget about venous access. What are some of the most commonly accessed veins?

The CFV is commonly accessed for procedures involving the IVC and iliac vessels and their branches for conditions such as May-Thurner, pelvic congestion syndrome, and IVC filter placement. Treatment of PE can also be performed through the CFV. The CFV can easily be compressed over the femoral head and is located medial to the CFA. Ultrasound guidance should be used to prevent arterial injury and backwalling.

The internal jugular vein can be accessed using US guidance (to prevent carotid injury; IJ is lateral to the carotid). IJ access is used most commonly for central venous catheters as well as IVC placement and filter retrieval. It is also an excellent access to treat pulmonary embolism via thrombolysis or thrombectomy. The IJ can be utilized to perform ovarian and internal iliac vein embolization. IJ is also the preferred access to perform TIPS, which is often of less interest to vascular surgeon.

The popliteal vein can be accessed with the patient in the prone position or the distal femoral vein in the supine position to diagnose and treat DVTs of the extremity veins. Ultrasound is also helpful to avoid arterial access and especially if the vein is thrombosed

Arm veins (cephalic/basilic) can be also readily accessed for vein mapping or fistula interventions.

Let's move on to access technique. Historically, there are two types of puncture needles: single-wall and double-wall. Can you talk about the differences?

Double wall needles were commonly used back in the day for femoral access. They have an outer hollow blunt-tipped needle and an inner sharp stylet. The needle was inserted through and through the artery and the stylet removed and the blunt hollow needle pulled back until blood is returned. These aren't favored anymore because they cause unnecessary backwalling of the artery. Double wall access kits are used in treating endoleaks from both a transcaval and translumbar routes to allow access into the aneurysm sack and needle removal to avoid puncturing the endograft. Single wall needles are typically the choice for diagnostic procedures. 18-gauge needles accommodate an 0.035 in wire and 21 gauge accommodates a 0.018 in wire.

Can you describe the micropuncture technique for percutaneous access?

Micropuncture technique is the most commonly used method for percutaneous access nowadays. The advantage of the micropuncture technique is the use of a small needle which can be removed and repositioned with a negligible risk of bleeding and minimal amount of manual compression needed.

Ultrasound is used to cannulate the artery with a 21-gauge needle. It is best to visualize the needle entering the artery and to be intraluminal without being against the wall Blood return is then seen and a floppy tip micropuncture (0.018) wire is inserted under fluoroscopic guidance to make sure the wire passes into the vessel easily. A 4 Fr introducer sheath is placed over the wire gently to avoid kinking the wire. The inner cannula of the sheath is removed, and a 0.035 guidewire is placed under fluoroscopic guidance. It is important to remember that there are two types of introducer sheath depending on amount of subcutaneous scar tissue containing either a soft or a stiffened cannula. The 4 Fr is removed over the wire while holding manual pressure and desired sheath (usually 5 or 6 Fr) is placed for definitive access. The side port is then aspirated for arterial blood and flushed with heparinized saline.

22.1 Complications

With any invasive procedure, there are risks of complications. What are some of the complications of percutaneous vascular access?

Hematomas are the most common complication and have an incidence of about 3%. Most of these hematomas are clinically insignificant but retroperitoneal hemorrhage from a high puncture above the inguinal ligament can be life-threatening. These may require conversion to open and direct repair of the vessel or covered stent placement (especially if the puncture is above the inguinal ligament). Proximal balloon occlusion can be helpful to control hemorrhage while the vessel is being repaired.

Groin hematomas are not uncommon and are usually self-limiting. An expanding hematoma that is seen early can be treated with simple manual pressure at the bedside. If the hematoma is large and compressing surrounding structures or threatening skin integrity or if the patient is hemodynamically unstable, then surgical evacuation may be necessary

Pseudoaneurysms are an uncommon complication with an incidence of about 0.6%. Most pseudoaneurysms are treated with ultrasound-guided compression or thrombin injection. Thrombin injection requires a narrow neck into the pseudoaneurysm. If the pseudoaneurysm is >2cm, compresses surrounding structures, threatens skin integrity, or has failed thrombin injection, then surgical repair is required. (Morgan and Belli 2003; Patrick A. Stone, Campbell, and AbuRahma 2014) For a representative image, see Section 20.5.1.1

Thrombosis of the CFA is a known complication but fortunately is rare with an incidence of 0.2%. This can result from manual compression of the CFA that has severe atherosclerotic disease or prior groin reconstruction. This generally requires a cutdown, endarterectomy, thrombectomy, and patch angioplasty.

Lastly, AV fistula can form and are usually between the femoral artery and vein with an incidence of 0.5-0.9%). These usually occur from a low puncture of the CFA bifurcation or profunda. They are usually asymptomatic and detected on exam (palpable thrill) and confirmed with duplex imaging. If the fistula is small, it generally can be observed with close duplex surveillance. If it enlarges or becomes symptomatic, then repair is indicated. Covered stent grafts can be placed with minimal morbidity, making this optimal for high risk patients. Open surgery also is highly successful. Deciding between non-operative, endovascular, or open treatment is up for debate and is up to the clinical judgement of the surgeon.

23 Rapid Review

Authors: Matthew Chia and Maggie Reilly

In this chapter we discuss a number of disparate topics that are often seen on the VSITE examination in the US, and often just require rote memorization. We have included links to the relevant chapters for you to see the information in more context if needed. This chapter is not meant as a stand a lone review.

23.1 Venous Disease

Memorization CEAP classification:

- C0: no visible or palpable signs of venous disease.
- C1: telangiectasias or reticular veins.
- C2: varicose veins.
- C3: edema.
- C4a: pigmentation and eczema.
- C4b: lipodermatosclerosis and atrophie blanche.
- C5: healed venous ulcer.
- C6: active venous ulcer.

EHIT (DVT after EVLA), endothermal heat-induced thrombosis

- 1. Thrombus without propagation into the deep vein: no treatment
 - a. Peripheral to superficial epigastric vein

b. Central to superficial epigastric vein, up to and including the deep vein junction: can consider antiplatelets

2. Thrombus propagation into the adjacent deep vein but comprising <50% of the deep vein lumen: weekly surveillance

- 3. Thrombus propagation into the adjacent deep vein but comprising >50% of the deep vein lumen: therapeutic anticoagation until resolution to the SFJ
- 4. 4. Occlusive deep vein thrombus contiguous with the treated superficial vein: treat like DVT

Reasons to treat superficial venous thrombosis (fondaparaneux x 45 days)

- Within 3-5 cm of deep system (e.g., SFJ)
- > 5cm long
- Propagates with conservative management

For more see Chapter 18

23.2 Vascular Lab

23.2.1 TCD

TCD temporal window:

- Toward probe is up on waveform (MCA, ICA)
- Away from probe is down on waveform (ACA)
- PCA is bidirectional (P1 before PCommA is up, P2 after PCommA is down)

TCD orbital window:

- Ophthalmic is toward probe
- ICA in siphon is bidirectional

TCD occipital window:

• vertebral and basilar away from probe (think direction on carotid duplex)

Consider shunt during CEA if MCA velocity drops by 50% or more A doubling of MCA mean velocity suggests cerebral hyperperfusion syndrome

Lindegaard ratio 3-6 indicates spasm, > 6 indicates severe spasm. PSV in MCA relative to extracranial ICA.

For more see Section 20.3

23.2.2 Basics

Power doppler is like amplitude, good for low flow evaluation but you lose directionality (think absolute values). Good for detecting string sign, low flow in renal parenchyma, or torsion. Not useful for directionality of flow (e.g., TCD or vertebral direction), and will not alias.

Depth resolution (synonymous with axial, longitudinal range) in PW mode is limited by half the spatial pulse length (SPL). As long as the resolution between objects is less than half the SPL, objects can be discriminated (because the total distance doubles going round). Shorter SPL means you can distinguish smaller things, which means better (smaller) resolution. Shorter SPL theoretically is achieved by shorter wavelength/higher frequency, more damping, shorter pulse duration. It is not technically affected by PRF/PRP.

PRP = time between pulses = imaging depth (ID) * 13 microseconds/cm note that <math>PRF = 1/PRP, and that PRFs are generally 5-40 kHz

Maximum Doppler shift detectable = Nyquist limit = PRF/2. Remember this only applies to PW Doppler. CW Doppler has no practical limit beyond which aliasing occurs. Use caution between depth resolution (2x SPL) and Nyquist limit (1/2 PRF)

For more, see Section 20.1

23.2.3 Sizes and Velocities

Intracranial aneurysm and carotid stenosis: no need to treat if < 10 mm



(Chaer et al. 2020)

AAA screening

- In any pt with suspected or confirmed AAA, physical exam of femoral + popliteal pulses
- Any pt with femoral or popliteal aneurysm should undergo AAA screening

• Men or women 65-75 years w/ h/o tobacco use (SVS recommendation)

AAA surveillance

- 2.5-3cm: 10 years
- 3-3.9cm: 3 years
- 4-4.9cm: 1 year
- 5-5.4cm: 6 months

Carotid velocities

- $\bullet ~> 70\%$
 - PSV > 230
 - EDV > 100
 - ICA/CCA > 4.0
- $\bullet ~> 50\%$
 - PSV > 125
 - EDV > 40
 - ICA /CCA > 2.0

CMI velocities

- CA PSV > 200
- SMA PSV > 275

Renal stenosis criteria

- PSV > 200 (some advocate for PSV > 285)
- RAR > 3.5, but a ortic velocity should be 40-100 $\rm cm/s$
- Parenchymal disease is RI > 0.7
 - (PSV-EDV)/PSV

23.3 Zebras

Indications for surgical management of vertebral disease

- At least 60% stenosis
 - In both arteries if patent
 - In one artery if contra is hypoplastic, occluded, or terminates in PICA
- If low flow: other etiologies have been ruled out (arrhythmia, emboli, inner ear dysfunction, electrolyte imbalance, intracranial tumor, hypotension)
- Posterior embolism should be considered for treatment
- Location
 - V1 can be treated by transposition onto CCA or vein bypass
 - Proximal vertebral artery reconstruction is associate with Horner's syndrome
 - V2 not usually accessible, if uncontrolled traumatic hemorrhage, ligate at V1+V3
 - V3 disease described treatments include by bypass, transposition of ECA or occipital artery, or transposition onto ICA

For more, see Section 1.5.2

Surgical exposure of supra-aortic vessels

- Median sternotomy for: innominate, proximal R SCA, R CCA, and L CCA
- L trapdoor (L anterolateral thoracotomy and supraclavicular incision) for proximal L SCA
 - Watch for L vagus n. and thoracic duct in chest, and watch for phrenic n. on anterior scalene

MALS: compression with EXPiration, normalization with INSPiration

Electrolyte abnormality with re-feeding

- Phos down (ATP)
- K, Mg down (taken intracellular for anabolism)
 - Mg down can lead to Ca down
- Torsades can be caused by low K, Mg, or Ca > treat with Mg

Popliteal artery disease

- Baker's cyst vs cystic adventitial disease (knee flexion)
 - Baker's cyst is contiguous with joint space
 - Cystic adventitial disease is within the arterial wall
- Popliteal entrapment (passive dorsiflexion or active plantar flexion)
- Popliteal artery aneurysms
 - If patients present with PAA, need screening for AAA
 - Above 2 cm should undergo repair, unless high clinical risk may wait until 3 cm
 - Can also consider repair at smaller sizes if high embolic risk
 - If life expectancy > 5 years, should undergo open repair
 - If acutely thrombosed
 - I + IIa should undergo lysis if tibial outflow bad
 - IIb should undergo thrombectomy of some fashion
 - III undergo primary amputation

For more, see Section 17.4

HITT

- 4 Ts
 - Thrombocytopenia Platelets fall > 50% and are less than 20k
 - Timing Clear onset between 5-10 days or > 1 day w/ previous heparin exposure
 - New Thrombosis
 - No other cause for Thrombocytopenia
- Bivalirudin if hepatic failure
- Argatroban for renal failure (OK for hepatic fail, but needs titration)

Obturator foramen

- Valentine/Wind
 - Nerve and vessels traverse the foramen superolateral corner
 - Tunnel should be created "centrally" by incising the medial portion of fascia
- Chaikof/Cambria

- Nerve and vessels traverse posterolaterally
- Tunnel should be created anteromedially
- Rutherford
 - Nerve and vessels are posterolateral
 - Tunnel should be created anteromedially

References

- Abou Ali, Adham N., Karim M. Salem, Louis H. Alarcon, Graciela Bauza, Emmanuel Pikoulis, Rabih A. Chaer, and Effhymios D. Avgerinos. 2017. "Vascular Shunts in Civilian Trauma." *Frontiers in Surgery* 4: 39. https://doi.org/10.3389/fsurg.2017.00039.
- Abou-Chebl, Alex, Jay S Yadav, Joel P Reginelli, Christopher Bajzer, Deepak Bhatt, and Derk W Krieger. 2004. "Intracranial Hemorrhage and Hyperperfusion Syndrome Following Carotid Artery Stenting: Risk Factors, Prevention, and Treatment." Journal of the American College of Cardiology 43 (9): 1596–1601. https://doi.org/10.1016/j.jacc.2003.12.039.
- Aboyans, Victor, Michael H. Criqui, Pierre Abraham, Matthew A. Allison, Mark A. Creager, Curt Diehm, F. Gerry R. Fowkes, et al. 2012. "Measurement and Interpretation of the Ankle-Brachial Index: A Scientific Statement From the American Heart Association." *Circulation* 126 (24): 2890–2909. https://doi.org/10.1161/CIR.0b013e318276fbcb.
- Aboyans, Victor, Ileana Desormais, Philippe Lacroix, Johanna Salazar, Michael H. Criqui, and Marc Laskar. 2010. "The General Prognosis of Patients With Peripheral Arterial Disease Differs According to the Disease Localization." Journal of the American College of Cardiology 55 (9): 898–903. https://doi.org/10.1016/j.jacc.2009.09.055.
- Aboyans, Victor, Jean-Baptiste Ricco, Marie-Louise E L Bartelink, Martin Björck, Marianne Brodmann, Tina Cohnert, Jean-Philippe Collet, et al. 2018. "2017 ESC Guidelines on the Diagnosis and Treatment of Peripheral Arterial Diseases, in Collaboration with the European Society for Vascular Surgery (ESVS)." European Heart Journal 39 (9): 763–816. https://doi.org/10.1093/eurheartj/ehx095.
- Abraha, Iosief, Maria Laura Luchetta, Rita De Florio, Francesco Cozzolino, Giovanni Casazza, Piergiorgio Duca, Basso Parente, et al. 2017. "Ultrasonography for Endoleak Detection After Endoluminal Abdominal Aortic Aneurysm Repair." The Cochrane Database of Systematic Reviews 6 (June): CD010296. https://doi.org/10.1002/14651858.CD010296.pub2.
- Abularrage, Christopher J., Robert S. Crawford, Mark F. Conrad, Hang Lee, Christopher J. Kwolek, David C. Brewster, Richard P. Cambria, and Glenn M. Lamuraglia. 2010. "Preoperative Variables Predict Persistent Type 2 Endoleak After Endovascular Aneurysm Repair." Journal of Vascular Surgery 52 (1): 19–24. https://doi.org/10.1016/j.jvs.2010.02.023.
- AbuRahma, Ali F., Efthymios D. Avgerinos, Robert W. Chang, R. Clement Darling, Audra A. Duncan, Thomas L. Forbes, Mahmoud B. Malas, et al. 2022. "Society for Vascular Surgery Clinical Practice Guidelines for Management of Extracranial Cerebrovascular Disease." Journal of Vascular Surgery 75 (1): 4S-22S. https://doi.org/10.1016/j.jvs.2021.04.073.
- Aburahma, Ali F., and Bruce A. Perler. 2022. Noninvasive Vascular Diagnosis: A Practical Textbook for Clinicians.

- AbuRahma, Ali F., Patrick A. Robinson, and Tucker G. Jennings. 2000. "Carotid-Subclavian Bypass Grafting with Polytetrafluoroethylene Grafts for Symptomatic Subclavian Artery Stenosis or Occlusion: A 20-Year Experience." *Journal of Vascular Surgery* 32 (3): 411–19. https://doi.org/10.1067/mva.2000.108644.
- AbuRahma, Ali F., Patrick A. Stone, Mohit Srivastava, L. Scott Dean, Tammi Keiffer, Stephen M. Hass, and Albeir Y. Mousa. 2012. "Mesenteric/Celiac Duplex Ultrasound Interpretation Criteria Revisited." *Journal of Vascular Surgery* 55 (2): 428-436.e6; discussion 435-436. https://doi.org/10.1016/j.jvs.2011.08.052.
- Acosta, Stefan, and Martin Björck. 2019. "136. Mesenteric Vascular Disease : Venous Thrombosis." In Rutherford's Vascular Surgery and Endovascular Therapy. Vol. 2. Elsevier Inc.
- Adam, D. J., J. D. Beard, T. Cleveland, J. Bell, A. W. Bradbury, J. F. Forbes, F. G. R. Fowkes, et al. 2005. "Bypass Versus Angioplasty in Severe Ischaemia of the Leg (BASIL): Multicentre, Randomised Controlled Trial." *Lancet (London, England)* 366 (9501): 1925–34. https://doi.org/10.1016/S0140-6736(05)67704-5.
- Adler, A. I., I. M. Stratton, H. A. Neil, J. S. Yudkin, D. R. Matthews, C. A. Cull, A. D. Wright, R. C. Turner, and R. R. Holman. 2000. "Association of Systolic Blood Pressure with Macrovascular and Microvascular Complications of Type 2 Diabetes (UKPDS 36): Prospective Observational Study." BMJ (Clinical Research Ed.) 321 (7258): 412–19. https://doi.org/10.1136/bmj.321.7258.412.
- Adler, Amanda I., Richard J. Stevens, Andrew Neil, Irene M. Stratton, Andrew J. M. Boulton, Rury R. Holman, and for the U.K. Prospective Diabetes Study Group. 2002. "UKPDS 59: Hyperglycemia and Other Potentially Modifiable Risk Factors for Peripheral Vascular Disease in Type 2 Diabetes." *Diabetes Care* 25 (5): 894–99. https://doi.org/10.2337/ diacare.25.5.894.
- Afifi, Rana O., Akiko Tanaka, Ibrahim Yazji, Hazim J. Safi, and Anthony L. Estrera. 2017. "Thoracoabdominal Aortic Aneurysm Repair in Marfan Syndrome: How We Do It." Annals of Cardiothoracic Surgery 6 (6): 709–11. https://doi.org/10.21037/acs.2017.11.04.
- Agarwal, Anil K. 2015. "Endovascular Interventions for Central Vein Stenosis." Kidney Research and Clinical Practice 34 (4): 228–32. https://doi.org/10.1016/j.krcp.2015.10.005.
- Agnelli, Giancarlo, Cecilia Becattini, Guy Meyer, Andres Muñoz, Menno V. Huisman, Jean M. Connors, Alexander Cohen, et al. 2020. "Apixaban for the Treatment of Venous Thromboembolism Associated with Cancer." The New England Journal of Medicine 382 (17): 1599–1607. https://doi.org/10.1056/NEJMoa1915103.
- Ahlborg, Gunvor, Alexey Shemyakin, Felix Böhm, Adrian Gonon, and John Pernow. 2007. "Dual Endothelin Receptor Blockade Acutely Improves Insulin Sensitivity in Obese Patients with Insulin Resistance and Coronary Artery Disease." *Diabetes Care* 30 (3): 591–96. https://doi.org/10.2337/dc06-1978.
- Aiello, Francesco, and Nicholas J. Morrissey. 2011. "Open and Endovascular Management of Subclavian and Innominate Arterial Pathology." Seminars in Vascular Surgery 24 (1): 31–35. https://doi.org/10.1053/j.semvascsurg.2011.04.001.
- Aitken, Emma, Andrew Jackson, Rachel Kearns, Mark Steven, John Kinsella, Marc Clancy, and Alan Macfarlane. 2016. "Effect of Regional Versus Local Anaesthesia on Outcome

After Arteriovenous Fistula Creation: A Randomised Controlled Trial." Lancet (London, England) 388 (10049): 1067–74. https://doi.org/10.1016/S0140-6736(16)30948-5.

- Aiyagari, Venkatesh, and Philip B. Gorelick. 2009. "Management of Blood Pressure for Acute and Recurrent Stroke." Stroke 40 (6): 2251–56. https://doi.org/10.1161/STROKEAHA. 108.531574.
- Akar, Ahmet Rüçhan. 2019. "138 Thromboangiitis Obliterans (Buerger Disease)." In Rutherford's Vascular Surgery and Endovascular Therapy, 27. Elsevier Inc.
- Akker, P. J. van den, R. van Schilfgaarde, R. Brand, J. Hajo van Bockel, and J. L. Terpstra. 1992. "Long-Term Results of Prosthetic and Non-Prosthetic Reconstruction for Obstructive Aorto-Iliac Disease." *European Journal of Vascular Surgery* 6 (1): 53–61. https://doi.org/ 10.1016/s0950-821x(05)80095-7.
- Al Shakarchi, J., J. G. Houston, R. G. Jones, and N. Inston. 2015. "A Review on the Hemodialysis Reliable Outflow (HeRO) Graft for Haemodialysis Vascular Access." *European Journal* of Vascular and Endovascular Surgery 50 (1): 108–13. https://doi.org/10.1016/j.ejvs.2015. 03.059.
- Al Shakarchi, Julien, and Nicholas Inston. 2019. "Early Cannulation Grafts for Haemodialysis: An Updated Systematic Review." The Journal of Vascular Access 20 (2): 123–27. https: //doi.org/10.1177/1129729818776571.
- Al Shakarchi, Julien, Jan Stolba, J. Graeme Houston, and Nicholas Inston. 2016. "Surgical Techniques for Haemodialysis Access-Induced Distal Ischaemia." The Journal of Vascular Access 17 (1): 40–46. https://doi.org/10.5301/jva.5000467.
- Alarhayem, Abdul Q., Todd E. Rasmussen, Behzad Farivar, Sungho Lim, Max Braverman, David Hardy, Donald J. Jenkins, Brian J. Eastridge, and Ramon F. Cestero. 2021. "Timing of Repair of Blunt Thoracic Aortic Injuries in the Thoracic Endovascular Aortic Repair Era." *Journal of Vascular Surgery* 73 (3): 896–902. https://doi.org/10.1016/j.jvs.2020.05.079.
- Al-Jaishi, Ahmed A., Aiden R. Liu, Charmaine E. Lok, Joyce C. Zhang, and Louise M. Moist. 2017. "Complications of the Arteriovenous Fistula: A Systematic Review." Journal of the American Society of Nephrology : JASN 28 (6): 1839–50. https://doi.org/10.1681/ASN. 2016040412.
- Allison, Matthew A., Michael H. Criqui, Robyn L. McClelland, JoAnn M. Scott, Mary M. McDermott, Kiang Liu, Aaron R. Folsom, et al. 2006. "The Effect of Novel Cardiovascular Risk Factors on the Ethnic-Specific Odds for Peripheral Arterial Disease in the Multi-Ethnic Study of Atherosclerosis (MESA)." Journal of the American College of Cardiology 48 (6): 1190–97. https://doi.org/10.1016/j.jacc.2006.05.049.
- Alqahtani, Khalid M., Munveer Bhangoo, Florin Vaida, Julie O. Denenberg, Matthew A. Allison, and Michael H. Criqui. 2018. "Predictors of Change in the Ankle-Brachial Index with Exercise." European Journal of Vascular and Endovascular Surgery : The Official Journal of the European Society for Vascular Surgery 55 (3): 399–404. https://doi.org/10. 1016/j.ejvs.2017.12.004.
- Al-Thani, Hassan, Ayman El-Menyar, Noora Al-Thani, Mohammad Asim, Ahmed Hussein, Ahmed Sadek, Ahmed Sharaf, and Amr Fares. 2017. "Characteristics, Management, and Outcomes of Surgically Treated Arteriovenous Fistula Aneurysm in Patients on Regular Hemodialysis." Annals of Vascular Surgery 41 (May): 46–55. https://doi.org/10.1016/j.

avsg.2016.08.046.

- Alvarez-Tostado, Javier A., Mireille A. Moise, James F. Bena, Mircea L. Pavkov, Roy K. Greenberg, Daniel G. Clair, and Vikram S. Kashyap. 2009. "The Brachial Artery: A Critical Access for Endovascular Procedures." *Journal of Vascular Surgery* 49 (2): 378–385; discussion 385. https://doi.org/10.1016/j.jvs.2008.09.017.
- Anand, Sonia S., Francois Caron, John W. Eikelboom, Jackie Bosch, Leanne Dyal, Victor Aboyans, Maria Teresa Abola, et al. 2018. "Major Adverse Limb Events and Mortality in Patients with Peripheral Artery Disease: The COMPASS Trial." Journal of the American College of Cardiology 71 (20): 2306–15. https://doi.org/10.1016/j.jacc.2018.03.008.
- Ananthan, K., S. Onida, and A. H. Davies. 2017. "Nutcracker Syndrome: An Update on Current Diagnostic Criteria and Management Guidelines." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 53 (6): 886–94. https://doi.org/10.1016/j.ejvs.2017.02.015.
- Anaya-Ayala, Javier E., Christopher J. Smolock, Benjamin D. Colvard, Joseph J. Naoum, Jean Bismuth, Alan B. Lumsden, Mark G. Davies, and Eric K. Peden. 2011. "Efficacy of Covered Stent Placement for Central Venous Occlusive Disease in Hemodialysis Patients." *Journal of Vascular Surgery* 54 (3): 754–59. https://doi.org/10.1016/j.jvs.2011.03.260.
- Angiletta, Domenico, Davide Marinazzo, Gloria Guido, Luigi Greco, and Guido Regina. 2011. "Spinal Cord, Bowel, and Buttock Ischemia After Endovascular Aneurysm Repair." Annals of Vascular Surgery 25 (7): 980.e15–19. https://doi.org/10.1016/j.avsg.2010.12.044.
- Ansari, Naheed. 2011. "Peritoneal Dialysis in Renal Replacement Therapy for Patients with Acute Kidney Injury." International Journal of Nephrology 2011 (June): 739794. https: //doi.org/10.4061/2011/739794.
- Anthenelli, Robert M., Neal L. Benowitz, Robert West, Lisa St Aubin, Thomas McRae, David Lawrence, John Ascher, Cristina Russ, Alok Krishen, and A. Eden Evins. 2016. "Neuropsychiatric Safety and Efficacy of Varenicline, Bupropion, and Nicotine Patch in Smokers with and Without Psychiatric Disorders (EAGLES): A Double-Blind, Randomised, Placebo-Controlled Clinical Trial." Lancet (London, England) 387 (10037): 2507–20. https: //doi.org/10.1016/S0140-6736(16)30272-0.
- Antoniou, G. A., M. K. Lazarides, G. S. Georgiadis, G. S. Sfyroeras, E. S. Nikolopoulos, and A. D. Giannoukas. 2009. "Lower-Extremity Arteriovenous Access for Haemodialysis: A Systematic Review." *European Journal of Vascular and Endovascular Surgery* 38 (3): 365–72. https://doi.org/10.1016/j.ejvs.2009.06.003.
- Antoniou, George A., David Murray, George S. Georgiadis, Stavros A. Antoniou, Andrew Schiro, Ferdinand Serracino-Inglott, and J. Vincent Smyth. 2012. "Percutaneous Transluminal Angioplasty and Stenting in Patients with Proximal Vertebral Artery Stenosis." *Journal of Vascular Surgery* 55 (4): 1167–77. https://doi.org/10.1016/j.jvs.2011.09.084.
- Arbique, Gary M. n.d. "The Fluoroscopic Sentinel Event: What to Do?" 3 (3): 13.
- Arhuidese, Isibor J., Besma Nejim, Susruth Chavali, Satinderjit Locham, Tammam Obeid, Caitlin W. Hicks, and Mahmoud B. Malas. 2017. "Endarterectomy Versus Stenting in Patients with Prior Ipsilateral Carotid Artery Stenting." Journal of Vascular Surgery 65 (5): 1418–28. https://doi.org/10.1016/j.jvs.2016.11.041.
- Arko, Frank R., Konstantinos A. Filis, Scott A. Siedel, Bonnie L. Johnson, Angelia R. Drake,

Thomas J. Fogarty, and Christopher K. Zarins. 2003. "Intrasac Flow Velocities Predict Sealing of Type II Endoleaks After Endovascular Abdominal Aortic Aneurysm Repair." *Journal of Vascular Surgery* 37 (1): 8–15. https://doi.org/10.1067/mva.2003.55.

- Armstrong, David G., Andrew J. M. Boulton, and Sicco A. Bus. 2017. "Diabetic Foot Ulcers and Their Recurrence." The New England Journal of Medicine 376 (24): 2367– 75. https://doi.org/10.1056/NEJMra1615439.
- Armstrong, David L, Timothy K Fisher, Brian Lepow, Matthew L White, and Joseph L Mills. 2012. "26Pathophysiology and Principles of Management of the Diabetic Foot." In Mechanisms of Vascular Disease: A Reference Book for Vascular Specialists. Adelaide: University of Adelaide Press. https://doi.org/10.1017/UPO9781922064004.
- Arnold, Maggie, and Bruce A. Perler. 2019. "91. Carotid Endarterectomy." In Rutherford's Vascular Surgery and Endovascular Therapy. Elsevier Inc.
- Arvela, Eva, Maria Söderström, Anders Albäck, Pekka-Sakari Aho, Maarit Venermo, and Mauri Lepäntalo. 2010. "Arm Vein Conduit Vs Prosthetic Graft in Infrainguinal Revascularization for Critical Leg Ischemia." Journal of Vascular Surgery 52 (3): 616–23. https://doi.org/10.1016/j.jvs.2010.04.013.
- Arya, Shipra, Anjali Khakharia, Zachary O. Binney, Randall R. DeMartino, Luke P. Brewster, Philip P. Goodney, and Peter W. F. Wilson. 2018. "Statins Have a Dose-Dependent Effect on Amputation and Survival in Peripheral Artery Disease Patients." *Circulation* 137 (14): 1435–46. https://doi.org/10.1161/CIRCULATIONAHA.117.032361.
- Ascher, Enrico, Natalia Markevich, Sreedhar Kallakuri, Richard W. Schutzer, and Anil P. Hingorani. 2004. "Intraoperative Carotid Artery Duplex Scanning in a Modern Series of 650 Consecutive Primary Endarterectomy Procedures." Journal of Vascular Surgery 39 (2): 416–20. https://doi.org/10.1016/j.jvs.2003.09.019.
- Asghar, M., K. Ahmed, S. S. Shah, M. K. Siddique, P. Dasgupta, and M. S. Khan. 2007. "Renal Vein Thrombosis." *European Journal of Vascular and Endovascular Surgery* 34 (2): 217–23. https://doi.org/10.1016/j.ejvs.2007.02.017.
- Atasoy, Mehmet Mahir. 2015. "Efficacy and Safety of Endovenous Laser Ablation in Very Large and Tortuous Great Saphenous Veins." Journal of Vascular and Interventional Radiology: JVIR 26 (9): 1347–52. https://doi.org/10.1016/j.jvir.2015.04.015.
- Avgerinos, Efthymios D., Ulka Sachdev, Abdallah Naddaf, Dannielle R. Doucet, Abhisekh Mohapatra, Steven A. Leers, Rabih A. Chaer, and Michel S. Makaroun. 2015. "Autologous Alternative Veins May Not Provide Better Outcomes Than Prosthetic Conduits for Below-Knee Bypass When Great Saphenous Vein Is Unavailable." *Journal of Vascular Surgery* 62 (2): 385–91. https://doi.org/10.1016/j.jvs.2015.03.025.
- Aylwin, Christopher, and Michael Jenkins. 2022. "18. Abdominal Aortic Trauma, Iliac and Visceral Vessel Injuries." In, 212–25. Elsevier. https://doi.org/10.1016/B978-0-323-69766-8.00018-4.
- Azizzadeh, Ali, Kourosh Keyhani, Charles C. Miller, Sheila M. Coogan, Hazim J. Safi, and Anthony L. Estrera. 2009. "Blunt Traumatic Aortic Injury: Initial Experience with Endovascular Repair." Journal of Vascular Surgery 49 (6): 1403–8. https://doi.org/10.1016/ j.jvs.2009.02.234.
- Azizzadeh, Ali, Jaime Valdes, Charles C. Miller, Louis L. Nguyen, Anthony L. Estrera,

Kristofer Charlton-Ouw, Sheila M. Coogan, John B. Holcomb, and Hazim J. Safi. 2011. "The Utility of Intravascular Ultrasound Compared to Angiography in the Diagnosis of Blunt Traumatic Aortic Injury." *Journal of Vascular Surgery* 53 (3): 608–14. https://doi.org/10.1016/j.jvs.2010.09.059.

- Bacharach, J. Michael, and David P. Slovut. 2008. "State of the Art: Management of Iliac Artery Aneurysmal Disease." Catheterization and Cardiovascular Interventions: Official Journal of the Society for Cardiac Angiography and Interventions 71 (5): 708–14. https: //doi.org/10.1002/ccd.21507.
- Bagir, Melih, Emrah Sayit, and Asli Tanrivermis Sayit. 2017. "Pseudoaneurysm of the Radial Artery on the Hand Secondary to Stabbing." Annals of Vascular Surgery 41 (May): 280.e7– 10. https://doi.org/10.1016/j.avsg.2016.09.027.
- Bahl, Vinita, Hsou Mei Hu, Peter K. Henke, Thomas W. Wakefield, Darrell A. Campbell, and Joseph A. Caprini. 2010. "A Validation Study of a Retrospective Venous Thromboembolism Risk Scoring Method." Annals of Surgery 251 (2): 344–50. https://doi.org/ 10.1097/SLA.0b013e3181b7fca6.
- Baig, m shadman, and Carlos H. Timaran. 2019. "84. Upper Extremity Aneurysms." In Rutherford's Vascular Surgery and Endovascular Therapy. Elsevier Inc.
- Bailey, Charles J., Seth Force, Ross Milner, Karthikeshwar Kasirajan, and Ravi K. Veeraswamy. 2011. "Thoracic Endovascular Repair as a Safe Management Strategy for Aortobronchial Fistulas." Journal of Vascular Surgery 53 (5): 1202–1209; discussion 1209. https://doi. org/10.1016/j.jvs.2010.10.103.
- Baiter, Stephen, Marvin Rosenstein, Donald L. Miller, Beth Schueler, and David Spelic. 2011. "Patient Radiation Dose Audits for Fluoroscopically Guided Interventional Procedures." *Medical Physics* 38 (3): 1611–18. https://doi.org/10.1118/1.3557868.
- Balderman, Joshua, Ahmmad A. Abuirqeba, Lindsay Eichaker, Cassandra Pate, Jeanne A. Earley, Michael M. Bottros, Senthil N. Jayarajan, and Robert W. Thompson. 2019. "Physical Therapy Management, Surgical Treatment, and Patient-Reported Outcomes Measures in a Prospective Observational Cohort of Patients with Neurogenic Thoracic Outlet Syndrome." Journal of Vascular Surgery 70 (3): 832–41. https://doi.org/10.1016/j.jvs.2018.12.027.
- Ball, E. L., S. R. Walsh, T. Y. Tang, R. Gohil, and J. M. F. Clarke. 2010. "Role of Ultrasonography in the Diagnosis of Temporal Arteritis." *The British Journal of Surgery* 97 (12): 1765–71. https://doi.org/10.1002/bjs.7252.
- Ballotta, Enzo, Gaetano Thiene, Claudio Baracchini, Mario Ermani, Carmelo Militello, Giuseppe Da Giau, Bruno Barbon, and Annalisa Angelini. 2005. "Surgical Vs Medical Treatment for Isolated Internal Carotid Artery Elongation with Coiling or Kinking in Symptomatic Patients: A Prospective Randomized Clinical Study." Journal of Vascular Surgery 42 (5): 838–46. https://doi.org/10.1016/j.jvs.2005.07.034.
- Balter, Stephen, John W. Hopewell, Donald L. Miller, Louis K. Wagner, and Michael J. Zelefsky. 2010. "Fluoroscopically Guided Interventional Procedures: A Review of Radiation Effects on Patients' Skin and Hair." *Radiology* 254 (2): 326–41. https://doi.org/10.1148/ radiol.2542082312.
- Bandyk, Dennis F. 2013. "Interpretation of Duplex Ultrasound Dialysis Access Testing." Seminars in Vascular Surgery 26 (2-3): 120–26. https://doi.org/10.1053/j.semvascsurg.2013.

12.001.

——. 2018. "The Diabetic Foot: Pathophysiology, Evaluation, and Treatment." Seminars in Vascular Surgery 31 (2-4): 43–48. https://doi.org/10.1053/j.semvascsurg.2019.02.001.

- Bandyk, Dennis F, Gary R Seabrook, Pamela Moldenhauer, Jane Lavin, Janis Edwards, Ruth Cato, and Jonathan B Towne. 1988. "Hemodynamics of Vein Graft Stenosis" 8 (6): 8.
- Banerjee, Subhash, Karan Sarode, Ariel Vinas, Avantika Banerjee, Atif Mohammad, and Emmanouil S. Brilakis. 2015. "The Role of Antiplatelet Therapy in Patients with Peripheral Artery Disease and Lower Extremity Peripheral Artery Revascularization." Current Opinion in Cardiology 30 (5): 525–35. https://doi.org/10.1097/HCO.00000000000208.
- Banno, Hiroshi, Shuta Ikeda, Yohei Kawai, Katsuaki Meshii, Noriko Takahashi, Masayuki Sugimoto, Akio Kodama, and Kimihiro Komori. 2020. "Early and midterm outcomes of celiac artery coverage during thoracic endovascular aortic repair." Journal of Vascular Surgery 72 (5): 1552–57. https://doi.org/10.1016/j.jvs.2020.02.025.
- Baril, Donald T., Robert Y. Rhee, Justine Kim, Michel S. Makaroun, Rabih A. Chaer, and Luke K. Marone. 2009. "Duplex Criteria for Determination of in-Stent Stenosis After Angioplasty and Stenting of the Superficial Femoral Artery." *Journal of Vascular Surgery* 49 (1): 133–39. https://doi.org/10.1016/j.jvs.2008.09.046.
- Barrons, Robert W., and J. Andrew Woods. 2016. "The Roles of ACE Inhibitors in Lower Extremity Peripheral Artery Disease." American Journal of Therapeutics 23 (1): e7–15. https://doi.org/10.1097/MJT.000000000000011.
- Bates, Shannon M., Anita Rajasekhar, Saskia Middeldorp, Claire McLintock, Marc A. Rodger, Andra H. James, Sara R. Vazquez, et al. 2018. "American Society of Hematology 2018 Guidelines for Management of Venous Thromboembolism: Venous Thromboembolism in the Context of Pregnancy." *Blood Advances* 2 (22): 3317–59. https://doi.org/10.1182/ bloodadvances.2018024802.
- Baumeister, Ruediger G. H., Weiss Mayo, Mike Notohamiprodjo, Jens Wallmichrath, Stephanie Springer, and Andreas Frick. 2016. "Microsurgical Lymphatic Vessel Transplantation." Journal of Reconstructive Microsurgery 32 (1): 34–41. https://doi.org/10.1055/s-0035-1554934.
- Bavry, Anthony A., R. David Anderson, Yan Gong, Scott J. Denardo, Rhonda M. Cooper-Dehoff, Eileen M. Handberg, and Carl J. Pepine. 2010. "Outcomes Among Hypertensive Patients with Concomitant Peripheral and Coronary Artery Disease: Findings from the INternational VErapamil-SR/Trandolapril STudy." Hypertension (Dallas, Tex.: 1979) 55 (1): 48–53. https://doi.org/10.1161/HYPERTENSIONAHA.109.142240.
- Beckman, Michele G., W. Craig Hooper, Sara E. Critchley, and Thomas L. Ortel. 2010. "Venous Thromboembolism: A Public Health Concern." *American Journal of Preventive Medicine* 38 (4 Suppl): S495–501. https://doi.org/10.1016/j.amepre.2009.12.017.
- Benjamin, Marshall E, and Kimberley J Hansen. 2019. "126: Renovascular Disease : Open Surgical Treatment." In Rutherford's Vascular Surgery and Endovascular Therapy, Ninth. Vol. 2. Elsevier Inc. https://doi.org/10.1016/B978-1-4557-5304-8.00145-X.
- Beretta, Federica, Salah A. Hemida, Norberto Andaluz, Mario Zuccarello, and Jeffrey T. Keller. 2006. "Exposure of the Cervical Internal Carotid Artery: Surgical Steps to the Cranial Base and Morphometric Study." *Neurosurgery* 59 (1 Suppl 1): ONS25-34; discussion ONS25-34.

https://doi.org/10.1227/01.NEU.0000219877.43072.49.

- Bergan, J., L. Pascarella, and L. Mekenas. 2006. "Venous Disorders: Treatment with Sclerosant Foam." The Journal of Cardiovascular Surgery 47 (1): 9–18.
- Berguer, R. 1985. "Distal Vertebral Artery Bypass: Technique, the 'Occipital Connection,' and Potential Uses." Journal of Vascular Surgery 2 (4): 621–26.
- Berguer, R., M. D. Morasch, and R. A. Kline. 1998. "Transthoracic Repair of Innominate and Common Carotid Artery Disease: Immediate and Long-Term Outcome for 100 Consecutive Surgical Reconstructions." *Journal of Vascular Surgery* 27 (1): 34–41; discussion 42. https: //doi.org/10.1016/s0741-5214(98)70289-7.
- Berguer, R., M. D. Morasch, R. A. Kline, A. Kazmers, and M. S. Friedland. 1999. "Cervical Reconstruction of the Supra-Aortic Trunks: A 16-Year Experience." *Journal of Vascular Surgery* 29 (2): 239-246; discussion 246-248. https://doi.org/10.1016/s0741-5214(99)70377-0.
- Berman, Scott S., and Andrew T. Gentile. 2001. "Impact of Secondary Procedures in Autogenous Arteriovenous Fistula Maturation and Maintenance." Journal of Vascular Surgery 34 (5): 866–71. https://doi.org/10.1067/mva.2001.118086.
- Bhatia, Shivank S., Shree Venkat, Ana Echenique, Caio Rocha-Lima, Mehul H. Doshi, Jason Salsamendi, Katuska Barbery, and Govindarajan Narayanan. 2015. "Proximal Splenic Artery Embolization in Chemotherapy-Induced Thrombocytopenia: A Retrospective Analysis of 13 Patients." Journal of Vascular and Interventional Radiology: JVIR 26 (8): 1205–11. https://doi.org/10.1016/j.jvir.2015.04.003.
- Bianchi, Christian, and Ahmed M Abou-zamzam Jr. 2018. "111. Lower Extremity Amputations: Epidemiology, Procedure Selection, and Rehabilitation Outcomes." In 111. Lower Extremity Amputations : Epidemiology, Procedure Selection, and Rehabilitation Outcomes. Elsevier Inc. https://doi.org/10.1016/B978-0-323-42791-3.00111-0.
- Biffl, Walter L., Ernest E. Moore, Patrick J. Offner, and Jon M. Burch. 2001. "Blunt Carotid and Vertebral Arterial Injuries." *World Journal of Surgery* 25 (8): 1036–43. https://doi.org/10.1007/s00268-001-0056-x.
- Bild, D. E. 2002. "Multi-Ethnic Study of Atherosclerosis: Objectives and Design." American Journal of Epidemiology 156 (9): 871–81. https://doi.org/10.1093/aje/kwf113.
- Bisdas, Theodosios, Giuseppe Panuccio, Masayuki Sugimoto, Giovanni Torsello, and Martin Austermann. 2015. "Risk Factors for Spinal Cord Ischemia After Endovascular Repair of Thoracoabdominal Aortic Aneurysms." Journal of Vascular Surgery 61 (6): 1408–16. https://doi.org/10.1016/j.jvs.2015.01.044.
- Bitzur, Rafael, Hofit Cohen, Yehuda Kamari, and Dror Harats. 2013. "Intolerance to Statins: Mechanisms and Management." *Diabetes Care* 36 Suppl 2 (August): S325–330. https://doi.org/10.2337/dcS13-2038.
- Black, James H. 2019. "140. Aneurysms Caused by Connective Tissue Abnormalities." In Rutherford's Vascular Surgery and Endovascular Therapy, 21. Elsevier Inc.
- Blauw, Juliette T. M., Robert Meerwaldt, Marjolein Brusse-Keizer, Jeroen J. Kolkman, Dick Gerrits, and Robert H. Geelkerken. 2014. "Retrograde Open Mesenteric Stenting for Acute Mesenteric Ischemia." Journal of Vascular Surgery 60 (3): 726–34. https://doi.org/ 10.1016/j.jvs.2014.04.001.

- Blessberger, Hermann, Juergen Kammler, Hans Domanovits, Oliver Schlager, Brigitte Wildner, Danyel Azar, Martin Schillinger, Franz Wiesbauer, and Clemens Steinwender. 2018. "Perioperative Beta-Blockers for Preventing Surgery-Related Mortality and Morbidity." The Cochrane Database of Systematic Reviews 3 (March): CD004476. https://doi.org/10.1002/ 14651858.CD004476.pub3.
- Blom, Aaron S., Douglas Troutman, Brian Beeman, Mark Yarchoan, Matthew J. Dougherty, and Keith D. Calligaro. 2012. "Duplex Ultrasound Imaging to Detect Limb Stenosis or Kinking of Endovascular Device." *Journal of Vascular Surgery* 55 (6): 1577–80. https: //doi.org/10.1016/j.jvs.2011.12.058.
- Blondon, Marc, Marc Righini, Henri Bounameaux, and David L. Veenstra. 2012. "Fondaparinux for Isolated Superficial Vein Thrombosis of the Legs: A Cost-Effectiveness Analysis." Chest 141 (2): 321–29. https://doi.org/10.1378/chest.11-0625.
- Bobadilla, Joseph L., John R. Hoch, Glen E. Leverson, and Girma Tefera. 2010. "The Effect of Warfarin Therapy on Endoleak Development After Endovascular Aneurysm Repair (EVAR) of the Abdominal Aorta." *Journal of Vascular Surgery* 52 (2): 267–71. https://doi.org/10. 1016/j.jvs.2010.02.290.
- Boezem, P. B. van den, T. M. a. L. Klem, E. le Cocq d'Armandville, and C. H. A. Wittens. 2011. "The Management of Superficial Venous Incompetence." *BMJ (Clinical Research Ed.)* 343 (August): d4489. https://doi.org/10.1136/bmj.d4489.
- Bogerijen, Guido H. W. van, Jip L. Tolenaar, Vincenzo Rampoldi, Frans L. Moll, Joost A. van Herwaarden, Frederik H. W. Jonker, Kim A. Eagle, and Santi Trimarchi. 2014. "Predictors of Aortic Growth in Uncomplicated Type b Aortic Dissection." *Journal of Vascular Surgery* 59 (4): 1134–43. https://doi.org/10.1016/j.jvs.2014.01.042.
- Boll, Julia M, and R. James Valentine. 2019. "122: Thoracic Outlet Syndrome: Arterial." In Rutherford's Vascular Surgery and Endovascular Therapy, Ninth. Vol. 2. Elsevier Health Sciences.
- Bongartz, Tim, and Eric L. Matteson. 2006. "Large-Vessel Involvement in Giant Cell Arteritis." Current Opinion in Rheumatology 18 (1): 10–17. https://doi.org/10.1097/01.bor. 0000197996.04709.4e.
- Boren, Clark H, Jonathan B Towne, Victor M Bernhard, and S Salles-Cunha. 1980. "Profundapopliteal Collateral Index. A Guide to Successful Profundaplasty." Archives of Surgery 115: 1366–72. https://doi.org/10.1001/archsurg.1980.01380110098015.
- Boulton, Andrew JM, Loretta Vileikyte, Gunnel Ragnarson-Tennvall, and Jan Apelqvist. 2005. "The Global Burden of Diabetic Foot Disease." *The Lancet* 366 (9498): 1719–24. https://doi.org/10.1016/S0140-6736(05)67698-2.
- Bountouris, Ioannis, Georgia Kritikou, Nikolaos Degermetzoglou, and Konstantinos Ioannis Avgerinos. 2018. "A Review of Percutaneous Transluminal Angioplasty in Hemodialysis Fistula." International Journal of Vascular Medicine 2018 (March): 1420136. https://doi. org/10.1155/2018/1420136.
- Bourquelot, Pierre, Marek Rawa, Olivier Van Laere, and Gilbert Franco. 2012. "Long-Term Results of Femoral Vein Transposition for Autogenous Arteriovenous Hemodialysis Access." *Journal of Vascular Surgery* 56 (2): 440–45. https://doi.org/10.1016/j.jvs.2012.01.068.
- Bowersox, J. C., R. M. Zwolak, D. B. Walsh, J. R. Schneider, A. Musson, F. E. LaBom-

bard, and J. L. Cronenwett. 1991. "Duplex Ultrasonography in the Diagnosis of Celiac and Mesenteric Artery Occlusive Disease." *Journal of Vascular Surgery* 14 (6): 780-786; discussion 786-788. https://doi.org/10.1067/mva.1991.33215.

- Bowyer, Mark W. 2015. "Lower Extremity Fasciotomy: Indications and Technique." Current Trauma Reports 1 (1): 35–44. https://doi.org/10.1007/s40719-014-0002-7.
- Bradaric, Christian, Kristin Kuhs, Philip Groha, Michael Dommasch, Nicolas Langwieser, Bernhard Haller, Ilka Ott, et al. 2015. "Endovascular Therapy for Steno-Occlusive Subclavian and Innominate Artery Disease: – Safety and Efficacy at a Single Center With ≫20 Years' Experience –." Circulation Journal 79 (3): 537–43. https://doi.org/10.1253/circj. CJ-14-0855.
- Bradbury, Andrew W., D. J. Adam, J. D. Beard, T. Cleveland, J. F. Forbes, F. G. R. Fowkes, I. Gillepsie, C. V. Ruckley, G. Raab, and H. Storkey. 2005. "Bypass Versus Angioplasty in Severe Ischaemia of the Leg (BASIL): Multicentre, Randomised Controlled Trial." *Lancet* 366 (9501): 1925–34. https://doi.org/10.1016/S0140-6736(05)67704-5.
- Bradshaw, Rhiannon J., S. Sadie Ahanchi, Obie Powell, Sebastian Larion, Colin Brandt, Michael C. Soult, and Jean M. Panneton. 2017. "Left Subclavian Artery Revascularization in Zone 2 Thoracic Endovascular Aortic Repair Is Associated with Lower Stroke Risk Across All Aortic Diseases." Journal of Vascular Surgery 65 (5): 1270–79. https: //doi.org/10.1016/j.jvs.2016.10.111.
- Branco, Bernardino C., Mina L. Boutrous, Joseph J. DuBose, Samuel S. Leake, Kristopher Charlton-Ouw, Peter Rhee, Joseph L. Mills, and Ali Azizzadeh. 2016. "Outcome Comparison Between Open and Endovascular Management of Axillosubclavian Arterial Injuries." *Journal of Vascular Surgery* 63 (3): 702–9. https://doi.org/10.1016/j.jvs.2015.08.117.
- Brewster, D. C., and R. C. Darling. 1978. "Optimal Methods of Aortoiliac Reconstruction." Surgery 84 (6): 739–48.
- Brewster, D. C., D. P. Franklin, R. P. Cambria, R. C. Darling, A. C. Moncure, G. M. Lamuraglia, W. M. Stone, and W. M. Abbott. 1991. "Intestinal Ischemia Complicating Abdominal Aortic Surgery." Surgery 109 (4): 447–54.
- Brightwell, Robert E., Victoria Pegna, and Nicholas Boyne. 2013. "Aortocaval Fistula: Current Management Strategies." ANZ Journal of Surgery 83 (1-2): 31–35. https://doi.org/ 10.1111/j.1445-2197.2012.06294.x.
- Brochado Neto, F., G. A. Sandri, M. J. Kalaf, M. F. Matielo, I. B. Casella, M. R. Godoy, M. V. Martins Cury, and R. Sacilotto. 2014. "Arm Vein as an Alternative Autogenous Conduit for Infragenicular Bypass in the Treatment of Critical Limb Ischaemia: A 15 Year Experience." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 47 (6): 609–14. https://doi.org/10. 1016/j.ejvs.2014.01.019.
- Bromberg, William J., Bryan C. Collier, Larry N. Diebel, Kevin M. Dwyer, Michelle R. Holevar, David G. Jacobs, Stanley J. Kurek, Martin A. Schreiber, Mark L. Shapiro, and Todd R. Vogel. 2010. "Blunt Cerebrovascular Injury Practice Management Guidelines: The Eastern Association for the Surgery of Trauma." Journal of Trauma: Injury, Infection & Critical Care 68 (2): 471–77. https://doi.org/10.1097/TA.0b013e3181cb43da.
- Brott, Thomas G., Robert W. Hobson, George Howard, Gary S. Roubin, Wayne M. Clark,

William Brooks, Ariane Mackey, et al. 2010. "Stenting Versus Endarterectomy for Treatment of Carotid-Artery Stenosis." *The New England Journal of Medicine* 363 (1): 11–23. https://doi.org/10.1056/NEJMoa0912321.

- Brott, Thomas G., George Howard, Gary S. Roubin, James F. Meschia, Ariane Mackey, William Brooks, Wesley S. Moore, et al. 2016. "Long-Term Results of Stenting Versus Endarterectomy for Carotid-Artery Stenosis." *The New England Journal of Medicine* 374 (11): 1021–31. https://doi.org/10.1056/NEJMoa1505215.
- Brown, Kellie R., and Eva Rzucidlo. 2011. "Acute and Chronic Radiation Injury." Journal of Vascular Surgery 53 (1 Suppl): 15S–21S. https://doi.org/10.1016/j.jvs.2010.06.175.
- Brownie, Evan R. 2016. "A Comparison of Patency and Interventions in Thigh Versus Hemodialysis Reliable Outflow Grafts for Chronic Hemodialysis Vascular Access." JOUR-NAL OF VASCULAR SURGERY 64 (5): 8.
- Brunkwall, J., P. Kasprzak, E. Verhoeven, R. Heijmen, P. Taylor, ADSORB Trialists, P. Alric, et al. 2014. "Endovascular Repair of Acute Uncomplicated Aortic Type b Dissection Promotes Aortic Remodelling: 1 Year Results of the ADSORB Trial." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 48 (3): 285–91. https://doi.org/10.1016/j.ejvs.2014.05.012.
- Bulas, D. I., A. M. Jones, J. J. Seibert, C. Driscoll, R. O'Donnell, and R. J. Adams. 2000. "Transcranial Doppler (TCD) Screening for Stroke Prevention in Sickle Cell Anemia: Pitfalls in Technique Variation." *Pediatric Radiology* 30 (11): 733–38. https://doi.org/10. 1007/s002470000317.
- Bulbulia, Richard, and Alison Halliday. 2017. "The Asymptomatic Carotid Surgery Trial-2 (ACST-2): An Ongoing Randomised Controlled Trial Comparing Carotid Endarterectomy with Carotid Artery Stenting to Prevent Stroke." *Health Technology Assessment* 21 (57): 1–40. https://doi.org/10.3310/hta21570.
- Bulger, Christopher M., Chad Jacobs, and Nilesh H. Patel. 2004. "Epidemiology of Acute Deep Vein Thrombosis." *Techniques in Vascular and Interventional Radiology* 7 (2): 50–54. https://doi.org/10.1053/j.tvir.2004.02.001.
- Bulger, Eileen M., David Snyder, Karen Schoelles, Cathy Gotschall, Drew Dawson, Eddy Lang, Nels D. Sanddal, et al. 2014. "An Evidence-Based Prehospital Guideline for External Hemorrhage Control: American College of Surgeons Committee on Trauma." *Prehospital Emergency Care* 18 (2): 163–73. https://doi.org/10.3109/10903127.2014.896962.
- Burkhardt, Gabriel E., Mitchell Cox, W. Darrin Clouse, Chantel Porras, Shaun M. Gifford, Ken Williams, Brandon W. Propper, and Todd E. Rasmussen. 2010. "Outcomes of Selective Tibial Artery Repair Following Combat-Related Extremity Injury." Journal of Vascular Surgery 52 (1): 91–96. https://doi.org/10.1016/j.jvs.2010.02.017.
- Busch, Kathryn. 2011. "Buerger's Disease (Thromboangiitis Obliterans): Clinical Features and Assessment by Colour Duplex Ultrasound." Australasian Journal of Ultrasound in Medicine 14 (4): 18–22. https://doi.org/10.1002/j.2205-0140.2011.tb00126.x.
- Busch, Kathryn J., Geoffrey H. White, James May, John P. Harris, Judith Doyle, Martin R. Forbes, Virginia Makeham, and Alison Burnett. 2009. "0230: Ultrasound Detection of Intermittent and Positional Dependent Endoleaks; Evidence for a Novel Mehanism of AAA Sac Growth After Endovascular Repair." Ultrasound in Medicine and Biology 35 (8):

S23. https://doi.org/10.1016/j.ultrasmedbio.2009.06.093.

- Bush, Ruth L, Paul Long, and Marvin D Atkins. 2019. "94. Carotid Artery Aneurysms." In Rutherford's Vascular Surgery and Endovascular Therapy, 2-Volume Set, 1242–1254e. Elsevier Inc. https://doi.org/10.1016/B978-0-323-42791-3.00094-3.
- Buth, Jacob, Peter L. Harris, Roel Hobo, Randolph van Eps, Philippe Cuypers, Lucien Duijm, and Xander Tielbeek. 2007. "Neurologic Complications Associated with Endovascular Repair of Thoracic Aortic Pathology: Incidence and Risk Factors. A Study from the European Collaborators on Stent/Graft Techniques for Aortic Aneurysm Repair (EU-ROSTAR) Registry." Journal of Vascular Surgery 46 (6): 1103-1110; discussion 1110-1111. https://doi.org/10.1016/j.jvs.2007.08.020.
- Byrne, J., R. C. Darling, S. P. Roddy, M. Mehta, P. S. K. Paty, P. B. Kreienberg, B. B. Chang, K. J. Ozsvath, Y. Sternbach, and D. M. Shah. 2007. "Long Term Outcome for Extra-Anatomic Arch Reconstruction. An Analysis of 143 Procedures." *European Journal of Vascular and Endovascular Surgery* 34 (4): 444–50. https://doi.org/10.1016/j.ejvs.2007.05.016.
- Cahill, Kate, Nicola Lindson-Hawley, Kyla H. Thomas, Thomas R. Fanshawe, and Tim Lancaster. 2016. "Nicotine Receptor Partial Agonists for Smoking Cessation." The Cochrane Database of Systematic Reviews, no. 5 (May): CD006103. https://doi.org/10.1002/14651858.CD006103.pub7.
- Calligaro, Keith D, and Matthew J Dougherty. 2019. "128: Renovascular Disease: Aneurysms and Arteriovenous Fistulae." In *Rutherford's Vascular Surgery and Endovascular Therapy*, 9th ed. Elsevier Inc.
- Cameron, Scott J., Hannah M. Russell, and A. Phillip Owens. 2018. "Antithrombotic Therapy in Abdominal Aortic Aneurysm: Beneficial or Detrimental?" Blood 132 (25): 2619–28. https://doi.org/10.1182/blood-2017-08-743237.
- Canaud, Ludovic, Baris A. Ozdemir, Benjamin O. Patterson, Peter J. E. Holt, Ian M. Loftus, and Matt M. Thompson. 2014. "Retrograde Aortic Dissection After Thoracic Endovascular Aortic Repair." Annals of Surgery 260 (2): 389–95. https://doi.org/10.1097/SLA. 000000000000585.
- Caridi, J. G., and I. F. Hawkins. 1997. "CO2 Digital Subtraction Angiography: Potential Complications and Their Prevention." Journal of Vascular and Interventional Radiology: JVIR 8 (3): 383–91. https://doi.org/10.1016/s1051-0443(97)70577-3.
- Caro Monroig, Angeliz, Shilpa N. Reddy, Jeffrey Forris Beecham Chick, Therese E. Sammarco, Jesse L. Chittams, and Scott O. Trerotola. 2018. "Fistulography of a Patent Hemodialysis Access: When Not to Treat and Implications for Establishing a Nontreatment Rate." Journal of Vascular and Interventional Radiology: JVIR 29 (3): 376–82. https://doi.org/10.1016/j.jvir.2017.11.012.
- Carr, Marcus P., Giles W. Becker, Mihra S. Taljanovic, and Wendy E. McCurdy. 2019.
 "Hypothenar Hammer Syndrome: Case Report and Literature Review." *Radiology Case Reports* 14 (7): 868–71. https://doi.org/10.1016/j.radcr.2019.04.015.
- Carter, Stefan A., and Robert B. Tate. 2001. "The Value of Toe Pulse Waves in Determination of Risks for Limb Amputation and Death in Patients with Peripheral Arterial Disease and Skin Ulcers or Gangrene." Journal of Vascular Surgery 33 (4): 708–14. https://doi.org/10.

1067/mva.2001.112329.

- Cauley, Ryan, Winona W. Wu, Andres Doval, Elliot Chaikof, Kalon K. L. Ho, and Matthew L. Iorio. 2019. "Identifying Complications and Optimizing Consultations Following Transradial Arterial Access for Cardiac Procedures." Annals of Vascular Surgery 56 (April): 87–96. https://doi.org/10.1016/j.avsg.2018.07.064.
- CDC. 2011. Prenatal Radiation Exposure: A Fact Sheet for Physicians. Radiation Emergencies. https://emergency.cdc.gov/radiation/pdf/prenatalphysician.pdf.
- Chaer, Rabih A., Christopher J. Abularrage, Dawn M. Coleman, Mohammad H. Eslami, Vikram S. Kashyap, Caron Rockman, and M. Hassan Murad. 2020. "The Society for Vascular Surgery Clinical Practice Guidelines on the Management of Visceral Aneurysms." Journal of Vascular Surgery 72 (1): 3S-39S. https://doi.org/10.1016/j.jvs.2020.01.039.
- Chaer, Rabih A., Anna Gushchin, Robert Rhee, Luke Marone, Jae S. Cho, Steven Leers, and Michel S. Makaroun. 2009. "Duplex Ultrasound as the Sole Long-Term Surveillance Method Post-Endovascular Aneurysm Repair: A Safe Alternative for Stable Aneurysms." *Journal of Vascular Surgery* 49 (4): 845-849; discussion 849-850. https://doi.org/10.1016/ j.jvs.2008.10.073.
- Chaikof, Elliot L., Ronald L. Dalman, Mark K. Eskandari, Benjamin M. Jackson, W. Anthony Lee, M. Ashraf Mansour, Tara M. Mastracci, Matthew Mell, M. Hassan Murad, Louis L. Nguyen, Gustavo S. Oderich, et al. 2018. "The Society for Vascular Surgery Practice Guidelines on the Care of Patients with an Abdominal Aortic Aneurysm." Journal of Vascular Surgery 67 (1): 2–77.e2. https://doi.org/10.1016/j.jvs.2017.10.044.
- Chaikof, Elliot L., Ronald L. Dalman, Mark K. Eskandari, Benjamin M. Jackson, W. Anthony Lee, M. Ashraf Mansour, Tara M. Mastracci, Matthew Mell, M. Hassan Murad, Louis L. Nguyen, and et al. 2018. "The Society for Vascular Surgery Practice Guidelines on the Care of Patients with an Abdominal Aortic Aneurysm." Journal of Vascular Surgery 67 (1): 2–77.e2. https://doi.org/10.1016/j.jvs.2017.10.044.
- Chandra, Ankur, and William J. Quinones-Baldrich. 2010. "Chronic Mesenteric Ischemia: How to Select Patients for Invasive Treatment." Seminars in Vascular Surgery 23 (1): 21–28. https://doi.org/10.1053/j.semvascsurg.2009.12.003.
- Chandra, Venita, Chelsea Dorsey, Amy B. Reed, Palma Shaw, Dawn Banghart, and Wei Zhou. 2013. "Monitoring of Fetal Radiation Exposure During Pregnancy." Journal of Vascular Surgery 58 (3): 710–14. https://doi.org/10.1016/j.jvs.2013.01.052.
- Chang, Benjamin B., Philip S. K. Paty, Dhiraj M. Shah, and Robert P. Leather. 1992. "The Lesser Saphenous Vein: An Underappreciated Source of Autogenous Vein." Journal of Vascular Surgery 15 (1): 152–57. https://doi.org/10.1016/0741-5214(92)70023-E.
- Chang, Benjamin B., Sean P. Roddy, R. Clement Darling, Dale Maharaj, Philip S. K. Paty, Paul B. Kreienberg, Kathleen J. Ozsvath, Manish Mehta, and Dhiraj M. Shah. 2003.
 "Upper Extremity Bypass Grafting for Limb Salvage in End-Stage Renal Failure." *Journal* of Vascular Surgery 38 (6): 1313–15. https://doi.org/10.1016/s0741-5214(03)00773-0.
- Chang, Kevin Z., Kendall Likes, Jasmine Demos, James H. Black, and Julie A. Freischlag. 2012. "Routine Venography Following Transaxillary First Rib Resection and Scalenectomy (FRRS) for Chronic Subclavian Vein Thrombosis Ensures Excellent Outcomes and Vein Patency." Vascular and Endovascular Surgery 46 (1): 15–20. https://doi.org/10.1177/

1538574411423982.

- Chatterjee, Saurav, Nishant Nerella, Saneka Chakravarty, and Jacob Shani. 2013. "Angioplasty Alone Versus Angioplasty and Stenting for Subclavian Artery Stenosis—A Systematic Review and Meta-analysis." American Journal of Therapeutics 20 (5): 520–23. https://doi.org/10.1097/MJT.0b013e31822831d8.
- Chawla, Yogesh K., and Vijay Bodh. 2015. "Portal Vein Thrombosis." Journal of Clinical and Experimental Hepatology 5 (1): 22–40. https://doi.org/10.1016/j.jceh.2014.12.008.
- Cheatham, Michael L., Manu L. N. G. Malbrain, Andrew Kirkpatrick, Michael Sugrue, Michael Parr, Jan De Waele, Zsolt Balogh, et al. 2007. "Results from the International Conference of Experts on Intra-Abdominal Hypertension and Abdominal Compartment Syndrome. II. Recommendations." *Intensive Care Medicine* 33 (6): 951–62. https://doi.org/10.1007/ s00134-007-0592-4.
- Chen, Allan, Davey Prout, and Neeraj Jain. 2014. "Bilateral Subclavian Stenosis as a Cause of Unexplained Hypotension." Journal of American College of Cardiology 12 Supplement (April): A698.
- Chen, Xiyang, Yang Liu, Jiarong Wang, Jichun Zhao, Niten Singh, and Wayne W. Zhang. 2020. "A Systematic Review and Meta-Analysis of the Risk of Death and Patency After Application of Paclitaxel-Coated Balloons in the Hemodialysis Access." Journal of Vascular Surgery 72 (6): 2186–2196.e3. https://doi.org/10.1016/j.jvs.2020.04.525.
- Cheun, Tracy J., Lalithapriya Jayakumar, Maureen K. Sheehan, Matthew J. Sideman, Lori L. Pounds, and Mark G. Davies. 2019. "Outcomes of Upper Extremity Interventions for Chronic Critical Ischemia." *Journal of Vascular Surgery* 69 (1): 120–128.e2. https: //doi.org/10.1016/j.jvs.2018.04.056.
- Chi, Yung-Wei, Christopher J. White, Stanley Thornton, and Richard V. Milani. 2009. "Ultrasound Velocity Criteria for Renal in-Stent Restensis." Journal of Vascular Surgery 50 (1): 119–23. https://doi.org/10.1016/j.jvs.2008.12.066.
- Chia, Matthew C., Gabriel A. Wallace, Brian T. Cheng, Karen J. Ho, and Mark K. Eskandari. 2020. "Identifying a Safe Carotid Stump Pressure Threshold for Selective Shunting During Carotid Endarterectomy." Annals of Vascular Surgery 69 (November): 158–62. https: //doi.org/10.1016/j.avsg.2020.06.002.
- Chimowitz, Marc I., Michael J. Lynn, Colin P. Derdeyn, Tanya N. Turan, David Fiorella, Bethany F. Lane, L. Scott Janis, et al. 2011. "Stenting Versus Aggressive Medical Therapy for Intracranial Arterial Stenosis." *The New England Journal of Medicine* 365 (11): 993– 1003. https://doi.org/10.1056/NEJMoa1105335.
- Chloros, George D., Nicholas N. Smerlis, Zhongyu Li, Thomas L. Smith, Beth P. Smith, and L. Andrew Koman. 2008. "Noninvasive Evaluation of Upper-Extremity Vascular Perfusion." The Journal of Hand Surgery 33 (4): 591–600. https://doi.org/10.1016/j.jhsa.2008.01.034.
- Cho, Kyung Jae. 2015. "Carbon Dioxide Angiography: Scientific Principles and Practice." Vascular Specialist International 31 (3): 67–80. https://doi.org/10.5758/vsi.2015.31.3.67.
- Cho, Kyung J., and Irvin F. Hawkins. 2011. "Discontinuation of the Plastic Bag Delivery System for Carbon Dioxide Angiography Will Increase Radiocontrast Nephropathy and Life-Threatening Complications." AJR. American Journal of Roentgenology 197 (5): W940–941. https://doi.org/10.2214/AJR.11.6639.

- Chongruksut, Wilaiwan, Tanat Vaniyapong, and Kittipan Rerkasem. 2014. "Routine or Selective Carotid Artery Shunting for Carotid Endarterectomy (and Different Methods of Monitoring in Selective Shunting)." The Cochrane Database of Systematic Reviews 2014 (6): CD000190. https://doi.org/10.1002/14651858.CD000190.pub3.
- Chopra, Atish, J. Gregory Modrall, Martyn Knowles, Herbert A. Phelan, R. James Valentine, and Jayer Chung. 2016. "Uncertain Patency of Covered Stents Placed for Traumatic Axillosubclavian Artery Injury." Journal of the American College of Surgeons 223 (1): 174–83. https://doi.org/10.1016/j.jamcollsurg.2016.02.007.
- Chow, Warren B., Wesley S Moore, and Glenn M. Lamuraglia. 2019. "95. Carotid Body Tumors." In Rutherford's Vascular Surgery and Endovascular Therapy, 1255–64. Elsevier Inc. https://doi.org/10.1016/B978-0-323-42791-3.00095-5.
- Chung, Jayer, Wilmer Valentine, Sherene E. Sharath, Amita Pathak, Neal R. Barshes, George Pisimisis, Panagiotis Kougias, and Joseph L. Mills. 2016. "Percutaneous Intervention for Carotid in-Stent Restenosis Does Not Improve Outcomes Compared with Nonoperative Management." Journal of Vascular Surgery 64 (5): 1286–1294.e1. https://doi.org/10.1016/ j.jvs.2016.05.086.
- Ciccone, Marco Matteo, Ilaria Dentamaro, Filippo Masi, Santa Carbonara, and Gabriella Ricci. 2016. "Advances in the Diagnosis of Acute Aortic Syndromes: Role of Imaging Techniques." Vascular Medicine (London, England) 21 (3): 239–50. https://doi.org/10. 1177/1358863X16631419.
- Cinà, Claudio S., Hussein A. Safar, Antonello Laganà, Goffredo Arena, and Catherine M. Clase. 2002. "Subclavian Carotid Transposition and Bypass Grafting: Consecutive Cohort Study and Systematic Review." Journal of Vascular Surgery 35 (3): 422–29. https://doi. org/10.1067/mva.2002.120035.
- Clavel, Pierre, Solène Hebert, Suzana Saleme, Charbel Mounayer, Aymeric Rouchaud, and Benoit Marin. 2019. "Cumulative Incidence of Restenosis in the Endovascular Treatment of Extracranial Carotid Artery Stenosis: A Meta-Analysis." Journal of Neurointerventional Surgery 11 (9): 916–23. https://doi.org/10.1136/neurintsurg-2018-014534.
- Clouse, W. Darrin, Luke K. Marone, J. Kenneth Davison, David J. Dorer, David C. Brewster, Glenn M. LaMuraglia, and Richard P. Cambria. 2003. "Late Aortic and Graft-Related Events After Thoracoabdominal Aneurysm Repair." Journal of Vascular Surgery 37 (2): 254–61. https://doi.org/10.1067/mva.2003.62.
- Clouse, W. Darrin, Todd E. Rasmussen, Michael A. Peck, Jonathan L. Eliason, Mitchell W. Cox, Andrew N. Bowser, Donald H. Jenkins, David L. Smith, and Norman M. Rich. 2007. "In-Theater Management of Vascular Injury: 2 Years of the Balad Vascular Registry." *Journal of the American College of Surgeons* 204 (4): 625–32. https://doi.org/10.1016/j.jamcollsurg.2007.01.040.
- Coelho, Andreia Pires, Miguel Lobo, Clara Nogueira, Ricardo Gouveia, Jacinta Campos, Rita Augusto, Nuno Coelho, Ana Carolina Semião, and Alexandra Canedo. 2019. "Overview of Evidence on Risk Factors and Early Management of Acute Carotid Stent Thrombosis During the Last Two Decades." Journal of Vascular Surgery 69 (3): 952–64. https://doi. org/10.1016/j.jvs.2018.09.053.
- Cohen, José E., John M. Gomori, Eyal Itshayek, Sergey Spektor, Yigal Shoshan, Guy Rosen-

thal, and Samuel Moscovici. 2012. "Single-Center Experience on Endovascular Reconstruction of Traumatic Internal Carotid Artery Dissections:" *The Journal of Trauma and Acute Care Surgery* 72 (1): 216–21. https://doi.org/10.1097/TA.0b013e31823f630a.

- Cohen, José E., J. Moshe Gomori, Haim Anner, and Eyal Itshayek. 2014. "Inadvertent Subclavian Artery Cannulation Treated by Percutaneous Closure." Journal of Clinical Neuroscience: Official Journal of the Neurosurgical Society of Australasia 21 (11): 1973– 75. https://doi.org/10.1016/j.jocn.2014.04.009.
- Cohn, Joseph D., Alberto Caggiati, and Keith F. Korver. 2006. "Accessory and Great Saphenous Veins as Coronary Artery Bypass Conduits." *Interactive Cardiovascular and Thoracic* Surgery 5 (5): 550–54. https://doi.org/10.1510/icvts.2005.125468.
- Coleman, Dawn M., Andrea Obi, Enrique Criado, Shipra Arya, and Ramon Berguer. 2013. "Contemporary Outcomes After Distal Vertebral Reconstruction." Journal of Vascular Surgery 58 (1): 152–57. https://doi.org/10.1016/j.jvs.2012.12.067.
- Coleman, Dawn M., and James C. Stanley. 2015. "Renal Artery Aneurysms." Journal of Vascular Surgery 62 (3): 779–85. https://doi.org/10.1016/j.jvs.2015.05.034.
- Colhoun, Helen M., D. John Betteridge, Paul N. Durrington, Graham A. Hitman, H. Andrew W. Neil, Shona J. Livingstone, Margaret J. Thomason, et al. 2004. "Primary Prevention of Cardiovascular Disease with Atorvastatin in Type 2 Diabetes in the Collaborative Atorvastatin Diabetes Study (CARDS): Multicentre Randomised Placebo-Controlled Trial." Lancet (London, England) 364 (9435): 685–96. https://doi.org/10.1016/S0140-6736(04)16895-5.
- Collaboration, Antithrombotic Trialists'. 2002. "Collaborative Meta-Analysis of Randomised Trials of Antiplatelet Therapy for Prevention of Death, Myocardial Infarction, and Stroke in High Risk Patients." BMJ : British Medical Journal 324 (7329): 71–86.
- Collaboration, Antithrombotic Trialists'(ATT), Colin Baigent, Lisa Blackwell, Rory Collins, Jonathan Emberson, Jon Godwin, Richard Peto, et al. 2009. "Aspirin in the Primary and Secondary Prevention of Vascular Disease: Collaborative Meta-Analysis of Individual Participant Data from Randomised Trials." *Lancet (London, England)* 373 (9678): 1849–60. https://doi.org/10.1016/S0140-6736(09)60503-1.
- Collaborators, Cholesterol Treatment Trialists'(CTT), P. M. Kearney, L. Blackwell, R. Collins, A. Keech, J. Simes, R. Peto, J. Armitage, and C. Baigent. 2008. "Efficacy of Cholesterol-Lowering Therapy in 18,686 People with Diabetes in 14 Randomised Trials of Statins: A Meta-Analysis." *Lancet (London, England)* 371 (9607): 117–25. https://doi.org/10.1016/ S0140-6736(08)60104-X.
- Comerota, Anthony J, and Faisal Aziz. 2019. "148. Acute Deep Venous Thrombosis : Surgical and Interventional Treatment." In Rutherford's Vascular Surgery and Endovascular Therapy. Elsevier Inc. http://dx.doi.org/10.1016/B978-1-4557-5304-8.00052-2.
- Commission, The Joint. 2019. Radiation Risks of Diagnostic Imaging and Fluoroscopy. Sentinel Event Alert. https://www.jointcommission.org/-/media/tjc/documents/resources/ patient-safety-topics/sentinel-event/sea_47_radiation_revised2_feb_20181.pdf.
- Committee, CAPRIE Steering. 1996a. "A Randomised, Blinded, Trial of Clopidogrel Versus Aspirin in Patients at Risk of Ischaemic Events (CAPRIE)." The Lancet 348 (9038): 1329– 39. https://doi.org/10.1016/S0140-6736(96)09457-3.

—. 1996b. "A Randomised, Blinded, Trial of Clopidogrel Versus Aspirin in Patients at Risk of Ischaemic Events (CAPRIE)." *The Lancet* 348 (9038): 1329–39. https://doi.org/10.1016/S0140-6736(96)09457-3.

- Connolly, Stuart J., Mark Crowther, John W. Eikelboom, C. Michael Gibson, John T. Curnutte, John H. Lawrence, Patrick Yue, et al. 2019. "Full Study Report of Andexanet Alfa for Bleeding Associated with Factor Xa Inhibitors." *The New England Journal of Medicine* 380 (14): 1326–35. https://doi.org/10.1056/NEJMoa1814051.
- Conte, Michael S., Dennis F. Bandyk, Alexander W. Clowes, Gregory L. Moneta, Lynn Seely, Todd J. Lorenz, Hamid Namini, et al. 2006. "Results of PREVENT III: A Multicenter, Randomized Trial of Edifoligide for the Prevention of Vein Graft Failure in Lower Extremity Bypass Surgery." Journal of Vascular Surgery 43 (4): 742–751; discussion 751. https: //doi.org/10.1016/j.jvs.2005.12.058.
- Conte, Michael S., Andrew W. Bradbury, Philippe Kolh, John V. White, Florian Dick, Robert Fitridge, Joseph L Mills, Jean-Baptiste Ricco, Kalkunte R. Suresh, and M. Hassan Murad. 2019. "Global Vascular Guidelines on the Management of Chronic Limb-Threatening Ischemia." European Journal of Vascular and Endovascular Surgery : The Official Journal of the European Society for Vascular Surgery 58 (1 Suppl): S1–S109.e33. https: //doi.org/10.1016/j.ejvs.2019.05.006.
- Conte, Michael S., Frank B. Pomposelli, Daniel G. Clair, Patrick J. Geraghty, James F. McKinsey, Joseph L. Mills, Gregory L. Moneta, M. Hassan Murad, Richard J. Powell, Amy B. Reed, Andres Schanzer, et al. 2015. "Society for Vascular Surgery Practice Guidelines for Atherosclerotic Occlusive Disease of the Lower Extremities: Management of Asymptomatic Disease and Claudication." Journal of Vascular Surgery 61 (3): 2S-41S.e1. https://doi.org/10.1016/j.jvs.2014.12.009.
- Conte, Michael S., Frank B. Pomposelli, Daniel G. Clair, Patrick J. Geraghty, James F. McKinsey, Joseph L. Mills, Gregory L. Moneta, M. Hassan Murad, Richard J. Powell, Amy B. Reed, and et al. 2015. "Society for Vascular Surgery Practice Guidelines for Atherosclerotic Occlusive Disease of the Lower Extremities: Management of Asymptomatic Disease and Claudication." Journal of Vascular Surgery 61 (3): 2S-41S.e1. https://doi.org/10.1016/j.jvs.2014.12.009.
- Cooper, Christopher J., Timothy P. Murphy, Donald E. Cutlip, Kenneth Jamerson, William Henrich, Diane M. Reid, David J. Cohen, et al. 2014. "Stenting and Medical Therapy for Atherosclerotic Renal-Artery Stenosis." New England Journal of Medicine 370 (1): 13–22. https://doi.org/10.1056/NEJMoa1310753.
- Cope, C., R. Salem, and L. R. Kaiser. 1999. "Management of Chylothorax by Percutaneous Catheterization and Embolization of the Thoracic Duct: Prospective Trial." Journal of Vascular and Interventional Radiology: JVIR 10 (9): 1248–54. https://doi.org/10.1016/ s1051-0443(99)70227-7.
- Coselli, Joseph S., Hiruni S. Amarasekara, Susan Y. Green, Matt D. Price, Ourania Preventza, Kim I. de la Cruz, Qianzi Zhang, and Scott A. LeMaire. 2017. "Open Repair of Thoracoabdominal Aortic Aneurysm in Patients 50 Years Old and Younger." *The Annals of Thoracic* Surgery 103 (6): 1849–57. https://doi.org/10.1016/j.athoracsur.2016.09.058.

Coselli, Joseph S., Scott A. LeMaire, Ourania Preventza, Kim I. de la Cruz, Denton A. Cooley,

Matt D. Price, Alan P. Stolz, Susan Y. Green, Courtney N. Arredondo, and Todd K. Rosengart. 2016. "Outcomes of 3309 Thoracoabdominal Aortic Aneurysm Repairs." *The Journal of Thoracic and Cardiovascular Surgery* 151 (5): 1323–37. https://doi.org/10.1016/j.jtcvs.2015.12.050.

- Cowan, John A, Justin B Dimick, Peter K Henke, Thomas S Huber, James C Stanley, and Gilbert R Upchurch. 2003. "Surgical Treatment of Intact Thoracoabdominal Aortic Aneurysms in the United States: Hospital and Surgeon Volume-Related Outcomes." Journal of Vascular Surgery 37 (6): 1169–74. https://doi.org/10.1016/S0741-5214(03)00085-5.
- Coward, Lucy J., Dominick J. H. McCabe, Joerg Ederle, Roland L. Featherstone, Andrew Clifton, Martin M. Brown, and CAVATAS Investigators. 2007. "Long-Term Outcome After Angioplasty and Stenting for Symptomatic Vertebral Artery Stenosis Compared with Medical Treatment in the Carotid and Vertebral Artery Transluminal Angioplasty Study (CAVATAS): A Randomized Trial." Stroke 38 (5): 1526–30. https://doi.org/10.1161/ STROKEAHA.106.471862.
- Cranley, John J. 1969. "Ischemic Rest Pain." Archives of Surgery 98 (2): 187. https://doi. org/10.1001/archsurg.1969.01340080079015.
- Crawford, E. S., J. L. Crawford, H. J. Safi, J. S. Coselli, K. R. Hess, B. Brooks, H. J. Norton, and D. H. Glaeser. 1986. "Thoracoabdominal Aortic Aneurysms: Preoperative and Intraoperative Factors Determining Immediate and Long-Term Results of Operations in 605 Patients." Journal of Vascular Surgery 3 (3): 389–404. https://doi.org/10.1067/mva. 1986.avs0030389.
- Creager, Mark A., John A. Kaufman, and Michael S. Conte. 2012. "Clinical Practice. Acute Limb Ischemia." The New England Journal of Medicine 366 (23): 2198–2206. https: //doi.org/10.1056/NEJMcp1006054.
- Criado, Enrique, Ramon Berguer, and Lazar Greenfield. 2010. "The Spectrum of Arterial Compression at the Thoracic Outlet." *Journal of Vascular Surgery* 52 (2): 406–11. https: //doi.org/10.1016/j.jvs.2010.03.009.
- Criado, Enrique, Gilbert R. Upchurch, Kate Young, John E. Rectenwald, Dawn M. Coleman, Jonathon L. Eliason, and Guillermo A. Escobar. 2012. "Endovascular Aortic Aneurysm Repair with Carbon Dioxide-Guided Angiography in Patients with Renal Insufficiency." Journal of Vascular Surgery 55 (6): 1570–75. https://doi.org/10.1016/j.jvs.2011.11.142.
- Criado, Frank J. 2007. "Iliac Arterial Conduits for Endovascular Access: Technical Considerations." Journal of Endovascular Therapy 14 (3): 347–51. https://doi.org/10.1583/06-2059.1.
- Cua, Bennett, Natasha Mamdani, David Halpin, Sunny Jhamnani, Sasanka Jayasuriya, and Carlos Mena-Hurtado. 2017. "Review of Coronary Subclavian Steal Syndrome." Journal of Cardiology 70 (5): 432–37. https://doi.org/10.1016/j.jjcc.2017.02.012.
- Cuker, Adam, Allison Burnett, Darren Triller, Mark Crowther, Jack Ansell, Elizabeth M. Van Cott, Diane Wirth, and Scott Kaatz. 2019. "Reversal of Direct Oral Anticoagulants: Guidance from the Anticoagulation Forum." American Journal of Hematology 94 (6): 697– 709. https://doi.org/10.1002/ajh.25475.
- Curl, G. R., G. L. Faggioli, A. Stella, M. D'Addato, and J. J. Ricotta. 1992. "Aneurysmal Change at or Above the Proximal Anastomosis After Infrarenal Aortic Grafting." *Journal*

of Vascular Surgery 16 (6): 855-859; discussion 859-860.

- D'Angelo, A., and J. Selhub. 1997. "Homocysteine and Thrombotic Disease." Blood 90 (1): 1–11.
- Daniel, Vijaya T., Arin L. Madenci, Louis L. Nguyen, Mohammad H. Eslami, Jeffrey A. Kalish, Alik Farber, and James T. McPhee. 2014. "Contemporary Comparison of Supra-Aortic Trunk Surgical Reconstructions for Occlusive Disease." Journal of Vascular Surgery 59 (6): 1577–1582.e2. https://doi.org/10.1016/j.jvs.2013.12.017.
- Darwood, Rosemary, David C Berridge, David O Kessel, Iain Robertson, and Rachel Forster. 2018. "Surgery Versus Thrombolysis for Initial Management of Acute Limb Ischaemia." *The Cochrane Database of Systematic Reviews* 2018 (8): CD002784. https://doi.org/10. 1002/14651858.CD002784.pub3.
- Dauer, Lawrence T., Donald L. Miller, Beth Schueler, James Silberzweig, Stephen Balter, Gabriel Bartal, Charles Chambers, et al. 2015. "Occupational Radiation Protection of Pregnant or Potentially Pregnant Workers in IR: A Joint Guideline of the Society of Interventional Radiology and the Cardiovascular and Interventional Radiological Society of Europe." Journal of Vascular and Interventional Radiology: JVIR 26 (2): 171–81. https://doi.org/10.1016/j.jvir.2014.11.026.
- Davies, Mark G., Javier E. Anaya-Ayala, and Hosam F. El-Sayed. 2016. "Equivalent Outcomes with Standard and Heparin-Bonded Expanded Polytetrafluoroethylene Grafts Used as Conduits for Hemodialysis Access." Journal of Vascular Surgery 64 (3): 715–18. https: //doi.org/10.1016/j.jvs.2016.03.443.
- Davies, Mark G., Wael E. Saad, Eric K. Peden, Imran T. Mohiuddin, Joseph J. Naoum, and Alan B. Lumsden. 2008. "The Long-Term Outcomes of Percutaneous Therapy for Renal Artery Fibromuscular Dysplasia." *Journal of Vascular Surgery* 48 (4): 865–71. https: //doi.org/10.1016/j.jvs.2008.05.030.
- Davila, Victor J., James M. Chang, William M. Stone, Richard J. Fowl, Thomas C. Bower, Michael L. Hinni, and Samuel R. Money. 2016. "Current Surgical Management of Carotid Body Tumors." *Journal of Vascular Surgery* 64 (6): 1703–10. https://doi.org/10.1016/j. jvs.2016.05.076.
- De Maeseneer, Marianne G., Stavros K. Kakkos, Thomas Aherne, Niels Baekgaard, Stephen Black, Lena Blomgren, Athanasios Giannoukas, et al. 2022. "Editor's Choice – European Society for Vascular Surgery (ESVS) 2022 Clinical Practice Guidelines on the Management of Chronic Venous Disease of the Lower Limbs." European Journal of Vascular and Endovascular Surgery 63 (2): 184–267. https://doi.org/10.1016/j.ejvs.2021.12.024.
- De Rango, P., A. L. Estrera, C. Miller, T.-Y. Lee, K. Keyhani, S. Abdullah, and H. Safi. 2011. "Operative Outcomes Using a Side-Branched Thoracoabdominal Aortic Graft (STAG) for Thoraco-Abdominal Aortic Repair." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 41 (1): 41–47. https://doi.org/10.1016/j.ejvs.2010.10.006.
- Dean, S. M., and B. Satiani. 2001. "Three Cases of Digital Ischemia Successfully Treated with Cilostazol." Vascular Medicine (London, England) 6 (4): 245–48. https://doi.org/10. 1177/1358836x0100600408.
- Debourdeau, P., D. Kassab Chahmi, G. Le Gal, I. Kriegel, E. Desruennes, M.-C. Douard,

I. Elalamy, et al. 2009. "2008 SOR Guidelines for the Prevention and Treatment of Thrombosis Associated with Central Venous Catheters in Patients with Cancer: Report from the Working Group." Annals of Oncology: Official Journal of the European Society for Medical Oncology 20 (9): 1459–71. https://doi.org/10.1093/annonc/mdp052.

- Decousus, Hervé, Isabelle Quéré, Emilie Presles, Francois Becker, Marie-Thérèse Barrellier, Myriam Chanut, Jean-Luc Gillet, et al. 2010. "Superficial Venous Thrombosis and Venous Thromboembolism: A Large, Prospective Epidemiologic Study." Annals of Internal Medicine 152 (4): 218–24. https://doi.org/10.7326/0003-4819-152-4-201002160-00006.
- DeGiovanni, Jason, Andrew Son, and Payam Salehi. 2020. "Transposition of External Jugular to Proximal Internal Jugular Vein for Relief of Venous Thoracic Outlet Syndrome and Maintenance of Arteriovenous Fistula Access for Chronic Hemodialysis: A New Approach." The Journal of Vascular Access 21 (1): 98–102. https://doi.org/10.1177/1129729819851063.
- Delaney, Sean W., Haoran Shi, Alireza Shokrani, and Uttam K. Sinha. 2017. "Management of Chyle Leak After Head and Neck Surgery: Review of Current Treatment Strategies." *International Journal of Otolaryngology* 2017: 8362874. https://doi.org/10.1155/2017/ 8362874.
- Demetriades, Demetrios, Peep Talving, and Kenji Inaba. 2022. "17. Blunt Thoracic Aortic Injury." In Rich's Vascular Trauma, 199–211. Elsevier. https://doi.org/10.1016/B978-0-323-69766-8.00017-2.
- Demetriades, Demetrios, George C. Velmahos, Thomas M. Scalea, Gregory J. Jurkovich, Riyad Karmy-Jones, Pedro G. Teixeira, Mark R. Hemmila, et al. 2008. "Diagnosis and Treatment of Blunt Thoracic Aortic Injuries: Changing Perspectives." *The Journal of Trauma* 64 (6): 1415-1418; discussion 1418-1419. https://doi.org/10.1097/TA.0b013e3181715e32.
- Desai, Kush R., James L. Laws, Riad Salem, Samdeep K. Mouli, Martin F. Errea, Jennifer K. Karp, Yihe Yang, Robert K. Ryu, and Robert J. Lewandowski. 2017. "Defining Prolonged Dwell Time: When Are Advanced Inferior Vena Cava Filter Retrieval Techniques Necessary? An Analysis in 762 Procedures." *Circulation. Cardiovascular Interventions* 10 (6): e003957. https://doi.org/10.1161/CIRCINTERVENTIONS.116.003957.
- Devereaux, P. J., and Daniel I. Sessler. 2015. "Cardiac Complications in Patients Undergoing Major Noncardiac Surgery." The New England Journal of Medicine 373 (23): 2258–69. https://doi.org/10.1056/NEJMra1502824.
- DeYoung, Elliot, and Jeet Minocha. 2016. "Inferior Vena Cava Filters: Guidelines, Best Practice, and Expanding Indications." Seminars in Interventional Radiology 33 (2): 65–70. https://doi.org/10.1055/s-0036-1581088.
- Di Nisio, Marcello, Iris M. Wichers, and Saskia Middeldorp. 2018. "Treatment for Superficial Thrombophlebitis of the Leg." *The Cochrane Database of Systematic Reviews* 2 (February): CD004982. https://doi.org/10.1002/14651858.CD004982.pub6.
- Diagnostic Medical Sonography, Society of. 2015. "Scope of Practice and Clinical Standards for the Diagnostic Medical Sonographer." https://www.sdms.org/docs/defaultsource/Resources/scope-of-practice-and-clinical-standards.pdf.
- Diehm, Curt, Stefan Lange, Hans Joachim Trampisch, Roman Haberl, Harald Darius, Berndt von Stritzky, David Pittrow, Gerhart Tepohl, Jens Rainer Allenberg, and GetABI Study Group. 2004. "Relationship Between Lipid Parameters and the Presence of Peripheral

Arterial Disease in Elderly Patients." Current Medical Research and Opinion 20 (12): 1873–75. https://doi.org/10.1185/030079904x12780.

- Dineen, Sean, Stephen Smith, and Frank R. Arko. 2007. "Successful Percutaneous Angioplasty and Stenting of the Radial Artery in a Patient with Chronic Upper Extremity Ischemia and Digital Gangrene." Journal of Endovascular Therapy: An Official Journal of the International Society of Endovascular Specialists 14 (3): 426–28. https://doi.org/10.1583/ 06-2017.1.
- Diwan, Aparna, Rajabrata Sarkar, James C. Stanley, Gerald B. Zelenock, and Thomas W. Wakefield. 2000a. "Incidence of Femoral and Popliteal Artery Aneurysms in Patients with Abdominal Aortic Aneurysms." Journal of Vascular Surgery 31 (5): 863–69. https://doi.org/10.1067/mva.2000.105955.
 - ——. 2000b. "Incidence of Femoral and Popliteal Artery Aneurysms in Patients with Abdominal Aortic Aneurysms." *Journal of Vascular Surgery* 31 (5): 863–69. https://doi.org/10.1067/mva.2000.105955.
- Dix, F. P., M. Titi, and H. Al-Khaffaf. 2005. "The Isolated Internal Iliac Artery Aneurysm–a Review." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 30 (2): 119–29. https://doi.org/10.1016/j. ejvs.2005.04.035.
- Donas, Konstantinos P., Jason T. Lee, Mario Lachat, Giovanni Torsello, and Frank J. Veith. 2015. "Collected World Experience About the Performance of the Snorkel/Chimney Endovascular Technique in the Treatment of Complex Aortic Pathologies: The PERICLES Registry." Annals of Surgery 262 (3): 546–53. https://doi.org/10.1097/SLA.000000000001405.
- Dong, Ying, Yi Wang, Rui-Lian Ma, Ming Liu, Jun-Zhen Gao, Wu-Yun Su, Li Yan, and Jian-Jun Sun. 2019. "Efficacy and Safety of Direct Oral Anticoagulants Versus Low-Molecular-Weight Heparin in Patients with Cancer: A Systematic Review and Meta-Analysis." Journal of Thrombosis and Thrombolysis 48 (3): 400–412. https://doi.org/10.1007/s11239-019-01871-4.
- Doty, J. R., J. H. Flores, and D. B. Doty. 1999. "Superior Vena Cava Obstruction: Bypass Using Spiral Vein Graft." *The Annals of Thoracic Surgery* 67 (4): 1111–16. https://doi. org/10.1016/s0003-4975(99)00145-9.
- DuBose, Joseph J., Ravi Rajani, Ramy Gilani, Zachary A. Arthurs, Jonathan J. Morrison, William D. Clouse, Todd E. Rasmussen, Endovascular Skills for Trauma, and Resuscitative Surgery Working Group. 2012. "Endovascular Management of Axillo-Subclavian Arterial Injury: A Review of Published Experience." *Injury* 43 (11): 1785–92. https://doi.org/10. 1016/j.injury.2012.08.028.
- Duffy, S T, M P Colgan, S Sultan, D J Moore, and G D Shanik. 1998. "Popliteal Aneurysms: A 10-Year Experience." *European Journal of Endovascular Surgery* 16: 218–22. https://doi.org/10.1016/s1078-5884(98)80223-2.
- Dumfarth, Julia, Alan S. Chou, Bulat A. Ziganshin, Rohan Bhandari, Sven Peterss, Maryann Tranquilli, Hamid Mojibian, Hai Fang, John A. Rizzo, and John A. Elefteriades. 2015.
 "Atypical Aortic Arch Branching Variants: A Novel Marker for Thoracic Aortic Disease." The Journal of Thoracic and Cardiovascular Surgery 149 (6): 1586–92. https://doi.org/10.

1016/j.jtcvs.2015.02.019.

- Earnshaw, Jonothan J. 2019. "100: Acute Limb Ischemia : Evaluation and Decision Making." In Rutherford's Vascular Surgery and Endovascular Therapy, 2-Volume Set, 1315–1325.e2. Elsevier Inc. https://doi.org/10.1016/B978-0-323-42791-3.00100-6.
- Edwards, Matthew S, and Christopher J Cooper. 2019. "127: Renovascular Disease : Endovascular Treatment." In Rutherford's Vascular Surgery and Endovascular Therapy, Ninth. Vol. 2. Elsevier Inc. https://doi.org/10.1016/B978-1-4557-5304-8.00146-1.
- Ehlert, Bryan A, and Christopher J. Abularrage. 2019b. "139. Takayasu Disease." In Rutherford's Vascular Surgery and Endovascular Therapy, Ninth, 17. Elsevier Inc.
- ———. 2019a. "139. Takayasu Disease." In *Rutherford's Vascular Surgery and Endovascular Therapy*, 9th ed., 17. Elsevier Inc.
- Eidt, John F, and Venkat R Kalapatapu. 2019. "112: Lower Extremity Amputation : Techniques and Results." In Rutherford's Vascular Surgery and Endovascular Therapy. Vol. 2. Elsevier Inc.
- Eikelboom, John W., Stuart J. Connolly, Jackie Bosch, Gilles R. Dagenais, Robert G. Hart, Olga Shestakovska, Rafael Diaz, et al. 2017. "Rivaroxaban with or Without Aspirin in Stable Cardiovascular Disease." *The New England Journal of Medicine* 377 (14): 1319–30. https://doi.org/10.1056/NEJMoa1709118.
- Einstein, Andrew J. 2012. "Effects of Radiation Exposure from Cardiac Imaging: How Good Are the Data?" Journal of the American College of Cardiology 59 (6): 553–65. https://doi.org/10.1016/j.jacc.2011.08.079.
- Eklof, Bo, Michel Perrin, Konstantinos T. Delis, Robert B. Rutherford, and Peter Gloviczki. 2009. "Updated Terminology of Chronic Venous Disorders: The VEIN-TERM Transatlantic Interdisciplinary Consensus Document." *Journal of Vascular Surgery* 49 (2): 498–501. https://doi.org/10.1016/j.jvs.2008.09.014.
- Eklöf, Bo, Robert B. Rutherford, John J. Bergan, Patrick H. Carpentier, Peter Gloviczki, Robert L. Kistner, Mark H. Meissner, et al. 2004. "Revision of the CEAP Classification for Chronic Venous Disorders: Consensus Statement." *Journal of Vascular Surgery* 40 (6): 1248–52. https://doi.org/10.1016/j.jvs.2004.09.027.
- Elkassaby, Mohammed, Nashaat Elsaadany, Khaled Mowaphy, and Mosaad Soliman. 2021. "Balloon-Assisted Maturation of Autogenous Arteriovenous Fistulae: A Randomized Controlled Prospective Study." Vascular 29 (5): 776–83. https://doi.org/10.1177/1708538120979872.
- Elkouri, Stephane, Peter Gloviczki, Michael A. McKusick, Jean M. Panneton, James Andrews, Thomas C. Bower, Audra A. Noel, William S. Harmsen, Tanya L. Hoskin, and Kenneth Cherry. 2004. "Perioperative Complications and Early Outcome After Endovascular and Open Surgical Repair of Abdominal Aortic Aneurysms." Journal of Vascular Surgery 39 (3): 497–505. https://doi.org/10.1016/j.jvs.2003.10.018.
- Elliott, Kirsten G. B., and Alan J. Johnstone. 2003. "Diagnosing Acute Compartment Syndrome." The Journal of Bone and Joint Surgery. British Volume 85 (5): 625–32.
- Emdin, Connor A., Simon G. Anderson, Thomas Callender, Nathalie Conrad, Gholamreza Salimi-Khorshidi, Hamid Mohseni, Mark Woodward, and Kazem Rahimi. 2015. "Usual Blood Pressure, Peripheral Arterial Disease, and Vascular Risk: Cohort Study of 4.2 Million

Adults." BMJ (Clinical Research Ed.) 351 (September): h4865. https://doi.org/10.1136/ bmj.h4865.

- Erben, Young, Peter Gloviczki, Manju Kalra, Haraldur Bjarnason, Nanette R. Reed, Audra A. Duncan, Gustavo S. Oderich, and Thomas C. Bower. 2015. "Treatment of Nutcracker Syndrome with Open and Endovascular Interventions." Journal of Vascular Surgery. Venous and Lymphatic Disorders 3 (4): 389–96. https://doi.org/10.1016/j.jvsv.2015.04.003.
- Erdoes, Luke S., Jessica B. Ezell, Stuart I. Myers, Michael B. Hogan, Christopher J. LeSar, and L. Richard Sprouse. 2011. "Pharmacomechanical Thrombolysis for Phlegmasia Cerulea Dolens." *The American Surgeon* 77 (12): 1606–12. https://doi.org/10.1177/000313481107701230.
- Eskandari, Mark K, and Courtney E Morgan. 2020. 185: Conditions Arising From Repetitive Trauma and Occupational Vascular Problems. Rutherford's Vascular Surgery and Endovascular Therapy, 2-Volume Set. Elsevier Inc. https://doi.org/10.1016/B978-0-323-42791-3.00185-7.
- Eslami, Mohammad H., Denis Rybin, Gheorghe Doros, Jeffrey A. Kalish, and Alik Farber. 2015. "Comparison of a Vascular Study Group of New England Risk Prediction Model with Established Risk Prediction Models of in-Hospital Mortality After Elective Abdominal Aortic Aneurysm Repair." Journal of Vascular Surgery 62 (5): 1125–1133.e2. https://doi. org/10.1016/j.jvs.2015.06.051.
- Eslami, Mohammad H., Maksim Zayaruzny, and Gordon A. Fitzgerald. 2007. "The Adverse Effects of Race, Insurance Status, and Low Income on the Rate of Amputation in Patients Presenting with Lower Extremity Ischemia." *Journal of Vascular Surgery* 45 (1): 55–59. https://doi.org/10.1016/j.jvs.2006.09.044.
- Estrera, Anthony L., Roy Sheinbaum, Charles C. Miller, Ali Azizzadeh, Jon-Cecil Walkes, Taek-Yeon Lee, Larry Kaiser, and Hazim J. Safi. 2009. "Cerebrospinal Fluid Drainage During Thoracic Aortic Repair: Safety and Current Management." *The Annals of Thoracic Surgery* 88 (1): 9–15; discussion 15. https://doi.org/10.1016/j.athoracsur.2009.03.039.
- Fabian, T. C., J. D. Richardson, M. A. Croce, J. S. Smith, G. Rodman, P. A. Kearney, W. Flynn, et al. 1997. "Prospective Study of Blunt Aortic Injury: Multicenter Trial of the American Association for the Surgery of Trauma." *The Journal of Trauma* 42 (3): 374-380; discussion 380-383. https://doi.org/10.1097/00005373-199703000-00003.
- Fabian, Timothy, and Stephanie Savage. 2021. "19. Inferior Vena Cava, Portal and Mesenteric Venous Systems." In Rich's Vascular Trauma. Ed Rasmussen, Todd and Tai, Nigel, 4th ed. Elsevier Inc. https://www.clinicalkey.com.au/dura/browse/bookChapter/3-s2.0-C20180031916.
- Fair, D. S., R. A. Marlar, and E. G. Levin. 1986. "Human Endothelial Cells Synthesize Protein s." Blood 67 (4): 1168–71.
- Fairman, Ronald M. 2019. "72. Aortoiliac Aneurysms: Endovascular Treatment." In Rutherford's Vascular Surgery and Endovascular Therapy. Elsevier Inc.
- Fakhry, Farzin, Sandra Spronk, Lijckle van der Laan, Jan J. Wever, Joep A. W. Teijink, Wolter H. Hoffmann, Taco M. Smits, et al. 2015. "Endovascular Revascularization and Supervised Exercise for Peripheral Artery Disease and Intermittent Claudication: A Randomized Clinical Trial." JAMA 314 (18): 1936–44. https://doi.org/10.1001/jama.2015.14851.

- Fakhry, F., E. V. Rouwet, P. T. den Hoed, M. G. M. Hunink, and S. Spronk. 2013. "Long-Term Clinical Effectiveness of Supervised Exercise Therapy Versus Endovascular Revascularization for Intermittent Claudication from a Randomized Clinical Trial." *The British Journal* of Surgery 100 (9): 1164–71. https://doi.org/10.1002/bjs.9207.
- Famularo, Marissa, Karol Meyermann, and Joseph V. Lombardi. 2017. "Aneurysmal Degeneration of Type b Aortic Dissections After Thoracic Endovascular Aortic Repair: A Systematic Review." Journal of Vascular Surgery 66 (3): 924–30. https://doi.org/10.1016/ j.jvs.2017.06.067.
- Fankhauser, Grant T., William M. Stone, Richard J. Fowl, Mark E. O'Donnell, Thomas C. Bower, Fredric B. Meyer, and Samuel R. Money. 2015. "Surgical and Medical Management of Extracranial Carotid Artery Aneurysms." *Journal of Vascular Surgery* 61 (2): 389–93. https://doi.org/10.1016/j.jvs.2014.07.092.
- Farber, Alik, Matthew T Menard, Michael S Conte, John A Kaufman, Richard J Powell, Niteesh K Choudhry, Taye H Hamza, et al. 2022. "Surgery or Endovascular Therapy for Chronic Limb-Threatening Ischemia." N Engl j Med, 13.
- Faries, P. L., F. W. Logerfo, S. Arora, M. C. Pulling, D. I. Rohan, C. M. Akbari, D. R. Campbell, G. W. Gibbons, and F. B. Pomposelli. 2000. "Arm Vein Conduit Is Superior to Composite Prosthetic-Autogenous Grafts in Lower Extremity Revascularization." Journal of Vascular Surgery 31 (6): 1119–27. https://doi.org/10.1067/mva.2000.106488.
- Fatima, Javairiah, Audra A. Duncan, Eileen de Grandis, Gustavo S. Oderich, Manju Kalra, Peter Gloviczki, and Thomas C. Bower. 2013. "Treatment Strategies and Outcomes in Patients with Infected Aortic Endografts." *Journal of Vascular Surgery* 58 (2): 371–79. https://doi.org/10.1016/j.jvs.2013.01.047.
- Fatima, Javairiah, Vincent P. Federico, Salvatore T. Scali, Robert J. Feezor, Scott A. Berceli, Kristina A. Giles, Dean J. Arnaoutakis, Martin R. Back, Gilbert R. Upchurch, and Thomas S. Huber. 2019. "Management of Patch Infections After Carotid Endarterectomy and Utility of Femoral Vein Interposition Bypass Graft." *Journal of Vascular Surgery* 69 (6): 1815–1823.e1. https://doi.org/10.1016/j.jvs.2018.09.036.
- Ferreira, Joana, Alexandra Canedo, Daniel Brandão, Miguel Maia, Sandrina Braga, Manuel Chaparro, Paulo Barreto, and Guedes Vaz. 2010. "Isolated Iliac Artery Aneurysms: Six-Year Experience." *Interactive Cardiovascular and Thoracic Surgery* 10 (2): 245–48. https: //doi.org/10.1510/icvts.2009.218305.
- Ferris, B. L., L. M. Taylor, K. Oyama, R. B. McLafferty, J. M. Edwards, G. L. Moneta, and J. M. Porter. 2000. "Hypothenar Hammer Syndrome: Proposed Etiology." *Journal of Vascular Surgery* 31 (1 Pt 1): 104–13. https://doi.org/10.1016/s0741-5214(00)70072-3.
- Fiore, Marco, Chiara Colombo, Piermarco Locati, Mattia Berselli, Stefano Radaelli, Carlo Morosi, Paolo G. Casali, and Alessandro Gronchi. 2012. "Surgical Technique, Morbidity, and Outcome of Primary Retroperitoneal Sarcoma Involving Inferior Vena Cava." Annals of Surgical Oncology 19 (2): 511–18. https://doi.org/10.1245/s10434-011-1954-2.
- Fischer, Urs, Adrian Baumgartner, Marcel Arnold, Krassen Nedeltchev, Jan Gralla, Gian Marco De Marchis, Liliane Kappeler, et al. 2010. "What Is a Minor Stroke?" Stroke 41 (4): 661–66. https://doi.org/10.1161/STROKEAHA.109.572883.
- Fleisher, Lee A., Kirsten E. Fleischmann, Andrew D. Auerbach, Susan A. Barnason, Joshua
A. Beckman, Biykem Bozkurt, Victor G. Davila-Roman, et al. 2014. "2014 ACC/AHA Guideline on Perioperative Cardiovascular Evaluation and Management of Patients Undergoing Noncardiac Surgery: Executive Summary: A Report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines." *Circulation* 130 (24): 2215–45. https://doi.org/10.1161/CIR.000000000000105.

- Fletcher, E., M. Kabeer, J. Sathianathan, I. Muir, D. Williams, and C. Lim. 2016. "Immediate Catheter Directed Thrombolysis for Thromboembolic Stroke During Carotid Endarterectomy." *EJVES Short Reports* 31 (April): 12–15. https://doi.org/10.1016/j.ejvssr.2016.03. 003.
- Foley, Matthew I., Gregory L. Moneta, Ahmed M. Abou-Zamzam, James M. Edwards, Lloyd M. Taylor, Richard A. Yeager, and John M. Porter. 2000. "Revascularization of the Superior Mesenteric Artery Alone for Treatment of Intestinal Ischemia." *Journal of Vascular Surgery* 32 (1): 37–47. https://doi.org/10.1067/mva.2000.107314.
- Fontaine, R., M. Kim, and R. Kieny. 1954. "[Surgical treatment of peripheral circulation disorders]." Helvetica Chirurgica Acta 21 (5-6): 499–533.
- Forauer, Andrew R., Joseph J. Gemmete, Narasimham L. Dasika, Kyung J. Cho, and David M. Williams. 2002. "Intravascular Ultrasound in the Diagnosis and Treatment of Iliac Vein Compression (May-Thurner) Syndrome." Journal of Vascular and Interventional Radiology: JVIR 13 (5): 523–27. https://doi.org/10.1016/s1051-0443(07)61535-8.
- Forbes, Thomas L., and Ahmed Kayssi. 2019. "143: Nonatheromatous Popliteal Artery Disease." In Rutherford's Vascular Surgery and Endovascular Therapy. Elsevier Inc.
- Foundation, National Kidney. 2015. "KDOQI Clinical Practice Guideline for Hemodialysis Adequacy: 2015 Update." American Journal of Kidney Diseases: The Official Journal of the National Kidney Foundation 66 (5): 884–930. https://doi.org/10.1053/j.ajkd.2015.07. 015.
- Fowkes, F G R, E Housley, E H H Cawood, C C A Macintyre, C V Ruckley, and R J Prescott. 1991. "Edinburgh Artery Study: Prevalence of Asymptomatic and Symptomatic Peripheral Arterial Disease in the General Population." *International Journal of Epidemiology* 20 (2): 384–92. https://doi.org/10.1093/ije/20.2.384.
- Fowkes, F. Gerald R. 2010. "Aspirin for Prevention of Cardiovascular Events in a General Population Screened for a Low Ankle Brachial IndexA Randomized Controlled Trial." JAMA 303 (9): 841. https://doi.org/10.1001/jama.2010.221.
- Fox, Nicole, Ravi R. Rajani, Faran Bokhari, William C. Chiu, Andrew Kerwin, Mark J. Seamon, David Skarupa, Eric Frykberg, and Eastern Association for the Surgery of Trauma. 2012. "Evaluation and Management of Penetrating Lower Extremity Arterial Trauma: An Eastern Association for the Surgery of Trauma Practice Management Guideline." The Journal of Trauma and Acute Care Surgery 73 (5 Suppl 4): S315–320. https://doi.org/10. 1097/TA.0b013e31827018e4.
- Frech, A., L. Pellegrini, G. Fraedrich, G. Goebel, and J. Klocker. 2016. "Long-Term Clinical Outcome and Functional Status After Arterial Reconstruction in Upper Extremity Injury." *European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery* 52 (1): 119–23. https://doi.org/10.1016/j.ejvs.2016. 03.028.

- Frederick, Michael, John Newman, and Jeffrey Kohlwes. 2010. "Leriche Syndrome." Journal of General Internal Medicine 25 (10): 1102–4. https://doi.org/10.1007/s11606-010-1412-z.
- Freischlag, Julie. 2018. "The Art of Caring in the Treatment of Thoracic Outlet Syndrome." Diagnostics 8 (2): 35. https://doi.org/10.3390/diagnostics8020035.
- Frink, Michael, Frank Hildebrand, Christian Krettek, Jurgen Brand, and Stefan Hankemeier. 2010. "Compartment Syndrome of the Lower Leg and Foot." *Clinical Orthopaedics and Related Research* 468 (4): 940–50. https://doi.org/10.1007/s11999-009-0891-x.
- Frohlich, Maxfield M., Ga-Young Suh, Johan Bondesson, Matthew Leineweber, Jason T. Lee, Michael D. Dake, and Christopher P. Cheng. 2020. "Thoracic Aortic Geometry Correlates with Endograft Bird-Beaking Severity." *Journal of Vascular Surgery* 72 (4): 1196–1205. https://doi.org/10.1016/j.jvs.2019.11.045.
- Fujii, Yuichi, Junko Soga, Takayuki Hidaka, Takaki Hata, Naomi Idei, Noritaka Fujimura, Shinsuke Mikami, et al. 2011. "Color Doppler Flows of Corkscrew Collaterals in Thromboangiitis Obliterans (Buerger's Disease) Using Color Duplex Ultrasonography." Journal of the American College of Cardiology 57 (25): 2539. https://doi.org/10.1016/j.jacc.2010. 07.071.
- Gagne, Paul J., Nicole Gagne, Taras Kucher, Michael Thompson, and Dana Bentley. 2019. "Long-Term Clinical Outcomes and Technical Factors with the Wallstent for Treatment of Chronic Iliofemoral Venous Obstruction." Journal of Vascular Surgery. Venous and Lymphatic Disorders 7 (1): 45–55. https://doi.org/10.1016/j.jvsv.2018.07.016.
- Gagne, Paul J., Antonios Gasparis, Stephen Black, Patricia Thorpe, Marc Passman, Suresh Vedantham, William Marston, and Mark Iafrati. 2018. "Analysis of Threshold Stenosis by Multiplanar Venogram and Intravascular Ultrasound Examination for Predicting Clinical Improvement After Iliofemoral Vein Stenting in the VIDIO Trial." Journal of Vascular Surgery. Venous and Lymphatic Disorders 6 (1): 48–56.e1. https://doi.org/10.1016/j.jvsv. 2017.07.009.
- Gaitini, Diana, Nira Beck Razi, Eduard Ghersin, Amos Ofer, and Michalle Soudack. 2008. "Sonographic Evaluation of Vascular Injuries." Journal of Ultrasound in Medicine: Official Journal of the American Institute of Ultrasound in Medicine 27 (1): 95–107. https://doi. org/10.7863/jum.2008.27.1.95.
- Galan, Anjela, Shawn E Cowper, and Richard Bucala. 2006. "Nephrogenic Systemic FIbrosis (Nephrogenic FIbrosing Dermopathy)." Current Opinion in Rheumatology 18: 614–17.
- Galbusera, Miriam, Giuseppe Remuzzi, and Paola Boccardo. 2009. "Treatment of Bleeding in Dialysis Patients." Seminars in Dialysis 22 (3): 279–86. https://doi.org/10.1111/j.1525-139X.2008.00556.x.
- Gallieni, Maurizio, Vladimir Matoussevitch, Tobias Steinke, Adrian Ebner, Silke Brunkwall, Maurizio Cariati, Santiago Gallo, Roman Reindl-Schwaighofer, and Gürkan Sengölge. 2020.
 "Multicenter Experience with the Surfacer Inside-Out Access Catheter System in Patients with Thoracic Venous Obstruction: Results from the SAVE Registry." Journal of Vascular and Interventional Radiology: JVIR 31 (10): 1654–1660.e1. https://doi.org/10.1016/j.jvir. 2020.06.020.
- García-Criado, Angeles, Rosa Gilabert, Annalisa Berzigotti, and Concepción Brú. 2009. "Doppler Ultrasound Findings in the Hepatic Artery Shortly After Liver Transplantation."

AJR. American Journal of Roentgenology 193 (1): 128–35. https://doi.org/10.2214/AJR. 07.3919.

- Gary G Wind, and R. James Valentine. 2013. Anatomic Exposures in Vascular Surgery. Second. Lippincott Williams & Wilkins.
- Garzon-Muvdi, Tomas, Wuyang Yang, Xiaoming Rong, Justin M. Caplan, Xiaobu Ye, Geoffrey P. Colby, Alexander L. Coon, Rafael J. Tamargo, and Judy Huang. 2016. "Restenosis After Carotid Endarterectomy: Insight into Risk Factors and Modification of Postoperative Management." World Neurosurgery 89 (May): 159–67. https://doi.org/10.1016/j.wneu. 2016.01.028.
- Ge, Benjamin H., Alexander Copelan, Dominic Scola, and Micah M. Watts. 2015. "Iatrogenic Percutaneous Vascular Injuries: Clinical Presentation, Imaging, and Management." Seminars in Interventional Radiology 32 (2): 108–22. https://doi.org/10.1055/s-0035-1549375.
- Geddes, Andrea E., Clay Cothren Burlew, Amy E. Wagenaar, Walter L. Biffl, Jeffrey L. Johnson, Fredric M. Pieracci, Eric M. Campion, and Ernest E. Moore. 2016. "Expanded Screening Criteria for Blunt Cerebrovascular Injury: A Bigger Impact Than Anticipated." The American Journal of Surgery 212 (6): 1167–74. https://doi.org/10.1016/j.amjsurg. 2016.09.016.
- Gelberman, R. H., S. R. Garfin, P. T. Hergenroeder, S. J. Mubarak, and J. Menon. 1981. "Compartment Syndromes of the Forearm: Diagnosis and Treatment." *Clinical Orthopaedics and Related Research*, no. 161 (December): 252–61.
- Geraldes, Pedro, and George L. King. 2010. "Activation of Protein Kinase c Isoforms and Its Impact on Diabetic Complications." *Circulation Research* 106 (8): 1319–31. https: //doi.org/10.1161/CIRCRESAHA.110.217117.
- Gerhard-Herman, Marie D., Heather L. Gornik, Coletta Barrett, Neal R. Barshes, Matthew A. Corriere, Douglas E. Drachman, Lee A. Fleisher, et al. 2017. "2016 AHA/ACC Guideline on the Management of Patients with Lower Extremity Peripheral Artery Disease: Executive Summary: A Report of the American College of Cardiology/American Heart Association Task Force on Clinical Practice Guidelines." *Circulation* 135 (12). https://doi.org/10.1161/CIR.00000000000470.
- Gerrickens, Michael W. M., Roel H. D. Vaes, Bastiaan Govaert, Magda van Loon, Jan H. M. Tordoir, Frank van Hoek, Joep A. W. Teijink, and Marc R. Scheltinga. 2018. "Three Year Patency and Recurrence Rates of Revision Using Distal Inflow with a Venous Interposition Graft for High Flow Brachial Artery Based Arteriovenous Fistula." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 55 (6): 874–81. https://doi.org/10.1016/j.ejvs.2018.03.014.
- Ghahramani, Grant K., Alison E. Seline, and Karolyn A. Wanat. 2016. "Postprocedural Blue Toes." JAMA 315 (13): 1396–97. https://doi.org/10.1001/jama.2016.1810.
- Giannopoulos, Stefanos, Pavlos Texakalidis, Anil Kumar Jonnalagadda, Theofilos Karasavvidis, Spyridon Giannopoulos, and Damianos G. Kokkinidis. 2018. "Revascularization of Radiation-Induced Carotid Artery Stenosis with Carotid Endarterectomy Vs. Carotid Artery Stenting: A Systematic Review and Meta-Analysis." *Cardiovascular Revascularization Medicine* 19 (5): 638–44. https://doi.org/10.1016/j.carrev.2018.01.014.

Gilmore, JoAnne. 2006. "KDOQI Clinical Practice Guidelines and Clinical Practice

Recommendations–2006 Updates." Nephrology Nursing Journal: Journal of the American Nephrology Nurses' Association 33 (5): 487–88.

- Ginat, Daniel T., Wael E. A. Saad, and Ulku C. Turba. 2009. "Transcatheter Renal Artery Embolization: Clinical Applications and Techniques." *Techniques in Vascular and Inter*ventional Radiology 12 (4): 224–39. https://doi.org/10.1053/j.tvir.2009.09.007.
- Giurato, Laura, Marco Meloni, Valentina Izzo, and Luigi Uccioli. 2017. "Osteomyelitis in Diabetic Foot: A Comprehensive Overview." World Journal of Diabetes 8 (4): 135–42. https://doi.org/10.4239/wjd.v8.i4.135.
- Glickman, Marc H. 2011. "HeRO Vascular Access Device." Seminars in Vascular Surgery 24 (2): 108–12. https://doi.org/10.1053/j.semvascsurg.2011.05.006.
- Glickman, Marc H., Jason Burgess, David Cull, Prabir Roy-Chaudhury, and Harry Schanzer. 2015. "Prospective Multicenter Study with a 1-Year Analysis of a New Vascular Graft Used for Early Cannulation in Patients Undergoing Hemodialysis." Journal of Vascular Surgery 62 (2): 434–41. https://doi.org/10.1016/j.jvs.2015.03.020.
- Gloviczki, Peter, Anthony J. Comerota, Michael C. Dalsing, Bo G. Eklof, David L. Gillespie, Monika L. Gloviczki, Joann M. Lohr, et al. 2011. "The Care of Patients with Varicose Veins and Associated Chronic Venous Diseases: Clinical Practice Guidelines of the Society for Vascular Surgery and the American Venous Forum." *Journal of Vascular Surgery* 53 (5 Suppl): 2S-48S. https://doi.org/10.1016/j.jvs.2011.01.079.
- Glund, Stephan, Kelly Coble, Dietmar Gansser, Joachim Stangier, Karin Hoermann, Charles V. Pollack, and Paul Reilly. 2019. "Pharmacokinetics of Idarucizumab and Its Target Dabigatran in Patients Requiring Urgent Reversal of the Anticoagulant Effect of Dabigatran." Journal of Thrombosis and Haemostasis 17 (8): 1319–28. https://doi.org/10.1111/jth.14476.
- Gohel, Manjit S, Jamie R Barwell, Maxine Taylor, Terry Chant, Chris Foy, Jonothan J Earnshaw, Brian P Heather, David C Mitchell, Mark R Whyman, and Keith R Poskitt. 2007.
 "Long Term Results of Compression Therapy Alone Versus Compression Plus Surgery in Chronic Venous Ulceration (ESCHAR): Randomised Controlled Trial." BMJ : British Medical Journal 335 (7610): 83. https://doi.org/10.1136/bmj.39216.542442.BE.
- Gohel, Manjit S., Francine Heatley, Xinxue Liu, Andrew Bradbury, Richard Bulbulia, Nicky Cullum, David M. Epstein, et al. 2018. "A Randomized Trial of Early Endovenous Ablation in Venous Ulceration." New England Journal of Medicine 378 (22): 2105–14. https://doi. org/10.1056/NEJMoa1801214.
- Gokkus, Kemal, Ergin Sagtas, Tamer Bakalim, Ertugrul Taskaya, and Ahmet Turan Aydin. 2014. "Popliteal Entrapment Syndrome. A Systematic Review of the Literature and Case Presentation." *Muscles, Ligaments and Tendons Journal* 4 (2): 141–48.
- Goldstein, Lee J., Joseph D. Ayers, Scott Hollenbeck, Jason A. Spector, and Ageliki G. Vouyouka. 2010. "Successful Revascularization for Delayed Presentation of Radiation-Induced Distal Upper Extremity Ischemia." Annals of Vascular Surgery 24 (2): 257.e5–8. https://doi.org/10.1016/j.avsg.2009.07.024.
- Gonzales, John P., Christopher Moran, and James E. Silberzweig. 2014. "Reduction of Operator Radiation Dose by an Extended Lower Body Shield." *Journal of Vascular and Interventional Radiology: JVIR* 25 (3): 462–68, 468.e1. https://doi.org/10.1016/j.jvir.2013.11.006.

- Goodney, Philip P., Brian W. Nolan, Jens Eldrup-Jorgensen, Donald S. Likosky, and Jack L. Cronenwett. 2010. "Restenosis After Carotid Endarterectomy in a Multicenter Regional Registry." Journal of Vascular Surgery 52 (4): 897–905.e2. https://doi.org/10.1016/j.jvs. 2010.05.005.
- Gordon, T., and W. B. Kannel. 1972. "Predisposition to Atherosclerosis in the Head, Heart, and Legs. The Framingham Study." JAMA 221 (7): 661–66.
- Gornik, Heather L., and Mark A. Creager. 2008. "Aortitis." *Circulation* 117 (23): 3039–51. https://doi.org/10.1161/CIRCULATIONAHA.107.760686.
- Goudeketting, Seline R., P. H. Ping Fung Kon Jin, Çağdaş Ünlü, and Jean-Paul P. M. de Vries. 2019. "Systematic Review and Meta-Analysis of Elective and Urgent Late Open Conversion After Failed Endovascular Aneurysm Repair." Journal of Vascular Surgery 70 (2): 615–628.e7. https://doi.org/10.1016/j.jvs.2018.11.022.
- Gouëffic, Yann, Alain Costargent, Benoît Dupas, Marie-Françoise Heymann, Philippe Chaillou, and Philippe Patra. 2002. "Superior Mesenteric Artery Dissection: Case Report." Journal of Vascular Surgery 35 (5): 1003–5. https://doi.org/10.1067/mva.2002.122152.
- Gould, Michael K., David A. Garcia, Sherry M. Wren, Paul J. Karanicolas, Juan I. Arcelus, John A. Heit, and Charles M. Samama. 2012. "Prevention of VTE in Nonorthopedic Surgical Patients: Antithrombotic Therapy and Prevention of Thrombosis, 9th Ed: American College of Chest Physicians Evidence-Based Clinical Practice Guidelines." Chest 141 (2 Suppl): e227S-e277S. https://doi.org/10.1378/chest.11-2297.
- Grada, Ayman A., and Tania J. Phillips. 2017. "Lymphedema: Diagnostic Workup and Management." Journal of the American Academy of Dermatology 77 (6): 995–1006. https: //doi.org/10.1016/j.jaad.2017.03.021.
- Gradman, Wayne S., Judith Laub, and William Cohen. 2005. "Femoral Vein Transposition for Arteriovenous Hemodialysis Access: Improved Patient Selection and Intraoperative Measures Reduce Postoperative Ischemia." Journal of Vascular Surgery 41 (2): 279–84. https://doi.org/10.1016/j.jvs.2004.10.039.
- Gray, Kelsey, Brian Beckord, Ashkan Moazzez, David Plurad, Nina Bowens, and Dennis Kim. 2017. "A Comparative Analysis of Open Versus Endovascular Techniques for Management of Non-Aortic Cervicothoracic Arterial Injuries." The American Surgeon 83 (10): 1054–58.
- Greenhalgh, Rm. 2004. "Comparison of Endovascular Aneurysm Repair with Open Repair in Patients with Abdominal Aortic Aneurysm (EVAR Trial 1), 30-Day Operative Mortality Results: Randomised Controlled Trial." *The Lancet* 364 (9437): 843–48. https://doi.org/ 10.1016/S0140-6736(04)16979-1.
- Greenhalgh, Roger M, Louise C Brown, Janet T Powell, Simon G Thompson, David M. Epstein, Mark J Sculpher, and UK EVAR Trial Investigators. 2010. "Endovascular Versus Open Repair of Abdominal Aortic Aneurysm." New England Journal of Medicine 362 (20): 1863– 71. https://doi.org/10.1056/NEJMoa0909305.
- Grenon, S. Marlene, Millie Hughes-Fulford, Joseph Rapp, and Michael S. Conte. 2012. "Polyunsaturated Fatty Acids and Peripheral Artery Disease." Vascular Medicine (London, England) 17 (1): 51–63. https://doi.org/10.1177/1358863X11429175.
- Group, CALISTO Study, Hervé Decousus, Paolo Prandoni, Patrick Mismetti, Rupert M. Bauersachs, Zoltán Boda, Benjamin Brenner, et al. 2010. "Fondaparinux for the Treat-

ment of Superficial-Vein Thrombosis in the Legs." *The New England Journal of Medicine* 363 (13): 1222–32. https://doi.org/10.1056/NEJMoa0912072.

- Group, EC/IC Bypass Study. 1985. "Failure of Extracranial-Intracranial Arterial Bypass to Reduce the Risk of Ischemic Stroke. Results of an International Randomized Trial." *The New England Journal of Medicine* 313 (19): 1191–1200. https://doi.org/10.1056/ NEJM198511073131904.
- Group, Heart Protection Study Collaborative. 2007. "Randomized Trial of the Effects of Cholesterol-Lowering with Simvastatin on Peripheral Vascular and Other Major Vascular Outcomes in 20,536 People with Peripheral Arterial Disease and Other High-Risk Conditions." Journal of Vascular Surgery 45 (4): 645–654.e1. https://doi.org/10.1016/j.jvs.2006. 12.054.
- Group, POISE Study, P. J. Devereaux, Homer Yang, Salim Yusuf, Gordon Guyatt, Kate Leslie, Juan Carlos Villar, et al. 2008. "Effects of Extended-Release Metoprolol Succinate in Patients Undergoing Non-Cardiac Surgery (POISE Trial): A Randomised Controlled Trial." Lancet (London, England) 371 (9627): 1839–47. https://doi.org/10.1016/S0140-6736(08)60601-7.
- Group, Vascular Access 2006 Work. 2006. "Clinical Practice Guidelines for Vascular Access." American Journal of Kidney Diseases: The Official Journal of the National Kidney Foundation 48 Suppl 1 (July): S176–247. https://doi.org/10.1053/j.ajkd.2006.04.029.
- Grundy, Scott M., Neil J. Stone, Alison L. Bailey, Craig Beam, Kim K. Birtcher, Roger S. Blumenthal, Lynne T. Braun, et al. 2019. "2018 AHA/ACC/AACVPR/AAPA/ABC/ACPM/ADA/AGS/APhA Guideline on the Management of Blood Cholesterol: A Report of the American College of Cardiology/American Heart Association Task Force on Clinical Practice Guidelines." Journal of the American College of Cardiology 73 (24): e285–350. https://doi.org/10.1016/j.jacc.2018.11.003.
- Grunwald, Michael R., and Lawrence V. Hofmann. 2004. "Catheter-Directed Thrombolysis for the Treatment of Symptomatic Deep Vein Thrombosis." *Circulation* 109 (2). https://doi.org/10.1161/01.CIR.0000111132.23469.70.
- Guan, Jingxia, Shaofeng Zhang, Qin Zhou, Chengyan Li, and Zuneng Lu. 2013. "Usefulness of Transcranial Doppler Ultrasound in Evaluating Cervical-Cranial Collateral Circulations." *Interventional Neurology* 2 (1): 8–18. https://doi.org/10.1159/000354732.
- Guesnier-Dopagne, Mélanie, Louis Boyer, Bruno Pereira, Joël Guersen, Pascal Motreff, and Michel D'Incan. 2019. "Incidence of Chronic Radiodermatitis After Fluoroscopically Guided Interventions: A Retrospective Study." Journal of Vascular and Interventional Radiology: JVIR 30 (5): 692–698.e13. https://doi.org/10.1016/j.jvir.2019.01.010.
- Gupta, NavYash, Theodore H. Yuo, Gerhardt Konig, Ellen Dillavou, Steven A. Leers, Rabih A. Chaer, Jae S. Cho, and Michel S. Makaroun. 2011. "Treatment Strategies of Arterial Steal After Arteriovenous Access." *Journal of Vascular Surgery* 54 (1): 162–67. https: //doi.org/10.1016/j.jvs.2010.10.134.
- Gutta, Rao, James Lopes, William R. Flinn, and David G. Neschis. 2008. "Endovascular Embolization of a Giant Renal Artery Aneurysm with Preservation of Renal Parenchyma." Angiology 59 (2): 240–43. https://doi.org/10.1177/0003319707304038.
- Hackam, Daniel G., and Sonia S. Anand. 2003. "Emerging Risk Factors for Atherosclerotic

Vascular Disease: A Critical Review of the Evidence." JAMA 290 (7): 932–40. https://doi.org/10.1001/jama.290.7.932.

- Hageman, David, Hugo JP Fokkenrood, Lindy NM Gommans, Marijn ML van den Houten, and Joep AW Teijink. 2018. "Supervised Exercise Therapy Versus Home-Based Exercise Therapy Versus Walking Advice for Intermittent Claudication." *The Cochrane Database* of Systematic Reviews 2018 (4): CD005263. https://doi.org/10.1002/14651858.CD005263. pub4.
- Haig, Ylva, Tone Enden, Ole Grøtta, Nils-Einar Kløw, Carl-Erik Slagsvold, Waleed Ghanima, Leiv Sandvik, et al. 2016. "Post-Thrombotic Syndrome After Catheter-Directed Thrombolysis for Deep Vein Thrombosis (CaVenT): 5-Year Follow-up Results of an Open-Label, Randomised Controlled Trial." The Lancet Haematology 3 (2): e64–71. https: //doi.org/10.1016/S2352-3026(15)00248-3.
- Halliday, Alison, Michael Harrison, Elizabeth Hayter, Xiangling Kong, Averil Mansfield, Joanna Marro, Hongchao Pan, et al. 2010. "10-Year Stroke Prevention After Successful Carotid Endarterectomy for Asymptomatic Stenosis (ACST-1): A Multicentre Randomised Trial." Lancet 376 (9746): 1074–84. https://doi.org/10.1016/S0140-6736(10)61197-X.
- Ham, Sung Wan, and Fred A. Weaver. 2014. "Ex Vivo Renal Artery Reconstruction for Complex Renal Artery Disease." Journal of Vascular Surgery 60 (1): 143–50. https: //doi.org/10.1016/j.jvs.2014.01.061.
- Hamada, Nobuyuki, Sisko Salomaa, Kathryn D. Held, and John D. Boice. 2017. "Introduction to the Bill Morgan Memorial Special Issue on Biology, Epidemiology, and Implications for Radiation Protection." *International Journal of Radiation Biology* 93 (10): 1003–8. https://doi.org/10.1080/09553002.2017.1342148.
- Han, Sun, Pil Won Seo, and Jae-Wook Ryu. 2017. "Surgical Outcomes of Forearm Loop Arteriovenous Fistula Formation Using Tapered Versus Non-Tapered Polytetrafluoroethylene Grafts." *The Korean Journal of Thoracic and Cardiovascular Surgery* 50 (1): 30–35. https://doi.org/10.5090/kjtcs.2017.50.1.30.
- Haqqani, Omar P., Prakhar K. Agarwal, Neil M. Halin, and Mark D. Iafrati. 2012. "Minimizing Radiation Exposure to the Vascular Surgeon." Journal of Vascular Surgery 55 (3): 799–805. https://doi.org/10.1016/j.jvs.2011.08.055.
- Harris, Sally A., Rahul Velineni, and Alun H. Davies. 2016. "Inferior Vena Cava Filters in Pregnancy: A Systematic Review." Journal of Vascular and Interventional Radiology: JVIR 27 (3): 354–360.e8. https://doi.org/10.1016/j.jvir.2015.11.024.
- Healthcare Inspections, Office of. 2014. Combined Assessment Program Review of the Canandaigua VA Medical Center. 14-00688-162. Department of Veterans Affairs Office of Inspector General. https://www.va.gov/oig/pubs/VAOIG-14-00688-162.pdf.
- Hedrick, Wayne R., David L. Hykes, and Dale E. Starchman. 2005. Ultrasound Physics and Instrumentation. 4th ed. Elsevier Mosby.
- Heidbuchel, Hein, Fred H. M. Wittkampf, Eliseo Vano, Sabine Ernst, Richard Schilling, Eugenio Picano, Lluis Mont, et al. 2014. "Practical Ways to Reduce Radiation Dose for Patients and Staff During Device Implantations and Electrophysiological Procedures." Europace: European Pacing, Arrhythmias, and Cardiac Electrophysiology: Journal of the Working Groups on Cardiac Pacing, Arrhythmias, and Cardiac Cellular Electrophysiology of the

European Society of Cardiology 16 (7): 946–64. https://doi.org/10.1093/europace/eut409.

- Heinen, Stefan G., Sanne W. de Boer, Daniel A. van den Heuvel, Wouter Huberts, Patricia Dekker, Frans N. van de Vosse, Tammo Delhaas, and Jean-Paul de Vries. 2018. "Hemodynamic Significance Assessment of Equivocal Iliac Artery Stenoses by Comparing Duplex Ultrasonography with Intra-Arterial Pressure Measurements." The Journal of Cardiovascular Surgery 59 (1): 37–44. https://doi.org/10.23736/S0021-9509.17.10186-2.
- Heit, John A., Catie E. Kobbervig, Andra H. James, Tanya M. Petterson, Kent R. Bailey, and L. Joseph Melton. 2005. "Trends in the Incidence of Venous Thromboembolism During Pregnancy or Postpartum: A 30-Year Population-Based Study." Annals of Internal Medicine 143 (10): 697–706. https://doi.org/10.7326/0003-4819-143-10-200511150-00006.
- Hellings, Willem E., Frans L. Moll, Jean-Paul P. M. De Vries, Rob G. A. Ackerstaff, Kees A. Seldenrijk, Rosemarie Met, Evelyn Velema, Wouter J. M. Derksen, Dominique P. V. De Kleijn, and Gerard Pasterkamp. 2008. "Atherosclerotic Plaque Composition and Occurrence of Restenosis After Carotid Endarterectomy." JAMA 299 (5): 547–54. https: //doi.org/10.1001/jama.299.5.547.
- Heneghan, Rachel E., Shahram Aarabi, Elina Quiroga, Martin L. Gunn, Niten Singh, and Benjamin W. Starnes. 2016. "Call for a New Classification System and Treatment Strategy in Blunt Aortic Injury." *Journal of Vascular Surgery* 64 (1): 171–76. https://doi.org/10. 1016/j.jvs.2016.02.047.
- Henke, Peter K. 2009. "Contemporary Management of Acute Limb Ischemia: Factors Associated with Amputation and in-Hospital Mortality." Seminars in Vascular Surgery 22 (1): 34–40. https://doi.org/10.1053/j.semvascsurg.2009.01.002.
- Hicks, Caitlin W., Joseph K. Canner, Nestoras Mathioudakis, Ronald Sherman, Mahmoud B. Malas, James H. Black, and Christopher J. Abularrage. 2018. "The Society for Vascular Surgery Wound, Ischemia, and Foot Infection (WIfI) Classification Independently Predicts Wound Healing in Diabetic Foot Ulcers." Journal of Vascular Surgery 68 (4): 1096–1103. https://doi.org/10.1016/j.jvs.2017.12.079.
- Hicks, Caitlin W., Katherine Talbott, Joseph K. Canner, Umair Qazi, Isibor Arhuidese, Natalia O. Glebova, Julie A. Freischlag, Bruce A. Perler, and Mahmoud B. Malas. 2015. "Risk of Disease Progression in Patients with Moderate Asymptomatic Carotid Artery Stenosis: Implications of Tobacco Use and Dual Antiplatelet Therapy." Annals of Vascular Surgery 29 (1): 1–8. https://doi.org/10.1016/j.avsg.2014.02.007.
- Higashi, Yukihito, Kensuke Noma, Masao Yoshizumi, and Yasuki Kihara. 2009. "Endothelial Function and Oxidative Stress in Cardiovascular Diseases." *Circulation Journal: Official Journal of the Japanese Circulation Society* 73 (3): 411–18. https://doi.org/10.1253/circj. cj-08-1102.
- Higdon, Mark L., and Jennifer A. Higdon. 2006. "Treatment of Oncologic Emergencies." American Family Physician 74 (11): 1873–80.
- Hiltunen, T. P., J. S. Luoma, T. Nikkari, and S. Ylä-Herttuala. 1998. "Expression of LDL Receptor, VLDL Receptor, LDL Receptor-Related Protein, and Scavenger Receptor in Rabbit Atherosclerotic Lesions: Marked Induction of Scavenger Receptor and VLDL Receptor Expression During Lesion Development." *Circulation* 97 (11): 1079–86. https: //doi.org/10.1161/01.cir.97.11.1079.

- Hingorani, Anil, Glenn M. LaMuraglia, Peter Henke, Mark H. Meissner, Lorraine Loretz, Kathya M. Zinszer, Vickie R. Driver, et al. 2016. "The Management of Diabetic Foot: A Clinical Practice Guideline by the Society for Vascular Surgery in Collaboration with the American Podiatric Medical Association and the Society for Vascular Medicine." Journal of Vascular Surgery 63 (2): 3S-21S. https://doi.org/10.1016/j.jvs.2015.10.003.
- Hiramoto, Jade, Kimberley J. Hansen, Xian Mang Pan, Matthew S. Edwards, Rajiv Sawhney, and Joseph H. Rapp. 2005. "Atheroemboli During Renal Artery Angioplasty: An Ex Vivo Study." Journal of Vascular Surgery 41 (6): 1026–30. https://doi.org/10.1016/j.jvs.2005. 02.042.
- Hirsch, Alan T., Ziv J. Haskal, Norman R. Hertzer, Curtis W. Bakal, Mark A. Creager, Jonathan L. Halperin, Loren F. Hiratzka, et al. 2006. "ACC/AHA 2005 Practice Guidelines for the Management of Patients With Peripheral Arterial Disease (Lower Extremity, Renal, Mesenteric, and Abdominal Aortic): A Collaborative Report from the American Association for Vascular Surgery/Society for Vascular Surgery,* Society for Cardiovascular Angiography and Interventions, Society for Vascular Medicine and Biology, Society of Interventional Radiology, and the ACC/AHA Task Force on Practice Guidelines (Writing Committee to Develop Guidelines for the Management of Patients With Peripheral Arterial Disease): Endorsed by the American Association of Cardiovascular and Pulmonary Rehabilitation; National Heart, Lung, and Blood Institute; Society for Vascular Nursing; TransAtlantic Inter-Society Consensus; and Vascular Disease Foundation." *Circulation* 113 (11). https://doi.org/10.1161/CIRCULATIONAHA.106.174526.
- Hirshfeld, John W., Stephen Balter, Jeffrey A. Brinker, Morton J. Kern, Lloyd W. Klein, Bruce D. Lindsay, Carl L. Tommaso, et al. 2005. "ACCF/AHA/HRS/SCAI Clinical Competence Statement on Physician Knowledge to Optimize Patient Safety and Image Quality in Fluoroscopically Guided Invasive Cardiovascular Procedures: A Report of the American College of Cardiology Foundation/American Heart Association/American College of Physicians Task Force on Clinical Competence and Training." *Circulation* 111 (4): 511–32. https://doi.org/10.1161/01.CIR.0000157946.29224.5D.
- Ho, Vivian, Douglas Wirthlin, Huifeng Yun, and Jeroan Allison. 2005. "Physician Supply, Treatment, and Amputation Rates for Peripheral Arterial Disease." Journal of Vascular Surgery 42 (1): 81–87. https://doi.org/10.1016/j.jvs.2005.03.023.
- Hoballah, Jamal J. 2021. "Inferior Mesenteric Artery Reimplantation." In Vascular Reconstructions, edited by Jamal J. Hoballah and Carlos F. Bechara, 373–79. New York, NY: Springer New York. https://doi.org/10.1007/978-1-0716-1089-3_17.
- Hoffmann, Ulrich, James M. Edwards, Stephen Carter, Martin L. Goldman, John D. Harley, Molly J. Zaccardi, and D. Eugene Strandness. 1991. "Role of Duplex Scanning for the Detection of Atherosclerotic Renal Artery Disease." *Kidney International* 39 (6): 1232–39. https://doi.org/10.1038/ki.1991.156.
- Hooten, W. M., H. K. Bruns, and J. T. Hays. 1998. "Inpatient Treatment of Severe Nicotine Dependence in a Patient with Thromboangiitis Obliterans (Buerger's Disease)." Mayo Clinic Proceedings 73 (6): 529–32. https://doi.org/10.4065/73.6.529.
- Hossny, Ahmed. 2014. "Partial Aneurysmectomy for Salvage of Autogenous Arteriovenous Fistula with Complicated Venous Aneurysms." Journal of Vascular Surgery 59 (4): 1073–

77. https://doi.org/10.1016/j.jvs.2013.10.083.

- Hrbáč, Tomáš, Vladimír Beneš, Pavel Širůček, Tomáš Jonszta, Roman Herzig, Václav Procházka, and David Skoloudík. 2012. "Safety and Efficacy of Surgical Treatment of Carotid Stump Syndrome: Pilot Study." Annals of Vascular Surgery 26 (6): 797–801. https://doi.org/10.1016/j.avsg.2011.11.034.
- Huang, Ying, Peter Gloviczki, Audra A. Duncan, Manju Kalra, Tanya L. Hoskin, Gustavo S. Oderich, Michael A. McKusick, and Thomas C. Bower. 2008. "Common Iliac Artery Aneurysm: Expansion Rate and Results of Open Surgical and Endovascular Repair." *Journal of Vascular Surgery* 47 (6): 1203-1210; discussion 1210-1211. https://doi.org/10.1016/ j.jvs.2008.01.050.
- Huber, Thomas S., Christa M. Hirneise, W.Anthony Lee, Timothy C. Flynn, and James M. Seeger. 2004. "Outcome After Autogenous Brachial-Axillary Translocated Superficial Femoropopliteal Vein Hemodialysis Access." *Journal of Vascular Surgery* 40 (2): 311–18. https://doi.org/10.1016/j.jvs.2004.04.018.
- Huber, Thomas S., C. Keith Ozaki, Timothy C. Flynn, W. Anthony Lee, Scott A. Berceli, Christa M. Hirneise, Lori M. Carlton, Jeffrey W. Carter, Edward A. Ross, and James M. Seeger. 2002. "Prospective Validation of an Algorithm to Maximize Native Arteriovenous Fistulae for Chronic Hemodialysis Access." Journal of Vascular Surgery 36 (3): 452–59. https://doi.org/10.1067/mva.2002.127342.
- Hughes, G. Chad, Nicholas D. Andersen, and Richard L. McCann. 2013. "Management of Acute Type b Aortic Dissection." *The Journal of Thoracic and Cardiovascular Surgery* 145 (3 Suppl): S202–207. https://doi.org/10.1016/j.jtcvs.2012.11.078.
- Hughes, John R., Lindsay F. Stead, Jamie Hartmann-Boyce, Kate Cahill, and Tim Lancaster. 2014. "Antidepressants for Smoking Cessation." The Cochrane Database of Systematic Reviews, no. 1 (January): CD000031. https://doi.org/10.1002/14651858.CD000031.pub4.
- Hughes, Kakra, Lucas Mota, Maria Nunez, Neil Sehgal, and Gezzer Ortega. 2019. "The Effect of Income and Insurance on the Likelihood of Major Leg Amputation." Journal of Vascular Surgery 70 (2): 580–87. https://doi.org/10.1016/j.jvs.2018.11.028.
- Humphries, Misty D, and Julie A Freischlag. 2019a. "123: Thoracic Outlet Syndrome: Venous." In Rutherford's Vascular Surgery and Endovascular Therapy, 2:1648–1655.e2. Elsevier Inc. https://doi.org/10.1016/j.mpsur.2018.12.010.
- ———. 2019b. "124: Thoracic Outlet Syndrome : Surgical Decompression of the Thoracic Outlet." In *Rutherford's Vascular Surgery and Endovascular Therapy*, Ninth. Elsevier Inc.
- Hunter, Glenn C, Scott S. Berman, and Eric M. Walser. 2019. "174. Acquired Arteriovenous Fistulas." In Rutherford's Vascular Surgery and Endovascular Therapy, Ninth, 20. Elsevier Inc.
- Hussain, Mohamad A., Mohammed Al-Omran, Mark A. Creager, Sonia S. Anand, Subodh Verma, and Deepak L. Bhatt. 2018. "Antithrombotic Therapy for Peripheral Artery Disease: Recent Advances." Journal of the American College of Cardiology 71 (21): 2450– 67. https://doi.org/10.1016/j.jacc.2018.03.483.
- Hwang, Ji Young. 2017. "Doppler Ultrasonography of the Lower Extremity Arteries: Anatomy and Scanning Guidelines." *Ultrasonography* 36 (2): 111–19. https://doi.org/10.14366/usg. 16054.

- IAC. 2021. "IAC Standards and Guidelines for Vascular Testing Accreditation (2021)." https://intersocietal.org/wp-content/uploads/2021/11/IACVascularTestingStandards2021.pdf.
- Iafrati, Mark D, and Thomas F O Donnell. 2019. "154: Varicose Veins : Surgical Treatment." In Rutherford's Vascular Surgery and Endovacular Therapy. Vol. 1. Elsevier Inc. http: //dx.doi.org/10.1016/B978-1-4557-5304-8.00057-1.
- Ikram, M. Arfan, Guy G. O. Brusselle, Sarwa Darwish Murad, Cornelia M. van Duijn, Oscar H. Franco, André Goedegebure, Caroline C. W. Klaver, et al. 2017. "The Rotterdam Study: 2018 Update on Objectives, Design and Main Results." *European Journal of Epidemiology* 32 (9): 807–50. https://doi.org/10.1007/s10654-017-0321-4.
- Illig, Karl A. 2019. "57. Upper Extremity Vascular Exposure." In *Rutherford's Vascular Surgery and Endovascular Therapy*, 12. Elsevier Inc.
- Illig, Karl A., and Adam J. Doyle. 2010. "A Comprehensive Review of Paget-Schroetter Syndrome." *Journal of Vascular Surgery* 51 (6): 1538–47. https://doi.org/10.1016/j.jvs. 2009.12.022.
- Illuminati, Giulio, Giulia Pizzardi, Francesco G. Calio, Federica Masci, Rocco Pasqua, Francesca Frezzotti, and Simone Peschillo. 2018. "Results of Subclavian to Carotid Artery Bypass for Occlusive Disease of the Common Carotid Artery: A Retrospective Cohort Study." International Journal of Surgery (London, England) 53 (May): 111–16. https://doi.org/10.1016/j.ijsu.2018.03.038.
- Investigators, ASTRAL. 2009. "Revascularization Versus Medical Therapy for Renal-Artery Stenosis." The New England Journal of Medicine, 10.
- Investigators, The STILE. 1994. "Results of a Prospective Randomized Trial Evaluating Surgery Versus Thrombolysis for Ischemia of the Lower Extremity. The STILE Trial." Annals of Surgery 220 (3): 251–68.
- Ishimaru, Shin. 2004. "Endografting of the Aortic Arch." Journal of Endovascular Therapy: An Official Journal of the International Society of Endovascular Specialists 11 Suppl 2 (December): II62–71. https://doi.org/10.1177/15266028040110S614.
- Itoga, Nathan K., Daniel S. Tawfik, Charles K. Lee, Satoshi Maruyama, Nicholas J. Leeper, and Tara I. Chang. 2018. "Association of Blood Pressure Measurements with Peripheral Artery Disease Events." *Circulation* 138 (17): 1805–14. https://doi.org/10.1161/ CIRCULATIONAHA.118.033348.
- Iwakoshi, Shinichi, Shigeo Ichihashi, Takeshi Inoue, Takashi Inoue, Futoshi Sakuragi, Shoji Sakaguchi, and Kimihiko Kichikawa. 2019. "Measuring the Greater Curvature Length of Virtual Stent Graft Can Provide Accurate Prediction of Stent Graft Position for Thoracic Endovascular Aortic Repair." Journal of Vascular Surgery 69 (4): 1021–27. https://doi.org/10.1016/j.jvs.2018.07.068.
- Jack L Cronenwett, Alik Farber, and Erica L. Mitchell. 2020. Vascular Decision Making: Medical, Endovascular, Surgical. Wolters Kluwer.
- Janssen, Stan P., Emile H. Comans, Alexandre E. Voskuyl, Willem Wisselink, and Yvo M. Smulders. 2008. "Giant Cell Arteritis: Heterogeneity in Clinical Presentation and Imaging Results." Journal of Vascular Surgery 48 (4): 1025–31. https://doi.org/10.1016/j.jvs.2008. 04.054.
- Jaquinandi, Vincent, Jean Picquet, Jean-Louis Saumet, Peyman Benharash, Georges Lefthe-

riotis, and Pierre Abraham. 2008. "Functional Assessment at the Buttock Level of the Effect of Aortobifemoral Bypass Surgery." Annals of Surgery 247 (5): 869–76. https://doi.org/10.1097/SLA.0b013e31816bcd75.

- Jenkins, Michael P., Stéphan Haulon, Roy K. Greenberg, and Thomas L. Forbes. 2011. "Debate Regarding the Best Surgical Option for Type IV Thoracoabdominal Aortic Aneurysms." Journal of Vascular Surgery 54 (1): 258–67. https: //doi.org/10.1016/j.jvs.2011.05.024.
- Jenkins, Thomas L., Mary Grace Baker, Aparna R. Baheti, Aditya M. Sharma, James T. Patrie, J. Fritz Angle, and Alan H. Matsumoto. 2015. "Outcomes of Hypertensive Patients with Renal Fibromuscular Dysplasia Compared with Patients with Concomitant Atherosclerotic Renal Artery Stenosis Following Endovascular Therapy." Journal of Vascular and Interventional Radiology 26 (5): 625–33. https://doi.org/10.1016/j.jvir.2015.01.027.
- Jeyabalan, Geetha, Justin R. Wallace, Rabih Antoine Chaer, Steven A. Leers, Luke Keith Marone, and Michel S. Makaroun. 2014. "Endovascular Strategies for Treatment of Embolizing Thoracoabdominal Aortic Lesions." Journal of Vascular Surgery 59 (5): 1256–64. https://doi.org/10.1016/j.jvs.2013.11.068.
- Jha, Manish K., Arman Qamar, Muthiah Vaduganathan, Dennis S. Charney, and James W. Murrough. 2019. "Screening and Management of Depression in Patients with Cardiovascular Disease: JACC State-of-the-Art Review." Journal of the American College of Cardiology 73 (14): 1827–45. https://doi.org/10.1016/j.jacc.2019.01.041.
- Joh, Jin Hyun, Woo-Shik Kim, In Mok Jung, Ki-Hyuk Park, Taeseung Lee, Jin Mo Kang, and Consensus Working Group. 2014. "Consensus for the Treatment of Varicose Vein with Radiofrequency Ablation." Vascular Specialist International 30 (4): 105–12. https: //doi.org/10.5758/vsi.2014.30.4.105.
- Johnson, D. R., J. Kyriou, E. J. Morton, A. Clifton, M. Fitzgerald, and E. Macsweeney. 2001. "Radiation Protection in Interventional Radiology." *Clinical Radiology* 56 (2): 99–106. https://doi.org/10.1053/crad.2000.0640.
- Jones, A. Kyle, and Alexander S. Pasciak. 2011. "Calculating the Peak Skin Dose Resulting from Fluoroscopically Guided Interventions. Part i: Methods." *Journal of Applied Clinical Medical Physics* 12 (4): 3670. https://doi.org/10.1120/jacmp.v12i4.3670.
- Jones, John E., Marvin D. Atkins, David C. Brewster, Thomas K. Chung, Christopher J. Kwolek, Glenn M. LaMuraglia, Thomas M. Hodgman, and Richard P. Cambria. 2007. "Persistent Type 2 Endoleak After Endovascular Repair of Abdominal Aortic Aneurysm Is Associated with Adverse Late Outcomes." Journal of Vascular Surgery 46 (1): 1–8. https://doi.org/10.1016/j.jvs.2007.02.073.
- Jongsma, Hidde, Joost A. Bekken, George P. Akkersdijk, Sanne E. Hoeks, Hence J. Verhagen, and Bram Fioole. 2017. "Angiosome-Directed Revascularization in Patients with Critical Limb Ischemia." Journal of Vascular Surgery 65 (4): 1208–1219.e1. https://doi.org/10. 1016/j.jvs.2016.10.100.
- Jongsma, Hidde, Joost A. Bekken, Fons van Buchem, Wouter J. J. Bekkers, Fahim Azizi, and Bram Fioole. 2016. "Secondary Interventions in Patients with Autologous Infrainguinal Bypass Grafts Strongly Improve Patency Rates." Journal of Vascular Surgery 63 (2): 385– 90. https://doi.org/10.1016/j.jvs.2015.08.100.

- Jordan, William D., Manish Mehta, David Varnagy, William M. Moore, Frank R. Arko, James Joye, Kenneth Ouriel, Jean-Paul de Vries, and Aneurysm Treatment using the Heli-FX Aortic Securement System Global Registry (ANCHOR) Workgroup Members. 2014. "Results of the ANCHOR Prospective, Multicenter Registry of EndoAnchors for Type Ia Endoleaks and Endograft Migration in Patients with Challenging Anatomy." Journal of Vascular Surgery 60 (4): 885–892.e2. https://doi.org/10.1016/j.jvs.2014.04.063.
- Jun Yan Wee, Ian, Ismail Heyder Mohamed, Amit Patel, and Andrew M. T. L. Choong. 2018. "A Systematic Review and Meta-Analysis of One-Stage Versus Two-Stage Brachiobasilic Arteriovenous Fistula Creation." Journal of Vascular Surgery 68 (1): 285–97. https://doi. org/10.1016/j.jvs.2018.03.428.
- Kabbani, Loay, Guillermo A. Escobar, Farah Mansour, Thomas W. Wakefield, and Peter K. Henke. 2011. "Longevity and Outcomes of Axillary Valve Transplantation for Severe Lower Extremity Chronic Venous Insufficiency." Annals of Vascular Surgery 25 (4): 496– 501. https://doi.org/10.1016/j.avsg.2011.02.002.
- Kabnick, Lowell S., Mikel Sadek, Haraldur Bjarnason, Dawn M. Coleman, Ellen D. Dillavou, Anil P. Hingorani, Brajesh K. Lal, Peter F. Lawrence, Rafael D. Malgor, and Alessandra Puggioni. 2021. "Classification and Treatment of Endothermal Heat-Induced Thrombosis: Recommendations from the American Venous Forum and the Society for Vascular Surgery." Journal of Vascular Surgery. Venous and Lymphatic Disorders 9 (1): 6–22. https://doi. org/10.1016/j.jvsv.2020.06.008.
- Kabnick, Lowell S, and Mikel Sadek. 2019. "155. Varicose Veins : Endovenous Ablation and Sclerotherapy." In Rutherford's Vascular Surgery and Endovascular Therapy. Elsevier Inc. http://dx.doi.org/10.1016/B978-1-4557-5304-8.00058-3.
- Kahn, Susan R., Anthony J. Comerota, Mary Cushman, Natalie S. Evans, Jeffrey S. Ginsberg, Neil A. Goldenberg, Deepak K. Gupta, et al. 2014. "The Postthrombotic Syndrome: Evidence-Based Prevention, Diagnosis, and Treatment Strategies: A Scientific Statement from the American Heart Association." *Circulation* 130 (18): 1636–61. https://doi.org/10. 1161/CIR.000000000000130.
- Kakish, M. E., M. M. Abu-Yousef, P. B. Brown, N. G. Warnock, T. J. Barloon, and R. E. Pelsang. 1996. "Pulsatile Lower Limb Venous Doppler Flow: Prevalence and Value in Cardiac Disease Diagnosis." Journal of Ultrasound in Medicine: Official Journal of the American Institute of Ultrasound in Medicine 15 (11): 747–53. https://doi.org/10.7863/jum.1996.15.11.747.
- Kakkos, Stavros K., Manjit Gohel, Niels Baekgaard, Rupert Bauersachs, Sergi Bellmunt-Montoya, Stephen A. Black, Arina J. ten Cate-Hoek, et al. 2021. "Editor's Choice – European Society for Vascular Surgery (ESVS) 2021 Clinical Practice Guidelines on the Management of Venous Thrombosis." European Journal of Vascular and Endovascular Surgery 61 (1): 9–82. https://doi.org/10.1016/j.ejvs.2020.09.023.
- Kaku, Yasuhiko, Shin-ichi Yoshimura, and Jouji Kokuzawa. 2004. "Factors Predictive of Cerebral Hyperperfusion After Carotid Angioplasty and Stent Placement." AJNR: American Journal of Neuroradiology 25 (8): 1403–8. https://www.ncbi.nlm.nih.gov/pmc/articles/ PMC7975476/.
- Kalaria, Vijay G., Sony Jacob, William Irwin, and Robert M. Schainfeld. 2005. "Duplex

Ultrasonography of Vertebral and Subclavian Arteries." Journal of the American Society of Echocardiography: Official Publication of the American Society of Echocardiography 18 (10): 1107–11. https://doi.org/10.1016/j.echo.2005.06.005.

- Kalish, Jeffrey, and Allen Hamdan. 2010. "Management of Diabetic Foot Problems." Journal of Vascular Surgery 51 (2): 476–86. https://doi.org/10.1016/j.jvs.2009.08.043.
- Kalra, Manju, Haraldur Bjarnason, and Peter Gloviczki. 2019. "162. Superior Vena Cava Occlusion and Management." In *Rutherford's Vascular Surgery and Endovascular Therapy*, 2130–2144.e3. Elsevier Inc. https://doi.org/10.32388/0d18ww.
- Kalra, Manju, Indrani Sen, and Peter Gloviczki. 2018. "Endovenous and Operative Treatment of Superior Vena Cava Syndrome." The Surgical Clinics of North America 98 (2): 321–35. https://doi.org/10.1016/j.suc.2017.11.013.
- Kamman, Arnoud V., Jan Brunkwall, Eric L. Verhoeven, Robin H. Heijmen, Santi Trimarchi, and ADSORB trialists. 2017. "Predictors of Aortic Growth in Uncomplicated Type b Aortic Dissection from the Acute Dissection Stent Grafting or Best Medical Treatment (ADSORB) Database." Journal of Vascular Surgery 65 (4): 964–971.e3. https://doi.org/ 10.1016/j.jvs.2016.09.033.
- Kang, Jeanwan, Mark F. Conrad, Virendra I. Patel, Shankha Mukhopadhyay, Ashu Garg, Matthew R. Cambria, Glenn M. LaMuraglia, and Richard P. Cambria. 2014. "Clinical and Anatomic Outcomes After Carotid Endarterectomy." *Journal of Vascular Surgery* 59 (4): 944–49. https://doi.org/10.1016/j.jvs.2013.10.059.
- Kannel, W. B., and D. L. McGee. 1985. "Update on Some Epidemiologic Features of Intermittent Claudication: The Framingham Study." Journal of the American Geriatrics Society 33 (1): 13–18. https://doi.org/10.1111/j.1532-5415.1985.tb02853.x.
- Kapetanios, Dimitrios, Nikolaos Kontopodis, Dimitrios Mavridis, Richard G. McWilliams, Athanasios D. Giannoukas, and George A. Antoniou. 2019. "Meta-Analysis of the Accuracy of Contrast-Enhanced Ultrasound for the Detection of Endoleak After Endovascular Aneurysm Repair." Journal of Vascular Surgery 69 (1): 280–294.e6. https://doi.org/10. 1016/j.jvs.2018.07.044.
- Karam, Jocelyne G., Lisel Loney-Hutchinson, and Samy I. McFarlane. 2008. "High-Dose Atorvastatin After Stroke or Transient Ischemic Attack: The Stroke Prevention by Aggressive Reduction in Cholesterol Levels (SPARCL) Investigators." Journal of the CardioMetabolic Syndrome 3 (1): 68–69. https://doi.org/10.1111/j.1559-4572.2008.07967.x.
- Karch, L. A., K. J. Hodgson, M. A. Mattos, W. T. Bohannon, D. E. Ramsey, and R. B. McLafferty. 2000. "Adverse Consequences of Internal Iliac Artery Occlusion During Endovascular Repair of Abdominal Aortic Aneurysms." *Journal of Vascular Surgery* 32 (4): 676–83. https://doi.org/10.1067/mva.2000.109750.
- Karthikesalingam, A., W. Al-Jundi, D. Jackson, J. R. Boyle, J. D. Beard, P. J. E. Holt, and M. M. Thompson. 2012. "Systematic Review and Meta-Analysis of Duplex Ultrasonography, Contrast-Enhanced Ultrasonography or Computed Tomography for Surveillance After Endovascular Aneurysm Repair." The British Journal of Surgery 99 (11): 1514–23. https://doi.org/10.1002/bjs.8873.
- Kashyap, Vikram S., Alexander H. King, Mazin I. Foteh, Matthew Janko, Jeffrey Jim, Raghu L. Motaganahalli, Jeffrey M. Apple, Saideep Bose, and Norman H. Kumins. 2019. "A

Multi-Institutional Analysis of Transcarotid Artery Revascularization Compared to Carotid Endarterectomy." *Journal of Vascular Surgery* 70 (1): 123–29. https://doi.org/10.1016/j.jvs.2018.09.060.

- Katsanos, Konstantinos, Tarun Sabharwal, Tom Carrell, Renato Dourado, and Andreas Adam. 2009. "Peripheral Endografts for the Treatment of Traumatic Arterial Injuries." *Emergency Radiology* 16 (3): 175–84. https://doi.org/10.1007/s10140-008-0771-9.
- Katsanos, Konstantinos, Stavros Spiliopoulos, Panagiotis Kitrou, Miltiadis Krokidis, and Dimitrios Karnabatidis. 2018. "Risk of Death Following Application of Paclitaxel-Coated Balloons and Stents in the Femoropopliteal Artery of the Leg: A Systematic Review and Meta-Analysis of Randomized Controlled Trials." Journal of the American Heart Association 7 (24): e011245. https://doi.org/10.1161/JAHA.118.011245.
- Kauvar, David S, and Larry W Kraiss. 2020. 184: Vascular Trauma: Extremity. Rutherford's Vascular Surgery and Endovascular Therapy, 2-Volume Set. Elsevier Inc. https://doi.org/ 10.1016/B978-0-323-42791-3.00184-5.
- Kauvar, David S., and Brandon W. Propper. 2022. "22. Lower Extremity Vascular Trauma." In, 273–87. Elsevier. https://doi.org/10.1016/B978-0-323-69766-8.00022-6.
- Kawachi, I., G. A. Colditz, M. J. Stampfer, W. C. Willett, J. E. Manson, B. Rosner, F. E. Speizer, and C. H. Hennekens. 1993. "Smoking Cessation and Decreased Risk of Stroke in Women." JAMA 269 (2): 232–36.
- Kawarada, Osami, Akihiro Higashimori, Miyuki Noguchi, Naoto Waratani, Masafumi Yoshida, Masahiko Fujihara, Yoshiaki Yokoi, Yasuhiro Honda, and Peter J. Fitzgerald. 2013. "Duplex Criteria for in-Stent Restenosis in the Superficial Femoral Artery." Catheterization and Cardiovascular Interventions: Official Journal of the Society for Cardiac Angiography and Interventions 81 (4): E199–205. https://doi.org/10.1002/ccd.24509.
- Kayssi, Ahmed, Talal Al-Atassi, George Oreopoulos, Graham Roche-Nagle, Kong Teng Tan, and Dheeraj K Rajan. 2016. "Drug-eluting Balloon Angioplasty Versus Uncoated Balloon Angioplasty for Peripheral Arterial Disease of the Lower Limbs." *The Cochrane Database* of Systematic Reviews 2016 (8): CD011319. https://doi.org/10.1002/14651858.CD011319. pub2.
- KAYSSI, AHMED, LEE C. ROGERS, and RICHARD F. NEVILLE. 2019. "113: General Considerations of Diabetic Foot Ulcers." In Rutherford's Vascular Surgery and Endovascular Therapy. Elsevier Inc. https://doi.org/10.1016/B978-0-323-42791-3.00113-4.
- Kazmers, A. 1998. "Operative Management of Acute Mesenteric Ischemia. Part 1." Annals of Vascular Surgery 12 (2): 187–97. https://doi.org/10.1007/s100169900139.
- Kearon, Clive, Elie A. Akl, Joseph Ornelas, Allen Blaivas, David Jimenez, Henri Bounameaux, Menno Huisman, Christopher S. King, Timothy A. Morris, Namita Sood, Scott M. Stevens, et al. 2016. "Antithrombotic Therapy for VTE Disease." *Chest* 149 (2): 315–52. https: //doi.org/10.1016/j.chest.2015.11.026.
- Kearon, Clive, Elie A. Akl, Joseph Ornelas, Allen Blaivas, David Jimenez, Henri Bounameaux, Menno Huisman, Christopher S. King, Timothy A. Morris, Namita Sood, and et al. 2016.
 "Antithrombotic Therapy for VTE Disease: CHEST Guideline and Expert Panel Report." Chest 149 (2): 315–52. https://doi.org/10.1016/j.chest.2015.11.026.
- Keen, Helen I., Janakan Krishnarajah, Timothy R. Bates, and Gerald F. Watts. 2014. "Statin

Myopathy: The Fly in the Ointment for the Prevention of Cardiovascular Disease in the 21st Century?" *Expert Opinion on Drug Safety* 13 (9): 1227–39. https://doi.org/10.1517/14740338.2014.937422.

- Kelso, Rebecca L., Sean P. Lyden, Brett Butler, Roy K. Greenberg, Matthew J. Eagleton, and Daniel G. Clair. 2009. "Late Conversion of Aortic Stent Grafts." *Journal of Vascular Surgery* 49 (3): 589–95. https://doi.org/10.1016/j.jvs.2008.10.020.
- Kempe, Kelly, Brett Starr, Jeanette M. Stafford, Arsalla Islam, Ashley Mooney, Emily Lagergren, Matthew A. Corriere, and Matthew S. Edwards. 2014. "Results of Surgical Management of Acute Thromboembolic Lower Extremity Ischemia." Journal of Vascular Surgery 60 (3): 702–7. https://doi.org/10.1016/j.jvs.2014.03.273.
- Kent, K. Craig, Robert M. Zwolak, Natalia N. Egorova, Thomas S. Riles, Andrew Manganaro, Alan J. Moskowitz, Annetine C. Gelijns, and Giampaolo Greco. 2010. "Analysis of Risk Factors for Abdominal Aortic Aneurysm in a Cohort of More Than 3 Million Individuals." *Journal of Vascular Surgery* 52 (3): 539–48. https://doi.org/10.1016/j.jvs.2010.05.090.
- Keudell, Arvind G. von, Michael J. Weaver, Paul T. Appleton, Donald S. Bae, George S. M. Dyer, Marilyn Heng, Jesse B. Jupiter, and Mark S. Vrahas. 2015. "Diagnosis and Treatment of Acute Extremity Compartment Syndrome." *Lancet (London, England)* 386 (10000): 1299–1310. https://doi.org/10.1016/S0140-6736(15)00277-9.
- Khandanpour, N., Y. K. Loke, F. J. Meyer, B. Jennings, and M. P. Armon. 2009. "Homocysteine and Peripheral Arterial Disease: Systematic Review and Meta-Analysis." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 38 (3): 316–22. https://doi.org/10.1016/j.ejvs.2009.05.007.
- Killewich, Lois A., Garietta Falls, Tara M. Mastracci, and Kellie R. Brown. 2011. "Factors Affecting Radiation Injury." Journal of Vascular Surgery 53 (1): 9S-14S. https://doi.org/ 10.1016/j.jvs.2010.07.025.
- Kim, Kyung Won, Jeong Yeon Cho, Seung Hyup Kim, Jeong-Hee Yoon, Dae Sik Kim, Jin Wook Chung, and Jae Hyung Park. 2011. "Diagnostic Value of Computed Tomographic Findings of Nutcracker Syndrome: Correlation with Renal Venography and Renocaval Pressure Gradients." *European Journal of Radiology* 80 (3): 648–54. https://doi.org/10. 1016/j.ejrad.2010.08.044.
- Kim, S. H., S. W. Cho, H. D. Kim, J. W. Chung, J. H. Park, and M. C. Han. 1996. "Nutcracker Syndrome: Diagnosis with Doppler US." *Radiology* 198 (1): 93–97. https://doi.org/10. 1148/radiology.198.1.8539413.
- Kim, Seung Hyup. 2019. "Doppler US and CT Diagnosis of Nutcracker Syndrome." Korean Journal of Radiology 20 (12): 1627–37. https://doi.org/10.3348/kjr.2019.0084.
- Kim, Young Chul, Jong Yun Won, Sun Young Choi, Heung-kyu Ko, Kwang-Hun Lee, Do Yun Lee, Byung-Chul Kang, and Seung-Jung Kim. 2009. "Percutaneous Treatment of Central Venous Stenosis in Hemodialysis Patients: Long-Term Outcomes." Cardiovascular and Interventional Radiology 32 (2): 271–78. https://doi.org/10.1007/s00270-009-9511-0.
- Kirkwood, Melissa L., Gary M. Arbique, Jeffrey B. Guild, Carlos Timaran, Jon A. Anderson, and R. James Valentine. 2015. "Deterministic Effects After Fenestrated Endovascular Aortic Aneurysm Repair." *Journal of Vascular Surgery* 61 (4): 902–6. https://doi.org/10. 1016/j.jvs.2014.11.044.

- Kirkwood, Melissa L., Gary M. Arbique, Jeffrey B. Guild, Carlos Timaran, Jayer Chung, Jon A. Anderson, and R. James Valentine. 2013. "Surgeon Education Decreases Radiation Dose in Complex Endovascular Procedures and Improves Patient Safety." Journal of Vascular Surgery 58 (3): 715–21. https://doi.org/10.1016/j.jvs.2013.04.004.
- Kirkwood, Melissa L., Gary M. Arbique, Jeffrey B. Guild, Carlos Timaran, R. James Valentine, and Jon A. Anderson. 2014. "Radiation-Induced Skin Injury After Complex Endovascular Procedures." Journal of Vascular Surgery 60 (3): 742–48. https://doi.org/10.1016/j.jvs. 2014.03.236.
- Kirkwood, Melissa L., Gary M. Arbique, Jeffrey B. Guild, Katie Zeng, Yin Xi, John Rectenwald, Jon A. Anderson, and Carlos Timaran. 2018. "Radiation Brain Dose to Vascular Surgeons During Fluoroscopically Guided Interventions Is Not Effectively Reduced by Wearing Lead Equivalent Surgical Caps." Journal of Vascular Surgery 68 (2): 567–71. https://doi.org/10.1016/j.jvs.2017.12.054.
- Kirsch, Jonathan D., Mahan Mathur, Michele H. Johnson, Gunabushanam Gowthaman, and Leslie M. Scoutt. 2013. "Advances in Transcranial Doppler US: Imaging Ahead." *Radiographics: A Review Publication of the Radiological Society of North America, Inc* 33 (1): E1–14. https://doi.org/10.1148/rg.331125071.
- Kistler, Justin M., Asif M. Ilyas, and Joseph J. Thoder. 2018. "Forearm Compartment Syndrome: Evaluation and Management." Hand Clinics 34 (1): 53–60. https://doi.org/10. 1016/j.hcl.2017.09.006.
- Klarin, Derek, Robert T. Lancaster, Emel Ergul, Daniel Bertges, Philip Goodney, Marc L. Schermerhorn, Richard P. Cambria, and Virendra I. Patel. 2016. "Perioperative and Long-Term Impact of Chronic Kidney Disease on Carotid Artery Interventions." Journal of Vascular Surgery 64 (5): 1295–1302. https://doi.org/10.1016/j.jvs.2016.04.038.
- Klausner, Jill Q., Peter F. Lawrence, Michael P. Harlander-Locke, Dawn M. Coleman, James C. Stanley, Naoki Fujimura, Nathan K. Itoga, et al. 2015. "The Contemporary Management of Renal Artery Aneurysms." *Journal of Vascular Surgery* 61 (4): 978–984.e1. https://doi.org/10.1016/j.jvs.2014.10.107.
- Knipp, Brian S., Guillermo A. Escobar, Sean English, Gilbert R. Upchurch, and Enrique Criado. 2010. "Endovascular Repair of Ruptured Aortic Aneurysms Using Carbon Dioxide Contrast Angiography." Annals of Vascular Surgery 24 (7): 845–50. https://doi.org/10. 1016/j.avsg.2010.05.013.
- Ko, Sae Hee, Dennis F. Bandyk, Kelley D. Hodgkiss-Harlow, Andrew Barleben, and John Lane. 2015. "Estimation of Brachial Artery Volume Flow by Duplex Ultrasound Imaging Predicts Dialysis Access Maturation." Journal of Vascular Surgery 61 (6): 1521–28. https: //doi.org/10.1016/j.jvs.2015.01.036.
- Kobayashi, M., M. Ito, A. Nakagawa, N. Nishikimi, and Y. Nimura. 1999. "Immunohistochemical Analysis of Arterial Wall Cellular Infiltration in Buerger's Disease (Endarteritis Obliterans)." Journal of Vascular Surgery 29 (3): 451–58. https://doi.org/10.1016/s0741-5214(99)70273-9.
- Kobayashi, Norihiro, Keisuke Hirano, Masahiro Yamawaki, Motoharu Araki, Tsuyoshi Sakai, Yasunari Sakamoto, Shinsuke Mori, Masakazu Tsutsumi, Yohsuke Honda, and Yoshiaki Ito. 2018. "Simple Classification and Clinical Outcomes of Angiographic Dissection After

Balloon Angioplasty for Femoropopliteal Disease." Journal of Vascular Surgery 67 (4): 1151–58. https://doi.org/10.1016/j.jvs.2017.08.092.

- Kovacs, M. J., S. R. Kahn, M. Rodger, D. R. Anderson, R. Andreou, J. E. Mangel, B. Morrow, A. M. Clement, and P. S. Wells. 2007. "A Pilot Study of Central Venous Catheter Survival in Cancer Patients Using Low-Molecular-Weight Heparin (Dalteparin) and Warfarin Without Catheter Removal for the Treatment of Upper Extremity Deep Vein Thrombosis (the Catheter Study)." Journal of Thrombosis and Haemostasis: JTH 5 (8): 1650–53. https://doi.org/10.1111/j.1538-7836.2007.02613.x.
- Kremkau, Frederick W. 2021. Sonography: Principles and Instruments. Tenth edition. Elsevier.
- Kret, Marcus R., Ronald L. Dalman, Jeffrey Kalish, and Matthew Mell. 2016. "Arterial Cutdown Reduces Complications After Brachial Access for Peripheral Vascular Intervention." *Journal of Vascular Surgery* 64 (1): 149–54. https://doi.org/10.1016/j.jvs.2016.02.019.
- Kris-Etherton, Penny M., William S. Harris, Lawrence J. Appel, and American Heart Association. Nutrition Committee. 2002. "Fish Consumption, Fish Oil, Omega-3 Fatty Acids, and Cardiovascular Disease." *Circulation* 106 (21): 2747–57. https://doi.org/10.1161/01. cir.0000038493.65177.94.
- Kristensen, Steen Dalby, Juhani Knuuti, Antti Saraste, Stefan Anker, Hans Erik Bøtker, Stefan De Hert, Ian Ford, et al. 2014. "2014 ESC/ESA Guidelines on Non-Cardiac Surgery: Cardiovascular Assessment and Management: The Joint Task Force on Non-Cardiac Surgery: Cardiovascular Assessment and Management of the European Society of Cardiology (ESC) and the European Society of Anaesthesiology (ESA)." European Heart Journal 35 (35): 2383–2431. https://doi.org/10.1093/eurheartj/ehu282.
- Kristmundsson, Thorarinn, Nuno Dias, Tim Resch, and Björn Sonesson. 2016. "Morphology of Small Abdominal Aortic Aneurysms Should Be Considered Before Continued Ultrasound Surveillance." Annals of Vascular Surgery 31 (February): 18–22. https://doi.org/10.1016/ j.avsg.2015.09.016.
- Kruger, Allan J., Philip J. Walker, Wallace J. Foster, Jason S. Jenkins, Nicholas S. Boyne, and Julie Jenkins. 2010. "Important Observations Made Managing Carotid Body Tumors During a 25-Year Experience." Journal of Vascular Surgery 52 (6): 1518–23. https://doi. org/10.1016/j.jvs.2010.06.153.
- Krupski, W. C., C. H. Selzman, R. Floridia, P. K. Strecker, M. R. Nehler, and T. A. Whitehill. 1998. "Contemporary Management of Isolated Iliac Aneurysms." Journal of Vascular Surgery 28 (1): 1-11; discussion 11-13. https://doi.org/10.1016/s0741-5214(98)70194-6.
- Kubota, Y., K. Kichikawa, H. Uchida, K. Nishimine, R. Hirohashi, and H. Ohishi. 1997. "Superselective Urokinase Infusion Therapy for Dorsalis Pedis Artery Occlusion in Buerger's Disease." *Cardiovascular and Interventional Radiology* 20 (5): 380–82. https://doi.org/10.1007/s002709900172.
- Kupelian, A. S., and M. S. B. Huda. 2007. "Pregnancy, Thrombophlebitis and Thromboembolism: What Every Obstetrician Should Know." Archives of Gynecology and Obstetrics 275 (3): 215–17. https://doi.org/10.1007/s00404-006-0217-0.
- Kurklinsky, Andrew K., and Thom W. Rooke. 2010. "Nutcracker Phenomenon and Nutcracker Syndrome." Mayo Clinic Proceedings 85 (6): 552–59. https://doi.org/10.4065/mcp.2009.

0586.

- Kurstjens, Ralph L. M., Timme M. A. J. van Vuuren, Mark A. F. de Wolf, Rick de Graaf, Carsten W. K. P. Arnoldussen, and Cees H. A. Wittens. 2016. "Abdominal and Pubic Collateral Veins as Indicators of Deep Venous Obstruction." Journal of Vascular Surgery. Venous and Lymphatic Disorders 4 (4): 426–33. https://doi.org/10.1016/j.jvsv.2016.06. 005.
- Kwolek, Christopher J., Michael R. Jaff, J. Ignacio Leal, L. Nelson Hopkins, Rasesh M. Shah, Todd M. Hanover, Sumaira Macdonald, et al. 2015. "Results of the ROADSTER Multicenter Trial of Transcarotid Stenting with Dynamic Flow Reversal." Journal of Vascular Surgery 62 (5): 1227–1234.e1. https://doi.org/10.1016/j.jvs.2015.04.460.
- Kwon, Deukwoo, Mark P. Little, and Donald L. Miller. 2011. "Reference Air Kerma and Kerma-Area Product as Estimators of Peak Skin Dose for Fluoroscopically Guided Interventions." *Medical Physics* 38 (7): 4196–4204. https://doi.org/10.1118/1.3590358.
- Kwon, Edward, Daniel Grabo, and George C. Velmahos. 2019. "Carotid Artery and Internal Jugular Vein Injuries." In Atlas of Surgical Techniques in Trauma, edited by Demetrics Demetriades, Kenji Inaba, and George Velmahos, 2nd ed., 51–58. Cambridge University Press. https://doi.org/10.1017/9781108698665.010.
- Kwong, Jonathan M, Caron B. Rockman, and Vikram S. Kashyap. n.d. "85. Splanchnic Artery Aneurysms." In *Rutherford's Vascular Surgery and Endovascular Therapy*, Ninth, 17. Elsevier Inc.
- Labropoulos, Nicos, Patrick T. Jasinski, Demetri Adrahtas, Antonios P. Gasparis, and Mark H. Meissner. 2017. "A Standardized Ultrasound Approach to Pelvic Congestion Syndrome." *Phlebology* 32 (9): 608–19. https://doi.org/10.1177/0268355516677135.
- Labropoulos, Nicos, Jay Tiongson, Landon Pryor, Apostolos K. Tassiopoulos, Steven S. Kang, M. Ashraf Mansour, and William H. Baker. 2003. "Definition of Venous Reflux in Lower-Extremity Veins." Journal of Vascular Surgery 38 (4): 793–98. https://doi.org/10.1016/ s0741-5214(03)00424-5.
- Laine, M T, S J Laukontaus, I Kantonen, and M Venermo. 2016. "Population-Based Study of Ruptured Abdominal Aortic Aneurysm." *British Journal of Surgery* 103 (12): 1634–39. https://doi.org/10.1002/bjs.10200.
- Laine, Matti T., Martin Björck, C. Barry Beiles, Zoltán Szeberin, Ian Thomson, Martin Altreuther, E. Sebastian Debus, Kevin Mani, Gábor Menyhei, and Maarit Venermo. 2017.
 "Few Internal Iliac Artery Aneurysms Rupture Under 4 Cm." Journal of Vascular Surgery 65 (1): 76–81. https://doi.org/10.1016/j.jvs.2016.06.109.
- Lajoie, Lidie. 2019. "12. Atherosclerotic Risk Factors: Diabetes." In *Rutherfords Vascular Surgery and Endovascular Therapy*, 12. Elsevier Inc.
- Lal, Brajesh K., Robert W. Hobson, Babak Tofighi, Indu Kapadia, Salvador Cuadra, and Zafar Jamil. 2008. "Duplex Ultrasound Velocity Criteria for the Stented Carotid Artery." Journal of Vascular Surgery 47 (1): 63–73. https://doi.org/10.1016/j.jvs.2007.09.038.
- LaMuraglia, Glenn M., Michael C. Stoner, David C. Brewster, Michael T. Watkins, Kendra L. Juhola, Christopher Kwolek, David J. Dorer, and Richard P. Cambria. 2005. "Determinants of Carotid Endarterectomy Anatomic Durability: Effects of Serum Lipids and Lipid-Lowering Drugs." Journal of Vascular Surgery 41 (5): 762–68. https://doi.org/10.

1016/j.jvs.2005.01.035.

- Landry, Gregory J. 2019. "141. Raynaud Phenomenon." In Rutherford Vascular Surgery and Endovascular Therapy. Elsevier Inc.
- Landry, Gregory J. 2013. "Current Medical and Surgical Management of Raynaud's Syndrome." Journal of Vascular Surgery 57 (6): 1710–16. https://doi.org/10.1016/j.jvs.2013. 03.012.
- Landry, Gregory J., Courtney J. Mostul, Daniel S. Ahn, Bryant J. McLafferty, Timothy K. Liem, Erica L. Mitchell, Enjae Jung, et al. 2018. "Causes and Outcomes of Finger Ischemia in Hospitalized Patients in the Intensive Care Unit." *Journal of Vascular Surgery* 68 (5): 1499–1504. https://doi.org/10.1016/j.jvs.2018.01.050.
- Lane, Risha, Amy Harwood, Lorna Watson, and Gillian C Leng. 2017. "Exercise for Intermittent Claudication." The Cochrane Database of Systematic Reviews 2017 (12): CD000990. https://doi.org/10.1002/14651858.CD000990.pub4.
- Langlois, Michel R., and Børge G. Nordestgaard. 2018. "Which Lipids Should Be Analyzed for Diagnostic Workup and Follow-up of Patients with Hyperlipidemias?" Current Cardiology Reports 20 (10): 88. https://doi.org/10.1007/s11886-018-1036-1.
- Laredo, James, Byung B Lee, and Richard F Neville. 2010. "Endovenous Thermal Ablation of the Anterior Accessory Great Saphenous Vein." *Endovascular Today*, March, 36–39.
- Laryea, Jonathan, and Bradley Champagne. 2013. "Venous Thromboembolism Prophylaxis." *Clinics in Colon and Rectal Surgery* 26 (3): 153–59. https://doi.org/10.1055/s-0033-1351130.
- Lather, Henry D., Heather L. Gornik, Jeffrey W. Olin, Xiaokui Gu, Steven T. Heidt, Esther S. H. Kim, Daniella Kadian-Dodov, et al. 2017. "Prevalence of Intracranial Aneurysm in Women with Fibromuscular Dysplasia." JAMA Neurology 74 (9): 1081–87. https: //doi.org/10.1001/jamaneurol.2017.1333.
- Lawenda, Brian D., Tammy E. Mondry, and Peter A. S. Johnstone. 2009. "Lymphedema: A Primer on the Identification and Management of a Chronic Condition in Oncologic Treatment." CA: A Cancer Journal for Clinicians 59 (1): 8–24. https://doi.org/10.3322/ caac.20001.
- Layton, K. F., D. F. Kallmes, H. J. Cloft, E. P. Lindell, and V. S. Cox. 2006. "Bovine Aortic Arch Variant in Humans: Clarification of a Common Misnomer." AJNR: American Journal of Neuroradiology 27 (7): 1541–42. https://www.ncbi.nlm.nih.gov/pmc/articles/ PMC7977516/.
- Lazarides, Miltos K., Christos Argyriou, Andreas Koutsoumpelis, Efstratios I. Georgakarakos, and George S. Georgiadis. 2018. "Thigh Arteriovenous Grafts. Quantitative Comparison with Alternative Options: A Meta-Analysis." *The Journal of Vascular Access* 19 (5): 430– 35. https://doi.org/10.1177/1129729818762991.
- Le Dorze, Matthieu, Adrien Bouglé, Stéphane Deruddre, and Jacques Duranteau. 2012. "Renal Doppler Ultrasound: A New Tool to Assess Renal Perfusion in Critical Illness." *Shock* (Augusta, Ga.) 37 (4): 360–65. https://doi.org/10.1097/SHK.0b013e3182467156.
- Le Joncour, Alexandre, Simon Soudet, Axelle Dupont, Olivier Espitia, Fabien Koskas, Philippe Cluzel, Pierre Yves Hatron, et al. 2018. "Long-term Outcome and Prognostic Factors of Complications in Thromboangiitis Obliterans (Buerger's Disease): A Multicenter Study of

224 Patients." Journal of the American Heart Association: Cardiovascular and Cerebrovascular Disease 7 (23): e010677. https://doi.org/10.1161/JAHA.118.010677.

- Le, Trong Binh, Taeg Ki Lee, Keun-Myoung Park, Yong Sun Jeon, Kee Chun Hong, and Soon Gu Cho. 2018. "Contralateral Deep Vein Thrombosis After Iliac Vein Stent Placement in Patients with May-Thurner Syndrome." Journal of Vascular and Interventional Radiology: JVIR 29 (6): 774–80. https://doi.org/10.1016/j.jvir.2018.01.771.
- Leake, Andrew E., Daniel G. Winger, Steven A. Leers, Navyash Gupta, and Ellen D. Dillavou. 2015. "Management and Outcomes of Dialysis Access-Associated Steal Syndrome." *Journal* of Vascular Surgery 61 (3): 754–61. https://doi.org/10.1016/j.jvs.2014.10.038.
- Lederle, Frank A., Tassos C. Kyriakides, Kevin T. Stroupe, Julie A. Freischlag, Frank T. Padberg, Jon S. Matsumura, Zhiping Huo, and Gary R. Johnson. 2019. "Open Versus Endovascular Repair of Abdominal Aortic Aneurysm." New England Journal of Medicine 380 (22): 2126–35. https://doi.org/10.1056/NEJMoa1715955.
- Lee, Chien-Chang, Meng-Tse Gabriel Lee, Yueh-Sheng Chen, Shih-Hao Lee, Yih-Sharng Chen, Shyr-Chyr Chen, and Shan-Chwen Chang. 2015. "Risk of Aortic Dissection and Aortic Aneurysm in Patients Taking Oral Fluoroquinolone." JAMA Internal Medicine 175 (11): 1839–47. https://doi.org/10.1001/jamainternmed.2015.5389.
- Lee, Chien-Chang, Meng-Tse Gabriel Lee, Ronan Hsieh, Lorenzo Porta, Wan-Chien Lee, Si-Huei Lee, and Shy-Shin Chang. 2018. "Oral Fluoroquinolone and the Risk of Aortic Dissection." Journal of the American College of Cardiology 72 (12): 1369–78. https://doi.org/10.1016/j.jacc.2018.06.067.
- Lee, James T., and Frederic S. Bongard. 2002. "Iliac Vessel Injuries." The Surgical Clinics of North America 82 (1): 21–48, xix. https://doi.org/10.1016/S0039-6109(03)00139-7.
- Lee, Kyongjune Benjamin, Jinny Lu, Robyn A. Macsata, Darshan Patel, Alexander Yang, John J. Ricotta, Richard L. Amdur, Anton N. Sidawy, and Bao-Ngoc Nguyen. 2019. "Inferior Mesenteric Artery Replantation Does Not Decrease the Risk of Ischemic Colitis After Open Infrarenal Abdominal Aortic Aneurysm Repair." Journal of Vascular Surgery 69 (6): 1825–30. https://doi.org/10.1016/j.jvs.2018.09.046.
- Lee, W. Anthony, Jon S. Matsumura, R. Scott Mitchell, Mark A. Farber, Roy K. Greenberg, Ali Azizzadeh, Mohammad Hassan Murad, and Ronald M. Fairman. 2011. "Endovascular Repair of Traumatic Thoracic Aortic Injury: Clinical Practice Guidelines of the Society for Vascular Surgery." Journal of Vascular Surgery 53 (1): 187–92. https://doi.org/10.1016/j. jvs.2010.08.027.
- Lee, Wai-Kit, Peter J. Mossop, Andrew F. Little, Gregory J. Fitt, Jhon I. Vrazas, Jenny K. Hoang, and Oliver F. Hennessy. 2008 Nov-Dec. "Infected (Mycotic) Aneurysms: Spectrum of Imaging Appearances and Management." *Radiographics: A Review Publication of the Radiological Society of North America, Inc* 28 (7): 1853–68. https://doi.org/10.1148/rg. 287085054.
- LeFevre, Michael L., and U. S. Preventive Services Task Force. 2014. "Screening for Abdominal Aortic Aneurysm: U.s. Preventive Services Task Force Recommendation Statement." *Annals of Internal Medicine* 161 (4): 281–90. https://doi.org/10.7326/M14-1204.
- Lejay, A., M. Ohana, J. T. Lee, Y. Georg, C. Delay, B. Lucereau, F. Thaveau, S. Gaertner, N. Chakfé, and Groupe Européen de Recherche sur les Prothèses Appliquées à la Chirurgie

Vasculaire (GEPROVAS). 2014. "Popliteal Artery Entrapment Syndrome." *The Journal of Cardiovascular Surgery* 55 (2 Suppl 1): 225–37.

- LeMaire, Scott A., Lin Zhang, Wei Luo, Pingping Ren, Alon R. Azares, Yidan Wang, Chen Zhang, Joseph S. Coselli, and Ying H. Shen. 2018. "Effect of Ciprofloxacin on Susceptibility to Aortic Dissection and Rupture in Mice." JAMA Surgery 153 (9): e181804. https://doi.org/10.1001/jamasurg.2018.1804.
- Lend, G. C., and F. G. R. Fowkes. 1992. "The Edinburgh Claudication Questionnaire: An Improved Version of the WHO/Rose Questionnaire for Use in Epidemiological Surveys." *Journal of Clinical Epidemiology* 45 (10): 1101–9. https://doi.org/10.1016/0895-4356(92) 90150-L.
- Léobon, Bertrand, Daniel Roux, Antoine Mugniot, Hervé Rousseau, Alain Cérene, Yves Glock, and Gérard Fournial. 2002. "Endovascular Treatment of Thoracic Aortic Fistulas." *The Annals of Thoracic Surgery* 74 (1): 247–49. https://doi.org/10.1016/s0003-4975(02)03466-5.
- Leopold, P. W., A. Shandall, A. M. Kupinkski, B. B. Chang, J. Kaufman, D. M. Shah, and R. P. Leather. 1989. "Role of b-Mode Venous Mapping in Infrainguinal in Situ Vein-Arterial Bypasses." *The British Journal of Surgery* 76 (3): 305–7. https://doi.org/10.1002/bjs. 1800760326.
- Lepäntalo, M., J. Apelqvist, C. Setacci, J.-B. Ricco, G. de Donato, F. Becker, H. Robert-Ebadi, et al. 2011. "Chapter v: Diabetic Foot." European Journal of Vascular and Endovascular Surgery 42 (December): S60–74. https://doi.org/10.1016/S1078-5884(11)60012-9.
- Leriche, René, and André Morel. 1948. "The Syndrome of Thrombotic Obliteration of the Aortic Bifurcation." Annals of Surgery 127 (2): 193–206. https://www.ncbi.nlm.nih.gov/ pmc/articles/PMC1513778/.
- Leversedge, Fraser J., Thomas J. Moore, Bret C. Peterson, and John G. Seiler. 2011. "Compartment Syndrome of the Upper Extremity." *The Journal of Hand Surgery* 36 (3): 544– 559; quiz 560. https://doi.org/10.1016/j.jhsa.2010.12.008.
- Li, Shuo, Brian N. King, Noel Velasco, Yogesh Kumar, and Nishant Gupta. 2017. "Cystic Adventitial Disease—Case Series and Review of Literature." Annals of Translational Medicine 5 (16): 327. https://doi.org/10.21037/atm.2017.05.04.
- Li, Zong-Zhuang, and Qiu-Yan Dai. 2012. "Pathogenesis of Abdominal Aortic Aneurysms: Role of Nicotine and Nicotinic Acetylcholine Receptors." *Mediators of Inflammation* 2012: 103120. https://doi.org/10.1155/2012/103120.
- Liang, Nathan L., Louis H. Alarcon, Geetha Jeyabalan, Efthymios D. Avgerinos, Michel S. Makaroun, and Rabih A. Chaer. 2016. "Contemporary Outcomes of Civilian Lower Extremity Arterial Trauma." Journal of Vascular Surgery 64 (3): 731–36. https://doi.org/ 10.1016/j.jvs.2016.04.052.
- Liang, Nathan L., Katherine M. Reitz, Michel S. Makaroun, Mahmoud B. Malas, and Edith Tzeng. 2018. "Comparable Perioperative Mortality Outcomes in Younger Patients Undergoing Elective Open and Endovascular Abdominal Aortic Aneurysm Repair." Journal of Vascular Surgery 67 (5): 1404–1409.e2. https://doi.org/10.1016/j.jvs.2017.08.057.
- Liddell, Robert P., and Natalie S. Evans. 2018. "May-Thurner Syndrome." Vascular Medicine (London, England) 23 (5): 493–96. https://doi.org/10.1177/1358863X18794276.

- Liebman, Howard A. 2019. "38. Hypercoagulable States." In *Rutherford's Vascular Surgery* and Endovascular Therapy, 19. Elsevier Inc.
- Lim, Sungho, Pegge M. Halandras, Carlos Bechara, Bernadette Aulivola, and Paul Crisostomo. 2019. "Contemporary Management of Acute Mesenteric Ischemia in the Endovascular Era." Vascular and Endovascular Surgery 53 (1): 42–50. https://doi.org/10.1177/ 1538574418805228.
- Lima Neto, Arlindo Cardoso, Roseli Bittar, Gabriel Scarabotolo Gattas, Edson Bor-Seng-Shu, Marcelo de Lima Oliveira, Rafael da Costa Monsanto, and Luis Felipe Bittar. 2017a. "Pathophysiology and Diagnosis of Vertebrobasilar Insufficiency: A Review of the Literature." International Archives of Otorhinolaryngology 21 (3): 302–7. https://doi.org/10. 1055/s-0036-1593448.
 - ——. 2017b. "Pathophysiology and Diagnosis of Vertebrobasilar Insufficiency: A Review of the Literature." *International Archives of Otorhinolaryngology* 21 (3): 302–7. https://doi.org/10.1055/s-0036-1593448.
- Lindsay, Thomas F. 2019a. "74. Ruptured Aortoiliac Aneurysms and Their Management." In *Rutherford's Vascular Surgery and Endovascular Therapy*, 9th ed., 21. Elsevier Inc.
- ——. 2019b. "Chapter 74 Ruptured Aortoiliac Aneurysms and Their Management." In *Rutherford's Vascular Surgery and Endovascular Therapy*, Ninth, 21. Elsevier Inc.
- Lipski, D. A., T. M. Bergamini, R. N. Garrison, and R. L. Fulton. 1996. "Intraoperative Duplex Scanning Reduces the Incidence of Residual Stenosis After Carotid Endarterectomy." *The Journal of Surgical Research* 60 (2): 317–20. https://doi.org/10.1006/jsre.1996.0050.
- Litzendorf, Maria E., and Bhagwan Satiani. 2011. "Superficial Venous Thrombosis: Disease Progression and Evolving Treatment Approaches." Vascular Health and Risk Management 7: 569–75. https://doi.org/10.2147/VHRM.S15562.
- Lo, Ruby C., and Marc L. Schermerhorn. 2016a. "Abdominal Aortic Aneurysms in Women." Journal of Vascular Surgery 63 (3): 839–44. https://doi.org/10.1016/j.jvs.2015.10.087.
- 2016b. "Abdominal Aortic Aneurysms in Women." Journal of Vascular Surgery 63
 (3): 839–44. https://doi.org/10.1016/j.jvs.2015.10.087.
- Lok, Charmaine E., Thomas S. Huber, Timmy Lee, Surendra Shenoy, Alexander S. Yevzlin, Kenneth Abreo, Michael Allon, et al. 2020. "KDOQI Clinical Practice Guideline for Vascular Access: 2019 Update." American Journal of Kidney Diseases 75 (4): S1-164. https://doi.org/10.1053/j.ajkd.2019.12.001.
- Lok, Charmaine E., Dheeraj K. Rajan, Jason Clement, Mercedeh Kiaii, Ravi Sidhu, Ken Thomson, George Buldo, et al. 2017. "Endovascular Proximal Forearm Arteriovenous Fistula for Hemodialysis Access: Results of the Prospective, Multicenter Novel Endovascular Access Trial (NEAT)." American Journal of Kidney Diseases 70 (4): 486–97. https: //doi.org/10.1053/j.ajkd.2017.03.026.
- Lombardi, Joseph V., Thomas G. Gleason, Jean M. Panneton, Benjamin W. Starnes, Michael D. Dake, Stephan Haulon, Peter J. Mossop, Mary Margaret Seale, and Qing Zhou. 2020. "STABLE II Clinical Trial on Endovascular Treatment of Acute, Complicated Type B Aortic Dissection with a Composite Device Design." Journal of Vascular Surgery 71 (4): 1077–1087.e2. https://doi.org/10.1016/j.jvs.2019.06.189.

Lombardi, Joseph V., G. Chad Hughes, Jehangir J. Appoo, Joseph E. Bavaria, Adam W. Beck,

Richard P. Cambria, Kristofer Charlton-Ouw, et al. 2020. "Society for Vascular Surgery (SVS) and Society of Thoracic Surgeons (STS) Reporting Standards for Type B Aortic Dissections." *Journal of Vascular Surgery*, 1–25. https://doi.org/10.1016/j.jvs.2019.11.013.

- Lugo, Joanelle, Adam Tanious, Paul Armstrong, Martin Back, Brad Johnson, Murray Shames, Neil Moudgill, Peter Nelson, and Karl A. Illig. 2015. "Acute Paget–Schroetter Syndrome: Does the First Rib Routinely Need to Be Removed After Thrombolysis?" Annals of Vascular Surgery 29 (6): 1073–77. https://doi.org/10.1016/j.avsg.2015.02.006.
- Lum, Ying Wei, Benjamin S. Brooke, George J. Arnaoutakis, Timothy K. Williams, and James H. Black. 2012. "Endovascular Procedures in Patients With Ehlers–Danlos Syndrome: A Review of Clinical Outcomes and Iatrogenic Complications." Annals of Vascular Surgery 26 (1): 25–33. https://doi.org/10.1016/j.avsg.2011.05.028.
- Lum, Ying Wei, Benjamin S. Brooke, Kendall Likes, Monica Modi, Holly Grunebach, Paul J. Christo, and Julie A. Freischlag. 2012. "Impact of Anterior Scalene Lidocaine Blocks on Predicting Surgical Success in Older Patients with Neurogenic Thoracic Outlet Syndrome." Journal of Vascular Surgery 55 (5): 1370–75. https://doi.org/10.1016/j.jvs.2011.11.132.
- Lurie, Fedor, Anthony Comerota, Bo Eklof, Robert L. Kistner, Nicos Labropoulos, Joann Lohr, William Marston, et al. 2012. "Multicenter Assessment of Venous Reflux by Duplex Ultrasound." Journal of Vascular Surgery 55 (2): 437–45. https://doi.org/10.1016/j.jvs. 2011.06.121.
- Lurie, Fedor, Marc Passman, Mark Meisner, Michael Dalsing, Elna Masuda, Harold Welch, Ruth L. Bush, et al. 2020. "The 2020 Update of the CEAP Classification System and Reporting Standards." Journal of Vascular Surgery: Venous and Lymphatic Disorders 8 (3): 342–52. https://doi.org/10.1016/j.jvsv.2019.12.075.
- Machan, Lindsay. 2018. "The Eyes Have It." Techniques in Vascular and Interventional Radiology 21 (1): 21–25. https://doi.org/10.1053/j.tvir.2017.12.005.
- Macsata, Robyn A, and Anton N Sidawy. 2019. "175: Hemodialysis Access: General Considerations and Strategies to Optimize Access Placement." In *Rutherford's Vascular Surgery* and Endovascular Therapy. Elsevier Inc.
- Magee, Gregory A., and Fred A. Weaver. 2022. "20. Neck and Thoracic Outlet." In Rich's Vascular Trauma, 241–51. Elsevier. https://doi.org/10.1016/B978-0-323-69766-8.00020-2.
- Mahmood, Syed S., Daniel Levy, Ramachandran S. Vasan, and Thomas J. Wang. 2014. "The Framingham Heart Study and the Epidemiology of Cardiovascular Diseases: A Historical Perspective." *Lancet* 383 (9921): 999–1008. https://doi.org/10.1016/S0140-6736(13)61752-3.
- Mahmoud, Mustafa Z, Mohammed Al-Saadi, Abdulwahab Abuderman, Khalid S Alzimami, Mohammed Alkhorayef, Babikir Almagli, and Abdelmoneim Sulieman. 2015. "'To-and-Fro' Waveform in the Diagnosis of Arterial Pseudoaneurysms." World Journal of Radiology 7 (5): 89–99. https://doi.org/10.4329/wjr.v7.i5.89.
- Maleti, O., and M. Perrin. 2011. "Reconstructive Surgery for Deep Vein Reflux in the Lower Limbs: Techniques, Results and Indications." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 41 (6): 837–48. https://doi.org/10.1016/j.ejvs.2011.02.013.
- Maleux, Geert, Lien Poorteman, Annouschka Laenen, Bertrand Saint-Lèbes, Sabrina

Houthoofd, Inge Fourneau, and Hervé Rousseau. 2017. "Incidence, Etiology, and Management of Type III Endoleak After Endovascular Aortic Repair." *Journal of Vascular Surgery* 66 (4): 1056–64. https://doi.org/10.1016/j.jvs.2017.01.056.

- Malgor, R. D., and N. Labropoulos. 2013. "Diagnosis of Venous Disease with Duplex Ultrasound." *Phlebology* 28 Suppl 1 (March): 158–61. https://doi.org/10.1177/ 0268355513476653.
- Malone, J. M., G. G. Anderson, S. G. Lalka, R. M. Hagaman, R. Henry, K. E. McIntyre, and V. M. Bernhard. 1987. "Prospective Comparison of Noninvasive Techniques for Amputation Level Selection." *American Journal of Surgery* 154 (2): 179–84. https://doi.org/10.1016/0002-9610(87)90174-7.
- Mann, Kenneth G. 2019. "36. Normal Coagulation." In Rutherford's Vascular Surgery and Endovascular Therapy, 25. Elsevier Inc.
- Mansukhani, Neel A., Kyle R. Miller, George E. Havelka, Hyde M. Russell, and Mark K. Eskandari. 2018. "Aorta-Innominate Bypass Through Ministernotomy." Journal of Vascular Surgery 68 (2): 607–10. https://doi.org/10.1016/j.jvs.2018.01.069.
- Mantese, Vito A., Carlos H. Timaran, David Chiu, Richard J. Begg, and Thomas G. Brott. 2010. "The Carotid Revascularization Endarterectomy Versus Stenting Trial (CREST) -Stenting Versus Carotid Endarterectomy for Carotid Disease." Stroke; a Journal of Cerebral Circulation 41 (10 Suppl): S31–34. https://doi.org/10.1161/STROKEAHA.110.595330.
- Markus, Hugh S., Christopher Levi, Alice King, Jeremy Madigan, and John Norris. 2019. "Antiplatelet Therapy Vs Anticoagulation Therapy in Cervical Artery Dissection." JAMA Neurology 76 (6): 657–64. https://doi.org/10.1001/jamaneurol.2019.0072.
- Marrewijk, C. J. van, G. Fransen, R. J. F. Laheij, P. L. Harris, J. Buth, and EUROSTAR Collaborators. 2004. "Is a Type II Endoleak After EVAR a Harbinger of Risk? Causes and Outcome of Open Conversion and Aneurysm Rupture During Follow-up." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 27 (2): 128–37. https://doi.org/10.1016/j.ejvs.2003.10.016.
- Marrocco-Trischitta, Massimiliano M., Benedetta Spampinato, Girolomina Mazzeo, Daniela Mazzaccaro, Valentina Milani, Moad Alaidroos, Federico Ambrogi, and Giovanni Nano. 2019. "Impact of the Bird-Beak Configuration on Postoperative Outcome After Thoracic Endovascular Aortic Repair: A Meta-Analysis." Journal of Endovascular Therapy: An Official Journal of the International Society of Endovascular Specialists 26 (6): 771–78. https://doi.org/10.1177/1526602819865906.
- Masden, Derek L., Mitchel Seruya, and James P. Higgins. 2012. "A Systematic Review of the Outcomes of Distal Upper Extremity Bypass Surgery with Arterial and Venous Conduits." *The Journal of Hand Surgery* 37 (11): 2362–67. https://doi.org/10.1016/j.jhsa.2012.07.028.
- Maskanakis, Anastasios, Nikolaos Patelis, Demetrios Moris, Diamantis I. Tsilimigras, Dimitrios Schizas, Maria Diakomi, Chris Bakoyiannis, Sotirios Georgopoulos, Chris Klonaris, and Theodoros Liakakos. 2018. "Stenting of Subclavian Artery True and False Aneurysms: A Systematic Review." Annals of Vascular Surgery 47 (February): 291–304. https://doi.org/ 10.1016/j.avsg.2017.08.013.
- Mason, Peter J., Binita Shah, Jacqueline E. Tamis-Holland, John A. Bittl, Mauricio G. Cohen, Jordan Safirstein, Douglas E. Drachman, et al. 2018. "An Update on Radial Artery Ac-

cess and Best Practices for Transradial Coronary Angiography and Intervention in Acute Coronary Syndrome: A Scientific Statement from the American Heart Association." *Circulation. Cardiovascular Interventions* 11 (9): e000035. https://doi.org/10.1161/HCV.0000000000035.

- Matsumura, Jon S., W. Anthony Lee, R. Scott Mitchell, Mark A. Farber, Mohammad Hassan Murad, Alan B. Lumsden, Roy K. Greenberg, Hazim J. Safi, Ronald M. Fairman, and Society for Vascular Surgery. 2009. "The Society for Vascular Surgery Practice Guidelines: Management of the Left Subclavian Artery with Thoracic Endovascular Aortic Repair." Journal of Vascular Surgery 50 (5): 1155–58. https://doi.org/10.1016/j.jvs.2009.08.090.
- Mawatari, K., Yoichi Muto, K. Komori, K. Kawasaki, J. Okazaki, D. H. Eguchi, S. Kuma, T. Ikeda, and K. Sugimachi. 2000. "Value of the Profundapopliteal Collateral Index for Selecting Between an in-Flow and Sequential Arterial Reconstruction in Patients with Multisegment Arterial Occlusive Disease." Journal of Cardiovascular Surgery 41 (January): 79–82.
- May, R., and J. Thurner. 1957. "The Cause of the Predominantly Sinistral Occurrence of Thrombosis of the Pelvic Veins." Angiology 8 (5): 419–27. https://doi.org/10.1177/ 000331975700800505.
- Maz, Mehrdad, Sharon A. Chung, Andy Abril, Carol A. Langford, Mark Gorelik, Gordon Guyatt, Amy M. Archer, et al. 2021. "2021 American College of Rheumatology/Vasculitis Foundation Guideline for the Management of Giant Cell Arteritis and Takayasu Arteritis." Arthritis Care & Research 73 (8): 1071–87. https://doi.org/10.1002/acr.24632.
- McClure, M. J., B. E. Kelly, N. S. Campbell, and P. H. Blair. 2000. "Duplex Doppler Ultrasonography of Lower Limb Veins: Detection of Cardiac Abnormalities." *Clinical Radiology* 55 (7): 533–36. https://doi.org/10.1053/crad.1999.0489.
- McDermott, Mary M., Jack M. Guralnik, Lu Tian, Melina R. Kibbe, Luigi Ferrucci, Lihui Zhao, Kiang Liu, Yihua Liao, Ying Gao, and Michael H. Criqui. 2016. "Incidence and Prognostic Significance of Depressive Symptoms in Peripheral Artery Disease." Journal of the American Heart Association 5 (3): e002959. https://doi.org/10.1161/JAHA.115. 002959.
- McGill, H. C., C. A. McMahan, A. W. Zieske, R. E. Tracy, G. T. Malcom, E. E. Herderick, and J. P. Strong. 2000. "Association of Coronary Heart Disease Risk Factors with Microscopic Qualities of Coronary Atherosclerosis in Youth." *Circulation* 102 (4): 374–79. https://doi. org/10.1161/01.cir.102.4.374.
- McMinn, Robert H. 2019. Last's Anatomy: Regional and Applied. Elsevier Inc.
- McQueen, M. M., and C. M. Court-Brown. 1996. "Compartment Monitoring in Tibial Fractures. The Pressure Threshold for Decompression." The Journal of Bone and Joint Surgery. British Volume 78 (1): 99–104.
- Medical, Merit. n.d. "HeRO Graft Instructions for Use." https://cloud.merit.com/catalog/ IFUs/403225012MLP.pdf.
- Meershoek, A. J. A., and G. J. de Borst. 2018. "Timing of Carotid Intervention." *The British Journal of Surgery* 105 (10): 1231–33. https://doi.org/10.1002/bjs.10950.
- Mehta, Manish, R. Clement Darling, Sean P. Roddy, Scott Fecteau, Kathleen J. Ozsvath, Paul B. Kreienberg, Philip S. K. Paty, Benjamin B. Chang, and Dhiraj M. Shah. 2005. "Factors

Associated with Abdominal Compartment Syndrome Complicating Endovascular Repair of Ruptured Abdominal Aortic Aneurysms." *Journal of Vascular Surgery* 42 (6): 1047–51. https://doi.org/10.1016/j.jvs.2005.08.033.

- Meissner, Mark H., Peter Gloviczki, Anthony J. Comerota, Michael C. Dalsing, Bo G. Eklof, David L. Gillespie, Joann M. Lohr, et al. 2012. "Early Thrombus Removal Strategies for Acute Deep Venous Thrombosis: Clinical Practice Guidelines of the Society for Vascular Surgery and the American Venous Forum." Journal of Vascular Surgery 55 (5): 1449–62. https://doi.org/10.1016/j.jvs.2011.12.081.
- Meissner, Mark H., Thomas W. Wakefield, Enrico Ascher, Joseph A. Caprini, Anthony J. Comerota, Bo Eklof, David L. Gillespie, et al. 2007. "Acute Venous Disease: Venous Thrombosis and Venous Trauma." *Journal of Vascular Surgery* 46 Suppl S (December): 25S-53S. https://doi.org/10.1016/j.jvs.2007.08.037.
- Melby, Spencer J., Suresh Vedantham, Vamsidhar R. Narra, George A. Paletta, Lynnette Khoo-Summers, Matt Driskill, and Robert W. Thompson. 2008. "Comprehensive Surgical Management of the Competitive Athlete with Effort Thrombosis of the Subclavian Vein (Paget-Schroetter Syndrome)." Journal of Vascular Surgery 47 (4): 809–820.e3. https: //doi.org/10.1016/j.jvs.2007.10.057.
- Merkus, J W S, C J A M Zeebregts, A J Hoitsma, W N J C van Asten, R A P Koene, and S H Skotnicki. 2005. "High Incidence of Arteriovenous Fistula After Biopsy of Kidney Allografts." British Journal of Surgery 80 (3): 310–12. https://doi.org/10.1002/bjs. 1800800313.
- Merrett, N. D., R. B. Wilson, P. Cosman, and A. V. Biankin. 2009. "Superior Mesenteric Artery Syndrome: Diagnosis and Treatment Strategies." Journal of Gastrointestinal Surgery: Official Journal of the Society for Surgery of the Alimentary Tract 13 (2): 287–92. https://doi.org/10.1007/s11605-008-0695-4.
- Metzger, Patrick Bastos, Fabio Henrique Rossi, Antônio Massamitsu Kambara, Nilo Mitsuru Izukawa, Mohamed Hassan Saleh, Ibraim M. F. Pinto, Jorge Eduardo Amorim, and Patricia E. Thorpe. 2016. "Criteria for Detecting Significant Chronic Iliac Venous Obstructions with Duplex Ultrasound." Journal of Vascular Surgery. Venous and Lymphatic Disorders 4 (1): 18–27. https://doi.org/10.1016/j.jvsv.2015.07.002.
- Meyer, Alexander, Cornelia Fiessler, Konstantinos Stavroulakis, Giovanni Torsello, Theodosios Bisdas, Werner Lang, and CRITISCH collaborators. 2018. "Outcomes of Dialysis Patients with Critical Limb Ischemia After Revascularization Compared with Patients with Normal Renal Function." Journal of Vascular Surgery 68 (3): 822–829.e1. https://doi.org/10.1016/ j.jvs.2017.12.048.
- Miller, Donald L., Eliseo Vañó, Gabriel Bartal, Stephen Balter, Robert Dixon, Renato Padovani, Beth Schueler, et al. 2010. "Occupational Radiation Protection in Interventional Radiology: A Joint Guideline of the Cardiovascular and Interventional Radiology Society of Europe and the Society of Interventional Radiology." Journal of Vascular and Interventional Radiology: JVIR 21 (5): 607–15. https://doi.org/10.1016/j.jvir.2010.01.007.
- Mills, Joseph L., Michael S. Conte, David G. Armstrong, Frank B. Pomposelli, Andres Schanzer, Anton N. Sidawy, and George Andros. 2014a. "The Society for Vascular Surgery Lower Extremity Threatened Limb Classification System: Risk Stratification

Based on Wound, Ischemia, and Foot Infection (WIfI)." Journal of Vascular Surgery 59 (1): 220–234.e2. https://doi.org/10.1016/j.jvs.2013.08.003.

- ——. 2014b. "The Society for Vascular Surgery Lower Extremity Threatened Limb Classification System: Risk Stratification Based on Wound, Ischemia, and Foot Infection (WIfI)." *Journal of Vascular Surgery* 59 (1): 220–234.e2. https://doi.org/10.1016/j.jvs.2013.08.003.
- Milner, Ross, Edward Y. Woo, and Jeffrey P. Carpenter. 2004. "Superior Mesenteric Artery Angioplasty and Stenting via a Retrograde Approach in a Patient with Bowel Ischemia: A Case Report." Vascular and Endovascular Surgery 38 (1): 89–91. https://doi.org/10.1177/ 153857440403800112.
- Min, Robert J., Neil M. Khilnani, and Piyush Golia. 2003. "Duplex Ultrasound Evaluation of Lower Extremity Venous Insufficiency." Journal of Vascular and Interventional Radiology: JVIR 14 (10): 1233–41. https://doi.org/10.1097/01.rvi.0000092663.72261.37.
- Mitchell, Erica L., and Patricia Furey. 2011. "Prevention of Radiation Injury from Medical Imaging." Journal of Vascular Surgery 53 (1): 22S–27S. https://doi.org/10.1016/j.jvs.2010. 05.139.
- Modarai, Bijan, Stéphan Haulon, Elizabeth Ainsbury, Dittmar Böckler, Eliseo Vano-Carruana, Joseph Dawson, Mark Farber, et al. 2022. "European Society for Vascular Surgery (ESVS) 2023 Clinical Practice Guidelines on Radiation Safety." European Journal of Vascular and Endovascular Surgery, September, S1078588422005469. https://doi.org/10.1016/j.ejvs. 2022.09.005.
- Modrall, J. Gregory, and Javid Sadjadi. 2003. "Early and Late Presentations of Radiation Arteritis." Seminars in Vascular Surgery 16 (3): 209–14. https://doi.org/10.1016/s0895-7967(03)00026-7.
- Moghadamyeghaneh, Zhobin, Michael D. Sgroi, Samuel L. Chen, Nii-Kabu Kabutey, Michael J. Stamos, and Roy M. Fujitani. 2016. "Risk Factors and Outcomes of Postoperative Ischemic Colitis in Contemporary Open and Endovascular Abdominal Aortic Aneurysm Repair." Journal of Vascular Surgery 63 (4): 866–72. https://doi.org/10.1016/j.jvs.2015.10.064.
- Mohan, Sathish, Andres Schanzer, William P. Robinson, and Francesco A. Aiello. 2016. "Endovascular Management of Radiation-Induced Subclavian and Axillary Artery Aneurysms." Journal of Vascular Surgery 64 (4): 1135–37. https://doi.org/10.1016/j.jvs.2015.08.073.
- Mohapatra, Abhisekh, Darve Robinson, Othman Malak, Michael C. Madigan, Efthimios D. Avgerinos, Rabih A. Chaer, Michael J. Singh, and Michel S. Makaroun. 2019. "Increasing Use of Open Conversion for Late Complications After Endovascular Aortic Aneurysm Repair." Journal of Vascular Surgery 69 (6): 1766–75. https://doi.org/10.1016/j.jvs.2018.09.049.
- Mohapatra, Abhisekh, Karim M. Salem, Emade Jaman, Darve Robinson, Efthymios D. Avgerinos, Michel S. Makaroun, and Mohammad H. Eslami. 2018. "Risk Factors for Perioperative Mortality Following Revascularization for Acute Aortic Occlusion." *Journal of Vascular Surgery* 68 (6): 1789–95. https://doi.org/10.1016/j.jvs.2018.04.037.
- Mohler, Emile R., Heather L. Gornik, Marie Gerhard-Herman, Sanjay Misra, Jeffrey W. Olin,
 R. Eugene Zierler, Michael J. Wolk, et al. 2012. "ACCF/ACR/AIUM/ASE/ASN/ICAVL/SCAI/SCCT/SIR,
 2012 Appropriate Use Criteria for Peripheral Vascular Ultrasound and Physiological Testing Part i: Arterial Ultrasound and Physiological Testing." Journal of the American

College of Cardiology 60 (3): 242–76. https://doi.org/10.1016/j.jacc.2012.02.009.

- Molina, J. Ernesto, David W. Hunter, and Charles A. Dietz. 2007. "Paget-Schroetter Syndrome Treated with Thrombolytics and Immediate Surgery." Journal of Vascular Surgery 45 (2): 328–34. https://doi.org/10.1016/j.jvs.2006.09.052.
- Mondry, Tammy E., Robert H. Riffenburgh, and Peter A. S. Johnstone. 2004. "Prospective Trial of Complete Decongestive Therapy for Upper Extremity Lymphedema After Breast Cancer Therapy." *Cancer Journal (Sudbury, Mass.)* 10 (1): 42-48; discussion 17-19. https: //doi.org/10.1097/00130404-200401000-00009.
- Moneta, Gregory L, Raymond W Lee, Richard A Yeager, Lloyd M Taylor Jr, and John M Porter. 1993. "Mesenteric Duplex Scanning: A Blinded Prospective Study." Journal of Vascular Surgery 17 (1): 79–86. https://doi.org/10.1067/mva.1993.41706.
- Moneta, Gregory L, Richard A Yeager, Ronald Dalman, and Ruza Antonovic. 1991. "Duplex Ultrasound Criteria for Diagnosis of Splanchnic Artery Stenosis or Occlusion" 14 (4): 10.
- Montori, V. M., S. J. Kavros, E. E. Walsh, and T. W. Rooke. 2002. "Intermittent Compression Pump for Nonhealing Wounds in Patients with Limb Ischemia. The Mayo Clinic Experience (1998-2000)." International Angiology: A Journal of the International Union of Angiology 21 (4): 360–66.
- Moore, Wesley S., Peter F. Lawrence, and Gustavo S. Oderich, eds. 2019. Vascular and Endovascular Surgery: A Comprehensive Review. Ninth edition. Philadelphia: Elsevier/Saunders.
- Morasch, Mark. 2019. "97. Vertebral Artery Dissection and Other Conditions." In Rutherford's Vascular Surgery and Endovascular Therapy, 18. Elsevier Inc.
- Morasch, Mark D. 2009a. "Technique for Subclavian to Carotid Transposition, Tips, and Tricks." Journal of Vascular Surgery 49 (1): 251–54. https://doi.org/10.1016/j.jvs.2008. 10.035.
- ———. 2009b. "Technique for Subclavian to Carotid Transposition, Tips, and Tricks." Journal of Vascular Surgery 49 (1): 251–54. https://doi.org/10.1016/j.jvs.2008.10.035.
- Mordasini, P., J. Gralla, D.-D. Do, J. Schmidli, B. Keserü, M. Arnold, U. Fischer, G. Schroth, and C. Brekenfeld. 2011. "Percutaneous and Open Retrograde Endovascular Stenting of Symptomatic High-Grade Innominate Artery Stenosis: Technique and Follow-up." AJNR: American Journal of Neuroradiology 32 (9): 1726–31. https://doi.org/10.3174/ajnr.A2598.
- Moreira, Carla C., Alexander D. Leung, Alik Farber, Denis Rybin, Gheorghe Doros, Jeffrey J. Siracuse, Jeffrey Kalish, and Mohammad H. Eslami. 2016. "Alternative Conduit for Infrageniculate Bypass in Patients with Critical Limb Ischemia." Journal of Vascular Surgery 64 (1): 131–139.e1. https://doi.org/10.1016/j.jvs.2016.01.042.
- Moreno-Sánchez, T., M. Moreno-Ramírez, F. H. Machancoses, P. Pardo-Moreno, P. F. Navarro-Vergara, and J. García-Revillo. 2020. "Efficacy of Paclitaxel Balloon for Hemodialysis Stenosis Fistulae After One Year Compared to High-Pressure Balloons: A Controlled, Multicenter, Randomized Trial." *Cardiovascular and Interventional Radiology* 43 (3): 382– 90. https://doi.org/10.1007/s00270-019-02372-w.
- Morgan, Robert, and Anna-Maria Belli. 2003. "Current Treatment Methods for Postcatheterization Pseudoaneurysms." Journal of Vascular and Interventional Radiology: JVIR 14 (6): 697–710. https://doi.org/10.1097/01.rvi.0000071089.76348.6a.

- Moriarty, John M., Dennis F. Bandyk, Daniel F. Broderick, Rebecca S. Cornelius, Karin E. Dill, Christopher J. Francois, Marie D. Gerhard-Herman, et al. 2015. "ACR Appropriateness Criteria Imaging in the Diagnosis of Thoracic Outlet Syndrome." Journal of the American College of Radiology: JACR 12 (5): 438–43. https://doi.org/10.1016/j.jacr.2015.01.016.
- Morimoto, A., A. Nakatani, K. Matsui, T. Hashimoto, M. Maeda, H. Uchida, and K. Dohi. 1995. "A Unique Case of Renovascular Hypertension Caused by Combined Renal Artery Disease." *Hypertension Research: Official Journal of the Japanese Society of Hypertension* 18 (3): 255–57. https://doi.org/10.1291/hypres.18.255.
- Moritz, Stefan, Piotr Kasprzak, Matthias Arlt, Kai Taeger, and Christoph Metz. 2007a. "Accuracy of Cerebral Monitoring in Detecting Cerebral Ischemia During Carotid Endarterectomy." Anesthesiology 107 (4): 563–69. https://doi.org/10.1097/01.anes. 0000281894.69422.ff.
 - ——. 2007b. "Accuracy of Cerebral Monitoring in Detecting Cerebral Ischemia During Carotid Endarterectomy: A Comparison of Transcranial Doppler Sonography, Near-Infrared Spectroscopy, Stump Pressure, and Somatosensory Evoked Potentials." *Anesthesiology* 107 (4): 563–69. https://doi.org/10.1097/01.anes.0000281894.69422.ff.
- Morrell, Rosalyn M., Michele Y. Halyard, Steven E. Schild, Muna S. Ali, Leonard L. Gunderson, and Barbara A. Pockaj. 2005. "Breast Cancer-Related Lymphedema." Mayo Clinic Proceedings 80 (11): 1480–84. https://doi.org/10.4065/80.11.1480.
- Morrissey, J. H. 2001. "Tissue Factor: An Enzyme Cofactor and a True Receptor." *Thrombosis* and Haemostasis 86 (1): 66–74.
- Mousa, Albeir Y., Ramez Morkous, Mike Broce, Michael Yacoub, Andrew Sticco, Ravi Viradia, Mark C. Bates, and Ali F. AbuRahma. 2017. "Validation of Subclavian Duplex Velocity Criteria to Grade Severity of Subclavian Artery Stenosis." Journal of Vascular Surgery 65 (6): 1779–85. https://doi.org/10.1016/j.jvs.2016.12.098.
- Muela Méndez, Miguel, Pilar Caridad Morata Barrado, Estrella Blanco Cañibano, Beatriz García Fresnillo, and Mercedes Guerra Requena. 2018. "Preoperative Mapping of the Aortoiliac Territory with Duplex Ultrasound in Patients with Peripheral Arterial Occlusive Disease." Journal of Vascular Surgery 68 (2): 503–9. https://doi.org/10.1016/j.jvs.2017. 11.082.
- Müller, Barbara Theresia, Lutz Reiher, Tomas Pfeiffer, Wolfram Müller, Waldemar Hort, Adina Voiculescu, Bernd Grabensee, Günther Fürst, and Wilhelm Sandmann. 2003. "Surgical Treatment of Renal Artery Dissection in 25 Patients: Indications and Results." Journal of Vascular Surgery 37 (4): 761–68. https://doi.org/10.1067/mva.2003.171.
- Müller, M., M. Voges, U. Piepgras, and K. Schimrigk. 1995. "Assessment of Cerebral Vasomotor Reactivity by Transcranial Doppler Ultrasound and Breath-Holding. A Comparison with Acetazolamide as Vasodilatory Stimulus." Stroke 26 (1): 96–100. https: //doi.org/10.1161/01.str.26.1.96.
- Munavalli, Girish S., and Robert A. Weiss. 2007. "Complications of Sclerotherapy." Seminars in Cutaneous Medicine and Surgery 26 (1): 22–28. https://doi.org/10.1016/j.sder.2006.12. 009.
- Murphy, Erin H., Blake Johns, Elliot Varney, William Buck, Arjun Jayaraj, and Seshadri Raju. 2017. "Deep Venous Thrombosis Associated with Caval Extension of Iliac Stents." *Journal*

of Vascular Surgery. Venous and Lymphatic Disorders 5 (1): 8–17. https://doi.org/10. 1016/j.jvsv.2016.09.002.

- Murphy, Timothy P., Donald E. Cutlip, Judith G. Regensteiner, Emile R. Mohler, David J. Cohen, Matthew R. Reynolds, Joseph M. Massaro, et al. 2012. "Supervised Exercise Versus Primary Stenting for Claudication Resulting from Aortoiliac Peripheral Artery Disease: Six-Month Outcomes from the Claudication: Exercise Versus Endoluminal Revascularization (CLEVER) Study." *Circulation* 125 (1): 130–39. https://doi.org/10.1161/ CIRCULATIONAHA.111.075770.
- ——, et al. 2015. "Supervised Exercise, Stent Revascularization, or Medical Therapy for Claudication Due to Aortoiliac Peripheral Artery Disease: The CLEVER Study." *Journal* of the American College of Cardiology 65 (10): 999–1009. https://doi.org/10.1016/j.jacc. 2014.12.043.
- Mustafa, Bisher O., Suman W. Rathbun, Thomas L. Whitsett, and Gary E. Raskob. 2002. "Sensitivity and Specificity of Ultrasonography in the Diagnosis of Upper Extremity Deep Vein Thrombosis: A Systematic Review." Archives of Internal Medicine 162 (4): 401–4. https://doi.org/10.1001/archinte.162.4.401.
- Mustapha, J. A., Sara M. Finton, Larry J. Diaz-Sandoval, Fadi A. Saab, and Larry E. Miller. 2016. "Percutaneous Transluminal Angioplasty in Patients with Infrapopliteal Arterial Disease: Systematic Review and Meta-Analysis." *Circulation: Cardiovascular Interventions* 9 (5): e003468. https://doi.org/10.1161/CIRCINTERVENTIONS.115.003468.
- Nachbur, B. H., R. G. Inderbitzi, and W. Bär. 1991. "Isolated Iliac Aneurysms." European Journal of Vascular Surgery 5 (4): 375–81. https://doi.org/10.1016/s0950-821x(05)80169-0.
- Nadeau, Mélissa, M. Patricia Rosas-Arellano, Kevin R. Gurr, Stewart I. Bailey, David C. Taylor, Ruby Grewal, D. Kirk Lawlor, and Chris S. Bailey. 2013. "The Reliability of Differentiating Neurogenic Claudication from Vascular Claudication Based on Symptomatic Presentation." *Canadian Journal of Surgery* 56 (6): 372–77. https://doi.org/10.1503/cjs.016512.
- Nasser, Felipe, Rafael Noronha Cavalcante, Francisco Leonardo Galastri, Jorge Eduardo de Amorim, Marcus Alexandre Politzer Telles, Fabiellen Berzoini Travassos, Bruna De Fina, and Breno Boueri Affonso. 2014. "Endovascular Stenting of Brachial Artery Occlusion in Critical Hand Ischemia." Annals of Vascular Surgery 28 (6): 1564.e1–3. https://doi.org/ 10.1016/j.avsg.2013.12.022.
- Naylor, A. R., R. L. Cuffe, P. M. Rothwell, and P. R. F. Bell. 2003. "A Systematic Review of Outcomes Following Staged and Synchronous Carotid Endarterectomy and Coronary Artery Bypass." *European Journal of Vascular and Endovascular Surgery* 25 (5): 380–89. https://doi.org/10.1053/ejvs.2002.1895.
- Naylor, A. Ross. 2015. "Why Is the Management of Asymptomatic Carotid Disease so Controversial?" *The Surgeon* 13 (1): 34–43. https://doi.org/10.1016/j.surge.2014.08.004.
- Naylor, A. R., J.-B. Ricco, G. J. de Borst, S. Debus, J. de Haro, A. Halliday, G. Hamilton, et al. 2018. "Editor's Choice - Management of Atherosclerotic Carotid and Vertebral Artery Disease: 2017 Clinical Practice Guidelines of the European Society for Vascular Surgery (ESVS)." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 55 (1): 3–81. https://doi.org/10.1016/j.ejvs.

2017.06.021.

- Naylor, Ross, Barbara Rantner, Stefano Ancetti, Gert J. de Borst, Marco De Carlo, Alison Halliday, Stavros K. Kakkos, et al. 2023. "Editor's Choice - European Society for Vascular Surgery (ESVS) 2023 Clinical Practice Guidelines on the Management of Atherosclerotic Carotid and Vertebral Artery Disease." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 65 (1): 7–111. https://doi.org/10.1016/j.ejvs.2022.04.011.
- Nazzal, M., M. El-Fedaly, V. Kazan, W. Qu, A. W. Renno, M. Al-Natour, and J. Abbas. 2015. "Incidence and Clinical Significance of Iliac Vein Compression." Vascular 23 (4): 337–43. https://doi.org/10.1177/1708538114551194.
- Neal, Joseph M., Michael J. Barrington, Michael R. Fettiplace, Marina Gitman, Stavros G. Memtsoudis, Eva E. Mörwald, Daniel S. Rubin, and Guy Weinberg. 2018. "The Third American Society of Regional Anesthesia and Pain Medicine Practice Advisory on Local Anesthetic Systemic Toxicity: Executive Summary 2017." Regional Anesthesia and Pain Medicine 43 (2): 113–23. https://doi.org/10.1097/AAP.000000000000720.
- Needleman, Laurence, John J. Cronan, Michael P. Lilly, Geno J. Merli, Srikar Adhikari, Barbara S. Hertzberg, M. Robert DeJong, Michael B. Streiff, and Mark H. Meissner. 2018. "Ultrasound for Lower Extremity Deep Venous Thrombosis: Multidisciplinary Recommendations from the Society of Radiologists in Ultrasound Consensus Conference." *Circulation* 137 (14): 1505–15. https://doi.org/10.1161/CIRCULATIONAHA.117.030687.
- Nelson, Howard A., Maryellyn Gilfeather, John M. Holman, Edward W. Nelson, and Hyo-Chun Yoon. 1999. "Gadolinium-Enhanced Breathhold Three-dimensional Time-of-Flight Renal MR Angiography in the Evaluation of Potential Renal Donors." Journal of Vascular and Interventional Radiology 10 (2): 175–81. https://doi.org/10.1016/S1051-0443(99)70462-8.
- Newton, Emily R., Adam W. Akerman, Paula D. Strassle, and Melina R. Kibbe. 2021. "Association of Fluoroquinolone Use With Short-term Risk of Development of Aortic Aneurysm." JAMA Surgery 156 (3): 264. https://doi.org/10.1001/jamasurg.2020.6165.
- Nguyen, T. H., T. D. Bui, I. L. Gordon, and S. E. Wilson. 2007. "Functional Patency of Autogenous AV Fistulas for Hemodialysis." The Journal of Vascular Access 8 (4): 275–80.
- Nicolaï, Saskia P. A., Wolfgang Viechtbauer, Lotte M. Kruidenier, Math J. J. M. Candel, Martin H. Prins, and Joep A. W. Teijink. 2009. "Reliability of Treadmill Testing in Peripheral Arterial Disease: A Meta-Regression Analysis." *Journal of Vascular Surgery* 50 (2): 322–29. https://doi.org/10.1016/j.jvs.2009.01.042.
- Nienaber, Christoph A., and Rachel E. Clough. 2015. "Management of Acute Aortic Dissection." Lancet (London, England) 385 (9970): 800–811. https://doi.org/10.1016/S0140-6736(14)61005-9.
- Nienaber, Christoph A., Hervé Rousseau, Holger Eggebrecht, Stephan Kische, Rossella Fattori, Tim C. Rehders, Günther Kundt, et al. 2009. "Randomized Comparison of Strategies for Type b Aortic Dissection: The INvestigation of STEnt Grafts in Aortic Dissection (INSTEAD) Trial." *Circulation* 120 (25): 2519–28. https://doi.org/10.1161/CIRCULATIONAHA.109.886408.
- Nierlich, Patrick, Florian K. Enzmann, Werner Dabernig, Julio Ellacuriaga San Martin, Fatema Akhavan, Klaus Linni, and Thomas Hölzenbein. 2019. "Small Saphenous Vein

and Arm Vein as Bypass Grafts for Upper Extremity Ischemia." Annals of Vascular Surgery 60 (October): 264–69. https://doi.org/10.1016/j.avsg.2019.02.015.

- Norgren, L., W. R. Hiatt, J. A. Dormandy, M. R. Nehler, K. A. Harris, and F. G. R. Fowkes. 2007. "Inter-Society Consensus for the Management of Peripheral Arterial Disease (TASC II)." Journal of Vascular Surgery 45 (1): S5–67. https://doi.org/10.1016/j.jvs.2006.12.037.
- North American Symptomatic Carotid Endarterectomy Trial Collaborators. 1991. "Beneficial Effect of Carotid Endarterectomy in Symptomatic Patients with High-Grade Carotid Stenosis." New England Journal of Medicine 325 (7): 445–53. https://doi.org/10.1056/ NEJM199108153250701.
- Notten, Pascale, Hugo Ten Cate, and Arina J. Ten Cate-Hoek. 2021. "Postinterventional Antithrombotic Management After Venous Stenting of the Iliofemoral Tract in Acute and Chronic Thrombosis: A Systematic Review." Journal of Thrombosis and Haemostasis: JTH 19 (3): 753–96. https://doi.org/10.1111/jth.15197.
- Nouh, Amre, Jessica Remke, and Sean Ruland. 2014. "Ischemic Posterior Circulation Stroke: A Review of Anatomy, Clinical Presentations, Diagnosis, and Current Management." Frontiers in Neurology 5 (April): 30. https://doi.org/10.3389/fneur.2014.00030.
- Nouvini, Rosa, and Sanjaykumar Hapani. 2013. "Images in Clinical Medicine. Enlarged Spleen with a Heterogeneous Pattern." The New England Journal of Medicine 368 (26): e37. https://doi.org/10.1056/NEJMicm1210836.
- O'Donnell, Thomas F., Marc A. Passman, William A. Marston, William J. Ennis, Michael Dalsing, Robert L. Kistner, Fedor Lurie, et al. 2014. "Management of Venous Leg Ulcers: Clinical Practice Guidelines of the Society for Vascular Surgery ® and the American Venous Forum." Journal of Vascular Surgery 60 (2 Suppl): 3S–59S. https://doi.org/10.1016/j.jvs. 2014.04.049.
- O'Donnell, Thomas F., John C. Rasmussen, and Eva M. Sevick-Muraca. 2017. "New Diagnostic Modalities in the Evaluation of Lymphedema." *Journal of Vascular Surgery. Venous* and Lymphatic Disorders 5 (2): 261–73. https://doi.org/10.1016/j.jvsv.2016.10.083.
- Obara, Andrzej, Miroslaw A. Dziekiewicz, Marek Maruszynski, Adam Witkowski, Maciej Dąbrowski, and Zbigniew Chmielak. 2014. "Lymphatic Complications After Vascular Interventions." Wideochirurgia I Inne Techniki Maloinwazyjne = Videosurgery and Other Miniinvasive Techniques 9 (3): 420–26. https://doi.org/10.5114/wiitm.2014.43021.
- Obara, Hideaki, Kentaro Matsubara, and Yuko Kitagawa. 2018. "Acute Limb Ischemia." Annals of Vascular Diseases 11 (4): 443–48. https://doi.org/10.3400/avd.ra.18-00074.
- Oderich, Gustavo S., Thomas C. Bower, Jan Hofer, Manju Kalra, Audra A. Duncan, John W. Wilson, Stephan Cha, and Peter Gloviczki. 2011. "In Situ Rifampin-Soaked Grafts with Omental Coverage and Antibiotic Suppression Are Durable with Low Reinfection Rates in Patients with Aortic Graft Enteric Erosion or Fistula." Journal of Vascular Surgery 53 (1): 99–106, 107.e1-7; discussion 106-107. https://doi.org/10.1016/j.jvs.2010.08.018.
- Oderich, Gustavo S., Mateus P. Correa, and Bernardo C. Mendes. 2014. "Technical Aspects of Repair of Juxtarenal Abdominal Aortic Aneurysms Using the Zenith Fenestrated Endovascular Stent Graft." *Journal of Vascular Surgery* 59 (5): 1456–61. https://doi.org/10. 1016/j.jvs.2013.10.060.
- Oderich, Gustavo S., Luke S. Erdoes, Christopher Lesar, Bernardo C. Mendes, Peter Gloviczki,

Stephen Cha, Audra A. Duncan, and Thomas C. Bower. 2013. "Comparison of Covered Stents Versus Bare Metal Stents for Treatment of Chronic Atherosclerotic Mesenteric Arterial Disease." *Journal of Vascular Surgery* 58 (5): 1316–23. https://doi.org/10.1016/j.jvs.2013.05.013.

- Oderich, Gustavo S., Peter Gloviczki, and Thomas C. Bower. 2010. "Open Surgical Treatment for Chronic Mesenteric Ischemia in the Endovascular Era: When It Is Necessary and What Is the Preferred Technique?" Seminars in Vascular Surgery 23 (1): 36–46. https://doi. org/10.1053/j.semvascurg.2009.12.005.
- Oderich, Gustavo S., Roy K. Greenberg, Mark Farber, Sean Lyden, Luis Sanchez, Ron Fairman, Feiyi Jia, and Priya Bharadwaj. 2014. "Results of the United States Multicenter Prospective Study Evaluating the Zenith Fenestrated Endovascular Graft for Treatment of Juxtarenal Abdominal Aortic Aneurysms." Journal of Vascular Surgery 60 (6): 1420– 1428.e5. https://doi.org/10.1016/j.jvs.2014.08.061.
- Oderich, Gustavo S., Rodrigo Macedo, David H. Stone, Edward Y. Woo, Jean M. Panneton, Timothy Resch, Nuno V. Dias, et al. 2018. "Multicenter Study of Retrograde Open Mesenteric Artery Stenting Through Laparotomy for Treatment of Acute and Chronic Mesenteric Ischemia." Journal of Vascular Surgery 68 (2): 470–480.e1. https://doi.org/10. 1016/j.jvs.2017.11.086.
- Oderich, Gustavo S, and Mauricio Ribeiro. 2019a. "132. Chronic Mesenteric Arterial Disease : Clinical Evaluation, Open Surgical and Endovascular Treatment." In *Rutherford Vascular Surgery and Endovascular Therapy*, Ninth. Elsevier Inc. https://doi.org/10.1016/B978-0-323-42791-3.00132-8.
 - —. 2019b. "132. Chronic Mesenteric Arterial Disease : Clinical Evaluation , Open Surgical and Endovascular Treatment." In *Rutherford Vascular Surgery and Endovascular Therapy*, 9th ed. Elsevier Inc. https://doi.org/10.1016/B978-0-323-42791-3.00132-8.
- Oglesby, Beverly, and Kathyrn Sorrell. 2006. "Acute and Isolated Dissection of the Superior Mesenteric Artery: Presentation and Follow-up with Duplex Ultrasound." Journal for Vascular Ultrasound 30 (3): 149–53. https://doi.org/10.1177/154431670603000306.
- Olin, J. W. 2000. "Thromboangiitis Obliterans (Buerger's Disease)." The New England Journal of Medicine 343 (12): 864–69. https://doi.org/10.1056/NEJM200009213431207.
- Olin, Jeffrey W. 2018. "Thromboangiitis Obliterans: 110 Years Old and Little Progress Made." Journal of the American Heart Association: Cardiovascular and Cerebrovascular Disease 7 (23): e011214. https://doi.org/10.1161/JAHA.118.011214.
- Olin, Jeffrey W., James Froehlich, Xiaokui Gu, J. Michael Bacharach, Kim Eagle, Bruce H. Gray, Michael R. Jaff, et al. 2012. "The United States Registry for Fibromuscular Dysplasia: Results in the First 447 Patients." *Circulation* 125 (25): 3182–90. https://doi. org/10.1161/CIRCULATIONAHA.112.091223.
- Olin, Jeffrey W., Heather L. Gornik, J. Michael Bacharach, Jose Biller, Lawrence J. Fine, Bruce H. Gray, William A. Gray, et al. 2014. "Fibromuscular Dysplasia: State of the Science and Critical Unanswered Questions: A Scientific Statement From the American Heart Association." *Circulation* 129 (9): 1048–78. https://doi.org/10.1161/01.cir.0000442577. 96802.8c.
- Olin, Jeffrey W., and Brett A. Sealove. 2011. "Diagnosis, Management, and Future Devel-

opments of Fibromuscular Dysplasia." Journal of Vascular Surgery 53 (3): 826–836.e1. https://doi.org/10.1016/j.jvs.2010.10.066.

- Oliver-Williams, Clare, Michael J. Sweeting, Jo Jacomelli, Lisa Summers, Anne Stevenson, Tim Lees, and Jonothan J. Earnshaw. 2019. "Safety of Men with Small and Medium Abdominal Aortic Aneurysms Under Surveillance in the NAAASP." *Circulation* 139 (11): 1371–80. https://doi.org/10.1161/CIRCULATIONAHA.118.036966.
- Ong, Kim-Thanh, Jérôme Perdu, Julie De Backer, Erwan Bozec, Patrick Collignon, Joseph Emmerich, Anne-Laure Fauret, et al. 2010. "Effect of Celiprolol on Prevention of Cardiovascular Events in Vascular Ehlers-Danlos Syndrome: A Prospective Randomised, Open, Blinded-Endpoints Trial." The Lancet 376 (9751): 1476–84. https://doi.org/10.1016/ S0140-6736(10)60960-9.
- Orion, Kristine Clodfelter, Robert J. Beaulieu, and James H. Black. 2016. "Aortocaval Fistula: Is Endovascular Repair the Preferred Solution?" Annals of Vascular Surgery 31 (February): 221–28. https://doi.org/10.1016/j.avsg.2015.09.006.
- Osgood, Michael J., Josh M. Heck, Eric J. Rellinger, Stacey L. Doran, C. Louis Garrard, Raul J. Guzman, Thomas C. Naslund, and Jeffery B. Dattilo. 2014. "Natural History of Grade i-II Blunt Traumatic Aortic Injury." *Journal of Vascular Surgery* 59 (2): 334–41. https://doi.org/10.1016/j.jvs.2013.09.007.
- Ouriel, Kenneth, Frank J. Veith, and Arthur A. Sasahara. 1998. "A Comparison of Recombinant Urokinase with Vascular Surgery as Initial Treatment for Acute Arterial Occlusion of the Legs." New England Journal of Medicine 338 (16): 1105–11. https://doi.org/10.1056/ NEJM199804163381603.
- Ouriel, K., A. E. McDonnell, C. E. Metz, and C. K. Zarins. 1982. "Critical Evaluation of Stress Testing in the Diagnosis of Peripheral Vascular Disease." Surgery 91 (6): 686–93.
- Owens, Christopher D. 2019. "4: Atherosclerosis." In Rutherford's Vascular Surgery and Endovascular Therapy, 14. Elsevier Inc.
- Ozbek, S. S., A. Memiş, R. Killi, E. Karaca, C. Kabasakal, and S. Mir. 1995. "Image-Directed and Color Doppler Ultrasonography in the Diagnosis of Postbiopsy Arteriovenous Fistulas of Native Kidneys." *Journal of Clinical Ultrasound: JCU* 23 (4): 239–42. https: //doi.org/10.1002/jcu.1870230406.
- Padberg, Frank T., Keith D. Calligaro, and Anton N. Sidawy. 2008. "Complications of Arteriovenous Hemodialysis Access: Recognition and Management." Journal of Vascular Surgery 48 (5): S55–80. https://doi.org/10.1016/j.jvs.2008.08.067.
- Palchik, Eugene, Andrew M. Bakken, Heather Y. Wolford, David L. Waldman, and Mark G. Davies. 2008. "Evolving Strategies in Treatment of Isolated Symptomatic Innominate Artery Disease." Vascular and Endovascular Surgery 42 (5): 440–45. https://doi.org/10. 1177/1538574408315994.
- Palma, E. C., and R. Esperon. 1960. "Vein Transplants and Grafts in the Surgical Treatment of the Postphlebitic Syndrome." The Journal of Cardiovascular Surgery 1 (July): 94–107.
- Papalambros, E. L., Y. P. Panayiotopoulos, E. Bastounis, G. Zavos, and P. Balas. 1989. "Prophylactic Fasciotomy of the Legs Following Acute Arterial Occlusion Procedures." *International Angiology: A Journal of the International Union of Angiology* 8 (3): 120–24.
- Papazoglou, Konstantinos O., George S. Sfyroeras, Neofytos Zambas, Konstantinos Konstan-

tinidis, Stavros K. Kakkos, and Maria Mitka. 2012. "Outcomes of Endovascular Aneurysm Repair with Selective Internal Iliac Artery Coverage Without Coil Embolization." *Journal of Vascular Surgery* 56 (2): 298–303. https://doi.org/10.1016/j.jvs.2011.08.063.

- Paravastu, S. C. V., J. Ghosh, D. Murray, F. G. Farquharson, F. Serracino-Inglott, and M. G. Walker. 2009. "A Systematic Review of Open Versus Endovascular Repair of Inflammatory Abdominal Aortic Aneurysms." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 38 (3): 291–97. https://doi.org/10.1016/j.ejvs.2009.05.005.
- Parekh, Vishal B., Vandana D. Niyyar, and Tushar J. Vachharajani. 2016. "Lower Extremity Permanent Dialysis Vascular Access." *Clinical Journal of the American Society of Nephrology* 11 (9): 1693–1702. https://doi.org/10.2215/CJN.01780216.
- Parish, J. M., R. F. Marschke, D. E. Dines, and R. E. Lee. 1981. "Etiologic Considerations in Superior Vena Cava Syndrome." *Mayo Clinic Proceedings* 56 (7): 407–13.
- Parkinson, Fran, Stuart Ferguson, Peter Lewis, Ian M. Williams, and Christopher P. Twine. 2015. "Rupture Rates of Untreated Large Abdominal Aortic Aneurysms in Patients Unfit for Elective Repair." *Journal of Vascular Surgery* 61 (6): 1606–12. https://doi.org/10. 1016/j.jvs.2014.10.023.
- Pasklinsky, Garri, Robert J. Meisner, Nicos Labropoulos, Luis Leon, Antonios P. Gasparis, David Landau, Apostolos K. Tassiopoulos, and Peter J. Pappas. 2011. "Management of True Aneurysms of Hemodialysis Access Fistulas." Journal of Vascular Surgery 53 (5): 1291–97. https://doi.org/10.1016/j.jvs.2010.11.100.
- Pasternak, Björn, Malin Inghammar, and Henrik Svanström. 2018. "Fluoroquinolone Use and Risk of Aortic Aneurysm and Dissection: Nationwide Cohort Study." BMJ (Clinical Research Ed.) 360 (March): k678. https://doi.org/10.1136/bmj.k678.
- Patel, Mitul S., Tiffany Street, Mark G. Davies, Eric K. Peden, and Joseph J. Naoum. 2015. "Evaluating and Treating Venous Outflow Stenoses Is Necessary for the Successful Open Surgical Treatment of Arteriovenous Fistula Aneurysms." *Journal of Vascular Surgery* 61 (2): 444–48. https://doi.org/10.1016/j.jvs.2014.07.033.
- Patel, Rajesh, Michael J Sweeting, Janet T Powell, and Roger M Greenhalgh. 2016. "Endovascular Versus Open Repair of Abdominal Aortic Aneurysm in 15-Years' Follow-up of the UK Endovascular Aneurysm Repair Trial 1 (EVAR Trial 1): A Randomised Controlled Trial." *The Lancet* 388 (10058): 2366–74. https://doi.org/10.1016/S0140-6736(16)31135-7.
- Pauletto, P., M. Puato, E. Faggin, N. Santipolo, V. Pagliara, M. Zoleo, G. P. Deriu, et al. 2000. "Specific Cellular Features of Atheroma Associated with Development of Neointima After Carotid Endarterectomy: The Carotid Atherosclerosis and Restenosis Study." *Circulation* 102 (7): 771–78. https://doi.org/10.1161/01.cir.102.7.771.
- Paulson, William D., Sunanda J. Ram, and Gazi B. Zibari. 2002. "Vascular Access: Anatomy, Examination, Management." Seminars in Nephrology 22 (3): 183–94. https://doi.org/10. 1053/snep.2002.31702.
- Pellerito, John S., and Joseph F. Polak. 2019. Introduction to Vascular Ultrasonography. 7th ed. Elsevier.
- Pengo, Vittorio, Gentian Denas, Giacomo Zoppellaro, Seena Padayattil Jose, Ariela Hoxha, Amelia Ruffatti, Laura Andreoli, et al. 2018. "Rivaroxaban Vs Warfarin in High-Risk
Patients with Antiphospholipid Syndrome." *Blood* 132 (13): 1365–71. https://doi.org/10. 1182/blood-2018-04-848333.

- Petrik, P. V., H. A. Gelabert, W. S. Moore, W. Quinones-Baldrich, and M. M. Law. 1995. "Cigarette Smoking Accelerates Carotid Artery Intimal Hyperplasia in a Dose-Dependent Manner." Stroke 26 (8): 1409–14. https://doi.org/10.1161/01.str.26.8.1409.
- Pfarr, K. M., A. Y. Debrah, S. Specht, and A. Hoerauf. 2009. "Filariasis and Lymphoedema." Parasite Immunology 31 (11): 664–72. https://doi.org/10.1111/j.1365-3024.2009.01133.x.
- Pham, Martin H., Rudy J. Rahme, Omar Arnaout, Michael C. Hurley, Richard A. Bernstein, H. Hunt Batjer, and Bernard R. Bendok. 2011. "Endovascular Stenting of Extracranial Carotid and Vertebral Artery Dissections: A Systematic Review of the Literature." Neurosurgery 68 (4): 856–66. https://doi.org/10.1227/NEU.0b013e318209ce03.
- Piffaretti, Gabriele, Matteo Tozzi, Chiara Lomazzi, Nicola Rivolta, Francesca Riva, Roberto Caronno, and Patrizio Castelli. 2007. "Splenic Artery Aneurysms: Postembolization Syndrome and Surgical Complications." *American Journal of Surgery* 193 (2): 166–70. https://doi.org/10.1016/j.amjsurg.2006.09.007.
- Pike, Steven L., Alik Farber, Nkiruka Arinze, Scott Levin, Thomas W. Cheng, Douglas W. Jones, Tze-Woei Tan, Mahmoud Malas, Denis Rybin, and Jeffrey J. Siracuse. 2019. "Patients with Lower Extremity Dialysis Access Have Poor Primary Patency and Survival." Journal of Vascular Surgery 70 (6): 1913–18. https://doi.org/10.1016/j.jvs.2019.03.037.
- Piper, Margaret A., Corinne V. Evans, Brittany U. Burda, Karen L. Margolis, Elizabeth O'Connor, Ning Smith, Elizabeth Webber, Leslie A. Perdue, Keshia D. Bigler, and Evelyn P. Whitlock. 2014. Screening for High Blood Pressure in Adults: A Systematic Evidence Review for the u.s. Preventive Services Task Force. U.s. Preventive Services Task Force Evidence Syntheses, Formerly Systematic Evidence Reviews. Agency for Healthcare Research; Quality (US). http://www.ncbi.nlm.nih.gov/books/NBK269495/.
- Plessers, Maarten, Isabelle Van Herzeele, Dimitri Hemelsoet, Nikil Patel, Emma M. L. Chung, Guy Vingerhoets, and Frank Vermassen. 2016. "Transcervical Carotid Stenting with Dynamic Flow Reversal Demonstrates Embolization Rates Comparable to Carotid Endarterectomy." Journal of Endovascular Therapy: An Official Journal of the International Society of Endovascular Specialists 23 (2): 249–54. https://doi.org/10.1177/1526602815626561.
- Popplewell, Matthew A, and Andrew W Bradbury. 2019. "110. Infrainguinal Disease : Endovascular Therapy." In Rutherford's Vascular Surgery and Endovascular Therapy. Elsevier Inc.
- Pourdeyhimi, Neda, and Zackery Bullard. 2014. "Warfarin-Induced Skin Necrosis." *Hospital Pharmacy* 49 (11): 1044–48. https://doi.org/10.1310/hpj4911-1044.
- Pourier, V. E. C., J. C. Welleweerd, L. J. Kappelle, G. J. E. Rinkel, Y. M. Ruigrok, H. B. van der Worp, T. H. Lo, M. L. Bots, F. L. Moll, and G. J. de Borst. 2018. "Experience of a Single Center in the Conservative Approach of 20 Consecutive Cases of Asymptomatic Extracranial Carotid Artery Aneurysms." *European Journal of Neurology* 25 (10): 1285–89. https://doi.org/10.1111/ene.13720.
- Power, Adam H., Thomas C. Bower, Jan Kasperbauer, Michael J. Link, Gustavo Oderich, Harry Cloft, William F. Young, and Peter Gloviczki. 2012. "Impact of Preoperative Embolization on Outcomes of Carotid Body Tumor Resections." Journal of Vascular Surgery

56 (4): 979–89. https://doi.org/10.1016/j.jvs.2012.03.037.

- Powers, William J., Alejandro A. Rabinstein, Teri Ackerson, Opeolu M. Adeoye, Nicholas C. Bambakidis, Kyra Becker, José Biller, et al. 2018. "2018 Guidelines for the Early Management of Patients With Acute Ischemic Stroke: A Guideline for Healthcare Professionals From the American Heart Association/American Stroke Association." Stroke 49 (3). https://doi.org/10.1161/STR.00000000000158.
- Prinssen, Monique, Philippe W M Cuypers, and Erik Buskens. 2004. "A Randomized Trial Comparing Conventional and Endovascular Repair of Abdominal Aortic Aneurysms." The New England Journal of Medicine, 13.
- Puges, Mathilde, Xavier Bérard, Jean-Baptiste Ruiz, Frederic Debordeaux, Arnaud Desclaux, Laurent Stecken, Sabine Pereyre, et al. 2019. "Retrospective Study Comparing WBC Scan and 18F-FDG PET/CT in Patients with Suspected Prosthetic Vascular Graft Infection." *European Journal of Vascular and Endovascular Surgery* 57 (6): 876–84. https://doi.org/ 10.1016/j.ejvs.2018.12.032.
- Quan, Reagan W., David L. Gillespie, Rory P. Stuart, Audrey S. Chang, David R. Whittaker, and Charles J. Fox. 2008. "The Effect of Vein Repair on the Risk of Venous Thromboembolic Events: A Review of More Than 100 Traumatic Military Venous Injuries." Journal of Vascular Surgery 47 (3): 571–77. https://doi.org/10.1016/j.jvs.2007.10.056.
- Quevedo, Henry C., Salman A. Arain, and Nidal Abi Rafeh. 2014. "Systematic Review of Endovascular Therapy for Nutcracker Syndrome and Case Presentation." *Cardiovascular Revascularization Medicine: Including Molecular Interventions* 15 (5): 305–7. https://doi. org/10.1016/j.carrev.2014.04.008.
- Quinones, Adriana, and Muhamed Saric. 2013. "The Cholesterol Emboli Syndrome in Atherosclerosis." Current Atherosclerosis Reports 15 (4): 315. https://doi.org/10.1007/ s11883-013-0315-y.
- Quinones-Baldrich, William, Ali Alktaifi, Fritz Eilber, and Frederick Eilber. 2012. "Inferior Vena Cava Resection and Reconstruction for Retroperitoneal Tumor Excision." Journal of Vascular Surgery 55 (5): 1386–1393; discussion 1393. https://doi.org/10.1016/j.jvs.2011. 11.054.
- Quintana, Alfonso L., Esther Martínez Aguilar, Alvaro Fernandez Heredero, Vicente Riambau, Laura Paul, and Francisco Acín. 2006. "Aortobronchial Fistula After Aortic Coartactation." *The Journal of Thoracic and Cardiovascular Surgery* 131 (1): 240–43. https://doi.org/10. 1016/j.jtcvs.2005.10.005.
- Quiroga, Elina, Benjamin W. Starnes, Nam T. Tran, and Niten Singh. 2019. "Implementation and Results of a Practical Grading System for Blunt Thoracic Aortic Injury." *Journal of Vascular Surgery* 70 (4): 1082–88. https://doi.org/10.1016/j.jvs.2019.01.048.
- Rabkin, Simon W. 2016. "The Effect of Nicotine and Tobacco on Aortic Matrix Metalloproteinases in the Production of Aortic Aneurysm." *Current Vascular Pharmacology* 14 (6): 514–22. https://doi.org/10.2174/1570161114666160625091205.

//doi.org/10.1056/NEJM200102083440603.

- Rajaraman, Preetha, Michele M. Doody, Chu Ling Yu, Dale L. Preston, Jeremy S. Miller, Alice J. Sigurdson, D. Michal Freedman, et al. 2016. "Cancer Risks in u.s. Radiologic Technologists Working with Fluoroscopically Guided Interventional Procedures, 1994-2008." *AJR. American Journal of Roentgenology* 206 (5): 1101–1108; quiz 1109. https://doi.org/ 10.2214/AJR.15.15265.
- Rajendran, Saissan, and James May. 2017. "Late Rupture of Abdominal Aortic Aneurysm After Endovascular Repair." Journal of Vascular Surgery 65 (1): 52–57. https://doi.org/ 10.1016/j.jvs.2016.05.090.
- Raju, Seshadri, Rikki Darcey, and Peter Neglén. 2010. "Unexpected Major Role for Venous Stenting in Deep Reflux Disease." Journal of Vascular Surgery 51 (2): 401–408; discussion 408. https://doi.org/10.1016/j.jvs.2009.08.032.
- Raju, Seshadri, Sam Owen, and Peter Neglen. 2002. "The Clinical Impact of Iliac Venous Stents in the Management of Chronic Venous Insufficiency." *Journal of Vascular Surgery* 35 (1): 8–15. https://doi.org/10.1067/mva.2002.121054.
- Ramirez, Claudio Alberto, Guillaume Febrer, Julien Gaudric, Salam Abou-Taam, Kenza Beloucif, Laurent Chiche, and Fabien Koskas. 2012. "Open Repair of Vertebral Artery: A 7-Year Single-Center Report." Annals of Vascular Surgery 26 (1): 79–85. https://doi.org/ 10.1016/j.avsg.2011.09.001.
- Ramirez, Joel L., Laura M. Drudi, and S. Marlene Grenon. 2018. "Review of Biologic and Behavioral Risk Factors Linking Depression and Peripheral Artery Disease." Vascular Medicine (London, England) 23 (5): 478–88. https://doi.org/10.1177/1358863X18773161.
- Rangel-Castilla, Leonardo, M. Yashar S. Kalani, Katherine Cronk, Joseph M. Zabramski, Jonathan J. Russin, and Robert F. Spetzler. 2015. "Vertebral Artery Transposition for Revascularization of the Posterior Circulation: A Critical Assessment of Temporary and Permanent Complications and Outcomes." Journal of Neurosurgery 122 (3): 671–77. https: //doi.org/10.3171/2014.9.JNS14194.
- Rantner, Barbara, Barbara Kollerits, Gary S Roubin, Peter A Ringleb, Olaf Jansen, George Howard, Jeroen Hendrikse, et al. 2017. "Early Endarterectomy Carries a Lower Procedural Risk Than Early Stenting in Patients With Symptomatic Stenosis of the Internal Carotid Artery." Stroke 48 (June): 1580–87. https://doi.org/10.1161/STROKEAHA.116.016233.
- Raskob, Gary E., Nick van Es, Peter Verhamme, Marc Carrier, Marcello Di Nisio, David Garcia, Michael A. Grosso, et al. 2018. "Edoxaban for the Treatment of Cancer-Associated Venous Thromboembolism." *The New England Journal of Medicine* 378 (7): 615–24. https: //doi.org/10.1056/NEJMoa1711948.
- Rasmussen, T. E., and J. W. Hallett. 1997. "Inflammatory Aortic Aneurysms. A Clinical Review with New Perspectives in Pathogenesis." Annals of Surgery 225 (2): 155–64. https: //doi.org/10.1097/00000658-199702000-00003.
- Rasmussen, Todd E, and Nigel R. M Tai. 2022. *Rich's Vascular Trauma*. https://www.clinicalkey.com.au/dura/browse/bookChapter/3-s2.0-C20180031916.
- Ratchford, Elizabeth V., and James H. Black. 2011. "Approach to Smoking Cessation in the Patient with Vascular Disease." *Current Treatment Options in Cardiovascular Medicine* 13 (2): 91–102. https://doi.org/10.1007/s11936-011-0109-x.

- Redfern, G., R. N. Rodseth, and B. M. Biccard. 2011. "Outcomes in Vascular Surgical Patients with Isolated Postoperative Troponin Leak: A Meta-Analysis: Troponin Leak in Vascular Surgical Patients." Anaesthesia 66 (7): 604–10. https://doi.org/10.1111/j.1365-2044.2011.06763.x.
- Reed, Amy B. 2019. "24. Radiation Safety." In Rutherford's Vascular Surgery and Endovascular Therapy, 265–273.e1. Elsevier Inc. https://doi.org/10.1016/B978-0-323-40126-5.00005-7.
- Reed, Nanette R., Manju Kalra, Thomas C. Bower, Terri J. Vrtiska, Joseph J. Ricotta, and Peter Gloviczki. 2009. "Left Renal Vein Transposition for Nutcracker Syndrome." *Journal* of Vascular Surgery 49 (2): 386-393; discussion 393-394. https://doi.org/10.1016/j.jvs.2008. 09.051.
- Regensteiner, Judith G., Sherita Golden, Amy G. Huebschmann, Elizabeth Barrett-Connor, Alice Y. Chang, Deborah Chyun, Caroline S. Fox, et al. 2015. "Sex Differences in the Cardiovascular Consequences of Diabetes Mellitus: A Scientific Statement from the American Heart Association." *Circulation* 132 (25): 2424–47. https://doi.org/10.1161/CIR. 000000000000343.
- Renaud, Sophie, Hélène Leray-Moraguès, Leila Chenine, Ludovic Canaud, Hélène Vernhet-Kovacsik, and Bernard Canaud. 2012. "Spontaneous Renal Artery Dissection with Renal Infarction." *Clinical Kidney Journal* 5 (3): 261–64. https://doi.org/10.1093/ckj/sfs047.
- Rerkasem, Amaraporn, Saritphat Orrapin, Dominic Pj Howard, Sothida Nantakool, and Kittipan Rerkasem. 2021. "Local Versus General Anaesthesia for Carotid Endarterectomy." The Cochrane Database of Systematic Reviews 10 (October): CD000126. https: //doi.org/10.1002/14651858.CD000126.pub5.
- Rerkasem, Kittipan, and Peter M Rothwell. 2009. "Patch Angioplasty Versus Primary Closure for Carotid Endarterectomy." The Cochrane Database of Systematic Reviews 2009 (4): CD000160. https://doi.org/10.1002/14651858.CD000160.pub3.
- Resnick, Helaine E., Robert S. Lindsay, Mary McGrae McDermott, Richard B. Devereux, Kristina L. Jones, Richard R. Fabsitz, and Barbara V. Howard. 2004. "Relationship of High and Low Ankle Brachial Index to All-Cause and Cardiovascular Disease Mortality: The Strong Heart Study." *Circulation* 109 (6): 733–39. https://doi.org/10.1161/01.CIR. 0000112642.63927.54.
- Reul, G. J., M. J. Jacobs, I. D. Gregoric, M. Calderon, J. M. Duncan, D. A. Ott, J. J. Livesay, and D. A. Cooley. 1991. "Innominate Artery Occlusive Disease: Surgical Approach and Long-Term Results." *Journal of Vascular Surgery* 14 (3): 405–12. https://doi.org/10.1067/ mva.1991.31287.
- Reutersberg, Benedikt, Matthias Trenner, Bernhard Haller, Sarah Geisbüsch, Christian Reeps, and Hans-Henning Eckstein. 2018. "The Incidence of Delayed Complications in Acute Type b Aortic Dissections Is Underestimated." *Journal of Vascular Surgery* 68 (2): 356–63. https://doi.org/10.1016/j.jvs.2017.11.089.
- Reyes Valdivia, Andrés, Africa Duque Santos, Marta Garnica Ureña, Asunción Romero Lozano, Enrique Aracil Sanus, Julia Ocaña Guaita, and Claudio Gandaria. 2017. "Anticoagulation Alone for Aortic Segment Treatment in Symptomatic Primary Aortic Mural Thrombus Patients." Annals of Vascular Surgery 43 (August): 121–26. https://doi.org/10.1016/j.

avsg.2017.01.005.

- Ribeiro, Mauricio, Gustavo S. Oderich, Thanila Macedo, Terri J. Vrtiska, Jan Hofer, Julia Chini, Bernardo Mendes, and Stephen Cha. 2017. "Assessment of Aortic Wall Thrombus Predicts Outcomes of Endovascular Repair of Complex Aortic Aneurysms Using Fenestrated and Branched Endografts." Journal of Vascular Surgery 66 (5): 1321–33. https: //doi.org/10.1016/j.jvs.2017.03.428.
- Ricco, Jean-Baptiste, Fabrice Schneider, Giulio Illuminati, and Russell H. Samson. 2013.
 "The Role of Completion Imaging Following Carotid Artery Endarterectomy." Journal of Vascular Surgery 57 (5): 1432–38. https://doi.org/10.1016/j.jvs.2013.02.001.
- Rice, Todd W., R. Michael Rodriguez, and Richard W. Light. 2006. "The Superior Vena Cava Syndrome: Clinical Characteristics and Evolving Etiology." *Medicine* 85 (1): 37–42. https://doi.org/10.1097/01.md.0000198474.99876.f0.
- Rickey, Ashley K, and Randolph L Geary. 2019a. "125: Renovascular Disease: Pathophysiology, Epidemiology, Clinical Presentation, and Medical Management." In *Rutherford's Vascular Surgery and Endovascular Therapy*, Ninth. Elsevier Inc. https://doi.org/10.1016/ B978-0-323-42791-3.00125-0.
- ———. 2019b. "125: Renovascular Disease: Pathophysiology, Epidemiology, Clinical Presentation, and Medical Management." In, 9th ed. Elsevier Inc. https://www-clinicalkeycom.cmich.idm.oclc.org/#!/content/book/3-s2.0-B9780323427913001250?indexOverride= GLOBAL.
- Ricotta, John J. 2019. "89. Cerebrovascular Disease: Decision Making Including Medical Therapy." In Rutherford's Vascular Surgery and Endovascular Therapy, 20.
- Ridker, Paul M., Eleanor Danielson, Francisco A. H. Fonseca, Jacques Genest, Antonio M. Gotto, John J. P. Kastelein, Wolfgang Koenig, et al. 2008. "Rosuvastatin to Prevent Vascular Events in Men and Women with Elevated c-Reactive Protein." *The New England Journal of Medicine* 359 (21): 2195–2207. https://doi.org/10.1056/NEJMoa0807646.
- Rinehardt, Elena K., John E. Scarborough, and Kyla M. Bennett. 2017. "Current Practice of Thoracic Outlet Decompression Surgery in the United States." *Journal of Vascular Surgery* 66 (3): 858–65. https://doi.org/10.1016/j.jvs.2017.03.436.
- Rizvi, Adnan Z., Manju Kalra, Haraldur Bjarnason, Thomas C. Bower, Cathy Schleck, and Peter Gloviczki. 2008. "Benign Superior Vena Cava Syndrome: Stenting Is Now the First Line of Treatment." *Journal of Vascular Surgery* 47 (2): 372–80. https://doi.org/10.1016/ j.jvs.2007.09.071.
- Roberts, Lauren, Alik Farber, Douglas W. Jones, Karen Woo, Mohammad H. Eslami, Jessica Simons, Mahmoud Malas, Tze-Woei Tan, Denis Rybin, and Jeffrey J. Siracuse. 2019.
 "Tapered Arteriovenous Grafts Do Not Provide Significant Advantage over Nontapered Grafts in Upper Extremity Dialysis Access." Journal of Vascular Surgery 69 (5): 1552–58. https://doi.org/10.1016/j.jvs.2018.08.181.
- Robertson, Vaux, Federica Poli, Ben Hobson, Athanasios Saratzis, and A. Ross Naylor. 2019. "A Systematic Review and Meta-Analysis of the Presentation and Surgical Management of Patients with Carotid Body Tumours." *European Journal of Vascular and Endovascular* Surgery 57 (4): 477–86. https://doi.org/10.1016/j.ejvs.2018.10.038.

Rockman, Caron B. 2019. "86. Cerebrovascular Disease: Epidemiology and Natural History."

In Rutherford's Vascular Surgery and Endovascular Theapy, 25. Elsevier Inc.

- Rockson, S. G., L. T. Miller, R. Senie, M. J. Brennan, J. R. Casley-Smith, E. Földi, M. Földi, et al. 1998. "American Cancer Society Lymphedema Workshop. Workgroup III: Diagnosis and Management of Lymphedema." *Cancer* 83 (12 Suppl American): 2882–85. https: //doi.org/10.1002/(sici)1097-0142(19981215)83:12b+%3C2882::aid-cncr45%3E3.0.co;2-0.
- Rockson, Stanley G. 2019. "168 Lymphedema: Evaluation and Decision Making." In *Ruther*ford's Vascular Surgery and Endovascular Therapy. Elsevier Inc.
- Rodriguez, Joleen Duarte. 2016. "Brachiocephalic Artery Disease Progression Resulting in Complex Steal Phenomena." Journal of Diagnostic Medical Sonography 32 (3): 173–80. https://doi.org/10.1177/8756479316649950.
- Rollo, Johnathon C., Steven M. Farley, Adam Z. Oskowitz, Karen Woo, and Brian G. DeRubertis. 2017. "Contemporary Outcomes After Venography-Guided Treatment of Patients with May-Thurner Syndrome." Journal of Vascular Surgery. Venous and Lymphatic Disorders 5 (5): 667–676.e1. https://doi.org/10.1016/j.jvsv.2017.02.009.
- Ronel, Daniel N., Estomih Mtui, and William B. Nolan. 2004. "Forearm Compartment Syndrome: Anatomical Analysis of Surgical Approaches to the Deep Space." *Plastic and Recon*structive Surgery 114 (3): 697–705. https://doi.org/10.1097/01.prs.0000130967.42426.23.
- Rooke, Thom W., Alan T. Hirsch, Sanjay Misra, Anton N. Sidawy, Joshua A. Beckman, Laura Findeiss, Jafar Golzarian, et al. 2013. "Management of Patients with Peripheral Artery Disease (Compilation of 2005 and 2011 ACCF/AHA Guideline Recommendations): A Report of the American College of Cardiology Foundation/American Heart Association Task Force on Practice Guidelines." Journal of the American College of Cardiology 61 (14): 1555–70. https://doi.org/10.1016/j.jacc.2013.01.004.
- Roon, Anthony J., and Norman Christensen. 1979. "Evaluation and Treatment of Penetrating Cervical Injuries:" The Journal of Trauma: Injury, Infection, and Critical Care 19 (6): 391–97. https://doi.org/10.1097/00005373-197906000-00001.
- Rose, G. A. 1962. "The Diagnosis of Ischaemic Heart Pain and Intermittent Claudication in Field Surveys." Bulletin of the World Health Organization 27 (6): 645–58.
- Rosenfield, Kenneth, Jon S. Matsumura, Seemant Chaturvedi, Tom Riles, Gary M. Ansel, D. Chris Metzger, Lawrence Wechsler, Michael R. Jaff, and William Gray. 2016. "Randomized Trial of Stent Versus Surgery for Asymptomatic Carotid Stenosis." New England Journal of Medicine 374 (11): 1011–20. https://doi.org/10.1056/NEJMoa1515706.
- Rossel, Anne, Helia Robert-Ebadi, Christophe Combescure, Olivier Grosgurin, Jérôme Stirnemann, Alfredo Addeo, Nicolas Garin, Thomas Agoritsas, Jean-Luc Reny, and Christophe Marti. 2019. "Anticoagulant Therapy for Acute Venous Thrombo-Embolism in Cancer Patients: A Systematic Review and Network Meta-Analysis." *PloS One* 14 (3): e0213940. https://doi.org/10.1371/journal.pone.0213940.
- Rothwell, Pm, M Eliasziw, Sa Gutnikov, Cp Warlow, and Hjm Barnett. 2004. "Endarterectomy for Symptomatic Carotid Stenosis in Relation to Clinical Subgroups and Timing of Surgery." *The Lancet* 363 (9413): 915–24. https://doi.org/10.1016/S0140-6736(04)15785-1.
- Rubin, Jonathan M., Jing Gao, Keith Hetel, and Robert Min. 2010. "Duplication Images in Vascular Sonography." Journal of Ultrasound in Medicine: Official Journal of the American

Institute of Ultrasound in Medicine 29 (10): 1385–90. https://doi.org/10.7863/jum.2010. 29.10.1385.

- Rueda, Carlos A., Emilia N. Bittenbinder, Clifford J. Buckley, William T. Bohannon, Marvin D. Atkins, and Ruth L. Bush. 2013. "The Management of Chronic Venous Insufficiency with Ulceration: The Role of Minimally Invasive Perforator Interruption." Annals of Vascular Surgery 27 (1): 89–95. https://doi.org/10.1016/j.avsg.2012.09.001.
- Rutherford, Robert B., J.Dennis Baker, Calvin Ernst, K.Wayne Johnston, John M. Porter, Sam Ahn, and Darrell N. Jones. 1997. "Recommended Standards for Reports Dealing with Lower Extremity Ischemia: Revised Version." Journal of Vascular Surgery 26 (3): 517–38. https://doi.org/10.1016/S0741-5214(97)70045-4.
- Ryer, Evan J., Robert P. Garvin, Travis P. Webb, David P. Franklin, and James R. Elmore. 2012. "Comparison of Outcomes with Coils Versus Vascular Plug Embolization of the Internal Iliac Artery for Endovascular Aortoiliac Aneurysm Repair." *Journal of Vascular Surgery* 56 (5): 1239–45. https://doi.org/10.1016/j.jvs.2012.05.001.
- Sadeghi-Azandaryani, M, D Bürklein, A Ozimek, C Geiger, N Mendl, B Steckmeier, and J Heyn. 2009. "Thoracic Outlet Syndrome: Do We Have Clinical Tests as Predictors for the Outcome After Surgery?" European Journal of Medical Research 14 (10): 443–46. https://doi.org/10.1186/2047-783X-14-10-443.
- Sadek, Mikel, Lowell S. Kabnick, Caron B. Rockman, Todd L. Berland, Di Zhou, Cara Chasin, Glenn R. Jacobowitz, and Mark A. Adelman. 2013. "Increasing Ablation Distance Peripheral to the Saphenofemoral Junction May Result in a Diminished Rate of Endothermal Heat-Induced Thrombosis." *Journal of Vascular Surgery. Venous and Lymphatic Disorders* 1 (3): 257–62. https://doi.org/10.1016/j.jvsv.2013.01.002.
- Safi, H. J., and C. C. Miller. 1999. "Spinal Cord Protection in Descending Thoracic and Thoracoabdominal Aortic Repair." The Annals of Thoracic Surgery 67 (6): 1937-1939; discussion 1953-1958. https://doi.org/10.1016/s0003-4975(99)00397-5.
- Saha, Tisa, Syed Yaseen Naqvi, Omar Abine Ayah, Daniel McCormick, and Sheldon Goldberg. 2017. "Subclavian Artery Disease: Diagnosis and Therapy." *The American Journal of Medicine* 130 (4): 409–16. https://doi.org/10.1016/j.amjmed.2016.12.027.
- Sailer, Anna M., Leonie Paulis, Laura Vergoossen, Joachim E. Wildberger, and Cécile R. L. P. N. Jeukens. 2019. "Optimizing Staff Dose in Fluoroscopy-Guided Interventions by Comparing Clinical Data with Phantom Experiments." Journal of Vascular and Interventional Radiology: JVIR 30 (5): 701–708.e1. https://doi.org/10.1016/j.jvir.2018.11.019.
- Sajid, Muhammad S., Naeem Ahmed, Mittal Desai, Daryll Baker, and George Hamilton. 2007. "Upper Limb Deep Vein Thrombosis: A Literature Review to Streamline the Protocol for Management." Acta Haematologica 118 (1): 10–18. https://doi.org/10.1159/000101700.
- Salhan, Navjeet, Misty Humphries, Samir Sheth, Charles DeMesa, and Julie Freischlag. 2016. "PC214. Ultrasound-Guided Anesthetic or Botulinum Toxin Injection of the Anterior and Middle Scalene Muscles to Assist With Diagnosis and Treatment of Neurogenic Thoracic Outlet Syndrome." Journal of Vascular Surgery 63 (6): 218S. https://doi.org/10.1016/j. jvs.2016.03.365.
- Saliou, Christophe, Alain Raynaud, François Blanc, Michel Azencot, and Jean-Noël Fabiani. 1998. "Idiopathic Renal Arteriovenous Fistula: Treatment with Embolization." Annals of

Vascular Surgery 12 (1): 75–77. https://doi.org/10.1007/s100169900119.

- Sanders, Richard J., Sharon L. Hammond, and Neal M. Rao. 2007. "Diagnosis of Thoracic Outlet Syndrome." Journal of Vascular Surgery 46 (3): 601–4. https://doi.org/10.1016/j. jvs.2007.04.050.
- Sandri, J. L., F. S. Barros, S. Pontes, C. Jacques, and S. X. Salles-Cunha. 1999. "Diameter-Reflux Relationship in Perforating Veins of Patients with Varicose Veins." *Journal of Vascular Surgery* 30 (5): 867–74. https://doi.org/10.1016/s0741-5214(99)70011-x.
- Sanyal, Rupan, Chandana G. Lall, Ramit Lamba, Sadhna Verma, Shetal N. Shah, Temel Tirkes, William A. Berry, and Kumaresan Sandrasegaran. 2012. "Orthotopic Liver Transplantation: Reversible Doppler US Findings in the Immediate Postoperative Period." Radiographics: A Review Publication of the Radiological Society of North America, Inc 32 (1): 199–211. https://doi.org/10.1148/rg.321115006.
- Sarac, Timur P., Connor Gibbons, Lina Vargas, Jane Liu, Sunita Srivastava, James Bena, Tara Mastracci, Vikram S. Kashyap, and Daniel Clair. 2012. "Long-Term Follow-up of Type II Endoleak Embolization Reveals the Need for Close Surveillance." Journal of Vascular Surgery 55 (1): 33–40. https://doi.org/10.1016/j.jvs.2011.07.092.
- Saric, Muhamed, and Itzhak Kronzon. 2012. "Aortic Atherosclerosis and Embolic Events." Current Cardiology Reports 14 (3): 342–49. https://doi.org/10.1007/s11886-012-0261-2.
- Sauve, J. Stephane. 1994. "Can Bruits Distinguish High-Grade from Moderate Symptomatic Carotid Stenosis?" Annals of Internal Medicine 120 (8): 633. https://doi.org/10.7326/ 0003-4819-120-8-199404150-00002.
- Saxon, David R., and Robert H. Eckel. 2016. "Statin Intolerance: A Literature Review and Management Strategies." Progress in Cardiovascular Diseases 59 (2): 153–64. https: //doi.org/10.1016/j.pcad.2016.07.009.
- Scali, Salvatore T., Diego Ayo, Kristina A. Giles, Sarah Gray, Paul Kubilis, Martin Back, Javairiah Fatima, et al. 2019. "Outcomes of Antegrade and Retrograde Open Mesenteric Bypass for Acute Mesenteric Ischemia." Journal of Vascular Surgery 69 (1): 129–40. https: //doi.org/10.1016/j.jvs.2018.04.063.
- Schabelman, Esteban, and Michael Witting. 2010. "The Relationship of Radiocontrast, Iodine, and Seafood Allergies: A Medical Myth Exposed." *The Journal of Emergency Medicine* 39 (5): 701–7. https://doi.org/10.1016/j.jemermed.2009.10.014.
- Schäberle, W., L. Leyerer, W. Schierling, and K. Pfister. 2016. "Ultrasound Diagnostics of Renal Artery Stenosis: Stenosis Criteria, CEUS and Recurrent in-Stent Stenosis." Gefasschirurgie: Zeitschrift Fur Vaskulare Und Endovaskulare Chirurgie: Organ Der Deutschen Und Der Osterreichischen Gesellschaft Fur Gefasschirurgie Unter Mitarbeit Der Schweizerischen Gesellschaft Fur Gefasschirurgie 21: 4–13. https://doi.org/10.1007/s00772-015-0060-3.
- Schanzer, Andres, Louis M. Messina, Kaushik Ghosh, Jessica P. Simons, William P. Robinson, Francesco A Aiello, Robert J. Goldberg, and Allison B. Rosen. 2015. "Follow-Up Compliance After Endovascular Abdominal Aortic Aneurysm Repair in Medicare Beneficiaries." *Journal of Vascular Surgery* 61 (1): 16–22.e1. https://doi.org/10.1016/j.jvs.2014.06.006.
- Schanzer, Harry, and David Eisenberg. 2004. "Management of Steal Syndrome Resulting from Dialysis Access." Seminars in Vascular Surgery 17 (1): 45–49. https://doi.org/10.1053/j.

semvascsurg.2003.11.003.

- Scher, L. A., F. J. Veith, H. Haimovici, R. H. Samson, E. Ascer, S. K. Gupta, and S. Sprayregen. 1984. "Staging of Arterial Complications of Cervical Rib: Guidelines for Surgical Management." Surgery 95 (6): 644–49.
- Schmidt, R., C. Bruns, M. Walter, and H. Erasmi. 1994. "Aorto-Caval Fistula–an Uncommon Complication of Infrarenal Aortic Aneurysms." *The Thoracic and Cardiovascular Surgeon* 42 (4): 208–11. https://doi.org/10.1055/s-2007-1016489.
- Schmidt, Wolfgang A. 2014. "Role of Ultrasound in the Understanding and Management of Vasculitis." Therapeutic Advances in Musculoskeletal Disease 6 (2): 39–47. https://doi. org/10.1177/1759720X13512256.
- Schneider, Darren B., Jon S. Matsumura, Jason T. Lee, Brian G. Peterson, Rabih A. Chaer, and Gustavo S. Oderich. 2017. "Prospective, Multicenter Study of Endovascular Repair of Aortoiliac and Iliac Aneurysms Using the Gore Iliac Branch Endoprosthesis." *Journal of Vascular Surgery* 66 (3): 775–85. https://doi.org/10.1016/j.jvs.2017.02.041.
- Schoolwerth, Anton C., Domenic A. Sica, Barbara J. Ballermann, and Christopher S. Wilcox. 2001. "Renal Considerations in Angiotensin Converting Enzyme Inhibitor Therapy: A Statement for Healthcare Professionals From the Council on the Kidney in Cardiovascular Disease and the Council for High Blood Pressure Research of the American Heart Association." *Circulation* 104 (16): 1985–91. https://doi.org/10.1161/hc4101.096153.
- Schreurs, Rachel H. P., Manuela A. Joore, Daisy P. De Bruijn-Geraets, Hugo Ten Cate, and Arina J. Ten Cate-Hoek. 2022. "A Realist Evaluation to Identify Targets to Improve the Organization of Compression Therapy for Deep Venous Thrombosis- and Chronic Venous Disease Patients." *PloS One* 17 (8): e0272566. https://doi.org/10.1371/journal.pone.0272566.
- Schrock, Theodore. 1968. "Management of Blunt Trauma to the Liver and Hepatic Veins." Archives of Surgery 96 (5): 698. https://doi.org/10.1001/archsurg.1968.01330230006002.
- Schwarz, A., M. Hiss, W. Gwinner, T. Becker, H. Haller, and M. Keberle. 2008. "Course and Relevance of Arteriovenous Fistulas After Renal Transplant Biopsies." *American Journal* of Transplantation 8 (4): 826–31. https://doi.org/10.1111/j.1600-6143.2008.02160.x.
- Scovell, Sherry D., Emel A. Ergul, and Mark F. Conrad. 2018. "Medical Management of Acute Superficial Vein Thrombosis of the Saphenous Vein." *Journal of Vascular Surgery. Venous* and Lymphatic Disorders 6 (1): 109–17. https://doi.org/10.1016/j.jvsv.2017.08.016.
- Sekhar, R. 2018. Yearbook of Vascular and Endovascular Surgery-2, 2018. Place of publication not identified: JAYPEE Brothers MEDICAL P.
- Selvin, Elizabeth, Keattiyoat Wattanakit, Michael W. Steffes, Josef Coresh, and A. Richey Sharrett. 2006. "HbA1c and Peripheral Arterial Disease in Diabetes: The Atherosclerosis Risk in Communities Study." *Diabetes Care* 29 (4): 877–82. https://doi.org/10.2337/ diacare.29.04.06.dc05-2018.
- Sen, Indrani, and Ramesh K. Tripathi. 2016. "Dialysis Access-Associated Steal Syndromes." Seminars in Vascular Surgery 29 (4): 212–26. https://doi.org/10.1053/j.semvascsurg.2017. 04.002.
- Setacci, C., G. Galzerano, F. Setacci, G. De Donato, P. Sirignano, V. Kamargianni, A. Cannizzaro, and A. Cappelli. 2012. "Endovascular Approach to Leriche Syndrome." *The Journal* of Cardiovascular Surgery 53 (3): 301–6.

- Sfyroeras, G. S., C. N. Antonopoulos, G. Mantas, K. G. Moulakakis, J. D. Kakisis, E. Brountzos, C. R. Lattimer, and G. Geroulakos. 2017. "A Review of Open and Endovascular Treatment of Superior Vena Cava Syndrome of Benign Aetiology." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 53 (2): 238–54. https://doi.org/10.1016/j.ejvs.2016.11.013.
- Sgroi, Michael D., Graeme McFarland, Nathan K. Itoga, Ehab Sorial, and Manuel Garcia-Toca. 2019. "Arteriovenous Fistula and Graft Construction in Patients with Implantable Cardiac Devices: Does Side Matter?" Annals of Vascular Surgery 54 (January): 66–71. https://doi.org/10.1016/j.avsg.2018.10.003.
- Shaalan, Wael E., Carl M. Wahlgren, Tina Desai, Giancarlo Piano, Christopher Skelly, and Hisham S. Bassiouny. 2008. "Reappraisal of Velocity Criteria for Carotid Bulb/Internal Carotid Artery Stenosis Utilizing High-Resolution b-Mode Ultrasound Validated with Computed Tomography Angiography." Journal of Vascular Surgery 48 (1): 104–13. https: //doi.org/10.1016/j.jvs.2008.02.068.
- Shalaby, Sherif Y., Trenton R. Foster, Michael R. Hall, Kirstyn E. Brownson, Penny Vasilas, Daniel G. Federman, Hamid R. Mojibian, and Alan Dardik. 2016. "Systemic Inflammatory Disease and Its Association with Type II Endoleak and Late Interventions After Endovascular Aneurysm Repair." JAMA Surgery 151 (2): 147–53. https://doi.org/10.1001/jamasurg. 2015.3219.
- Shalhub, Sherene, Benjamin W. Starnes, Thomas S. Hatsukami, Riyad Karmy-Jones, and Nam T. Tran. 2011. "Repair of Blunt Thoracic Outlet Arterial Injuries: An Evolution from Open to Endovascular Approach." *The Journal of Trauma* 71 (5): E114–121. https: //doi.org/10.1097/TA.0b013e31822cf359.
- Shamblin, W. R., W. H. ReMine, S. G. Sheps, and E. G. Harrison. 1971. "Carotid Body Tumor (Chemodectoma). Clinicopathologic Analysis of Ninety Cases." *American Journal* of Surgery 122 (6): 732–39. https://doi.org/10.1016/0002-9610(71)90436-3.
- Shanmugam, Victoria K. 2019. "137. Vasculitis and Other Uncommon Arteriopathies." In Rutherford's Vascular Surgery and Endovascular Therapy, 16.
- Sharafuddin, Mel J., and Anna E. Marjan. 2017. "Current Status of Carbon Dioxide Angiography." Journal of Vascular Surgery 66 (2): 618–37. https://doi.org/10.1016/j.jvs.2017.03. 446.
- Sharma, Amit, and Robert A. Schwartz. 2012. "Stewart-Treves Syndrome: Pathogenesis and Management." Journal of the American Academy of Dermatology 67 (6): 1342–48. https://doi.org/10.1016/j.jaad.2012.04.028.
- Shaw, Meagan, David J. Del Pizzo, William C. Pevec, James W. Holcroft, and David L. Dawson. 2007. "Case Study: Cystic Adventitial Disease of the Popliteal Artery." Journal for Vascular Ultrasound 31 (1): 45–48. https://doi.org/10.1177/154431670703100109.
- Shaw, Palma, Audra Duncan, Ageliki Vouyouka, and Kathleen Ozsvath. 2011. "Radiation Exposure and Pregnancy." Journal of Vascular Surgery 53 (1): 28S-34S. https://doi.org/ 10.1016/j.jvs.2010.05.140.
- Shaydakov, E., O. Porembskaya, and G. Geroulakos. 2015. "The May-Husni Procedure: A Reappraisal." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 50 (4): 513–17. https://doi.org/10.

1016/j.ejvs.2015.05.010.

- Sheikh, M. A., B. B. Fernandez, B. H. Gray, L. M. Graham, and Teresa L. Carman. 2005. "Endovascular Stenting of Nonmalignant Superior Vena Cava Syndrome." *Catheterization and Cardiovascular Interventions: Official Journal of the Society for Cardiac Angiography and Interventions* 65 (3): 405–11. https://doi.org/10.1002/ccd.20458.
- Shemesh, David, Ilya Goldin, Jamal Hijazi, Ibrahim Zaghal, Daniel Berelowitz, Anthony Verstandig, and Oded Olsha. 2015. "A Prospective Randomized Study of Heparin-Bonded Graft (Propaten) Versus Standard Graft in Prosthetic Arteriovenous Access." Journal of Vascular Surgery 62 (1): 115–22. https://doi.org/10.1016/j.jvs.2015.01.056.
- Shemesh, David, Ilya Goldin, Ibrahim Zaghal, Daniel Berlowitz, David Raveh, and Oded Olsha. 2008. "Angioplasty with Stent Graft Versus Bare Stent for Recurrent Cephalic Arch Stenosis in Autogenous Arteriovenous Access for Hemodialysis: A Prospective Randomized Clinical Trial." Journal of Vascular Surgery 48 (6): 1524–1531.e2. https://doi.org/10.1016/ j.jvs.2008.07.071.
- Shenoy, R. K. 2008. "Clinical and Pathological Aspects of Filarial Lymphedema and Its Management." The Korean Journal of Parasitology 46 (3): 119–25. https://doi.org/10. 3347/kjp.2008.46.3.119.
- Sheth, Sharvil U., Tejas R. Shah, Ziqing Wang, Tahmina Ferdous, Mikel Sadek, and Firas F. Mussa. 2015. "Endovascular Treatment of Acute Renal Failure Secondary to Caval Thrombosis and Suprarenal Filter Migration." Journal of Vascular Surgery. Venous and Lymphatic Disorders 3 (2): 198–200. https://doi.org/10.1016/j.jvsv.2014.04.008.
- Shi, Yan, Peter I. Tsai, Matthew J. Wall, and Ramyar Gilani. 2015. "Intravascular Ultrasound Enhanced Aortic Sizing for Endovascular Treatment of Blunt Aortic Injury." *The Journal of Trauma and Acute Care Surgery* 79 (5): 817–21. https://doi.org/10.1097/TA. 00000000000858.
- Shuja, Fahad. 2019. "117. Upper Extremity Arterial Disease: Epidemiology, Etiology, and Diagnostic Evaluation." In, 11. Elsevier Inc.
- Sidawy, Anton N., Lawrence M. Spergel, Anatole Besarab, Michael Allon, William C. Jennings, Frank T. Padberg, M. Hassan Murad, et al. 2008. "The Society for Vascular Surgery: Clinical Practice Guidelines for the Surgical Placement and Maintenance of Arteriovenous Hemodialysis Access." Journal of Vascular Surgery 48 (5): S2–25. https://doi.org/10.1016/ j.jvs.2008.08.042.
- Silver, Frank L., Ariane Mackey, Wayne M. Clark, William Brooks, Carlos H. Timaran, David Chiu, Larry B. Goldstein, et al. 2011. "Safety of Stenting and Endarterectomy by Symptomatic Status in the Carotid Revascularization Endarterectomy Vs. Stenting Trial (CREST)." Stroke; a Journal of Cerebral Circulation 42 (3): 675–80. https://doi.org/ 10.1161/STROKEAHA.110.610212.
- Singer, Gregory M., Michael S. Remetz, Jeptha P. Curtis, and John F. Setaro. 2009. "Impact of Baseline Renal Function on Outcomes of Renal Artery Stenting in Hypertensive Patients." *The Journal of Clinical Hypertension* 11 (11): 615–20. https://doi.org/10.1111/j.1751-7176.2009.00167.x.
- Singh, Tejas P., Shannon A. Wong, Joseph V. Moxon, T. Christian Gasser, and Jonathan Golledge. 2019. "Systematic Review and Meta-Analysis of the Association Between Intra-

luminal Thrombus Volume and Abdominal Aortic Aneurysm Rupture." Journal of Vascular Surgery 70 (6): 2065–2073.e10. https://doi.org/10.1016/j.jvs.2019.03.057.

- Sinha, Sidhartha, Jon Houghton, Peter J. Holt, Matt M. Thompson, Ian M. Loftus, and Robert J. Hinchliffe. 2012. "Popliteal Entrapment Syndrome." Journal of Vascular Surgery 55 (1): 252–262.e30. https://doi.org/10.1016/j.jvs.2011.08.050.
- Sinnakirouchenan, Ramapriya, and Jean L. Holley. 2011. "Peritoneal Dialysis Versus Hemodialysis: Risks, Benefits, and Access Issues." Advances in Chronic Kidney Disease 18 (6): 428–32. https://doi.org/10.1053/j.ackd.2011.09.001.
- Siracuse, Jeffrey J., Thomas W. Cheng, Nkiruka V. Arinze, Scott R. Levin, Douglas W. Jones, Mahmoud B. Malas, Jeffrey A. Kalish, Denis Rybin, and Alik Farber. 2019. "Snuffbox Arteriovenous Fistulas Have Similar Outcomes and Patency as Wrist Arteriovenous Fistulas." *Journal of Vascular Surgery* 70 (2): 554–61. https://doi.org/10.1016/j.jvs.2018.11.030.
- Siracuse, Jeffrey J., Heather L. Gill, Inkyong Parrack, Zhen S. Huang, Darren B. Schneider, Peter H. Connolly, and Andrew J. Meltzer. 2014. "Variability in Anesthetic Considerations for Arteriovenous Fistula Creation." *The Journal of Vascular Access* 15 (5): 364–69. https: //doi.org/10.5301/jva.5000215.
- Siracuse, Jeffrey J., Paul C. Johnston, Douglas W. Jones, Heather L. Gill, Peter H. Connolly, Andrew J. Meltzer, and Darren B. Schneider. 2015. "Infraclavicular First Rib Resection for the Treatment of Acute Venous Thoracic Outlet Syndrome." Journal of Vascular Surgery: Venous and Lymphatic Disorders 3 (4): 397–400. https://doi.org/10.1016/j.jvsv.2015.06. 002.
- Siscovick, David S., Thomas A. Barringer, Amanda M. Fretts, Jason H. Y. Wu, Alice H. Lichtenstein, Rebecca B. Costello, Penny M. Kris-Etherton, et al. 2017. "Omega-3 Polyunsaturated Fatty Acid (Fish Oil) Supplementation and the Prevention of Clinical Cardiovascular Disease: A Science Advisory from the American Heart Association." *Circulation* 135 (15): e867–84. https://doi.org/10.1161/CIR.00000000000482.
- Size, Gail P, Laurie Lozanski, Troy Russo, Eileen French-Sherry, and Christopher L Skelly. 2013. Inside Ultrasound Vascular Reference Guide.
- Smeds, Matthew R., Audra A. Duncan, Michael P. Harlander-Locke, Peter F. Lawrence, Sean Lyden, Javariah Fatima, Mark K. Eskandari, and Vascular Low-Frequency Disease Consortium. 2016. "Treatment and Outcomes of Aortic Endograft Infection." Journal of Vascular Surgery 63 (2): 332–40. https://doi.org/10.1016/j.jvs.2015.08.113.
- Smith, George E., Risha Gohil, and Ian C. Chetter. 2012. "Factors Affecting the Patency of Arteriovenous Fistulas for Dialysis Access." Journal of Vascular Surgery 55 (3): 849–55. https://doi.org/10.1016/j.jvs.2011.07.095.
- Sörelius, Karl, Anders Wanhainen, Mia Furebring, Martin Björck, Peter Gillgren, Kevin Mani, and Swedish Collaborator Group for Mycotic Abdominal Aortic Aneurysms^{*}. 2016. "Nationwide Study of the Treatment of Mycotic Abdominal Aortic Aneurysms Comparing Open and Endovascular Repair." *Circulation* 134 (23): 1822–32. https://doi.org/10.1161/ CIRCULATIONAHA.116.024021.
- Sorrentino, Sabato, Phong Nguyen, Nadia Salerno, Alberto Polimeni, Jolanda Sabatino, Angela Makris, Annemarie Hennessy, et al. 2020. "Standard Versus Ultrasound-Guided Cannulation of the Femoral Artery in Patients Undergoing Invasive Procedures: A Meta-

Analysis of Randomized Controlled Trials." Journal of Clinical Medicine 9 (3): 677. https://doi.org/10.3390/jcm9030677.

- Sperry, Jason L., Ernest E. Moore, Raul Coimbra, Martin Croce, James W. Davis, Riyad Karmy-Jones, Robert C. McIntyre, et al. 2013. "Western Trauma Association Critical Decisions in Trauma: Penetrating Neck Trauma." Journal of Trauma and Acute Care Surgery 75 (6): 936–40. https://doi.org/10.1097/TA.0b013e31829e20e3.
- Spinelli, F., F. Benedetto, G. Passari, M. La Spada, G. Carella, F. Stilo, G. De Caridi, and S. Lentini. 2010. "Bypass Surgery for the Treatment of Upper Limb Chronic Ischaemia." *European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery* 39 (2): 165–70. https://doi.org/10.1016/j.ejvs.2009. 10.015.
- Stamler, J., O. Vaccaro, J. D. Neaton, and D. Wentworth. 1993. "Diabetes, Other Risk Factors, and 12-Yr Cardiovascular Mortality for Men Screened in the Multiple Risk Factor Intervention Trial." *Diabetes Care* 16 (2): 434–44. https://doi.org/10.2337/diacare.16.2. 434.
- Stanley, James C, Frank Veith, and Thomas W. Wakefield. 2014. Current Therapy in Vascular and Endovascular Surgery. Elsevier Health Sciences.
- Starnes, Benjamin W., Rachel S. Lundgren, Martin Gunn, Samantha Quade, Thomas S. Hatsukami, Nam T. Tran, Nahush Mokadam, and Gabriel Aldea. 2012. "A New Classification Scheme for Treating Blunt Aortic Injury." *Journal of Vascular Surgery* 55 (1): 47–54. https://doi.org/10.1016/j.jvs.2011.07.073.
- Stecker, Michael S., Stephen Balter, Richard B. Towbin, Donald L. Miller, Eliseo Vañó, Gabriel Bartal, J. Fritz Angle, et al. 2009. "Guidelines for Patient Radiation Dose Management." *Journal of Vascular and Interventional Radiology: JVIR* 20 (7 Suppl): S263–273. https: //doi.org/10.1016/j.jvir.2009.04.037.
- Stępniak, Anna. 2018. "Uterine Artery Embolization in the Treatment of Symptomatic Fibroids State of the Art 2018." Menopause Review 17 (4): 141–43. https://doi.org/10.5114/pm.2018.81733.
- Stevens, J W, E Simpson, S Harnan, H Squires, Y Meng, S Thomas, J Michaels, and G Stansby. 2012. "Systematic Review of the Efficacy of Cilostazol, Naftidrofuryl Oxalate and Pentoxifylline for the Treatment of Intermittent Claudication." British Journal of Surgery 99 (12): 1630–38. https://doi.org/10.1002/bjs.8895.
- Stokmans, R. A., E. M. Willigendael, J. a. W. Teijink, J. A. Ten Bosch, M. R. H. M. van Sambeek, and Ph W. M. Cuypers. 2013. "Challenging the Evidence for Pre-Emptive Coil Embolisation of the Internal Iliac Artery During Endovascular Aneurysm Repair." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 45 (3): 220–26. https://doi.org/10.1016/j.ejvs.2012. 12.001.
- Stone, Patrick A., John E. Campbell, and Ali F. AbuRahma. 2014. "Femoral Pseudoaneurysms After Percutaneous Access." Journal of Vascular Surgery 60 (5): 1359–66. https://doi.org/10.1016/j.jvs.2014.07.035.
- Stone, Patrick A, and Stephen M Hass. 2019. "21. Vascular Laboratory: Arterial Duplex Scanning." In Rutherford's Vascular Surgery and Endovascular Therapy, Ninth. Elsevier

Inc. https://doi.org/10.1016/B978-0-323-42791-3.00021-9.

- Stone, Patrick A., Mohit Srivastava, John E. Campbell, Albeir Y. Mousa, Stephen H. Hass, Hamza Kazmi, Daniel D. Dearing, and Ali F. AbuRahma. 2011. "A 10-Year Experience of Infection Following Carotid Endarterectomy with Patch Angioplasty." *Journal of Vascular Surgery* 53 (6): 1473–77. https://doi.org/10.1016/j.jvs.2011.02.020.
- Strandness, D. E., and J. W. Bell. 1964. "AN EVALUATION OF THE HEMODYNAMIC RESPONSE OF THE CLAUDICATING EXTREMITY TO EXERCISE." Surgery, Gynecology and Obstetrics 119 (December): 1237–42.
- Strandness, D. E., and David S. Sumner. 1975. Hemodynamics for Surgeons. Modern Surgical Monographs. New York: Grune & Stratton.
- Strobl, Frederik F., Klaus Brechtel, Jörg Schmehl, Thomas Zeller, Maximilian F. Reiser, Claus D. Claussen, and Gunnar Tepe. 2013. "Twelve-Month Results of a Randomized Trial Comparing Mono with Dual Antiplatelet Therapy in Endovascularly Treated Patients with Peripheral Artery Disease." Journal of Endovascular Therapy: An Official Journal of the International Society of Endovascular Specialists 20 (5): 699–706. https://doi.org/10.1583/13-4275MR.1.
- Subramanian, Anuradha, Gary Vercruysse, Christopher Dente, Amy Wyrzykowski, Erin King, and David V. Feliciano. 2008. "A Decade's Experience with Temporary Intravascular Shunts at a Civilian Level i Trauma Center." *The Journal of Trauma* 65 (2): 316-324; discussion 324-326. https://doi.org/10.1097/TA.0b013e31817e5132.
- Sueyoshi, Eijun, Hiroki Nagayama, Takeshi Hayashida, Ichiro Sakamoto, and Masataka Uetani. 2013. "Comparison of Outcome in Aortic Dissection with Single False Lumen Versus Multiple False Lumens: CT Assessment." Radiology 267 (2): 368–75. https: //doi.org/10.1148/radiol.12121274.
- Sugawara, Yoshifumi, Takanori Kikuchi, Toshihiro Ueda, Mamoru Nishizaki, Shigeru Nakata, Teruhito Mochizuki, and Junpei Ikezoe. n.d. "Usefulness of Brain SPECT to Evaluate Brain Tolerance and Hemodynamic Changes During Temporary Balloon Occlusion Test and After Permanent Carotid Occlusion," 8.
- Sule, J. A., and R. B. Dharmaraj. 2016. "Surgeon Modified Fenestrated Endovascular Abdominal Aortic Repair (f-EVAR) for Subacute Multifocal Mycotic Abdominal and Iliac Artery Saccular Aneurysms." *EJVES Short Reports* 32: 7–11. https://doi.org/10.1016/j.ejvssr. 2016.03.007.
- Sztajnkrycer, Matthew D. 2019. "Local Anesthetics." In Goldfrank's Toxicologic Emergencies, 11e, edited by Lewis S. Nelson, Mary Ann Howland, Neal A. Lewin, Silas W. Smith, Lewis R. Goldfrank, and Robert S. Hoffman. McGraw-Hill Education. accesspharmacy.mhmedical.com/content.aspx?aid=1163013831.
- Talapatra, Kaustav, Soumadip Panda, Sandeep Goyle, Kallol Bhadra, and Rajesh Mistry. 2016. "Superior Vena Cava Syndrome: A Radiation Oncologist's Perspective." Journal of Cancer Research and Therapeutics 12 (2): 515–19. https://doi.org/10.4103/0973-1482.177503.
- Talebi, Mahnaz, Behzad Salari, Hossein Ghannadan, Farzad Kakaei, and Sima Abedi Azar. 2011. "Nerve Conduction Changes Following Arteriovenous Fistula Construction in Hemodialysis Patients." *International Urology and Nephrology* 43 (3): 849–53. https://doi.org/10.1007/s11255-010-9740-9.

- Tanaka, Akiko, Ross Milner, and Takeyoshi Ota. 2015. "Kommerell's Diverticulum in the Current Era: A Comprehensive Review." General Thoracic and Cardiovascular Surgery 63 (5): 245–59. https://doi.org/10.1007/s11748-015-0521-3.
- Tansavatdi, Katharine, Arthur B. Dublin, Paul J. Donald, and Brian Dahlin. 2015. "Combined Balloon Test Occlusion and SPECT Analysis for Carotid Sacrifice: Angiographic Predictors for Success or Failure?" Journal of Neurological Surgery. Part B, Skull Base 76 (4): 249–51. https://doi.org/10.1055/s-0035-1547363.
- Tashiro, Teruko, Sorin V. Pislaru, Jodi M. Blustin, Vuyisile T. Nkomo, Martin D. Abel, Christopher G. Scott, and Patricia A. Pellikka. 2014. "Perioperative Risk of Major Non-Cardiac Surgery in Patients with Severe Aortic Stenosis: A Reappraisal in Contemporary Practice." *European Heart Journal* 35 (35): 2372–81. https://doi.org/10.1093/eurheartj/ ehu044.
- Taudorf, M., L. P. Jensen, K. C. Vogt, J. Grønvall, T. V. Schroeder, and L. Lönn. 2014. "Endograft Limb Occlusion in EVAR: Iliac Tortuosity Quantified by Three Different Indices on the Basis of Preoperative CTA." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 48 (5): 527–33. https://doi.org/10.1016/j.ejvs.2014.04.018.
- Taylor, L. M., J. M. Edwards, B. Brant, E. S. Phinney, and J. M. Porter. 1987. "Autogenous Reversed Vein Bypass for Lower Extremity Ischemia in Patients with Absent or Inadequate Greater Saphenous Vein." American Journal of Surgery 153 (5): 505–10. https://doi.org/ 10.1016/0002-9610(87)90803-8.
- Teixeira, Pedro Gr, Karen Woo, Adam W. Beck, Salvatore T. Scali, Fred A. Weaver, and Vascular Quality Initiative (VQI)[®] Society for Vascular Surgery. 2017. "Association of Left Subclavian Artery Coverage Without Revascularization and Spinal Cord Ischemia in Patients Undergoing Thoracic Endovascular Aortic Repair: A Vascular Quality Initiative[®] Analysis." Vascular 25 (6): 587–97. https://doi.org/10.1177/1708538116681910.
- Tembey, Raina Anil, Aneeta S. Bajaj, Prasad K. Wagle, and Abdul Samad Ansari. 2015. "Real-Time Ultrasound: Key Factor in Identifying Celiac Artery Compression Syndrome." *The Indian Journal of Radiology and Imaging* 25 (2): 202–5. https://doi.org/10.4103/0971-3026.155882.
- Ten Cate-Hoek, Arina J., Elham E. Amin, Annemieke C. Bouman, Karina Meijer, Lidwine W. Tick, Saskia Middeldorp, Guy J. M. Mostard, et al. 2018. "Individualised Versus Standard Duration of Elastic Compression Therapy for Prevention of Post-Thrombotic Syndrome (IDEAL DVT): A Multicentre, Randomised, Single-Blind, Allocation-Concealed, Non-Inferiority Trial." *The Lancet. Haematology* 5 (1): e25–33. https://doi.org/10.1016/S2352-3026(17)30227-2.
- Teodorescu, Victoria, Susan Gustavson, and Harry Schanzer. 2012. "Duplex Ultrasound Evaluation of Hemodialysis Access: A Detailed Protocol." *International Journal of Nephrology* 2012: 508956. https://doi.org/10.1155/2012/508956.
- Tepe, Gunnar, Rüdiger Bantleon, Klaus Brechtel, Jörg Schmehl, Thomas Zeller, Claus D. Claussen, and Frederik F. Strobl. 2012. "Management of Peripheral Arterial Interventions with Mono or Dual Antiplatelet Therapy-the MIRROR Study: A Randomised and Double-Blinded Clinical Trial." European Radiology 22 (9): 1998–2006. https://doi.org/10.1007/

s00330-012-2441-2.

- Teter, Katherine, Ezra Schrem, Neel Ranganath, Mark Adelman, Jeffrey Berger, Rebecca Sussman, Bhama Ramkhelawon, Caron Rockman, and Thomas S. Maldonado. 2019. "Presentation and Management of Inferior Vena Cava Thrombosis." Annals of Vascular Surgery 56 (April): 17–23. https://doi.org/10.1016/j.avsg.2018.08.082.
- Textor, Stephen C., Sanjay Misra, and Gustavo S. Oderich. 2013. "Percutaneous Revascularization for Ischemic Nephropathy: The Past, Present, and Future." *Kidney International* 83 (1): 28–40. https://doi.org/10.1038/ki.2012.363.
- Thom, Christopher, and Mary Warlaumont. 2017. "A Necrotizing Fasciitis Fake Out on Point-of-Care Ultrasound-Watch the Shadow." *The Journal of Emergency Medicine* 52 (4): 523–26. https://doi.org/10.1016/j.jemermed.2016.09.007.
- Thomas, Matthew, Andrew Hollingsworth, and Reza Mofidi. 2019. "Endovascular Management of Acute Lower Limb Deep Vein Thrombosis: A Systematic Review and Meta-Analysis." Annals of Vascular Surgery 58 (July): 363–70. https://doi.org/10.1016/j.avsg. 2018.12.067.
- Thompson, S. G., L. C. Brown, M. J. Sweeting, M. J. Bown, L. G. Kim, M. J. Glover, M. J. Buxton, and J. T. Powell. 2013. "Systematic Review and Meta-Analysis of the Growth and Rupture Rates of Small Abdominal Aortic Aneurysms: Implications for Surveillance Intervals and Their Cost-Effectiveness." *Health Technology Assessment (Winchester, England)* 17 (41): 1–118. https://doi.org/10.3310/hta17410.
- Thomsen, Thordis, Nete Villebro, and Ann Merete Møller. 2014. "Interventions for Preoperative Smoking Cessation." *The Cochrane Database of Systematic Reviews*, no. 3 (March): CD002294. https://doi.org/10.1002/14651858.CD002294.pub4.
- Timaran, Carlos H., Takao Ohki, Soo J. Rhee, Frank J. Veith, Nicholas J. Gargiulo, Hisako Toriumi, Mahmood B. Malas, William D. Suggs, Reese A. Wain, and Evan C. Lipsitz. 2004. "Predicting Aneurysm Enlargement in Patients with Persistent Type II Endoleaks." *Journal of Vascular Surgery* 39 (6): 1157–62. https://doi.org/10.1016/j.jvs.2003.12.033.
- Tisi, Paul V., and Mary M. Than. 2014. "Type of Incision for Below Knee Amputation." The Cochrane Database of Systematic Reviews, no. 4 (April): CD003749. https://doi.org/10. 1002/14651858.CD003749.pub3.
- Tjaden, Bruce L., Harleen Sandhu, Charles Miller, Dennis Gable, Santi Trimarchi, Fred Weaver, and Ali Azizzadeh. 2018. "Outcomes from the Gore Global Registry for Endovascular Aortic Treatment in Patients Undergoing Thoracic Endovascular Aortic Repair for Type b Dissection." Journal of Vascular Surgery 68 (5): 1314–23. https://doi.org/10. 1016/j.jvs.2018.03.391.
- Tosti, Rick, Sezai Özkan, Robert M. Schainfeld, and Kyle R. Eberlin. 2017. "Radial Artery Pseudoaneurysm." The Journal of Hand Surgery 42 (4): 295.e1–6. https://doi.org/10. 1016/j.jhsa.2017.01.024.
- Tracci, MARGARET C, Rishi Roy, and Gilbert R. Upchurch. n.d. "70. Aortoiliac Aneurysms: Evaluation, Decision Making, and Medical Management." In *Rutherford's Vascular Surgery* and Endovascular Therapy, 9th Edition, 884–93.
- Tran, Daniel D., and Charles A. Andersen. 2011. "Axillary Sheath Hematomas Causing Neurologic Complications Following Arterial Access." Annals of Vascular Surgery 25 (5):

697.e5-8. https://doi.org/10.1016/j.avsg.2010.12.024.

- Tran, Huyen A., Sanjeev D. Chunilal, Paul L. Harper, Huy Tran, Erica M. Wood, Alex S. Gallus, Australasian Society of Thrombosis, and Haemostasis (ASTH). 2013. "An Update of Consensus Guidelines for Warfarin Reversal." *The Medical Journal of Australia* 198 (4): 198–99. https://doi.org/10.5694/mja12.10614.
- Tran, Victor, Alexander Shiferson, Anil P. Hingorani, Enrico Ascher, Natalie Marks, Kapil Gopal, Nirav Patel, and Theresa Jacob. 2009. "Use of the StarClose Device for Closure of Inadvertent Subclavian Artery Punctures." Annals of Vascular Surgery 23 (5): 688.e11–13. https://doi.org/10.1016/j.avsg.2009.06.005.
- Treitl, Karla Maria, Cosima König, Maximilian F. Reiser, and Marcus Treitl. 2015. "Complications of Transbrachial Arterial Access for Peripheral Endovascular Interventions." Journal of Endovascular Therapy: An Official Journal of the International Society of Endovascular Specialists 22 (1): 63–70. https://doi.org/10.1177/1526602814564363.
- Trimarchi, Santi, Christoph A. Nienaber, Vincenzo Rampoldi, Truls Myrmel, Toru Suzuki, Eduardo Bossone, Valerio Tolva, et al. 2006. "Role and Results of Surgery in Acute Type b Aortic Dissection: Insights from the International Registry of Acute Aortic Dissection (IRAD)." Circulation 114 (1_supplement). https://doi.org/10.1161/CIRCULATIONAHA. 105.000620.
- Tritschler, Tobias, Noémie Kraaijpoel, Grégoire Le Gal, and Philip S. Wells. 2018. "Venous Thromboembolism: Advances in Diagnosis and Treatment." *JAMA* 320 (15): 1583–94. https://doi.org/10.1001/jama.2018.14346.
- Trompeter, Markus, Thurid Brazda, Christopher T. Remy, Thomas Vestring, and Peter Reimer. 2002. "Non-Occlusive Mesenteric Ischemia: Etiology, Diagnosis, and Interventional Therapy." *European Radiology* 12 (5): 1179–87. https://doi.org/10.1007/s00330-001-1220-2.
- Tsimikas, Sotirios, Emmanouil S. Brilakis, Elizabeth R. Miller, Joseph P. McConnell, Ryan J. Lennon, Kenneth S. Kornman, Joseph L. Witztum, and Peter B. Berger. 2005. "Oxidized Phospholipids, Lp(a) Lipoprotein, and Coronary Artery Disease." *The New England Journal of Medicine* 353 (1): 46–57. https://doi.org/10.1056/NEJMoa043175.
- Tuveson, Viktoria, Hedvig E Löfdahl, and Rebecka Hultgren. 2016. "Patients with Abdominal Aortic Aneurysm Have a High Prevalence of Popliteal Artery Aneurysms." Vascular Medicine 21 (4): 369–75. https://doi.org/10.1177/1358863X16648404.
- Ultee, Klaas H. J., Peter A. Soden, Sara L. Zettervall, Jeremy Darling, Hence J. M. Verhagen, and Marc L. Schermerhorn. 2016a. "Conversion from Endovascular to Open Abdominal Aortic Aneurysm Repair." *Journal of Vascular Surgery* 64 (1): 76–82. https://doi.org/10. 1016/j.jvs.2015.12.055.
- ———. 2016b. "Conversion from EVAR to Open Abdominal Aortic Aneurysm Repair." Journal of Vascular Surgery 64 (1): 76–82. https://doi.org/10.1016/j.jvs.2015.12.055.
- United States Surgeon General. 2014. "The Health Consequences of Smoking 50 Years of Progress: A Report of the Surgeon General: (510072014-001)." American Psychological Association. https://doi.org/10.1037/e510072014-001.
- University, © Stanford, Stanford, and California 94305 Copyright Complaints. n.d. "Radiation Protection Guidance for Hospital Staff – Stanford Environmental Health and Safety." https://

//ehs.stanford.edu/manual/radiation-protection-guidance-hospital-staff.

- Urschel, Harold C., and Amit N. Patel. 2008. "Surgery Remains the Most Effective Treatment for Paget-Schroetter Syndrome: 50 Years' Experience." The Annals of Thoracic Surgery 86 (1): 254–60. https://doi.org/10.1016/j.athoracsur.2008.03.021.
- Vacirca, Andrea, Gianluca Faggioli, Rodolfo Pini, Enrico Gallitto, Chiara Mascoli, Laura Maria Cacioppa, Mauro Gargiulo, and Andrea Stella. 2019. "The Outcome of Technical Intraoperative Complications Occurring in Standard Aortic Endovascular Repair." Annals of Vascular Surgery 56 (April): 153–62. https://doi.org/10.1016/j.avsg.2018.08.092.
- Vahdatpour, Babak, Razieh Maghroori, Mojgan Mortazavi, and Saeid Khosrawi. 2012. "Evaluation of Ulnar Neuropathy on Hemodialysis Patients." Journal of Research in Medical Sciences : The Official Journal of Isfahan University of Medical Sciences 17 (10): 905–10. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC3698646/.
- Valentine, R. James, J. Gregory Modrall, and G. Patrick Clagett. 2005. "Hand Ischemia After Radial Artery Cannulation." Journal of the American College of Surgeons 201 (1): 18–22. https://doi.org/10.1016/j.jamcollsurg.2005.01.011.
- Valls, Joan, Maranta Peiro-Chamarro, Serafí Cambray, Jessica Molina-Seguin, Ikram Benabdelhak, and Francisco Purroy. 2017. "A Current Estimation of the Early Risk of Stroke After Transient Ischemic Attack: A Systematic Review and Meta-Analysis of Recent Intervention Studies." Cerebrovascular Diseases (Basel, Switzerland) 43 (1-2): 90–98. https://doi.org/10.1159/000452978.
- van Gulik, Thomas M., and Ivo Schoots. 2005. "Anastomosis of Riolan Revisited: The Meandering Mesenteric Artery." Archives of Surgery (Chicago, Ill.: 1960) 140 (12): 1225– 29. https://doi.org/10.1001/archsurg.140.12.1225.
- Vascular Surgery Lower Extremity Guidelines Writing Group, Society for, Michael S. Conte, Frank B. Pomposelli, Daniel G. Clair, Patrick J. Geraghty, James F. McKinsey, Joseph L. Mills, et al. 2015. "Society for Vascular Surgery Practice Guidelines for Atherosclerotic Occlusive Disease of the Lower Extremities: Management of Asymptomatic Disease and Claudication." Journal of Vascular Surgery 61 (3 Suppl): 2S-41S. https://doi.org/10.1016/ j.jvs.2014.12.009.
- Vascular Ultrasound, Society for. 2019. Lower Extremity Venous Duplex Evaluation for Thrombosis. Vascular Technology Professional Performance Guidelines. https://higherlogicdownload.s3.amazonaws.com/SVUNET/c9a8d83b-2044-4a4e-b3eccd4b2f542939/UploadedImages/PPG_Docs/13_Lower_Extremity_Venous_Duplex_ Evaluation__Updated_2019_.pdf.
- Vasquez, Michael A., Eberhard Rabe, Robert B. McLafferty, Cynthia K. Shortell, William A. Marston, David Gillespie, Mark H. Meissner, and Robert B. Rutherford. 2010. "Revision of the Venous Clinical Severity Score: Venous Outcomes Consensus Statement: Special Communication of the American Venous Forum Ad Hoc Outcomes Working Group." Journal of Vascular Surgery 52 (5): 1387–96. https://doi.org/10.1016/j.jvs.2010.06.161.
- Vedantham, Suresh, Samuel Z. Goldhaber, Jim A. Julian, Susan R. Kahn, Michael R. Jaff, David J. Cohen, Elizabeth Magnuson, et al. 2017. "Pharmacomechanical Catheter-Directed Thrombolysis for Deep-Vein Thrombosis." New England Journal of Medicine 377 (23): 2240–52. https://doi.org/10.1056/NEJMoa1615066.

- Veith, Frank J., Mario Lachat, Dieter Mayer, Martin Malina, Jan Holst, Manish Mehta, Eric L. G. Verhoeven, et al. 2009. "Collected World and Single Center Experience with Endovascular Treatment of Ruptured Abdominal Aortic Aneurysms." Annals of Surgery 250 (5): 818–24. https://doi.org/10.1097/SLA.0b013e3181bdd7f5.
- Velazquez-ramirez, Gabriela, and Matthew A Corriere. 2019. "129: Renovascular Disease : Acute Occlusive and Ischemic Events." In Rutherford's Vascular Surgery and Endovascular Therapy, Ninth. Elsevier Inc. https://doi.org/10.1016/B978-0-323-42791-3.00129-8.
- Velmahos, George C., and Konstantinos G. Toutouzas. 2002. "Vascular Trauma and Compartment Syndromes." The Surgical Clinics of North America 82 (1): 125–41, xxi. https: //doi.org/10.1016/S0039-6109(03)00145-2.
- Vemuri, Chandu, Lauren N. McLaughlin, Ahmmad A. Abuirqeba, and Robert W. Thompson. 2017. "Clinical Presentation and Management of Arterial Thoracic Outlet Syndrome." *Journal of Vascular Surgery* 65 (5): 1429–39. https://doi.org/10.1016/j.jvs.2016.11.039.
- Vemuri, Chandu, Payam Salehi, Jaime Benarroch-Gampel, Lauren N. McLaughlin, and Robert W. Thompson. 2016. "Diagnosis and Treatment of Effort-Induced Thrombosis of the Axillary Subclavian Vein Due to Venous Thoracic Outlet Syndrome." Journal of Vascular Surgery: Venous and Lymphatic Disorders 4 (4): 485–500. https://doi.org/10.1016/j.jvsv. 2016.01.004.
- Verma, Himanshu, Narendranadh Meda, Simit Vora, Robbie K. George, and Ramesh K. Tripathi. 2014. "Contemporary Management of Symptomatic Primary Aortic Mural Thrombus." Journal of Vascular Surgery 60 (6): 1524–34. https://doi.org/10.1016/j.jvs.2014.08. 057.
- Viazzi, Francesca, Giovanna Leoncini, Lorenzo E. Derchi, and Roberto Pontremoli. 2014. "Ultrasound Doppler Renal Resistive Index: A Useful Tool for the Management of the Hypertensive Patient." Journal of Hypertension 32 (1): 149–53. https://doi.org/10.1097/ HJH.0b013e328365b29c.
- Vierhout, B. P., C. J. Zeebregts, J. J. A. M. van den Dungen, and M. M. P. J. Reijnen. 2010. "Changing Profiles of Diagnostic and Treatment Options in Subclavian Artery Aneurysms." *European Journal of Vascular and Endovascular Surgery* 40 (1): 27–34. https://doi.org/ 10.1016/j.ejvs.2010.03.011.
- Vilela, Marcelo D., Robert Goodkin, David A. Lundin, and David W. Newell. 2005. "Rotational Vertebrobasilar Ischemia: Hemodynamic Assessment and Surgical Treatment." *Neurosurgery* 56 (1): 36-43; discussion 43-45. https://doi.org/10.1227/01.neu.0000146441. 93026.ce.
- Villalta, SB, P Bagatella, A Piccioli, A Lensing, MH Prins, and P Prandoni. 1994. "Assessment of Validity and Reproducibility of a Clinical Scale for the Post-Thrombotic Syndrome." *Haemostasis* 24 (Suppl. 1): 158a. https://doi.org/10.1159/000314254.
- Vitti, M. J., D. V. Robinson, M. Hauer-Jensen, B. W. Thompson, T. J. Ranval, G. Barone, R. W. Barnes, and J. F. Eidt. 1994. "Wound Healing in Forefoot Amputations: The Predictive Value of Toe Pressure." Annals of Vascular Surgery 8 (1): 99–106. https: //doi.org/10.1007/BF02133411.
- Vliet, J. A. van der, H. W. Palamba, D. M. Scharn, S. F. S. van Roye, and F. G. M. Buskens. 1995. "Arterial Reconstruction for Subclavian Obstructive Disease: A Comparison of Ex-

trathoracic Procedures." European Journal of Vascular and Endovascular Surgery 9 (4): 454–58. https://doi.org/10.1016/S1078-5884(05)80015-2.

- Voeks, Jenifer H., George Howard, Gary S. Roubin, Mahmoud B. Malas, David J. Cohen, W. Charles Sternbergh, Herbert D. Aronow, et al. 2011. "Age and Outcomes After Carotid Stenting and Endarterectomy: The Carotid Revascularization Endarterectomy Versus Stenting Trial." Stroke 42 (12): 3484–90. https://doi.org/10.1161/STROKEAHA. 111.624155.
- Vogeler, Tiffany, Claire McClain, and Kirk E. Evoy. 2016. "Combination Bupropion SR and Varenicline for Smoking Cessation: A Systematic Review." *The American Journal of Drug* and Alcohol Abuse 42 (2): 129–39. https://doi.org/10.3109/00952990.2015.1117480.
- Voigt, Soraya L., Muath Bishawi, David Ranney, Babatunde Yerokun, Richard L. McCann, and G. Chad Hughes. 2019. "Outcomes of Carotid-Subclavian Bypass Performed in the Setting of Thoracic Endovascular Aortic Repair." Journal of Vascular Surgery 69 (3): 701–9. https://doi.org/10.1016/j.jvs.2018.07.022.
- Vuoncino, Matthew, Joseph M. White, and William Darrin Clouse. 2022. "21. Upper Extremity and Junctional Zone Injuries." In, 252–72. Elsevier. https://doi.org/10.1016/B978-0-323-69766-8.00021-4.
- Wagner, Jason K., Ellen Dillavou, Uttara Nag, Adham Abou Ali, Sandra Truong, Rabih Chaer, Eric Hager, Theodore Yuo, Michel Makaroun, and Efthymios D. Avgerinos. 2019. "Immediate-Access Grafts Provide Comparable Patency to Standard Grafts, with Fewer Reinterventions and Catheter-Related Complications." Journal of Vascular Surgery 69 (3): 883–89. https://doi.org/10.1016/j.jvs.2018.06.204.
- Wagner, L. K., M. D. McNeese, M. V. Marx, and E. L. Siegel. 1999. "Severe Skin Reactions from Interventional Fluoroscopy: Case Report and Review of the Literature." *Radiology* 213 (3): 773–76. https://doi.org/10.1148/radiology.213.3.r99dc16773.
- Wahlberg, Eric, Jerry Goldstone, and Pär Olofsson. 2006. Emergency Vascular Surgery: A Practical Guide. Medicine (Springer-11650; ZDB-2-SME). Springer-Verlag. Springer e-books.
- Walker, Michael D. 1995. "Endarterectomy for Asymptomatic Carotid Artery Stenosis." JAMA: The Journal of the American Medical Association 273 (18): 1421. https://doi. org/10.1001/jama.1995.03520420037035.
- Wang, John C., Ann H. Kim, and Vikram S. Kashyap. 2016. "Open Surgical or Endovascular Revascularization for Acute Limb Ischemia." Journal of Vascular Surgery 63 (1): 270–78. https://doi.org/10.1016/j.jvs.2015.09.055.
- Wang, Linda J., Satinderjit Locham, Omar Al-Nouri, Matthew J. Eagleton, W. Darrin Clouse, and Mahmoud B. Malas. 2020. "Endovascular Repair of Ruptured Abdominal Aortic Aneurysm Is Superior to Open Repair: Propensity-matched Analysis in the Vascular Quality Initiative." Journal of Vascular Surgery 72 (2): 498–507. https://doi.org/10.1016/j.jvs. 2019.11.063.
- Wang, Wenda, Rui Sun, Yuexin Chen, and Changwei Liu. 2018. "Meta-Analysis and Systematic Review of Percutaneous Mechanical Thrombectomy for Lower Extremity Deep Vein Thrombosis." Journal of Vascular Surgery. Venous and Lymphatic Disorders 6 (6): 788–800. https://doi.org/10.1016/j.jvsv.2018.08.002.

- Wang, Xiaobai, Yan Zhang, Chengzhi Li, and Hong Zhang. 2012. "Results of Endovascular Treatment for Patients with Nutcracker Syndrome." Journal of Vascular Surgery 56 (1): 142–48. https://doi.org/10.1016/j.jvs.2012.01.007.
- Wanhainen, Anders, Jonas Lundkvist, David Bergqvist, and Martin Björck. 2006. "Cost-Effectiveness of Screening Women for Abdominal Aortic Aneurysm." Journal of Vascular Surgery 43 (5): 908–914; discussion 914. https://doi.org/10.1016/j.jvs.2005.12.064.
- Ward, Natalie C., Gerald F. Watts, and Robert H. Eckel. 2019. "Statin Toxicity." Circulation Research 124 (2): 328–50. https://doi.org/10.1161/CIRCRESAHA.118.312782.
- Ward, Thomas J., Stuart Cohen, Rahul S. Patel, Edward Kim, Aaron M. Fischman, Francis S. Nowakowski, Sharif H. Ellozy, Peter L. Faries, Michael L. Marin, and Robert A. Lookstein. 2014. "Anatomic Risk Factors for Type-2 Endoleak Following EVAR: A Retrospective Review of Preoperative CT Angiography in 326 Patients." Cardiovascular and Interventional Radiology 37 (2): 324–28. https://doi.org/10.1007/s00270-013-0646-7.
- Warncke, Emily S., Dorissa L. Gursahaney, Margherita Mascolo, and Elizabeth Dee. 2019. "Superior Mesenteric Artery Syndrome: A Radiographic Review." Abdominal Radiology (New York) 44 (9): 3188–94. https://doi.org/10.1007/s00261-019-02066-4.
- Warren, Anne G., Håkan Brorson, Loren J. Borud, and Sumner A. Slavin. 2007. "Lymphedema: A Comprehensive Review." Annals of Plastic Surgery 59 (4): 464–72. https://doi.org/10.1097/01.sap.0000257149.42922.7e.
- Wattanakit, Keattiyoat, Aaron R. Folsom, Elizabeth Selvin, Beth D. Weatherley, James S. Pankow, Frederick L. Brancati, and Alan T. Hirsch. 2005. "Risk Factors for Peripheral Arterial Disease Incidence in Persons with Diabetes: The Atherosclerosis Risk in Communities (ARIC) Study." Atherosclerosis 180 (2): 389–97. https://doi.org/10.1016/j.atherosclerosis. 2004.11.024.
- Weaver, Fred A., S.Ram Kumar, Albert E. Yellin, Scott Anderson, Douglas B. Hood, Vincent L. Rowe, Rodanthi C. Kitridou, Roy D. Kohl, and Jason Alexander. 2004. "Renal Revascularization in Takayasu Arteritis-induced Renal Artery Stenosis." *Journal of Vascular Surgery* 39 (4): 749–57. https://doi.org/10.1016/j.jvs.2003.12.022.
- Weaver, Marvin V., Rami O. Tadros, Varinder S. Phangureh, Peter L. Faries, Robert A. Lookstein, and Michael L. Marin. 2014. "A Novel Approach to the Management of a Recurrent Lymphocele Following a Femoral-Femoral Bypass." *Journal of Vascular Surgery* 59 (4): 1109–11. https://doi.org/10.1016/j.jvs.2013.04.049.
- Weinstein, Stefanie, Marc C. Mabray, Riz Aslam, Tom Hope, Judy Yee, and Christopher Owens. 2015. "Intraoperative Sonography During Carotid Endarterectomy: Normal Appearance and Spectrum of Complications." Journal of Ultrasound in Medicine: Official Journal of the American Institute of Ultrasound in Medicine 34 (5): 885–94. https: //doi.org/10.7863/ultra.34.5.885.
- Welleweerd, J. C., H. M. den Ruijter, B. G. L. Nelissen, M. L. Bots, L. J. Kappelle, G. J. E. Rinkel, F. L. Moll, and G. J. de Borst. 2015. "Management of Extracranial Carotid Artery Aneurysm." *European Journal of Vascular and Endovascular Surgery* 50 (2): 141–47. https://doi.org/10.1016/j.ejvs.2015.05.002.
- Welsch, Thilo, Markus W. Büchler, and Peter Kienle. 2007. "Recalling Superior Mesenteric Artery Syndrome." Digestive Surgery 24 (3): 149–56. https://doi.org/10.1159/000102097.

- Werth, Sebastian, Rupert Bauersachs, Horst Gerlach, Eberhard Rabe, Sebastian Schellong, and Jan Beyer-Westendorf. 2016. "Superficial Vein Thrombosis Treated for 45 Days with Rivaroxaban Versus Fondaparinux: Rationale and Design of the SURPRISE Trial." Journal of Thrombosis and Thrombolysis 42 (2): 197–204. https://doi.org/10.1007/s11239-016-1354-3.
- Weyand, Cornelia M., and Jörg J. Goronzy. 2003. "Medium- and Large-Vessel Vasculitis." The New England Journal of Medicine 349 (2): 160–69. https://doi.org/10.1056/ NEJMra022694.
- ———. 2014. "Giant-Cell Arteritis and Polymyalgia Rheumatica." Edited by Caren G. Solomon. New England Journal of Medicine 371 (1): 50–57. https://doi.org/10.1056/ NEJMcp1214825.
- Wickström, J.-E., M. Laivuori, E. Aro, R. T. Sund, O. Hautero, M. Venermo, J. Jalkanen, and H. Hakovirta. 2017. "Toe Pressure and Toe Brachial Index Are Predictive of Cardiovascular Mortality, Overall Mortality, and Amputation Free Survival in Patients with Peripheral Artery Disease." *European Journal of Vascular and Endovascular Surgery* 53 (5): 696–703. https://doi.org/10.1016/j.ejvs.2017.02.012.
- Wigley, Fredrick M., and Nicholas A. Flavahan. 2016. "Raynaud's Phenomenon." The New England Journal of Medicine 375 (6): 556–65. https://doi.org/10.1056/NEJMra1507638.
- Wilmink, A. B. M., M. Forshaw, C. R. G. Quick, C. S. Hubbard, and N. E. Day. 2002. "Accuracy of Serial Screening for Abdominal Aortic Aneurysms by Ultrasound." *Journal of Medical Screening* 9 (3): 125–27. https://doi.org/10.1136/jms.9.3.125.
- Wilson, Jonathan J., Catherine A. Zahn, and Harold Newman. 1990. "Fibrinolytic Therapy for Idiopathic Subclavian-Axillary Vein Thrombosis." The American Journal of Surgery 159 (2): 208–10. https://doi.org/10.1016/S0002-9610(05)80262-4.
- Wilson, P. W., J. M. Hoeg, R. B. D'Agostino, H. Silbershatz, A. M. Belanger, H. Poehlmann, D. O'Leary, and P. A. Wolf. 1997. "Cumulative Effects of High Cholesterol Levels, High Blood Pressure, and Cigarette Smoking on Carotid Stenosis." *The New England Journal* of Medicine 337 (8): 516–22. https://doi.org/10.1056/NEJM199708213370802.
- Winn, Justin, Charan Singh, Adam Cloud, and Harlan Stock. 2015. "Radiological Case: Cystic Adventitial Disease." Applied Radiology, June, 22–23. https://doi.org/10.37549/ AR2193.
- Wiske, Clay, Isibor Arhuidese, Mahmoud Malas, and Robert Patterson. 2018. "Comparing the Efficacy of Shunting Approaches and Cerebral Monitoring During Carotid Endarterectomy Using a National Database." Journal of Vascular Surgery 68 (2): 416–25. https://doi.org/ 10.1016/j.jvs.2017.11.077.
- Witrak, B. J., and P. Sprawls. 1984. "Maternity Lead Apron." *Radiology* 150 (2): 597. https://doi.org/10.1148/radiology.150.2.6691124.
- Wittens, C., A. H. Davies, N. Bækgaard, R. Broholm, A. Cavezzi, S. Chastanet, M. de Wolf, et al. 2015. "Editor's Choice - Management of Chronic Venous Disease: Clinical Practice Guidelines of the European Society for Vascular Surgery (ESVS)." European Journal of Vascular and Endovascular Surgery: The Official Journal of the European Society for Vascular Surgery 49 (6): 678–737. https://doi.org/10.1016/j.ejvs.2015.02.007.

Witztum, J. L. 1994. "The Oxidation Hypothesis of Atherosclerosis." Lancet (London, Eng-

land) 344 (8925): 793–95. https://doi.org/10.1016/s0140-6736(94)92346-9.

- Wu, S. M., D. P. Morris, and D. W. Stafford. 1991. "Identification and Purification to Near Homogeneity of the Vitamin k-Dependent Carboxylase." *Proceedings of the National Academy of Sciences of the United States of America* 88 (6): 2236–40. https://doi.org/10. 1073/pnas.88.6.2236.
- Wyers, Mark C and Martin, Michelle C. 2019. "133: Acute Mesenteric Arterial Disease." In Rutherford's Vascular Surgery and Endovascular Therapy, Ninth. Elsevier Inc. https: //doi.org/10.1016/B978-0-323-42791-3.00133-X.
- Wyers, Mark C, and Michelle C Martin. 2019. "133: Acute Mesenteric Arterial Disease." In Rutherford's Vascular Surgery and Endovascular Therapy, 9th ed. Elsevier Inc. https: //doi.org/10.1016/B978-0-323-42791-3.00133-X.
- Xenos, Eleftherios S., George A. Davis, Qiang He, Amanda Green, and Susan S. Smyth. 2019. "The Implementation of a Pulmonary Embolism Response Team in the Management of Intermediate- or High-Risk Pulmonary Embolism." Journal of Vascular Surgery. Venous and Lymphatic Disorders 7 (4): 493–500. https://doi.org/10.1016/j.jvsv.2018.11.014.
- Xenos, Eleftherios S, Michael Freeman, Scott Stevens, David Cassada, John Pacanowski, and Mitchell Goldman. 2003. "Covered Stents for Injuries of Subclavian and Axillary Arteries." Journal of Vascular Surgery 38 (3): 451–54. https://doi.org/10.1016/S0741-5214(03)00553-6.
- Yanase, Yohsuke, Nobuyoshi Kawaharada, Toshiyuki Maeda, Tetsuya Koyanagi, Toshiro Ito, Yoshihiko Kurimoto, and Tetsuya Higami. 2012. "Treatment of Delayed Neurological Deficits After Surgical Repair of Thoracic Aortic Aneurysm." Annals of Thoracic and Cardiovascular Surgery: Official Journal of the Association of Thoracic and Cardiovascular Surgeons of Asia 18 (3): 271–74. https://doi.org/10.5761/atcs.cr.11.01741.
- Yao, S T, J T Hobbs, and W T Irivne. 2005. "Ankle Systolic Pressure Measurements in Arterial Disease Affecting the Lower Extremities." *British Journal of Surgery* 56 (9): 676– 79. https://doi.org/10.1002/bjs.1800560910.
- Yevzlin, Alexander S., Micah R. Chan, and Arif Asif. 2016. "Hand Ischemia in a Patient with an Arteriovenous Fistula." American Journal of Kidney Diseases: The Official Journal of the National Kidney Foundation 67 (3): 512–15. https://doi.org/10.1053/j.ajkd.2015.07. 041.
- Yi, Jeniann A., Charles J. Fox, and Ernest E. Moore. 2022. "11. Resuscitative Endovascular Balloon Occlusion of the Aorta." In *Rich's Vascular Trauma*, 126–36. Elsevier. https: //doi.org/10.1016/B978-0-323-69766-8.00011-1.
- Yin, Yanqi, Yaxue Shi, Tianlei Cui, Hua Li, Jianghua Chen, Lihong Zhang, Zhengya Yu, et al. 2021. "Efficacy and Safety of Paclitaxel-Coated Balloon Angioplasty for Dysfunctional Arteriovenous Fistulas: A Multicenter Randomized Controlled Trial." American Journal of Kidney Diseases: The Official Journal of the National Kidney Foundation 78 (1): 19–27.e1. https://doi.org/10.1053/j.ajkd.2020.11.022.
- Yoon, Dustin Y., Suman Annambhotla, Scott A. Resnick, Mark K. Eskandari, and Heron E. Rodriguez. 2015. "Inadvertent Arterial Placement of Central Venous Catheters: Diagnostic and Therapeutic Strategies." Annals of Vascular Surgery 29 (8): 1567–74. https: //doi.org/10.1016/j.avsg.2015.05.030.

- Yun, Seong Jong, Jung Min Lee, Deok Ho Nam, Jung Kyu Ryu, and Sun Hwa Lee. 2016. "Discriminating Renal Nutcracker Syndrome from Asymptomatic Nutcracker Phenomenon Using Multidetector Computed Tomography." Abdominal Radiology (New York) 41 (8): 1580–88. https://doi.org/10.1007/s00261-016-0717-8.
- Yusof, Nur Nazifah Mohd, Andrew McCann, Peter J. Little, and Hang T. Ta. 2019. "Non-Invasive Imaging Techniques for the Differentiation of Acute and Chronic Thrombosis." *Thrombosis Research* 177 (May): 161–71. https://doi.org/10.1016/j.thromres.2019.03.009.
- Zanow, Jurgen, Ulf Kruger, and Hans Scholz. 2006. "Proximalization of the Arterial Inflow: A New Technique to Treat Access-Related Ischemia." Journal of Vascular Surgery 43 (6): 1216–21. https://doi.org/10.1016/j.jvs.2006.01.025.
- Zhu, Chenmou, Jichun Zhao, Ding Yuan, Bin Huang, Yi Yang, Yukui Ma, and Fei Xiong. 2019. "Endovascular and Surgical Management of Intact Splenic Artery Aneurysm." Annals of Vascular Surgery 57 (May): 75–82. https://doi.org/10.1016/j.avsg.2018.08.088.
- Zhu, F P, S Luo, Z J Wang, Z Y Jin, L J Zhang, and G M Lu. 2012. "Takayasu Arteritis: Imaging Spectrum at Multidetector CT Angiography." *The British Journal of Radiology* 85 (1020): e1282–92. https://doi.org/10.1259/bjr/25536451.
- Zierler, R. Eugene, and David L. Dawson, eds. 2016. Strandness's Duplex Scanning in Vascular Disorders. Fifth edition. Philadelphia: Wolters Kluwer.
- Zierler, R. Eugene, William D. Jordan, Brajesh K. Lal, Firas Mussa, Steven Leers, Joseph Fulton, William Pevec, Andrew Hill, and M. Hassan Murad. 2018. "The Society for Vascular Surgery Practice Guidelines on Follow-up After Vascular Surgery Arterial Procedures." *Journal of Vascular Surgery* 68 (1): 256–84. https://doi.org/10.1016/j.jvs.2018.04.018.
- Ziomek, S., W. J. Quiñones-Baldrich, R. W. Busuttil, J. D. Baker, H. I. Machleder, and W. S. Moore. 1986. "The Superiority of Synthetic Arterial Grafts over Autologous Veins in Carotid-Subclavian Bypass." *Journal of Vascular Surgery* 3 (1): 140–45.
- Zito, Concetta, Giuseppe Dattilo, Giuseppe Oreto, Gianluca Di Bella, Annalisa Lamari, Raffaella Iudicello, Olimpia Trio, et al. 2009. "Patent Foramen Ovale: Comparison Among Diagnostic Strategies in Cryptogenic Stroke and Migraine." *Echocardiography (Mount Kisco,* N.Y.) 26 (5): 495–503. https://doi.org/10.1111/j.1540-8175.2008.00852.x.
- Zwolak, Robert M., Mark F. Fillinger, Daniel B.. Walsh, F. Alizabeth LaBombard, Anne Musson, Chad E. Darling, and Jack L. Cronenwett. 1998. "Mesenteric and Celiac Duplex Scanning: A Validation Study." *Journal of Vascular Surgery* 27 (6): 1078–88. https: //doi.org/10.1016/S0741-5214(98)60010-0.